Mathematical analysis of evolution equations in curved thin domains or on moving surfaces

Tatsu-Hiko Miura

Graduate School of Mathematical Sciences, The University of Tokyo, 3-8-1 Komaba Meguro-ku Tokyo 153-8914, Japan

Contents

1	Intr	oduction	1						
	1.1	Purpose of the thesis	1						
	1.2	Introduction to Chapter 2	2						
	1.3	Introduction to Chapter 3	2						
	1.4	Introduction to Chapter 4	3						
	1.5	Introduction to Chapter 5	3						
	1.6	Introduction to Chapter 6	4						
	Ackı	nowledgments	5						
	Refe	rences	5						
2	Zere	o width limit of the heat equation on moving thin domains	7						
	2.1	Introduction	7						
	2.2	Evolving surfaces and moving thin domains	10						
	2.3	Heat equation on moving thin domains	12						
	2.4	Basic function spaces on evolving surfaces	17						
	2.5	Average operator	19						
		2.5.1 Definition and basic properties of the average operator	19						
		2.5.2 Tangential gradient of the average operator	22						
		2.5.3 Material derivative of the average operator	24						
	2.6	Convergence and characterization of the limit	28						
		2.6.1 Variational formulations of the average of solutions to the heat equation	28						
		2.6.2 Estimates for the average $M_{\varepsilon}u^{\varepsilon}$ in the space W_T	29						
		2.6.3 Limit equation on evolving surfaces and weak convergence of $M_{\varepsilon}u^{\varepsilon}$.	31						
		2.6.4 Estimates for the difference between solutions to the heat equation and							
		the limit equation	35						
	2.A	Heuristic derivation of the limit equation	37						
	$2.\mathrm{B}$	Elementary facts on integrals over evolving surfaces	40						
	$2.\mathrm{C}$	Calculations involving the differential geometry of tubular neighborhoods	45						
	Refe	rences	50						
3	Energetic variational approach for nonlinear diffusion in moving thin do-								
	mai	ns	52						
	3.1	Introduction	52						
	3.2	Quantities on a moving surface	55						
	3.3	Transport equation in a moving domain and on a moving surface	57						
	3.4	Zero width limit for nonlinear diffusion equations	58						
	3.5	Energy law	61						
	3.6	Energetic variation for derivation of Darcy's law	68						

	Refe	rences .		73					
4	Singular limit equations for incompressible fluids in moving thin domains 75								
	4.1	Introd	$uction \ldots \ldots$	75					
	4.2	Prelim	inaries \ldots	78					
		4.2.1	Moving surfaces and moving thin domains	78					
		4.2.2	Notations and formulas for quantities on fixed surfaces	79					
		4.2.3	Material derivatives and differentiation of composite functions with						
			the closest point mapping $\ldots \ldots \ldots$	83					
	4.3	Limit	equations of the Euler equations	84					
	4.4	.4 Limit equations of the Navier–Stokes equations							
	4.5	Energy	y identities \ldots	94					
		4.5.1	Euler equations	94					
		4.5.2	Navier–Stokes equations	98					
	4.A	Elemen	ntary calculations of various quantities on surfaces	101					
	4.B	Compa	arison of vector Laplacians	107					
	Refe	rences .		108					
5	Nav	ier–St	okes equations in a curved thin domain	111					
	5.1	Introd	\mathbf{uction}	111					
	5.2	Prelim	inaries	122					
		5.2.1	Closed surface	123					
		5.2.2	Curved thin domain	130					
	5.3	Funda	mental tools for analysis	133					
		5.3.1	Sobolev inequalities	133					
		5.3.2	Consequences of the boundary conditions	137					
		5.3.3	Impermeable extension of surface vector fields	141					
	5.4	Korn i	nequalities on a thin domain and a surface	144					
		5.4.1	Uniform Korn inequalities on a thin domain	144					
		5.4.2	Korn inequalities on a surface	152					
	5.5	Stokes	operator with slip boundary conditions	154					
		5.5.1	Helmholtz–Leray projection on a thin domain	154					
		5.5.2	Definition and basic properties of the Stokes operator	158					
		5.5.3	Uniform regularity estimates for the Stokes operator	160					
		5.5.4	Uniform a priori estimate for the vector Laplacian	164					
	5.6	Averag	ge operators in the thin direction	171					
		5.6.1	Definition and basic inequalities of the average operators	171					
		5.6.2	Tangential derivatives of the average operators	175					
		5.6.3	Decomposition of vector fields into the average and residual parts	182					
		5.6.4	Average of bilinear and trilinear forms	187					
	5.7	Estima	ate for the trilinear term	193					
	5.8	Global	existence and uniform estimates of a strong solution	198					
	5.9	Weight	ted solenoidal spaces on a surface	206					
	0.0	5.9.1	Nečas inequality on a surface	206					
		5.9.2	Characterization of the annihilator of a weighted solenoidal space	212					
		5.9.3	Weighted Helmholtz–Leray decomposition of tangential vector fields	214					
		5.9.4	Solenoidal spaces of general vector fields	219					
	5.10	Singul	ar limit problem on degeneration of a thin domain	223					
		5.10.1	Weak formulation for the bulk system	223					

		5.10.2 Average of the weak formulation	223
		5.10.3 Energy estimate for the average of a solution	231
		5.10.4 Estimate for the time derivative of the average	235
		5.10.5 Weak convergence of the average and characterization of the limit	238
		5.10.6 Strong convergence of the average and error estimates	247
	5.A	Notations and basic formulas on vectors and matrices	251
	$5.\mathrm{B}$	Calculations involving differential geometry of surfaces	255
	$5.\mathrm{C}$	Riemannian connection on a surface	269
	Refe	rences	274
6	Har	nilton–Jacobi equations on an evolving surface	278
	6.1	Introduction	278
		6.1.1 A motivating example	278
		6.1.2 Background	279
		6.1.3 Outline	279
	6.2	Preliminaries	280
		6.2.1 Tangential derivatives of functions on fixed surfaces	280
		6.2.2 Time dependent surfaces	280
		6.2.3 Triangulated surface	281
	6.3	Viscosity solutions: Uniqueness	282
	6.4	Finite volume scheme	284
	6.5	Convergence to viscosity solutions	292
	6.6	Error bound	297
	Refe	rences	300

Chapter 1

Introduction

1.1 Purpose of the thesis

The purpose of this thesis is to study five problems related to evolution equations in a curved thin domain or on a moving surface. They are completely independent and studied separately in each chapter. References are listed at the end of each chapter and cited only in that chapter. Also, notations are different from chapter to chapter.

In Chapters 2–5 we consider parabolic equations in curved thin domains degenerating into stationary or moving hypersurfaces. Partial differential equations (PDEs) in thin domains appear in many problems of natural sciences such as solid mechanics (thin elastic bodies) and fluid mechanics (lubrication, meteorology, ocean dynamics). In the mathematical study of PDEs in thin domains we are mainly interested in two problems. One problem is to investigate the relation between the existence, uniqueness, and long time behavior of solutions to PDEs and the smallness of the width of thin domains. For example, in the study of the Navier–Stokes equations in a three-dimensional thin domain we expect to show the global-intime existence of a strong solution for large data, since a three-dimensional thin domain with very small width can be considered almost two-dimensional. Another problem is a singular limit problem for a PDE in a thin domain as it degenerates into a lower dimensional set. We are concerned with derivation of a limit equation on the limit set and comparison of the original and limit equations. Such problems were first studied by Hale and Raugel [6,7], who considered damped wave and reaction-diffusion equations in a flat thin domain degenerating into a lower dimensional domain. Since then, many researchers have studied PDEs, mainly a reaction-diffusion equation and the Navier-Stokes equations, in flat thin domains. In the case of curved thin domains whose limit sets are a lower dimensional manifold, there are several works on the asymptotic behavior of eigenvalues of the Laplacian on a curved thin domain around a hypersurface (see [9] and the references cited therein). However, evolution equations in curved thin domains have not been studied well, except for a few works on a reaction-diffusion equation in a curved thin domain [17] and the Navier–Stokes equations in a thin spherical shell [20]. It is also important to consider PDEs in moving thin domains for applications in physical problems and a few researchers have studied them. We refer to [5] for the study of a diffuse interface model in a moving thin domain for an advection-diffusion equation on a moving surface and to [16] for the study of a singular limit problem for a reaction-diffusion equation in a moving thin domain degenerating into a lower dimensional stationary domain. However, there is no literature that derives an unknown limit equation of a PDE in a moving thin domain whose limit set also moves in time, even in the case of a linear diffusion equation. The difficulties in the analysis of PDEs in curved or moving thin domains arise from the geometry and motion of the boundary of the thin domains and their limit sets. Our purpose in Chapters 2–5 is to provide mathematical methods for dealing with such difficulties and to investigate the effects of the geometry and motion of the curved thin domains and their limit surfaces on the original and limit equations.

In Chapter 6 we study the first order Hamilton–Jacobi equation on a moving surface. PDEs on moving surfaces arise in many applications in biology, fluid mechanics, and material sciences as frequently as those in thin domains. For example, an advection-diffusion equation on a moving surface describes transport of surfactants on the interface between two fluids. They provide mathematically interesting problems such as the well-posedness, numerical analysis, and interaction between the evolution of a surface and the behavior of a solution to a surface PDE. Many researchers have studied various kinds of PDEs on moving surfaces in recent years. We refer to [4] and the references cited therein for the mathematical and numerical study of PDEs on moving surfaces and its applications. In this thesis we consider the Hamilton–Jacobi equation as a new kind of PDE on a moving surface. Our goal is to establish the existence and uniqueness of a viscosity solution as well as to provide a numerical scheme with an error bound.

1.2 Introduction to Chapter 2

In Chapter 2 we study a singular limit problem for the heat equation in a moving thin domain. We consider the Neumann type problem of the heat equation in a moving thin domain that degenerates into a closed moving hypersurface as the width of the domain tends to zero. Here the Neumann type boundary condition is imposed to make the total amount of heat in the moving thin domain conserved in time.

Our purpose in Chapter 2 is to investigate the behavior of a solution to the heat equation as the width of the thin domain tends to zero and to derive a limit equation on the moving surface. To this end, we use a change of variables formula (a co-area formula) to transform an integral over the thin domain into integrals over the limit surface and its normal direction, and analyze the weighted average in the thin direction of a variational solution to the heat equation. Then, under suitable assumptions, we prove the weak convergence of the weighted average of a solution in an appropriate function space on the limit surface. Moreover, we show that the weak limit is a unique variational solution to a limit equation on the moving limit surface, which is a linear diffusion equation involving the mean curvature and normal velocity of the surface. We also estimate the difference between a solution to the heat equation and a solution to the limit equation. This is the first result that derives an a priori unknown limit equation of a PDE in a thin domain whose limit set evolves in time.

Chapter 2 is based on the work [13].

1.3 Introduction to Chapter 3

The subject of Chapter 3 is nonlinear diffusion equations of porous media type in a moving thin domain and on a moving surface. After a brief review of the transport equations, we consider the thin width limit of a nonlinear diffusion equation in a moving thin domain that consists of the transport equation, Darcy's law, and the boundary condition corresponding to the situation that there is no exchange of mass between the domain and its surroundings. Under the assumption that the moving thin domain shrinks to a moving closed hypersurface as its width tends to zero, we formally derive a limit nonlinear diffusion equation on the moving surface based on calculations of the Taylor series for bulk quantities with respect to the signed distance from the limit surface. In particular, we see that in the thin width limit the transport equation and Darcy's law in the moving thin domain become those on the moving surface. We also show that the thin width limit of the energy law for the nonlinear diffusion equation in the moving thin domain is that for the limit equation on the moving surface. Then we discuss an energetic variational approach to derivation of nonlinear diffusion equations in a moving domain and on a moving surface, and observe that the energetic variation commutes with the thin width limit.

Chapter 3 is based on the joint work [15] with Yoshikazu Giga and Chun Liu.

1.4 Introduction to Chapter 4

Chapter 4 is devoted to formal derivation of and discussions on singular limit equations of the incompressible Euler and Navier–Stokes equations in a three-dimensional moving thin domain. For a given closed moving surface, we define a moving thin domain as its tubular neighborhood of small radius. We consider the Euler equations with impermeable boundary condition and the Navier–Stokes equations with Navier's perfect slip boundary conditions in the moving thin domain. Then we formally derive their limit equations on the moving limit surface by calculations of the Taylor series for the bulk velocity and pressure with respect to the signed distance from the surface. Our limit equations are basically the same as incompressible fluid equations on a moving surface derived from local conservation laws of mass and linear momentum for a surface fluid [10] and by a global energetic variational approach [12]. We also observe that in the thin width limit the energy identities of the Euler and Navier–Stokes equations in the moving thin domain become those of the corresponding limit equations on the moving surface.

The limit equations on the moving surface involve the first and second order derivatives of tangential and normal vector fields on the surface. To understand the structure of the limit equations, we give several formulas on the derivatives of vector fields on an embedded surface in the three-dimensional Euclidean space. We use them to show that our limit equations are the same as the Euler and Navier–Stokes equations on a manifold introduced by Arnold [1,2] and Taylor [19] when the limit surface is stationary.

Chapter 4 is based on the work [14].

1.5 Introduction to Chapter 5

In Chapter 5 we study the three-dimensional Navier–Stokes equations in a stationary curved thin domain, which is defined as a region between two very close parametrized surfaces of a given two-dimensional closed surface. Under the assumption that the curved thin domain degenerates into the given surface as its width tends to zero, we consider the Navier–Stokes equations with Navier's slip boundary conditions.

The Navier–Stokes equations in thin domains have been studied in the case of a flat thin product domain [18], a flat thin domain whose top and bottom boundaries are given by the graph of functions on a two-dimensional domain [8], and a thin spherical shell given as a region between two concentric spheres of near radii [20]. Our goal is not just to generalize the results in the previous works, but to study the effect of the curvatures of a general limit surface on the bulk and limit equations.

The first result in Chapter 5 is the global-in-time existence of a strong solution to the Naiver–Stokes equations. Under suitable assumptions on the limit surface and friction coefficients appearing in the slip boundary conditions, we establish the global existence and uniform estimates of a strong solution for very large data according to the smallness of the width of the thin domain. Main tools for the proof of the global existence are an average operator in the normal direction of the limit surface and an extension of a surface vector field to the thin domain that satisfies the impermeable boundary condition. Using them, the slip boundary conditions, and Sobolev type inequalities on the thin domain and the surface we derive a good product estimate for the inertial and viscous terms in the Navier–Stokes equations. A key idea is to decompose a vector field on the thin domain into the average part, to which we can use a product estimate for a function on the thin domain and that on the surface, and the residual part, to which a good L^{∞} -estimate is applicable.

The second result is concerning a singular limit problem for the Navier–Stokes equations as the curved thin domain degenerates into the closed surface. We show that, under suitable assumptions on given data, the averaged tangential component of a strong solution to the bulk Navier–Stokes equations converges weakly in an appropriate function space on the limit surface, and that the weak limit is a unique weak solution to limit equations, which are the damped and weighted Navier–Stokes equations on the limit surface with viscous term involving the Gaussian curvature of the surface and the functions that parametrize the boundaries of the thin domain. To prove these results, we approximate a weak formulation for the bulk equations by that for the averaged tangential component of a strong solution and derive its energy estimate. In approximation of the weak formulation, we use the average operator and change of variables formulas for integrals over the thin domain and its boundary. We also employ the impermeable extension of a surface vector field and a uniform estimate for the gradient part of the Helmholtz–Leray decomposition on the thin domain to construct an appropriate test function for the bulk equations from a test function for the limit equations. To derive the energy estimate for the averaged tangential component of a strong solution to the bulk equations, we would like to take it as a test function for its weak formulation. However, it is not allowed since the averaged tangential component is not in the space of test functions for the limit equations, which is a weighted solenoidal space on the surface. To overcome this difficultly, we use the weighted Helmholtz–Leray projection on the surface.

Besides the weak convergence and characterization of the limit, we estimate the difference between a strong solution to the Navier–Stokes equations and a weak solution to the limit equations. It is worth noting that the normal derivative (with respect to the surface) of a strong solution to the Navier–Stokes equations is compared with a surface vector field given by a weak solution to the limit equations and the Weingarten map (or shape operator) of the surface. In particular, it is not necessarily small even though the thin domain and its limit surface are stationary.

1.6 Introduction to Chapter 6

In Chapter 6 we consider the first order Hamilton–Jacobi equation on a moving closed surface in the three-dimensional Euclidean space. One motivation for considering such an equation is to describe the motion of a curve on an evolving surface. The aim of Chapter 6 is to study the well-posedness as well as to provide a numerical scheme and an error bound.

We first extend the definition of viscosity sub- and supersolutions to a moving surface by using the material and tangential derivative operators. Here the material derivative is the time derivative along the total velocity of the surface, which consists of the outward normal velocity and a given tangential velocity. Then we prove a comparison principle by a standard doubling of variables method that yields the uniqueness of a viscosity solution.

To establish the existence of a viscosity solution as well as to give a numerical scheme, we

consider discretization of the Hamilton–Jacobi equation in space and time. We approximate the smooth moving surface by a triangulated surface with moving vertices. Then, following the idea of the work [11] in the case of a flat stationary domain, we introduce a finite volume scheme based on the viscous approximation and discretization of surface integrals, and prove its monotonicity and consistency. We point out that to prove the monotonicity and consistency we require only the regularity of the triangulation. In particular, it is not necessary to assume that the triangulation is acute, which is very important for implementation of the numerical scheme.

Using the monotonicity and consistence of our scheme, we prove the existence of a viscosity solution by the half-relaxed limits method. Moreover, by a doubling of variables method we establish an error bound between a viscosity solution and a numerical solution of the same order as in the case of a flat stationary domain.

Chapter 6 is based on the joint work [3] with Klaus Deckelnick and Charles M. Elliott.

Acknowledgments

The author is grateful to his supervisor Professor Yoshikazu Giga for his valuable comments on this work. He also would like to express his gratitude to Professor Charles M. Elliott, Professor Klaus Deckelnick, Dr. Björn Stinner, and Mr. Lewis Church for their kind hospitality during his stay at the University of Warwick. The author would like to thank Professor Chun Liu, Dr. Yohei Tsutsui, Dr. Martin Bolkart, and Dr. Takuya Suzuki for fruitful discussions.

The work of the author was supported by Grant-in-Aid for JSPS Fellows No. 16J02664 and the Program for Leading Graduate Schools, MEXT, Japan.

References

- V. Arnold, Sur la géométrie différentielle des groupes de Lie de dimension infinie et ses applications à l'hydrodynamique des fluides parfaits, Ann. Inst. Fourier (Grenoble) 16 (1966), no. fasc. 1, 319–361 (French).
- [2] V. I. Arnol'd, Mathematical methods of classical mechanics, Graduate Texts in Mathematics, vol. 60, Springer-Verlag, New York, 1989. Translated from the 1974 Russian original by K. Vogtmann and A. Weinstein; Corrected reprint of the second (1989) edition.
- [3] K. Deckelnick, C. M. Elliott, and T.-H. Miura, Hamilton-Jacobi equations on an evolving surface, submitted.
- [4] G. Dziuk and C. M. Elliott, Finite element methods for surface PDEs, Acta Numer. 22 (2013), 289–396.
- [5] C. M. Elliott and B. Stinner, Analysis of a diffuse interface approach to an advection diffusion equation on a moving surface, Math. Models Methods Appl. Sci. 19 (2009), no. 5, 787–802.
- [6] J. K. Hale and G. Raugel, A damped hyperbolic equation on thin domains, Trans. Amer. Math. Soc. 329 (1992), no. 1, 185–219.
- [7] J. K. Hale and G. Raugel, Reaction-diffusion equation on thin domains, J. Math. Pures Appl. (9) 71 (1992), no. 1, 33–95.
- [8] D. Iftimie, G. Raugel, and G. R. Sell, Navier-Stokes equations in thin 3D domains with Navier boundary conditions, Indiana Univ. Math. J. 56 (2007), no. 3, 1083–1156.
- [9] S. Jimbo and K. Kurata, Asymptotic behavior of eigenvalues of the Laplacian on a thin domain under the mixed boundary condition, Indiana Univ. Math. J. 65 (2016), no. 3, 867–898.
- [10] T. Jankuhn, M. A. Olshanskii, and A. Reusken, *Incompressible fluid problems on embedded surfaces: Modeling and variational formulations*, arXiv:1702.02989 [math-ph].
- [11] K. Kim and Y. Li, Convergence of finite volume schemes for Hamilton-Jacobi equations with Dirichlet boundary conditions, J. Comput. Math. 33 (2015), no. 3, 227–247.

- [12] H. Koba, C. Liu, and Y. Giga, Energetic variational approaches for incompressible fluid systems on an evolving surface, Quart. Appl. Math. 75 (2017), no. 2, 359–389.
- [13] T.-H. Miura, Zero width limit of the heat equation on moving thin domains, Interfaces Free Bound. 19 (2017), no. 1, 31–77.
- [14] T.-H. Miura, On singular limit equations for incompressible fluids in moving thin domains, Quart. Appl. Math. 76 (2018), 215–251.
- [15] T.-H. Miura, Y. Giga, and C. Liu, An energetic variational approach for nonlinear diffusion equations in moving thin domains, Adv. Math. Sci. Appl., to appear.
- [16] J. V. Pereira and R. P. Silva, Reaction-diffusion equations in a noncylindrical thin domain, Bound. Value Probl. (2013), 2013:248, 10.
- [17] M. Prizzi, M. Rinaldi, and K. P. Rybakowski, Curved thin domains and parabolic equations, Studia Math. 151 (2002), no. 2, 109–140.
- [18] G. Raugel and G. R. Sell, Navier-Stokes equations on thin 3D domains. I. Global attractors and global regularity of solutions, J. Amer. Math. Soc. 6 (1993), no. 3, 503–568.
- [19] M. E. Taylor, Analysis on Morrey spaces and applications to Navier-Stokes and other evolution equations, Comm. Partial Differential Equations 17 (1992), no. 9-10, 1407–1456.
- [20] R. Temam and M. Ziane, Navier-Stokes equations in thin spherical domains, Optimization methods in partial differential equations (South Hadley, MA, 1996), Contemp. Math., vol. 209, Amer. Math. Soc., Providence, RI, 1997, pp. 281–314.

Chapter 2

Zero width limit of the heat equation on moving thin domains

2.1 Introduction

For $t \in [0, T]$, T > 0, let $\Omega_{\varepsilon}(t)$ be a moving thin domain in \mathbb{R}^n , $n \ge 2$, with width of order $\varepsilon > 0$ that converges to an evolving closed hypersurface $\Gamma(t)$ as $\varepsilon \to 0$. We consider the Neumann type problem of the heat equation of the form

$$\begin{cases} \partial_t u^{\varepsilon} - \Delta u^{\varepsilon} = 0 & \text{in } Q_{\varepsilon,T}, \\ \partial_{\nu_{\varepsilon}} u^{\varepsilon} + v_{\varepsilon}^N u^{\varepsilon} = 0 & \text{on } \partial_{\ell} Q_{\varepsilon,T}, \\ u^{\varepsilon}(0) = u_0^{\varepsilon} & \text{in } \Omega_{\varepsilon}(0). \end{cases}$$
(*H*_{\varepsilon})

Here $Q_{\varepsilon,T} := \bigcup_{t \in (0,T)} \Omega_{\varepsilon}(t) \times \{t\}$, $\partial_{\ell} Q_{\varepsilon,T} := \bigcup_{t \in (0,T)} \partial \Omega_{\varepsilon}(t) \times \{t\}$, and $\nu_{\varepsilon}, v_{\varepsilon}^{N}$ are the unit outward normal vector field of $\partial \Omega_{\varepsilon}(t)$ and the outer normal velocity of $\partial \Omega_{\varepsilon}(t)$, respectively. The term $v_{\varepsilon}^{N} u^{\varepsilon}$ in the boundary condition is added so that the total amount of heat $\int_{\Omega_{\varepsilon}(t)} u^{\varepsilon} dx$ is conserved, see the beginning of Section 2.3. Also, if u^{ε} denotes the concentration of some chemicals, the boundary condition says that chemicals near the boundary move along it and do not go into and out of the moving thin domain.

We are interested in the behavior of a solution u^{ε} to (H_{ε}) as $\varepsilon \to 0$. Our goal is to characterize its limit as well as its convergence. Let us explain the simplest case when $\Omega_{\varepsilon}(t)$ is the set of all points in \mathbb{R}^n with distance less than ε from $\Gamma(t)$ so that the width of $\Omega_{\varepsilon}(t)$ is 2ε . Let ν be the unit outward normal vector field of $\Gamma(t)$ and $V_{\Gamma} = v_{\Gamma}^{N}\nu + V_{\Gamma}^{T}$ be the total velocity of $\Gamma(t)$, where v_{Γ}^{N} and V_{Γ}^{T} are the outer normal velocity of $\Gamma(t)$ and a given tangential velocity field. Then our main result formally implies that, under suitable assumptions on the initial data u_{0}^{ε} of (H_{ε}) , the limit v is a solution to

$$\partial^{\circ} v - v_{\Gamma}^{N} H v - \Delta_{\Gamma(t)} v = 0 \quad \text{on} \quad S_{T}.$$
 (2.1.1)

Here $S_T := \bigcup_{t \in (0,T)} \Gamma(t) \times \{t\}$ and $\partial^\circ v = \partial_t v + v_{\Gamma}^N \nu \cdot \nabla v$ is the normal time derivative of v. (The notation ∂° is used in [2,5]. We refer to [3] for the normal time derivative.) Also, $H := -\operatorname{div}_{\Gamma(t)} \nu$ and $\Delta_{\Gamma(t)} := \operatorname{div}_{\Gamma(t)} \nabla_{\Gamma(t)}$ are the mean curvature of $\Gamma(t)$ and the Laplace-Beltrami operator on $\Gamma(t)$, where $\operatorname{div}_{\Gamma(t)}$ and $\nabla_{\Gamma(t)}$ are the surface divergence operator and the tangential gradient on $\Gamma(t)$, respectively (see Section 2.2 for their definitions). We will give a heuristic derivation of the limit equation (2.1.1) in Appendix 2.A. The equation (2.1.1) is equivalent to

$$\partial^{\bullet} v + (\operatorname{div}_{\Gamma(t)} V_{\Gamma}) v - \Delta_{\Gamma(t)} v - \operatorname{div}_{\Gamma(t)} (v V_{\Gamma}^{T}) = 0 \quad \text{on} \quad S_{T},$$
(2.1.2)

which we will actually derive in Section 2.6. Here $\partial^{\bullet} v = \partial^{\circ} v + V_{\Gamma}^{T} \cdot \nabla_{\Gamma(t)} v$ denotes the material derivative of v (see Section 2.4 for its precise definition). Note that the equation (2.1.1) is independent of the tangential velocity V_{Γ}^{T} . In other words, the evolution of the limit v is not affected by advection along $\Gamma(t)$. Such a phenomenon does not occur in an advection-diffusion equation widely studied in recent years [2, 4–9, 19, 28]:

$$\partial^{\bullet} v + (\operatorname{div}_{\Gamma(t)} V_{\Gamma}) v - \Delta_{\Gamma(t)} v = 0 \quad \text{on} \quad S_T.$$
(2.1.3)

This equation is derived from a conservation law such that

$$\frac{d}{dt} \int_{\mathcal{M}(t)} v \, d\mathcal{H}^{n-1} = -\int_{\partial \mathcal{M}(t)} q \cdot \mu \, d\mathcal{H}^{n-2}$$

for an arbitrary portion $\mathcal{M}(t)$ of $\Gamma(t)$, where \mathcal{H}^k is the k-dimensional Hausdorff measure for $k \in \mathbb{N}$, μ is the co-normal to the boundary $\partial \mathcal{M}(t)$, and q is the surface flux, see [4, Section 3] and [5, Section 3.1] for details.

Partial differential equations on thin domains are studied over the years [12–16, 20–24, 26, 27], and many researchers deal with a nonmoving thin domain of the form

$$\Omega_{\varepsilon} = \{ (x', x_n) \in \mathbb{R}^{n-1} \times \mathbb{R} \mid x' \in \omega, \, \varepsilon g_0(x') < x_n < \varepsilon g_1(x') \}, \quad \varepsilon > 0,$$
(2.1.4)

where ω is a domain in \mathbb{R}^{n-1} and g_0, g_1 are functions on ω . In their pioneering works [12,13], Hale and Raugel compared the dynamics of reaction-diffusion equations and damped wave equations on Ω_{ε} of the form (2.1.4) (with $g_0 = 0$ and slightly modified g_1) and that of corresponding limit equations on ω by the scaling argument. They transformed the equations on Ω_{ε} into scaled equations on a fixed reference domain $\Omega_0 = \omega \times (0,1)$ by the change of variables, and formally derived the limit equations on ω by letting $\varepsilon \to 0$ in the scaled equations on Ω_0 and omitting divergent terms. Then they compared the dynamics of the scaled equations on Ω_0 and that of the limit equations on ω by analyzing weighted bilinear forms that appear in variational formulations of the scaled equations and the limit equations. Their scaling argument is applicable to more general thin domains such as a thin L-shaped domain [14] and a moving thin domain of the form (2.1.4) where $g_0 = 0$ and g_1 depends on time [20]. Prizzi and Rybakowski [22] generalized the scaling argument in [12, 13] to study reaction-diffusion equations on a (nonmoving) thin domain with holes around a lower dimensional domain. The generalized scaling argument in [22] is also valid for a (nonmoving) thin domain with holes around a lower dimensional manifold [21, 23]. We refer to [24] and references therein for other examples of thin domains.

In contrast to the above papers, the limit hypersurface $\Gamma(t)$ of our thin domain $\Omega_{\varepsilon}(t)$ evolves. Such a situation has been considered only in the paper [8], which deals with a diffuse interface model for the advection-diffusion equation (2.1.3). See also [9] for numerical computations of the advection-diffusion equation (2.1.3) based on the diffuse interface model. In [8], however, the limit equation (2.1.3) on the evolving surface is given and a bulk equation on the moving thin domain involves a weight function that vanishes on the boundary of the domain. Therefore, there is no literature on initial-boundary value problems of partial differential equations on moving thin domains around evolving surfaces whose limit equations are unknown in advance, even in the case of the heat equation.

The difficulty caused by the evolution of the hypersurface $\Gamma(t)$ is in transforming equations on $\Omega_{\varepsilon}(t)$ and $\Gamma(t)$ into equations on fixed (in time and width) domain and hypersurface. In particular, transformations of differential operators on $\Gamma(t)$ into those on a fixed hypersurface is so complicated that we can hardly find a limit equation on the fixed hypersurface and convert it into an equation on $\Gamma(t)$, see [7] for the actual transformations of differential operators.

To avoid this difficulty, we employ another method that does not require transformations of $\Omega_{\varepsilon}(t)$ and $\Gamma(t)$. Let us explain our idea of derivation of a limit equation on $\Gamma(t)$. We start from a variational formulation of (H_{ε}) (see (2.3.2)) that consists of integrals over the noncylindrical domain $Q_{\varepsilon,T}$ of a variational solution u^{ε} to (H_{ε}) and a test function defined on $Q_{\varepsilon,T}$. In this variational formulation, we take a test function independent of the normal direction of $\Gamma(t)$ and apply the co-area formula (see (2.5.1)) and a weighted average operator M_{ε} (see Definition 2.5.1) to get a variational formulation (with some residual term) of the average $M_{\varepsilon}u^{\varepsilon}$ (see (2.6.1)) that consists of integrals over the space-time manifold S_T of $M_{\varepsilon}u^{\varepsilon}$ and a test function defined on S_T . Then we obtain a variational formulation of a limit equation on $\Gamma(t)$ (see (2.6.13)) by omitting the residual term in the variational formulation of $M_{\varepsilon}u^{\varepsilon}$. Moreover, we prove that $M_{\varepsilon}u^{\varepsilon}$ converges weakly in a function space on S_T as $\varepsilon \to 0$ and that the limit is a unique variational solution to the limit equation (see Theorem 2.6.9), and estimate the $L^2(Q_{\varepsilon,T})$ -norm of the difference between variational solutions to (H_{ε}) and the limit equation (see Theorem 2.6.12). These results indicate that our limit equation on $\Gamma(t)$ derived as above is indeed the "limit" of (H_{ε}) .

In our derivation of a limit equation, Lemma 2.5.6 and Lemma 2.5.13 play an important role. In Lemma 2.5.6 we approximate an H^1 -bilinear form on $\Omega_{\varepsilon}(t)$ for each $t \in [0,T]$ by that on $\Gamma(t)$ with the tangential gradient of the average $M_{\varepsilon}u$ of a function u on $\Omega_{\varepsilon}(t)$. The proof of Lemma 2.5.6 is based on simple representations of the gradient in \mathbb{R}^n and the tangential gradient on $\Gamma(t)$ under a special local coordinate system for each fixed point on $\Gamma(t)$. On the other hand, Lemma 2.5.13 gives an integral formula that formally represents a relation between the weak time derivative of a function u on $Q_{\varepsilon,T}$ and the weak material derivative of its average $M_{\varepsilon}u$ (in fact, we do not explicitly deal with the time derivative of u). Lemma 2.5.13 essentially follows from Lemma 2.5.11, which gives a relation between the time derivative of functions defined on S_T .

Average operators in the thin direction were originally introduced by Hale and Raugel [12,13], but they took the average of functions on the scaled domain $\Omega_0 = \omega \times (0, 1)$. Average operators on actual thin domains Ω_{ε} appears in the study of the Navier–Stokes equations on three-dimensional thin domains [15, 16, 26, 27]. Temam and Ziane [26, 27] first employed them to study the global existence of strong solutions to the Navier–Stokes equations for large initial data and external forces and the behavior of solutions as $\varepsilon \to 0$ when Ω_{ε} is a three-dimensional thin product domain $\Omega_{\varepsilon} = \omega \times (0, \varepsilon)$ with a bounded domain ω in \mathbb{R}^2 and a thin spherical domain $\Omega_{\varepsilon} = \{x \in \mathbb{R}^3 \mid a < |x| < (1+\varepsilon)a\}$ with a constant a > 0. In [15,16], average operators were employed to study the dynamics of the Navier–Stokes equations on Ω_{ε} of the form (2.1.4). In particular, the authors of [16] compared the dynamics of the Navier–Stokes equations with that of limit equations by estimating the difference of the average of solutions to the Navier–Stokes equations.

We point out that our weighted average operator given in Definition 2.5.1 is a generalization of average operators given in [15, 16, 26] and that its weight function is different from that of an average operator given in [27]. In fact, the weight function of our average operator is a Jacobian that appears when we change variables of integrals over a tubular neighborhood of $\Gamma(t)$ in terms of the normal coordinate system around $\Gamma(t)$. Our choice of the weight function enables us to avoid including the material derivative of a test function in the estimate for the residual term in the variational formulation of the average of a variational solution to (H_{ε}) , which is essential for derivation of its energy estimate, see Lemma 2.5.13 and Remark 2.5.14. We also note that, contrary to our case, Kublik, Tanushev, and Tsai [17] employed the same Jacobian and co-area formula to transform integrals over boundaries of domains into those over their tubular neighborhoods. Based on this transformation, they proposed a new approach to numerical computations of boundary integrals without explicit parametrizations of boundaries and a simple formulation for constructing boundary integral methods to solve Poisson's equation. Their method of the numerical computations of boundary integrals is also applicable to integrals over nonclosed manifolds of higher codimension, such as curves in \mathbb{R}^3 with different endpoints, see [18] for details.

Finally we mention variational formulations of partial differential equations on evolving surfaces. There are several kinds of variational frameworks for equations on evolving surfaces, mainly the advection-diffusion equation (2.1.3), see [4, 19, 28] for example. In addition, Alphonse, Elliott, and Stinner [1, 2] proposed an abstract variational setting with evolving Hilbert spaces and applied it to some equations on moving domains and evolving surfaces. Among these variational frameworks, we adopt the one introduced by Olshanskii, Reusken, and Xu [19]. Their variational formulation is imposed on function spaces on S_T , which is suitable for our calculation of bilinear forms on function spaces on S_T and $Q_{\varepsilon,T}$ performed in Section 2.5 and Section 2.6.

This chapter is organized as follows. In Section 2.2 we introduce notations related to the evolving surface $\Gamma(t)$ and define the moving thin domain $\Omega_{\varepsilon}(t)$. In Section 2.3 we define a variational solution to (H_{ε}) and prove its existence and uniqueness. We also derive an energy estimate of a variational solution to (H_{ε}) with a constant independent of ε . In Section 2.4 we define function spaces on S_T introduced in [19] and give their properties. In Section 2.5 we define the weighted average operator M_{ε} and establish estimates and formulas related to M_{ε} . In Section 2.6 we derive a limit equation on $\Gamma(t)$ of the form (2.1.2) via its variational formulation and prove our main theorems (Theorem 2.6.9 and Theorem 2.6.12). In Appendix 2.A we give a heuristic derivation of the limit equation (2.1.1) when $\Omega_{\varepsilon}(t)$ is the set of all points in \mathbb{R}^n with distance less than ε from $\Gamma(t)$. In Appendix 2.B we give complete proofs of some results in Section 2.4 related to integrals over $\Gamma(t)$. In Appendix 2.C we show detailed calculations in proofs of some lemmas in Section 2.5 involving the differential geometry of tubular neighborhoods of $\Gamma(t)$.

2.2 Evolving surfaces and moving thin domains

For each $t \in [0, T]$, let $\Gamma(t)$ be a closed (that is, compact and without boundary), connected and oriented smooth hypersurface in \mathbb{R}^n . We set $\Gamma_0 := \Gamma(0)$ and define a space-time manifold $S_T \subset \mathbb{R}^{n+1}$ as $S_T := \bigcup_{t \in (0,T)} \Gamma(t) \times \{t\}$. We assume that each point y on $\Gamma(t)$ evolves with velocity $V_{\Gamma}(y, t)$, which is not necessarily normal to $\Gamma(t)$, and the velocity field $V_{\Gamma} : \overline{S_T} \to \mathbb{R}^n$ is smooth. Let $\Phi(\cdot, t) : \Gamma_0 \to \Gamma(t)$ be the flow map of V_{Γ} , that is, $\Phi(\cdot, t)$ is a diffeomorphism from Γ_0 onto $\Gamma(t)$ with its inverse $\Phi^{-1}(\cdot, t)$ for each $t \in [0, T]$ and satisfies

$$\Phi(Y,0) = Y, \quad \frac{\partial \Phi}{\partial t}(Y,t) = V_{\Gamma}(\Phi(Y,t),t) \quad \text{for all} \quad Y \in \Gamma_0, \, t \in [0,T].$$

We assume that Φ and Φ^{-1} are smooth on $\Gamma_0 \times [0, T]$ and $\overline{S_T}$, respectively. Due to this assumption, $\overline{S_T}$ is a compact smooth manifold in \mathbb{R}^{n+1} .

Let $\nu : \overline{S_T} \to \mathbb{R}^n$ be the unit outward normal vector field of $\Gamma(t)$. The velocity V_{Γ} is decomposed into $V_{\Gamma} = v_{\Gamma}^N \nu + V_{\Gamma}^T$, where $v_{\Gamma}^N : \overline{S_T} \to \mathbb{R}$ is the outer normal velocity and $V_{\Gamma}^T : \overline{S_T} \to \mathbb{R}^n$ is a tangential velocity field. Note that to describe the geometric motion of $\Gamma(t)$ it is sufficient to prescribe the normal velocity. However, to describe a limit equation on $\Gamma(t)$ we will derive in Section 2.6, we also need to consider a tangential velocity, which represents advection along $\Gamma(t)$.

For each $t \in [0, T]$, let $d(\cdot, t)$ be the signed distance function from $\Gamma(t)$ that increases in the direction of $\nu(\cdot, t)$. By the smoothness (in space and time) and compactness of $\Gamma(t)$, there is an open set N(t) in \mathbb{R}^n of the form $N(t) = \{x \in \mathbb{R}^n \mid -\delta < d(x,t) < \delta\}$ for each $t \in [0, T]$, where $\delta > 0$ is a constant independent of t, that satisfies the following conditions:

• The signed distance function d is smooth on $\overline{N_T}$, where

$$N_T := \bigcup_{t \in (0,T)} N(t) \times \{t\} \subset \mathbb{R}^{n+1}$$

• For each $(x,t) \in \overline{N_T}$, there is a unique point $p(x,t) \in \Gamma(t)$ such that

$$x = p(x,t) + d(x,t)\nu(p(x,t),t), \quad \nabla d(x,t) = \nu(p(x,t),t).$$

The set N(t) is called a tubular neighborhood of $\Gamma(t)$. Based on the above equality, we extend the outward normal ν to $\overline{N_T}$ by setting $\nu(x,t) := \nabla d(x,t)$ for $(x,t) \in \overline{N_T}$. Then, by the smoothness of d, the extended outward normal ν and the projection mapping p are smooth on $\overline{N_T}$. Also, the normal velocity v_{Γ}^N of $\Gamma(t)$ is given by $v_{\Gamma}^N = -\partial_t d$ on $\overline{S_T}$.

Next, we give definitions of differential operators on evolving surfaces. For a function v and a vector field F on S_T , we define the tangential gradient of v and the surface divergence of F as

$$\nabla_{\Gamma(t)} v(y,t) := [I_n - \nu(y,t) \otimes \nu(y,t)] \nabla \overline{v}(y,t),$$

$$\operatorname{div}_{\Gamma(t)} F(y,t) := \operatorname{trace}[\{I_n - \nu(y,t) \otimes \nu(y,t)\} \nabla \overline{F}(y,t)]$$

for $(y,t) \in S_T$. Here I_n is the identity matrix of size n and $\nu \otimes \nu := (\nu_i \nu_j)_{i,j}$ is the tensor product of ν . Also, $\overline{\nu}$ and \overline{F} are the constant extensions of v and F in the normal direction of $\Gamma(t)$ given by

$$\overline{v}(x,t) := v(p(x,t),t), \quad \overline{F}(x,t) := F(p(x,t),t), \quad (x,t) \in N_T.$$

By definition, $\nu \cdot \nabla_{\Gamma(t)} v = 0$ holds. Hereafter we use the same notations for functions and vector fields on $\Gamma(t)$ with each fixed $t \in [0, T]$.

Finally, we define a moving thin domain. Let g_0 and g_1 be smooth functions on $\overline{S_T}$. We assume that there is a constant c > 0 such that

$$g(y,t) := g_1(y,t) - g_0(y,t) \ge c \text{ for all } (y,t) \in \overline{S_T}.$$
 (2.2.1)

Then we define a moving thin domain $\Omega_{\varepsilon}(t) \subset \mathbb{R}^n$ as

$$\Omega_{\varepsilon}(t) := \{ y + \rho\nu(y,t) \mid y \in \Gamma(t), \, \varepsilon g_0(y,t) < \rho < \varepsilon g_1(y,t) \}, \quad t \in [0,T], \, \varepsilon > 0$$

and a space-time noncylindrical domain $Q_{\varepsilon,T} \subset \mathbb{R}^{n+1}$ as $Q_{\varepsilon,T} := \bigcup_{t \in (0,T)} \Omega_{\varepsilon}(t) \times \{t\}$. Note that $\Omega_{\varepsilon}(t)$ does not necessarily include $\Gamma(t)$, since we do not assume that g_0 is negative and g_1 is positive. Since g_0 and g_1 are smooth and thus bounded on the compact manifold $\overline{S_T}$, there is a positive number ε_0 such that $\overline{\Omega_{\varepsilon}(t)} \subset N(t)$ for all $\varepsilon \in (0, \varepsilon_0)$ and $t \in [0, T]$. Hereafter we assume that $\varepsilon \in (0, \varepsilon_0)$.

2.3 Heat equation on moving thin domains

In this section, we consider the initial-boundary problem (H_{ε}) of the heat equation on the moving thin domain $\Omega_{\varepsilon}(t)$. First we show that the boundary condition of (H_{ε}) yields the conservation of the total amount of heat. Suppose that u^{ε} satisfies the heat equation in $Q_{\varepsilon,T}$. By the Reynolds transport theorem and Green's formula (see [10, Appendix C]) we have

$$\frac{d}{dt} \int_{\Omega_{\varepsilon}(t)} u^{\varepsilon} dx = \int_{\Omega_{\varepsilon}(t)} \partial_{t} u^{\varepsilon} dx + \int_{\partial\Omega_{\varepsilon}(t)} v^{N}_{\varepsilon} u^{\varepsilon} d\mathcal{H}^{n-1}$$
$$= \int_{\Omega_{\varepsilon}(t)} \Delta u^{\varepsilon} dx + \int_{\partial\Omega_{\varepsilon}(t)} v^{N}_{\varepsilon} u^{\varepsilon} d\mathcal{H}^{n-1}$$
$$= \int_{\partial\Omega_{\varepsilon}(t)} (\partial_{\nu_{\varepsilon}} u^{\varepsilon} + v^{N}_{\varepsilon} u^{\varepsilon}) d\mathcal{H}^{n-1}.$$

Hence if u^{ε} additionally satisfies the boundary condition of (H_{ε}) , then $\frac{d}{dt} \int_{\Omega_{\varepsilon}(t)} u^{\varepsilon} dx = 0$ for all $t \in (0, T)$, that is, the total amount of heat $\int_{\Omega_{\varepsilon}(t)} u^{\varepsilon} dx$ is conserved.

Next, we give a definition of a variational solution to (H_{ε}) . For each $\varepsilon > 0$, we define a function space $L^2_{H^1(\varepsilon)}$ on $Q_{\varepsilon,T}$ and an inner product on $L^2_{H^1(\varepsilon)}$ as

$$L^{2}_{H^{1}(\varepsilon)} := \{ u \in L^{2}(Q_{\varepsilon,T}) \mid \nabla u \in L^{2}(Q_{\varepsilon,T}) \},\$$

$$(u_{1}, u_{2})_{L^{2}_{H^{1}(\varepsilon)}} := \int_{0}^{T} \int_{\Omega_{\varepsilon}(t)} (u_{1}u_{2} + \nabla u_{1} \cdot \nabla u_{2}) \, dx \, dt.$$

$$(2.3.1)$$

The space $L^2_{H^1(\varepsilon)}$ is a Hilbert space endowed with the above inner product. Let $\|\cdot\|_{L^2_{H^1(\varepsilon)}}$ denote the norm of $L^2_{H^1(\varepsilon)}$ induced by the inner product $(\cdot, \cdot)_{L^2_{H^1(\varepsilon)}}$.

Definition 2.3.1. Let $u_0^{\varepsilon} \in L^2(\Omega_{\varepsilon}(0))$. A function $u^{\varepsilon} \in L^2_{H^1(\varepsilon)}$ is said to be a variational solution to the initial-boundary value problem (H_{ε}) if it satisfies

$$\int_0^T \int_{\Omega_{\varepsilon}(t)} (-u^{\varepsilon} \partial_t w + \nabla u^{\varepsilon} \cdot \nabla w) \, dx \, dt - \int_{\Omega_{\varepsilon}(0)} u_0^{\varepsilon} w(0) \, dx = 0 \tag{2.3.2}$$

for all $w \in C^1(\overline{Q_{\varepsilon,T}})$ with w(T) = 0 in $\Omega_{\varepsilon}(T)$.

The variational formulation (2.3.2) is derived as follows. Suppose that u^{ε} is a classical solution to (H_{ε}) . We multiply both sides of the heat equation in $Q_{\varepsilon,T}$ by an arbitrary function $w \in C^1(\overline{Q_{\varepsilon,T}})$ with w(T) = 0 in $\Omega_{\varepsilon}(T)$ and integrate them over $Q_{\varepsilon,T}$ to get

$$\int_0^T \int_{\Omega_{\varepsilon}(t)} (\partial_t u^{\varepsilon} - \Delta u^{\varepsilon}) w \, dx \, dt = 0.$$

We calculate the left-hand side of the above equality. By the Reynolds transport theorem and the conditions $u^{\varepsilon}(0) = u_0^{\varepsilon}$ in $\Omega_{\varepsilon}(0)$ and w(T) = 0 in $\Omega_{\varepsilon}(T)$, we have

$$\int_0^T \int_{\Omega_{\varepsilon}(t)} (\partial_t u^{\varepsilon}) w \, dx \, dt = -\int_0^T \int_{\Omega_{\varepsilon}(t)} u^{\varepsilon} \partial_t w \, dx \, dt \\ -\int_0^T \int_{\partial\Omega_{\varepsilon}(t)} v_{\varepsilon}^N u^{\varepsilon} w \, d\mathcal{H}^{n-1} \, dt - \int_{\Omega_{\varepsilon}(0)} u_0^{\varepsilon} w(0) \, dx.$$

On the other hand, by integration by parts,

$$-\int_{\Omega_{\varepsilon}(t)} (\Delta u^{\varepsilon}) w \, dx \, dt = \int_{\Omega_{\varepsilon}(t)} \nabla u^{\varepsilon} \cdot \nabla w \, dx - \int_{\partial\Omega_{\varepsilon}(t)} (\partial_{\nu_{\varepsilon}} u^{\varepsilon}) w \, d\mathcal{H}^{n-1}$$

Hence it follows that

$$\int_0^T \int_{\Omega_{\varepsilon}(t)} (-u^{\varepsilon} \partial_t w + \nabla u^{\varepsilon} \cdot \nabla w) \, dx \, dt - \int_0^T \int_{\partial\Omega_{\varepsilon}(t)} (\partial_{\nu_{\varepsilon}} u^{\varepsilon} + v_{\varepsilon}^N u^{\varepsilon}) w \, d\mathcal{H}^{n-1} \, dt \\ - \int_{\Omega_{\varepsilon}(0)} u_0^{\varepsilon} w(0) \, dx = 0$$

and we obtain (2.3.2) by applying the boundary condition of (H_{ε}) to the second term of the left-hand side in the above equality.

Our goal in this section is to obtain a unique variational solution to (H_{ε}) that satisfies an energy estimate with a constant independent of ε . To this end, we transform (2.3.2) into a variational formulation of some equation on the fixed (in time) domain $\Omega_{\varepsilon}(0)$ with the aid of a suitable diffeomorphism between $\Omega_{\varepsilon}(0)$ and $\Omega_{\varepsilon}(t)$.

Lemma 2.3.2. For each $t \in [0,T]$, there exists a diffeomorphism $\Psi_{\varepsilon}(\cdot,t): \Omega_{\varepsilon}(0) \to \Omega_{\varepsilon}(t)$ with its inverse $\Psi_{\varepsilon}^{-1}(\cdot,t): \Omega_{\varepsilon}(t) \to \Omega_{\varepsilon}(0)$ such that Ψ_{ε} and Ψ_{ε}^{-1} are smooth on $\overline{\Omega_{\varepsilon}(0)} \times [0,T]$ and $\overline{Q_{\varepsilon,T}}$, respectively, and $\Psi_{\varepsilon}(\cdot,0)$ is the identity mapping on $\Omega_{\varepsilon}(0)$. Moreover, there exists a constant c > 0 independent of ε such that

$$|\partial_X^{\alpha} \partial_t^k \Psi_{\varepsilon}(X, t)| \le c, \quad |\partial_x^{\alpha} \partial_t^k \Psi_{\varepsilon}^{-1}(x, t)| \le c$$
(2.3.3)

for all $(X,t) \in \Omega_{\varepsilon}(0) \times (0,T)$, $(x,t) \in Q_{\varepsilon,T}$, and $|\alpha| + k \leq 2$, k = 0, 1, 2.

Proof. We observe that for each $X \in \Omega_{\varepsilon}(0)$ there is a unique $\theta \in (0,1)$ such that

$$X = p(X,0) + \varepsilon \{ (1-\theta)g_0(p(X,0),0) + \theta g_1(p(X,0),0) \} \nu(p(X,0),0),$$
(2.3.4)

that is, X divides the line segment A_0A_1 internally in the ratio $\theta: 1 - \theta$, where

$$A_i := p(X,0) + \varepsilon g_i(p(X,0),0)\nu(p(X,0),0), \quad i = 0, 1.$$

Based on this observation we define $\Psi_{\varepsilon}(X,t) \in \Omega_{\varepsilon}(t)$ as

$$\Psi_{\varepsilon}(X,t) := \Phi(p(X,0),t) + \varepsilon\{(1-\theta)g_0(\Phi(p(X,0),t),t) + \theta g_1(\Phi(p(X,0),t),t)\}\nu(\Phi(p(X,0),t),t), \quad (2.3.5)$$

that is, $\Psi_{\varepsilon}(X,t)$ divides the line segment B_0B_1 internally in the ratio $\theta: 1-\theta$, where

$$B_i := \Phi(p(X,0),t) + \varepsilon g_i(\Phi(p(X,0),t),t)\nu(\Phi(p(X,0),t),t), \quad i = 0, 1.$$

To eliminate θ in (2.3.5), we take the inner product of both sides of (2.3.4) and the vector $\nu(p(X,0),0)$. Then

$$\{X - p(X,0)\} \cdot \nu(p(X,0),0) = \varepsilon\{(1-\theta)g_0(p(X,0),0) + \theta g_1(p(X,0),0)\}.$$

Since $\{X - p(X, 0)\} \cdot \nu(p(X, 0), 0) = d(X, 0)$ and $g_1 - g_0 = g > 0$, it follows that

$$\theta = \frac{d(X,0) - \varepsilon g_0(p(X,0),0)}{\varepsilon g(p(X,0),0)}$$

Hence, by substituting this for θ in (2.3.5), we obtain

$$\Psi_{\varepsilon}(X,t) = \Phi(p(X,0),t) + \{d(X,0)\phi_1(X,t) + \varepsilon\phi_2(X,t)\}\nu(\Phi(p(X,0),t),t)$$
(2.3.6)

for $X \in \Omega_{\varepsilon}(0)$ and $t \in [0, T]$, where

$$\phi_1(X,t) := \frac{g(\Phi(p(X,0),t),t)}{g(p(X,0),0)}, \quad \phi_2(X,t) := g_0(\Phi(p(X,0),t),t) - \phi_1(X,t)g_0(p(X,0),0)$$

Similarly we define a mapping Ψ_{ε}^{-1} as

$$\Psi_{\varepsilon}^{-1}(x,t) := \Phi^{-1}(p(x,t),t) + \{d(x,t)\phi_3(x,t) + \varepsilon\phi_4(x,t)\}\nu(\Phi^{-1}(p(x,t),t),0)$$
(2.3.7)

for $(x,t) \in Q_{\varepsilon,T}$, where

$$\phi_3(x,t) := \frac{g(\Phi^{-1}(p(x,t),t),0)}{g(p(x,t),t)}, \quad \phi_4(x,t) := g_0(\Phi^{-1}(p(x,t),t),0) - \phi_3(x,t)g_0(p(x,t),t).$$

By definition, $\Psi_{\varepsilon}(\cdot, t) \colon \Omega_{\varepsilon}(0) \to \Omega_{\varepsilon}(t)$ is a bijection with its inverse $\Psi_{\varepsilon}^{-1}(\cdot, t) \colon \Omega_{\varepsilon}(t) \to \Omega_{\varepsilon}(0)$ for each $t \in [0, T]$. Also, since $\Phi(\cdot, 0)$ is the identity mapping on Γ_0 , we have $\phi_1(X, 0) = 1$, $\phi_2(X, 0) = 0$ and thus

$$\Psi_{\varepsilon}(X,0) = p(X,0) + d(X,0)\nu(p(X,0),0) = X \quad \text{for all} \quad X \in \Omega_{\varepsilon}(0),$$

that is, $\Psi_{\varepsilon}(\cdot, 0)$ is the identity mapping on $\Omega_{\varepsilon}(0)$. Due to the smoothness of Φ , Φ^{-1} , d, $\underline{p}, \underline{g}_0$, and \underline{g}_1 , the right-hand sides of (2.3.6) and (2.3.7) are smooth on the compact sets $\overline{N(0)} \times [0, T]$ and $\overline{N_T}$, respectively, and thus bounded independently of ε along with their derivatives. From this fact and the inclusion $\overline{\Omega_{\varepsilon}(t)} \subset N(t)$ for each $t \in [0, T]$, it follows that Ψ_{ε} and Ψ_{ε}^{-1} are smooth on $\overline{\Omega_{\varepsilon}(0)} \times [0, T]$ and $\overline{Q_{\varepsilon,T}}$, respectively, and that the inequalities (2.3.3) hold with a constant c > 0 independent of ε . In particular, $\Psi_{\varepsilon}(\cdot, t) \colon \Omega_{\varepsilon}(0) \to \Omega_{\varepsilon}(t)$ is a diffeomorphism for each $t \in [0, T]$.

Let Ψ_{ε} and Ψ_{ε}^{-1} be the mappings given by Lemma 2.3.2. In (2.3.2), we set

$$U^{\varepsilon}(X,t) := u^{\varepsilon}(\Psi_{\varepsilon}(X,t),t), \quad W(X,t) := w(\Psi_{\varepsilon}(X,t),t), \quad (X,t) \in \Omega_{\varepsilon}(0) \times (0,T).$$

Then, by the change of variables $x = \Psi_{\varepsilon}(X, t)$, we transform (2.3.2) into

$$\int_0^T \{-(U^{\varepsilon}(t), J^{\varepsilon}(t)\partial_t W(t))_{L^2} + (A^{\varepsilon}(t)\nabla U^{\varepsilon}(t) - U^{\varepsilon}(t)B^{\varepsilon}(t), \nabla W(t))_{L^2}\} dt - (u_0^{\varepsilon}, W(0))_{L^2} = 0. \quad (2.3.8)$$

Here $(\cdot, \cdot)_{L^2}$ denotes the inner product of $L^2(\Omega_{\varepsilon}(0))$ and

$$J^{\varepsilon}(X,t) := |\det \nabla \Psi_{\varepsilon}(X,t)| \in \mathbb{R},$$

$$A^{\varepsilon}(X,t) := J^{\varepsilon}(X,t) \nabla \Psi_{\varepsilon}^{-1}(\Psi_{\varepsilon}(X,t),t) [\nabla \Psi_{\varepsilon}^{-1}(\Psi_{\varepsilon}(X,t),t)]^{T} \in \mathbb{R}^{n \times n},$$

$$B^{\varepsilon}(X,t) := J^{\varepsilon}(X,t) \partial_{t} \Psi_{\varepsilon}^{-1}(\Psi_{\varepsilon}(X,t),t) \in \mathbb{R}^{n}$$

for $(X,t) \in \Omega_{\varepsilon}(0) \times (0,T)$, where

$$\nabla \Psi_{\varepsilon}^{-1} := \begin{pmatrix} \partial_1(\Psi_{\varepsilon}^{-1})_1 & \dots & \partial_n(\Psi_{\varepsilon}^{-1})_1 \\ \vdots & \ddots & \vdots \\ \partial_1(\Psi_{\varepsilon}^{-1})_n & \dots & \partial_n(\Psi_{\varepsilon}^{-1})_n \end{pmatrix} \quad \text{for} \quad \Psi_{\varepsilon}^{-1} = \begin{pmatrix} (\Psi_{\varepsilon}^{-1})_1 \\ \vdots \\ (\Psi_{\varepsilon}^{-1})_n \end{pmatrix}$$

and $[\nabla \Psi_{\varepsilon}^{-1}]^T$ denotes the transposed matrix of $\nabla \Psi_{\varepsilon}^{-1}$. Note that the vector field B^{ε} comes from the differentiation of $w(x,t) = W(\Psi_{\varepsilon}^{-1}(x,t),t)$ with respect to t holding $x \in \Omega_{\varepsilon}(t)$ fixed:

$$\partial_t w(x,t) = \partial_t W(\Psi_{\varepsilon}^{-1}(x,t),t) + \partial_t \Psi_{\varepsilon}^{-1}(x,t) \cdot \nabla W(\Psi_{\varepsilon}^{-1}(x,t),t).$$

Since w(T) = 0 in $\Omega_{\varepsilon}(T)$ and $\Psi_{\varepsilon}(\cdot, 0)$ is the identity mapping on $\Omega_{\varepsilon}(0)$, we have W(T) = 0and $J^{\varepsilon}(0) = 1$ in $\Omega_{\varepsilon}(0)$. Thus, by integration by parts with respect to t, we further transform (2.3.8) into

$$\int_0^T \{_{(H^1)'} \langle \partial_t U^{\varepsilon}(t), J^{\varepsilon}(t) W(t) \rangle_{H^1} + (U^{\varepsilon}(t), W(t) \partial_t J^{\varepsilon}(t))_{L^2} + (A^{\varepsilon}(t) \nabla U^{\varepsilon}(t) - U^{\varepsilon}(t) B^{\varepsilon}(t), \nabla W(t))_{L^2} \} dt = 0.$$
(2.3.9)

Here $_{(H^1)'}\langle \cdot, \cdot \rangle_{H^1}$ is the duality product between $H^1(\Omega_{\varepsilon}(0))$ and its dual space $(H^1(\Omega_{\varepsilon}(0)))'$.

Theorem 2.3.3. For every $u_0^{\varepsilon} \in L^2(\Omega_{\varepsilon}(0))$, there exists a unique function

$$U^{\varepsilon} \in L^{\infty}(0,T; L^{2}(\Omega_{\varepsilon}(0))) \cap L^{2}(0,T; H^{1}(\Omega_{\varepsilon}(0))) \quad with \quad \partial_{t}U^{\varepsilon} \in L^{2}(0,T; (H^{1}(\Omega_{\varepsilon}(0)))')$$

that satisfies (2.3.9) for all $W \in L^2(0,T; H^1(\Omega_{\varepsilon}(0)))$ and $U^{\varepsilon}(0) = u_0^{\varepsilon}$ in $L^2(\Omega_{\varepsilon}(0))$. Moreover, there exists a constant c > 0 independent of u_0^{ε} , U^{ε} , and ε such that

$$\sup_{t \in (0,T)} \|U^{\varepsilon}(t)\|_{L^{2}(\Omega_{\varepsilon}(0))}^{2} + \int_{0}^{T} \|\nabla U^{\varepsilon}(t)\|_{L^{2}(\Omega_{\varepsilon}(0))}^{2} dt \le c \|u_{0}^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon}(0))}^{2}.$$
(2.3.10)

Proof. For i, j = 1, ..., n, let A_{ij}^{ε} be the (i, j)-entry of A^{ε} and B_i^{ε} be the *i*-th component of B^{ε} . Suppose that there is a positive constant C independent of ε such that

$$C^{-1} \le J^{\varepsilon}(X, t) \le C, \tag{2.3.11}$$

$$|\nabla J^{\varepsilon}(X,t)| \le C, \quad |\partial_t J^{\varepsilon}(X,t)| \le C, \quad |A_{ij}^{\varepsilon}(X,t)| \le C, \quad |B_i^{\varepsilon}(X,t)| \le C, \quad (2.3.12)$$

$$A^{\varepsilon}(X,t)\zeta \cdot \zeta \ge C|\zeta|^2 \tag{2.3.13}$$

for all $(X,t) \in \Omega_{\varepsilon}(0) \times (0,T)$, $\zeta \in \mathbb{R}^n$, and i, j = 1, ..., n. Then the theorem is proved by a standard Galerkin method and Gronwall argument, see [10, Section 7.1] for details. In particular, the constant c in (2.3.10) depends only on the above C and thus it is independent of ε .

Let us prove (2.3.11), (2.3.12), and (2.3.13). The inequalities (2.3.12) and the right-hand inequality of (2.3.11) immediately follow from (2.3.3). For all $(X,t) \in \Omega_{\varepsilon}(0) \times (0,T)$ since $\nabla \Psi_{\varepsilon}^{-1}(\Psi_{\varepsilon}(X,t),t) \nabla \Psi_{\varepsilon}(X,t) = I_n$ it follows that

$$|\det \nabla \Psi_{\varepsilon}^{-1}(\Psi_{\varepsilon}(X,t),t)|J^{\varepsilon}(X,t) = 1, \quad [\nabla \Psi_{\varepsilon}(X,t)]^{T}[\nabla \Psi_{\varepsilon}^{-1}(\Psi_{\varepsilon}(X,t),t)]^{T} = I_{n}.$$

The first equality yields the left-hand inequality of (2.3.11) because $|\det \nabla \Psi_{\varepsilon}^{-1}|$ is bounded on $Q_{\varepsilon,T}$ independently of ε by (2.3.3). Moreover, the above equality and (2.3.3) imply that, for all $(X,t) \in \Omega_{\varepsilon}(0) \times (0,T)$ and $\zeta \in \mathbb{R}^{n}$,

$$\begin{split} |\zeta|^2 &= |[\nabla \Psi_{\varepsilon}(X,t)]^T [\nabla \Psi_{\varepsilon}^{-1}(\Psi_{\varepsilon}(X,t),t)]^T \zeta|^2 \\ &\leq c |[\nabla \Psi_{\varepsilon}^{-1}(\Psi_{\varepsilon}(X,t),t)]^T \zeta|^2 \\ &= c \{\nabla \Psi_{\varepsilon}^{-1}(\Psi_{\varepsilon}(X,t),t) [\nabla \Psi_{\varepsilon}^{-1}(\Psi_{\varepsilon}(X,t),t)]^T \zeta\} \cdot \zeta \\ &= c |\det \nabla \Psi_{\varepsilon}^{-1}(\Psi_{\varepsilon}(X,t),t)| A^{\varepsilon}(X,t) \zeta \cdot \zeta \leq c A^{\varepsilon}(X,t) \zeta \cdot \zeta \end{split}$$

with a constant c > 0 independent of ε . Thus (2.3.13) follows.

Now we can show the existence and uniqueness of a variational solution to (H_{ε}) and its energy estimate with a constant independent of ε .

Theorem 2.3.4. For every $u_0^{\varepsilon} \in L^2(\Omega_{\varepsilon}(0))$, there exists a unique variational solution u^{ε} to (H_{ε}) . Moreover, u^{ε} satisfies that $u^{\varepsilon}(0) = u_0^{\varepsilon}$ in $L^2(\Omega_{\varepsilon}(0))$ and

$$\sup_{t \in (0,T)} \|u^{\varepsilon}(t)\|_{L^{2}(\Omega_{\varepsilon}(t))}^{2} + \int_{0}^{T} \|\nabla u^{\varepsilon}(t)\|_{L^{2}(\Omega_{\varepsilon}(t))}^{2} dt \le c \|u_{0}^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon}(0))}^{2}$$
(2.3.14)

with a constant c > 0 independent of u_0^{ε} , u^{ε} , and ε .

Proof. For each $u_0^{\varepsilon} \in L^2(\Omega_{\varepsilon}(0))$, let U^{ε} be the unique function given by Theorem 2.3.3 and we set

$$u^{\varepsilon}(x,t) := U^{\varepsilon}(\Psi_{\varepsilon}^{-1}(x,t),t), \quad (x,t) \in Q_{\varepsilon,T}.$$

Since $\Psi_{\varepsilon}(\cdot, 0)$ is the identity mapping on $\Omega_{\varepsilon}(0)$ by Lemma 2.3.2 and $U^{\varepsilon}(0) = u_0^{\varepsilon}$ in $L^2(\Omega_{\varepsilon}(0))$ by Theorem 2.3.3, we have $u^{\varepsilon}(0) = u_0^{\varepsilon}$ in $L^2(\Omega_{\varepsilon}(0))$. Let us show that u^{ε} satisfies (2.3.2) for all $w \in C^1(\overline{Q_{\varepsilon,T}})$ with w(T) = 0 in $\Omega_{\varepsilon}(T)$. Since Ψ_{ε} is smooth on $\overline{\Omega_{\varepsilon}(0)} \times [0,T]$, a function

$$W(X,t) := w(\Psi_{\varepsilon}(X,t),t), \quad (X,t) \in \overline{\Omega_{\varepsilon}(0)} \times [0,T]$$

is in $C^1(\overline{\Omega_{\varepsilon}(0)} \times [0,T])$ and satisfies W(T) = 0 in $\Omega_{\varepsilon}(0)$. Hence we can substitute it for W in (2.3.9) and integrate by parts with respect to t to get (2.3.8). By changing variables $X = \Psi_{\varepsilon}^{-1}(x,t)$ in (2.3.8), we obtain (2.3.2).

Next we prove the energy estimate (2.3.14). By the change of variables $x = \Psi_{\varepsilon}(X, t)$ we have

$$\int_{\Omega_{\varepsilon}(t)} |u^{\varepsilon}(x,t)|^2 dx = \int_{\Omega_{\varepsilon}(0)} |U^{\varepsilon}(X,t)|^2 |\det \nabla \Psi_{\varepsilon}(X,t)| dX,$$
$$\int_{\Omega_{\varepsilon}(t)} |\nabla u^{\varepsilon}(x,t)|^2 dx = \int_{\Omega_{\varepsilon}(0)} |[\nabla \Psi_{\varepsilon}^{-1}(\Psi_{\varepsilon}(X,t),t)]^T \nabla U^{\varepsilon}(X,t)|^2 |\det \nabla \Psi_{\varepsilon}(X,t)| dX.$$

for all $t \in [0, T]$. Hence the inequalities (2.3.3) yield

$$\|u^{\varepsilon}(t)\|_{L^{2}(\Omega_{\varepsilon}(t))}^{2} \leq c\|U^{\varepsilon}(t)\|_{L^{2}(\Omega_{\varepsilon}(0))}^{2}, \quad \|\nabla u^{\varepsilon}(t)\|_{L^{2}(\Omega_{\varepsilon}(t))}^{2} \leq c\|\nabla U^{\varepsilon}(t)\|_{L^{2}(\Omega_{\varepsilon}(0))}^{2}$$

with a constant c > 0 independent of ε . By these inequalities and (2.3.10), we obtain (2.3.14) and thus $u^{\varepsilon} \in L^2_{H^1(\varepsilon)}$. Hence u^{ε} is a variational solution to (H_{ε}) .

Finally, the uniqueness of a variational solution to (H_{ε}) follows from that of a function given by Theorem 2.3.3. The proof is complete.

Remark 2.3.5. Let u^{ε} be the unique variational solution to (H_{ε}) with initial data $u_0^{\varepsilon} \in L^2(\Omega_{\varepsilon}(0))$. Then it immediately follows from (2.3.14) that

$$\|u^{\varepsilon}\|_{L^{2}_{H^{1}(\varepsilon)}} \le c \|u^{\varepsilon}_{0}\|_{L^{2}(\Omega_{\varepsilon}(0))}, \qquad (2.3.15)$$

where c > 0 is a constant independent of u_0^{ε} , u^{ε} , and ε . We will use this inequality in Section 2.6.

2.4 Basic function spaces on evolving surfaces

In this section, we define function spaces on the space-time manifold S_T introduced by Olshanskii, Reusken, and Xu [19] and give their properties. These spaces will give an appropriate variational formulation of a limit equation on $\Gamma(t)$ we will derive in Section 2.6. All results in this section are originally obtained in [19] for the three-dimensional case. They can be easily extended for arbitrary dimensions and we give proofs of them for the readers' convenience.

For each fixed T > 0, we define a function space H_T and an inner product on H_T as

$$H_T := \{ v \in L^2(S_T) \mid \nabla_{\Gamma(t)} v \in L^2(S_T) \},\$$

$$(v_1, v_2)_{H_T} := \int_0^T \int_{\Gamma(t)} \{ v_1(y, t) v_2(y, t) + \nabla_{\Gamma(t)} v_1(y, t) \cdot \nabla_{\Gamma(t)} v_2(y, t) \} d\mathcal{H}^{n-1}(y) dt.$$

(2.4.1)

This inner product induces the norm $\|\cdot\|_{H_T}$ that is equivalent to the one induced by the inner product $\int_{S_T} \{v_1(\sigma)v_2(\sigma) + \nabla_{\Gamma(t)}v_1(\sigma) \cdot \nabla_{\Gamma(t)}v_2(\sigma)\} d\mathcal{H}^n(\sigma)$, since the identity

$$\int_{0}^{T} \int_{\Gamma(t)} f(y,t) \, d\mathcal{H}^{n-1}(y) \, dt = \int_{S_T} f(\sigma) (1+|v_{\Gamma}^N(\sigma)|^2)^{-1/2} \, d\mathcal{H}^n(\sigma) \tag{2.4.2}$$

holds and v_{Γ}^N is bounded on S_T . This identity is stated in [19] without proof. We give the proof of (2.4.2) in Appendix 2.B for the readers' convenience. If $T_1 < T_2$, then H_{T_2} is continuously embedded into H_{T_1} just by restricting elements of H_{T_2} on S_{T_1} .

Next we define an auxiliary space. Let $H^1(\Gamma_0) := \{V \in L^2(\Gamma_0) \mid \nabla_{\Gamma_0} V \in L^2(\Gamma_0)\}$ with the inner product $(V_1, V_2)_{H^1(\Gamma_0)} := \int_{\Gamma_0} (V_1 V_2 + \nabla_{\Gamma_0} V_1 \cdot \nabla_{\Gamma_0} V_2) d\mathcal{H}^{n-1}$, where ∇_{Γ_0} is the tangential gradient on Γ_0 . Then we define a Hilbert space \hat{H}_T as

$$\widehat{H}_T := L^2(0,T; H^1(\Gamma_0)), \quad (V_1, V_2)_{\widehat{H}_T} := \int_0^T (V_1(t), V_2(t))_{H^1(\Gamma_0)} dt$$

and let $\|\cdot\|_{\widehat{H}_T}$ denote the norm of \widehat{H}_T induced by the inner product $(\cdot, \cdot)_{\widehat{H}_T}$.

Let $\Phi(\cdot, t) \colon \Gamma_0 \to \Gamma(t)$ be the flow map of V_{Γ} and $\Phi^{-1}(\cdot, t)$ be its inverse mapping (see Section 2.2). For a function V on $\Gamma_0 \times (0, T)$, we define a function v = LV on S_T as

$$v(y,t) := V(\Phi^{-1}(y,t),t), \quad (y,t) \in S_T.$$
 (2.4.3)

Also, for a function v on S_T , we define a function $V = L^{-1}v$ on $\Gamma_0 \times (0,T)$ as

$$V(Y,t) := v(\Phi(Y,t),t), \quad (Y,t) \in \Gamma_0 \times (0,T).$$

Clearly L and L^{-1} are linear mappings and satisfy $L^{-1}(LV) = V$ and $L(L^{-1}v) = v$ for all functions V on $\Gamma_0 \times (0,T)$ and v on S_T .

Lemma 2.4.1. The linear mapping L given by (2.4.3) defines an isomorphism between \hat{H}_T and H_T .

A short proof is given in [19]. We give a detailed proof in Appendix 2.B for the completeness.

Let $C_0^1(S_T)$ be the space of all functions in $C^1(S_T)$ with compact support in S_T . That is, each function in $C_0^1(S_T)$ vanishes near t = 0 and t = T.

Lemma 2.4.2. The space H_T is a Hilbert space and $C_0^1(S_T)$ is dense in H_T .

Proof. Since \widehat{H}_T is a Hilbert space, Lemma 2.4.1 implies that H_T is a Hilbert space. Also, since $C_0^1(\Gamma_0 \times (0,T))$ is dense in \widehat{H}_T (see [19, Lemma 3.1]) and $C_0^1(S_T)$ includes $L[C_0^1(\Gamma_0 \times (0,T))]$, Lemma 2.4.1 again implies that $C_0^1(S_T)$ is dense in H_T .

The space H_T is continuously embedded into $L^2(S_T)$. Moreover, H_T is dense in $L^2(S_T)$ since it includes a dense subspace $C_0^1(S_T)$ of $L^2(S_T)$. Hence we have continuous and dense embeddings $H_T \hookrightarrow L^2(S_T) \hookrightarrow H'_T$, where H'_T is the dual space of H_T .

For $v \in C^1(S_T)$, we define its (strong) material derivative $\partial^{\bullet} v$ as

$$\partial^{\bullet} v(\Phi(Y,t),t) := \frac{d}{dt} (v(\Phi(Y,t),t)), \quad (Y,t) \in \Gamma_0 \times (0,T).$$
(2.4.4)

From the Leibniz formula (see [4, Lemma 2.2])

$$\frac{d}{dt} \int_{\Gamma(t)} v \, d\mathcal{H}^{n-1} = \int_{\Gamma(t)} (\partial^{\bullet} v + v \operatorname{div}_{\Gamma(t)} V_{\Gamma}) \, d\mathcal{H}^{n-1}, \quad v \in C^{1}(S_{T}),$$

we have the integration by parts identity

-

$$\int_{0}^{T} \int_{\Gamma(t)} (v_{2}\partial^{\bullet}v_{1} + v_{1}\partial^{\bullet}v_{2} + v_{1}v_{2}\operatorname{div}_{\Gamma(t)}V_{\Gamma}) d\mathcal{H}^{n-1} dt$$
$$= \int_{\Gamma(T)} v_{1}(T)v_{2}(T) d\mathcal{H}^{n-1} - \int_{\Gamma(0)} v_{1}(0)v_{2}(0) d\mathcal{H}^{n-1} \quad (2.4.5)$$

for all $v_1, v_2 \in C^1(S_T)$. Based on this identity, we define the weak material derivative of $v \in H_T$ as a functional $\partial^{\bullet} v$ such that

$$\langle \partial^{\bullet} v, \psi \rangle_T := -\int_0^T \int_{\Gamma(t)} (v \,\partial^{\bullet} \psi + v\psi \operatorname{div}_{\Gamma(t)} V_{\Gamma}) \, d\mathcal{H}^{n-1} \, dt, \quad \psi \in C_0^1(S_T).$$
(2.4.6)

If $v \in C^1(S_T)$, then its weak material derivative agrees with the strong one. We set

$$\|\partial^{\bullet}v\|_{H'_T} := \sup_{\psi \in C_0^1(S_T) \setminus \{0\}} \frac{\langle \partial^{\bullet}v, \psi \rangle_T}{\|\psi\|_{H_T}}, \quad v \in H_T.$$

If $\|\partial^{\bullet}v\|_{H_T}$ is finite for some $v \in H_T$, then $\partial^{\bullet}v$ can be extended to a bounded linear functional on H_T because $C_0^1(S_T)$ is dense in H_T (see Lemma 2.4.2). In this case, we say that $\partial^{\bullet}v$ is in H_T' and we define a function space W_T and its norm as

$$W_T := \{ v \in H_T \mid \partial^{\bullet} v \in H_T' \}, \quad \|v\|_{W_T} := \left(\|v\|_{H_T}^2 + \|\partial^{\bullet} v\|_{H_T'}^2 \right)^{1/2}.$$
(2.4.7)

For $T_1 < T_2$, the space W_{T_2} is continuously embedded into W_{T_1} since $C_0^1(S_{T_1}) \subset C_0^1(S_{T_2})$ and H_{T_2} is continuously embedded into H_{T_1} .

To investigate properties of W_T , we define an auxiliary Hilbert space and its norm as

$$\widehat{W}_T := \{ V \in \widehat{H}_T \mid \partial_t V \in \widehat{H}'_T \}, \quad \|V\|_{\widehat{W}_T} := \left(\|V\|_{\widehat{H}_T}^2 + \|\partial_t V\|_{\widehat{H}'_T}^2 \right)^{1/2}$$

Here \hat{H}'_T is the dual space of \hat{H}_T and $\partial_t V$ is the weak time derivative of $V \in \hat{H}_T$ defined as

$$[\partial_t V, \Psi]_T := -\int_0^T \int_{\Gamma_0} V \,\partial_t \Psi \,d\mathcal{H}^{n-1} \,dt, \quad \Psi \in C_0^1(\Gamma_0 \times (0, T)),$$

and we say $\partial_t V \in \widehat{H}'_T$ if $\|\partial_t V\|_{\widehat{H}'_T} := \sup_{\Psi \in C^1_0(\Gamma_0 \times (0,T)) \setminus \{0\}} [\partial_t V, \Psi]_T / \|\Psi\|_{\widehat{H}_T}$ is finite.

Lemma 2.4.3. The linear mapping L given by (2.4.3) defines an isomorphism between \widehat{W}_T and W_T .

A proof for the three-dimensional case is given in [19] and easily extended for arbitrary dimensions. We give a complete proof in Appendix 2.B for the readers' convenience.

Lemma 2.4.3 shows that W_T has similar properties to those of W_T .

Lemma 2.4.4. The space W_T is a Hilbert space and $C^1(S_T)$ is dense in W_T . Moreover, the trace operator $v \mapsto v(t)$ from $C^1(S_T)$ into $L^2(\Gamma(t))$ for each $t \in [0,T]$ can be extended to a bounded linear operator from W_T to $L^2(\Gamma(t))$ and there exists a constant c > 0 such that

$$\max_{t \in [0,T]} \|v(t)\|_{L^2(\Gamma(t))} \le c \|v\|_{W_T}$$

for all $v \in W_T$.

Proof. Since \widehat{W}_T is a Hilbert space, Lemma 2.4.3 implies that W_T is a Hilbert space. For the rest of the proof, see [19, Theorem 3.6].

Finally we show an integration by parts formula which we will use in Section 2.6.

Lemma 2.4.5. For all $v_1, v_2 \in W_T$, we have

$$\begin{aligned} \langle \partial^{\bullet} v_1, v_2 \rangle_T + \langle \partial^{\bullet} v_2, v_1 \rangle_T + \int_0^T \int_{\Gamma(t)} v_1 v_2 \operatorname{div}_{\Gamma(t)} V_{\Gamma} \, d\mathcal{H}^{n-1} \, dt \\ &= \int_{\Gamma(T)} v_1(T) v_2(T) \, d\mathcal{H}^{n-1} - \int_{\Gamma_0} v_1(0) v_2(0) \, d\mathcal{H}^{n-1}. \end{aligned}$$
(2.4.8)

Note that, by Lemma 2.4.4, $v_i(0)$ and $v_i(T)$, i = 1, 2, are well-defined as functions in $L^2(\Gamma_0)$ and $L^2(\Gamma(T))$, respectively.

Proof. For $v \in C^1(S_T)$, its weak material derivative agrees with the strong one. Thus we have

$$\langle \partial^{\bullet} v, \psi \rangle_T = \int_0^T \int_{\Gamma(t)} (\partial^{\bullet} v) \psi \, d\mathcal{H}^{n-1} \, dt, \quad \psi \in C_0^1(S_T).$$

Moreover, since $C_0^1(S_T)$ is dense in H_T (see Lemma 2.4.2), the above equality holds for all $\psi \in H_T$ and thus (2.4.8) follows from (2.4.5) when $v_1, v_2 \in C^1(S_T)$. Since $C^1(S_T)$ is dense in W_T (see Lemma 2.4.4), a density argument shows that (2.4.8) holds for general $v_1, v_2 \in W_T$.

2.5 Average operator

2.5.1 Definition and basic properties of the average operator

In this section we define and investigate a weighted average operator. Lemma 2.5.6 and Lemma 2.5.13 are fundamental to derivation of a limit equation of (H_{ε}) in Section 2.6. Other results in this section are also useful themselves.

For $(y,t) \in \overline{S_T}$, let $\kappa_1(y,t), \ldots, \kappa_{n-1}(y,t)$ be the principal curvatures of $\Gamma(t)$ at y (see [11, Section 14.6]). We define a function J on $\overline{S_T} \times (-\delta, \delta)$ as

$$J(y,t,\rho) := \prod_{i=1}^{n-1} \{1 - \rho \kappa_i(y,t)\}, \quad (y,t) \in \overline{S_T}, \ \rho \in (-\delta,\delta).$$

Here $\delta > 0$ is a half of the width of the tubular neighborhood N(t) of $\Gamma(t)$, which is independent of $t \in [0, T]$ (see Section 2.2). The function J is the Jacobian appearing in the transformation formula

$$\int_{\Omega_{\varepsilon}(t)} u(x) \, dx = \int_{\Gamma(t)} \int_{\varepsilon g_0(y,t)}^{\varepsilon g_1(y,t)} u(y + \rho \nu(y,t)) J(y,t,\rho) \, d\rho \, d\mathcal{H}^{n-1}(y) \tag{2.5.1}$$

for a function u on $\Omega_{\varepsilon}(t)$ with each fixed $t \in [0, T]$, see (14.98) in [11]. This formula can be viewed as a co-area formula. Based on this formula, we define a weighted average operator M_{ε} as follows.

Definition 2.5.1. For a function u on $Q_{\varepsilon,T}$, we define its weighted average $M_{\varepsilon}u$ as

$$M_{\varepsilon}u(y,t) := \frac{1}{\varepsilon g(y,t)} \int_{\varepsilon g_0(y,t)}^{\varepsilon g_1(y,t)} u(y+\rho\nu(y,t),t) J(y,t,\rho) \, d\rho, \quad (y,t) \in S_T.$$
(2.5.2)

We use the same notation $M_{\varepsilon}u$ for the average of a function u on $\Omega_{\varepsilon}(t)$ with each fixed $t \in [0, T]$.

Before starting to derive properties of the average operator, we give inequalities which we use throughout Section 2.5 and Section 2.6. Since $\kappa_1, \ldots, \kappa_{n-1}$ are smooth on $\overline{S_T}$, they are bounded on $\overline{S_T}$ along with their derivatives. Hence, by taking $\delta > 0$ sufficiently small, we may assume that there exists a constant c > 0 such that

$$c^{-1} \le 1 - \rho \kappa_i(y, t) \le c \quad \text{for all} \quad (y, t) \in \overline{S_T}, \ \rho \in (-\delta, \delta), \ i = 1, \dots, n-1.$$
(2.5.3)

Then J is smooth and bounded on $\overline{S_T} \times (-\delta, \delta)$ along with its derivatives and satisfies

$$c^{-1} \le J(y,t,\rho) \le c$$
 for all $(y,t) \in \overline{S_T}, \rho \in (-\delta,\delta).$ (2.5.4)

Moreover, since $J(y, t, \rho)$ is of the form

$$J(y,t,\rho) = 1 - \rho \sum_{i=1}^{n-1} \kappa_i(y,t) + \rho^2 P(\kappa_1(y,t),\dots,\kappa_{n-1}(y,t),\rho),$$

where P(z) is a polynomial in $z = (z_1, \ldots, z_n) \in \mathbb{R}^n$, we have

$$|1 - J(y, t, \rho)| \le c\varepsilon, \quad |\nabla_{\Gamma(t)} J(y, t, \rho)| \le c\varepsilon, \quad \left|\frac{\partial J}{\partial \rho}(y, t, \rho)\right| \le c$$
(2.5.5)

for all $(y,t) \in \overline{S_T}$ and $\rho \in (\varepsilon g_0(y,t), \varepsilon g_1(y,t))$ with a constant c > 0 independent of ε .

Now let us derive properties of the average operator M_{ε} . For a function u on $Q_{\varepsilon,T}$, we set

$$u^{\sharp}(y,t,\rho) := u(y + \rho\nu(y,t),t), \quad (y,t) \in S_T, \, \rho \in (\varepsilon g_0(y,t), \varepsilon g_1(y,t)).$$
(2.5.6)

For simplicity, we omit arguments of functions unless we need to specify them. For example, the co-area formula (2.5.1) is referred to as

$$\int_{\Omega_{\varepsilon}(t)} u \, dx = \int_{\Gamma(t)} \int_{\varepsilon g_0}^{\varepsilon g_1} u^{\sharp} J \, d\rho \, d\mathcal{H}^{n-1}.$$

Throughout the rest of this subsection and the next subsection, we fix $t \in [0, T]$ and omit it. For example, we refer to $\Gamma(t)$ as Γ . Also, let *c* denote a general positive constant independent of *t*. **Lemma 2.5.2.** If $v \in L^2(\Gamma)$, then its constant extension \overline{v} in the normal direction of Γ is in $L^2(\Omega_{\varepsilon})$. Moreover, there exists a constant c > 0 independent of ε such that

$$\|\overline{v}\|_{L^2(\Omega_{\varepsilon})} \le c\varepsilon^{1/2} \|v\|_{L^2(\Gamma)}.$$
(2.5.7)

Proof. By the co-area formula (2.5.1) and (2.5.4),

$$\|\overline{v}\|_{L^{2}(\Omega_{\varepsilon})}^{2} = \int_{\Gamma} \int_{\varepsilon g_{1}}^{\varepsilon g_{0}} |v|^{2} J \, d\rho \, d\mathcal{H}^{n-1} \le c \int_{\Gamma} \varepsilon g |v|^{2} \, d\mathcal{H}^{n-1} \le c\varepsilon \|v\|_{L^{2}(\Gamma)}^{2}.$$

Thus (2.5.7) follows.

Lemma 2.5.3. If $u \in L^2(\Omega_{\varepsilon})$, then $M_{\varepsilon}u \in L^2(\Gamma)$ and

$$\|M_{\varepsilon}u\|_{L^{2}(\Gamma)} \leq c\varepsilon^{-1/2} \|u\|_{L^{2}(\Omega_{\varepsilon})}$$

$$(2.5.8)$$

with a constant c > 0 independent of ε .

Proof. By Hölder's inequality, (2.5.4), (2.2.1), and the co-area formula (2.5.1),

$$\begin{split} \int_{\Gamma} |M_{\varepsilon}u|^2 \, d\mathcal{H}^{n-1} &\leq \int_{\Gamma} (\varepsilon g)^{-2} \bigg(\int_{\varepsilon g_0}^{\varepsilon g_1} J \, d\rho \bigg) \bigg(\int_{\varepsilon g_0}^{\varepsilon g_1} |u^{\sharp}|^2 J \, d\rho \bigg) d\mathcal{H}^{n-1} \\ &\leq c \int_{\Gamma} (\varepsilon g)^{-1} \int_{\varepsilon g_0}^{\varepsilon g_1} |u^{\sharp}|^2 J \, d\rho \, d\mathcal{H}^{n-1} \leq c \varepsilon^{-1} \int_{\Omega_{\varepsilon}} |u|^2 \, dx. \end{split}$$

Thus (2.5.8) follows.

By Lemma 2.5.2 and Lemma 2.5.3, the constant extension of $M_{\varepsilon}u$ in the normal direction of Γ is in $L^2(\Omega_{\varepsilon})$ for all $u \in L^2(\Omega_{\varepsilon})$. Let us estimate the difference between u and $M_{\varepsilon}u$ in $L^2(\Omega_{\varepsilon})$.

Lemma 2.5.4. There exists a constant c > 0 independent of ε such that

$$\left\| u - \overline{M_{\varepsilon} u} \right\|_{L^{2}(\Omega_{\varepsilon})} \le c \varepsilon \| u \|_{H^{1}(\Omega_{\varepsilon})}$$
(2.5.9)

for all $u \in H^1(\Omega_{\varepsilon})$. Here $\overline{M_{\varepsilon}u}$ is the constant extension of $M_{\varepsilon}u$ in the normal direction of Γ .

Proof. For $y \in \Gamma$ and $\rho \in (\varepsilon g_0(y), \varepsilon g_1(y))$, we set

$$I_1(y,\rho) = \frac{1}{\varepsilon g(y)} \int_{\varepsilon g_0(y)}^{\varepsilon g_1(y)} \{ u^{\sharp}(y,\rho) - u^{\sharp}(y,r) \} dr,$$
$$I_2(y) = \frac{1}{\varepsilon g(y)} \int_{\varepsilon g_0(y)}^{\varepsilon g_1(y)} u^{\sharp}(y,r) \{ 1 - J(y,r) \} dr.$$

Then we have $u^{\sharp}(y,\rho) - M_{\varepsilon}u(y) = I_1(y,\rho) + I_2(y)$. Let us estimate I_1 and I_2 . Since

$$\begin{aligned} |u^{\sharp}(y,\rho) - u^{\sharp}(y,r)| &= \left| \int_{r}^{\rho} \frac{d}{d\eta} (u(y+\eta\nu(y))) \, d\eta \right| \\ &= \left| \int_{r}^{\rho} \nu(y) \cdot \nabla u(y+\eta\nu(y)) \, d\eta \right| \le \int_{\varepsilon g_{0}(y)}^{\varepsilon g_{1}(y)} |(\nabla u)^{\sharp}(y,\eta)| \, d\eta \end{aligned}$$

for all $\rho, r \in (\varepsilon g_0(y), \varepsilon g_1(y))$ and the right-hand side of the above inequality is independent of r,

$$|I_1(y,\rho)| \le \int_{\varepsilon g_0(y)}^{\varepsilon g_1(y)} |(\nabla u)^{\sharp}(y,\eta)| \, d\eta.$$

On the other hand, by (2.2.1) and (2.5.5) we have

$$|I_2(y)| \le c \int_{\varepsilon g_0(y)}^{\varepsilon g_1(y)} |u^{\sharp}(y,r)| \, dr.$$

These inequalities and Hölder's inequality yield

$$\begin{aligned} |u^{\sharp}(y,\rho) - M_{\varepsilon}u(y)| &\leq |I_{1}(y,\rho)| + |I_{2}(y)| \leq c \int_{\varepsilon g_{0}(y)}^{\varepsilon g_{1}(y)} (|u^{\sharp}(y,r)| + |(\nabla u)^{\sharp}(y,r)|) \, dr \\ &\leq c \left(\varepsilon g(y) \int_{\varepsilon g_{0}(y)}^{\varepsilon g_{1}(y)} (|u^{\sharp}(y,r)|^{2} + |(\nabla u)^{\sharp}(y,r)|^{2}) \, dr \right)^{1/2}. \end{aligned}$$

Here the last term is independent of ρ . Hence by the co-area formula (2.5.1) and (2.5.4) we obtain

$$\begin{split} \left\| u - \overline{M_{\varepsilon}u} \right\|_{L^{2}(\Omega_{\varepsilon})}^{2} &= \int_{\Gamma} \int_{\varepsilon g_{0}(y)}^{\varepsilon g_{1}(y)} |u^{\sharp}(y,\rho) - M_{\varepsilon}u(y)|^{2} J(y,\rho) \, d\rho \, d\mathcal{H}^{n-1}(y) \\ &\leq c \int_{\Gamma} \{\varepsilon g(y)\}^{2} \int_{\varepsilon g_{0}(y)}^{\varepsilon g_{1}(y)} (|u^{\sharp}(y,r)|^{2} + |(\nabla u)^{\sharp}(y,r)|^{2}) \, dr \, d\mathcal{H}^{n-1}(y) \\ &\leq c \varepsilon^{2} \int_{\Gamma} \int_{\varepsilon g_{0}(y)}^{\varepsilon g_{1}(y)} (|u^{\sharp}(y,r)|^{2} + |(\nabla u)^{\sharp}(y,r)|^{2}) J(y,r) \, dr \, d\mathcal{H}^{n-1}(y) \\ &= c \varepsilon^{2} ||u||_{H^{1}(\Omega_{\varepsilon})}^{2}. \end{split}$$

Thus (2.5.9) follows.

2.5.2 Tangential gradient of the average operator

In this subsection, we investigate relations between the usual gradient operator in Ω_{ε} and the tangential gradient operator on Γ . We first establish estimates for the gradient of the constant extension of a function on Γ in the normal direction of Γ .

Lemma 2.5.5. If $v \in H^1(\Gamma)$, then its constant extension \overline{v} in the normal direction of Γ is in $H^1(\Omega_{\varepsilon})$. Moreover, there exists a constant c > 0 independent of ε such that

$$\|\nabla \overline{v}\|_{L^{2}(\Omega_{\varepsilon})} \leq c\varepsilon^{1/2} \|\nabla_{\Gamma}v\|_{L^{2}(\Gamma)}, \quad \left\|\nabla \overline{v} - \overline{\nabla_{\Gamma}v}\right\|_{L^{2}(\Omega_{\varepsilon})} \leq c\varepsilon^{3/2} \|\nabla_{\Gamma}v\|_{L^{2}(\Gamma)}.$$
(2.5.10)

Proof. The first inequality of (2.5.10) and Lemma 2.5.2 imply $\overline{v} \in H^1(\Omega_{\varepsilon})$ for all $v \in H^1(\Gamma)$. The inequalities (2.5.10) follow from the co-area formula (2.5.1), (2.5.4), and the inequalities

$$|\nabla \overline{v}(y + \rho\nu(y))| \le c |\nabla_{\Gamma}v(y)|, \quad |\nabla \overline{v}(y + \rho\nu(y)) - \nabla_{\Gamma}v(y)| \le c\varepsilon |\nabla_{\Gamma}v(y)|$$
(2.5.11)

for all $y \in \Gamma$ and $\rho \in (\varepsilon g_0(y), \varepsilon g_1(y))$. We prove (2.5.11) in Appendix 2.C. Here we give the main idea for the proof. We fix each $y_0 \in \Gamma$. By a rotation of coordinates, we can take

a smooth function $f: U \to \mathbb{R}$ with an open set U in \mathbb{R}^{n-1} such that Γ is described as the graph of f near y_0 and

$$\nabla' f(s_0) = 0, \quad (\nabla')^2 f(s_0) = \operatorname{diag}[\kappa_1(y_0), \dots, \kappa_{n-1}(y_0)],$$

where $y_0 = (s_0, f(s_0))$ with $s_0 \in U$ and ∇' is the gradient in $s \in \mathbb{R}^{n-1}$ (see [11, Section 14.6]). Then (2.5.11) at y_0 is proved by direct calculations under this local coordinate system. \Box

Next we approximate an H^1 -bilinear form on Ω_{ε} by that on Γ with the tangential gradient of the weighted average of a function on Ω_{ε} .

Lemma 2.5.6. For $u \in C^{\infty}(\Omega_{\varepsilon}) \cap H^{1}(\Omega_{\varepsilon})$ and $\varphi \in H^{1}(\Gamma)$, let

$$I_{\varepsilon}^{1}(u,\varphi) := \int_{\Omega_{\varepsilon}} \nabla u \cdot \nabla \overline{\varphi} \, dx - \varepsilon \int_{\Gamma} g \nabla_{\Gamma} M_{\varepsilon} u \cdot \nabla_{\Gamma} \varphi \, d\mathcal{H}^{n-1}.$$
(2.5.12)

Then there exists a constant c > 0 independent of u, φ , and ε such that

$$|I_{\varepsilon}^{1}(u,\varphi)| \le c\varepsilon^{3/2} \|u\|_{H^{1}(\Omega_{\varepsilon})} \|\nabla_{\Gamma}\varphi\|_{L^{2}(\Gamma)}.$$
(2.5.13)

Remark 2.5.7. The bilinear form $I_{\varepsilon}^{1}(u,\varphi)$ is well-defined for $u \in C^{\infty}(\Omega_{\varepsilon}) \cap H^{1}(\Omega_{\varepsilon})$ and $\varphi \in H^{1}(\Gamma)$, since $\overline{\varphi} \in H^{1}(\Omega_{\varepsilon})$ by Lemma 2.5.5 and $M_{\varepsilon}u$ is smooth on Γ and thus in $H^{1}(\Gamma)$ by the compactness of Γ . We will observe later that $I_{\varepsilon}^{1}(u,\varphi)$ is well-defined and (2.5.13) holds for all $u \in H^{1}(\Omega_{\varepsilon})$ and $\varphi \in H^{1}(\Gamma)$, see Remark 2.5.9.

Proof of Lemma 2.5.6. By (2.5.1) we have $I^1_{\varepsilon}(u, \varphi) = \int_{\Gamma} I(y) \, d\mathcal{H}^{n-1}(y)$, where

$$I(y) := \int_{\varepsilon g_0(y)}^{\varepsilon g_1(y)} (\nabla u)^{\sharp}(y,\rho) \cdot (\nabla \overline{\varphi})^{\sharp}(y,\rho) J(y,\rho) \, d\rho - \varepsilon g(y) \nabla_{\Gamma} M_{\varepsilon} u(y) \cdot \nabla_{\Gamma} \varphi(y).$$

Here we used the notation (2.5.6). Suppose that there is a constant c > 0 independent of ε such that

$$|I(y)| \le c\varepsilon |\nabla_{\Gamma}\varphi(y)| \int_{\varepsilon g_0(y)}^{\varepsilon g_1(y)} (|u^{\sharp}(y,\rho)| + |(\nabla u)^{\sharp}(y,\rho)|) \, d\rho \tag{2.5.14}$$

for all $y \in \Gamma$. Then, by (2.5.14), Hölder's inequality, and (2.5.4) we have

$$\begin{split} |I_{\varepsilon}^{1}(u,\varphi)| &\leq c\varepsilon \int_{\Gamma} |\nabla_{\Gamma}\varphi| \int_{\varepsilon g_{0}}^{\varepsilon g_{1}} (|u^{\sharp}| + |(\nabla u)^{\sharp}|) \, d\rho \, d\mathcal{H}^{n-1} \\ &\leq c\varepsilon \Big(\int_{\Gamma} |\nabla_{\Gamma}\varphi|^{2} \, d\mathcal{H}^{n-1} \Big)^{1/2} \Big\{ \int_{\Gamma} \Big(\int_{\varepsilon g_{0}}^{\varepsilon g_{1}} (|u^{\sharp}| + |(\nabla u)^{\sharp}|) \, d\rho \Big)^{2} d\mathcal{H}^{n-1} \Big\}^{1/2} \\ &\leq c\varepsilon \|\nabla_{\Gamma}\varphi\|_{L^{2}(\Gamma)} \Big(\int_{\Gamma} \varepsilon g \int_{\varepsilon g_{0}}^{\varepsilon g_{1}} (|u^{\sharp}|^{2} + |(\nabla u)^{\sharp}|^{2}) J \, d\rho \, d\mathcal{H}^{n-1} \Big)^{1/2} \\ &\leq c\varepsilon^{3/2} \|\nabla_{\Gamma}\varphi\|_{L^{2}(\Gamma)} \|u\|_{H^{1}(\Omega_{\varepsilon})}. \end{split}$$

Hence (2.5.13) holds. The inequality (2.5.14) is proved by direct calculations under the local coordinate system we took in the proof of Lemma 2.5.5. We give a complete proof in Appendix 2.C.

Lemma 2.5.6 gives an estimate for the $L^2(\Gamma)$ -norm of $\nabla_{\Gamma} M_{\varepsilon} u$ for $u \in H^1(\Omega_{\varepsilon})$.

Lemma 2.5.8. If $u \in H^1(\Omega_{\varepsilon})$, then $M_{\varepsilon}u \in H^1(\Gamma)$ and

$$\|\nabla_{\Gamma} M_{\varepsilon} u\|_{L^{2}(\Gamma)} \leq c \varepsilon^{-1/2} \|u\|_{H^{1}(\Omega_{\varepsilon})}$$

$$(2.5.15)$$

with a constant c > 0 independent of ε .

Proof. First, we show (2.5.15) for all $u \in C^{\infty}(\Omega_{\varepsilon}) \cap H^{1}(\Omega_{\varepsilon})$. For such u, its average $M_{\varepsilon}u$ is smooth on Γ and thus in $H^{1}(\Gamma)$ by the compactness of Γ . We substitute $M_{\varepsilon}u$ for φ in (2.5.12), (2.5.13) to get

$$\int_{\Gamma} g |\nabla_{\Gamma} M_{\varepsilon} u|^2 d\mathcal{H}^{n-1} = \varepsilon^{-1} \bigg(\int_{\Omega_{\varepsilon}} \nabla u \cdot \nabla \overline{M_{\varepsilon} u} \, dx - I_{\varepsilon}^1(u, M_{\varepsilon} u) \bigg), |I_{\varepsilon}^1(u, M_{\varepsilon} u)| \le c \varepsilon^{3/2} \|u\|_{H^1(\Omega_{\varepsilon})} \|\nabla_{\Gamma} M_{\varepsilon} u\|_{L^2(\Gamma)}.$$

Hence, by (2.2.1), Hölder's inequality, and (2.5.10) we obtain

$$\begin{aligned} \|\nabla_{\Gamma} M_{\varepsilon} u\|_{L^{2}(\Gamma)}^{2} &\leq c \int_{\Gamma} g |\nabla_{\Gamma} M_{\varepsilon} u|^{2} \, d\mathcal{H}^{n-1} \\ &\leq c \varepsilon^{-1} \left(\|\nabla u\|_{L^{2}(\Omega_{\varepsilon})} \left\| \nabla \overline{M_{\varepsilon} u} \right\|_{L^{2}(\Omega_{\varepsilon})} + |I_{\varepsilon}^{1}(u, M_{\varepsilon} u)| \right) \\ &\leq c \varepsilon^{-1} (\varepsilon^{1/2} + \varepsilon^{3/2}) \|u\|_{H^{1}(\Omega_{\varepsilon})} \|\nabla_{\Gamma} M_{\varepsilon} u\|_{L^{2}(\Gamma)} \\ &\leq c \varepsilon^{-1/2} \|u\|_{H^{1}(\Omega_{\varepsilon})} \|\nabla_{\Gamma} M_{\varepsilon} u\|_{L^{2}(\Gamma)} \end{aligned}$$

and thus (2.5.15) follows when $u \in C^{\infty}(\Omega_{\varepsilon}) \cap H^{1}(\Omega_{\varepsilon})$. Since Ω_{ε} is bounded, $C^{\infty}(\Omega_{\varepsilon}) \cap H^{1}(\Omega_{\varepsilon})$ is dense in $H^{1}(\Omega_{\varepsilon})$, see [10, Section 5.3.2] for the proof. Hence a density argument together with Lemma 2.5.3 yields that $M_{\varepsilon}u \in H^{1}(\Gamma)$ and (2.5.15) holds for all $u \in H^{1}(\Omega_{\varepsilon})$.

Remark 2.5.9. By Lemma 2.5.5 and Lemma 2.5.8, the bilinear form $I_{\varepsilon}^{1}(u, \varphi)$ given by (2.5.12) is well-defined for all $u \in H^{1}(\Omega_{\varepsilon})$ and $\varphi \in H^{1}(\Gamma)$. Moreover, since $C^{\infty}(\Omega_{\varepsilon}) \cap H^{1}(\Omega_{\varepsilon})$ is dense in $H^{1}(\Omega_{\varepsilon})$, a density argument implies that (2.5.13) also holds for all $u \in H^{1}(\Omega_{\varepsilon})$ and $\varphi \in H^{1}(\Gamma)$.

2.5.3 Material derivative of the average operator

Now let us return to the evolving surface $\Gamma(t)$. Recall the function spaces $L^2_{H^1(\varepsilon)}$ and H_T given by (2.3.1) and (2.4.1), respectively. By Lemma 2.5.3 and Lemma 2.5.8 we immediately get the next lemma.

Lemma 2.5.10. If $u \in L^2_{H^1(\varepsilon)}$, then $M_{\varepsilon}u \in H_T$ and

$$\|M_{\varepsilon}u\|_{H_T} \le c\varepsilon^{-1/2} \|u\|_{L^2_{H^1(\varepsilon)}}$$

with a constant c > 0 independent of ε .

Lemma 2.5.10 enables us to consider the weak material derivative of $M_{\varepsilon}u \in H_T$ for $u \in L^2_{H^1(\varepsilon)}$. Our goal in this subsection is to give a relation between the weak time derivative of u and the weak material derivative of $M_{\varepsilon}u$. To this end, we show an auxiliary statement about the material derivative of a function on S_T .

Lemma 2.5.11. Let $\varphi \in C^1(S_T)$ and $\overline{\varphi}$ be its constant extension in the normal direction of $\Gamma(t)$. Then

$$\partial^{\bullet}\varphi(p(x,t),t) = \partial_t\overline{\varphi}(x,t) + \{V_{\Gamma}(p(x,t),t) + a(x,t)\} \cdot \nabla\overline{\varphi}(x,t)$$
(2.5.16)

holds for all $(x,t) \in N_T$ with a vector field $a: N_T \to \mathbb{R}^n$ given by

$$a(x,t) := d(x,t) \{ \partial_t \nu(p(x,t),t) + \nabla \nu(p(x,t),t) V_{\Gamma}(p(x,t),t) \}.$$
(2.5.17)

Here $\nabla \nu := (\partial \nu_i / \partial x_j)_{i,j}$ is the gradient matrix of ν .

Proof. For $X \in N(0)$ and $t \in (0, T)$ we set

$$\Psi(X,t) := \Phi(p(X,0),t) + d(X,0)\nu(\Phi(p(X,0),t),t),$$

where $\Phi(\cdot, t) \colon \Gamma_0 \to \Gamma(t)$ is the flow map of V_{Γ} (see Section 2.2). By the definition of the constant extension $\overline{\varphi}$ and the formula $p(\Psi(X, t), t) = \Phi(p(X, 0), t)$ we have

$$\overline{\varphi}(\Psi(X,t),t) = \varphi(\Phi(p(X,0),t),t)$$

for all $X \in N(0)$ and $t \in (0, T)$. Differentiating both sides with respect to t and observing that each $x \in N(t)$ is represented as $x = \Psi(X, t)$ with a unique $X \in N(0)$, we get the formula (2.5.16). For detailed calculations, see Appendix 2.C.

Remark 2.5.12. Let $\varphi \in C^1(S_T)$. Since p(y,t) = y and d(y,t) = 0 for all $(y,t) \in S_T$, we have

$$\partial^{\bullet}\varphi = \partial_t \overline{\varphi} + V_{\Gamma} \cdot \nabla \overline{\varphi} = \partial_t \overline{\varphi} + v_{\Gamma}^N \nu \cdot \nabla \overline{\varphi} + V_{\Gamma}^T \cdot \nabla_{\Gamma(t)} \overline{\varphi} \quad \text{on} \quad S_T$$

by Lemma 2.5.11. Here the last equality follows from the fact that V_{Γ}^{T} is tangent to $\Gamma(t)$. Based on this equality, the material derivative operator acting on functions on $\Gamma(t)$ is formally represented as $\partial^{\bullet} = \partial_{t} + V_{\Gamma} \cdot \nabla = \partial_{t} + v_{\Gamma}^{N} \nu \cdot \nabla + V_{\Gamma}^{T} \cdot \nabla_{\Gamma(t)}$.

Using Lemma 2.5.11, we derive an integral formula related to the weak time derivative of a function $u \in L^2_{H^1(\varepsilon)}$ and the weak material derivative of its average $M_{\varepsilon} u \in H_T$.

Lemma 2.5.13. Let $u \in L^2_{H^1(\varepsilon)}$, $\varphi \in C^1_0(S_T)$, and $\overline{\varphi}$ be its constant extension. Then we have

$$\int_{0}^{T} \int_{\Omega_{\varepsilon}(t)} u \,\partial_{t} \overline{\varphi} \,dx \,dt = -\varepsilon \langle \partial^{\bullet} M_{\varepsilon} u, g\varphi \rangle_{T} - \varepsilon \int_{0}^{T} \int_{\Gamma(t)} (\partial^{\bullet} g + g \operatorname{div}_{\Gamma(t)} V_{\Gamma}) (M_{\varepsilon} u) \varphi \,d\mathcal{H}^{n-1} \,dt \\ - \varepsilon \int_{0}^{T} \int_{\Gamma(t)} g(M_{\varepsilon} u) V_{\Gamma}^{T} \cdot \nabla_{\Gamma(t)} \varphi \,d\mathcal{H}^{n-1} \,dt + I_{\varepsilon}^{2}(u, \varphi; T).$$
(2.5.18)

Here $I^2_{\varepsilon}(u,\varphi;T)$ is a residual term that satisfies

$$|I_{\varepsilon}^{2}(u,\varphi;T)| \leq c\varepsilon^{3/2} \int_{0}^{T} \|u(t)\|_{L^{2}(\Omega_{\varepsilon}(t))} \|\nabla_{\Gamma(t)}\varphi(t)\|_{L^{2}(\Gamma(t))} dt \qquad (2.5.19)$$

with a constant c > 0 independent of u, φ , and ε .

Note that the tangential velocity V_{Γ}^{T} appears instead of the total velocity V_{Γ} in the third term of the right-hand side of (2.5.18), see Remark 2.5.15 below.

Proof. By (2.5.16), we have $\overline{\partial^{\bullet}\varphi} = \partial_t \overline{\varphi} + (\overline{V_{\Gamma}} + a) \cdot \nabla \overline{\varphi}$ on N_T , where *a* is the vector field on N_T given by (2.5.17). Hence if we set

$$I_{\varepsilon}^{2}(u,\varphi;T) := -\int_{0}^{T} \int_{\Omega_{\varepsilon}(t)} u\left\{a \cdot \nabla\overline{\varphi} + \overline{V_{\Gamma}} \cdot \left(\nabla\overline{\varphi} - \overline{\nabla_{\Gamma(t)}\varphi}\right)\right\} dx \, dt,$$

then we have

$$\int_0^T \int_{\Omega_{\varepsilon}(t)} u \,\partial_t \overline{\varphi} \,dx \,dt = \int_0^T \int_{\Omega_{\varepsilon}(t)} u \Big(\overline{\partial^{\bullet} \varphi} - \overline{V_{\Gamma}} \cdot \overline{\nabla_{\Gamma(t)} \varphi}\Big) \,dx \,dt + I_{\varepsilon}^2(u,\varphi;T). \tag{2.5.20}$$

Let us compute the first term of the right-hand side of (2.5.20). By the co-area formula (2.5.1) and the definition of the weighted average $M_{\varepsilon}u$ (see (2.5.2)),

$$\int_{\Omega_{\varepsilon}(t)} u(x,t)\overline{\partial^{\bullet}\varphi}(x,t) \, dx = \int_{\Gamma(t)} \int_{\varepsilon g_0(y,t)}^{\varepsilon g_1(y,t)} u(y+\rho\nu(y,t),t)\partial^{\bullet}\varphi(y,t)J(y,t,\rho) \, d\rho \, d\mathcal{H}^{n-1}(y)$$
$$= \varepsilon \int_{\Gamma(t)} g(y,t)M_{\varepsilon}u(y,t)\partial^{\bullet}\varphi(y,t) \, d\mathcal{H}^{n-1}(y)$$

for all $t \in (0, T)$. On the other hand, since the weak material derivative is given by (2.4.6),

$$\begin{aligned} \langle \partial^{\bullet} M_{\varepsilon} u, g\varphi \rangle_{T} &= -\int_{0}^{T} \int_{\Gamma(t)} \{ (M_{\varepsilon} u) \partial^{\bullet} (g\varphi) + (M_{\varepsilon} u) g\varphi \operatorname{div}_{\Gamma(t)} V_{\Gamma} \} d\mathcal{H}^{n-1} dt \\ &= -\int_{0}^{T} \int_{\Gamma(t)} \{ (\partial^{\bullet} g + g \operatorname{div}_{\Gamma(t)} V_{\Gamma}) (M_{\varepsilon} u) \varphi + g (M_{\varepsilon} u) \partial^{\bullet} \varphi \} d\mathcal{H}^{n-1} dt \end{aligned}$$

Thus it follows that

$$\int_{0}^{T} \int_{\Omega_{\varepsilon}(t)} u \,\overline{\partial^{\bullet}\varphi} \, dx \, dt = -\varepsilon \langle \partial^{\bullet} M_{\varepsilon} u, g\varphi \rangle_{T} -\varepsilon \int_{0}^{T} \int_{\Gamma(t)} (\partial^{\bullet} g + g \operatorname{div}_{\Gamma(t)} V_{\Gamma}) (M_{\varepsilon} u) \varphi \, d\mathcal{H}^{n-1} \, dt. \quad (2.5.21)$$

Since $V_{\Gamma} = v_{\Gamma}^{N} \nu + V_{\Gamma}^{T}$ and $\nu \cdot \nabla_{\Gamma(t)} \varphi = 0$, we have $V_{\Gamma} \cdot \nabla_{\Gamma(t)} \varphi = V_{\Gamma}^{T} \cdot \nabla_{\Gamma(t)} \varphi$ on S_{T} . This equality together with the co-area formula (2.5.1) yields

$$\begin{split} \int_{\Omega_{\varepsilon}(t)} u(x,t) \overline{V_{\Gamma}}(x,t) \cdot \overline{\nabla_{\Gamma(t)}\varphi}(x,t) \, dx \\ &= \int_{\Gamma(t)} \int_{\varepsilon g_0(y,t)}^{\varepsilon g_1(y,t)} u(y + \rho \nu(y,t),t) V_{\Gamma}(y,t) \cdot \nabla_{\Gamma(t)}\varphi(y,t) J(y,t,\rho) \, d\rho \, d\mathcal{H}^{n-1}(y) \\ &= \varepsilon \int_{\Gamma(t)} g(y,t) M_{\varepsilon} u(y,t) V_{\Gamma}^T(y,t) \cdot \nabla_{\Gamma(t)}\varphi(y,t) \, d\mathcal{H}^{n-1}(y) \end{split}$$

for all $t \in (0, T)$ and thus

$$\int_0^T \int_{\Omega_{\varepsilon}(t)} u\left(\overline{V_{\Gamma}} \cdot \overline{\nabla_{\Gamma(t)}\varphi}\right) dx \, dt = \varepsilon \int_0^T \int_{\Gamma(t)} g(M_{\varepsilon}u) V_{\Gamma}^T \cdot \nabla_{\Gamma(t)}\varphi \, d\mathcal{H}^{n-1} \, dt.$$
(2.5.22)

Substituting (2.5.21) and (2.5.22) for (2.5.20), we obtain the equality (2.5.18).

$$|a(x,t)| \le c |d(x,t)| \le c\varepsilon \max_{i=1,2} \sup_{(y,\tau)\in \overline{S_T}} |g_i(y,\tau)| \le c\varepsilon$$

for all $(x,t) \in Q_{\varepsilon,T}$. By this inequality, Hölder's inequality, and (2.5.10) we obtain

$$\begin{split} |I_{\varepsilon}^{2}(u,\varphi;T)| &\leq c \int_{0}^{T} \|u(t)\|_{L^{2}(\Omega_{\varepsilon}(t))} \bigg(\varepsilon \|\nabla\overline{\varphi}(t)\|_{L^{2}(\Omega_{\varepsilon}(t))} + \left\|\nabla\overline{\varphi}(t) - \overline{\nabla_{\Gamma(t)}\varphi(t)}\right\|_{L^{2}(\Omega_{\varepsilon}(t))}\bigg) dt \\ &\leq c\varepsilon^{3/2} \int_{0}^{T} \|u(t)\|_{L^{2}(\Omega_{\varepsilon}(t))} \|\nabla_{\Gamma(t)}\varphi(t)\|_{L^{2}(\Gamma(t))} dt. \end{split}$$
Thus (2.5.19) holds.

Thus (2.5.19) holds.

Remark 2.5.14. If $M_{\varepsilon}u$ is in the Hilbert space W_T given by (2.4.7), then the right-hand side of (2.5.18) is well-defined for $\varphi \in H_T$ since $C_0^1(S_T)$ is dense in H_T (see Lemma 2.4.2). In particular, we can substitute $M_{\varepsilon}u$ for φ in the right-hand side of (2.5.18). This fact is essential for derivation of the energy estimate for the weighted average of a variational solution to (H_{ε}) (see Lemma 2.6.4). If we replace M_{ε} in (2.5.18) by a usual unweighted average operator

$$\mathcal{M}_{\varepsilon}u(y,t) := \frac{1}{\varepsilon g(y,t)} \int_{\varepsilon g_0(y,t)}^{\varepsilon g_1(y,t)} u(y + \rho\nu(y,t),t) \, d\rho,$$

then the estimate for the residual term becomes

$$|I_{\varepsilon}^{2}(u,\varphi;T)| \leq c\varepsilon^{3/2} \int_{0}^{T} \|u(t)\|_{L^{2}(\Omega_{\varepsilon}(t))} \Big(\|\nabla_{\Gamma(t)}\varphi(t)\|_{L^{2}(\Gamma(t))} + \|\partial^{\bullet}\varphi(t)\|_{L^{2}(\Gamma(t))} \Big) dt.$$

Because of the term $\|\partial^{\bullet}\varphi(t)\|_{L^{2}(\Gamma(t))}$ in the above inequality, the right-hand side of (2.5.18) with M_{ε} replaced by $\mathcal{M}_{\varepsilon}$ is not well-defined for $\varphi \in H_T$. Therefore we can not derive the energy estimate for the unweighted average of a variational solution to (H_{ε}) .

Remark 2.5.15. Let $\Gamma \subset \mathbb{R}^n$ be a closed, connected, and oriented smooth hypersurface. Then, since $\partial \Gamma = \emptyset$, the integral formula (see [25, Section 7.2])

$$\int_{\Gamma} \operatorname{div}_{\Gamma} V \, d\mathcal{H}^{n-1} = -\int_{\Gamma} (V \cdot \nu) H \, d\mathcal{H}^{n-1}$$

holds for smooth vector fields $V: \Gamma \to \mathbb{R}^n$. Here ν is the unit outward normal vector of Γ and $H := -\operatorname{div}_{\Gamma} \nu$ is the mean curvature of Γ . This formula yields the equality

$$\int_{\Gamma} V \cdot \nabla_{\Gamma} \varphi \, d\mathcal{H}^{n-1} = -\int_{\Gamma} \{ \operatorname{div}_{\Gamma} V + (V \cdot \nu) H \} \varphi \, d\mathcal{H}^{n-1}$$

for smooth functions φ on Γ . In this equality we decompose $V = v^N \nu + V^T$ into the normal component $v^N := V \cdot \nu$ and the tangential component $V^T := V - (V \cdot \nu)\nu$. Then, since

$$\nu \cdot \nabla_{\Gamma} \varphi = 0$$
, $\operatorname{div}_{\Gamma}(v^{N}\nu) = \nabla_{\Gamma} v^{N} \cdot \nu + v^{N} \operatorname{div}_{\Gamma} \nu = 0 + v^{N} \cdot (-H) = -(V \cdot \nu)H$,

we obtain a usual integration by parts formula

$$\int_{\Gamma} V^T \cdot \nabla_{\Gamma} \varphi \, d\mathcal{H}^{n-1} = -\int_{\Gamma} \varphi \operatorname{div}_{\Gamma} V^T \, d\mathcal{H}^{n-1}, \qquad (2.5.23)$$

which we will use to recover a limit equation on $\Gamma(t)$ from its variational formulation. This is the reason the tangential velocity V_{Γ}^{T} appears in (2.5.18) instead of the total velocity V_{Γ} of $\Gamma(t)$.

2.6 Convergence and characterization of the limit

2.6.1 Variational formulations of the average of solutions to the heat equation

Let us return to the initial-boundary value problem (H_{ε}) of the heat equation. By Theorem 2.3.4, for every $u_0^{\varepsilon} \in L^2(\Omega_{\varepsilon}(0))$ there exists a unique variational solution $u^{\varepsilon} \in L^2_{H^1(\varepsilon)}$ to (H_{ε}) .

Let M_{ε} be the weighted average operator defined in Definition 2.5.1. Our goal in this subsection is to derive a variational formulation of $M_{\varepsilon}u^{\varepsilon}$.

Lemma 2.6.1. Let $u_0^{\varepsilon} \in L^2(\Omega_{\varepsilon}(0))$ and $u^{\varepsilon} \in L^2_{H^1(\varepsilon)}$ be the unique variational solution to (H_{ε}) given by Theorem 2.3.4. Then $M_{\varepsilon}u^{\varepsilon} \in H_T$ and it satisfies

$$\langle \partial^{\bullet} M_{\varepsilon} u^{\varepsilon}, g\varphi \rangle_{T} + \int_{0}^{T} \int_{\Gamma(t)} (\partial^{\bullet} g + g \operatorname{div}_{\Gamma(t)} V_{\Gamma}) (M_{\varepsilon} u^{\varepsilon}) \varphi \, d\mathcal{H}^{n-1} \, dt + \int_{0}^{T} \int_{\Gamma(t)} g \{ \nabla_{\Gamma(t)} M_{\varepsilon} u^{\varepsilon} + (M_{\varepsilon} u^{\varepsilon}) V_{\Gamma}^{T} \} \cdot \nabla_{\Gamma(t)} \varphi \, d\mathcal{H}^{n-1} \, dt = I_{\varepsilon} (u^{\varepsilon}, \varphi; T)$$
(2.6.1)

for all $\varphi \in C_0^1(S_T)$. Here $I_{\varepsilon}(u^{\varepsilon}, \varphi; T)$ is a residual term that satisfies

$$|I_{\varepsilon}(u^{\varepsilon},\varphi;T)| \le c\varepsilon^{1/2} \int_0^T \|u^{\varepsilon}(t)\|_{H^1(\Omega_{\varepsilon}(t))} \|\nabla_{\Gamma(t)}\varphi(t)\|_{L^2(\Gamma(t))} dt$$
(2.6.2)

with a constant c > 0 independent of u^{ε} , φ , and ε .

Proof. Since $u^{\varepsilon} \in L^2_{H^1(\varepsilon)}$, we have $M_{\varepsilon}u^{\varepsilon} \in H_T$ by Lemma 2.5.10. For each $\varphi \in C^1_0(S_T)$, its constant extension $\overline{\varphi}$ is in $C^1(\overline{Q_{\varepsilon,T}})$ and satisfies $\overline{\varphi}(0) = 0$ in $\Omega_{\varepsilon}(0)$ and $\overline{\varphi}(T) = 0$ in $\Omega_{\varepsilon}(T)$. Thus, by substituting $\overline{\varphi}$ for w in the variational formulation (2.3.2) we obtain

$$\int_0^T \int_{\Omega_{\varepsilon}(t)} (-u^{\varepsilon} \partial_t \overline{\varphi} + \nabla u^{\varepsilon} \cdot \nabla \overline{\varphi}) \, dx \, dt = 0.$$
(2.6.3)

Moreover, from Lemma 2.5.6 and Lemma 2.5.13 we have

$$\int_0^T \int_{\Omega_{\varepsilon}(t)} \nabla u^{\varepsilon} \cdot \nabla \overline{\varphi} \, dx \, dt = \varepsilon \int_0^T \int_{\Gamma(t)} g \nabla_{\Gamma(t)} M_{\varepsilon} u^{\varepsilon} \cdot \nabla_{\Gamma(t)} \varphi \, d\mathcal{H}^{n-1} \, dt + I_{\varepsilon}^1(u^{\varepsilon}, \varphi; T) \quad (2.6.4)$$

and

$$\int_{0}^{T} \int_{\Omega_{\varepsilon}(t)} u^{\varepsilon} \partial_{t} \overline{\varphi} \, dx \, dt = -\varepsilon \langle \partial^{\bullet} M_{\varepsilon} u^{\varepsilon}, g\varphi \rangle_{T} - \varepsilon \int_{0}^{T} \int_{\Gamma(t)} (\partial^{\bullet} g + g \operatorname{div}_{\Gamma(t)} V_{\Gamma}) (M_{\varepsilon} u^{\varepsilon}) \varphi \, d\mathcal{H}^{n-1} \, dt \\ - \varepsilon \int_{0}^{T} \int_{\Gamma(t)} g(M_{\varepsilon} u^{\varepsilon}) V_{\Gamma}^{T} \cdot \nabla_{\Gamma(t)} \varphi \, d\mathcal{H}^{n-1} \, dt + I_{\varepsilon}^{2} (u^{\varepsilon}, \varphi; T), \quad (2.6.5)$$

where $I_{\varepsilon}^{1}(u^{\varepsilon},\varphi;T)$ and $I_{\varepsilon}^{2}(u^{\varepsilon},\varphi;T)$ satisfy

$$|I_{\varepsilon}^{k}(u^{\varepsilon},\varphi;T)| \le c\varepsilon^{3/2} \int_{0}^{T} \|u^{\varepsilon}(t)\|_{H^{1}(\Omega_{\varepsilon}(t))} \|\nabla_{\Gamma(t)}\varphi(t)\|_{L^{2}(\Gamma(t))} dt, \quad k = 1, 2,$$
(2.6.6)

with some constant c > 0 independent of ε . Hence, by substituting (2.6.4) and (2.6.5) for (2.6.3) and dividing both sides by ε , we obtain (2.6.1) with the residual term

$$I_{\varepsilon}(u^{\varepsilon},\varphi;T) := \varepsilon^{-1} \big\{ I_{\varepsilon}^{2}(u^{\varepsilon},\varphi;T) - I_{\varepsilon}^{1}(u^{\varepsilon},\varphi;T) \big\},\$$

which satisfies (2.6.2) because $I_{\varepsilon}^1(u^{\varepsilon}, \varphi; T)$ and $I_{\varepsilon}^2(u^{\varepsilon}, \varphi; T)$ satisfy (2.6.6).

2.6.2 Estimates for the average $M_{\varepsilon}u^{\varepsilon}$ in the space W_T

In this subsection, we estimate $M_{\varepsilon}u^{\varepsilon}$ in the Hilbert space W_T given by (2.4.7).

Lemma 2.6.2. Let $u_0^{\varepsilon} \in L^2(\Omega_{\varepsilon}(0))$ and $u^{\varepsilon} \in L^2_{H^1(\varepsilon)}$ be the unique variational solution to (H_{ε}) given by Theorem 2.3.4. Then $M_{\varepsilon}u^{\varepsilon} \in W_T$ and there exists a constant c > 0 independent of u^{ε} and ε such that

$$\|\partial^{\bullet} M_{\varepsilon} u^{\varepsilon}\|_{H_{T}} \leq c \Big(\|M_{\varepsilon} u^{\varepsilon}\|_{H_{T}} + \varepsilon^{1/2} \|u^{\varepsilon}\|_{L^{2}_{H^{1}(\varepsilon)}}\Big).$$

$$(2.6.7)$$

Proof. Let φ be an arbitrary function in $C_0^1(S_T)$. By substituting $g^{-1}\varphi \in C_0^1(S_T)$ for φ in (2.6.1), we obtain $\langle \partial^{\bullet} M_{\varepsilon} u^{\varepsilon}, \varphi \rangle_T = I(u^{\varepsilon}, \varphi) + I_{\varepsilon}(u^{\varepsilon}, g^{-1}\varphi; T)$, where

$$\begin{split} I(u^{\varepsilon},\varphi) &:= \int_0^T \int_{\Gamma(t)} \{g^{-1}(V_{\Gamma}^T \cdot \nabla_{\Gamma(t)}g - \partial^{\bullet}g) - \operatorname{div}_{\Gamma(t)}V_{\Gamma}\}(M_{\varepsilon}u^{\varepsilon})\varphi \, d\mathcal{H}^{n-1} \, dt \\ &- \int_0^T \int_{\Gamma(t)} \{\nabla_{\Gamma(t)}M_{\varepsilon}u^{\varepsilon} + (M_{\varepsilon}u^{\varepsilon})V_{\Gamma}^T\} \cdot \nabla_{\Gamma(t)}\varphi \, d\mathcal{H}^{n-1} \, dt \\ &+ \int_0^T \int_{\Gamma(t)} g^{-1}(\nabla_{\Gamma(t)}g \cdot \nabla_{\Gamma(t)}M_{\varepsilon}u^{\varepsilon})\varphi \, d\mathcal{H}^{n-1} \, dt. \end{split}$$

Since g and V_{Γ} are smooth on $\overline{S_T}$, they are bounded on S_T along with their derivatives. Moreover, g^{-1} and V_{Γ}^T are bounded on S_T . Thus we have $|I(u^{\varepsilon}, \varphi)| \leq c ||M_{\varepsilon}u^{\varepsilon}||_{H_T} ||\varphi||_{H_T}$ with a constant c > 0 independent of u^{ε}, φ , and ε . Also, by (2.6.2),

$$\begin{aligned} |I_{\varepsilon}(u^{\varepsilon}, g^{-1}\varphi; T)| &\leq c\varepsilon^{1/2} \int_{0}^{T} \|u^{\varepsilon}(t)\|_{H^{1}(\Omega_{\varepsilon}(t))} \|\nabla_{\Gamma(t)}(g^{-1}\varphi)(t)\|_{L^{2}(\Gamma(t))} dt \\ &\leq c\varepsilon^{1/2} \int_{0}^{T} \|u^{\varepsilon}(t)\|_{H^{1}(\Omega_{\varepsilon}(t))} \Big(\|\varphi(t)\|_{L^{2}(\Gamma(t))} + \|\nabla_{\Gamma(t)}\varphi(t)\|_{L^{2}(\Gamma(t))} \Big) dt \\ &\leq c\varepsilon^{1/2} \|u^{\varepsilon}\|_{L^{2}_{H^{1}(\varepsilon)}} \|\varphi\|_{H_{T}} \end{aligned}$$

with some c > 0 independent of u^{ε} , φ , and ε . Hence we obtain

$$|\langle \partial^{\bullet} M_{\varepsilon} u^{\varepsilon}, \varphi \rangle_{T}| \leq |I(u^{\varepsilon}, \varphi)| + |I_{\varepsilon}(u^{\varepsilon}, g^{-1}\varphi; T)| \leq c \Big(\|M_{\varepsilon} u^{\varepsilon}\|_{H_{T}} + \varepsilon^{1/2} \|u^{\varepsilon}\|_{L^{2}_{H^{1}(\varepsilon)}} \Big) \|\varphi\|_{H_{T}}$$

for all $\varphi \in C_0^1(S_T)$, which implies $M_{\varepsilon} u^{\varepsilon} \in W_T$ and the inequality (2.6.7).

Remark 2.6.3. Since $M_{\varepsilon}u^{\varepsilon} \in W_T$ and $C_0^1(S_T)$ is dense in H_T (see Lemma 2.4.2), the equality (2.6.1) also holds for all $\varphi \in H_T$. Moreover, since W_{T_1} is continuously embedded into W_{T_2} when $T_1 > T_2$, we have $M_{\varepsilon}u^{\varepsilon} \in W_{\tau}$ for each $\tau \in [0, T]$. Hence (2.6.1) and (2.6.2) with T replaced by each $\tau \in [0, T]$ are also valid for all $\varphi \in H_{\tau}$.

Lemma 2.6.4. Let u_0^{ε} and u^{ε} be as in Lemma 2.6.2. Then there exists a constant c > 0 independent of u_0^{ε} , u^{ε} , and ε such that the energy estimate

$$\begin{split} \|M_{\varepsilon}u^{\varepsilon}(\tau)\|_{L^{2}(\Gamma(\tau))}^{2} + \int_{0}^{\tau} \|\nabla_{\Gamma(t)}M_{\varepsilon}u^{\varepsilon}(t)\|_{L^{2}(\Gamma(t))}^{2} dt \\ & \leq c \Big(\|M_{\varepsilon}u_{0}^{\varepsilon}\|_{L^{2}(\Gamma_{0})}^{2} + \varepsilon \|u_{0}^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon}(0))}^{2}\Big) \quad (2.6.8) \end{split}$$

holds for all $\tau \in [0, T]$.

Proof. As we mentioned in Remark 2.6.3, the equality (2.6.1) holds with T replaced by each $\tau \in [0, T]$. Hence, by substituting $g^{-1}M_{\varepsilon}u^{\varepsilon} \in H_{\tau}$ for φ in (2.6.1) with T replaced by τ , we obtain

$$\begin{split} \langle \partial^{\bullet} M_{\varepsilon} u^{\varepsilon}, M_{\varepsilon} u^{\varepsilon} \rangle_{\tau} &+ \int_{0}^{\tau} \| \nabla_{\Gamma(t)} M_{\varepsilon} u^{\varepsilon}(t) \|_{L^{2}(\Gamma(t))}^{2} dt \\ &+ \int_{0}^{\tau} \int_{\Gamma(t)} \{ g^{-1} (\partial^{\bullet} g - V_{\Gamma}^{T} \cdot \nabla_{\Gamma(t)} g) + \operatorname{div}_{\Gamma(t)} V_{\Gamma} \} |M_{\varepsilon} u^{\varepsilon}|^{2} d\mathcal{H}^{n-1} dt \\ &+ \int_{0}^{\tau} \int_{\Gamma(t)} M_{\varepsilon} u^{\varepsilon} (V_{\Gamma}^{T} - g^{-1} \nabla_{\Gamma(t)} g) \cdot \nabla_{\Gamma(t)} M_{\varepsilon} u^{\varepsilon} d\mathcal{H}^{n-1} dt = I_{\varepsilon} (u^{\varepsilon}, g^{-1} M_{\varepsilon} u^{\varepsilon}; \tau). \end{split}$$

Moreover, from (2.4.8) with T replaced by τ ,

$$\begin{split} \langle \partial^{\bullet} M_{\varepsilon} u^{\varepsilon}, M_{\varepsilon} u^{\varepsilon} \rangle_{\tau} &= -\frac{1}{2} \int_{0}^{\tau} \int_{\Gamma(t)} |M_{\varepsilon} u^{\varepsilon}|^{2} \operatorname{div}_{\Gamma(t)} V_{\Gamma} d\mathcal{H}^{n-1} dt \\ &+ \frac{1}{2} \|M_{\varepsilon} u^{\varepsilon}(\tau)\|_{L^{2}(\Gamma(\tau))}^{2} - \frac{1}{2} \|M_{\varepsilon} u^{\varepsilon}(0)\|_{L^{2}(\Gamma_{0})}^{2}. \end{split}$$

Applying this equality and the relation $u^{\varepsilon}(0) = u_0^{\varepsilon}$ in $L^2(\Omega_{\varepsilon}(0))$ (see Theorem 2.3.4) to the above equality, we have

$$\frac{1}{2} \|M_{\varepsilon}u^{\varepsilon}(\tau)\|_{L^{2}(\Gamma(\tau))}^{2} + \int_{0}^{\tau} \|\nabla_{\Gamma(t)}M_{\varepsilon}u^{\varepsilon}(t)\|_{L^{2}(\Gamma(t))}^{2} dt$$

$$= \frac{1}{2} \|M_{\varepsilon}u_{0}^{\varepsilon}\|_{L^{2}(\Gamma_{0})}^{2} + I_{1}(\tau) + I_{2}(\tau) + I_{\varepsilon}(u^{\varepsilon}, g^{-1}M_{\varepsilon}u^{\varepsilon}; \tau), \quad (2.6.9)$$

where

$$I_{1}(\tau) := -\frac{1}{2} \int_{0}^{\tau} \int_{\Gamma(t)} \{ 2g^{-1} (\partial^{\bullet}g - V_{\Gamma}^{T} \cdot \nabla_{\Gamma(t)}g) + \operatorname{div}_{\Gamma(t)}V_{\Gamma} \} |M_{\varepsilon}u^{\varepsilon}|^{2} d\mathcal{H}^{n-1} dt,$$

$$I_{2}(\tau) := -\int_{0}^{\tau} \int_{\Gamma(t)} M_{\varepsilon}u^{\varepsilon} (V_{\Gamma}^{T} - g^{-1}\nabla_{\Gamma(t)}g) \cdot \nabla_{\Gamma(t)}M_{\varepsilon}u^{\varepsilon} d\mathcal{H}^{n-1} dt.$$

Since g and V_{Γ} are smooth on $\overline{S_T}$, they are bounded on S_T along with their derivatives. Also, g^{-1} and V_{Γ}^T are bounded on S_T . Thus it follows that

$$|I_{1}(\tau)| \leq c \int_{0}^{\tau} \|M_{\varepsilon}u^{\varepsilon}(t)\|_{L^{2}(\Gamma(t))}^{2} dt,$$

$$|I_{2}(\tau)| \leq c \int_{0}^{\tau} \|M_{\varepsilon}u^{\varepsilon}(t)\|_{L^{2}(\Gamma(t))} \|\nabla_{\Gamma(t)}M_{\varepsilon}u^{\varepsilon}(t)\|_{L^{2}(\Gamma(t))} dt.$$
(2.6.10)

On the other hand, the inequality (2.6.2) with T replaced by τ yields

$$|I_{\varepsilon}(u^{\varepsilon}, g^{-1}M_{\varepsilon}u^{\varepsilon}; \tau)| \leq c\varepsilon^{1/2} \int_{0}^{\tau} \|u^{\varepsilon}(t)\|_{H^{1}(\Omega_{\varepsilon}(t))} \|\nabla_{\Gamma(t)}(g^{-1}M_{\varepsilon}u^{\varepsilon})(t)\|_{L^{2}(\Gamma(t))} dt$$
$$\leq c\varepsilon^{1/2} \int_{0}^{\tau} \|u^{\varepsilon}(t)\|_{H^{1}(\Omega_{\varepsilon}(t))} \Big(\|M_{\varepsilon}u^{\varepsilon}(t)\|_{L^{2}(\Gamma(t))} + \|\nabla_{\Gamma(t)}M_{\varepsilon}u^{\varepsilon}(t)\|_{L^{2}(\Gamma(t))} \Big) dt. \quad (2.6.11)$$

Thus, by applying (2.6.10) and (2.6.11) to (2.6.9), we obtain

$$\begin{split} \frac{1}{2} \|M_{\varepsilon}u^{\varepsilon}(\tau)\|_{L^{2}(\Gamma(\tau))}^{2} + \int_{0}^{\tau} \|\nabla_{\Gamma(t)}M_{\varepsilon}u^{\varepsilon}(t)\|_{L^{2}(\Gamma(t))}^{2} dt \\ &\leq \frac{1}{2} \|M_{\varepsilon}u_{0}^{\varepsilon}\|_{L^{2}(\Gamma_{0})}^{2} + \frac{1}{2}\int_{0}^{\tau} \|\nabla_{\Gamma(t)}M_{\varepsilon}u^{\varepsilon}(t)\|_{L^{2}(\Gamma(t))}^{2} dt \\ &\quad + c\int_{0}^{\tau} \left(\|M_{\varepsilon}u^{\varepsilon}(t)\|_{L^{2}(\Gamma(t))}^{2} + \varepsilon\|u^{\varepsilon}(t)\|_{H^{1}(\Omega_{\varepsilon}(t))}^{2}\right) dt. \end{split}$$

We multiply both sides by two and subtract $\int_0^\tau \|\nabla_{\Gamma(t)} M_\varepsilon u^\varepsilon(t)\|_{L^2(\Gamma(t))}^2 dt$ to get

$$\begin{split} \|M_{\varepsilon}u^{\varepsilon}(\tau)\|_{L^{2}(\Gamma(\tau))}^{2} + \int_{0}^{\tau} \|\nabla_{\Gamma(t)}M_{\varepsilon}u^{\varepsilon}(t)\|_{L^{2}(\Gamma(t))}^{2} dt \\ & \leq \|M_{\varepsilon}u_{0}^{\varepsilon}\|_{L^{2}(\Gamma_{0})}^{2} + c\int_{0}^{\tau} \left(\|M_{\varepsilon}u^{\varepsilon}(t)\|_{L^{2}(\Gamma(t))}^{2} + \varepsilon\|u^{\varepsilon}(t)\|_{H^{1}(\Omega_{\varepsilon}(t))}^{2}\right) dt. \end{split}$$

Hence Gronwall's inequality implies

$$\|M_{\varepsilon}u^{\varepsilon}(\tau)\|_{L^{2}(\Gamma(\tau))}^{2} + \int_{0}^{\tau} \|\nabla_{\Gamma(t)}M_{\varepsilon}u^{\varepsilon}(t)\|_{L^{2}(\Gamma(t))}^{2} dt \le c \Big(\|M_{\varepsilon}u_{0}^{\varepsilon}\|_{L^{2}(\Gamma_{0})}^{2} + \varepsilon \|u^{\varepsilon}\|_{L^{2}_{H^{1}(\varepsilon)}}^{2}\Big)$$

for all $\tau \in [0,T]$, and we obtain (2.6.8) by applying (2.3.15) to the second term of the right-hand side of the above inequality.

Lemma 2.6.5. Let u_0^{ε} and u^{ε} be as in Lemma 2.6.2. Then there exists a constant c > 0 independent of u_0^{ε} , u^{ε} , and ε such that

$$\|M_{\varepsilon}u^{\varepsilon}\|_{W_{T}} \leq c \Big(\|M_{\varepsilon}u_{0}^{\varepsilon}\|_{L^{2}(\Gamma_{0})} + \varepsilon^{1/2}\|u_{0}^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon}(0))}\Big).$$

$$(2.6.12)$$

Proof. It follows from (2.6.8) that

$$\|M_{\varepsilon}u^{\varepsilon}\|_{H_{T}} \leq c \Big(\|M_{\varepsilon}u_{0}^{\varepsilon}\|_{L^{2}(\Gamma_{0})} + \varepsilon^{1/2}\|u_{0}^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon}(0))}\Big).$$

Moreover, by applying this inequality and (2.3.15) to (2.6.7) we have

$$\|\partial^{\bullet} M_{\varepsilon} u^{\varepsilon}\|_{H_{T}'} \leq c \Big(\|M_{\varepsilon} u^{\varepsilon}\|_{H_{T}} + \varepsilon^{1/2} \|u^{\varepsilon}\|_{L^{2}_{H^{1}(\varepsilon)}}\Big) \leq c \Big(\|M_{\varepsilon} u^{\varepsilon}_{0}\|_{L^{2}(\Gamma_{0})} + \varepsilon^{1/2} \|u^{\varepsilon}_{0}\|_{L^{2}(\Omega_{\varepsilon}(0))}\Big).$$

Thus we obtain (2.6.12).

2.6.3 Limit equation on evolving surfaces and weak convergence of $M_{\varepsilon}u^{\varepsilon}$

Assume that $I_{\varepsilon}(u^{\varepsilon}, \varphi; T) = 0$ holds for all $\varphi \in C_0^1(S_T)$ and $v = M_{\varepsilon}u^{\varepsilon}$ is independent of ε in the variational formulation (2.6.1). Then v satisfies

$$\begin{split} \langle \partial^{\bullet} v, g\varphi \rangle_{T} &+ \int_{0}^{T} \int_{\Gamma(t)} (\partial^{\bullet} g + g \operatorname{div}_{\Gamma(t)} V_{\Gamma}) v \,\varphi \, d\mathcal{H}^{n-1} \, dt \\ &+ \int_{0}^{T} \int_{\Gamma(t)} g(\nabla_{\Gamma(t)} v + v V_{\Gamma}^{T}) \cdot \nabla_{\Gamma(t)} \varphi \, d\mathcal{H}^{n-1} \, dt = 0 \quad (2.6.13) \end{split}$$

for all $\varphi \in C_0^1(S_T)$. In addition we assume that v is sufficiently smooth. Since vector fields gvV_{Γ}^T and $g\nabla_{\Gamma(t)}v$ are tangent to $\Gamma(t)$ for each $t \in [0, T]$, we can apply the integration by parts formula (2.5.23) to obtain

$$\langle \partial^{\bullet} v, g\varphi \rangle_{T} + \int_{0}^{T} \int_{\Gamma(t)} \left\{ (\partial^{\bullet} g + g \operatorname{div}_{\Gamma(t)} V_{\Gamma}) v - \operatorname{div}_{\Gamma(t)} \left[g(\nabla_{\Gamma(t)} v + v V_{\Gamma}^{T}) \right] \right\} \varphi \, d\mathcal{H}^{n-1} \, dt = 0.$$

Since this equality holds for all $\varphi \in C_0^1(S_T)$, we conclude that v satisfies

 $\partial^{\bullet}(gv) + (g\operatorname{div}_{\Gamma(t)}V_{\Gamma})v - \operatorname{div}_{\Gamma(t)}[g(\nabla_{\Gamma(t)}v + vV_{\Gamma}^{T})] = 0 \quad \text{on} \quad S_{T}.$

This is the limit equation of (H_{ε}) . To justify the above argument, we employ a variational framework introduced by Olshanskii, Reusken, and Xu [19].

Definition 2.6.6. Let $v_0 \in L^2(\Gamma_0)$. A function $v \in W_T$ is said to be a variational solution to the initial value problem

$$\begin{cases} \partial^{\bullet}(gv) + (g\operatorname{div}_{\Gamma(t)}V_{\Gamma})v - \operatorname{div}_{\Gamma(t)}[g(\nabla_{\Gamma(t)}v + vV_{\Gamma}^{T})] = 0 & \text{on } S_{T}, \\ v(0) = v_{0} & \text{on } \Gamma_{0}, \end{cases}$$
(H₀)

if v satisfies (2.6.13) for all $\varphi \in H_T$ and $v(0) = v_0$ in $L^2(\Gamma_0)$.

Note that the condition $v(0) = v_0$ in $L^2(\Gamma_0)$ makes sense for $v \in W_T$ by Lemma 2.4.4.

Remark 2.6.7. Suppose that $v \in W_T$ is a variational solution to (H_0) . Then we have $v \in W_\tau$ for each $\tau \in [0, T]$ since W_T is continuously embedded into W_τ . Moreover, by taking test functions φ from $C_0^1(S_\tau)$ we observe that v is a variational solution to (H_0) with T replaced by τ .

We first prove the uniqueness of a variational solution to the initial value problem (H_0) .

Lemma 2.6.8. For each $v_0 \in L^2(\Gamma_0)$, there is at most one variational solution to (H_0) .

Proof. Since (H_0) is linear, it is sufficient to show that if $v \in W_T$ is a variational solution to (H_0) with zero initial data then v = 0.

Let v be a variational solution to (H_0) with v(0) = 0 in $L^2(\Gamma_0)$. For each $\tau \in [0, T]$, we substitute $g^{-1}v \in H_{\tau}$ for φ in (2.6.13) with T replaced by τ and compute as in the proof of Lemma 2.6.4 (replace $M_{\varepsilon}u^{\varepsilon}$ by v and omit $I_{\varepsilon}(u^{\varepsilon}, \varphi; \tau)$). Then we have

$$\|v(\tau)\|_{L^{2}(\Gamma(\tau))}^{2} + \int_{0}^{\tau} \|\nabla_{\Gamma(t)}v(t)\|_{L^{2}(\Gamma(t))}^{2} dt \le \|v(0)\|_{L^{2}(\Gamma_{0})}^{2} + c \int_{0}^{\tau} \|v(t)\|_{L^{2}(\Gamma(t))}^{2} dt.$$

Since v(0) = 0 in $L^2(\Gamma_0)$, the above inequality yields

$$\|v(\tau)\|_{L^{2}(\Gamma(\tau))}^{2} \leq \int_{0}^{\tau} \|v(t)\|_{L^{2}(\Gamma(t))}^{2} dt$$

Hence by Gronwall's inequality we obtain $v(\tau) = 0$ in $L^2(\Gamma(\tau))$ for all $\tau \in [0, T]$.

Now let us show that $\{M_{\varepsilon}u^{\varepsilon}\}_{\varepsilon}$ converges weakly in W_T and that the limit is a unique variational solution to the initial value problem (H_0) .

Theorem 2.6.9. Let $u_0^{\varepsilon} \in L^2(\Omega_{\varepsilon}(0))$ and $u^{\varepsilon} \in L^2_{H^1(\varepsilon)}$ be the unique variational solution to (H_{ε}) given by Theorem 2.3.4. Suppose that the following two conditions are satisfied:
- (a) There exist c > 0 and $\gamma \in (0, 1/2)$ such that $\|u_0^{\varepsilon}\|_{L^2(\Omega_{\varepsilon}(0))} \leq c\varepsilon^{-\gamma}$ for all $\varepsilon > 0$.
- (b) There exists $v_0 \in L^2(\Gamma_0)$ such that $\{M_{\varepsilon}u_0^{\varepsilon}\}_{\varepsilon}$ converges weakly to v_0 in $L^2(\Gamma_0)$ as $\varepsilon \to 0$.

Then $\{M_{\varepsilon}u^{\varepsilon}\}_{\varepsilon}$ converges weakly in W_T as $\varepsilon \to 0$. Moreover, the weak limit $v \in W_T$ of $\{M_{\varepsilon}u^{\varepsilon}\}_{\varepsilon}$ is the unique variational solution to (H_0) with initial data v_0 .

Proof. By the condition (b), $\{M_{\varepsilon}u_0^{\varepsilon}\}_{\varepsilon}$ is bounded in $L^2(\Gamma_0)$. From this fact, the inequality (2.6.12), and the condition (a) it follows that

$$\|M_{\varepsilon}u^{\varepsilon}\|_{W_{T}} \le c \Big(\|M_{\varepsilon}u_{0}^{\varepsilon}\|_{L^{2}(\Gamma_{0})} + \varepsilon^{1/2}\|u_{0}^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon}(0))}\Big) \le c(1 + \varepsilon^{-\gamma + 1/2}) \le c$$
(2.6.14)

with some constant c > 0 independent of ε . Here the last inequality follows from the condition $\gamma \in (0, 1/2)$. Hence $\{M_{\varepsilon}u^{\varepsilon}\}_{\varepsilon}$ is bounded in the Hilbert space W_T and there exist $v \in W_T$ and a sequence $\{\varepsilon_k\}_k$ of positive numbers with $\lim_{k\to\infty} \varepsilon_k = 0$ such that $\{M_{\varepsilon_k}u^{\varepsilon_k}\}_k$ converges weakly to v in W_T as $k \to \infty$.

Let us show that v is the unique variational solution to (H_0) with initial data v_0 . First we show that v satisfies the variational formulation (2.6.13) for all $\varphi \in H_T$. To this end, we return to the variational formulation (2.6.1) of $M_{\varepsilon_k} u^{\varepsilon_k}$:

$$\langle \partial^{\bullet} M_{\varepsilon_{k}} u^{\varepsilon_{k}}, g\varphi \rangle_{T} + \int_{0}^{T} \int_{\Gamma(t)} (\partial^{\bullet} g + g \operatorname{div}_{\Gamma(t)} V_{\Gamma}) (M_{\varepsilon_{k}} u^{\varepsilon_{k}}) \varphi \, d\mathcal{H}^{n-1} \, dt + \int_{0}^{T} \int_{\Gamma(t)} g\{ \nabla_{\Gamma(t)} M_{\varepsilon_{k}} u^{\varepsilon_{k}} + (M_{\varepsilon_{k}} u^{\varepsilon_{k}}) V_{\Gamma}^{T} \} \cdot \nabla_{\Gamma(t)} \varphi \, d\mathcal{H}^{n-1} \, dt = I_{\varepsilon_{k}} (u^{\varepsilon_{k}}, \varphi; T).$$
 (2.6.15)

Let $k \to \infty$ in (2.6.15). Since $\{M_{\varepsilon_k} u^{\varepsilon_k}\}_k$ converges weakly to v in W_T as $k \to \infty$ and g, V_{Γ} are bounded on S_T along with their derivatives, the left-hand side of (2.6.15) converges to

$$\begin{split} \langle \partial^{\bullet} v, g\varphi \rangle_{T} &+ \int_{0}^{T} \int_{\Gamma(t)} (\partial^{\bullet} g + g \operatorname{div}_{\Gamma(t)} V_{\Gamma}) v\varphi \, d\mathcal{H}^{n-1} \, dt \\ &+ \int_{0}^{T} \int_{\Gamma(t)} g(\nabla_{\Gamma(t)} v + v V_{\Gamma}^{T}) \cdot \nabla_{\Gamma(t)} \varphi \, d\mathcal{H}^{n-1} \, dt. \end{split}$$

On the other hand, it follows from (2.6.2) and (2.3.15) that

$$|I_{\varepsilon_{k}}(u^{\varepsilon_{k}},\varphi;T)| \leq c\varepsilon_{k}^{1/2} \int_{0}^{T} \|u^{\varepsilon_{k}}(t)\|_{H^{1}(\Omega_{\varepsilon_{k}}(t))} \|\nabla_{\Gamma(t)}\varphi(t)\|_{L^{2}(\Gamma(t))} dt$$
$$\leq c\varepsilon_{k}^{1/2} \|u^{\varepsilon_{k}}\|_{L^{2}_{H^{1}(\varepsilon_{k})}} \|\varphi\|_{H_{T}} \leq c\varepsilon_{k}^{1/2} \|u^{\varepsilon_{k}}_{0}\|_{L^{2}(\Omega_{\varepsilon_{k}}(0))} \|\varphi\|_{H_{T}}$$

with a constant c > 0 independent of ε_k . This inequality and the condition (a) imply that

$$|I_{\varepsilon_k}(u^{\varepsilon_k},\varphi;T)| \le c\varepsilon_k^{-\gamma+1/2} \|\varphi\|_{H_T} \to 0 \quad \text{as} \quad k \to \infty,$$
(2.6.16)

since $\gamma \in (0, 1/2)$ and c is independent of ε_k . Hence v satisfies (2.6.13) for all $\varphi \in H_T$.

Next we show that $v(0) = v_0$ in $L^2(\Gamma_0)$. Let $\eta \in C^{\infty}([0,T])$ satisfy $\eta(0) = 1$ and $\eta(T) = 0$. We take an arbitrary $\varphi_0 \in C^{\infty}(\Gamma_0)$ and set $\varphi(y,t) := \varphi_0(\Phi^{-1}(y,t))\eta(t)$ for $(y,t) \in \overline{S_T}$, where $\Phi^{-1}(\cdot,t)$ is the inverse mapping of the flow map $\Phi(\cdot,t) : \Gamma_0 \to \Gamma(t)$ (see Section 2.2). Due to the smoothness of Φ^{-1} , the function φ is smooth on $\overline{S_T}$ and thus $\varphi \in W_T$. Moreover, it satisfies $\varphi(0) = \varphi_0$ on Γ_0 and $\varphi(T) = 0$ on $\Gamma(T)$. Substituting $g^{-1}\varphi$ for φ in (2.6.13) and (2.6.15), we have

$$\begin{split} \langle \partial^{\bullet} v, \varphi \rangle_{T} &+ \int_{0}^{T} \int_{\Gamma(t)} (g^{-1} \partial^{\bullet} g + \operatorname{div}_{\Gamma(t)} V_{\Gamma}) v \,\varphi \, d\mathcal{H}^{n-1} \, dt \\ &+ \int_{0}^{T} \int_{\Gamma(t)} g(\nabla_{\Gamma(t)} v + v V_{\Gamma}^{T}) \cdot \nabla_{\Gamma(t)} (g^{-1} \varphi) \, d\mathcal{H}^{n-1} \, dt = 0 \end{split}$$

and

$$\langle \partial^{\bullet} M_{\varepsilon_{k}} u^{\varepsilon_{k}}, \varphi \rangle_{T} + \int_{0}^{T} \int_{\Gamma(t)} (g^{-1} \partial^{\bullet} g + \operatorname{div}_{\Gamma(t)} V_{\Gamma}) (M_{\varepsilon_{k}} u^{\varepsilon_{k}}) \varphi \, d\mathcal{H}^{n-1} \, dt + \int_{0}^{T} \int_{\Gamma(t)} g \{ \nabla_{\Gamma(t)} M_{\varepsilon_{k}} u^{\varepsilon_{k}} + (M_{\varepsilon_{k}} u^{\varepsilon_{k}}) V_{\Gamma}^{T} \} \cdot \nabla_{\Gamma(t)} (g^{-1} \varphi) \, d\mathcal{H}^{n-1} \, dt = I_{\varepsilon_{k}} (u^{\varepsilon_{k}}, g^{-1} \varphi; T).$$

Since φ , v, and $M_{\varepsilon_k} u^{\varepsilon_k}$ are in W_T , we can apply the identity (2.4.8) to get

$$\langle \partial^{\bullet} v, \varphi \rangle_{T} = -\langle \partial^{\bullet} \varphi, v \rangle_{T} - \int_{\Gamma_{0}} v(0) \varphi_{0} \, d\mathcal{H}^{n-1} - \int_{0}^{T} \int_{\Gamma(t)} v\varphi \operatorname{div}_{\Gamma(t)} V_{\Gamma} \, d\mathcal{H}^{n-1} \, dt$$

and the same identity for $M_{\varepsilon_k} u^{\varepsilon_k}$. Here we used the conditions $\varphi(0) = \varphi_0$ on Γ_0 and $\varphi(T) = 0$ on $\Gamma(T)$. Thus we have

$$-\langle \partial^{\bullet}\varphi, v \rangle_{T} + \int_{0}^{T} \int_{\Gamma(t)} g^{-1} (\partial^{\bullet}g) v\varphi \, d\mathcal{H}^{n-1} \, dt + \int_{0}^{T} \int_{\Gamma(t)} g(\nabla_{\Gamma(t)}v + vV_{\Gamma}^{T}) \cdot \nabla_{\Gamma(t)} (g^{-1}\varphi) \, d\mathcal{H}^{n-1} \, dt = \int_{\Gamma_{0}} v(0)\varphi_{0} \, d\mathcal{H}^{n-1} \quad (2.6.17)$$

and

$$-\langle \partial^{\bullet}\varphi, M_{\varepsilon_{k}}u^{\varepsilon_{k}}\rangle_{T} + \int_{0}^{T}\int_{\Gamma(t)}g^{-1}(\partial^{\bullet}g)(M_{\varepsilon_{k}}u^{\varepsilon_{k}})\varphi \,d\mathcal{H}^{n-1}\,dt + \int_{0}^{T}\int_{\Gamma(t)}g\{\nabla_{\Gamma(t)}M_{\varepsilon_{k}}u^{\varepsilon_{k}} + (M_{\varepsilon_{k}}u^{\varepsilon_{k}})V_{\Gamma}^{T}\}\cdot\nabla_{\Gamma(t)}(g^{-1}\varphi)\,d\mathcal{H}^{n-1}\,dt = \int_{\Gamma_{0}}(M_{\varepsilon_{k}}u_{0}^{\varepsilon_{k}})\varphi_{0}\,d\mathcal{H}^{n-1} + I_{\varepsilon_{k}}(u^{\varepsilon_{k}},g^{-1}\varphi;T). \quad (2.6.18)$$

Let $k \to \infty$ in (2.6.18). Since $\{M_{\varepsilon}u_0^{\varepsilon}\}_{\varepsilon}$ converges weakly to v_0 in $L^2(\Gamma_0)$ as $\varepsilon \to 0$,

$$\lim_{k \to \infty} \int_{\Gamma_0} M_{\varepsilon_k} u_0^{\varepsilon_k} \varphi_0 \, d\mathcal{H}^{n-1} = \int_{\Gamma_0} v_0 \, \varphi_0 \, d\mathcal{H}^{n-1}.$$

Moreover, since $\{M_{\varepsilon_k}u^{\varepsilon_k}\}_k$ converges weakly to v in W_T as $k \to \infty$ and (2.6.16) holds with φ replaced by $g^{-1}\varphi$, both sides of (2.6.18) converge to

$$-\langle \partial^{\bullet} \varphi, v \rangle_{T} + \int_{0}^{T} \int_{\Gamma(t)} g^{-1} (\partial^{\bullet} g) v \varphi \, d\mathcal{H}^{n-1} \, dt + \int_{0}^{T} \int_{\Gamma(t)} g(\nabla_{\Gamma(t)} v + v V_{\Gamma}^{T}) \cdot \nabla_{\Gamma(t)} (g^{-1} \varphi) \, d\mathcal{H}^{n-1} \, dt = \int_{\Gamma_{0}} v_{0} \, \varphi_{0} \, d\mathcal{H}^{n-1}. \quad (2.6.19)$$

Comparing (2.6.17) and (2.6.19), we obtain

$$\int_{\Gamma_0} v(0)\varphi_0 \, d\mathcal{H}^{n-1} = \int_{\Gamma_0} v_0 \, \varphi_0 \, d\mathcal{H}^{n-1} \quad \text{for all} \quad \varphi_0 \in C^\infty(\Gamma_0).$$

Since $C^{\infty}(\Gamma_0)$ is dense in $L^2(\Gamma_0)$, it follows that $v(0) = v_0$ in $L^2(\Gamma_0)$. Hence v is the unique variational solution to (H_0) with initial data v_0 . Here the uniqueness follows from Lemma 2.6.8.

Finally, using the boundedness of $\{M_{\varepsilon}u^{\varepsilon}\}_{\varepsilon}$ in W_T (see (2.6.14)) and the uniqueness of a variational solution to (H_0) (see Lemma 2.6.8), we can prove by contradiction that the full sequence $\{M_{\varepsilon}u^{\varepsilon}\}_{\varepsilon}$ converges weakly to v in W_T as $\varepsilon \to 0$. The argument is standard and thus we omit the details.

Corollary 2.6.10. For every $v_0 \in L^2(\Gamma_0)$, there exists a unique variational solution to (H_0) . *Proof.* For each $\varepsilon > 0$, we define a function u_0^{ε} on $\Omega_{\varepsilon}(0)$ as

$$u_0^{\varepsilon}(X) := \frac{v_0(p(X,0))}{J(p(X,0),0,d(X,0))}, \quad X \in \Omega_{\varepsilon}(0).$$

Clearly $M_{\varepsilon}u_0^{\varepsilon} = v_0$ holds on Γ_0 . Moreover, by the co-area formula (2.5.1) and (2.5.4) we have

$$\begin{aligned} \|u_0^{\varepsilon}\|_{L^2(\Omega_{\varepsilon}(0))} &= \left(\int_{\Gamma_0} \int_{\varepsilon g_0(Y,0)}^{\varepsilon g_1(Y,0)} |v_0(Y)|^2 J(Y,0,\rho)^{-1} \, d\rho \, d\mathcal{H}^{n-1}(Y)\right)^{1/2} \\ &\leq c \left(\int_{\Gamma_0} \varepsilon g(Y,0) |v_0(Y)|^2 \, d\mathcal{H}^{n-1}(Y)\right)^{1/2} \leq c \varepsilon^{1/2} \|v_0\|_{L^2(\Gamma_0)} \end{aligned}$$

with a constant c > 0 independent of ε . Hence $u_0^{\varepsilon} \in L^2(\Omega_{\varepsilon}(0))$ and u_0^{ε} , v_0 satisfy the conditions (a) and (b) of Theorem 2.6.9. Thus the corollary follows from Theorem 2.3.4 and Theorem 2.6.9.

Remark 2.6.11. Let $H = -\operatorname{div}_{\Gamma(t)}\nu$ be the mean curvature of $\Gamma(t)$. Since the material derivative operator is formally of the form $\partial^{\bullet} = \partial_t + v_{\Gamma}^N \nu \cdot \nabla + V_{\Gamma}^T \cdot \nabla_{\Gamma(t)}$ (see Remark 2.5.12) and the formula $\operatorname{div}_{\Gamma(t)}(v_{\Gamma}^N \nu) = -v_{\Gamma}^N H$ holds (see Remark 2.5.15), the limit equation (H_0) is formally equivalent to

$$\partial^{\circ}(gv) - v_{\Gamma}^{N}Hgv - \operatorname{div}_{\Gamma(t)}(g\nabla_{\Gamma(t)}v) = 0 \quad \text{on} \quad S_{T}.$$

Here $\partial^{\circ} = \partial_t + v_{\Gamma}^N \nu \cdot \nabla$ is the normal time derivative (see [2,3,5]). This equation depends on v_{Γ}^N , ν , and H, which represent the geometric motion of $\Gamma(t)$. On the other hand, it is independent of the tangential velocity V_{Γ}^T , which represents advection along $\Gamma(t)$. Hence, as we mentioned in Section 2.1, the evolution of the limit v given by Theorem 2.6.9 is not affected by advection along $\Gamma(t)$, but the geometric motion of $\Gamma(t)$.

2.6.4 Estimates for the difference between solutions to the heat equation and the limit equation

Let us estimate the difference between variational solutions to (H_{ε}) and (H_0) . For a function v on S_T , let \overline{v} be its constant extension in the normal direction of $\Gamma(t)$. For a function u on $Q_{\varepsilon,T}$, we set

$$||u||_{L^2(Q_{\varepsilon,T})} := \left(\int_0^T \int_{\Omega_{\varepsilon}(t)} |u|^2 \, dx \, dt\right)^{1/2}.$$

Theorem 2.6.12. Let $u_0^{\varepsilon} \in L^2(\Omega_{\varepsilon}(0))$ and $u^{\varepsilon} \in L^2_{H^1(\varepsilon)}$ be a unique variational solution to (H_{ε}) . Also, let $v_0 \in L^2(\Gamma_0)$ and $v \in W_T$ be a unique variational solution to (H_0) . Then there exists a constant c > 0 independent of u_0^{ε} , u^{ε} , v_0 , v, and ε such that

$$\|u^{\varepsilon} - \overline{v}\|_{L^2(Q_{\varepsilon,T})} \le c \Big(\|u_0^{\varepsilon} - \overline{v_0}\|_{L^2(\Omega_{\varepsilon}(0))} + \varepsilon^{3/2} \|v_0\|_{L^2(\Gamma_0)}\Big).$$

$$(2.6.20)$$

In particular, for each $\alpha \in [0, 3/2)$ we have

$$\lim_{\varepsilon \to 0} \varepsilon^{-\alpha} \| u^{\varepsilon} - \overline{v} \|_{L^2(Q_{\varepsilon,T})} = 0 \quad provided \quad \lim_{\varepsilon \to 0} \varepsilon^{-\alpha} \| u_0^{\varepsilon} - \overline{v_0} \|_{L^2(\Omega_{\varepsilon}(0))} = 0.$$

We first estimate the difference between $M_{\varepsilon}u^{\varepsilon}$ and v in the space W_T .

Lemma 2.6.13. Let u_0^{ε} , u^{ε} , v_0 , and v be as in Theorem 2.6.12. Then there exists a constant c > 0 independent of u_0^{ε} , u^{ε} , v_0 , v, and ε such that

$$\|M_{\varepsilon}u^{\varepsilon} - v\|_{W_T} \le c \Big(\|M_{\varepsilon}u_0^{\varepsilon} - v_0\|_{L^2(\Gamma_0)} + \varepsilon^{1/2}\|u_0^{\varepsilon}\|_{L^2(\Omega_{\varepsilon})}\Big).$$
(2.6.21)

In particular, if $\lim_{\varepsilon \to 0} \|M_{\varepsilon}u_0^{\varepsilon} - v_0\|_{L^2(\Gamma_0)} = 0$ and $\lim_{\varepsilon \to 0} \varepsilon^{1/2} \|u_0^{\varepsilon}\|_{L^2(\Omega_{\varepsilon}(0))} = 0$, then $\{M_{\varepsilon}u^{\varepsilon}\}_{\varepsilon}$ converges strongly to v in W_T .

Proof. For each $\tau \in [0, T]$, we subtract both sides of (2.6.13) with T replaced by τ from those of (2.6.1). Then we have

$$\begin{aligned} \langle \partial^{\bullet} (M_{\varepsilon} u^{\varepsilon} - v), g\varphi \rangle_{\tau} &+ \int_{0}^{\tau} \int_{\Gamma(t)} (\partial^{\bullet} g + g \operatorname{div}_{\Gamma(t)} V_{\Gamma}) (M_{\varepsilon} u^{\varepsilon} - v) \varphi \, d\mathcal{H}^{n-1} \, dt \\ &+ \int_{0}^{\tau} \int_{\Gamma(t)} g \{ \nabla_{\Gamma(t)} (M_{\varepsilon} u^{\varepsilon} - v) + (M_{\varepsilon} u^{\varepsilon} - v) V_{\Gamma}^{T} \} \cdot \nabla_{\Gamma(t)} \varphi \, d\mathcal{H}^{n-1} \, dt = I_{\varepsilon} (u^{\varepsilon}, \varphi; \tau) \end{aligned}$$

for all $\varphi \in H_{\tau}$. Hence, by calculating as in the proof of Lemma 2.6.2, Lemma 2.6.4, and Lemma 2.6.5 (replace $M_{\varepsilon}u^{\varepsilon}$ by $M_{\varepsilon}u^{\varepsilon} - v$), we obtain (2.6.21).

Proof of Theorem 2.6.12. First we show the inequality

$$\|u^{\varepsilon} - \overline{v}\|_{L^{2}(Q_{\varepsilon,T})} \le c\varepsilon^{1/2} \Big(\|M_{\varepsilon}u^{\varepsilon}_{0} - v_{0}\|_{L^{2}(\Gamma_{0})} + \varepsilon^{1/2} \|u^{\varepsilon}_{0}\|_{L^{2}(\Omega_{\varepsilon}(0))} \Big).$$
(2.6.22)

To this end, we use the triangle inequality

$$\|u^{\varepsilon} - \overline{v}\|_{L^{2}(Q_{\varepsilon,T})} \leq \left\|u^{\varepsilon} - \overline{M_{\varepsilon}u^{\varepsilon}}\right\|_{L^{2}(Q_{\varepsilon,T})} + \left\|\overline{M_{\varepsilon}u^{\varepsilon}} - \overline{v}\right\|_{L^{2}(Q_{\varepsilon,T})}$$

and estimate the right-hand side of the above inequality. By (2.5.9) and (2.3.15), we have

$$\left\| u^{\varepsilon} - \overline{M_{\varepsilon} u^{\varepsilon}} \right\|_{L^{2}(Q_{\varepsilon,T})} \le c\varepsilon \| u^{\varepsilon} \|_{L^{2}_{H^{1}(\varepsilon)}} \le c\varepsilon \| u^{\varepsilon}_{0} \|_{L^{2}(\Omega_{\varepsilon}(0))}$$

with a constant c > 0 independent of ε . On the other hand, by (2.5.7) and (2.6.21),

$$\left\|\overline{M_{\varepsilon}u^{\varepsilon}}-\overline{v}\right\|_{L^{2}(Q_{\varepsilon,T})} \leq c\varepsilon^{1/2} \|M_{\varepsilon}u^{\varepsilon}-v\|_{H_{T}} \leq c\varepsilon^{1/2} \Big(\|M_{\varepsilon}u_{0}^{\varepsilon}-v_{0}\|_{L^{2}(\Gamma_{0})} + \varepsilon^{1/2} \|u_{0}^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon})}\Big).$$

Hence (2.6.22) follows.

Next we estimate the right-hand side of (2.6.22) to get (2.6.20). We use the notation

$$(u_0^{\varepsilon})^{\sharp}(Y,\rho) := u_0^{\varepsilon}(Y+\rho\nu(Y,0)), \quad Y \in \Gamma_0, \, \rho \in (\varepsilon g_0(Y,0), \varepsilon g_1(Y,0)),$$

and omit the variables Y, ρ , and t = 0. We set

$$I_1 := \frac{1}{\varepsilon g} \int_{\varepsilon g_0}^{\varepsilon g_1} \{ (u_0^{\varepsilon})^{\sharp} - v_0 \} J \, d\rho, \quad I_2 := \frac{v_0}{\varepsilon g} \int_{\varepsilon g_0}^{\varepsilon g_1} (J-1) \, d\rho.$$

Then $M_{\varepsilon}u_0^{\varepsilon} - v_0 = I_1 + I_2$ on Γ_0 . By Hölder's inequality and (2.2.1), (2.5.4), we have

$$|I_1|^2 \leq \frac{1}{\varepsilon g} \int_{\varepsilon g_0}^{\varepsilon g_1} |(u_0^\varepsilon)^{\sharp} - v_0|^2 J^2 \, d\rho \leq c\varepsilon^{-1} \int_{\varepsilon g_0}^{\varepsilon g_1} |(u_0^\varepsilon)^{\sharp} - v_0|^2 J \, d\rho.$$

On the other hand, (2.5.5) yields $|I_2| \leq c\varepsilon |v_0|$. Hence

$$\begin{split} \|M_{\varepsilon}u_{0}^{\varepsilon} - v_{0}\|_{L^{2}(\Gamma_{0})}^{2} &\leq c \int_{\Gamma_{0}} (|I_{1}|^{2} + |I_{2}|^{2}) \, d\mathcal{H}^{n-1} \\ &\leq c \int_{\Gamma_{0}} \left(\varepsilon^{-1} \int_{\varepsilon g_{0}}^{\varepsilon g_{1}} |(u_{0}^{\varepsilon})^{\sharp} - v_{0}|^{2} J \, d\rho + \varepsilon^{2} |v_{0}|^{2} \right) d\mathcal{H}^{n-1} \\ &= c \Big(\varepsilon^{-1} \|u_{0}^{\varepsilon} - \overline{v_{0}}\|_{L^{2}(\Omega_{\varepsilon}(0))}^{2} + \varepsilon^{2} \|v_{0}\|_{L^{2}(\Gamma_{0})}^{2} \Big). \end{split}$$

Here we used the co-area formula (2.5.1) in the last equality. The above inequality is equivalent to

$$\|M_{\varepsilon}u_{0}^{\varepsilon} - v_{0}\|_{L^{2}(\Gamma_{0})} \leq c \Big(\varepsilon^{-1/2} \|u_{0}^{\varepsilon} - \overline{v_{0}}\|_{L^{2}(\Omega_{\varepsilon}(0))} + \varepsilon \|v_{0}\|_{L^{2}(\Gamma_{0})}\Big).$$
(2.6.23)

Moreover, by the triangle inequality and (2.5.7),

$$\begin{aligned} \|u_{0}^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon}(0))} &\leq \|u_{0}^{\varepsilon} - \overline{v_{0}}\|_{L^{2}(\Omega_{\varepsilon}(0))} + \|\overline{v_{0}}\|_{L^{2}(\Omega_{\varepsilon}(0))} \\ &\leq \|u_{0}^{\varepsilon} - \overline{v_{0}}\|_{L^{2}(\Omega_{\varepsilon}(0))} + c\varepsilon^{1/2} \|v_{0}\|_{L^{2}(\Gamma_{0})}. \end{aligned}$$
(2.6.24)

Finally, by applying (2.6.23) and (2.6.24) to (2.6.22), we obtain

$$\begin{aligned} \|u^{\varepsilon} - \overline{v}\|_{L^{2}(Q_{\varepsilon,T})} &\leq c\varepsilon^{1/2} \Big(\|M_{\varepsilon}u_{0}^{\varepsilon} - v_{0}\|_{L^{2}(\Gamma_{0})} + \varepsilon^{1/2} \|u_{0}^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon}(0))} \Big) \\ &\leq c\varepsilon^{1/2} \Big((\varepsilon^{-1/2} + \varepsilon^{1/2}) \|u_{0}^{\varepsilon} - \overline{v_{0}}\|_{L^{2}(\Omega_{\varepsilon}(0))} + \varepsilon \|v_{0}\|_{L^{2}(\Gamma_{0})} \Big) \\ &\leq c \Big(\|u_{0}^{\varepsilon} - \overline{v_{0}}\|_{L^{2}(\Omega_{\varepsilon}(0))} + \varepsilon^{3/2} \|v_{0}\|_{L^{2}(\Gamma_{0})} \Big) \end{aligned}$$

with a constant c > 0 independent of ε . Hence (2.6.20) holds.

2.A Heuristic derivation of the limit equation

Let us give a heuristic derivation of the limit equation (2.1.1) from (H_{ε}) when $\Omega_{\varepsilon}(t)$ is of the form $\Omega_{\varepsilon}(t) = \{x \in \mathbb{R}^n \mid -\varepsilon < d(x,t) < \varepsilon\}$. In this case, the unit outward normal vector field ν_{ε} of $\partial\Omega_{\varepsilon}(t)$ and the outer normal velocity v_{ε}^N of $\partial\Omega_{\varepsilon}(t)$ are of the form

$$\nu_{\varepsilon}(x,t) = \pm \nu(p(x,t),t), \quad v_{\varepsilon}^{N}(x,t) = \pm v_{\Gamma}^{N}(p(x,t),t), \quad (x,t) \in \partial_{\ell}Q_{\varepsilon,T},$$

according to $d(x,t) = \pm \varepsilon$ (double-sign corresponds). Thus we start from the heat equation

$$\partial_t u^{\varepsilon}(x,t) - \Delta u^{\varepsilon}(x,t) = 0, \quad (x,t) \in Q_{\varepsilon,T}$$

with the boundary condition

$$\nu(p(x,t),t) \cdot \nabla u^{\varepsilon}(x,t) + v_{\Gamma}^{N}(p(x,t),t)u^{\varepsilon}(x,t) = 0, \quad (x,t) \in \partial_{\ell}Q_{\varepsilon,T}.$$
(2.A.1)

To derive the limit equation, we make the following assumptions:

- (1) The signed distance d(x,t) of $x \in \Omega_{\varepsilon}(t)$ is negligible $(d(x,t) \approx 0)$, although the quantity $\varepsilon^{-1}d(x,t)$ is not negligible.
- (2) The relation $v_{\Gamma}^{N}(p(x,t),t) \approx -\partial_{t}d(x,t)$ holds for all $(x,t) \in Q_{\varepsilon,T}$.
- (3) The boundary condition (2.A.1) also holds in the noncylindrical domain $Q_{\varepsilon,T}$.

These assumptions come from the smallness of the width 2ε of $\Omega_{\varepsilon}(t)$. Taking the third assumption into account, we consider the two equations

$$\partial_t u^{\varepsilon}(x,t) - \Delta u^{\varepsilon}(x,t) = 0, \qquad (2.A.2)$$

$$\nu(p(x,t),t) \cdot \nabla u^{\varepsilon}(x,t) + v_{\Gamma}^{N}(p(x,t),t)u^{\varepsilon}(x,t) = 0$$
(2.A.3)

for $(x,t) \in Q_{\varepsilon,T}$. Recall that each $x \in \Omega_{\varepsilon}(t)$ is represented as

$$x=p(x,t)+d(x,t)\nu(p(x,t),t),\quad \nabla d(x,t)=\nu(x,t)=\nu(p(x,t),t).$$

First, we consider the gradient matrix of the projection p(x,t) onto $\Gamma(t)$ given by

$$\nabla p = \begin{pmatrix} \partial_1 p_1 & \dots & \partial_n p_1 \\ \vdots & \ddots & \vdots \\ \partial_1 p_n & \dots & \partial_n p_n \end{pmatrix} \quad \text{for} \quad p = \begin{pmatrix} p_1 \\ \vdots \\ p_n \end{pmatrix}.$$

By differentiating both sides of $x = p(x,t) + d(x,t)\nu(x,t)$ and using $\nabla d(x,t) = \nu(x,t)$, we have

 $I_n = \nabla p(x,t) + \nu(x,t) \otimes \nu(x,t) + d(x,t) \nabla \nu(x,t).$

According to the assumption (1), the above equality reads

$$\nabla p(x,t) \approx I_n - \nu(x,t) \otimes \nu(x,t) = I_n - \nu(p(x,t),t) \otimes \nu(p(x,t),t).$$
(2.A.4)

We define a function $v: S_T \times (-1, 1) \to \mathbb{R}$ as

$$v(y,t,r) := u^{\varepsilon}(y + \varepsilon r \nu(y,t),t), \quad (y,t) \in S_T, \ r \in (-1,1).$$

Then u^{ε} is represented by v as

$$u^{\varepsilon}(x,t) = v(p(x,t), t, \varepsilon^{-1}d(x,t)), \quad (x,t) \in Q_{\varepsilon,T}.$$
(2.A.5)

For abbreviation, we write p and d for p(x,t) and d(x,t) in arguments of functions unless we would like to emphasize them. For example, we write $\nu(p,t)$ for $\nu(p(x,t),t)$ and $\nu(p,t,\varepsilon^{-1}d)$ for $\nu(p(x,t),t,\varepsilon^{-1}d(x,t))$. By the chain rule of differentiation we have

$$\nabla u^{\varepsilon}(x,t) = [\nabla p(x,t)]^T \nabla v(p,t,\varepsilon^{-1}d) + \varepsilon^{-1}\partial_r v(p,t,\varepsilon^{-1}d) \nabla d(x,t) + \varepsilon^{-1}\partial_r v(p,t,\varepsilon^{-1}d) + \varepsilon^{-1}\partial_$$

By (2.A.4) and $\nabla d(x,t) = \nu(p(x,t),t)$, this equality reads

$$\nabla u^{\varepsilon}(x,t) \approx \nabla_{\Gamma(t)} v(p,t,\varepsilon^{-1}d) + \varepsilon^{-1} \partial_r v(p,t,\varepsilon^{-1}d) \nu(p,t).$$
(2.A.6)

Here we abused the definition of the tangential gradient $\nabla_{\Gamma(t)} = (I_n - \nu \otimes \nu)\nabla$. Applying (2.A.6) to (2.A.3) and observing that $\nu(p,t) \cdot \nabla_{\Gamma(t)} v(p,t,\varepsilon^{-1}d) = 0$, we obtain

$$\varepsilon^{-1}\partial_r v(p,t,\varepsilon^{-1}d) \approx -v_{\Gamma}^N(p,t)v(p,t,\varepsilon^{-1}d)$$
 (2.A.7)

and thus (2.A.6) becomes

$$\nabla u^{\varepsilon}(x,t) \approx \nabla_{\Gamma(t)} v(p,t,\varepsilon^{-1}d) - v_{\Gamma}^{N}(p,t) v(p,t,\varepsilon^{-1}d) \nu(p,t).$$
(2.A.8)

Next we compute $\Delta u^{\varepsilon} = \operatorname{div} \nabla u^{\varepsilon}$. For a vector field F on $\Omega_{\varepsilon}(t)$ with each fixed $t \in [0, T]$,

$$\operatorname{div} F(x) = \operatorname{trace}[\nabla F(x)]$$

=
$$\operatorname{trace}[\{I_n - \nu(x,t) \otimes \nu(x,t)\} \nabla F(x)] + \operatorname{trace}[\nu(x,t) \otimes \nu(x,t) \nabla F(x)]$$

=
$$\operatorname{div}_{\Gamma(t)} F(x) + \nu(x,t) \cdot \partial_{\nu} F(x)$$

holds since $\nu \otimes \nu$ is a projection matrix onto the ν -direction. Hence we have

$$\operatorname{div}\left[\nabla_{\Gamma(t)}v(p,t,\varepsilon^{-1}d)\right] = \operatorname{div}_{\Gamma(t)}\left[\nabla_{\Gamma(t)}v(p,t,\varepsilon^{-1}d)\right] + \nu(x,t) \cdot \partial_{\nu}\left[\nabla_{\Gamma(t)}v(p,t,\varepsilon^{-1}d)\right].$$

Moreover, since $p(x + h\nu(x,t), t) = p(x,t)$ and $d(x + h\nu(x,t), t) = d(x,t) + h$ for sufficiently small $h \in \mathbb{R}$, it follows that

$$\nabla_{\Gamma(t)}v(p(x+h\nu(x,t),t),t,\varepsilon^{-1}d(x+h\nu(x,t),t)) = \nabla_{\Gamma(t)}v(p(x,t),t,\varepsilon^{-1}d(x,t)+\varepsilon^{-1}h)$$

and thus

$$\partial_{\nu} \left[\nabla_{\Gamma(t)} v(p, t, \varepsilon^{-1} d) \right] = \varepsilon^{-1} \partial_{r} \left[\nabla_{\Gamma(t)} v(p, t, \varepsilon^{-1} d) \right]$$

by the formula $\partial_{\nu} f(x) = \lim_{h \to 0} \{f(x + h\nu(x, t)) - f(x)\}/h$ for functions f on $\Omega_{\varepsilon}(t)$ with fixed $t \in [0, T]$. Hence we obtain

$$\operatorname{div}\left[\nabla_{\Gamma(t)}v(p,t,\varepsilon^{-1}d)\right] = \operatorname{div}_{\Gamma(t)}\left[\nabla_{\Gamma(t)}v(p,t,\varepsilon^{-1}d)\right] + \varepsilon^{-1}\nu(x,t) \cdot \partial_r\left[\nabla_{\Gamma(t)}v(p,t,\varepsilon^{-1}d)\right].$$

Similarly we have

$$\begin{aligned} \operatorname{div} \begin{bmatrix} v_{\Gamma}^{N}(p,t)v(p,t,\varepsilon^{-1}d)\nu(p,t) \end{bmatrix} \\ &= \operatorname{div}_{\Gamma(t)} \begin{bmatrix} v_{\Gamma}^{N}(p,t)v(p,t,\varepsilon^{-1}d)\nu(p,t) \end{bmatrix} + \nu(x,t) \cdot \{\varepsilon^{-1}v_{\Gamma}^{N}(p,t)\partial_{r}v(p,t,\varepsilon^{-1}d)\nu(p,t) \} \\ &\approx \operatorname{div}_{\Gamma(t)} \begin{bmatrix} v_{\Gamma}^{N}(p,t)v(p,t,\varepsilon^{-1}d)\nu(p,t) \end{bmatrix} - \{v_{\Gamma}^{N}(p,t)\}^{2}v(p,t,\varepsilon^{-1}d). \end{aligned}$$

Here the last approximation follows from $\nu(x,t) = \nu(p(x,t),t)$ and (2.A.7). Hence, by (2.A.8),

$$\Delta u^{\varepsilon}(x,t) \approx \operatorname{div}_{\Gamma(t)} \left[\nabla_{\Gamma(t)} v(p,t,\varepsilon^{-1}d) \right] + \varepsilon^{-1} \nu(x,t) \cdot \partial_r \left[\nabla_{\Gamma(t)} v(p,t,\varepsilon^{-1}d) \right] - \operatorname{div}_{\Gamma(t)} \left[v_{\Gamma}^N(p,t) v(p,t,\varepsilon^{-1}d) \nu(p,t) \right] + \{ v_{\Gamma}^N(p,t) \}^2 v(p,t,\varepsilon^{-1}d). \quad (2.A.9)$$

On the other hand, we differentiate both sides of (2.A.5) with respect to t to get

$$\partial_t u^{\varepsilon}(x,t) = \partial_t p(x,t) \cdot \nabla v(p,t,\varepsilon^{-1}d) + \partial_t v(p,t,\varepsilon^{-1}d) + \varepsilon^{-1} \partial_t d(x,t) \partial_r v(p,t,\varepsilon^{-1}d).$$

To this equality we apply (2.A.7) and

$$\partial_t p(x,t) = -\partial_t d(x,t)\nu(x,t) - d(x,t)\partial_t \nu(x,t) \approx v_{\Gamma}^N(p,t)\nu(p,t),$$

where the last approximation follows from the assumptions (1), (2), and $\nu(x,t) = \nu(p(x,t),t)$. Then we have

$$\partial_t u^{\varepsilon}(x,t) \approx v_{\Gamma}^N(p,t)\nu(p,t) \cdot \nabla v(p,t,\varepsilon^{-1}d) + \partial_t v(p,t,\varepsilon^{-1}d) + \{v_{\Gamma}^N(p,t)\}^2 v(p,t,\varepsilon^{-1}d).$$
(2.A.10)

Substituting (2.A.9) and (2.A.10) for the equation (2.A.2), we obtain

$$\begin{split} \partial_t v(p,t,\varepsilon^{-1}d) + v_{\Gamma}^N(p,t)\nu(p,t)\cdot\nabla v(p,t,\varepsilon^{-1}d) + \operatorname{div}_{\Gamma(t)} \big[v_{\Gamma}^N(p,t)v(p,t,\varepsilon^{-1}d)\nu(p,t) \big] \\ &- \operatorname{div}_{\Gamma(t)} \big[\nabla_{\Gamma(t)}v(p,t,\varepsilon^{-1}d) \big] - \varepsilon^{-1}\nu(x,t)\cdot\partial_r \big[\nabla_{\Gamma(t)}v(p,t,\varepsilon^{-1}d) \big] = 0. \end{split}$$

Now let us make an additional assumption: the function v(y, t, r) is independent of the variable r. Then, the above equation reads

$$\partial_t v(y,t) + v_{\Gamma}^N(y,t)\nu(y,t) \cdot \nabla v(y,t) + \operatorname{div}_{\Gamma(t)} \left[v_{\Gamma}^N(y,t)v(y,t)\nu(y,t) \right] - \operatorname{div}_{\Gamma(t)} \left[\nabla_{\Gamma(t)}v(y,t) \right] = 0$$

with $y = p(x, t) \in \Gamma(t)$. Finally we observe that

$$\operatorname{div}_{\Gamma(t)}(v_{\Gamma}^{N}v\nu) = \nabla_{\Gamma(t)}(v_{\Gamma}^{N}v) \cdot \nu + v_{\Gamma}^{N}v \operatorname{div}_{\Gamma(t)}\nu = 0 + v_{\Gamma}^{N}v \cdot (-H) = -v_{\Gamma}^{N}Hv,$$

where $H = -\text{div}_{\Gamma(t)}\nu$ is the mean curvature of $\Gamma(t)$, to obtain

$$\partial_t v(y,t) + v_{\Gamma}^N(y,t)\nu(y,t) \cdot \nabla v(y,t) - v_{\Gamma}^N(y,t)H(y,t)v(y,t) - \Delta_{\Gamma(t)}v(y,t) = 0$$

for $(y,t) \in S_T$. This is the limit equation (2.1.1) we mentioned in Section 2.1.

2.B Elementary facts on integrals over evolving surfaces

In this appendix we give complete proofs of several facts on integrals over evolving surfaces which are essentially known or easily proved but there is no detailed proof for the readers' convenience. We first show the transformation formula (2.4.2).

Proof of (2.4.2). By a localization argument with a partition of unity of S_T , it is sufficient to show

$$\int_{I} \int_{\mu_{t}(U)} f(y,t) \, d\mathcal{H}^{n-1}(y) \, dt = \int_{\zeta(U \times I)} f(\sigma) (1 + |v_{\Gamma}^{N}(\sigma)|^{2})^{-1/2} \, d\mathcal{H}^{n}(\sigma), \tag{2.B.1}$$

where I is an open interval in (0,T), U is an open set in \mathbb{R}^{n-1} , $\mu_t \colon U \to \Gamma(t)$ is a smooth local parametrization of $\Gamma(t)$ for each $t \in I$, and $\zeta \colon U \times I \to S_T$ is given by $\zeta(s,t) = (\mu_t(s),t)$. Moreover, by rotating coordinates and taking I sufficiently small, we may assume that there exists a smooth function h on $U \times I$ such that $\mu_t(s) = (s, h(s, t))$ for all $s \in U$ and $t \in I$. Then $\zeta(s,t) = (s, h(s,t), t)$ and the outward normal velocity v_{Γ}^N of $\Gamma(t)$ is given by

$$v_{\Gamma}^{N}(\mu_{t}(s),t) = \frac{\partial_{t}h(s,t)}{\sqrt{1+|\nabla' h(s,t)|^{2}}}, \quad (s,t) \in U \times I.$$
(2.B.2)

Here ∇' is the gradient in $s \in \mathbb{R}^{n-1}$ and we assume that the *n*-th component of the normal ν is positive on $\zeta(U \times I)$. For $t \in I$ the Riemannian metric on $\Gamma(t)$ is locally given by

$$\frac{\partial \mu_t}{\partial s_i}(s) \cdot \frac{\partial \mu_t}{\partial s_j}(s) = \delta_{ij} + \frac{\partial h}{\partial s_i}(s,t) \frac{\partial h}{\partial s_j}(s,t), \quad s \in U, \, i, j = 1, \dots, n-1,$$

where δ_{ij} is the Kronecker delta. Hence the left-hand side of (2.B.1) is

$$\int_{I} \int_{\mu_{t}(U)} f(y,t) \, d\mathcal{H}^{n-1}(y) \, dt = \int_{I} \int_{U} f(\mu_{t}(s),t) \sqrt{1 + |\nabla' h(s,t)|^{2}} \, ds \, dt.$$
(2.B.3)

On the other hand, since the Riemannian metric on S_T is locally given by

$$\begin{split} \frac{\partial \zeta}{\partial s_i}(s,t) \cdot \frac{\partial \zeta}{\partial s_j}(s,t) &= \delta_{ij} + \frac{\partial h}{\partial s_i}(s,t) \frac{\partial h}{\partial s_i}(s,t), \\ \frac{\partial \zeta}{\partial s_i}(s,t) \cdot \frac{\partial \zeta}{\partial t}(s,t) &= \frac{\partial h}{\partial s_i}(s,t) \frac{\partial h}{\partial t}(s,t), \quad \frac{\partial \zeta}{\partial t}(s,t) \cdot \frac{\partial \zeta}{\partial t}(s,t) = 1 + \left| \frac{\partial h}{\partial t}(s,t) \right|^2 \end{split}$$

for $s \in U$, $t \in I$, and i, j = 1, ..., n - 1, the right-hand side of (2.B.1) is

$$\int_{\zeta(U\times I)} f(\sigma)(1+|v_{\Gamma}^{N}(\sigma)|^{2})^{-1/2} d\mathcal{H}^{n}(\sigma)$$

=
$$\int_{U\times I} f(\mu_{t}(s),t)(1+|v_{\Gamma}^{N}(\mu_{t}(s),t)|^{2})^{-1/2} \sqrt{\det A(s,t)} \, ds \, dt. \quad (2.B.4)$$

Here A is a matrix of the form

$$A = \begin{pmatrix} I_{n-1} + \nabla' h \otimes \nabla' h & \partial_t h \nabla' h \\ \partial_t h (\nabla' h)^T & 1 + |\partial_t h|^2 \end{pmatrix},$$

where $(\nabla' h)^T$ is the transpose of the column vector $\nabla' h$. By elementary row operations we have

$$\det A = \det \begin{pmatrix} I_{n-1} + \{1 - |\partial_t h|^2 / (1 + |\partial_t h|^2)\} \nabla' h \otimes \nabla' h & 0 \\ \partial_t h (\nabla' h)^T & 1 + |\partial_t h|^2 \end{pmatrix}$$
$$= (1 + |\partial_t h|^2) \det \left[I_{n-1} + \left(1 - \frac{|\partial_t h|^2}{1 + |\partial_t h|^2}\right) \nabla' h \otimes \nabla' h \right]$$
$$= (1 + |\partial_t h|^2) \left\{ 1 + \left(1 - \frac{|\partial_t h|^2}{1 + |\partial_t h|^2}\right) |\nabla' h|^2 \right\}$$
$$= \left(1 + \frac{|\partial_t h|^2}{1 + |\nabla' h|^2}\right) (1 + |\nabla' h|^2).$$

Hence, by (2.B.2),

$$\det A(s,t) = (1 + |v_{\Gamma}^{N}(\mu_{t}(s),t)|^{2})(1 + |\nabla' h(s,t)|^{2}).$$

Substituting this for the right-hand side of (2.B.4) and applying Fubini's theorem, we get the right-hand side of (2.B.3) and thus conclude that (2.B.1) holds.

Next we give complete proofs of Lemma 2.4.1 and Lemma 2.4.3. Before starting to prove, let us construct a partition of unity of $\Gamma(t)$ by that of Γ_0 . Since Γ_0 is compact, we can take a finite family $\{U_k\}_{k=1}^N$ of open sets in \mathbb{R}^{n-1} and smooth local parametrizations $\mu_0^k \colon U_k \to \Gamma_0$, k = 1, ..., N such that $\{\mu_0^k(U_k)\}_{k=1}^N$ is an open covering of Γ_0 . Let $\{\psi_0^k\}_{k=1}^N$ be a partition of unity of Γ_0 subordinate to the covering $\{\mu_0^k(U_k)\}_{k=1}^N$. For k = 1, ..., N and $t \in [0, T]$ we set

$$\mu_t^k(s) := \Phi(\mu_0^k(s), t), \quad s \in U_k, \quad \psi_t^k := \psi_0^k \circ \mu_0^k \circ (\mu_t^k)^{-1}, \tag{2.B.5}$$

where $\Phi(\cdot, t) \colon \Gamma_0 \to \Gamma(t)$ is the flow map of V_{Γ} (see Section 2.2). Then for each $k = 1, \ldots, N$ the mapping $\mu_t^k \colon U_k \to \Gamma(t)$ is a local parametrization of $\Gamma(t)$ and $\{\mu_t^k(U_k)\}_{k=1}^N$ is an open covering of $\Gamma(t)$. Moreover, $\{\psi_t^k\}_{k=1}^N$ is a partition of unity of $\Gamma(t)$ subordinate to $\{\mu_t^k(U_k)\}_{k=1}^N$. We use these partitions of unity to localize integrals over $\Gamma(t)$.

Proof of Lemma 2.4.1. Let V be a function on $\Gamma_0 \times (0,T)$ and v := LV. Our goal is to show

$$c_1 \|V(t)\|_{L^2(\Gamma_0)} \le \|v(t)\|_{L^2(\Gamma(t))} \le c_2 \|V(t)\|_{L^2(\Gamma_0)},$$

$$c_1 \|\nabla_{\Gamma_0} V(t)\|_{L^2(\Gamma_0)} \le \|\nabla_{\Gamma(t)} v(t)\|_{L^2(\Gamma(t))} \le c_2 \|\nabla_{\Gamma_0} V(t)\|_{L^2(\Gamma_0)}$$

for all $t \in (0,T)$ with some positive constants c_1 , c_2 independent of t. These inequalities yield $c_1 ||V||_{\widehat{H}_T} \leq ||v||_{H_T} \leq c_2 ||V||_{\widehat{H}_T}$, which means that L is an isomorphism between \widehat{H}_T and H_T . By a localization argument with the partitions of unity given by (2.B.5), it is sufficient to show that

$$c_{1} \int_{\mu_{0}(Q)} |V(t)|^{2} d\mathcal{H}^{n-1} \leq \int_{\mu_{t}(Q)} |v(t)|^{2} d\mathcal{H}^{n-1} \leq c_{2} \int_{\mu_{0}(Q)} |V(t)|^{2} d\mathcal{H}^{n-1}, \qquad (2.B.6)$$

$$c_{1} \int_{\mu_{0}(Q)} |\nabla_{\Gamma_{0}} V(t)|^{2} d\mathcal{H}^{n-1} \leq \int_{\mu_{t}(Q)} |\nabla_{\Gamma(t)} v(t)|^{2} d\mathcal{H}^{n-1} \leq c_{2} \int_{\mu_{0}(Q)} |\nabla_{\Gamma_{0}} V(t)|^{2} d\mathcal{H}^{n-1} \qquad (2.B.7)$$

for all $t \in (0,T)$ and all V supported in $\mu_0(Q) \times (0,T)$. Here $\mu_0: U \to \Gamma_0$ be a smooth local parametrization of Γ_0 with an open set U in \mathbb{R}^{n-1} , Q is a compact subset of U, and $\mu_t: U \to \Gamma(t)$ is the local parametrization of $\Gamma(t)$ given by $\mu_t(s) := \Phi(\mu_0(s), t)$ for $s \in U$. Note that in this case v = LV is supported in $\bigcup_{t \in (0,T)} \mu_t(Q) \times \{t\}$. Let $\theta_t = (\theta_{t,ij})_{i,j}$ be a matrix given by

$$\theta_{t,ij}(s) := \frac{\partial \mu_t}{\partial s_i}(s) \cdot \frac{\partial \mu_t}{\partial s_j}(s), \quad (s,t) \in U \times [0,T], \ i,j = 1, \dots, n-1,$$
(2.B.8)

and $\theta_t^{-1} = (\theta_t^{ij})_{ij}$ be the inverse matrix of θ_t . By the definition of integrals over hypersurfaces,

$$\int_{\mu_0(Q)} |V(Y,t)|^2 d\mathcal{H}^{n-1}(Y) = \int_Q |V(\mu_0(s),t)|^2 \sqrt{\det \theta_0(s)} \, ds,$$
$$\int_{\mu_t(Q)} |v(y,t)|^2 d\mathcal{H}^{n-1}(y) = \int_Q |v(\mu_t(s),t)|^2 \sqrt{\det \theta_t(s)} \, ds.$$

Since $\sqrt{\det \theta_t(s)}$ is continuous and does not vanish as a function of (s, t) on the compact set $Q \times [0, T]$, there is a constant c > 0 such that

$$c^{-1} \le \sqrt{\det \theta_t(s)} \le c \quad \text{for all} \quad (s,t) \in Q \times [0,T].$$
 (2.B.9)

Moreover, by the definitions of L and μ_t ,

$$v(\mu_t(s), t) = V(\Phi^{-1}(\mu_t(s), t), t) = V(\Phi^{-1}(\Phi(\mu_0(s), t), t), t) = V(\mu_0(s), t)$$
(2.B.10)

for all $(s,t) \in U \times [0,T]$. Hence (2.B.6) follows. Similarly, by (2.B.9) and the equality

$$\int_{\mu_0(Q)} |\nabla_{\Gamma_0} V(Y,t)|^2 \, d\mathcal{H}^{n-1}(Y) = \int_Q |\nabla_{\Gamma_0} V(\mu_0(s),t)|^2 \sqrt{\det \theta_0(s)} \, ds,$$
$$\int_{\mu_t(Q)} |\nabla_{\Gamma(t)} v(y,t)|^2 \, d\mathcal{H}^{n-1}(y) = \int_Q |\nabla_{\Gamma(t)} v(\mu_t(s),t)|^2 \sqrt{\det \theta_t(s)} \, ds,$$

it is sufficient for (2.B.7) to show that

$$c_1 |\nabla_{\Gamma_0} V(\mu_0(s), t)|^2 \le |\nabla_{\Gamma(t)} v(\mu_t(s), t)|^2 \le c_2 |\nabla_{\Gamma_0} V(\mu_0(s), t)|^2$$
(2.B.11)

for all $(s,t) \in Q \times [0,T]$. The tangential gradients $\nabla_{\Gamma_0} V$ and $\nabla_{\Gamma(t)} v$ are locally expressed as (see [6, Section 2.1 and Section 2.2] for example)

$$\nabla_{\Gamma_0} V(\mu_0(s), t) = \sum_{i,j=1}^{n-1} \theta_0^{ij}(s) \frac{\partial}{\partial s_j} (V(\mu_0(s), t)) \frac{\partial \mu_0}{\partial s_i}(s),$$
$$\nabla_{\Gamma(t)} v(\mu_t(s), t) = \sum_{i,j=1}^{n-1} \theta_t^{ij}(s) \frac{\partial}{\partial s_j} (v(\mu_t(s), t)) \frac{\partial \mu_t}{\partial s_i}(s)$$

for $(s,t) \in U \times [0,T]$ and their Euclidean norms are

$$\begin{aligned} |\nabla_{\Gamma_0} V(\mu_0(s), t)|^2 &= \sum_{i,j=1}^{n-1} \theta_0^{ij}(s) \frac{\partial}{\partial s_i} (V(\mu_0(s), t)) \frac{\partial}{\partial s_j} (V(\mu_0(s), t)), \\ |\nabla_{\Gamma(t)} v(\mu_t(s), t)|^2 &= \sum_{i,j=1}^{n-1} \theta_t^{ij}(s) \frac{\partial}{\partial s_i} (v(\mu_t(s), t)) \frac{\partial}{\partial s_j} (v(\mu_t(s), t)). \end{aligned}$$

Then, by (2.B.10), it is sufficient for (2.B.11) to show

$$c_1 \theta_0^{-1}(s) a \cdot a \le \theta_t^{-1}(s) a \cdot a \le c_2 \theta_0^{-1}(s) a \cdot a \quad \text{for all} \quad (s, t, a) \in Q \times [0, T] \times \mathbb{R}^{n-1}.$$
 (2.B.12)

To this end, we consider a real-valued function

$$F(s,t,a) := \theta_t^{-1}(s)a \cdot a, \quad (s,t,a) \in Q \times [0,T] \times \mathbb{R}^{n-1}.$$

It is continuous on $Q \times [0,T] \times \mathbb{R}^{n-1}$ and satisfies $F(s,t,a) = |B(s,t,a)|^2$, where

$$B(s,t,a) := \sum_{i=1}^{n-1} b_i(s,t,a) \frac{\partial \mu_t}{\partial s_i}(s), \quad b = (b_1, \dots, b_{n-1}) := \theta_t^{-1}(s)a.$$

For $a \neq 0$ we have $b \neq 0$ and thus $B \neq 0$. Hence F does not vanish on the compact set $Q \times [0,T] \times S^{n-2}$, where S^{n-2} is the unit sphere in \mathbb{R}^{n-1} . From this fact and the continuity of F there is a constant c > 0 such that $c^{-1} \leq F(s,t,a) \leq c$ for all $(s,t,a) \in Q \times [0,T] \times S^{n-2}$ and thus

$$c^{-1}|a|^2 \le \theta_t^{-1}(s)a \cdot a \le c|a|^2$$
 for all $(s,t,a) \in Q \times [0,T] \times \mathbb{R}^{n-1}$.

This inequality yields (2.B.12) and we conclude that (2.B.7) is valid.

Proof of Lemma 2.4.3. First we give transformation formulas of integrals over Γ_0 and $\Gamma(t)$. Let U be an open set in \mathbb{R}^{n-1} and $\mu_0: U \to \Gamma_0$ be a smooth local parametrization of Γ_0 . Moreover, let $\mu_t: U \to \Gamma(t)$ be the local parametrization of $\Gamma(t)$ given by $\mu_t(s) := \Phi(\mu_0(s), t)$. We set

$$\Lambda(\mu_0(s),t) := \sqrt{\frac{\det \theta_t(s)}{\det \theta_0(s)}}, \quad \lambda(\mu_t(s),t) := \sqrt{\frac{\det \theta_0(s)}{\det \theta_t(s)}}, \quad (s,t) \in U \times [0,T],$$

where $\theta_t = (\theta_{t,ij})_{ij}$ is given by (2.B.8). We can show that the right-hand sides of the above definitions are independent of the choice of the local parametrization μ_0 . From this fact and the smoothness assumption on Φ , the functions Λ and λ are well-defined and smooth on the compact manifolds $\Gamma_0 \times [0, T]$ and $\overline{S_T}$, respectively. In particular, they are bounded on $\Gamma_0 \times [0, T]$ and $\overline{S_T}$ along with their derivatives. Moreover, by a localization argument with the partitions of unity given by (2.B.5), we get the integral transformation formulas

$$\int_{\Gamma(t)} v(y,t) \, d\mathcal{H}^{n-1}(y) = \int_{\Gamma_0} V(Y,t) \Lambda(Y,t) \, d\mathcal{H}^{n-1}(Y), \tag{2.B.13}$$

$$\int_{\Gamma_0} V(Y,t) \, d\mathcal{H}^{n-1}(Y) = \int_{\Gamma(t)} v(y,t)\lambda(y,t) \, d\mathcal{H}^{n-1}(y) \tag{2.B.14}$$

for all functions V on $\Gamma_0 \times (0,T)$ and all $t \in (0,T)$, where v = LV.

Now let us prove the statement of Lemma 2.4.3. For $V \in \widehat{W}_T$ we set v := LV. Then Lemma 2.4.1 yields $v \in H_T$ and $\|v\|_{H_T} \leq c \|V\|_{\widehat{H}_T}$. We next show that $\partial^{\bullet} v \in H'_T$ and $\|\partial^{\bullet} v\|_{H'_T} \leq c \|V\|_{\widehat{W}_T}$. Let $\psi \in C_0^1(S_T)$. Then $\Psi := L^{-1}\psi$ is in $C_0^1(\Gamma_0 \times (0,T))$ and $\partial^{\bullet} \psi(\Phi(Y,t),t) = \partial_t \Psi(Y,t)$ for all $Y \in \Gamma_0$. Hence (2.B.13) yields

$$\begin{split} \langle \partial^{\bullet} v, \psi \rangle_{T} &= -\int_{0}^{T} \int_{\Gamma(t)} \left(v \, \partial^{\bullet} \psi + v \psi \operatorname{div}_{\Gamma(t)} V_{\Gamma} \right) d\mathcal{H}^{n-1} dt \\ &= -\int_{0}^{T} \int_{\Gamma_{0}} \left(V \, \partial_{t} \Psi + V \Psi F \right) \Lambda \, d\mathcal{H}^{n-1} \, dt, \end{split}$$

where $F := L^{-1}(\operatorname{div}_{\Gamma(t)}V_{\Gamma}) \in C^{\infty}(\Gamma_0 \times [0,T])$. Moreover, since $\Psi \Lambda \in C_0^1(\Gamma_0 \times (0,T))$,

$$-\int_0^T \int_{\Gamma_0} V\Lambda \,\partial_t \Psi \,d\mathcal{H}^{n-1} \,dt = [\partial_t V, \Psi\Lambda]_T + \int_0^T \int_{\Gamma_0} V\Psi \,\partial_t \Lambda \,d\mathcal{H}^{n-1} \,dt$$

by the definition of the weak time derivative $\partial_t V$. From these formulas and the boundedness of F and Λ on $\Gamma_0 \times (0, t)$ along with their derivatives, it follows that

$$\begin{aligned} |\langle \partial^{\bullet} v, \psi \rangle_{T}| &= \left| [\partial_{t} V, \Psi \Lambda]_{T} + \int_{0}^{T} \int_{\Gamma_{0}} (V \Psi \, \partial_{t} \Lambda - V \Psi \Lambda F) \, d\mathcal{H}^{n-1} \, dt \right| \\ &\leq c(\|\partial_{t} V\|_{\widehat{H}_{T}} \|\Psi \Lambda\|_{\widehat{H}_{T}} + \|V\|_{\widehat{H}_{T}} \|\Psi\|_{\widehat{H}_{T}}) \leq c \|V\|_{\widehat{W}_{T}} \|\psi\|_{H_{T}} \end{aligned}$$

with a constant c > 0 independent of V and ψ , which implies $\partial^{\bullet} v \in H'_T$ and $\|\partial^{\bullet} v\|_{H'_T} \le c\|V\|_{\widehat{W}_T}$. Hence v = LV is in W_T and $\|v\|_{W_T} \le c\|V\|_{\widehat{W}_T}$ for every $V \in \widehat{W}_T$.

Similarly, by (2.B.14) and the smoothness of λ on $\overline{S_T}$ we can show that $V := L^{-1}v$ is in \widehat{W}_T and $\|V\|_{\widehat{W}_T} \leq c \|v\|_{W_T}$ for every $v \in W_T$. Hence L is an isomorphism between \widehat{W}_T and W_T .

2.C Calculations involving the differential geometry of tubular neighborhoods

The purpose of this appendix is to show detailed calculations in the proofs of Lemma 2.5.5, Lemma 2.5.6, and Lemma 2.5.11. We fix $t \in [0, T]$ and omit it until the end of the proof of Lemma 2.5.6.

The proofs of Lemma 2.5.5 and Lemma 2.5.6 involve calculations of the usual gradient in N and the tangential gradient on Γ under a local coordinate system. Let $\mu: U \to \Gamma$ be a local parametrization of Γ with an open set U in \mathbb{R}^{n-1} . We set

$$\theta_{ij}(s) := \frac{\partial \mu}{\partial s_i}(s) \cdot \frac{\partial \mu}{\partial s_j}(s), \quad s \in U, \, i, j = 1, \dots, n-1.$$

Then, the tangential gradient of a function v on Γ is locally expressed as

$$\nabla_{\Gamma} v(y) = \sum_{i,j=1}^{n-1} \theta^{ij}(s) \frac{\partial \widetilde{v}}{\partial s_j}(s) \frac{\partial \mu}{\partial s_i}(s), \quad y = \mu(s) \in \mu(U), \tag{2.C.1}$$

where $\widetilde{v}(s) := v(\mu(s))$ and $\theta^{-1} = (\theta^{ij})_{i,j}$ denotes the inverse matrix of $\theta = (\theta_{ij})_{i,j}$. We define a mapping $M : U \times (-\delta, \delta) \to N$ as $M(s, \rho) := \mu(s) + \rho \nu(\mu(s))$ for $(s, \rho) \in U \times (-\delta, \delta)$ and set

$$\Theta_{ij}(s,\rho) := \frac{\partial M}{\partial s_i}(s,\rho) \cdot \frac{\partial M}{\partial s_j}(s,\rho), \quad (s,\rho) \in U \times (-\delta,\delta), \, i,j = 1, \dots, n,$$

where $s_n := \rho$. Then the gradient (in \mathbb{R}^n) of a function u on N is locally expressed as

$$\nabla u(x) = \sum_{i,j=1}^{n} \Theta^{ij}(s,\rho) \frac{\partial \widetilde{u}}{\partial s_j}(s,\rho) \frac{\partial M}{\partial s_i}(s,\rho), \quad x = M(s,\rho) \in M(U \times (-\delta,\delta)), \quad (2.C.2)$$

where $\widetilde{u}(s,\rho) := u(M(s,\rho))$ and $\Theta^{-1} = (\Theta^{ij})_{i,j}$ is the inverse matrix of $\Theta = (\Theta_{ij})_{i,j}$.

Let v be a function on Γ and \overline{v} be its constant extension in the normal direction of Γ . Then their local representations $\tilde{v} := v \circ \mu$ and $\tilde{\overline{v}} := \overline{v} \circ M$ satisfy

$$\overline{v}(s,\rho) = \overline{v}(p(M(s,\rho))) = v(\mu(s)) = \widetilde{v}(s), \quad (s,\rho) \in U \times (-\delta,\delta).$$

Hereafter we use this fact without mention.

Proof of Lemma 2.5.5. Let $v \in H^1(\Gamma)$. Our goal is to show the inequalities

$$|\nabla \overline{v}(y + \rho\nu(y))| \le c |\nabla_{\Gamma}v(y)|, \quad |\nabla \overline{v}(y + \rho\nu(y)) - \nabla_{\Gamma}v(y)| \le c\varepsilon |\nabla_{\Gamma}v(y)|$$
(2.C.3)

for all $y \in \Gamma$ and $\rho \in (\varepsilon g_0(y), \varepsilon g_1(y))$ with a constant c > 0 independent of y, ρ , and ε . For each fixed $y_0 \in \Gamma$, by a rotation of coordinates we can take an open set U in \mathbb{R}^{n-1} and a local parametrization $\mu: U \to \Gamma$ such that $y_0 = \mu(s_0)$ with $s_0 \in U$ and μ is of the form $\mu(s) = (s, f(s))$ with a smooth function f on U satisfying

$$\nabla' f(s_0) = 0, \quad (\nabla')^2 f(s_0) = \operatorname{diag}[\kappa_1, \dots, \kappa_{n-1}], \qquad (2.C.4)$$

where ∇' is the gradient in $s \in \mathbb{R}^{n-1}$ and $\kappa_i := \kappa_i(y_0)$ for $i = 1, \ldots, n-1$ (see [11, Section 14.6]). We set the direction of $\nu(y_0)$ in the positive direction of the x_n -axis to get

$$\nu(\mu(s)) = \frac{(-\nabla' f(s), 1)}{\sqrt{1 + |\nabla' f(s)|^2}}, \quad s \in U.$$

Then we have $\nu(y_0) = \nu(\mu(s_0)) = e_n$ and

$$\frac{\partial \mu}{\partial s_i}(s_0) = e_i, \quad \frac{\partial}{\partial s_i} \left(\nu(\mu(s)) \right) \Big|_{s=s_0} = -\kappa_i e_i, \quad i = 1, \dots, n-1$$
(2.C.5)

by (2.C.4), where $\{e_i\}_{i=1}^n$ is the standard basis of \mathbb{R}^n . This equality yields

$$\frac{\partial M}{\partial s_i}(s_0,\rho) = (1-\rho\kappa_i)e_i, \quad i = 1,\dots, n-1, \quad \frac{\partial M}{\partial \rho}(s_0,\rho) = \nu(\mu(s_0)) = e_n.$$
(2.C.6)

Hence we have $\theta(s_0) = I_{n-1}$, $\Theta(s_0, \rho) = \text{diag}[(1 - \rho \kappa_1)^2, \dots, (1 - \rho \kappa_{n-1})^2, 1]$, and

$$\theta^{-1}(s_0) = I_{n-1}, \quad \Theta^{-1}(s_0, \rho) = \text{diag}[(1 - \rho \kappa_1)^{-2}, \dots, (1 - \rho \kappa_{n-1})^{-2}, 1].$$
 (2.C.7)

Applying (2.C.5), (2.C.6), and (2.C.7) to (2.C.1) and (2.C.2) with $u = \overline{v}$, we obtain

$$\nabla_{\Gamma} v(y_0) = \sum_{i=1}^{n-1} \frac{\partial \widetilde{v}}{\partial s_i}(s_0) e_i, \quad \nabla \overline{v}(y_0 + \rho \nu(y_0)) = \sum_{i=1}^{n-1} (1 - \rho \kappa_i)^{-1} \frac{\partial \widetilde{v}}{\partial s_i}(s_0) e_i$$

and thus (2.5.3) implies that

$$|\nabla \overline{v}(y_0 + \rho \nu(y_0))|^2 = \sum_{i=1}^{n-1} (1 - \rho \kappa_i)^{-2} \left(\frac{\partial \widetilde{v}}{\partial s_i}(s_0)\right)^2 \le c \sum_{i=1}^{n-1} \left(\frac{\partial \widetilde{v}}{\partial s_i}(s_0)\right)^2 = |\nabla_{\Gamma} v(y_0)|^2,$$

which yields the first inequality of (2.C.3) with y replaced by y_0 . Moreover, by (2.5.3) we have

$$|(1 - \rho \kappa_i)^{-1} - 1| = |\rho \kappa_i (1 - \rho \kappa_i)^{-1}| \le c\varepsilon$$

for all $\rho \in (\varepsilon g_0(y_0), \varepsilon g_1(y_0))$ and $i = 1, \dots, n-1$ and thus

$$|\nabla \overline{v}(y_0 + \rho \nu(y_0)) - \nabla_{\Gamma} v(y_0)|^2 = \sum_{i=1}^{n-1} \{(1 - \rho \kappa_i)^{-1} - 1\}^2 \left(\frac{\partial \widetilde{v}}{\partial s_i}(s_0)\right)^2 \le c\varepsilon^2 |\nabla_{\Gamma} v(y_0)|^2.$$

Hence the second inequality of (2.C.3) with y replaced by y_0 is valid.

To prove Lemma 2.5.6, we need a differentiation formula of the average operator under a local coordinate system. Let U be an open set in \mathbb{R}^{n-1} and $\mu: U \to \Gamma$ be a local parametrization of Γ . The weighted average of a function u on Ω_{ε} is locally expressed as

$$\widetilde{M_{\varepsilon}u}(s) = \frac{1}{\varepsilon \widetilde{g}(s)} \int_{\varepsilon \widetilde{g}_0(s)}^{\varepsilon \widetilde{g}_1(s)} \widetilde{u}(s,\rho) \widetilde{J}(s,\rho) \, d\rho, \quad s \in U,$$
(2.C.8)

where $\widetilde{M_{\varepsilon}u}(s) = M_{\varepsilon}u(\mu(s)), \ \widetilde{u}(s,\rho) = u(M(s,\rho))$, and

$$\widetilde{J}(s,\rho) := J(\mu(s),\rho) = \prod_{i=1}^{n-1} \{1 - \rho \kappa_i(\mu(s))\}.$$
(2.C.9)

Lemma 2.C.1. Let $u \in H^1(\Omega_{\varepsilon})$. Then

$$\frac{\partial \widetilde{M_{\varepsilon}u}}{\partial s_{i}}(s) = \frac{1}{\varepsilon \widetilde{g}(s)} \int_{\varepsilon \widetilde{g}_{0}(s)}^{\varepsilon \widetilde{g}_{1}(s)} \left\{ \frac{\partial \widetilde{u}}{\partial s_{i}}(s,\rho) \widetilde{J}(s,\rho) + \widetilde{u}(s,\rho) \frac{\partial \widetilde{J}}{\partial s_{i}}(s,\rho) \right\} d\rho
+ \frac{1}{\varepsilon \widetilde{g}(s)} \int_{\varepsilon \widetilde{g}_{0}(s)}^{\varepsilon \widetilde{g}_{1}(s)} \left\{ \frac{\partial \widetilde{u}}{\partial \rho}(s,\rho) \widetilde{J}(s,\rho) + \widetilde{u}(s,\rho) \frac{\partial \widetilde{J}}{\partial \rho}(s,\rho) \right\} \chi_{i}(s,\rho) d\rho \quad (2.C.10)$$

for all $s \in U$ and $i = 1, \ldots, n - 1$, where

$$\chi_i(s,\rho) := \frac{1}{\widetilde{g}(s)} \left\{ (\rho - \varepsilon \widetilde{g}_0(s)) \frac{\partial \widetilde{g}_1}{\partial s_i}(s) + (\varepsilon \widetilde{g}_1(s) - \rho) \frac{\partial \widetilde{g}_0}{\partial s_i}(s) \right\}.$$
 (2.C.11)

Proof. For simplicity, we set $\partial_i = \partial/\partial s_i$ and $\partial_\rho = \partial/\partial \rho$. For each $i = 1, \ldots, n-1$, we differentiate both sides of (2.C.8) with respect to s_i to get

$$\partial_{i}\widetilde{M_{\varepsilon}u} = \frac{I}{\varepsilon\widetilde{g}} - \frac{\partial_{i}\widetilde{g}}{\varepsilon(\widetilde{g})^{2}} \int_{\varepsilon\widetilde{g}_{0}}^{\varepsilon\widetilde{g}_{1}} \widetilde{u}\widetilde{J}\,d\rho + \frac{1}{\varepsilon\widetilde{g}} \int_{\varepsilon\widetilde{g}_{0}}^{\varepsilon\widetilde{g}_{1}} \{(\partial_{i}\widetilde{u})\widetilde{J} + \widetilde{u}(\partial_{i}\widetilde{J})\}\,d\rho,\tag{2.C.12}$$

where I = I(s) is given by

$$I(s) := \varepsilon \partial_i \widetilde{g}_1(s) \widetilde{u}(s, \varepsilon \widetilde{g}_1(s)) \widetilde{J}(s, \varepsilon \widetilde{g}_1(s)) - \varepsilon \partial_i \widetilde{g}_0(s) \widetilde{u}(s, \varepsilon \widetilde{g}_0(s)) \widetilde{J}(s, \varepsilon \widetilde{g}_0(s)).$$

Since $I = \left[\widetilde{u}(\rho)\widetilde{J}(\rho)\chi_i(\rho)\right]_{\rho=\widetilde{e}\widetilde{g}_0}^{\widetilde{e}\widetilde{g}_1} = \int_{\widetilde{e}\widetilde{g}_0}^{\widetilde{e}\widetilde{g}_1} \partial_{\rho}(\widetilde{u}\widetilde{J}\chi_i) \, d\rho$ and $\partial_{\rho}\chi_i = \partial_i \widetilde{g}/\widetilde{g}$, we have

$$\frac{I}{\varepsilon \widetilde{g}} = \frac{\partial_i \widetilde{g}}{\varepsilon (\widetilde{g})^2} \int_{\varepsilon \widetilde{g}_0}^{\varepsilon \widetilde{g}_1} \widetilde{u} \widetilde{J} \, d\rho + \frac{1}{\varepsilon \widetilde{g}} \int_{\varepsilon \widetilde{g}_0}^{\varepsilon \widetilde{g}_1} \{ (\partial_\rho \widetilde{u}) \widetilde{J} + \widetilde{u} (\partial_\rho \widetilde{J}) \} \chi_i \, d\rho.$$
(2.C.13)

Substituting (2.C.13) for (2.C.12), we obtain (2.C.10).

Proof of Lemma 2.5.6. As in the proof of Lemma 2.C.1, we write $\partial_i = \partial/\partial s_i$ and $\partial_{\rho} = \partial/\partial \rho$. Let $u \in C^{\infty}(\Omega_{\varepsilon}) \cap H^1(\Omega_{\varepsilon}), \varphi \in H^1(\Gamma)$, and

$$I(y) := \int_{\varepsilon g_0(y)}^{\varepsilon g_1(y)} (\nabla u)^{\sharp}(y,\rho) \cdot (\nabla \overline{\varphi})^{\sharp}(y,\rho) J(y,\rho) \, d\rho - \varepsilon g(y) \nabla_{\Gamma} M_{\varepsilon} u(y) \cdot \nabla_{\Gamma} \varphi(y).$$

Here we used the notation (2.5.6). Our goal is to show

$$|I(y)| \le c\varepsilon |\nabla_{\Gamma}\varphi(y)| \int_{\varepsilon g_0(y)}^{\varepsilon g_1(y)} (|u^{\sharp}(y,\rho)| + |(\nabla u)^{\sharp}(y,\rho)|) \, d\rho \tag{2.C.14}$$

for all $y \in \Gamma$ with a constant c > 0 independent of y and ε . As in the proof of Lemma 2.5.5, we fix $y_0 \in \Gamma$ and take a local parametrization $\mu(s) = (s, f(s))$ of Γ near $y_0 = \mu(s_0), s_0 \in U$, where U is an open set in \mathbb{R}^{n-1} and f is a smooth function on U satisfying (2.C.4). We set the direction of $\nu(y_0)$ in the positive direction of the x_n -axis. Then by (2.C.5), (2.C.6), and (2.C.7) we have

$$(\nabla u)^{\sharp}(y_{0},\rho) = \sum_{i=1}^{n-1} (1-\rho\kappa_{i})^{-1} \partial_{i} \widetilde{u}(s_{0},\rho) e_{i} + \partial_{\rho} \widetilde{u}(s_{0},\rho) e_{n}, \quad \nabla_{\Gamma} M_{\varepsilon} u(y_{0}) = \sum_{i=1}^{n-1} \partial_{i} \widetilde{M_{\varepsilon}} u(s_{0}) e_{i},$$
$$(\nabla \overline{\varphi})^{\sharp}(y_{0},\rho) = \sum_{i=1}^{n-1} (1-\rho\kappa_{i})^{-1} \partial_{i} \widetilde{\varphi}(s_{0}) e_{i}, \quad \nabla_{\Gamma} \varphi(y_{0}) = \sum_{i=1}^{n-1} \partial_{i} \widetilde{\varphi}(s_{0}) e_{i},$$

where $\{e_i\}_{i=1}^n$ is the standard basis of \mathbb{R}^n and $\kappa_i := \kappa_i(y_0), i = 1, \ldots, n-1$. Hereafter we omit the variables ρ and s_0 unless we need to specify them. The above equality yields

$$(\nabla u)^{\sharp}(y_{0},\rho) \cdot (\nabla\overline{\varphi})^{\sharp}(y_{0},\rho) = \sum_{i=1}^{n-1} (1-\rho\kappa_{i})^{-2} \partial_{i}\widetilde{u} \,\partial_{i}\widetilde{\varphi}, \qquad (2.C.15)$$
$$\varepsilon g(y_{0}) \nabla_{\Gamma} M_{\varepsilon} u(y_{0}) \cdot \nabla_{\Gamma} \varphi(y_{0}) = \sum_{i=1}^{n-1} \varepsilon \widetilde{g} \left(\partial_{i} \widetilde{M_{\varepsilon} u} \right) \partial_{i}\widetilde{\varphi}.$$

Moreover, (2.C.10) implies that

$$\varepsilon \widetilde{g} \Big(\partial_i \widetilde{M_{\varepsilon} u} \Big) = \int_{\varepsilon \widetilde{g}_0}^{\varepsilon \widetilde{g}_1} \{ (\partial_i \widetilde{u}) \widetilde{J} + \widetilde{u} (\partial_i \widetilde{J}) + (\partial_\rho \widetilde{u}) \widetilde{J} \chi_i + \widetilde{u} (\partial_\rho \widetilde{J}) \chi_i \} d\rho,$$

where χ_i is given by (2.C.11), and thus

$$\begin{split} \varepsilon g(y_0) \nabla_{\Gamma} M_{\varepsilon} u(y_0) \cdot \nabla_{\Gamma} \varphi(y_0) &= \int_{\varepsilon \widetilde{g}_0}^{\varepsilon \widetilde{g}_1} \widetilde{J} \sum_{i=1}^{n-1} \partial_i \widetilde{u} \, \partial_i \widetilde{\varphi} \, d\rho + \int_{\varepsilon \widetilde{g}_0}^{\varepsilon \widetilde{g}_1} \widetilde{u} \sum_{i=1}^{n-1} \partial_i \widetilde{J} \, \partial_i \widetilde{\varphi} \, d\rho \\ &+ \int_{\varepsilon \widetilde{g}_0}^{\varepsilon \widetilde{g}_1} \{ (\partial_{\rho} \widetilde{u}) \widetilde{J} + \widetilde{u} (\partial_{\rho} \widetilde{J}) \} \sum_{i=1}^{n-1} \chi_i \, \partial_i \widetilde{\varphi} \, d\rho. \end{split}$$

From this equality and (2.C.15), we obtain $I(y_0) = I_1 + I_2 + I_3$ with

$$I_{1} = \int_{\varepsilon \widetilde{g}_{0}}^{\varepsilon \widetilde{g}_{1}} \widetilde{J} \sum_{i=1}^{n-1} \{ (1 - \rho \kappa_{i})^{-2} - 1 \} \partial_{i} \widetilde{u} \, \partial_{i} \widetilde{\varphi} \, d\rho,$$

$$I_{2} = -\int_{\varepsilon \widetilde{g}_{0}}^{\varepsilon \widetilde{g}_{1}} \widetilde{u} \sum_{i=1}^{n-1} \partial_{i} \widetilde{J} \, \partial_{i} \widetilde{\varphi} \, d\rho, \quad I_{3} = -\int_{\varepsilon \widetilde{g}_{0}}^{\varepsilon \widetilde{g}_{1}} \{ (\partial_{\rho} \widetilde{u}) \widetilde{J} + \widetilde{u} (\partial_{\rho} \widetilde{J}) \} \sum_{i=1}^{n-1} \chi_{i} \, \partial_{i} \widetilde{\varphi} \, d\rho.$$

Let us estimate these integrals. By the definition of \widetilde{J} (see (2.C.9)), we have

$$\nabla_{\Gamma} J(y_0,\rho) = \sum_{i=1}^{n-1} \partial_i \widetilde{J}(s_0,\rho) e_i, \quad \sum_{i=1}^{n-1} \partial_i \widetilde{J}(s_0,\rho) \partial_i \widetilde{\varphi}(s_0) = \nabla_{\Gamma} J(y_0,\rho) \cdot \nabla_{\Gamma} \varphi(y_0).$$

Hence I_2 is of the form

$$I_2 = -\int_{\varepsilon g_0(y_0)}^{\varepsilon g_1(y_0)} u^{\sharp}(y_0,\rho) \nabla_{\Gamma} J(y_0,\rho) \cdot \nabla_{\Gamma} \varphi(y_0) \, d\rho$$

and by applying (2.5.5) to the right-hand side we obtain

$$|I_2| \le c\varepsilon |\nabla_{\Gamma}\varphi(y_0)| \int_{\varepsilon g_0(y_0)}^{\varepsilon g_1(y_0)} |u^{\sharp}(y_0,\rho)| \, d\rho.$$
(2.C.16)

Next we estimate I_3 . By the definitions of \tilde{u} , \tilde{J} , and χ_i (see (2.C.9) and (2.C.11)),

$$\partial_{\rho}\widetilde{u}(s_{0},\rho) = \nu(y_{0}) \cdot (\nabla u)^{\sharp}(y_{0},\rho), \quad \partial_{\rho}\widetilde{J}(s_{0},\rho) = \partial_{\rho}J(y_{0},\rho),$$
$$\sum_{i=1}^{n-1} \chi_{i}(s_{0},\rho)\partial_{i}\widetilde{\varphi}(s_{0}) = \chi_{\varepsilon}(y_{0},\rho) \cdot \nabla_{\Gamma}\varphi(y_{0}),$$

where

$$\chi_{\varepsilon}(y_0,\rho) := \frac{(\rho - \varepsilon g_0(y_0))\nabla_{\Gamma} g_1(y_0) + (\varepsilon g_1(y_0) - \rho)\nabla_{\Gamma} g_0(y_0)}{g(y_0)}$$

Hence I_3 is of the form

$$I_{3} = -\int_{\varepsilon g_{0}(y_{0})}^{\varepsilon g_{1}(y_{0})} \chi_{\varepsilon}(y_{0},\rho) \cdot \nabla_{\Gamma}\varphi(y_{0}) \{\nu(y_{0}) \cdot (\nabla u)^{\sharp}(y_{0},\rho)J(y_{0},\rho) + u^{\sharp}(y_{0},\rho)\partial_{\rho}J(y_{0},\rho)\}\,d\rho.$$

Since $\nabla_{\Gamma} g_0$, $\nabla_{\Gamma} g_1$ are bounded and $g_1 - g_0 = g$,

$$|\chi_{\varepsilon}(y_0,\rho)| \leq \frac{|\nabla_{\Gamma}g_0(y_0)| + |\nabla_{\Gamma}g_1(y_0)|}{g(y_0)} \{(\rho - \varepsilon g_0(y_0)) + (\varepsilon g_1(y_0) - \rho)\} \leq c\varepsilon$$

for all $\rho \in (\varepsilon g_0(y_0), \varepsilon g_1(y_0))$. This inequality together with (2.5.4) and (2.5.5) yields

$$|I_3| \le c\varepsilon |\nabla_{\Gamma}\varphi(y_0)| \int_{\varepsilon g_0(y_0)}^{\varepsilon g_1(y_0)} (|u^{\sharp}(y_0,\rho)| + |(\nabla u)^{\sharp}(y_0,\rho)|) \, d\rho.$$
(2.C.17)

Let us estimate I_1 . For all $\rho \in (\varepsilon g_0(y_0), \varepsilon g_1(y_0))$ and $i = 1, \ldots, n-1$, we have

$$|(1 - \rho \kappa_i)^{-2} - 1| = |\rho \kappa_i (2 - \rho \kappa_i) (1 - \rho \kappa_i)^{-2}| \le c\varepsilon$$

by (2.5.3). From this inequality, Hölder's inequality, and (2.5.3),

$$\begin{aligned} \left|\sum_{i=1}^{n-1} \{(1-\rho\kappa_i)^{-2} - 1\} \partial_i \widetilde{u} \,\partial_i \widetilde{\varphi} \right| &\leq c \varepsilon \left(\sum_{i=1}^{n-1} (\partial_i \widetilde{u})^2 \right)^{1/2} \left(\sum_{i=1}^{n-1} (\partial_i \widetilde{\varphi})^2 \right)^{1/2} \\ &\leq c \varepsilon \left(\sum_{i=1}^{n-1} (1-\rho\kappa_i)^{-2} (\partial_i \widetilde{u})^2 \right)^{1/2} \left(\sum_{i=1}^{n-1} (\partial_i \widetilde{\varphi})^2 \right)^{1/2} \\ &\leq c \varepsilon |(\nabla u)^{\sharp}(y_0,\rho)| |\nabla_{\Gamma} \varphi(y_0)|. \end{aligned}$$

Using this inequality and (2.5.4) we obtain

$$|I_1| \le c\varepsilon |\nabla_{\Gamma}\varphi(y_0)| \int_{\varepsilon g_0(y_0)}^{\varepsilon g_1(y_0)} |(\nabla u)^{\sharp}(y_0,\rho)| \, d\rho.$$
(2.C.18)

By (2.C.16), (2.C.17), and (2.C.18) we conclude that (2.C.14) with y replaced by y_0 holds. \Box

Finally we give the complete proof of Lemma 2.5.11.

Proof of Lemma 2.5.11. Let $\Phi(\cdot, t) \colon \Gamma_0 \to \Gamma(t)$ be the flow map of V_{Γ} and $\Phi^{-1}(\cdot, t)$ be its inverse mapping (see Section 2.2). For $X \in N(0)$ and $t \in (0, T)$ we set

$$\Psi(X,t) := \Phi(p(X,0),t) + d(X,0)\nu(\Phi(p(X,0),t),t).$$
(2.C.19)

For each $t \in (0,T)$ the mapping $\Psi(\cdot,t) \colon N(0) \to N(t)$ is a bijection whose inverse is given by

$$\Psi^{-1}(x,t) := \Phi^{-1}(p(x,t),t) + d(x,t)\nu(\Phi^{-1}(p(x,t),t),0), \quad (x,t) \in N_T.$$

Let $\varphi \in C^1(S_T)$ and $\overline{\varphi}$ be its constant extension in the normal direction of $\Gamma(t)$. By the definition of $\overline{\varphi}$ and the formula $p(\Psi(X,t),t) = \Phi(p(X,0),t)$ we have

$$\overline{\varphi}(\Psi(X,t),t) = \varphi(\Phi(p(X,0),t),t), \quad (X,t) \in N(0) \times (0,T)$$

We differentiate both sides with respect to t. The time derivative of the left-hand side is

$$\partial_t \overline{\varphi}(\Psi(X,t),t) + \partial_t \Psi(X,t) \cdot \nabla \overline{\varphi}(\Psi(X,t),t).$$

On the other hand, the time derivative of the right-hand side is

$$\partial^{\bullet}\varphi(\Phi(p(X,0),t),t) = \partial^{\bullet}\varphi(p(\Psi(X,t),t),t)$$

by the definition of the strong material derivative (see (2.4.4)). Hence

$$\partial^{\bullet}\varphi(p(\Psi(X,t),t),t) = \partial_t \overline{\varphi}(\Psi(X,t),t) + \partial_t \Psi(X,t) \cdot \nabla \overline{\varphi}(\Psi(X,t),t)$$

for all $(X,t) \in N(0) \times (0,T)$. Substituting $\Psi^{-1}(x,t)$ for X in this equality we further get

$$\partial^{\bullet}\varphi(p(x,t),t) = \partial_t\overline{\varphi}(x,t) + \partial_t\Psi(\Psi^{-1}(x,t),t) \cdot \nabla\overline{\varphi}(x,t)$$
(2.C.20)

for all $(x,t) \in N_T$. Let us show

$$\partial_t \Psi(\Psi^{-1}(x,t),t) = V_{\Gamma}(p(x,t),t) + a(x,t), \qquad (2.C.21)$$

where a(x,t) is given by (2.5.17). We differentiate both sides of (2.C.19) with respect to t to get

$$\begin{split} \partial_t \Psi(X,t) &= \partial_t \Phi(p(X,0),t) \\ &\quad + d(X,0) \{ \partial_t \nu(\Phi(p(X,0),t),t) + \nabla \nu(\Phi(p(X,0),t),t) \partial_t \Phi(p(X,0),t) \} \end{split}$$

for $(X,t) \in N(0) \times (0,T)$. Moreover, since

$$d(X,0) = d(\Psi(X,t),t), \quad \Phi(p(X,0),t) = p(\Psi(X,t),t), \\ \partial_t \Phi(p(X,0),t) = V_{\Gamma}(\Phi(p(X,0),t),t) = V_{\Gamma}(p(\Psi(X,t),t),t),$$

it follows that

$$\partial_t \Psi(X,t) = V_{\Gamma}(p(\Psi(X,t),t),t) + d(\Psi(X,t),t)\{\partial_t \nu(p(\Psi(X,t),t),t) + \nabla \nu(p(\Psi(X,t),t),t)V_{\Gamma}(p(\Psi(X,t),t),t)\}.$$

Substituting $\Psi^{-1}(x,t)$ for X in this equality we obtain (2.C.21). Finally, the formula (2.5.16) follows from (2.C.20) and (2.C.21).

References

- A. Alphonse, C. M. Elliott, and B. Stinner, An abstract framework for parabolic PDEs on evolving spaces, Port. Math. 72 (2015), no. 1, 1–46.
- [2] A. Alphonse, C. M. Elliott, and B. Stinner, On some linear parabolic PDEs on moving hypersurfaces, Interfaces Free Bound. 17 (2015), no. 2, 157–187.
- [3] P. Cermelli, E. Fried, and M. E. Gurtin, Transport relations for surface integrals arising in the formulation of balance laws for evolving fluid interfaces, J. Fluid Mech. 544 (2005), 339–351.

- [4] G. Dziuk and C. M. Elliott, Finite elements on evolving surfaces, IMA J. Numer. Anal. 27 (2007), no. 2, 262–292.
- [5] G. Dziuk and C. M. Elliott, L²-estimates for the evolving surface finite element method, Math. Comp. 82 (2013), no. 281, 1–24.
- [6] G. Dziuk and C. M. Elliott, Finite element methods for surface PDEs, Acta Numer. 22 (2013), 289–396.
- [7] C. M. Elliott and H. Fritz, Time-periodic solutions of advection-diffusion equations on moving hypersurfaces, SIAM J. Math. Anal. 47 (2015), no. 3, 1693–1718.
- [8] C. M. Elliott and B. Stinner, Analysis of a diffuse interface approach to an advection diffusion equation on a moving surface, Math. Models Methods Appl. Sci. 19 (2009), no. 5, 787–802.
- [9] C. M. Elliott, B. Stinner, V. Styles, and R. Welford, Numerical computation of advection and diffusion on evolving diffuse interfaces, IMA J. Numer. Anal. 31 (2011), no. 3, 786–812.
- [10] L. C. Evans, Partial differential equations, 2nd ed., Graduate Studies in Mathematics, vol. 19, American Mathematical Society, Providence, RI, 2010.
- [11] D. Gilbarg and N. S. Trudinger, *Elliptic partial differential equations of second order*, Classics in Mathematics, Springer-Verlag, Berlin, 2001. Reprint of the 1998 edition.
- [12] J. K. Hale and G. Raugel, A damped hyperbolic equation on thin domains, Trans. Amer. Math. Soc. 329 (1992), no. 1, 185–219.
- [13] J. K. Hale and G. Raugel, Reaction-diffusion equation on thin domains, J. Math. Pures Appl. (9) 71 (1992), no. 1, 33–95.
- [14] J. K. Hale and G. Raugel, A reaction-diffusion equation on a thin L-shaped domain, Proc. Roy. Soc. Edinburgh Sect. A 125 (1995), no. 2, 283–327.
- [15] L. T. Hoang and G. R. Sell, Navier-Stokes equations with Navier boundary conditions for an oceanic model, J. Dynam. Differential Equations 22 (2010), no. 3, 563–616.
- [16] D. Iftimie, G. Raugel, and G. R. Sell, Navier-Stokes equations in thin 3D domains with Navier boundary conditions, Indiana Univ. Math. J. 56 (2007), no. 3, 1083–1156.
- [17] C. Kublik, N. M. Tanushev, and R. Tsai, An implicit interface boundary integral method for Poisson's equation on arbitrary domains, J. Comput. Phys. 247 (2013), 279–311.
- [18] C. Kublik and R. Tsai, Integration over curves and surfaces defined by the closest point mapping, Res. Math. Sci. 3 (2016), Paper No. 3, 17.
- [19] M. A. Olshanskii, A. Reusken, and X. Xu, An Eulerian space-time finite element method for diffusion problems on evolving surfaces, SIAM J. Numer. Anal. 52 (2014), no. 3, 1354–1377.
- [20] J. V. Pereira and R. P. Silva, Reaction-diffusion equations in a noncylindrical thin domain, Bound. Value Probl. (2013), 2013:248, 10.
- [21] M. Prizzi, M. Rinaldi, and K. P. Rybakowski, Curved thin domains and parabolic equations, Studia Math. 151 (2002), no. 2, 109–140.
- [22] M. Prizzi and K. P. Rybakowski, The effect of domain squeezing upon the dynamics of reaction-diffusion equations, J. Differential Equations 173 (2001), no. 2, 271–320.
- [23] M. Prizzi and K. P. Rybakowski, On inertial manifolds for reaction-diffusion equations on genuinely high-dimensional thin domains, Studia Math. 154 (2003), no. 3, 253–275.
- [24] G. Raugel, Dynamics of partial differential equations on thin domains, Dynamical systems (Montecatini Terme, 1994), Lecture Notes in Math., vol. 1609, Springer, Berlin, 1995, pp. 208–315.
- [25] L. Simon, Lectures on geometric measure theory, Proceedings of the Centre for Mathematical Analysis, Australian National University, vol. 3, Australian National University, Centre for Mathematical Analysis, Canberra, 1983.
- [26] R. Temam and M. Ziane, Navier-Stokes equations in three-dimensional thin domains with various boundary conditions, Adv. Differential Equations 1 (1996), no. 4, 499–546.
- [27] R. Temam and M. Ziane, Navier-Stokes equations in thin spherical domains, Optimization methods in partial differential equations (South Hadley, MA, 1996), Contemp. Math., vol. 209, Amer. Math. Soc., Providence, RI, 1997, pp. 281–314.
- [28] M. Vierling, Parabolic optimal control problems on evolving surfaces subject to point-wise box constraints on the control—theory and numerical realization, Interfaces Free Bound. 16 (2014), no. 2, 137–173.

Chapter 3

An energetic variational approach for nonlinear diffusion equations in moving thin domains

3.1 Introduction

In this chapter, we are interested in deriving diffusion equations on a moving surface, by regarding it as a thin width limit of the problem in a moving thin domain around the moving surface.

Let us begin with an equation of the conservation of mass ρ with velocity u in a moving domain $\Omega(t), t \in (0,T)$ in $\mathbb{R}^n, n \geq 2$ of the form

$$\partial_t \rho + \operatorname{div}(\rho u) = 0 \quad \text{in} \quad \Omega(t), \, t \in (0, T),$$

$$(3.1.1)$$

which represents the local conservation of mass. Considering the situation that there is no exchange of mass on the boundary, i.e.

$$u \cdot \nu_{\Omega} = V_{\Omega}^{N}$$
 on $\partial \Omega(t), t \in (0, T),$ (3.1.2)

where V_{Ω}^{N} is the normal velocity of the boundary $\partial\Omega(t)$ in the direction of the outward normal vector field ν_{Ω} of $\partial\Omega(t)$. Similar conservation law of mass η with velocity v on a moving surface $\Gamma(t)$ can be derived from the local conservation of mass. It turns out (see Section 3.3) that, when the normal component of v is equal to the outward normal velocity V_{Γ}^{N} of the moving surface $\Gamma(t)$, the resulting equation is of the form:

$$\partial^{\circ}\eta - V_{\Gamma}^{N}H\eta + \operatorname{div}_{\Gamma}(\eta v^{T}) = 0 \quad \text{on} \quad \Gamma(t), \ t \in (0, T),$$
(3.1.3)

where $\partial^{\circ} = \partial_t + V_{\Gamma}^N \nu_{\Gamma} \cdot \nabla$ is the normal time derivative, ν_{Γ} is the outward normal vector field of $\Gamma(t)$, H is the (n-1 times) mean curvature of $\Gamma(t)$, $\operatorname{div}_{\Gamma}$ is the surface divergence operator on $\Gamma(t)$, and v^T is a tangential vector field satisfying $v = V_{\Gamma}^N \nu_{\Gamma} + v^T$. Note that this equation is obtained as the zero width limit of the corresponding equation (3.1.1) in a moving thin domain $\Omega_{\varepsilon}(t)$ defined as the set of all points in \mathbb{R}^n with distance less than ε from $\Gamma(t)$ (see Remark 3.4.2).

The conventional diffusion equations, or even the porous-media equations, can be viewed as the combination of incompressible fluids with the damping in the form of Darcy's law. Take the usual Darcy's law for the velocity u in the moving thin domain $\Omega_{\varepsilon}(t)$:

$$-\rho u = \nabla p(\rho) \quad \text{in} \quad \Omega_{\varepsilon}(t), \, t \in (0, T), \tag{3.1.4}$$

where p is the pressure, then we can prove (see Theorem 3.4.1) that the zero width limit of the diffusion equations (3.1.1) and (3.1.4) yields diffusion equations on the moving surface $\Gamma(t)$: (3.1.3) and Darcy's law

$$-\eta v^T = \nabla_{\Gamma} p(\eta) \quad \text{on} \quad \Gamma(t), \ t \in (0, T).$$
(3.1.5)

Here v^T is the tangential component of the velocity v and ∇_{Γ} is the tangential gradient operator on $\Gamma(t)$.

The diffusion equations (3.1.1), (3.1.4) and (3.1.3), (3.1.5) possess specific energy identities. It can be easily proven (see Section 3.5) that for ρ and u satisfying (3.1.1) and (3.1.4) the energy identity

$$\frac{d}{dt} \int_{\Omega(t)} \omega(\rho) \, dx = -\int_{\Omega(t)} \rho |u|^2 \, dx - \int_{\partial \Omega(t)} p(\rho) V_{\Omega}^N \, d\mathcal{H}^{n-1} \tag{3.1.6}$$

holds. Here ω is a function satisfying $p(\rho) = \omega'(\rho)\rho - \omega(\rho)$. Similarly, for η and v satisfying (3.1.3) and (3.1.5) we have

$$\frac{d}{dt} \int_{\Gamma(t)} \omega(\eta) \, d\mathcal{H}^{n-1} = -\int_{\Gamma(t)} \eta |v^T|^2 \, d\mathcal{H}^{n-1} + \int_{\Gamma(t)} p(\eta) V_{\Gamma}^N H \, d\mathcal{H}^{n-1}, \tag{3.1.7}$$

where \mathcal{H}^{n-1} is the (n-1)-dimensional Hausdorff measure. Fortunately, the energy identity (3.1.7) on the moving surface can be derived as the zero width limit of the energy identity (3.1.6) in the moving thin domain (see Theorem 3.5.3).

With the results from this chapter, we can also note that the passing of zero width limit commutes with an energetic variational approach originated from the works of Lord Rayleigh [24] and Onsager [16, 17] and developed by Liu and others [3, 12, 25] (see Section 3.6). In summary, we show that the diagram below is commutative.



A standard approach for finding the limit of a thin domain problem is a rescaling argument: one transforms a partial differential equation in a thin domain into that in a fixed in width reference domain by the change of variables and then gets a limit equation by assuming that a rescaled solution is independent of variables in thin directions. In our case where the thin domain and the surface both move, one may transform the moving thin domain into a fixed in time and width reference domain. However, it yields tedious calculations because of the geometry of the limit moving surface and it is difficult to bring a limit equation obtained on a stationary reference surface back to an equation on the original moving surface. One other method is to rescale the width of the moving thin domain without fixing time, which is used in [15] to find the limit of the Neumann type problem of the heat equation (equations (3.1.1), (3.1.2), and (3.1.4) with $p(\rho) = \rho$) in moving thin domains. However, it is still complicated and requires a questionable assumption that the boundary condition holds in a middle of the moving thin domain. It is also artificial in the sense that we have to make rescaled solutions constant in the thin direction at an "appropriate" point to derive the limit energy identity and if we take a wrong point then we get a wrong limit (see Remarks 3.5.5).

To derive a thin width limit with more straightforward calculations we consider the Taylor series of a function on $\Omega_{\varepsilon}(t)$ in powers of the signed distance from $\Gamma(t)$. We assume that $\Omega_{\varepsilon}(t)$ admits the normal coordinate system around $\Gamma(t)$, i.e. for each $x \in \Omega_{\varepsilon}(t)$ there exists a unique point $\pi(x,t) \in \Gamma(t)$ such that

$$x = \pi(x, t) + d(x, t)\nu_{\Gamma}(\pi(x, t), t),$$

where d is the signed distance function from $\Gamma(t)$ increasing in the direction of ν_{Γ} . Based on the normal coordinate system we consider expansions of

$$\begin{split} \rho(x,t) &= \rho(\pi(x,t) + d(x,t)\nu_{\Gamma}(\pi(x,t),t),t), \\ u(x,t) &= u(\pi(x,t) + d(x,t)\nu_{\Gamma}(\pi(x,t),t),t) \end{split}$$

in powers of the signed distance d(x, t):

$$\rho(x,t) = \eta(\pi(x,t),t) + d(x,t)\eta^{1}(\pi(x,t),t) + d(x,t)^{2}\eta^{2}(\pi(x,t),t) + \cdots,$$

$$u(x,t) = v(\pi(x,t),t) + d(x,t)v^{1}(\pi(x,t),t) + d(x,t)^{2}v^{2}(\pi(x,t),t) + \cdots.$$
(3.1.8)

In these expansions we assume that η , v, and the coefficients of the powers of d(x,t) are functions on $\Gamma(t)$ and independent of ε (note that the functions ρ and u on $\Omega_{\varepsilon}(t)$ depend on ε). Under this and other suitable assumptions, we obtain the limit equations (3.1.3) and (3.1.5) as the zeroth order terms of expansions in powers of d(x,t) (or ε) of the bulk equations (3.1.1), (3.1.2), and (3.1.4) by differentiating (3.1.8) and substituting them for (3.1.1), (3.1.2), and (3.1.4) (see Section 3.4). Note that, if we take the average of (3.1.8) in the normal direction of $\Gamma(t)$, then we get

$$\frac{1}{2\varepsilon} \int_{-\varepsilon}^{\varepsilon} \rho(y + r\nu_{\Gamma}(y, t), t) \, dr = \eta(y, r) + (\text{higher order terms in } \varepsilon), \quad y \in \Gamma(t)$$

and a similar equality for u. Thus, formally speaking, we derive the limit equations (3.1.3) and (3.1.5) as equations on $\Gamma(t)$ satisfied by the limit as $\varepsilon \to 0$ of the averages of ρ and u in the thin direction.

The idea mentioned above also applies to derivation of the energy identity (3.1.7) on the moving surface from that in the moving thin domain (3.1.6) (see Section 3.5). To get the limit energy identity we use integral transformation formulas from surface integrals over the level-set surfaces $\{x \in \mathbb{R}^n \mid d(x,t) = r\}$ ($-\varepsilon < r < \varepsilon$) into that over the zero level-set surface $\Gamma(t)$ (see Lemma 3.5.4).

There is a long history in the study of partial differential equations in thin domains, such as the pioneering work by Hale and Raugel [8,9], where they investigated damped hyperbolic equations and reaction-diffusion equations in a flat stationary thin domain of the form

$$\Omega_{\varepsilon} = \{ (x', x_n) \in \mathbb{R}^n \mid x' \in \omega, \ 0 < x_n < \varepsilon g(x') \},$$
(3.1.9)

where ω is an open set in \mathbb{R}^{n-1} and g is a function on ω . There is also a large number of the literature on reaction-diffusion equations in various types of thin domains such as a thin L-shaped domain [10], a moving flat thin domain of the form (3.1.9) with g time-dependent

[18], and flat and curved thin domains with holes [19–21] (here a curved thin domain is a thin domain degenerating into a lower dimensional manifold). A main subject in the above literature is to compare the dynamics of equations in thin domains with that of limit equations in their limit sets rather than to find the limit equations of the original equations in the thin domains, since their limit sets are stationary and thus the rescaling argument works well for finding the limit equations. The Navier–Stokes equations in thin domains has been also studied well [11, 13, 23, 26, 27] since fluid flows in thin domains often appear in natural sciences like the flow of water in a large lake, geophysical flows, etc. Researchers are especially interested in the relation between the smallness of the width of thin domains. We refer to [22] and references therein for other types of thin domains degenerating into stationary sets and mathematical analysis of partial differential equations in such thin domains.

In the case where the limit set of a thin domain moves, derivation of the limit of a partial differential equation in the thin domain is more complicated since the geometry of the limit set changes as it moves. Such a problem was first considered in [15] where the author derived both formally and rigorously the limit equation of the Neumann type problem of the heat equation (equations (3.1.1), (3.1.2), and (3.1.4) with $p(\rho) = \rho$) in a moving thin domain degenerating into a closed smooth moving surface. He also found that the normal velocity and the mean curvature of the degenerate moving surface affects the limit equation, which is not observed in the case where the limit set of a thin domain does not move.

The rest of this chapter is organized as follows. In Section 3.2, we fix notations on various quantities related to the moving surface. In Section 3.3, we briefly observe that the transport equations in the moving domain and the moving surface are equivalent to the local mass conservation. In Section 3.4, we derive the limit equations (3.1.3) and (3.1.5) on the moving surface from the diffusion equations (3.1.1), (3.1.2), and (3.1.4) on the moving thin domain by means of expansion in terms of the signed distance. In Section 3.5, we derive the energy identities (3.1.6) and (3.1.7) from corresponding diffusion equations and then show that the energy identity (3.1.7) on the moving surface is the zero width limit of the energy identity (3.1.6) on the moving thin domain. In Section 3.6, we apply an energetic variational approach to the energy identities (3.1.6) and (3.1.7) to obtain Darcy's laws (3.1.4) and (3.1.5).

3.2 Quantities on a moving surface

We start with several notations for a moving surface. Let $\Gamma(t)$, $t \in [0, T]$ be an (n - 1)dimensional closed (that is, compact and without boundary), connected, oriented and smooth moving surface in \mathbb{R}^n with $n \ge 2$. Also, let

$$S_T := \bigcup_{t \in (0,T)} \Gamma(t) \times \{t\} \subset \mathbb{R}^{n+1}$$

be a space-time hypersurface associated with the moving surface $\Gamma(t)$. For each $t \in [0, T]$ we write $\nu_{\Gamma}(\cdot, t)$, $V_{\Gamma}^{N}(\cdot, t)$, and $d(\cdot, t)$ for the unit outward normal vector field of $\Gamma(t)$, the scalar outward normal velocity of $\Gamma(t)$, and the signed distance function from $\Gamma(t)$, respectively. Note that to describe the evolution of a closed surface it is sufficient to give the normal velocity. Since the smooth closed surface $\Gamma(t)$ varies smoothly in time, the principal curvatures $\kappa_1(\cdot, t), \ldots, \kappa_{n-1}(\cdot, t)$ of $\Gamma(t)$ are bounded uniformly in $t \in [0, T]$. Then there exists a constant $\delta > 0$ independent of t such that for each $t \in [0, T]$ the tubular neighborhood of $\Gamma(t)$ of the form

$$N(t) := \{ x \in \mathbb{R}^n \mid \operatorname{dist}(x, \Gamma(t)) < \delta \}$$

admits the normal coordinate system

$$x = \pi(x, t) + d(x, t)\nu_{\Gamma}(\pi(x, t), t), \quad x \in N(t),$$
(3.2.1)

where $\pi(x,t)$ is the closest point on $\Gamma(t)$ to x (see [6, Section 14.6] for example). For each $t \in [0,T]$ we suppose that $d(\cdot,t)$ increases along the direction of $\nu_{\Gamma}(\cdot,t)$. Then we have

$$\nabla d(x,t) = \nu_{\Gamma}(\pi(x,t),t), \quad (x,t) \in N(t),$$

$$\partial_t d(y,t) = -V_{\Gamma}^N(y,t), \quad (y,t) \in \Gamma(t),$$
(3.2.2)

Moreover, differentiating both sides of

$$d(x,t) = \{x - \pi(x,t)\} \cdot \nabla d(x,t), \quad d(\pi(x,t),t) = 0$$

with respect to t we easily obtain

$$\partial_t d(x,t) = \partial_t d(\pi(x,t),t) = -V_{\Gamma}^N(\pi(x,t),t), \quad (x,t) \in N_T,$$
(3.2.3)

where $N_T := \bigcup_{t \in (0,T)} N(t) \times \{t\}.$

Next we fix $t \in [0, T]$ and give differential operators on the surface $\Gamma(t)$. We define the orthogonal projection onto the tangent plane of $\Gamma(t)$ by

$$P_{\Gamma}(y,t) := I_n - \nu_{\Gamma}(y,t) \otimes \nu_{\Gamma}(y,t), \quad y \in \Gamma(t).$$

Here I_n denotes the identity matrix of size n and $a \otimes b = (a_i b_j)_{i,j}$ is the tensor product of two vectors $a = (a_1, \ldots, a_n)$ and $b = (b_1, \ldots, b_n)$ in \mathbb{R}^n . For a function $f: \Gamma(t) \to \mathbb{R}$ and a vector field $F: \Gamma(t) \to \mathbb{R}^n$ we define the tangential gradient of f and the surface divergence of F as

$$\nabla_{\Gamma} f(y) := P_{\Gamma}(y, t) \nabla \tilde{f}(y), \quad \operatorname{div}_{\Gamma} F(y) := \operatorname{tr}[P_{\Gamma}(y, t) \nabla \widetilde{F}(y)], \quad y \in \Gamma(t).$$

Here \tilde{f} and \tilde{F} are extensions of f and F to N(t) satisfying $\tilde{f} = f$ and $\tilde{F} = F$ on $\Gamma(t)$. Also, tr[M] denotes the trace of a square matrix M and we use the notation

$$\nabla G = \begin{pmatrix} \partial_1 G_1 & \dots & \partial_1 G_n \\ \vdots & \ddots & \vdots \\ \partial_n G_1 & \dots & \partial_n G_n \end{pmatrix}$$

for the gradient matrix of a vector field $G = (G_1, \ldots, G_n)$. Note that the tangential gradient of f and the surface divergence of F do not depend on a choice of extensions (see e.g. [5, Lemma 2.4]). Moreover, for any function f on $\Gamma(t)$ we easily see that

$$\nabla_{\Gamma} f(y) \cdot \nu_{\Gamma}(y, t) = 0, \quad y \in \Gamma(t).$$
(3.2.4)

We define the (n-1 times) mean curvature H of $\Gamma(t)$ as

$$H(y,t) := -\operatorname{div}_{\Gamma} \nu_{\Gamma}(y,t), \quad y \in \Gamma(t).$$
(3.2.5)

Note that the mean curvature is equal to the sum of the principal curvatures:

$$H(y,t) = \sum_{i=1}^{n-1} \kappa_i(y,t), \quad y \in \Gamma(t).$$
(3.2.6)

Finally, for a function f on the space-time hypersurface S_T we define the normal time derivative (the time derivative along the normal velocity) as

$$\partial^{\circ} f(y,t) := \partial_t \tilde{f}(y,t) + V_{\Gamma}(y,t)\nu_{\Gamma}(y,t) \cdot \nabla \tilde{f}(y,t), \quad (y,t) \in S_T$$

Here \tilde{f} is an extension of f to N_T satisfying $\tilde{f} = f$ on S_T . Note that the value of $\partial^{\circ} f$ does not depend on a choice of an extension of f and the formula

$$\partial^{\circ} f(y,t) = \frac{d}{dt} \left(f(\pi(y,t),t) \right), \quad (y,t) \in S_T$$
(3.2.7)

holds (see [2, Section 3.4] for details).

3.3 Transport equation in a moving domain and on a moving surface

In this section we give the transport equation for a scalar quantity in a moving domain and on a moving surface. We use some of the same terminology and techniques as in [14]. We first consider transportation of a scalar quantity in a bounded moving domain $\Omega(t)$ in \mathbb{R}^n . Let $\rho(x,t)$ and u(x,t) be the density and the velocity field of the scalar quantity at $x \in \Omega(t)$, respectively. Our starting point is the local mass conservation

$$\frac{d}{dt} \int_{U(t)} \rho \, dx = 0 \tag{3.3.1}$$

for any portion U(t) (relatively open set) of $\Omega(t)$ moving with velocity $u(\cdot, t)$ and whose closure (in \mathbb{R}^n) is contained in $\Omega(t)$. Since the left-hand side is equal to $\int_{U(t)} \{\partial_t \rho + \operatorname{div}(\rho u)\} dx$ by the Reynolds transport theorem [7] and the divergence theorem, the condition (3.3.1) for any U(t) is equivalent to the transport equation

$$\partial_t \rho + \operatorname{div}(\rho u) = 0$$
 in $Q_T := \bigcup_{t \in (0,T)} \Omega(t) \times \{t\}.$ (3.3.2)

To make the total mass $\int_{\Omega(t)} \rho \, dx$ conserved, we impose the boundary condition

$$u \cdot \nu_{\Omega} = V_{\Omega}^{N}$$
 on $\partial_{\ell} Q_{T} := \bigcup_{t \in (0,T)} \partial \Omega(t) \times \{t\},$ (3.3.3)

where $\nu_{\Omega}(\cdot, t)$ and $V_{\Omega}^{N}(\cdot, t)$ are the unit outward normal vector field and the scalar outward normal velocity of $\partial \Omega(t)$, respectively. The boundary condition (3.3.3) physically means that the quantity in $\Omega(t)$ moves along the boundary of $\Omega(t)$ and it does not go into and out of $\Omega(t)$.

Next we give the transport equation for a scalar quantity on a moving surface. Let $\Gamma(t)$ be a closed, connected, oriented moving surface in \mathbb{R}^n . As in Section 3.2, we write $\nu_{\Gamma}(\cdot, t)$ and $V_{\Gamma}(\cdot, t)$ for the outward normal vector field and the scalar outward normal velocity of

 $\Gamma(t)$, respectively. Suppose that for each $t \in (0, T)$ a scalar quantity on $\Gamma(t)$ has the density $\eta(y, t)$ at $y \in \Gamma(t)$ and moves with velocity

$$v(y,t) = V_{\Gamma}(y,t)\nu_{\Gamma}(y,t) + v^{T}(y,t), \quad y \in \Gamma(t),$$

where $v^T(\cdot, t)$ is a given tangential velocity field on $\Gamma(t)$. Then its local mass conservation is expressed as

$$\int_{U(t)} \eta \, d\mathcal{H}^{n-1} = 0 \tag{3.3.4}$$

for any portion U(t) (relatively open set) of $\Gamma(t)$ moving with velocity $v(\cdot, t)$. The Leibniz formula [4, Lemma 2.2] yields

$$\frac{d}{dt} \int_{U(t)} \eta \, d\mathcal{H}^{n-1} = \int_{U(t)} \{\partial^{\circ} \eta - V_{\Gamma}^{N} H \eta + \operatorname{div}_{\Gamma}(\eta v^{T})\} \, d\mathcal{H}^{n-1}.$$

From this formula, the condition (3.3.4) for any U(t) is equivalent to

$$\partial^{\circ} \eta - V_{\Gamma}^{N} H \eta + \operatorname{div}_{\Gamma}(\eta v^{T}) = 0 \quad \text{on} \quad S_{T}.$$
 (3.3.5)

This is the transport equation on the moving surface $\Gamma(t)$.

3.4 Zero width limit for nonlinear diffusion equations

Let us consider nonlinear diffusion of a scalar quantity in $\Omega(t)$ with density ρ and velocity u. Suppose that the diffusion process is described by the transport equation (3.3.2) and Darcy's law $-\rho u = \nabla p(\rho)$, where

$$p(\rho) := \omega'(\rho)\rho - \omega(\rho) \tag{3.4.1}$$

is the pressure with a given function $\omega(\rho)$, $\rho \in \mathbb{R}$. We impose the boundary condition (3.3.3). Hence the nonlinear diffusion equations we deal with are

$$\partial_t \rho + \operatorname{div}(\rho u) = 0$$
 in Q_T , (3.4.2)

$$-\rho u = \nabla p(\rho) \quad \text{in} \quad Q_T, \tag{3.4.3}$$

$$u \cdot \nu_{\Omega} = V_{\Omega}^{N}$$
 on $\partial_{\ell} Q_{T}$. (3.4.4)

We consider these equations in a moving thin domain. For sufficiently small $\varepsilon > 0$, we define a moving thin domain $\Omega_{\varepsilon}(t)$ as the set of all points in \mathbb{R}^n with distance less than ε from the moving surface $\Gamma(t)$:

$$\Omega_{\varepsilon}(t) := \{ x \in \mathbb{R}^n \mid \operatorname{dist}(x, \Gamma(t)) < \varepsilon \}.$$
(3.4.5)

We write $Q_{\varepsilon,T}$ and $\partial_{\ell}Q_{\varepsilon,T}$ for Q_T and $\partial_{\ell}Q_T$ with $\Omega(t) = \Omega_{\varepsilon}(t)$. Our goal in this section is to find the limit equations of (3.4.2)–(3.4.4) in $\Omega_{\varepsilon}(t)$ as ε goes to zero, that is, the moving thin domain $\Omega_{\varepsilon}(t)$ degenerates into the moving surface $\Gamma(t)$. According to the normal coordinate system (3.2.1), we expand ρ and u in powers of the signed distance d(x, t) as

$$\rho(x,t) = \eta(\pi(x,t),t) + d(x,t)\eta^{1}(\pi(x,t),t) + R(d(x,t)^{2}), \qquad (3.4.6)$$

$$u(x,t) = v(\pi(x,t),t) + d(x,t)v^{1}(\pi(x,t),t) + R(d(x,t)^{2})$$
(3.4.7)

for $(x,t) \in Q_{\varepsilon,T}$ and assume that η , v, and the coefficients of $d(x,t)^k$ for each $k \in \mathbb{N}$ in (3.4.6) and (3.4.7) are independent of ε . Here $R(d(x,t)^k)$ $(k \in \mathbb{N})$ is the sum of the terms of order equal to or higher than k with respect to small d(x,t). In particular, R(f(x,t)) for a function f(x,t) can be of the form

$$R(f(x,t)) = f(x,t)g(x,t)$$

with some (bounded) function g(x,t). Note that we can differentiate $R(d(x,t)^k)$ and its *j*-th order derivative is of the form $R(d(x,t)^{k-j})$ for $j \leq k$ although we cannot differentiate $O(d(x,t)^k)$ since it only represents a quantity whose absolute value is bounded above by $|d(x,t)|^k$. Also, since d(x,t) is of order ε on $Q_{\varepsilon,T}$, we have $R(d(x,t)^k) = O(\varepsilon^k)$ for $(x,t) \in Q_{\varepsilon,T}$ and $k \in \mathbb{N}$.

Under the expansions (3.4.6) and (3.4.7), the limit equations of (3.4.2)–(3.4.4) in $\Omega_{\varepsilon}(t)$ as ε goes to zero are given as equations on $\Gamma(t)$ satisfied by η and v.

Theorem 3.4.1. Let ρ and u satisfy the equations (3.4.2)–(3.4.4) in the moving thin domain $\Omega(t) = \Omega_{\varepsilon}(t)$ given by (3.4.5). Also, let η and v be the zeroth order terms in the expansions (3.4.6) and (3.4.7) of ρ and u, respectively. Then v is of the form

$$v = V_{\Gamma}^{N} \nu_{\Gamma} + v^{T} \quad on \quad S_{T} \tag{3.4.8}$$

with some tangential velocity field v^T on $\Gamma(t)$, and η and v satisfy the equations

$$\partial^{\circ} \eta - V_{\Gamma}^{N} H \eta + \operatorname{div}_{\Gamma}(\eta v^{T}) = 0 \qquad on \quad S_{T}, \qquad (3.4.9)$$

$$-\eta v^T = \nabla_{\Gamma} p(\eta) \quad on \quad S_T. \tag{3.4.10}$$

Proof. For the sake of simplicity, we use the abbreviations

$$f(\pi, t) = f(\pi(x, t), t), \quad R(d^k) = R(d(x, t)^k)$$
 (3.4.11)

for a function f on S_T , $(x,t) \in Q_{\varepsilon,T}$, and $k \in \mathbb{N}$. We also abbreviate the product of several functions with the same argument like

$$[u_1 \cdot u_2](x,t) = u_1(x,t) \cdot u_2(x,t)$$
(3.4.12)

for vector fields u_1 and u_2 on $Q_{\varepsilon,T}$. First we show that v is of the form (3.4.8). By the definition (3.4.5) of the moving thin domain $\Omega_{\varepsilon}(t)$, the unit outward normal vector and the outward normal velocity of its boundary are given by

$$\nu_{\Omega}(x,t) = \pm \nu_{\Gamma}(\pi,t), \quad V_{\Omega}^{N}(x,t) = \pm V_{\Gamma}^{N}(\pi,t)$$
(3.4.13)

for $(x,t) \in \partial_{\ell} Q_{\varepsilon,T}$ with $d(x,t) = \pm \varepsilon$ (double-sign corresponds). Hence the boundary condition (3.4.4) reads

$$u(x,t) \cdot \nu_{\Gamma}(\pi,t) = V_{\Gamma}^{N}(\pi,t), \quad (x,t) \in \partial_{\ell} Q_{\varepsilon,T}.$$

We substitute (3.4.7) for u in the above equality. Then

$$[v \cdot \nu](\pi, t) \pm \varepsilon [v^1 \cdot \nu](\pi, t) + O(\varepsilon^2) = V_{\Gamma}^N(\pi, t).$$

Since v, v^1, ν_{Γ} , and V_{Γ}^N are independent of ε , it follows that

$$[v \cdot \nu](\pi(x,t),t) = V_{\Gamma}(\pi(x,t),t), \quad [v^1 \cdot \nu](\pi(x,t),t) = 0$$

for all $(x,t) \in \partial_{\ell} Q_{\varepsilon,T}$, which imply that

$$[v \cdot \nu](y,t) = V_{\Gamma}^{N}(y,t), \quad (y,t) \in S_{T},$$
(3.4.14)

$$[v^1 \cdot \nu](y,t) = 0, \qquad (y,t) \in S_T.$$
(3.4.15)

Hence v is of the form (3.4.8) with some tangential velocity field v^T on $\Gamma(t)$.

Next we derive the equations (3.4.9)–(3.4.10). Let $(x,t) \in Q_{\varepsilon,T}$. We differentiate both sides of (3.4.6) with respect to t and apply (3.2.3) and (3.2.7) to get

$$\partial_t \rho(x,t) = \partial^{\circ} \eta(\pi,t) - [V_{\Gamma}^N \eta^1](\pi,t) + R(d).$$
(3.4.16)

Let us compute the divergence of ρu . We differentiate $\pi(x,t) = x - d(x,t)\nu_{\Gamma}(\pi,t)$ with respect to x and apply (3.2.2) to get

$$\nabla \pi(x,t) = P_{\Gamma}(\pi,t) + R(d).$$
 (3.4.17)

From the expansions (3.4.6) and (3.4.7),

$$[\rho u](x,t) = V(\pi,t) + d(x,t)V^{1}(\pi,t) + R(d^{2}), \qquad (3.4.18)$$

where

$$V(\pi, t) := [\eta v](\pi, t), \tag{3.4.19}$$

$$V^{1}(\pi,t) := [\eta v^{1}](\pi,t) + [\eta^{1}v](\pi,t).$$
(3.4.20)

We differentiate both sides of (3.4.18) with respect to x. Then by (3.2.2) and (3.4.17),

$$[\nabla(\rho u)](x,t) = \nabla \pi(x,t) \nabla V(\pi,t) + \nabla d(x,t) \otimes V^{1}(\pi,t) + R(d)$$
$$= [P_{\Gamma} \nabla V](\pi,t) + [\nu_{\Gamma} \otimes V^{1}](\pi,t) + R(d).$$

From this formula and $\operatorname{tr}[\nu_{\Gamma} \otimes V^1] = \nu_{\Gamma} \cdot V^1$, the divergence of ρu is

$$[\operatorname{div}(\rho u)](x,t) = \operatorname{div}_{\Gamma} V(\pi,t) + [\nu_{\Gamma} \cdot V^{1}](\pi,t) + R(d)$$

Since v is of the form (3.4.8) and V is given by (3.4.19),

$$div_{\Gamma}V = div_{\Gamma}[\eta(V_{\Gamma}^{N}\nu_{\Gamma} + v^{T})] = \nabla_{\Gamma}(\eta V_{\Gamma}^{N}) \cdot \nu_{\Gamma} + \eta V_{\Gamma}^{N}div_{\Gamma}\nu_{\Gamma} + div_{\Gamma}(\eta v^{T})$$
$$= -\eta V_{\Gamma}^{N}H + div_{\Gamma}(\eta v^{T})$$

on S_T by (3.2.4) and (3.2.5). We also have

$$[\nu_{\Gamma} \cdot V^1](\pi, t) = [\eta^1 V_{\Gamma}^N](\pi, t)$$

by (3.4.14), (3.4.15), and (3.4.20). Therefore,

$$[\operatorname{div}(\rho u)](x,t) = -[V_{\Gamma}^{N}H\eta](\pi,t) + [\operatorname{div}_{\Gamma}(\eta v^{T})](\pi,t) + [\eta^{1}V_{\Gamma}^{N}](\pi,t) + R(d).$$
(3.4.21)

Substituting (3.4.16) and (3.4.21) for (3.4.2), we obtain

$$\partial^{\circ}\eta(\pi,t) - [V_{\Gamma}^{N}H\eta](\pi,t) + [\operatorname{div}_{\Gamma}(\eta v^{T})](\pi,t) = R(d).$$

Here each term on the left-hand side is independent of d = d(x, t). Hence

$$\partial^{\circ}\eta(\pi(x,t),t) - [V_{\Gamma}^{N}H\eta](\pi(x,t),t) + [\operatorname{div}_{\Gamma}(\eta v^{T})](\pi(x,t),t) = 0$$

for all $(x,t) \in Q_{\varepsilon,T}$, which shows that η and $v = V_{\Gamma}^{N} \nu_{\Gamma} + v^{T}$ satisfy (3.4.9) on S_{T} . Let us derive (3.4.10). We expand the pressure $p(\rho)$ in d(x,t) as

$$p(\rho(x,t)) = p^{0}(\pi,t) + d(x,t)p^{1}(\pi,t) + R(d^{2}), \quad (x,t) \in Q_{\varepsilon,T}.$$
(3.4.22)

Then it follows form the expansions (3.4.6) and (3.4.22) that

$$p^{0}(\pi(x,t),t) = p(\eta(\pi(x,t),t))$$

for all $(x,t) \in Q_{\varepsilon,T}$, which implies that

$$p^{0}(y,t) = p(\eta(y,t)), \quad (y,t) \in S_{T}.$$
 (3.4.23)

Moreover, differentiating (3.4.22) in x and applying (3.2.2) and (3.4.17) we get

$$\nabla p(\rho(x,t)) = \nabla \pi(x,t) \nabla p^0(\pi,t) + p^1(\pi,t) \nabla d(x,t) + R(d)$$
$$= \nabla_{\Gamma} p^0(\pi,t) + [p^1 \nu_{\Gamma}](\pi,t) + R(d).$$

for $(x,t) \in Q_{\varepsilon,T}$. We substitute this for (3.4.3) and apply (3.4.8). Then we have

$$-[\eta v^T](\pi,t) - [\eta V_{\Gamma}^N \nu_{\Gamma}](\pi,t) + R(d) = \nabla_{\Gamma} p^0(\pi,t) + [p^1 \nu_{\Gamma}](\pi,t) + R(d).$$

Since all terms except of R(d) are independent of d = d(x, t) and the vectors v^T and $\nabla_{\Gamma} p^0$ are tangential to $\Gamma(t)$, it follows that

$$-[\eta v^{T}](\pi(x,t),t) = \nabla_{\Gamma} p^{0}(\pi(x,t),t), \quad -[\eta V_{\Gamma}^{N}](\pi(x,t),t) = p^{1}(\pi(x,t),t)$$

for all $(x,t) \in Q_{\varepsilon,T}$. Therefore, we get

$$-[\eta v^{T}](y,t) = \nabla_{\Gamma} p^{0}(y,t), \quad (y,t) \in S_{T},$$
(3.4.24)

$$-[\eta V_{\Gamma}^{N}](y,t) = p^{1}(y,t), \qquad (y,t) \in S_{T}.$$
(3.4.25)

By (3.4.23) and (3.4.24) we conclude that η and v satisfy (3.4.10) on S_T .

Remark 3.4.2. By the proof of Theorem 3.4.1 we observe that the transport equation (3.3.5) on the moving surface $\Gamma(t)$ can be derived as the limit of the transport equation (3.3.2) in the moving thin domain $\Omega(t) = \Omega_{\varepsilon}(t)$ with the boundary condition (3.3.3) as ε goes to zero.

3.5 Energy law

The subject in this section is the energy law for nonlinear diffusion equations (3.4.2)–(3.4.4) and (3.4.9)–(3.4.10). As in Section 3.4, the pressure $p(\rho)$ is given by (3.4.1) with a given function $\omega(\rho)$.

Proposition 3.5.1. Assume that ρ and u satisfy (3.4.2)–(3.4.4). Then

$$\frac{d}{dt} \int_{\Omega(t)} \omega(\rho) \, dx = -\int_{\Omega(t)} \rho |u|^2 \, dx - \int_{\partial \Omega(t)} p(\rho) V_{\Omega}^N \, d\mathcal{H}^{n-1}. \tag{3.5.1}$$

Proof. By the Reynolds transport theorem,

$$\frac{d}{dt} \int_{\Omega(t)} \omega(\rho) \, d\mathcal{H}^{n-1} = \int_{\Omega(t)} \partial_t \omega(\rho) \, dx + \int_{\partial \Omega(t)} \omega(\rho) V_{\Omega}^N \, d\mathcal{H}^{n-1}.$$

Since $\partial_t \omega(\rho) = \omega'(\rho) \partial_t \rho$ and the transport equation (3.4.2) is satisfied,

$$\partial_t \omega(\rho) = -\omega'(\rho) \operatorname{div}(\rho u) = -\operatorname{div}(\omega'(\rho)\rho u) + \nabla \omega'(\rho) \cdot (\rho u)$$

Hence the divergence theorem and (3.4.4) yield

$$\int_{\Omega(t)} \partial_t \omega(\rho) \, dx = -\int_{\partial\Omega(t)} \omega'(\rho) \rho V_{\Omega}^N \, d\mathcal{H}^{n-1} + \int_{\Omega(t)} \nabla \omega'(\rho) \cdot (\rho u) \, dx$$

Using this formula we get

$$\frac{d}{dt} \int_{\Omega(t)} \omega(\rho) \, dx = \int_{\Omega(t)} \nabla \omega'(\rho) \cdot (\rho u) \, dx - \int_{\partial \Omega(t)} \{\omega'(\rho)\rho - \omega(\rho)\} V_{\Omega}^N \, d\mathcal{H}^{n-1}.$$

The energy law (3.5.1) follows from this equality, (3.4.1), and

$$u = -\frac{\nabla p(\rho)}{\rho} = -\nabla \omega'(\rho)$$

by (3.4.1) and (3.4.3).

Proposition 3.5.2. Suppose that η and v of the form (3.4.8) satisfy (3.4.9) and (3.4.10). Then

$$\frac{d}{dt} \int_{\Gamma(t)} \omega(\eta) \, d\mathcal{H}^{n-1} = -\int_{\Gamma(t)} \eta |v^T|^2 \, d\mathcal{H}^{n-1} + \int_{\Gamma(t)} p(\eta) V_{\Gamma}^N H \, d\mathcal{H}^{n-1}. \tag{3.5.2}$$

Proof. By the Leibniz formula [4, Lemma 2.2],

$$\frac{d}{dt} \int_{\Gamma(t)} \omega(\eta) \, d\mathcal{H}^{n-1} = I - \int_{\Gamma(t)} \omega(\eta) V_{\Gamma}^N H \, d\mathcal{H}^{n-1},$$

where

$$I = \int_{\Gamma(t)} \{\partial^{\circ} \omega(\eta) + \operatorname{div}_{\Gamma}(\omega(\eta)v^{T})\} d\mathcal{H}^{n-1}.$$

Since $\partial^{\circ}\omega(\eta) = \omega'(\eta)\partial^{\circ}\eta$, the transport equation (3.4.9) implies that

$$\partial^{\circ}\omega(\eta) = \omega'(\eta)\{V_{\Gamma}^{N}H\eta - \operatorname{div}_{\Gamma}(\eta v^{T})\} = \omega'(\eta)V_{\Gamma}^{N}H\eta + \nabla_{\Gamma}\omega'(\eta)\cdot(\eta v^{T}) - \operatorname{div}_{\Gamma}(\omega'(\eta)\eta v^{T}).$$

Hence

Hence

$$I = \int_{\Gamma(t)} \{\omega'(\eta) V_{\Gamma}^{N} H \eta + \nabla_{\Gamma} \omega'(\eta) \cdot (\eta v^{T}) \} d\mathcal{H}^{n-1} + \int_{\Gamma(t)} \operatorname{div}_{\Gamma} [(\omega(\eta) - \omega'(\eta)\eta) v^{T}] d\mathcal{H}^{n-1}.$$

The second integral on the right-hand side vanishes by the Stokes formula and the fact that v^T is tangential and $\Gamma(t)$ has no boundary. Therefore,

$$\frac{d}{dt} \int_{\Gamma(t)} \omega(\eta) \, d\mathcal{H}^{n-1} = \int_{\Gamma(t)} \nabla_{\Gamma} \omega'(\eta) \cdot (\eta v^T) \, d\mathcal{H}^{n-1} + \int_{\Gamma(t)} \{\omega'(\eta)\eta - \omega(\eta)\} V_{\Gamma}^N H \, d\mathcal{H}^{n-1}.$$

Applying

$$v^T = -\frac{\nabla_{\Gamma} p(\eta)}{\eta} = -\nabla_{\Gamma} \omega'(\eta),$$

which follows from (3.4.1) and (3.4.10), to the first term on the right-hand side and (3.4.1)to the second term, we get the energy identity (3.5.2).

Next we derive the energy law (3.5.2) as a limit of the energy law (3.5.1) with the moving thin domain $\Omega(t) = \Omega_{\varepsilon}(t)$ when ε goes to zero. As in Section 3.4, we expand ρ and u in powers of the signed distance as (3.4.6) and (3.4.7) and determine an equality satisfied by η and v.

Theorem 3.5.3. Let ρ and u satisfy the energy law (3.5.1) in the moving thin domain $\Omega(t) = \Omega_{\varepsilon}(t)$ given by (3.4.5). Also, let η and v be the zeroth order terms in the expansions (3.4.6) and (3.4.7) of ρ and u, respectively. Assume that v is of the form (3.4.8) with some tangential velocity field v^T on $\Gamma(t)$ and Darcy's law (3.4.3) holds in $\Omega_{\varepsilon}(t)$. Then η and v satisfy the energy law (3.5.2).

We give change of variables formulas for integrals which we use in the proof of Theorem 3.5.3. For $y \in \Gamma(t)$ and $\rho \in [-\varepsilon, \varepsilon]$ we set

$$J(y,t,r) := \prod_{i=1}^{n-1} \{1 - r\kappa_i(y,t)\},$$
(3.5.3)

where $\kappa_1(\cdot, t), \ldots, \kappa_{n-1}(\cdot, t)$ are the principal curvatures of $\Gamma(t)$. It is the Jacobian that appears when we change variables of integrals over a tubular neighborhood $\{x \in \mathbb{R}^n \mid -r < d(x,t) < r\}$ (r > 0) of $\Gamma(t)$ and a level-set surface $\{x \in \mathbb{R}^n \mid d(x,t) = s\}$ $(s \in \mathbb{R})$ in terms of the normal coordinate system around $\Gamma(t)$ (see [6, Section 14.6] for example). The first formula in Lemma 3.5.4 is often called the co-area formula.

Lemma 3.5.4. For functions f on $\Omega_{\varepsilon}(t)$ and g on $\partial \Omega_{\varepsilon}(t)$ we have

$$\int_{\Omega_{\varepsilon}(t)} f(x) \, dx = \int_{\Gamma(t)} \int_{-\varepsilon}^{\varepsilon} f(y + r\nu_{\Gamma}(y, t)) J(y, t, r) \, dr \, d\mathcal{H}^{n-1}(y) \tag{3.5.4}$$

and

$$\int_{\partial\Omega_{\varepsilon}(t)} g(x) \, d\mathcal{H}^{n-1}(x) = \int_{\Gamma(t)} g(y + \varepsilon\nu_{\Gamma}(y,t)) J(y,t,\varepsilon) \, d\mathcal{H}^{n-1}(y) \\ + \int_{\Gamma(t)} g(y - \varepsilon\nu_{\Gamma}(y,t)) J(y,t,-\varepsilon) \, d\mathcal{H}^{n-1}(y). \quad (3.5.5)$$

Proof of Theorem 3.5.3. As in the proof of Theorem 3.4.1, we use the abbreviations (3.4.11) and (3.4.12). Let us calculate each term of (3.5.1). We expand $\omega(\rho)$ in powers of the signed distance d(x,t) as

$$\omega(\rho(x,t)) = \omega(\eta(\pi,t)) + d(x,t)\omega^1(\pi,t) + R(d^2), \quad (x,t) \in Q_{\varepsilon,T}.$$

Here the zeroth order term is $\omega(\eta(\pi, t))$ since the zeroth order term of $\rho(x, t)$ is $\eta(\pi, t)$. We divide the integral of $\omega(\rho)$ over $\Omega_{\varepsilon}(t)$ as

$$\int_{\Omega_{\varepsilon}(t)} \omega(\rho(x,t)) \, dx = I_1 + I_2 + I_3,$$

where

$$I_1 := \int_{\Omega_{\varepsilon}(t)} \omega(\eta(\pi, t)) \, dx, \quad I_2 := \int_{\Omega_{\varepsilon}(t)} d(x, t) \omega^1(\pi, t) \, dx, \quad I_3 := \int_{\Omega_{\varepsilon}(t)} R(d(x, t)^2) \, dx.$$

By the co-area formula (3.5.4) and the fact that J(y,t,r) is a polynomial in r with J(y,t,0) = 1 whose coefficients are polynomials in the principal curvatures, we have

$$I_1 = \int_{\Gamma(t)} \int_{-\varepsilon}^{\varepsilon} \omega(\eta(y,t)) J(y,t,r) \, dr \, d\mathcal{H}^{n-1}(y) = 2\varepsilon \int_{\Gamma(t)} \omega(\eta(y,t)) \, d\mathcal{H}^{n-1}(y) + \varepsilon^2 f_1(\varepsilon,t),$$

where $f_1(\varepsilon, t)$ is a polynomial in ε with time-dependent coefficients. Therefore,

$$\frac{dI_1}{dt} = 2\varepsilon \frac{d}{dt} \int_{\Gamma(t)} \omega(y, t) \, d\mathcal{H}^{n-1}(y) + O(\varepsilon^2). \tag{3.5.6}$$

Similarly we have

$$I_2 = \int_{\Gamma(t)} \int_{-\varepsilon}^{\varepsilon} r \omega^1(y, t) J(y, t, r) \, dr \, d\mathcal{H}^{n-1}(y) = \varepsilon^2 f_2(\varepsilon, t)$$

with a polynomial $f_2(\varepsilon, t)$ in ε with time-dependent coefficients and thus

$$\frac{dI_2}{dt} = O(\varepsilon^2). \tag{3.5.7}$$

We apply the Reynolds transport theorem to the time derivative of I_3 . Then, since the time derivative of $R(d(x,t)^2)$ is R(d(x,t)), we have

$$\frac{dI_3}{dt} = \int_{\Omega_{\varepsilon}(t)} R(d(x,t)) \, dx + \int_{\partial\Omega_{\varepsilon}(t)} R(d(x,t)^2) V_{\Omega}^N(x,t) \, d\mathcal{H}^{n-1}(x).$$

Since J(y, t, r) is bounded independently of ε , the co-area formula (3.5.4) yields

$$\int_{\Omega_{\varepsilon}(t)} R(d(x,t)) \, dx = \int_{\Gamma(t)} \int_{-\varepsilon}^{\varepsilon} R(r) J(y,t,r) \, dr \, d\mathcal{H}^{n-1}(y) = O(\varepsilon^2).$$

Moreover, applying (3.4.13) and (3.5.5) to the integral of $R(d(x,t)^2)V_{\Omega}^N(x,t)$ over $\partial\Omega_{\varepsilon}(t)$ and observing that $R(d(x,t)^2) = R(\varepsilon^2)$ holds for $x \in \partial\Omega_{\varepsilon}(t)$ we have

$$\int_{\partial\Omega_{\varepsilon}(t)} R(d(x,t)^2) V_{\Omega}^N(x,t) \, d\mathcal{H}^{n-1}(x) = O(\varepsilon^2).$$

Thus, we get the estimate

$$\frac{dI_3}{dt} = O(\varepsilon^2). \tag{3.5.8}$$

Since the integral of $\omega(\rho)$ over $\Omega_{\varepsilon}(t)$ is the sum of I_1 , I_2 , and I_3 , it follows from (3.5.6), (3.5.7), and (3.5.8) that

$$\frac{d}{dt} \int_{\Omega_{\varepsilon}(t)} \omega(\rho(x,t)) \, dx = 2\varepsilon \frac{d}{dt} \int_{\Gamma(t)} \omega(\eta(y,t)) \, d\mathcal{H}^{n-1}(y) + O(\varepsilon^2). \tag{3.5.9}$$

Next we calculate the first term on the right-hand side of (3.5.1). From the expansions (3.4.6) and (3.4.7), the product $\rho |u|^2$ is of the form

$$[\rho|u|^2](x,t) = [\eta|v|^2](\pi,t) + R(d), \quad (x,t) \in Q_{\varepsilon,T}$$

Hence, by (3.5.4),

$$\int_{\Omega_{\varepsilon}(t)} [\rho |u|^{2}](x,t) dx = \int_{\Gamma(t)} \int_{-\varepsilon}^{\varepsilon} \{ [\eta |v|^{2}](y,t) + R(r) \} J(y,t,r) dr d\mathcal{H}^{n-1}(y)$$

$$= 2\varepsilon \int_{\Gamma(t)} [\eta |v|^{2}](y,t) d\mathcal{H}^{n-1}(y) + O(\varepsilon^{2}).$$
(3.5.10)

Let us compute the last term on the right-hand side of (3.5.1). We expand the pressure $p(\rho)$ in d(x,t) as (3.4.22). Then, by the assumption that v is of the form (3.4.8) and Darcy's law (3.4.3) holds, we get (3.4.23) and (3.4.25) as in the proof of Theorem 3.4.1 and thus we can write

$$p(\rho(x,t)) = p(\eta(\pi,t)) - d(x,t)[\eta V_{\Gamma}^{N}](\pi,t) + R(d^{2}), \quad (x,t) \in Q_{\varepsilon,T}.$$

Therefore, by (3.4.13) and (3.5.5),

$$\int_{\partial\Omega_{\varepsilon}(t)} [p(\rho)V_{\Omega}^{N}](x,t) \, d\mathcal{H}^{n-1}(x) = J_1 + J_2 + O(\varepsilon^2),$$

where

$$J_1 := \int_{\Gamma(t)} [p(\eta)V_{\Gamma}^N](y,t) \{ J(y,t,\varepsilon) - J(y,t,-\varepsilon) \} d\mathcal{H}^{n-1}(y),$$

$$J_2 := -\varepsilon \int_{\Gamma(t)} [\eta|V_{\Gamma}^N|^2](y,t) \{ J(y,t,\varepsilon) + J(y,t,-\varepsilon) \} d\mathcal{H}^{n-1}(y).$$

By (3.5.3) and (3.2.6) we have

$$\begin{split} J(y,t,\varepsilon) &- J(y,t,-\varepsilon) = -2\varepsilon H(y,t) + O(\varepsilon^2), \\ J(y,t,\varepsilon) &+ J(y,t,-\varepsilon) = 2 + O(\varepsilon). \end{split}$$

Hence it follows that

$$J_1 = -2\varepsilon \int_{\Gamma(t)} [p(\eta)V_{\Gamma}^N H](y,t) d\mathcal{H}^{n-1}(y) + O(\varepsilon^2),$$

$$J_2 = -2\varepsilon \int_{\Gamma(t)} [\eta|V_{\Gamma}^N|^2](y,t) d\mathcal{H}^{n-1}(y) + O(\varepsilon^2)$$

and the integral of $p(\rho)V_{\Omega}^{N}$ over $\partial\Omega_{\varepsilon}(t)$ becomes

$$\int_{\partial\Omega_{\varepsilon}(t)} [p(\rho)V_{\Omega}^{N}](x,t) \, d\mathcal{H}^{n-1}(x) = -2\varepsilon \int_{\Gamma(t)} [p(\eta)V_{\Gamma}^{N}H](y,t) \, d\mathcal{H}^{n-1}(y) - 2\varepsilon \int_{\Gamma(t)} [\eta|V_{\Gamma}^{N}|^{2}](y,t) \, d\mathcal{H}^{n-1}(y) + O(\varepsilon^{2}). \quad (3.5.11)$$

Finally, substituting (3.5.9), (3.5.10), and (3.5.11) for (3.5.1), dividing both sides by 2ε , and observing that $|v|^2 = |V_{\Gamma}^N|^2 + |v^T|^2$ we obtain

$$\begin{split} \frac{d}{dt} \int_{\Gamma(t)} \omega(\eta(y,t)) \, d\mathcal{H}^{n-1}(y) &= -\int_{\Gamma(t)} [\eta |v^T|^2](y,t) \, d\mathcal{H}^{n-1}(y) \\ &+ \int_{\Gamma(t)} [p(\eta) V_{\Gamma}^N H](y,t) \, d\mathcal{H}^{n-1}(y) + O(\varepsilon). \end{split}$$

In the above equality all terms except for $O(\varepsilon)$ are independent of ε . Hence we conclude that η and v satisfy the energy law (3.5.2).

Remark 3.5.5 (A failure of a simple rescaling argument for a moving surface). It is possible to derive the limit energy identity by a rescaling argument. However, derivation by a rescaling argument is somewhat misleading. Let ρ and u satisfy the energy identity (3.5.1) in the moving thin domain $\Omega(t) = \Omega_{\varepsilon}(t)$. We set

$$\eta(y,t,r) := \rho(y + \varepsilon r \nu_{\Gamma}(y,t),t), \quad v(y,t,r) := u(y + \varepsilon r \nu_{\Gamma}(y,t),t)$$

for $(y,t) \in S_T$ and $r \in (-1,1)$. Then by (3.4.13) and the integral transformation formulas (3.5.4) and (3.5.5) we can write (3.5.1) in terms of η and v as

$$\varepsilon \frac{d}{dt} \int_{\Gamma(t)} \int_{-1}^{1} \omega(\eta(y,r)) J(y,\varepsilon r) \, dr \, d\mathcal{H}^{n-1}(y)$$

$$= -\varepsilon \int_{\Gamma(t)} \int_{-1}^{1} [\eta|v|^2](y,r) J(y,\varepsilon r) \, dr \, d\mathcal{H}^{n-1}(y) - \int_{\Gamma(t)} p(\eta(y,1)) V_{\Gamma}^N(y) J(y,\varepsilon) \, d\mathcal{H}^{n-1}(y)$$

$$+ \int_{\Gamma(t)} p(\eta(y,-1)) V_{\Gamma}^N(y) J(y,-\varepsilon) \, d\mathcal{H}^{n-1}(y). \quad (3.5.12)$$

Here we used the abbreviation (3.4.12) and suppressed the argument t of functions.

In formal derivation of a thin width limit by a rescaling argument we usually assume that rescaled functions are independent of the thin direction to get limit relations on a limit set. However, making the assumption at an inappropriate point may result in a wrong limit. To see this, let us assume that η and v are independent of the variable r in (3.5.12). Then since

$$J(y,\varepsilon) - J(y,-\varepsilon) = -2\varepsilon H(y) + O(\varepsilon^2)$$

by (3.5.3) and (3.2.6), it follows that

$$\begin{aligned} 2\varepsilon \frac{d}{dt} \int_{\Gamma(t)} \omega(\eta(y)) \, d\mathcal{H}^{n-1}(y) &= -2\varepsilon \int_{\Gamma(t)} [\eta|v|^2](y) \, d\mathcal{H}^{n-1}(y) \\ &+ 2\varepsilon \int_{\Gamma(t)} [p(\eta)V_{\Gamma}^N H](y) \, d\mathcal{H}^{n-1}(y) + O(\varepsilon^2). \end{aligned}$$

Dividing both sides by 2ε and taking the principal term we obtain

$$\frac{d}{dt} \int_{\Gamma(t)} \omega(\eta) \, d\mathcal{H}^{n-1} = -\int_{\Gamma(t)} \eta |v|^2 \, d\mathcal{H}^{n-1} + \int_{\Gamma(t)} p(\eta) V_{\Gamma}^N H \, d\mathcal{H}^{n-1}.$$

In this equality v should be of the form $v = V_{\Gamma}^{N} \nu_{\Gamma} + v^{T}$ with some tangential velocity field v^{T} , since it is the velocity of a substance on the moving surface $\Gamma(t)$ with normal velocity V_{Γ}^{N} . Hence we get

$$\frac{d}{dt} \int_{\Gamma(t)} \omega(\eta) \, d\mathcal{H}^{n-1} = -\int_{\Gamma(t)} \eta(|V_{\Gamma}^N|^2 + |v^T|^2) \, d\mathcal{H}^{n-1} + \int_{\Gamma(t)} p(\eta) V_{\Gamma}^N H \, d\mathcal{H}^{n-1}$$

which includes an additional term $\int_{\Gamma(t)} \eta |V_{\Gamma}^{N}|^2 d\mathcal{H}^{n-1}$ compared to the limit energy identity (3.5.2). This improper term appears because we ignore the difference between $p(\eta(y,t,1))$ and $p(\eta(y,t,-1))$ in (3.5.12) by assuming that η is independent of the variable r. Of course it vanishes if the shape of the surface does not change, i.e. $V_{\Gamma}^{N} = 0$. This is the reason why this simple rescaling argument is popular to derive a thin width limit problem in a formal level when the limit set of a thin domain does not change its shape.

Remark 3.5.6 (Corrected rescaling argument). To obtain the correct limit (3.5.2) we should take into account the difference between $p(\eta(y,t,1))$ and $p(\eta(y,t,-1))$ in (3.5.12). Let us rewrite the sum of the last two terms in the right-hand side of (3.5.12) into the sum of

$$I_{1} = -\int_{\Gamma(t)} \{p(\eta(y,1)) - p(\eta(y,-1))\} V_{\Gamma}^{N}(y) \, d\mathcal{H}^{n-1}(y),$$

$$I_{2} = \varepsilon \int_{\Gamma(t)} \{p(\eta(y,1)) + p(\eta(y,-1))\} [V_{\Gamma}^{N}H](y) \, d\mathcal{H}^{n-1}(y),$$

and a residual term $O(\varepsilon^2)$ and calculate them properly (here we again suppressed the argument t of functions). For I_2 we merely assume that η is independent of r to get

$$I_2 = 2\varepsilon \int_{\Gamma(t)} [p(\eta)V_{\Gamma}^N H](y) \, d\mathcal{H}^{n-1}(y). \tag{3.5.13}$$

For a proper calculation of I_1 we need to impose Darcy's law (3.4.3) in $\Omega_{\varepsilon}(t)$ and describe it in terms of the rescaled functions. By the definition of η ,

$$p(\rho(x)) = p(\eta(\pi(x), \varepsilon^{-1}d(x))), \quad x \in \Omega_{\varepsilon}(t).$$

We differentiate both sides in x and use (3.2.2) and (3.4.17). Then

$$\nabla p(\rho(x)) = \nabla_{\Gamma} p(\eta(\pi, \varepsilon^{-1}d)) + \varepsilon^{-1} \partial_r p(\eta(\pi, \varepsilon^{-1}d)) \nu_{\Gamma}(\pi) + O(\varepsilon),$$

where we abbreviate $\pi(x)$ and d(x) to π and d in the right-hand side. Substituting this for (3.4.3) and taking the normal component of the resulting equation we obtain

$$\partial_r p(\eta(y,r)) = -\varepsilon[\eta v](y,r) \cdot \nu_{\Gamma}(y) + O(\varepsilon^2)$$
(3.5.14)

for $y \in \Gamma(t)$ and $r \in (-1, 1)$. We apply the mean value theorem and (3.5.14) to the difference between $p(\eta(y, 1))$ and $p(\eta(y, -1))$. Then

$$p(\eta(y,1)) - p(\eta(y,-1)) = 2\partial_r p(\eta(y,\theta)) = -2\varepsilon[\eta v](y,\theta) \cdot \nu_{\Gamma}(y) + O(\varepsilon^2)$$

with some $\theta = \theta(y, t) \in (-1, 1)$. Hence I_1 is expressed as

$$I_1 = 2\varepsilon \int_{\Gamma(t)} [\eta v](y,\theta) \cdot \nu_{\Gamma}(y) V_{\Gamma}^N(y) \, d\mathcal{H}^{n-1}(y) + O(\varepsilon^2).$$

Now we assume that η and v are independent of the argument r. Then we have

$$I_1 = 2\varepsilon \int_{\Gamma} [\eta(v \cdot \nu_{\Gamma}) V_{\Gamma}^N](y) \, d\mathcal{H}^{n-1}(y) + O(\varepsilon^2). \tag{3.5.15}$$

We substitute (3.5.13) and (3.5.15) for (3.5.12), assume that the rescaled functions are constant in the variable r for the left-hand side and the first term on the right-hand side, and divide both sides by 2ε after calculations. Then the principal term on the resulting equation is

$$\begin{aligned} \frac{d}{dt} \int_{\Gamma(t)} \omega(\eta) \, d\mathcal{H}^{n-1} &= -\int_{\Gamma(t)} \eta |v|^2 \, d\mathcal{H}^{n-1} \\ &+ \int_{\Gamma(t)} \eta(v \cdot \nu_{\Gamma}) V_{\Gamma}^N \, d\mathcal{H}^{n-1} + \int_{\Gamma(t)} p(\eta) V_{\Gamma}^N H \, d\mathcal{H}^{n-1}. \end{aligned}$$

Finally we suppose that v is of the form $v = V_{\Gamma}^{N}\nu_{\Gamma} + v^{T}$ with some tangential velocity v^{T} , which is natural since it is the velocity of a substance on the moving surface $\Gamma(t)$ with normal velocity V_{Γ}^{N} as we mentioned in Remark 3.5.5. Then we obtain the proper limit energy identity (3.5.2) from the above equality.

3.6 Energetic variation for derivation of Darcy's law

In this section we discuss the energetic variational approach [3, 12, 25] for nonlinear diffusion equations in a moving domain and on a moving surface. For a general non-equilibrium thermodynamic system, if the system is isothermal, then the combination of the first and second laws of thermodynamics yields

$$\frac{d}{dt}E^{\text{total}} = \dot{W} - \Delta,$$

where $E^{\text{total}} = \mathcal{K} + \mathcal{F}$ is the sum of the kinetic energy \mathcal{K} and the Helmholtz free energy \mathcal{F} , Δ is the entropy production, and \dot{W} is the rate of change of work done by the external environment. If the system is closed, i.e. $\dot{W} = 0$, we further get the energy dissipation law

$$\frac{d}{dt}E^{\text{total}} = -2\mathcal{D},$$

where $\mathcal{D} = \Delta/2$ is sometimes called the energy dissipation. For a conservative system $(\Delta = 0)$, the principle of least action (LAP) [1] states that the variation of the kinetic and the free energies with respect to the flow map in Lagrangian coordinates yield the internal force F_i and the conservative force F_c . Formally it can be written as

$$\delta\left(\int_0^T \mathcal{K} \, dt\right) = \int_0^T \int (F_i \cdot \delta x) \, dx \, dt,$$
$$\delta\left(\int_0^T \mathcal{F} \, dt\right) = \int_0^T \int (F_c \cdot \delta x) \, dx \, dt,$$

where δ represents the procedure of variation. Sometimes such a calculation is also referred to as the principle of virtual work. Based on the LAP, the equation of motion for a conservative system is described by balance of forces:

$$F_i = F_c$$

For a dissipative system, we use the maximum dissipation principle (MDP) [16,17] to the dissipative force F_d : by taking the variation of the dissipation with respect to the velocity in Eulerian coordinates, we have

$$\delta \mathcal{D} = F_d \cdot \delta u.$$

When all forces are derived, the equation of motion for a dissipative system is formulated as balance of forces (Newton's third law):

$$F_i = F_c + F_d.$$

Let us apply the above energetic variational framework to the energy laws (3.5.1) and (3.5.2). For (3.5.1) we have

$$\mathcal{K} = 0, \quad \mathcal{F} = \int_{\Omega(t)} \omega(\rho) \, dx,$$
$$\mathcal{D} = \frac{1}{2} \int_{\Omega(t)} \rho |u|^2 \, dx, \quad \dot{W} = -\int_{\partial \Omega(t)} p(\rho) V_{\Omega}^N \, d\mathcal{H}^{n-1}.$$
Let $x: \Omega(0) \times [0,T] \to \mathbb{R}^n$ be the flow map of the velocity field u, i.e. for each $t \in [0,T]$ the mapping $x(\cdot,t)$ is a diffeomorphism from $\Omega(0)$ onto $\Omega(t)$ and

$$x(X,0) = X, \quad \frac{\partial}{\partial t}x(X,t) = u(x(X,t),t), \quad (X,t) \in \Omega(0) \times (0,T).$$

We write F for the deformation matrix of x:

$$F(X,t) = \frac{\partial x}{\partial X}(X,t), \quad (X,t) \in \Omega(0) \times (0,T).$$

The MDP gives the dissipative force

$$\frac{\delta \mathcal{D}}{\delta u} = \rho u. \tag{3.6.1}$$

On the other hand, the LAP shows that the conservative force is given by the gradient of the pressure.

Lemma 3.6.1. Suppose that ρ and u satisfy the transport equation (3.3.2). Then

$$\frac{\delta \mathcal{F}}{\delta x} = \nabla p(\rho), \qquad (3.6.2)$$

where $p(\rho)$ is given by (3.4.1).

Proof. Throughout the proof we use the notation

$$f^{\sharp}(X,t) = f(x(X,t),t), \quad (X,t) \in \Omega(0) \times (0,T)$$

for a function f on Q_T . Since the transport equation (3.3.2) is satisfied, the density ρ is given by

$$\rho(x(X,t),t) = \frac{\rho_0(X)}{\det F(X,t)}, \quad (X,t) \in \Omega(0) \times (0,T)$$
(3.6.3)

with the initial density ρ_0 and thus the free energy $\mathcal{F}(x) = \mathcal{F}(x(\cdot, t))$ is of the form

$$\mathcal{F}(x) = \int_{\Omega(0)} \omega\left(\frac{\rho_0(X)}{\det F(X,t)}\right) \det F(X,t) \, dX, \quad t \in (0,T).$$
(3.6.4)

Let $\{x^{\varepsilon}\}_{\varepsilon}$ be a family of flow maps and $u^{\varepsilon} = \partial x^{\varepsilon} / \partial t$ such that

$$\begin{aligned} x^{\varepsilon}(\cdot,0) &= x(\cdot,0), \quad x^{\varepsilon}(\cdot,T) = x(\cdot,T) \quad \text{for all} \quad \varepsilon, \\ x^{\varepsilon}(\cdot,t)|_{\varepsilon=0} &= x(\cdot,t), \quad u^{\varepsilon}(\cdot,t)|_{\varepsilon=0} = u(\cdot,t), \quad \left. \frac{d}{d\varepsilon} x^{\varepsilon}(\cdot,t) \right|_{\varepsilon=0} = w(x(\cdot,t),t) \end{aligned}$$

with any given vector field $w: Q_T \to \mathbb{R}^n$. We write F^{ε} for the deformation matrix of x^{ε} . Suppose that ρ^{ε} and u^{ε} satisfy the transport equation (3.3.2) with the same initial density ρ_0 . Then the relation (3.6.3) with ρ , x, and F replaced by ρ^{ε} , x^{ε} , and F^{ε} holds and by (3.6.4) the free energy \mathcal{F} with respect to the perturbed flow map x^{ε} is given by

$$\mathcal{F}(x^{\varepsilon}) = \int_{\Omega(0)} \omega\left(\frac{\rho_0}{\det F^{\varepsilon}}\right) \det F^{\varepsilon} \, dX. \tag{3.6.5}$$

Note that the argument $t \in (0,T)$ is suppressed in the above equality. We differentiate $\int_0^T \mathcal{F}(x^{\varepsilon}) dt$ with respect to ε at $\varepsilon = 0$. Since $F^{\varepsilon}|_{\varepsilon=0} = F$ and

$$\left. \frac{dF^{\varepsilon}}{d\varepsilon} \right|_{\varepsilon=0} = \frac{\partial}{\partial X} \frac{dx^{\varepsilon}}{d\varepsilon} \right|_{\varepsilon=0} = \frac{\partial w^{\sharp}}{\partial X}$$

the derivative of the determinant of F^{ε} with respect to ε at $\varepsilon = 0$ is

$$\frac{d}{d\varepsilon} \det F^{\varepsilon} \Big|_{\varepsilon=0} = \operatorname{tr} \left((F^{\varepsilon})^{-1} \frac{dF^{\varepsilon}}{d\varepsilon} \right) \det F^{\varepsilon} \Big|_{\varepsilon=0}$$

$$= \operatorname{tr} \left(F^{-1} \frac{\partial w^{\sharp}}{\partial X} \right) \det F = (\operatorname{div} w)^{\sharp} \det F,$$
(3.6.6)

where F^{-1} and $(F^{\varepsilon})^{-1}$ are the inverse matrix of F and F^{ε} , respectively. We differentiate the integrand of (3.6.5) at $\varepsilon = 0$ and apply (3.6.3), (3.6.6), and $F^{\varepsilon}|_{\varepsilon=0} = F$ to obtain

$$\frac{d}{d\varepsilon} \left(\omega \left(\frac{\rho_0}{\det F^{\varepsilon}} \right) \det F^{\varepsilon} \right) \Big|_{\varepsilon=0} = \{ -\omega'(\rho^{\sharp})\rho^{\sharp} + \omega(\rho^{\sharp}) \} (\operatorname{div} w)^{\sharp} \det F.$$

Therefore,

$$\frac{d}{d\varepsilon} \int_0^T \mathcal{F}(x^{\varepsilon}) dt \Big|_{\varepsilon=0} = \int_0^T \int_{\Omega(0)} \{-\omega'(\rho^{\sharp})\rho^{\sharp} + \omega(\rho^{\sharp})\} (\operatorname{div} w)^{\sharp} \operatorname{det} F \, dX \, dt$$
$$= \int_0^T \int_{\Omega(t)} \{-\omega'(\rho)\rho + \omega(\rho)\} \operatorname{div} w \, dx \, dt$$
$$= \int_0^T \int_{\Omega(t)} \nabla[\omega'(\rho)\rho - \omega(\rho)] \cdot w \, dx \, dt$$

and (3.6.2) follows.

By (3.6.1), (3.6.2), and $\mathcal{K} = 0$, the balance of forces

$$\frac{\delta \mathcal{K}}{\delta x} = \frac{\delta \mathcal{F}}{\delta x} + \frac{\delta \mathcal{D}}{\delta u}$$

is of the form

$$0 = \nabla p(\rho) + \rho u$$
, i.e. $-\rho u = \nabla p(\rho)$,

which is exactly Darcy's law in a moving domain. Combining this with the transport equation (3.3.2), we obtain the nonlinear diffusion equations

$$\partial_t \rho + \operatorname{div}(\rho u) = 0, \quad -\rho u = \nabla p(\rho)$$

in the moving domain $\Omega(t)$, where $p(\rho)$ is given by (3.4.1).

From the above discussion, we expect that the energetic variational approach for (3.5.2) yields Darcy' law (3.4.10) on a moving surface. For (3.5.2) we have

$$\mathcal{K} = 0, \quad \mathcal{F} = \int_{\Gamma(t)} \omega(\eta) \, d\mathcal{H}^{n-1},$$
$$\mathcal{D} = \frac{1}{2} \int_{\Gamma(t)} \eta |v^T|^2 \, d\mathcal{H}^{n-1}, \quad \dot{W} = \int_{\Gamma(t)} p(\eta) H V_{\Gamma}^N \, d\mathcal{H}^{n-1}.$$

The variation of $\mathcal D$ with respect to the total velocity $v = V_\Gamma^N \nu + v^T$ gives

$$\frac{\delta \mathcal{D}}{\delta v} = \eta v^T, \tag{3.6.7}$$

since $v^T = P_{\Gamma} v$. Let us apply the LAP to the free energy \mathcal{F} .

Lemma 3.6.2. Suppose that η and v of the form (3.4.8) satisfy the transport equation (3.3.5). Then

$$\frac{\delta \mathcal{F}}{\delta y} = \nabla_{\Gamma} p(\eta), \qquad (3.6.8)$$

where $p(\eta)$ is given by (3.4.1).

We localize integrals over $\Gamma(t)$ with a partition of unity of $\Gamma(t)$ as in [14, Section 2.4] and take the variation of \mathcal{F} with respect to a flow map in "local Lagrangian coordinates." Let U be an open set in \mathbb{R}^{n-1} . We call a mapping $y: U \times [0,T] \to \mathbb{R}^n$ the flow map of the velocity $v = V_{\Gamma}\nu_{\Gamma} + v^T$ in local Lagrangian coordinates if $y(\cdot, t): U \to \Gamma(t)$ is a smooth local parametrization of $\Gamma(t)$ for each $t \in [0,T]$ and

$$y(Y,0) \in \Gamma(0), \quad \frac{\partial}{\partial t} y(Y,t) = v(y(Y,t),t), \quad (Y,t) \in U \times (0,T).$$
(3.6.9)

We consider a localized surface integral

$$\mathcal{F}(y) = \mathcal{F}(y(\cdot, t)) = \int_{y(U,t)} \omega(\eta) \, d\mathcal{H}^{n-1} \tag{3.6.10}$$

and take its variation with respect to y. Let $\{y^{\varepsilon}\}_{\varepsilon}$ be a family of flow maps in local Lagrangian coordinates and $v^{\varepsilon} = \partial y^{\varepsilon}/\partial t$ such that

$$y^{\varepsilon}(\cdot, 0) = y(\cdot, 0), \quad y^{\varepsilon}(\cdot, T) = y(\cdot, T) \quad \text{for all} \quad \varepsilon,$$

$$y^{\varepsilon}(\cdot, t)|_{\varepsilon=0} = y(\cdot, t), \quad v^{\varepsilon}(\cdot, t)|_{\varepsilon=0} = v(\cdot, t), \quad \frac{d}{d\varepsilon}y^{\varepsilon}(\cdot, t)\Big|_{\varepsilon=0} = w(y(\cdot, t), t)$$
(3.6.11)

with any given vector field $w: S_T \to \mathbb{R}^n$ such that $w(\cdot, t)$ is tangential on $\Gamma(t)$ for each $t \in (0, T)$. For a function f on S_T we use the notation

$$f^{\sharp}(Y,t) = f(y(Y,t),t), \quad (Y,t) \in U \times (0,T).$$
 (3.6.12)

Lemma 3.6.3. Let $g = (g_{ij})_{i,j}$ be a matrix given by

$$g_{ij} = \frac{\partial y}{\partial Y_i} \cdot \frac{\partial y}{\partial Y_j}, \quad i, j = 1, \dots, n-1$$
 (3.6.13)

and $g^{\varepsilon} = (g_{ij}^{\varepsilon})_{i,j}$ be a matrix given as above with y replaced by y^{ε} . Then

$$\frac{d}{d\varepsilon}\sqrt{\det g^{\varepsilon}}\Big|_{\varepsilon=0} = (\operatorname{div}_{\Gamma}w)^{\sharp}\sqrt{\det g}.$$
(3.6.14)

Proof. Since $g^{\varepsilon}|_{\varepsilon=0} = g$ and

$$\frac{d}{d\varepsilon} \det g^{\varepsilon} = \operatorname{tr}\left((g^{\varepsilon})^{-1} \frac{dg^{\varepsilon}}{d\varepsilon} \right) \det g^{\varepsilon},$$

where $(g^{\varepsilon})^{-1}$ is the inverse matrix of g^{ε} , we have

$$\frac{d}{d\varepsilon}\sqrt{\det g^{\varepsilon}}\Big|_{\varepsilon=0} = \frac{1}{2}\mathrm{tr}\left(g^{-1}\frac{dg^{\varepsilon}}{d\varepsilon}\Big|_{\varepsilon=0}\right)\sqrt{\det g},\tag{3.6.15}$$

where $g^{-1} = (g^{ij})_{i,j}$ is the inverse matrix of g. Moreover, since

$$\frac{dg_{ij}^{\varepsilon}}{d\varepsilon}\Big|_{\varepsilon=0} = \left(\frac{\partial}{\partial Y_i}\frac{dy^{\varepsilon}}{d\varepsilon} \cdot \frac{\partial y^{\varepsilon}}{\partial Y_j} + \frac{\partial y^{\varepsilon}}{\partial Y_j} \cdot \frac{\partial}{\partial Y_j}\frac{dy^{\varepsilon}}{d\varepsilon}\right)\Big|_{\varepsilon=0} = \frac{\partial w^{\sharp}}{\partial Y_i} \cdot \frac{\partial y}{\partial Y_j} + \frac{\partial y}{\partial Y_i} \cdot \frac{\partial w^{\sharp}}{\partial Y_j}$$

for each i, j = 1, ..., n - 1, where we used the notation (3.6.12), and g^{-1} is symmetric,

$$\operatorname{tr}\left(g^{-1}\frac{dg^{\varepsilon}}{d\varepsilon}\Big|_{\varepsilon=0}\right) = \sum_{i,j=1}^{n-1} g^{ij} \left(\frac{\partial w^{\sharp}}{\partial Y_{i}} \cdot \frac{\partial y}{\partial Y_{j}} + \frac{\partial y}{\partial Y_{i}} \cdot \frac{\partial w^{\sharp}}{\partial Y_{j}}\right)$$
$$= 2\sum_{i,j=1}^{n-1} g^{ij}\frac{\partial w^{\sharp}}{\partial Y_{i}} \cdot \frac{\partial y}{\partial Y_{j}} = 2(\operatorname{div}_{\Gamma}w)^{\sharp}.$$

Substituting this for (3.6.15), we get (3.6.14).

Proof of Lemma 3.6.2. We first express the free energy \mathcal{F} in "local Lagrangian coordinates." Let U be an open set in \mathbb{R}^{n-1} and $y: U \times [0,T] \to \mathbb{R}^n$ be the flow map of the velocity $v = V_{\Gamma}\nu_{\Gamma} + v^T$ in local Lagrangian coordinates. Also, let $g = (g_{ij})_{i,j}$ be the matrix given by (3.6.13). For every open subset U' of U the integral

$$\int_{U'(t)} \eta(y,t) \, d\mathcal{H}^{n-1}(y) = \int_{U'} \eta(y(Y,t),t) \sqrt{\det g(Y,t)} \, dY \quad (U'(t) := y(U',t))$$

is constant in t, since η and v satisfy the transport equation (3.3.5) and U'(t) moves with velocity v. Hence for $(Y,t) \in U \times (0,T)$ we have

$$\eta(y(Y,t),t)\sqrt{\det g(Y,t)} = \eta(y(Y,0),0)\sqrt{\det g(Y,0)}$$
(3.6.16)

and the localized surface integral (3.6.10) is expressed as

$$\mathcal{F}(y) = \int_{U} \omega(\eta(y(Y,t),t)) \sqrt{\det g(Y,t)} \, dY$$

=
$$\int_{U} \omega\left(\frac{\eta_0(Y)}{\sqrt{\det g(Y,t)}}\right) \sqrt{\det g(Y,t)} \, dY,$$
 (3.6.17)

where $\eta_0(Y)$ is given by the right-hand side of (3.6.16).

Next we take a variation of \mathcal{F} with respect to the flow map y. Let $\{y^{\varepsilon}\}_{\varepsilon}$ be a family of flow maps in local Lagrangian coordinates satisfying (3.6.11) with $v^{\varepsilon} = \partial y^{\varepsilon}/\partial t$. Also, let $g^{\varepsilon} = (g_{ij}^{\varepsilon})_{i,j}$ be given by (3.6.13) with y replaced by y^{ε} . Suppose that η^{ε} and v^{ε} satisfy the transport equation (3.3.5) and $\eta^{\varepsilon}|_{t=0} = \eta|_{t=0}$ holds on $y^{\varepsilon}(U,0) = y(U,0)$. Then the relation (3.6.16) with η^{ε} , y^{ε} , and g^{ε} holds and by (3.6.17) the free energy \mathcal{F} with respect to the perturbed flow map y^{ε} is given by

$$\mathcal{F}(y^{\varepsilon}) = \int_{U} \omega\left(\frac{\eta_{0}}{\sqrt{\det g^{\varepsilon}}}\right) \sqrt{\det g^{\varepsilon}} \, dY.$$

Here the the argument $t \in (0, T)$ is suppressed. Note that the right-hand side of (3.6.16) with η^{ε} , y^{ε} , and g^{ε} is equal to $\eta_0(Y)$ since $\eta^{\varepsilon}|_{t=0} = \eta|_{t=0}$, $y^{\varepsilon}|_{t=0} = y|_{t=0}$, and $g^{\varepsilon}|_{t=0} = g|_{t=0}$. We differentiate the integrand of the right-hand side with respect to ε at $\varepsilon = 0$. Then by (3.6.14), (3.6.17), and $g^{\varepsilon}|_{\varepsilon=0} = g$ we get

$$\frac{d}{d\varepsilon} \left(\omega \left(\frac{\eta_0}{\sqrt{\det g^{\varepsilon}}} \right) \sqrt{\det g^{\varepsilon}} \right) \Big|_{\varepsilon=0} = \{ -\omega'(\eta^{\sharp})\eta^{\sharp} + \omega(\eta^{\sharp}) \} (\operatorname{div}_{\Gamma} w)^{\sharp} \sqrt{\det g}.$$

Here we used the notation (3.6.12). Hence

$$\begin{aligned} \frac{d}{d\varepsilon} \int_0^T \mathcal{F}(y^\varepsilon) \, dt \Big|_{\varepsilon=0} &= \int_0^T \int_U \{-\omega'(\eta^\sharp)\eta^\sharp + \omega(\eta^\sharp)\} (\operatorname{div}_{\Gamma} w)^\sharp \sqrt{\det g} \, dY \, dt \\ &= \int_0^T \int_{U(t)} \{-\omega'(\eta)\eta + \omega(\eta)\} \operatorname{div}_{\Gamma} w \, d\mathcal{H}^{n-1} \, dt \\ &= \int_0^T \int_{U(t)} \nabla_{\Gamma} [\omega'(\eta)\eta - \omega(\eta)] \cdot w \, d\mathcal{H}^{n-1} \, dt, \end{aligned}$$

where U(t) = y(U,t) and the last equality follows from the Stokes theorem and the fact that the vector field w is tangential on $\Gamma(t)$. (Note that we may assume that $\omega'(\eta)\eta - \omega(\eta)$ has a compact support in U(t) since we localize the surface integral by using a partition of unity of $\Gamma(t)$.) Since w is an arbitrary tangential vector field on $\Gamma(t)$, we conclude from the above equality that (3.6.8) holds.

By (3.6.7), (3.6.8), and $\mathcal{K} = 0$, the balance of forces

$$\frac{\delta \mathcal{K}}{\delta y} = \frac{\delta \mathcal{F}}{\delta y} + \frac{\delta \mathcal{D}}{\delta v}$$

is of the form

$$0 = \nabla_{\Gamma} p(\eta) + \eta v^T, \quad \text{i.e.} \quad -\eta v^T = \nabla_{\Gamma} p(\eta),$$

which is Darcy's law on a moving surface as we expected. Finally, combining this with the transport equation (3.3.5) we obtain the nonlinear diffusion equations

$$\partial^{\circ} \eta - V_{\Gamma}^{N} H \eta + \operatorname{div}_{\Gamma}(\eta v^{T}) = 0, \quad -\eta v^{T} = \nabla_{\Gamma} p(\eta)$$

on the moving surface $\Gamma(t)$, where $p(\eta)$ is given by (3.4.1).

References

- V. I. Arnol'd, Mathematical methods of classical mechanics, Graduate Texts in Mathematics, vol. 60, Springer-Verlag, New York, 1989. Translated from the 1974 Russian original by K. Vogtmann and A. Weinstein; Corrected reprint of the second (1989) edition.
- [2] P. Cermelli, E. Fried, and M. E. Gurtin, Transport relations for surface integrals arising in the formulation of balance laws for evolving fluid interfaces, J. Fluid Mech. 544 (2005), 339–351.
- [3] Q. Du, C. Liu, R. Ryham, and X. Wang, Energetic variational approaches in modeling vesicle and fluid interactions, Phys. D 238 (2009), no. 9-10, 923–930.
- [4] G. Dziuk and C. M. Elliott, Finite elements on evolving surfaces, IMA J. Numer. Anal. 27 (2007), no. 2, 262–292.
- [5] G. Dziuk and C. M. Elliott, Finite element methods for surface PDEs, Acta Numer. 22 (2013), 289–396.

- [6] D. Gilbarg and N. S. Trudinger, Elliptic partial differential equations of second order, Classics in Mathematics, Springer-Verlag, Berlin, 2001. Reprint of the 1998 edition.
- [7] M. E. Gurtin, An introduction to continuum mechanics, Mathematics in Science and Engineering, vol. 158, Academic Press, Inc. [Harcourt Brace Jovanovich, Publishers], New York-London, 1981.
- [8] J. K. Hale and G. Raugel, A damped hyperbolic equation on thin domains, Trans. Amer. Math. Soc. 329 (1992), no. 1, 185–219.
- [9] J. K. Hale and G. Raugel, Reaction-diffusion equation on thin domains, J. Math. Pures Appl. (9) 71 (1992), no. 1, 33–95.
- [10] J. K. Hale and G. Raugel, A reaction-diffusion equation on a thin L-shaped domain, Proc. Roy. Soc. Edinburgh Sect. A 125 (1995), no. 2, 283–327.
- [11] L. T. Hoang and G. R. Sell, Navier-Stokes equations with Navier boundary conditions for an oceanic model, J. Dynam. Differential Equations 22 (2010), no. 3, 563–616.
- [12] Y. Hyon, D. Y. Kwak, and C. Liu, Energetic variational approach in complex fluids: maximum dissipation principle, Discrete Contin. Dyn. Syst. 26 (2010), no. 4, 1291–1304.
- [13] D. Iftimie, G. Raugel, and G. R. Sell, Navier-Stokes equations in thin 3D domains with Navier boundary conditions, Indiana Univ. Math. J. 56 (2007), no. 3, 1083–1156.
- [14] H. Koba, C. Liu, and Y. Giga, Energetic variational approaches for incompressible fluid systems on an evolving surface, Quart. Appl. Math. 75 (2017), no. 2, 359–389.
- [15] T.-H. Miura, Zero width limit of the heat equation on moving thin domains, Interfaces Free Bound. 19 (2017), no. 1, 31–77.
- [16] L. Onsager, Reciprocal relations in irreversible processes. I, Physical Review 37 (1931), no. 4, 405–426.
- [17] L. Onsager, Reciprocal relations in irreversible processes. II, Physical Review 38 (1931), no. 12, 2265– 2279.
- [18] J. V. Pereira and R. P. Silva, Reaction-diffusion equations in a noncylindrical thin domain, Bound. Value Probl. (2013), 2013:248, 10.
- [19] M. Prizzi, M. Rinaldi, and K. P. Rybakowski, Curved thin domains and parabolic equations, Studia Math. 151 (2002), no. 2, 109–140.
- [20] M. Prizzi and K. P. Rybakowski, The effect of domain squeezing upon the dynamics of reaction-diffusion equations, J. Differential Equations 173 (2001), no. 2, 271–320.
- [21] M. Prizzi and K. P. Rybakowski, On inertial manifolds for reaction-diffusion equations on genuinely high-dimensional thin domains, Studia Math. 154 (2003), no. 3, 253–275.
- [22] G. Raugel, Dynamics of partial differential equations on thin domains, Dynamical systems (Montecatini Terme, 1994), Lecture Notes in Math., vol. 1609, Springer, Berlin, 1995, pp. 208–315.
- [23] G. Raugel and G. R. Sell, Navier-Stokes equations on thin 3D domains. I. Global attractors and global regularity of solutions, J. Amer. Math. Soc. 6 (1993), no. 3, 503–568.
- [24] J. W. Strutt, Some general theorems relating to vibrations, Proceedings of the London Mathematical Society s1-4 (1871), no. 1, 357–368.
- [25] H. Sun and C. Liu, On energetic variational approaches in modeling the nematic liquid crystal flows, Discrete Contin. Dyn. Syst. 23 (2009), no. 1-2, 455–475.
- [26] R. Temam and M. Ziane, Navier-Stokes equations in three-dimensional thin domains with various boundary conditions, Adv. Differential Equations 1 (1996), no. 4, 499–546.
- [27] R. Temam and M. Ziane, Navier-Stokes equations in thin spherical domains, Optimization methods in partial differential equations (South Hadley, MA, 1996), Contemp. Math., vol. 209, Amer. Math. Soc., Providence, RI, 1997, pp. 281–314.

Chapter 4

On singular limit equations for incompressible fluids in moving thin domains

4.1 Introduction

Fluid flows in a thin domain appear in many problems of natural sciences, e.g. ocean dynamics, geophysical fluid dynamics, and fluid flows in cell membranes. In the study of the incompressible Navier–Stokes equations in a three-dimensional thin domain mathematical researchers are mainly interested in global existence of a strong solution for large data since a three-dimensional thin domain with sufficiently small width can be considered "almost two-dimensional." It is also important to investigate the behavior of a solution as the width of a thin domain goes to zero. We may naturally ask whether we can derive limit equations as a thin domain degenerates into a two-dimensional set and compare properties of solutions to the original three-dimensional equations and the corresponding two-dimensional limit equations. There are several works studying such problems with a three-dimensional flat thin domain [15, 16, 29, 33] of the form

$$\Omega_{\varepsilon} = \{ x = (x', x_3) \in \mathbb{R}^3 \mid x' \in \omega, \ \varepsilon g_0(x') < x_3 < \varepsilon g_1(x') \}$$

for small $\varepsilon > 0$, where ω is a two-dimensional domain and g_0 and g_1 are functions on ω , and a three-dimensional thin spherical domain [34] which is a region between two concentric spheres of near radii. (We also refer to [28] for the strategy of analysis of the Euler equations in a flat and spherical thin domain and its limit equations.) However, mathematical studies of an incompressible fluid in a thin domain have not been done in the case where a thin domain and its limit set have more complicated geometric structures. (See [27] for the mathematical analysis of a reaction-diffusion equation in a thin domain degenerating into a lower dimensional manifold.)

In this chapter we are concerned with the incompressible Euler and Navier–Stokes equations in a three-dimensional thin domain that moves in time. The purpose of this chapter is to give a heuristic derivation of singular limits of these equations as a moving thin domain degenerates into a two-dimensional moving closed surface. We also investigate relations between the energy structures of the incompressible fluid systems in a moving thin domain and the corresponding limit systems on a moving closed surface.

Here let us explain our results on limit equations and strategy to derive them. Let $\Gamma(t)$ be an evolving closed surface in \mathbb{R}^3 and $V_{\Gamma}^N(\cdot, t)$ and $\nu(\cdot, t)$ its (scalar) outward normal velocity and unit outward normal vector field, respectively. We assume that $\Gamma(t)$ does not change its topology. Also, let $\Omega_{\varepsilon}(t)$ be a tubular neighborhood of $\Gamma(t)$ of radius ε in \mathbb{R}^3 with sufficiently small $\varepsilon > 0$. We consider the Euler equations

$$\partial_t u + (u \cdot \nabla)u + \nabla p = 0 \quad \text{in} \quad \Omega_{\varepsilon}(t), \, t \in (0, T), \tag{4.1.1}$$

div
$$u = 0$$
 in $\Omega_{\varepsilon}(t), t \in (0, T),$ (4.1.2)

$$u \cdot \nu_{\varepsilon} = V_{\varepsilon}^{N}$$
 on $\partial \Omega_{\varepsilon}(t), t \in (0, T)$ (4.1.3)

and the Navier–Stokes equations with (perfect slip) Navier boundary condition

$$\partial_t u + (u \cdot \nabla)u + \nabla p = \mu_0 \Delta u \quad \text{in} \quad \Omega_\varepsilon(t), \ t \in (0, T), \tag{4.1.4}$$

div
$$u = 0$$
 in $\Omega_{\varepsilon}(t), t \in (0, T),$ (4.1.5)

$$u \cdot \nu_{\varepsilon} = V_{\varepsilon}^{N}$$
 on $\partial \Omega_{\varepsilon}(t), t \in (0, T),$ (4.1.6)

$$[D(u)\nu_{\varepsilon}]_{tan} = 0 \qquad \text{on} \quad \partial\Omega_{\varepsilon}(t), t \in (0,T).$$
(4.1.7)

Here ν_{ε} and V_{ε}^{N} denote the unit outward normal vector field and the (scaler) outward normal velocity of $\partial \Omega_{\varepsilon}(t)$. Also, $\mu_{0} > 0$ is the viscosity coefficient and $D(u) := \{\nabla u + (\nabla u)^{T}\}/2$ is the strain rate tensor with $(\nabla u)^{T}$ the transpose of the gradient matrix ∇u . We suppose that $\Omega_{\varepsilon}(t)$ admits the normal coordinate system $x = \pi(x,t) + d(x,t)\nu(\pi(x,t),t)$ for $x \in \Omega_{\varepsilon}(t)$, where $\pi(\cdot,t)$ is the closest point mapping onto $\Gamma(t)$ and $d(\cdot,t)$ is the signed distance from $\Gamma(t)$ increasing in the direction of $\nu(\cdot,t)$. Based on the normal coordinates, we expand the velocity field u(x,t) on $\Omega_{\varepsilon}(t)$ in powers of the signed distance d(x,t) as

$$u(x,t) = v(\pi(x,t),t) + d(x,t)v^{1}(\pi(x,t),t) + \cdots, \quad x \in \Omega_{\varepsilon}(t)$$
(4.1.8)

and the pressure p(x,t) similarly. We substitute them for the equations in $\Omega_{\varepsilon}(t)$ and determine equations on $\Gamma(t)$ that the zeroth order term v in (4.1.8) satisfies. Then we obtain limit equations of the Euler equations (4.1.1)–(4.1.3):

$$\partial_v^{\bullet} v + \nabla_{\Gamma} q + q^1 \nu = 0 \qquad \text{on} \quad \Gamma(t), \, t \in (0, T), \tag{4.1.9}$$

$$\operatorname{div}_{\Gamma} v = 0 \qquad \text{on} \quad \Gamma(t), \ t \in (0, T), \tag{4.1.10}$$

$$v \cdot \nu = V_{\Gamma}^N$$
 on $\Gamma(t), t \in (0, T).$ (4.1.11)

Here $\partial_v^{\bullet} = \partial_t + v \cdot \nabla$ is the material derivative along the velocity field v and ∇_{Γ} and div_{\Gamma} denote the tangential gradient and the surface divergence on $\Gamma(t)$, respectively (see Section 4.2 for their definitions). Similarly, we get limit equations of the Navier–Stokes equations (4.1.4)–(4.1.7):

$$\partial_v^{\bullet} v + \nabla_{\Gamma} q + q^1 \nu = 2\mu_0 \operatorname{div}_{\Gamma}(P_{\Gamma} D^{tan}(v) P_{\Gamma}) \quad \text{on} \quad \Gamma(t), \, t \in (0, T),$$
(4.1.12)

$$\operatorname{div}_{\Gamma} v = 0$$
 on $\Gamma(t), t \in (0, T),$ (4.1.13)

$$v \cdot \nu = V_{\Gamma}^N$$
 on $\Gamma(t), t \in (0, T).$ (4.1.14)

Here $D^{tan}(v) := \{\nabla_{\Gamma} v + (\nabla_{\Gamma} v)^T\}/2$ and P_{Γ} is the orthogonal projection onto the tangent plane of $\Gamma(t)$. Note that if we take the average of (4.1.8) in the normal direction of $\Gamma(t)$ then

$$\frac{1}{2\varepsilon}\int_{-\varepsilon}^{\varepsilon}u(y+\rho\nu(y,t),t)\,d\rho=v(y,t)+(\text{higher order terms in }\varepsilon),\quad y\in\Gamma(t).$$

Therefore, formally speaking, our limit equations are equations satisfied by the limit of the average in the thin direction of a solution to the original Euler or Navier–Stokes equations in

 $\Omega_{\varepsilon}(t)$ as ε goes to zero. (The above method is also applied in [23] to derive a limit equation of a nonlinear diffusion equation in a moving thin domain.)

In the equations (4.1.9) and (4.1.12) the scalar function q^1 , which comes from the normal derivative of the bulk pressure p (see the expansion (4.3.5) of p and (4.3.17) in the proof of Theorem 4.3.1), is determined by the normal component of (4.1.9) and (4.1.12). Therefore, the limit Euler system (4.1.9)–(4.1.11) is intrinsically equivalent to

$$P_{\Gamma}\partial_{v}^{\bullet}v + \nabla_{\Gamma}q = 0, \quad \operatorname{div}_{\Gamma}v = 0, \quad v \cdot \nu = V_{\Gamma}^{N}$$

$$(4.1.15)$$

and the limit Navier–Stokes system (4.1.12)-(4.1.14) is equivalent to

$$P_{\Gamma}\partial_{v}^{\bullet}v + \nabla_{\Gamma}q = 2\mu_{0}P_{\Gamma}\operatorname{div}_{\Gamma}(P_{\Gamma}D^{tan}(v)P_{\Gamma}), \quad \operatorname{div}_{\Gamma}v = 0, \quad v \cdot \nu = V_{\Gamma}^{N}.$$
(4.1.16)

We note that these tangential surface fluid systems were also derived in [17, 18] recently. The derivation of the Navier–Stokes equations on a moving surface in [17] is based on local conservation laws of mass and linear momentum for a surface fluid. On the other hand, the authors of [18] applied a global energetic variational approach to derive several kinds of equations for an incompressible fluid on an evolving surface.

The viscous term $2\mu_0 \operatorname{div}_{\Gamma}(P_{\Gamma}D^{tan}(v)P_{\Gamma})$ in the momentum equation (4.1.12) of the limit Navier–Stokes system appears in the Boussinesq–Scriven surface fluid model which was first described by Boussinesq [7] and generalized by Scriven [30] to an arbitrary curved moving surface (see also [1, Chapter 10] for derivation of the Boussinesq–Scriven surface fluid model). In [4] the Boussinesq–Scriven surface fluid model was considered to formulate a continuum model for fluid membranes in a bulk fluid, which contains equations for a viscous fluid on a curved moving surface, and study the effect of membrane viscosity in the dynamics of fluid membranes. It was also studied in the context of two-phase flows [5,6,25] in which equations for a surface fluid are considered as the boundary condition on a fluid interface.

Since we consider an incompressible fluid on a moving surface or in its tubular neighborhood, some constraints on the motion of the surface are necessary. For the existence of a surface incompressible fluid it is required that the area of the moving surface is preserved in time. To consider a bulk incompressible fluid in the ε -tubular neighborhood of the moving surface for all $\varepsilon > 0$ sufficiently small, we need another constraint on the moving surface besides the area preserving condition. However, it is automatically satisfied by the Gauss–Bonnet theorem and the assumption that the moving surface does not change its topology. See Remark 4.3.3 for details.

When the surface does not move in time, our tangential limit system (4.1.15) of the Euler equations is the same as the Euler system on a fixed manifold derived by Arnol'd [2,3], who applied the Lie group of diffeomorphisms of a manifold (see also Ebin and Marsden [12]). Also, for a stationary surface our tangential limit system (4.1.16) of the Navier–Stokes equations is the same as the Navier–Stokes system on a manifold derived by Taylor [31], although the authors of [18] claim that (4.1.16) is different from Taylor's system (see Remark 4.4.3). For detailed comparison of our limit systems and the systems derived in previous works see Remarks 4.3.2 and 4.4.2. We further note that the function q^1 in the limit momentum equations (4.1.9) and (4.1.12), which is determined by the normal component of these equations, does not vanish even if the surface is stationary. See Remarks 4.3.2 and 4.4.2 for details.

Finally we note that our results are based on formal calculations and thus mathematical justification is required. There are a few works that present rigorous derivation of limit equations in the case where a limit set is a hypersurface or a manifold. Temam and Ziane [34] derived limit equations for the Navier–Stokes equations in a thin spherical domain by characterizing the thin width limit of a solution to the original equations as a solution to the limit equations. In [27], Prizzi, Rinaldi, and Rybakowski compared the dynamics of a reaction-diffusion equation in a thin domain and that of a limit equation when a thin domain degenerates into a lower dimensional manifold. Recently, the present author derived a limit equation of the heat equation in a moving thin domain shrinking to a moving closed hypersurface by characterization of the thin width limit of a solution [22]. Although there are several tools and methods introduced in the above papers, it seems that mathematical justification of our results is difficult because of the nonlinearity of the equations and the evolution of the shape of the limit surface, and that we need some new techniques.

This chapter is organized as follows. In Section 4.2 we give notations and formulas on quantities related to a moving surface and a moving thin domain. In Sections 4.3 and 4.4 we derive the limit equations of the Euler and Navier–Stokes equations in a moving thin domain, respectively. In Section 4.5 we derive the energy identities of the Euler and Navier–Stokes equations and the corresponding limit equations and investigate relations between them. In Appendices 4.A and 4.B we give proofs of lemmas in Section 4.2 involving the differential geometry of a surface embedded in the Euclidean space.

4.2 Preliminaries

We fix notations on various quantities of a moving surface and give formulas on them. All functions appearing in this section are assumed to be sufficiently smooth.

Lemmas in this section are proved by straightforward calculations. To avoid making this section too long we give proofs of them in Appendix 4.A, except for the proofs of Lemmas 4.2.4 and 4.2.5. Also, a proof of the formula (4.2.15) in Lemma 4.2.4 is given in Appendix 4.B. Although we are concerned with a two-dimensional surface in this chapter, all notations and formulas in this section apply to hypersurfaces of any dimension with easy modifications.

4.2.1 Moving surfaces and moving thin domains

Let $\Gamma(t)$, $t \in [0,T]$ be a two-dimensional closed (i.e. compact and without boundary), connected, and oriented moving surface in \mathbb{R}^3 . The unit outward normal vector and the (scalar) outward normal velocity of $\Gamma(t)$ are denoted by $\nu(\cdot, t)$ and $V_{\Gamma}^N(\cdot, t)$, respectively. Also, let $S_T := \bigcup_{t \in (0,T)} \Gamma(t) \times \{t\}$ be a space-time hypersurface associated with $\Gamma(t)$. We assume that $\Gamma(t)$ is smooth at each $t \in [0,T]$ and moves smoothly in time. In particular, $\Gamma(t)$ does not change its topology. By the smoothness assumption on $\Gamma(t)$, the (outward) principal curvatures $\kappa_1(\cdot, t)$ and $\kappa_2(\cdot, t)$ of $\Gamma(t)$ are bounded uniformly with respect to t. Hence there is a tubular neighborhood

$$N(t) := \{ x \in \mathbb{R}^3 \mid \operatorname{dist}(x, \Gamma(t)) < \delta \}$$

of radius $\delta > 0$ independent of t that admits the normal coordinate system

$$x = \pi(x,t) + d(x,t)\nu(\pi(x,t),t), \quad x \in N(t),$$
(4.2.1)

where $\pi(\cdot, t)$ is the closest point mapping onto $\Gamma(t)$ and $d(\cdot, t)$ is the signed distance function from $\Gamma(t)$ (see e.g. [11, Lemma 2.8]). Moreover, the mapping π and the signed distance d are smooth in the closure (in \mathbb{R}^4) of a space-time noncylindrical domain $N_T := \bigcup_{t \in (0,T)} N(t) \times$ $\{t\}$. We assume that $d(\cdot, t)$ increases in the direction of $\nu(\cdot, t)$. Therefore,

$$\nabla d(x,t) = \nu(\pi(x,t),t), \quad (x,t) \in N_T, \tag{4.2.2}$$

$$\partial_t d(y,t) = -V_{\Gamma}^N(y,t), \quad (y,t) \in S_T.$$

$$(4.2.3)$$

Moreover, differentiating both sides of

$$d(x,t) = \{x - \pi(x,t)\} \cdot \nabla d(x,t), \quad d(\pi(x,t),t) = 0$$

with respect to t and using (4.2.2) and (4.2.3) we easily get

$$\partial_t d(x,t) = \partial_t d(\pi(x,t),t) = -V_{\Gamma}^N(\pi(x,t),t), \quad (x,t) \in N_T.$$
 (4.2.4)

For a sufficiently small $\varepsilon > 0$ we define a moving thin domain $\Omega_{\varepsilon}(t)$ in \mathbb{R}^3 as

$$\Omega_{\varepsilon}(t) := \{ x \in \mathbb{R}^3 \mid \operatorname{dist}(x, \Gamma(t)) < \varepsilon \}$$

and a space-time noncylindrical domain $Q_{\varepsilon,T}$ and its lateral boundary $\partial_{\ell}Q_{\varepsilon,T}$ as

$$Q_{\varepsilon,T}:=\bigcup_{t\in(0,T)}\Omega_{\varepsilon}(t)\times\{t\},\quad \partial_{\ell}Q_{\varepsilon,T}:=\bigcup_{t\in(0,T)}\partial\Omega_{\varepsilon}(t)\times\{t\}$$

Since $\Omega_{\varepsilon}(t)$ is a tubular neighborhood of $\Gamma(t)$, the unit outward normal vector $\nu_{\varepsilon}(\cdot, t)$ and the outward normal velocity $V_{\varepsilon}^{N}(\cdot, t)$ of its boundary are given by

$$\nu_{\varepsilon}(x,t) = \begin{cases} \nu(\pi(x,t),t) & \text{if } d(x,t) = \varepsilon, \\ -\nu(\pi(x,t),t) & \text{if } d(x,t) = -\varepsilon, \end{cases}$$
(4.2.5)

$$V_{\varepsilon}^{N}(x,t) = \begin{cases} V_{\Gamma}^{N}(\pi(x,t),t) & \text{if } d(x,t) = \varepsilon, \\ -V_{\Gamma}^{N}(\pi(x,t),t) & \text{if } d(x,t) = -\varepsilon. \end{cases}$$
(4.2.6)

4.2.2 Notations and formulas for quantities on fixed surfaces

In this subsection we fix and suppress the time $t \in [0, T]$. Hence Γ denotes a two-dimensional closed, connected, oriented and smooth surface in \mathbb{R}^3 . Let us give notations and formulas for several quantities on the fixed surface Γ . (In the sequel we use the same notations given in this subsection for the moving surface $\Gamma(t)$.) Let P_{Γ} be the orthogonal projection onto the tangent plane of Γ at each point on Γ given by

$$P_{\Gamma}(y) := I_3 - \nu(y) \otimes \nu(y), \quad y \in \Gamma,$$

where I_3 is the identity matrix of three dimension and $a \otimes b$ for $a, b \in \mathbb{R}^3$ denotes the tensor product of a and b given by

$$a \otimes b := \begin{pmatrix} a_1b_1 & a_1b_2 & a_1b_3 \\ a_2b_1 & a_2b_2 & a_2b_3 \\ a_3b_1 & a_3b_2 & a_3b_3 \end{pmatrix}, \quad a = (a_1, a_2, a_3), \ b = (b_1, b_2, b_3).$$

For a function f on Γ we define its tangential gradient $\nabla_{\Gamma} f$ as

$$\nabla_{\Gamma} f(y) := P_{\Gamma}(y) \nabla f(y), \quad y \in \Gamma.$$

Here \tilde{f} is an extension of f to N satisfying $\tilde{f}|_{\Gamma} = f$. Note that the tangential gradient of f is independent of the choice of its extension (see e.g. [11, Lemma 2.4]). Also, it is easy to see that $\nabla_{\Gamma} f \cdot \nu = 0$ and $P_{\Gamma} \nabla_{\Gamma} f = \nabla_{\Gamma} f$ hold on Γ . The tangential derivative operators are given by

$$\partial_i^{tan} f(y) := \sum_{j=1}^3 \{ \delta_{ij} - \nu_i(y) \nu_j(y) \} \partial_j \tilde{f}(y), \quad i = 1, 2, 3$$

so that $\nabla_{\Gamma} = (\partial_1^{tan}, \partial_2^{tan}, \partial_3^{tan})$, which are again independent of the choice of an extension \tilde{f} of f. For example, we may take the constant extension in the normal direction of Γ given by $\bar{f}(x) := f(\pi(x))$ for $x \in N$.

For vector fields $F = (F_1, F_2, F_3)$ on N and $G = (G_1, G_2, G_3)$ on Γ , we define the gradient matrix and the divergence of F as

$$\nabla F := \begin{pmatrix} \partial_1 F_1 & \partial_1 F_2 & \partial_1 F_3 \\ \partial_2 F_1 & \partial_2 F_2 & \partial_2 F_3 \\ \partial_3 F_1 & \partial_3 F_2 & \partial_3 F_3 \end{pmatrix}, \quad \operatorname{div} F := \sum_{i=1}^3 \partial_i F_i$$

and the tangential gradient matrix and the surface divergence of G as

$$\nabla_{\Gamma}G := \begin{pmatrix} \partial_1^{tan}G_1 & \partial_1^{tan}G_2 & \partial_1^{tan}G_3\\ \partial_2^{tan}G_1 & \partial_2^{tan}G_2 & \partial_2^{tan}G_3\\ \partial_3^{tan}G_1 & \partial_3^{tan}G_2 & \partial_3^{tan}G_3 \end{pmatrix}, \quad \operatorname{div}_{\Gamma}G := \sum_{i=1}^3 \partial_i^{tan}G_i.$$

These notations are consistent with the formula $\nabla_{\Gamma}G = P_{\Gamma}\nabla \widetilde{G}$ on Γ , where \widetilde{G} is an arbitrary extension of G to N with $\widetilde{G}|_{\Gamma} = G$. For a function f on Γ we denote by $\nabla_{\Gamma}^2 f$ the tangential Hessian matrix of f whose (i, j)-entry is given by $\partial_i^{tan} \partial_j^{tan} f$ (i, j = 1, 2, 3). Let M be a 3×3 matrix-valued function defined on N or on Γ of the form

$$M = (M_{ij})_{i,j} = \begin{pmatrix} M_{11} & M_{12} & M_{13} \\ M_{21} & M_{22} & M_{23} \\ M_{31} & M_{32} & M_{33} \end{pmatrix}.$$

We define the divergence div M on N or the surface divergence div_{Γ}M on Γ as a vector field whose *j*-th component is given by

$$[\operatorname{div} M]_j := \sum_{i=1}^3 \partial_i M_{ij} \quad \text{or} \quad [\operatorname{div}_{\Gamma} M]_j := \sum_{i=1}^3 \partial_i^{tan} M_{ij}, \quad j = 1, 2, 3.$$

Finally we set

$$A := -\nabla_{\Gamma}\nu = (-\partial_i^{tan}\nu_j)_{i,j}, \quad \Delta_{\Gamma} := \operatorname{div}_{\Gamma}\nabla_{\Gamma} = \sum_{i=1}^3 (\partial_i^{tan})^2,$$
$$H := -\operatorname{div}_{\Gamma}\nu = \operatorname{tr}[A], \quad K := \kappa_1\kappa_2$$

and call them the Weingarten map of Γ , the Laplace–Beltrami operator on Γ , (twice) the mean curvature of Γ , and the Gaussian curvature of Γ , respectively. The usual Laplacian Δ and the Laplace–Beltrami operator Δ_{Γ} acting on vector fields are understood to be componentwise operators.

Lemma 4.2.1. For all $y \in \Gamma$ we have

$$A(y)\nu(y) = 0, (4.2.7)$$

$$A(y)P_{\Gamma}(y) = P_{\Gamma}(y)A(y) = A(y),$$
 (4.2.8)

$$A(y) = -\nabla^2 d(y).$$
 (4.2.9)

By (4.2.7) we see that A has the eigenvalue 0. Note that the other eigenvalues of A are κ_1 and κ_2 (see e.g. [19, Section VII.5]) and thus

$$H(y) = \kappa_1(y) + \kappa_2(y), \quad y \in \Gamma.$$

$$(4.2.10)$$

Also, A is symmetric (i.e. $\partial_i^{tan}\nu_j = \partial_j^{tan}\nu_i$) and $H = -\Delta d$ holds on Γ by (4.2.9). The tangential derivatives ∂_i^{tan} (i = 1, 2, 3) are noncommutative in general. An exchange formula for them includes the unit outward normal of the surface.

Lemma 4.2.2. Let f be a function on Γ . For each i, j = 1, 2, 3 we have

$$\partial_i^{tan} \partial_j^{tan} f - \partial_j^{tan} \partial_i^{tan} f = [A \nabla_{\Gamma} f]_i \nu_j - [A \nabla_{\Gamma} f]_j \nu_i.$$
(4.2.11)

Here $[A\nabla_{\Gamma} f]_i$ denotes the *i*-th component of the vector field $A\nabla_{\Gamma} f$.

The next formula is a consequence of (4.2.11), which we use in Section 4.4 to express a viscous term of limit equations of the Navier–Stokes equations in terms of the Laplace– Beltrami operator. For a vector field v on Γ we set

$$D^{tan}(v) := \frac{\nabla_{\Gamma} v + (\nabla_{\Gamma} v)^T}{2}.$$
(4.2.12)

The matrices $D^{tan}(v)$ and $P_{\Gamma}D^{tan}(v)P_{\Gamma}$ are called a tangential strain rate and a projected strain rate in [18], respectively.

Lemma 4.2.3. Let v be a (not necessarily tangential) vector field on Γ . Then

$$2\operatorname{div}_{\Gamma}(P_{\Gamma}D^{tan}(v)P_{\Gamma}) = 2\operatorname{tr}[A\nabla_{\Gamma}v]\nu + P_{\Gamma}(\Delta_{\Gamma}v) + \nabla_{\Gamma}(\operatorname{div}_{\Gamma}v) + H(\nabla_{\Gamma}v)\nu \qquad (4.2.13)$$

holds on Γ (note that $(\nabla_{\Gamma} v)\nu = P_{\Gamma}(\nabla_{\Gamma} v)\nu$ on the right-hand side is tangential).

To compare our limit systems with the incompressible fluid systems on a fixed manifold derived by Arnol'd [2,3] and Taylor [31] we need formulas on the Levi-Civita connection. Let $\overline{\nabla}$ be the Levi-Civita connection on Γ with respect to the metric on Γ induced by the Euclidean metric of \mathbb{R}^3 (see e.g. [9, Section 2.3] and [24, Sections 3.3.1 and 4.1.2] for the definition of the Levi-Civita connection). Hence for tangential vector fields X and Y on Γ the covariant derivative of X along Y is denoted by $\overline{\nabla}_Y X$, which is again a tangential vector field on Γ . The Levi-Civita connection is considered as a mapping

$$\overline{\nabla} \colon C^{\infty}(T\Gamma) \to C^{\infty}(T^*\Gamma \otimes T\Gamma), \quad X \mapsto \overline{\nabla}X,$$

where $T\Gamma$ and $T^*\Gamma$ are the tangent and cotangent bundle of Γ , respectively, and for a vector bundle E over Γ we denote by $C^{\infty}(E)$ the set of all smooth sections of E. (Hence $C^{\infty}(T\Gamma)$) denotes the set of all smooth tangential vector fields on Γ . We refer to [20, Chapter 10] for the definitions of a vector bundle and a section.) Also, for a tangential vector field X on Γ the notation $\overline{\nabla}X$ stands for a mapping $Y \mapsto \overline{\nabla}_Y X$ from $C^{\infty}(T\Gamma)$ into itself. Then we write $\overline{\nabla}^*: C^{\infty}(T^*\Gamma \otimes T\Gamma) \to C^{\infty}(T\Gamma)$ for the formal adjoint operator of $\overline{\nabla}$ (see [24, Section 10.1.3]) and set $\Delta_B := -\overline{\nabla}^* \overline{\nabla}$. The operator $\Delta_B : C^{\infty}(T\Gamma) \to C^{\infty}(T\Gamma)$ is called the Bochner Laplacian (note that there is another definition of the Bochner Laplacian where the sign is taken opposite).

Lemma 4.2.4. Let X and Y are tangential vector fields on Γ . Then

$$(Y \cdot \nabla)\widetilde{X} = \overline{\nabla}_Y X + (AX \cdot Y)\nu, \qquad (4.2.14)$$

$$\Delta_B X = P_{\Gamma}(\Delta_{\Gamma} X) + A^2 X \tag{4.2.15}$$

hold on Γ . Here \widetilde{X} is an extension of X to N with $\widetilde{X}|_{\Gamma} = X$ and $(Y \cdot \nabla)\widetilde{X}$ denotes the directional derivative of \widetilde{X} along Y in \mathbb{R}^3 , i.e.

$$(Y \cdot \nabla)\widetilde{X} = \left(\sum_{i=1}^{3} Y_i \partial_i \widetilde{X}_1, \sum_{i=1}^{3} Y_i \partial_i \widetilde{X}_2, \sum_{i=1}^{3} Y_i \partial_i \widetilde{X}_3\right).$$

Also, the left-hand side of (4.2.14) is independent of the choice of the extension X.

The formula (4.2.14) is well-known as the Gauss formula (see e.g. [9, Section 4.2] and [19, Section VII.3]) and we omit its proof. Note that $(Y \cdot \nabla)\widetilde{X} = (Y \cdot \nabla_{\Gamma})X$ on Γ since Y is tangential. Hence the Gauss formula (4.2.14) is also expressed as

$$(Y \cdot \nabla_{\Gamma})X = \overline{\nabla}_Y X + (AX \cdot Y)\nu \quad \text{on} \quad \Gamma$$
(4.2.16)

for tangential vector fields X and Y on Γ . We also call (4.2.16) the Gauss formula.

A proof of the formula (4.2.15) is given in Appendix 4.B. Note that (4.2.15) is useful by itself since it gives a global expression under the fixed Cartesian coordinate system of the Bochner Laplacian acting on tangential vector fields on Γ , which is originally defined intrinsically and represented under only local coordinate systems.

Combining Lemmas 4.2.3 and 4.2.4 we get the following formula on the surface divergence of the projected strain rate, which is crucial for comparison of our limit Navier–Stokes system and the incompressible viscous fluid system on a manifold derived by Taylor [31] (see Remark 4.4.2).

Lemma 4.2.5. For a tangential vector field v on Γ satisfying div_{Γ}v = 0 we have

$$2P_{\Gamma} \operatorname{div}_{\Gamma}(P_{\Gamma} D^{tan}(v) P_{\Gamma}) = \Delta_B v + K v \quad on \quad \Gamma.$$

$$(4.2.17)$$

Proof. Let v be a tangential vector field on Γ satisfying $\operatorname{div}_{\Gamma} v = 0$. Then

$$(\nabla_{\Gamma} v)\nu = \nabla_{\Gamma} (v \cdot \nu) - (\nabla_{\Gamma} \nu)v = Av$$

by $v \cdot \nu = 0$ and $-\nabla_{\Gamma} \nu = A$. Applying this and

$$P_{\Gamma}(\operatorname{tr}[A\nabla_{\Gamma}v]\nu) = \operatorname{tr}[A\nabla_{\Gamma}v]P_{\Gamma}\nu = 0, \quad \operatorname{div}_{\Gamma}v = 0$$

to the formula (4.2.13), and observing that $(\nabla_{\Gamma} v)\nu = Av$ is tangential, we have

$$2P_{\Gamma} \operatorname{div}_{\Gamma}(P_{\Gamma} D^{tan}(v) P_{\Gamma}) = P_{\Gamma}(\Delta_{\Gamma} v) + HAv.$$
(4.2.18)

Moreover, since A is symmetric and has the eigenvalues 0, κ_1 , and κ_2 , where the eigenvector corresponding to the eigenvalue 0 is ν (see Lemma 4.2.1), for each $y \in \Gamma$ we can take an orthonormal basis $\{e_1, e_2\}$ of the tangent plane of Γ at y such that $Ae_i = \kappa_i e_i$, i = 1, 2. (The vectors e_1 and e_2 are called the principal directions at y. See e.g. [19, Section VII.5] for details.) Expressing the tangential vector v as a linear combination of e_1 and e_2 and using $H = \kappa_1 + \kappa_2$ and $K = \kappa_1 \kappa_2$ we easily obtain $HAv = Kv + A^2v$. Applying this and (4.2.15) to (4.2.18) we obtain (4.2.17).

Besides derivation of limit equations, we are also interested in thin width limits of energy identities for the Euler and Navier–Stokes equations. To derive limit energy identities we give change of variables formulas for integrals over level-set surfaces and tubular neighborhoods of Γ . For $y \in \Gamma$ and $\rho \in [-\varepsilon, \varepsilon]$ we set

$$J(y,\rho) := \{1 - \rho\kappa_1(y)\}\{1 - \rho\kappa_2(y)\} = 1 - \rho H(y) + \rho^2 K(y).$$
(4.2.19)

Here the second equality follows from the definition of the Gaussian curvature and (4.2.10). The function J is the Jacobian appearing in the following change of variables formulas (see [13, Section 14.6] or Appendix 4.A). **Lemma 4.2.6.** For a function f on $\overline{\Omega_{\varepsilon}}$ we have

$$\int_{\Omega_{\varepsilon}} f(x) \, dx = \int_{\Gamma} \int_{-\varepsilon}^{\varepsilon} f(y + \rho\nu(y)) J(y,\rho) \, d\rho \, d\mathcal{H}^2(y) \tag{4.2.20}$$

and

$$\int_{\partial\Omega_{\varepsilon}} f(x) \, d\mathcal{H}^2(x) = \int_{\Gamma} f(y + \varepsilon\nu(y)) J(y,\varepsilon) \, d\mathcal{H}^2(y) + \int_{\Gamma} f(y - \varepsilon\nu(y)) J(y,-\varepsilon) \, d\mathcal{H}^2(y). \quad (4.2.21)$$

Here \mathcal{H}^2 denotes the two-dimensional Hausdorff measure.

When we use Lemma 4.2.6 with the moving surface $\Gamma(t)$ we write $J(y, t, \rho)$ for the Jacobian given by (4.2.19).

4.2.3 Material derivatives and differentiation of composite functions with the closest point mapping

Now let us return to the moving surface $\Gamma(t)$. We first give a material time derivative of a function on S_T . Let v be a vector field on S_T with $v \cdot \nu = V_{\Gamma}^N$. Suppose that there exists the flow map Φ_v of v, i.e. $\Phi_v(\cdot, t) \colon \Gamma(0) \to \mathbb{R}^3$ is a diffeomorphism onto its range for each $t \in [0, T]$ and

$$\Phi_v(Y,0) = Y, \quad \frac{d\Phi_v}{dt}(Y,t) = v(\Phi_v(Y,t),t) \quad \text{for} \quad (Y,t) \in \Gamma(0) \times (0,T).$$

Note that $\Phi_v(\cdot, t)$ is a diffeomorphism from $\Gamma(0)$ onto $\Phi_v(\Gamma(0), t) = \Gamma(t)$ for each $t \in [0, T]$ since the normal component of v is equal to the outward normal velocity V_{Γ}^N of the moving surface $\Gamma(t)$, which completely determines the change of the shape of $\Gamma(t)$. We define the material derivative of a function f on S_T along the velocity field v as

$$\partial_v^{\bullet} f(\Phi_v(Y,t),t) := \frac{d}{dt} \big(f(\Phi_v(Y,t),t) \big), \quad (Y,t) \in \Gamma(0) \times (0,T).$$

By the chain rule of differentiation it is also represented as

$$\partial_v^{\bullet} f(y,t) = \partial_t \tilde{f}(y,t) + v(y,t) \cdot \nabla \tilde{f}(y,t), \quad (y,t) \in S_T,$$
(4.2.22)

where \tilde{f} is an arbitrary extension of f to N_T satisfying $\tilde{f}|_{S_T} = f$. We write ∂° for ∂_v^\bullet with $v = V_{\Gamma}^N \nu$ and call it the normal time derivative. Note that the normal time derivative of a function f on S_T is equal to the time derivative of its constant extension \bar{f} in the normal direction, i.e.

$$\partial^{\circ} f(y,t) = \partial_t \bar{f}(y,t) = \frac{d}{dt} (f(\pi(y,t),t)), \quad (y,t) \in S_T.$$

Also, for a tangential vector field v^T on S_T the material derivative of f along the velocity field of the form $v = V_{\Gamma}^N \nu + v^T$ is expressed as

$$\partial_v^{\bullet} f = \partial^{\circ} f + v^T \cdot \nabla_{\Gamma} f \quad \text{on} \quad S_T \tag{4.2.23}$$

by (4.2.22) and $v^T \cdot \nabla \tilde{f} = v^T \cdot \nabla_{\Gamma} f$ on S_T since v^T is tangential. See also [8, Section 3] for the time derivative of functions on a moving surface.

In the following sections we frequently differentiate the composition of a function on $\Gamma(t)$ and the closest point mapping $\pi(\cdot, t)$. To avoid repetition of the same calculations we give several formulas on derivatives of composite functions with π .

Let f(x,t) be a function on $Q_{\varepsilon,T}$. Based on the normal coordinate system $x = \pi(x,t) + d(x,t)\nu(\pi(x,t),t)$ for $x \in \Omega_{\varepsilon}(t)$, we expand f(x,t) in powers of the signed distance d(x,t):

$$f(x,t) = g(\pi(x,t),t) + d(x,t)g^{1}(\pi(x,t),t) + \cdots$$

Here g, g^1 , and the coefficients of higher order terms in d(x,t) are considered as functions on S_T . Also, for $k \in \mathbb{N}$ we write $R(d(x,t)^k)$ for the terms of order higher than k-1 with respect to small d(x,t), i.e.

$$f(x,t) = g(\pi(x,t),t) + \dots + d(x,t)^{k-1}g^{k-1}(\pi(x,t),t) + R(d(x,t)^k),$$

$$R(d(x,t)^k) = d(x,t)^k g^k(\pi(x,t),t) + d(x,t)^{k+1}g^{k+1}(\pi(x,t),t) + \dots.$$
(4.2.24)

In the sequel, we also use Landau's symbol $O(\varepsilon^k)$ (as $\varepsilon \to 0$) for a nonnegative integer k, i.e. $O(\varepsilon^k)$ is a quantity satisfying $|O(\varepsilon^k)| \leq C\varepsilon^k$ for small $\varepsilon > 0$ with a constant C > 0 independent of ε . Note that, contrary to $O(\varepsilon^k)$, we may differentiate $R(d(x,t)^k)$ with respect to x and t since it just stands for the higher order terms in the expansion (4.2.24) with respect to small d(x,t), and the l-th order derivative of $R(d(x,t)^k)$ is $R(d(x,t)^{k-l})$ for $l \leq k$. Also, $R(d(x,t)^k) = O(\varepsilon^k)$ for $(x,t) \in Q_{\varepsilon,T}$ and $k \in \mathbb{N}$ by $|d(x,t)| < \varepsilon$ on $Q_{\varepsilon,T}$. We use the same notations on the expansion (4.2.24) for functions on $\Omega_{\varepsilon}(t)$ with each fixed $t \in [0,T]$.

Lemma 4.2.7. Let f be a scalar- or vector-valued function on S_T . The derivatives of the composite function $f(\pi(x,t),t)$ with respect to x and t are of the form

$$\nabla(f(\pi,t)) = \nabla_{\Gamma} f(\pi,t) + d(x,t) [A \nabla_{\Gamma} f](\pi,t) + R(d(x,t)^2), \qquad (4.2.25)$$

$$\partial_t \big(f(\pi, t) \big) = \partial^\circ f(\pi, t) + d(x, t) [(\nabla_\Gamma V_\Gamma^N \cdot \nabla_\Gamma) f](\pi, t) + R(d(x, t)^2)$$
(4.2.26)

for $(x,t) \in Q_{\varepsilon,T}$. Here we abbreviate $\pi(x,t)$ to π .

We also give an expansion formula for the divergence of a matrix-valued function which we need to derive limit equations of the Navier–Stokes equations.

Lemma 4.2.8. Let S and S¹ be 3×3 matrix-valued functions on $\Gamma(t)$ with each fixed $t \in (0,T)$. For $x \in \Omega_{\varepsilon}(t)$ we set

$$D(x) = S(\pi(x,t)) + d(x,t)S^{1}(\pi(x,t)) + R(d(x,t)^{2}).$$

Then we have

$$\operatorname{div} D(x) = \operatorname{div}_{\Gamma} S(\pi(x,t)) + \left(S^{1}(\pi(x,t))\right)^{T} \nu(\pi,t) + R(d(x,t)).$$
(4.2.27)

for $x \in \Omega_{\varepsilon}(t)$. Here $(S^1)^T$ denotes the transpose of the matrix S^1 .

4.3 Limit equations of the Euler equations

We consider the incompressible Euler equations in $\Omega_{\varepsilon}(t)$:

$$\partial_t u + (u \cdot \nabla)u + \nabla p = 0$$
 in $Q_{\varepsilon,T}$, (4.3.1)

$$\operatorname{div} u = 0 \quad \text{in} \quad Q_{\varepsilon,T}, \tag{4.3.2}$$

$$u \cdot \nu_{\varepsilon} = V_{\varepsilon}^{N}$$
 on $\partial_{\ell} Q_{\varepsilon,T}$. (4.3.3)

Here $u = (u_1, u_2, u_3)$ is the velocity of a bulk fluid and p is the pressure. The goal of this section is to derive limit equations of the Euler equations as ε goes to zero. According to the normal coordinate system (4.2.1), we expand u and p with respect to the signed distance d(x,t) as

$$u(x,t) = v(\pi(x,t),t) + d(x,t)v^{1}(\pi(x,t),t) + R(d(x,t)^{2}),$$
(4.3.4)

$$p(x,t) = q(\pi(x,t),t) + d(x,t)q^{1}(\pi(x,t),t) + R(d(x,t)^{2}).$$
(4.3.5)

Here we used the notation (4.2.24). The limit equations are given as the principal term in the expansion with respect to d(x,t) of the Euler equations in $\Omega_{\varepsilon}(t)$.

Theorem 4.3.1. Let u and p satisfy the Euler equations (4.3.1)–(4.3.3) in the moving thin domain $\Omega_{\varepsilon}(t)$. Then the normal component of the zeroth order term v in the expansion (4.3.4) is equal to the outward normal velocity of the moving surface $\Gamma(t)$, i.e. $v \cdot \nu = V_{\Gamma}^{N}$. Moreover, v and the zeroth order term q and the first order term q^{1} in the expansion (4.3.5) satisfy

$$\partial_v^{\bullet} v + \nabla_{\Gamma} q + q^1 \nu = 0 \quad on \quad S_T, \tag{4.3.6}$$

$$\operatorname{div}_{\Gamma} v = 0 \quad on \quad S_T. \tag{4.3.7}$$

Before starting to prove Theorem 4.3.1 we give remarks on the limit equations (4.3.6)–(4.3.7) and necessary conditions on the motion of $\Gamma(t)$ for the existence of incompressible fluids in $\Gamma(t)$ and $\Omega_{\varepsilon}(t)$ for all $\varepsilon > 0$.

Remark 4.3.2. Let us explain how the limit equations (4.3.6) and (4.3.7) determine v, q, and q^1 . As stated in Theorem 4.3.1, the normal component of v is equal to the outward normal velocity of the moving surface. The tangential component of v and the scalar function q are determined by the equations

$$P_{\Gamma}\partial_v^{\bullet}v + \nabla_{\Gamma}q = 0, \quad \text{div}_{\Gamma}v = 0 \quad \text{on} \quad S_T.$$
(4.3.8)

Finally the scalar function q^1 is given just by the inner product of (4.3.6) and ν :

$$q^1 = -\partial_v^{\bullet} v \cdot \nu \quad \text{on} \quad S_T. \tag{4.3.9}$$

Note that q^1 comes from the normal derivative of the pressure p of the bulk fluid in the moving thin domain (see (4.3.17) below).

The system (4.3.8) is the same as the incompressible Euler system (II) in [18] with the constant density. When the surface $\Gamma(t) = \Gamma$ is stationary, the limit velocity v is tangential $(v \cdot \nu = V_{\Gamma}^N = 0)$ and $P_{\Gamma}\{(v \cdot \nabla)v\} = \overline{\nabla}_v v$ holds on Γ by the Gauss formula (4.2.14), where $\overline{\nabla}_v v$ is the covariant derivative. From this and the fact that P_{Γ} is independent of the time it follows that

$$P_{\Gamma}\partial_{v}^{\bullet}v = P_{\Gamma}\partial_{t}v + P_{\Gamma}\{(v\cdot\nabla)v\} = \partial_{t}v + \overline{\nabla}_{v}v \quad \text{on} \quad \Gamma.$$
(4.3.10)

Hence the tangential limit system (4.3.8) becomes

$$\partial_t v + \overline{\nabla}_v v + \nabla_{\Gamma} q = 0, \quad \operatorname{div}_{\Gamma} v = 0 \quad \text{on} \quad \Gamma \times (0, T),$$

which is the same as the Euler system on a manifold derived by Arnol'd [2,3] (see also Ebin and Marsden [12]). Also, applying $v \cdot \nu = 0$, (4.2.14), and the fact that ν is independent of time to (4.3.9) we obtain

$$q^{1} = -\partial_{v}^{\bullet} v \cdot \nu = -\partial_{t} (v \cdot \nu) - \{ (v \cdot \nabla)v \} \cdot \nu = -Av \cdot v, \qquad (4.3.11)$$

which does not vanish in general even if the surface is stationary.

Remark 4.3.3. For the existence of a surface incompressible fluid obeying (4.3.7) it is required that the area of the moving surface $\Gamma(t)$ is preserved in time. Indeed, by the Leibniz formula (see [10, Lemma 2.2]) with a velocity field v on S_T satisfying $v \cdot v = V_{\Gamma}^N$ and (4.3.7) we have

$$\frac{d}{dt}|\Gamma(t)| = \frac{d}{dt}\int_{\Gamma(t)} 1\,d\mathcal{H}^2 = \int_{\Gamma(t)} \operatorname{div}_{\Gamma} v\,d\mathcal{H}^2 = 0, \qquad (4.3.12)$$

where $|\Gamma(t)|$ is the area of $\Gamma(t)$. Similarly, when the moving thin domain $\Omega_{\varepsilon}(t)$ is filled with an incompressible fluid satisfying (4.3.2) and the impermeable boundary condition (4.3.3), its volume $|\Omega_{\varepsilon}(t)|$ must remain constant by the Reynolds transport theorem (see e.g. [14]):

$$\frac{d}{dt}|\Omega_{\varepsilon}(t)| = \frac{d}{dt}\int_{\Omega_{\varepsilon}(t)} 1\,dx = \int_{\partial\Omega_{\varepsilon}(t)} V_{\varepsilon}^{N}\,d\mathcal{H}^{2} = \int_{\partial\Omega_{\varepsilon}(t)} u\cdot\nu_{\varepsilon}\,d\mathcal{H}^{2} = \int_{\Omega_{\varepsilon}(t)} \operatorname{div} u\,dx = 0.$$

By the change of variables formula (4.2.20) the volume of $\Omega_{\varepsilon}(t)$ is expressed as

$$|\Omega_{\varepsilon}(t)| = \int_{\Omega_{\varepsilon}(t)} 1 \, dx = \int_{\Gamma(t)} \int_{-\varepsilon}^{\varepsilon} J(y, t, \rho) \, d\rho \, d\mathcal{H}^2 = 2\varepsilon |\Gamma(t)| + \frac{2}{3}\varepsilon^3 \int_{\Gamma(t)} K \, d\mathcal{H}^2.$$

Hence we need to assume

$$\frac{d}{dt}|\Gamma(t)| = 0, \quad \frac{d}{dt}\int_{\Gamma(t)} K \, d\mathcal{H}^2 = 0$$

for the existence of an incompressible fluid in the ε -tubular neighborhood $\Omega_{\varepsilon}(t)$ of $\Gamma(t)$ for all $\varepsilon > 0$. However, by the Gauss–Bonnet theorem we have

$$\int_{\Gamma(t)} K \, d\mathcal{H}^2 = 2\pi \chi(\Gamma(t)),$$

where $\chi(\Gamma(t))$ is the Euler characteristic of $\Gamma(t)$ (see e.g. [32, Section C.5]). Since the Euler characteristic is a topological invariant and the moving surface $\Gamma(t)$ does not change its topology, the integral of the Gaussian curvature K over $\Gamma(t)$ is constant in time. Therefore, only the area preserving condition (4.3.12) on $\Gamma(t)$ is necessary for the existence of incompressible fluids on $\Gamma(t)$ and in $\Omega_{\varepsilon}(t)$ for all $\varepsilon > 0$. Note that this assertion is valid only for a moving surface in \mathbb{R}^3 or a moving hypersurface in \mathbb{R}^4 . Indeed, when $\Gamma(t)$ is a moving hypersurface in \mathbb{R}^n with n > 4, the Jacobian $J(y, t, \rho)$ is a polynomial in ρ of degree greater than three (see e.g. [13, Section 14.6] and [22, Section 5.1]) and thus we need more constraints on the motion of $\Gamma(t)$.

Proof of Theorem 4.3.1. For the sake of simplicity, we use the abbreviations

$$f(\pi, t) = f(\pi(x, t), t), \quad R(d^k) = R(d(x, t)^k)$$
(4.3.13)

for a function f on S_T and $k \in \mathbb{N}$. Since ν_{ε} and V_{ε}^N are given by (4.2.5) and (4.2.6), the boundary condition (4.3.3) reads

$$u(x,t) \cdot \nu(\pi,t) = V_{\Gamma}^{N}(\pi,t), \quad x \in \partial \Omega_{\varepsilon}(t).$$

We substitute (4.3.4) for u in the above equality. Then

$$v(\pi,t) \cdot \nu(\pi,t) \pm \varepsilon v^1(\pi,t) \cdot \nu(\pi,t) + O(\varepsilon^2) = V_{\Gamma}^N(\pi,t)$$

when $d(x,t) = \pm \varepsilon$ (double-sign corresponds). Since $v(\pi,t)$, $v^1(\pi,t)$, $\nu(\pi,t)$, and $V_{\Gamma}^N(\pi,t)$ are independent of ε , it follows from the above equation that

$$v(\pi, t) \cdot \nu(\pi, t) = V_{\Gamma}^{N}(\pi, t),$$
 (4.3.14)

$$v^{1}(\pi, t) \cdot \nu(\pi, t) = 0.$$
 (4.3.15)

The first statement of the theorem follows from the equality (4.3.14). Let us write $v = V_{\Gamma}^{N}\nu + v^{T}$ with a tangential velocity field v^{T} on $\Gamma(t)$ and derive the equations (4.3.6) and (4.3.7). By (4.2.2) and (4.2.25) we have

$$\nabla u(x,t) = \nabla (v(\pi,t)) + \nabla d(x,t) \otimes v^1(\pi,t) + R(d)$$

= $\nabla_{\Gamma} v(\pi,t) + \nu(\pi,t) \otimes v^1(\pi,t) + R(d)$ (4.3.16)

and

$$\nabla p(x,t) = \nabla_{\Gamma} q(\pi,t) + q^{1}(\pi,t)\nu(\pi,t) + R(d).$$
(4.3.17)

Also, by (4.2.4) and (4.2.26),

$$\partial_t u(x,t) = \partial_t (v(\pi,t)) + \partial_t d(x,t) v^1(\pi,t) + R(d)$$

= $\partial^\circ v(\pi,t) - V_{\Gamma}^N(\pi,t) v^1(\pi,t) + R(d).$ (4.3.18)

From (4.3.16) the gradient of the *j*-th component of *u* is

$$\nabla u_j(x,t) = \nabla_{\Gamma} v_j(\pi,t) + v_j^1(\pi,t)\nu(\pi,t) + R(d).$$

We take the inner product of this equation and (4.3.4), and then apply (4.3.14) and $v \cdot \nabla_{\Gamma} v_j = v^T \cdot \nabla_{\Gamma} v_j$ to get the *j*-th component of the inertia term

$$u(x,t) \cdot \nabla u_j(x,t) = v^T(\pi,t) \cdot \nabla_{\Gamma} v_j(\pi,t) + V^N_{\Gamma}(\pi,t) v_j^1(\pi,t) + R(d).$$

Hence the inertia term $(u \cdot \nabla)u$ is of the form

$$[(u \cdot \nabla)u](x,t) = [(v^T \cdot \nabla_{\Gamma})v](\pi,t) + V_{\Gamma}^N(\pi,t)v^1(\pi,t) + R(d).$$
(4.3.19)

Substituting (4.3.17), (4.3.18), and (4.3.19) for (4.3.1) and applying (4.2.23) we obtain

$$\partial_v^{\bullet} v(\pi, t) + \nabla_{\Gamma} q(\pi, t) + q^1(\pi, t) \nu(\pi, t) = R(d).$$

In this equation, each term on the left-hand side is independent of d. Therefore, the equation (4.3.6) should be satisfied.

Finally, by (4.3.15) and (4.3.16) we have

$$\operatorname{div} u(x,t) = \operatorname{tr}[\nabla u(x,t)] = \operatorname{div}_{\Gamma} v(\pi,t) + \nu(\pi,t) \cdot v^{1}(\pi,t) + R(d) = \operatorname{div}_{\Gamma} v(\pi,t) + R(d)$$

and thus the equation (4.3.2) reads $\operatorname{div}_{\Gamma} v(\pi, t) = R(d)$. Since the left-hand side is independent of d, we conclude that v satisfies the equation (4.3.7).

4.4 Limit equations of the Navier–Stokes equations

In this section, we consider the incompressible Navier–Stokes equations in $\Omega_{\varepsilon}(t)$:

$$\partial_t u + (u \cdot \nabla)u + \nabla p = \mu_0 \Delta u \quad \text{in} \quad Q_{\varepsilon,T}, \tag{4.4.1}$$

$$\operatorname{div} u = 0 \qquad \text{in} \quad Q_{\varepsilon,T}. \tag{4.4.2}$$

Here $u = (u_1, u_2, u_3)$ is the velocity of a bulk fluid, p is the pressure, and $\mu_0 > 0$ is the viscosity coefficient. On these equations we impose the (perfect slip) Navier boundary condition of the form

$$u \cdot \nu_{\varepsilon} = V_{\varepsilon}^{N}$$
 on $\partial_{\ell} Q_{\varepsilon,T}$, (4.4.3)

$$[D(u)\nu_{\varepsilon}]_{\tan} = 0 \qquad \text{on} \quad \partial_{\ell}Q_{\varepsilon,T}.$$
(4.4.4)

Here $[a]_{tan}$ denotes the tangential component to $\partial \Omega_{\varepsilon}(t)$ of a vector $a \in \mathbb{R}^3$ and D(u) is the strain rate tensor given by

$$D(u) := \frac{\nabla u + (\nabla u)^T}{2},$$

where $(\nabla u)^T$ is the transposed matrix of ∇u .

In order to derive limit equations of the Navier–Stokes equations (4.4.1)–(4.4.4) we expand the velocity field u with respect to the signed distance d(x,t) as

$$u(x,t) = v(\pi(x,t),t) + d(x,t)v^{1}(\pi(x,t),t) + d(x,t)^{2}v^{2}(\pi(x,t),t) + R(d(x,t)^{3})$$
(4.4.5)

and the pressure p as (4.3.5). We need to expand u up to the second order term in d(x,t) since the momentum equation (4.4.1) has the second order derivatives of u.

Theorem 4.4.1. Let u and p satisfy the Navier–Stokes equations (4.4.1)–(4.4.4) in the moving thin domain $\Omega_{\varepsilon}(t)$. Then the normal component of the zeroth order term v in the expansion (4.4.5) is equal to the outward normal velocity of the moving surface $\Gamma(t)$, i.e. $v \cdot \nu = V_{\Gamma}^N$. Moreover, the velocity field v and the zeroth and first order terms q and q¹ in the expansion (4.3.5) satisfy

$$\partial_v^{\bullet} v + \nabla_{\Gamma} q + q^1 \nu = 2\mu_0 \operatorname{div}_{\Gamma} (P_{\Gamma} D^{tan}(v) P_{\Gamma}) \quad on \quad S_T,$$
(4.4.6)

$$\operatorname{div}_{\Gamma} v = 0 \qquad \qquad on \quad S_T. \tag{4.4.7}$$

Here $D^{tan}(v)$ is the tangential strain rate given by (4.2.12).

Remark 4.4.2. As in Remark 4.3.2, the normal component of v is equal to V_{Γ}^{N} , the tangential component of v and the scalar function q are determined by

$$P_{\Gamma}\partial_v^{\bullet}v + \nabla_{\Gamma}q = 2\mu_0 P_{\Gamma} \operatorname{div}_{\Gamma}(P_{\Gamma}D^{tan}(v)P_{\Gamma}), \quad \operatorname{div}_{\Gamma}v = 0 \quad \text{on} \quad S_T,$$
(4.4.8)

and the scalar function q^1 is given by the normal component of (4.4.6). The tangential system (4.4.8) is the same as the tangential incompressible Navier–Stokes–Scriven–Koba (NSSK) system in [18] with constant density (see (4.4) in [18]).

When $\Gamma(t) = \Gamma$ is fixed in time, the tangential system (4.4.8) is the same as the incompressible Navier–Stokes system on a fixed manifold derived by Taylor [31]

$$\partial_t v + \overline{\nabla}_v v + \nabla_\Gamma q = \mu_0 (\Delta_B v + K v), \quad \operatorname{div}_\Gamma v = 0 \quad \text{on} \quad \Gamma \times (0, T)$$
(4.4.9)

for a tangential velocity field v on Γ , although the authors of [18] claim that the system (4.4.8) on the stationary surface Γ is different from Taylor's model (4.4.9) (see Remark 4.4.3 below). Indeed, when the surface Γ is stationary, i.e. $V_{\Gamma}^{N} = 0$, the velocity field v in the system (4.4.8) is tangential and by applying (4.3.10) to the left-hand side of the first equation in (4.4.8) we obtain

$$\partial_t v + \overline{\nabla}_v v + \nabla_{\Gamma} q = 2\mu_0 P_{\Gamma} \operatorname{div}_{\Gamma} (P_{\Gamma} D^{tan}(v) P_{\Gamma}), \quad \operatorname{div}_{\Gamma} v = 0 \quad \text{on} \quad \Gamma \times (0, T).$$

Moreover, since v is tangential and satisfies $\operatorname{div}_{\Gamma} v = 0$, the right-hand side of the first equation in the above system is the same as that in Taylor's system (4.4.9) by (4.2.17). Hence the tangential incompressible Navier–Stokes system (4.4.8) on the stationary surface Γ agrees with the system (4.4.9) given by Taylor.

As in the case of the Euler equations (see Remark 4.3.2), when the surface is stationary the function q^1 in (4.4.6) is given by

$$q^{1} = \{-\partial_{v}^{\bullet}v + 2\mu_{0}\operatorname{div}_{\Gamma}(P_{\Gamma}D^{tan}(v)P_{\Gamma})\} \cdot \nu = -Av \cdot v + 2\mu_{0}\operatorname{tr}[A\nabla_{\Gamma}v],$$

where the second equality follows from (4.2.13) and (4.3.11). From this formula we observe that q^1 does not vanish in general even if the surface is stationary.

Remark 4.4.3. The authors of [18] argue that the tangential incompressible Navier–Stokes system (4.4.8) on a stationary surface Γ is different from the Navier–Stokes system (4.4.9) on a manifold given by Taylor [31], which is inconsistent with our argument in Remark 4.4.2. Unfortunately, there seems to be a flaw in derivation of Taylor's system (4.4.9) in [18, Section 5]. The authors of [18] applied an energetic variational approach with the dissipation energy given by the tangential strain rate $D^{tan}(v) = \{\nabla_{\Gamma}v + (\nabla_{\Gamma}v)^T\}/2$ to obtain (4.4.9). In their derivation of (4.4.9) they claim that $P_{\Gamma} \operatorname{div}_{\Gamma} (P_{\Gamma} D^{tan}(v)) = \Delta_B v + Kv$ holds on Γ when Γ is stationary and v is tangential and satisfies $\operatorname{div}_{\Gamma}v = 0$ (see the argument after [18, Theorem 5.1]). However, we have

$$2P_{\Gamma} \operatorname{div}_{\Gamma} (P_{\Gamma} D^{tan}(v)) = \Delta_B v + Kv - A^2 v$$

for any tangential vector field v on Γ satisfying $\operatorname{div}_{\Gamma} v = 0$, since the sum of the first two terms on the right-hand side is equal to $2P_{\Gamma}\operatorname{div}_{\Gamma}(P_{\Gamma}D^{tan}(v)P_{\Gamma})$ by (4.2.17) and

$$2P_{\Gamma}\operatorname{div}_{\Gamma}\left(P_{\Gamma}D^{tan}(v)\right) - 2P_{\Gamma}\operatorname{div}_{\Gamma}\left(P_{\Gamma}D^{tan}(v)P_{\Gamma}\right) = 2P_{\Gamma}\operatorname{div}_{\Gamma}\left(P_{\Gamma}D^{tan}(v)(\nu\otimes\nu)\right) = -A^{2}v$$

holds by the same calculations as in the proof of Lemma 4.2.3 (see Appendix 4.A).

It seems that their choice of the dissipation energy for derivation of (4.4.9) comes from a subtle misunderstanding of the strain rate tensor in Taylor's model, which is called the deformation tensor in [21,31]. Taylor [31] defined the deformation tensor Def v for a tangential vector field v on Γ as a symmetric tensor field of type (0, 2) on the manifold Γ (see e.g. [20, Chapter 12] for tensor fields) satisfying

$$(\operatorname{Def} v)(X,Y) = \frac{1}{2} \Big(\overline{\nabla}_X v \cdot Y + X \cdot \overline{\nabla}_Y v \Big), \quad X,Y \in C^{\infty}(T\Gamma),$$
(4.4.10)

where $C^{\infty}(T\Gamma)$ is the set of all smooth tangential vector fields on Γ . (See also (2.3) in [21]. Note that (2.3) in [21] is a formula for one-forms on Γ and here we identify tangential vector fields on Γ with one-forms on Γ via raising and lowering indices.) Let us show that the right-hand side of (4.4.10) is equal to $\{D^{tan}(v)X\} \cdot Y$. By the Gauss formula (4.2.16) and the fact that the covariant derivative $\overline{\nabla}_X v$ is tangential,

$$\overline{\nabla}_X v = P_{\Gamma}\{(X \cdot \nabla_{\Gamma})v\} = P_{\Gamma}(\nabla_{\Gamma}v)^T X \quad \text{on} \quad \Gamma,$$

where the second equality just follows from our notation on the tangential gradient matrix (see Section 4.2). From this formula and the facts that P_{Γ} is symmetric and that Y is tangential it follows that

$$\overline{\nabla}_X v \cdot Y = \{ P_{\Gamma} (\nabla_{\Gamma} v)^T X \} \cdot Y = \{ (\nabla_{\Gamma} v)^T X \} \cdot (P_{\Gamma} Y) = \{ (\nabla_{\Gamma} v)^T X \} \cdot Y.$$

Similarly we have $X \cdot \overline{\nabla}_Y v = X \cdot \{(\nabla_{\Gamma} v)^T Y\} = \{(\nabla_{\Gamma} v)X\} \cdot Y$ and thus

$$\frac{1}{2} \left(\overline{\nabla}_X v \cdot Y + X \cdot \overline{\nabla}_Y v \right) = \frac{1}{2} \left(\{ \nabla_\Gamma v + (\nabla_\Gamma v)^T \} X \right) \cdot Y = \{ D^{tan}(v)X \} \cdot Y.$$

Therefore, for any $X, Y \in C^{\infty}(T\Gamma)$ the equality

$$(\text{Def } v)(X, Y) = \{ D^{tan}(v)X \} \cdot Y$$
(4.4.11)

holds. Therefore, the deformation tensor Def v can be identified with the restriction on $C^{\infty}(T\Gamma) \times C^{\infty}(T\Gamma)$ of the symmetric bilinear map

$$\mathcal{T}_{D^{tan}(v)} \colon C^{\infty}(\Gamma)^3 \times C^{\infty}(\Gamma)^3 \to C^{\infty}(\Gamma), \quad (F,G) \mapsto \{D^{tan}(v)F\} \cdot G$$

Here $C^{\infty}(\Gamma)$ denotes the set of all smooth functions on Γ and $C^{\infty}(\Gamma)^3$ is the set of all smooth three-dimensional vector fields on Γ not necessarily tangential. However, it does not mean that Def v can be identified with the matrix $D^{tan}(v)$. Since Def v is a tensor field of type (0,2) on the manifold Γ , for any $X \in C^{\infty}(T\Gamma)$ the mapping

$$(\operatorname{Def} v)(X, \cdot) \colon C^{\infty}(T\Gamma) \to C^{\infty}(\Gamma), \quad Y \mapsto (\operatorname{Def} v)(X, Y)$$

is a linear map from $C^{\infty}(T\Gamma)$ into $C^{\infty}(\Gamma)$, i.e. a one-form on Γ . By identifying one-forms on Γ with tangential vector fields on Γ via raising and lowering indices, we may consider $(\text{Def } v)(X, \cdot) = (\text{Def } v)X$ as a tangential vector field on Γ . On the other hand, for a tangential vector field X on Γ the vector field $D^{tan}(v)X$ is not tangential in general, even if v is tangential to Γ . Indeed, since $(\nabla_{\Gamma}v)^{T}\nu = (\nabla_{\Gamma}v)^{T}P_{\Gamma}\nu = 0$ and $(\nabla_{\Gamma}v)\nu = -(\nabla_{\Gamma}\nu)v = Av$, where the second relation follows from the fact that v is tangential, we have

$$D^{tan}(v)\nu = \frac{1}{2}\{(\nabla_{\Gamma}v)\nu + (\nabla_{\Gamma}v)^{T}\nu\} = \frac{1}{2}Av.$$

From this equality and the symmetry of the matrix $D^{tan}(v)$ it follows that

$$D^{tan}(v)X \cdot \nu = X \cdot D^{tan}(v)\nu = \frac{1}{2}X \cdot Av$$

for any tangential vector field X on Γ . The last term does not vanish and thus the vector field $D^{tan}(v)X$ is not tangential on Γ in general.

To give a proper interpretation of the deformation tensor as a matrix, we observe that in (4.4.11) the vector fields X and Y are tangential to Γ and thus

$$\{D^{tan}(v)X\} \cdot Y = \{D^{tan}(v)P_{\Gamma}X\} \cdot (P_{\Gamma}Y) = \{P_{\Gamma}D^{tan}(v)P_{\Gamma}X\} \cdot Y$$

by the symmetry of the orthogonal projection P_{Γ} . Then (4.4.11) becomes

$$(\operatorname{Def} v)(X, Y) = \{P_{\Gamma} D^{tan}(v) P_{\Gamma} X\} \cdot Y$$

for all tangential vector fields X and Y on Γ . Moreover, the matrix $P_{\Gamma}D^{tan}(v)P_{\Gamma}$ is symmetric and for any $X \in C^{\infty}(T\Gamma)$ the vector field $P_{\Gamma}D^{tan}(v)P_{\Gamma}X$ is tangential to Γ . Therefore, we may identify the deformation tensor

Def
$$v = \mathcal{T}_{D^{tan}(v)}|_{C^{\infty}(T\Gamma) \times C^{\infty}(T\Gamma)} \colon C^{\infty}(T\Gamma) \times C^{\infty}(T\Gamma) \to C^{\infty}(\Gamma)$$

with the symmetric matrix $P_{\Gamma}D^{tan}(v)P_{\Gamma}$.

The matrix $P_{\Gamma}D^{tan}(v)P_{\Gamma}$ is called a projected strain rate in [18] and employed to define the dissipation energy in their energetic variational method for derivation of the incompressible NSSK system on the moving surface (see [18, Lemma 3.4 and Section 4]). Therefore, the strain rate tensor in Taylor's system (4.4.9) is the same as that in the tangential incompressible Navier–Stokes system (4.4.8).

Proof of Theorem 4.4.1. As in the proof of Theorem 4.3.1 we use the abbreviations (4.3.13). Due to the first boundary condition (4.4.3) we have

$$v(\pi, t) \cdot \nu(\pi, t) = V_{\Gamma}^{N}(\pi, t),$$
 (4.4.12)

$$v^{1}(\pi,t) \cdot \nu(\pi,t) = 0, \qquad (4.4.13)$$

$$v^{2}(\pi,t) \cdot \nu(\pi,t) = 0 \tag{4.4.14}$$

and the surface divergence-free condition (4.4.7) for v by the same argument as in the proof of Theorem 4.3.1. Moreover, we already calculated the expansion of the left-hand side of (4.4.1) in the proof of Theorem 4.3.1:

$$\partial_t u(x,t) + [(u \cdot \nabla)u](x,t) + \nabla p(x,t) = \partial_v^{\bullet} v(\pi,t) + \nabla_{\Gamma} q(\pi,t) + q^1(\pi,t)\nu(\pi,t) + R(d). \quad (4.4.15)$$

Let us show that the expansion of the viscous term Δu is of the form

$$\Delta u(x,t) = 2[\operatorname{div}_{\Gamma}(P_{\Gamma}D^{tan}(v)P_{\Gamma})](\pi,t) + R(d).$$
(4.4.16)

Since $\Delta u = 2 \operatorname{div} D(u)$ holds by the divergence-free condition (4.4.2), we consider the expansion in powers of d of the strain rate tensor D(u). We differentiate both sides of (4.4.5) with respect to x and apply (4.2.2) and (4.2.25) to get

$$\nabla u(x,t) = \nabla_{\Gamma} v(\pi,t) + [\nu \otimes v^{1}](\pi,t) + d(x,t) \{ [A \nabla_{\Gamma} v](\pi,t) + \nabla_{\Gamma} v^{1}(\pi,t) + 2[\nu \otimes v^{2}](\pi,t) \} + R(d^{2}). \quad (4.4.17)$$

Hence the strain rate tensor of u is expressed as

$$D(u)(x,t) = S(\pi,t) + d(x,t)S^{1}(\pi,t) + R(d^{2}), \qquad (4.4.18)$$

where

$$S := D^{tan}(v) + \frac{\nu \otimes v^1 + v^1 \otimes \nu}{2}, \tag{4.4.19}$$

$$S^{1} := \frac{A\nabla_{\Gamma}v + (A\nabla_{\Gamma}v)^{T}}{2} + D^{tan}(v^{1}) + \nu \otimes v^{2} + v^{2} \otimes \nu.$$
(4.4.20)

Let us write the second boundary condition (4.4.4) in terms of S and S¹. By (4.2.5) and (4.2.6) the boundary condition (4.4.4) reads

$$P_{\Gamma}(\pi,t)D(u)(x,t)\nu(\pi,t) = 0, \quad x \in \partial \Omega_{\varepsilon}(t).$$

We substitute (4.4.18) for the above D(u)(x,t) to obtain

$$P_{\Gamma}(\pi,t)S(\pi,t)\nu(\pi,t) \pm \varepsilon P_{\Gamma}(\pi,t)S^{1}(\pi,t)\nu(\pi,t) + O(\varepsilon^{2}) = 0$$

according to $d(x,t) = \pm \varepsilon$ (double-sign corresponds). Since the matrices $S(\pi,t)$, $S^1(\pi,t)$, $P_{\Gamma}(\pi,t)$, and the vector $\nu(\pi,t)$ are independent of ε , we have

$$P_{\Gamma}(\pi, t)S(\pi, t)\nu(\pi, t) = 0 \tag{4.4.21}$$

$$P_{\Gamma}(\pi, t)S^{1}(\pi, t)\nu(\pi, t) = 0.$$
(4.4.22)

Substituting (4.4.19) for S in (4.4.21) and observing

$$(\nu \otimes v^1)\nu = (v^1 \cdot \nu)\nu = 0, \quad (v^1 \otimes \nu)\nu = (\nu \cdot \nu)v^1 = v^1, \quad P_{\Gamma}v^1 = v^1$$

by (4.4.13) we get

$$v^{1}(\pi,t) = -2P_{\Gamma}(\pi,t)D^{tan}(v)(\pi,t)\nu(\pi,t).$$
(4.4.23)

Moreover, we multiply ν by S^1 given by (4.4.20) and apply

$$(A\nabla_{\Gamma}v)^{T}\nu = (\nabla_{\Gamma}v)^{T}A\nu = 0, \quad (\nabla_{\Gamma}v^{1})^{T}\nu = (\nabla_{\Gamma}v^{1})^{T}P_{\Gamma}\nu = 0$$

by the symmetry of A and P_{Γ} , $\nabla_{\Gamma} = P_{\Gamma} \nabla_{\Gamma}$, and (4.2.7), and then use $(\nu \otimes v^2)\nu = 0$ and $(v^2 \otimes \nu)\nu = v^2$ by (4.4.14) to obtain

$$S^{1}\nu = \frac{1}{2}(A\nabla_{\Gamma}v + \nabla_{\Gamma}v^{1})\nu + v^{2}.$$
(4.4.24)

It is tangential to $\Gamma(t)$ by $\nabla_{\Gamma} = P_{\Gamma} \nabla$, (4.2.7) and (4.4.14). Hence (4.4.22) yields

$$S^{1}(\pi, t)\nu(\pi, t) = 0. \tag{4.4.25}$$

Now we apply the formula (4.2.27) to the expansion (4.4.18). Then by the symmetry of S^1 (see (4.4.20)) and the equality (4.4.25) we get

$$\operatorname{div} D(u)(x,t) = \operatorname{div}_{\Gamma} S(\pi,t) + R(d).$$
(4.4.26)

Let us write S in terms of v. Substituting (4.4.23) for (4.4.19), using the formulas

$$(Ma) \otimes b = M(a \otimes b), \quad a \otimes (Mb) = (a \otimes b)M^T$$

for a square matrix M of order three and three-dimensional vectors a and b, and observing $(P_{\Gamma}D^{tan}(v))^T = D^{tan}(v)P_{\Gamma}$ by the symmetry of P_{Γ} and $D^{tan}(v)$, we have

$$S = D^{tan}(v) - (\nu \otimes \nu)D^{tan}(v)P_{\Gamma} - P_{\Gamma}D^{tan}(v)(\nu \otimes \nu)$$

= $P_{\Gamma}D^{tan}(v)P_{\Gamma} + (\nu \otimes \nu)D^{tan}(v)(\nu \otimes \nu).$

Here the second term on the last line vanishes by $(\nu \otimes \nu)\nabla_{\Gamma}v = (\nabla_{\Gamma}v)^T(\nu \otimes \nu) = 0$. Hence it follows that

$$S(\pi, t) = P_{\Gamma}(\pi, t) D^{tan}(v)(\pi, t) P_{\Gamma}(\pi, t)$$
(4.4.27)

and we obtain (4.4.16) by applying (4.4.26) and (4.4.27) to $\Delta u = 2 \operatorname{div} D(u)$. Finally, we substitute (4.4.15) and (4.4.16) for the momentum equation (4.4.1) to get

$$\partial_{v}^{\bullet} v(\pi, t) + \nabla_{\Gamma} q(\pi, t) + q^{1}(\pi, t) \nu(\pi, t) + R(d) = 2\mu_{0} [\operatorname{div}_{\Gamma} (P_{\Gamma} D^{tan}(v) P_{\Gamma})](\pi, t) + R(d).$$

Since all terms except for R(d) are independent of d, we conclude that the equation (4.4.6) should be satisfied.

Remark 4.4.4. We may replace the perfect slip condition (4.4.4) by the partial slip condition

$$[D(u)\nu_{\varepsilon}]_{\tan} + k(u^T - v_{\Omega}^T) = 0 \quad \text{on} \quad \partial_{\ell}Q_{\varepsilon,T},$$

where $u^T = (I_3 - \nu_{\varepsilon} \otimes \nu_{\varepsilon})u$, k > 0 is a constant, and $v_{\Omega}^T(\cdot, t)$ is a given tangential velocity field on $\partial \Omega_{\varepsilon}(t)$. However, it makes the limit velocity overdetermined. Indeed, suppose that v_{Ω}^T is given by

$$v_{\Omega}^{T}(x,t) = \begin{cases} v_{\text{outer}}(\pi(x,t),t) & \text{if } d(x,t) = \varepsilon, \\ v_{\text{inner}}(\pi(x,t),t) & \text{if } d(x,t) = -\varepsilon. \end{cases}$$

where $v_{outer}(\cdot, t)$ and $v_{inner}(\cdot, t)$ are given tangential velocity fields on $\Gamma(t)$. Then the same calculations as in the proof of Theorem 4.4.1 yield

$$v = V_{\Gamma}^{N} \nu + \frac{v_{\text{outer}} + v_{\text{inner}}}{2}.$$

Hence the limit velocity v is completely determined by given velocities while it should satisfy similar equations to (4.4.6) and (4.4.7).

Remark 4.4.5. In the proof of Theorem 4.4.1 we obtained the expansion (4.4.16) of the viscous term Δu by using the expansion of the strain rate tensor D(u). Here let us expand Δu by direct calculations. In what follows, we abbreviate $\pi(x, t)$ and R(d(x, t)) to π and R(d) for $x \in \Omega_{\varepsilon}(t)$ and suppress the argument t. By (4.4.17) the gradient of the j-th component of u (j = 1, 2, 3) is

$$\nabla u_j(x) = \nabla_{\Gamma} v_j(\pi) + v_j^1(\pi) \nu(\pi) + d(x) F_j(\pi) + R(d^2), \qquad (4.4.28)$$

where $F_j = A \nabla_{\Gamma} v_j + \nabla_{\Gamma} v_j^1 + 2v_j^2 \nu$. We differentiate both sides of (4.4.28) with respect to x and apply (4.2.2), (4.2.25), and $\nabla_{\Gamma} \nu = -A$ to get

$$\nabla^2 u_j(x) = \nabla^2_{\Gamma} v_j(\pi) + [\nabla_{\Gamma} v_j^1 \otimes \nu](\pi) - v_j^1(\pi) A(\pi) + [\nu \otimes F_j](\pi) + R(d).$$

Taking the trace of both sides and observing $A \nabla_{\Gamma} v_j \cdot \nu = \nabla_{\Gamma} v_j^1 \cdot \nu = 0$ we obtain

$$\Delta u_j(x) = \Delta_{\Gamma} v_j(\pi) - v_j^1(\pi) H(\pi) + 2v_j^2(\pi) + R(d)$$

for each j = 1, 2, 3 and thus

$$\Delta u_j(x) = \Delta_{\Gamma} v(\pi) - H(\pi) v^1(\pi) + 2v^2(\pi) + R(d).$$

Let us express v^1 and v^2 in terms of v. The first order term v^1 is given by (4.4.23), $\nabla_{\Gamma} = P_{\Gamma} \nabla_{\Gamma}$, and $(\nabla_{\Gamma} v)^T \nu = (\nabla_{\Gamma} v)^T P_{\Gamma} \nu = 0$:

$$v^1 = -2P_{\Gamma}D^{tan}(v)\nu = -(\nabla_{\Gamma}v)\nu.$$

By (4.4.24) and (4.4.25) we can represent v^2 in terms of v and v^1 as

$$v^2 = -\frac{1}{2}(A\nabla_{\Gamma}v + \nabla_{\Gamma}v^1)\nu.$$

From this it follows that $v^2 = 0$ since $v^1 = -(\nabla_{\Gamma} v)\nu$ is tangential and thus

$$(\nabla_{\Gamma} v^{1})\nu = \nabla_{\Gamma} (v^{1} \cdot \nu) - (\nabla_{\Gamma} \nu)v^{1} = Av^{1} = -A(\nabla_{\Gamma} v)\nu.$$

Hence we obtain another expansion formula of the viscous term

$$\Delta u(x) = \Delta_{\Gamma} v(\pi) + [H(\nabla_{\Gamma} v)\nu](\pi) + R(d).$$
(4.4.29)

Comparing the expansions (4.4.16) and (4.4.29) we expect that the equality

$$2\operatorname{div}_{\Gamma}(P_{\Gamma}D^{tan}(v)P_{\Gamma}) = \Delta_{\Gamma}v + H(\nabla_{\Gamma}v)\nu$$
(4.4.30)

holds for the limit velocity v. Let us prove this equality. By the formula (4.2.13) for the left-hand side, the proof of (4.4.30) reduces to showing

$$\nabla_{\Gamma}(\operatorname{div}_{\Gamma} v) = 0, \quad 2\operatorname{tr}[A\nabla_{\Gamma} v] = (\Delta_{\Gamma} v) \cdot \nu. \tag{4.4.31}$$

The first equality follows from the surface divergence-free condition (4.4.7) for the limit velocity v. To obtain the second equality we need to observe the expansion of the divergence-free condition (4.4.2) in powers of the signed distance d up to the first order term. Taking the trace of (4.4.17) and using $v^1 \cdot \nu = 0$ and $v^2 = 0$ we have

$$\operatorname{div} u(x) = \operatorname{div}_{\Gamma} v(\pi) + d(x) \{ \operatorname{tr}[A\nabla_{\Gamma} v](\pi) + \operatorname{div}_{\Gamma} v^{1}(\pi) \} + R(d^{2}).$$

Since the left-hand side vanishes for all $x \in \Omega_{\varepsilon}(t)$ by (4.4.2), observing the first order term in d(x) on the right-hand side we obtain

$$\operatorname{tr}[A\nabla_{\Gamma}v] + \operatorname{div}_{\Gamma}v^{1} = 0. \tag{4.4.32}$$

To the second term on the left-hand side we apply $v^1 = -(\nabla_{\Gamma} v)\nu$. Then since

$$div_{\Gamma}[(\nabla_{\Gamma}v)\nu] = (div_{\Gamma}\nabla_{\Gamma}v) \cdot \nu + tr[(\nabla_{\Gamma}\nu)^{T}\nabla_{\Gamma}v],$$

= $(\Delta_{\Gamma}v) \cdot \nu - tr[A^{T}\nabla_{\Gamma}v]$

and the Weingarten map A is symmetric, the equality (4.4.32) becomes

$$2\mathrm{tr}[A\nabla_{\Gamma}v] - (\Delta_{\Gamma}v) \cdot \nu = 0.$$

Hence the second equality in (4.4.31) holds and (4.4.30) follows.

4.5 Energy identities

The purpose of this section is to find a relation between energy identities of the Euler and Navier–Stokes equations in the moving thin domains and those of the limit equations on the moving surface. We first derive the energy identities from the equations and then show that the energy identities of the limit surface equations are also derived as thin width limits of those of the original bulk equations.

4.5.1 Euler equations

Lemma 4.5.1. Let u and p satisfy the Euler equations (4.3.1)–(4.3.3) in the moving thin domain $\Omega_{\varepsilon}(t)$. Then we have

$$\frac{d}{dt} \int_{\Omega_{\varepsilon}(t)} \frac{|u|^2}{2} dx = -\int_{\partial\Omega_{\varepsilon}(t)} p V_{\varepsilon}^N d\mathcal{H}^2.$$
(4.5.1)

The identity (4.5.1) means that the rate of change of the kinetic energy of the incompressible perfect fluid in a moving domain is equal to the rate of work done by the pressure caused by the motion of the boundary.

Proof. By the Reynolds transport theorem (see [14]) and (4.3.1),

$$\frac{d}{dt} \int_{\Omega_{\varepsilon}(t)} \frac{|u|^2}{2} dx = \int_{\Omega_{\varepsilon}(t)} u \cdot \partial_t u \, dx + \int_{\partial\Omega_{\varepsilon}(t)} \frac{|u|^2}{2} V_{\varepsilon}^N \, d\mathcal{H}^2 \qquad (4.5.2)$$
$$= \int_{\Omega_{\varepsilon}(t)} u \cdot \{-(u \cdot \nabla)u - \nabla p\} \, dx + \int_{\partial\Omega_{\varepsilon}(t)} \frac{|u|^2}{2} V_{\varepsilon}^N \, d\mathcal{H}^2.$$

By integration by parts and the equations (4.3.2) and (4.3.3) we have

$$\int_{\Omega_{\varepsilon}(t)} u \cdot (u \cdot \nabla) u \, dx = \int_{\partial\Omega_{\varepsilon}(t)} |u|^2 (u \cdot \nu_{\varepsilon}) \, d\mathcal{H}^2 - \int_{\Omega_{\varepsilon}(t)} \{u \cdot (u \cdot \nabla) u + |u|^2 \operatorname{div} u\} \, dx$$
$$= \int_{\partial\Omega_{\varepsilon}(t)} |u|^2 V_{\varepsilon}^N \, d\mathcal{H}^2 - \int_{\Omega_{\varepsilon}(t)} u \cdot (u \cdot \nabla) u \, dx.$$

Therefore,

$$\int_{\Omega_{\varepsilon}(t)} u \cdot (u \cdot \nabla) u \, dx = \int_{\partial \Omega_{\varepsilon}(t)} \frac{|u|^2}{2} V_{\varepsilon}^N \, d\mathcal{H}^2.$$
(4.5.3)

On the other hand, by integration by parts

$$\int_{\Omega_{\varepsilon}(t)} u \cdot \nabla p \, dx = \int_{\partial\Omega_{\varepsilon}(t)} (u \cdot \nu_{\varepsilon}) p \, d\mathcal{H}^2 - \int_{\Omega_{\varepsilon}(t)} (\operatorname{div} u) p \, dx$$

and we apply (4.3.2) and (4.3.3) to the right-hand side to get

$$\int_{\Omega_{\varepsilon}(t)} u \cdot \nabla p \, dx = \int_{\partial\Omega_{\varepsilon}(t)} p V_{\varepsilon}^{N} \, d\mathcal{H}^{2}.$$
(4.5.4)

Substituting (4.5.3) and (4.5.4) for (4.5.2) we obtain the energy identity (4.5.1). \Box

Lemma 4.5.2. Let v, q, and q^1 satisfy the limit equations (4.3.6) and (4.3.7) of the Euler equations. Suppose that the normal component of v is equal to the outward normal velocity of $\Gamma(t)$, i.e. $v \cdot \nu = V_{\Gamma}^N$. Then we have

$$\frac{d}{dt} \int_{\Gamma(t)} \frac{|v|^2}{2} d\mathcal{H}^2 = \int_{\Gamma(t)} (qH - q^1) V_{\Gamma}^N d\mathcal{H}^2.$$
(4.5.5)

The right-hand side of (4.5.5) represents the rate of work done by the moving surface to the fluid. Note that it contains the scalar function q^1 , which corresponds to the normal derivative of the surface pressure.

Proof. By the assumption we can write $v = V_{\Gamma}^{N}\nu + v^{T}$ with a tangential velocity field v^{T} on $\Gamma(t)$. We apply the Leibniz formula (see [10, Lemma 2.2]) with $v = V_{\Gamma}^{N}\nu + v^{T}$ to the integral of $|v|^{2}/2$ over $\Gamma(t)$. (Note that the tangential velocity v^{T} does not affect the change of the shape of $\Gamma(t)$.) Then we have

$$\frac{d}{dt} \int_{\Gamma(t)} \frac{|v|^2}{2} d\mathcal{H}^2 = \int_{\Gamma(t)} \left\{ \partial_v^{\bullet} \left(\frac{|v|^2}{2} \right) + \frac{|v|^2}{2} \operatorname{div}_{\Gamma} v \right\} d\mathcal{H}^2$$
$$= \int_{\Gamma(t)} v \cdot \partial_v^{\bullet} v \, d\mathcal{H}^2 + \int_{\Gamma(t)} \frac{|v|^2}{2} \operatorname{div}_{\Gamma} v \, d\mathcal{H}^2.$$

To the last line we apply the equations (4.3.6) and (4.3.7). Then

$$\frac{d}{dt} \int_{\Gamma(t)} \frac{|v|^2}{2} d\mathcal{H}^2 = -\int_{\Gamma(t)} v \cdot (\nabla_{\Gamma} q + q^1 \nu) d\mathcal{H}^2.$$
(4.5.6)

For the first term on the right-hand side,

$$v \cdot \nabla_{\Gamma} q = \operatorname{div}_{\Gamma}(qv) + q \operatorname{div}_{\Gamma} v = -qHV_{\Gamma}^{N} + \operatorname{div}_{\Gamma}(qv^{T})$$

by $v = V_{\Gamma}^{N} \nu + v^{T}$, $\nabla_{\Gamma}(qV_{\Gamma}^{N}) \cdot \nu = 0$, $\operatorname{div}_{\Gamma} \nu = -H$, and (4.3.7). Moreover, the integral of the surface divergence of the tangential vector field qv^{T} over $\Gamma(t)$ vanishes by Stokes' theorem since $\Gamma(t)$ is closed. Hence we have

$$\int_{\Gamma(t)} v \cdot \nabla_{\Gamma} q \, d\mathcal{H}^2 = -\int_{\Gamma(t)} q H V_{\Gamma}^N \, d\mathcal{H}^2.$$
(4.5.7)

For the second term we have

$$\int_{\Gamma(t)} v \cdot (q^1 \nu) \, d\mathcal{H}^2 = \int_{\Gamma(t)} q^1 V_{\Gamma}^N \, d\mathcal{H}^2 \tag{4.5.8}$$

by $v \cdot \nu = V_{\Gamma}^{N}$. The energy identity (4.5.5) follows from (4.5.6), (4.5.7), and (4.5.8).

Let us show that the energy identity (4.5.5) on the moving surface can be derived as a thin width limit of that in the moving thin domain (4.5.1). As in Section 4.3 we expand the velocity u and the pressure p in powers of the signed distance d as (4.3.4) and (4.3.5) and determine the zeroth order term in ε of the energy identity (4.5.1).

Theorem 4.5.3. Let u and p satisfy the energy identity (4.5.1). Then the zeroth order term v in the expansion (4.3.4) and the zeroth and first order terms q and q^1 in the expansion (4.3.5) satisfy the energy identity (4.5.5).

Proof. From the expansion (4.3.4) we have

$$\frac{|u(x,t)|^2}{2} = \frac{|v(\pi(x,t),t)|^2}{2} + d(x,t)V(\pi(x,t),t) + R(d(x,t)^2)$$

for $x \in \Omega_{\varepsilon}(t)$, where $V := v \cdot v^1$. Using this expansion we write

$$\int_{\Omega_{\varepsilon}(t)} \frac{|u(x,t)|^2}{2} \, dx = I_1 + I_2 + I_3,$$

where

$$I_1 := \int_{\Omega_{\varepsilon}(t)} \frac{|v(\pi(x,t),t)|^2}{2} dx,$$
$$I_2 := \int_{\Omega_{\varepsilon}(t)} d(x,t) V(\pi(x,t),t) dx, \quad I_3 := \int_{\Omega_{\varepsilon}(t)} R(d(x,t)^2) dx.$$

To I_1 and I_2 we apply the change of variables formula (4.2.20) to get

$$\begin{split} I_1 &= \int_{\Gamma(t)} \int_{-\varepsilon}^{\varepsilon} \frac{|v(y,t)|^2}{2} J(y,t,\rho) \, d\rho \, d\mathcal{H}^2(y) = 2\varepsilon \int_{\Gamma(t)} \frac{|v(y,t)|^2}{2} \, d\mathcal{H}^2(y) + \varepsilon^2 f_1(\varepsilon,t), \\ I_2 &= \int_{\Gamma(t)} \int_{-\varepsilon}^{\varepsilon} \rho V(y,t) J(y,t,\rho) \, d\rho \, d\mathcal{H}^2(y) = \varepsilon^2 f_2(\varepsilon,t), \end{split}$$

where f_1 and f_2 are polynomials in ε with time-dependent coefficients. (Note that the Jacobian $J(y, t, \rho)$ given by (4.2.19) is a polynomial in ρ and the principal curvatures of $\Gamma(t)$.) Hence

$$\frac{dI_1}{dt} = 2\varepsilon \frac{d}{dt} \int_{\Gamma(t)} \frac{|v(y,t)|^2}{2} d\mathcal{H}^2(y) + O(\varepsilon^2), \quad \frac{dI_2}{dt} = O(\varepsilon^2).$$
(4.5.9)

For I_3 , using the Reynolds transport theorem and observing that the first order time derivative of $R(d(x,t)^2)$ is R(d(x,t)) we have

$$\frac{dI_3}{dt} = \int_{\Omega_{\varepsilon}(t)} R(d(x,t)) \, dx + \int_{\partial\Omega_{\varepsilon}(t)} R(d(x,t)^2) V_{\varepsilon}^N(x,t) \, d\mathcal{H}^2(x)$$

We apply the change of variables formula (4.2.20) to the first term on the right-hand side of the above equality. Then by $R(d(x,t)) = R(\rho) = O(\varepsilon)$ and $J(y,t,\rho) = O(1)$ for $d(x,t) = \rho \in (-\varepsilon,\varepsilon)$ with $x \in \Omega_{\varepsilon}(t)$ to get

$$\int_{\Omega_{\varepsilon}(t)} R(d(x,t)) \, dx = \int_{\Gamma(t)} \int_{-\varepsilon}^{\varepsilon} R(\rho) J(y,t,\rho) \, d\rho \, d\mathcal{H}^2(y) = O(\varepsilon^2).$$

Moreover, by $R(d(x,t)^2) = O(\varepsilon^2)$ for $x \in \partial \Omega_{\varepsilon}(t)$ and

$$|V_{\varepsilon}^{N}(x,t)| = |V_{\Gamma}^{N}(\pi(x,t),t)| = O(1), \quad x \in \partial\Omega_{\varepsilon}(t)$$

which follows from (4.2.6) and the fact that V_{Γ}^{N} is independent of ε , and the change of variables formula (4.2.21) and $J(y, t, \pm \varepsilon) = O(1)$, we have

$$\int_{\partial\Omega_{\varepsilon}(t)} R(d(x,t)^2) V_{\varepsilon}^N(x,t) \, d\mathcal{H}^2(x) = \sum_{\rho=\pm\varepsilon} \int_{\Gamma(t)} O(\varepsilon^2) J(y,t,\rho) \, d\mathcal{H}^2(y) = O(\varepsilon^2).$$

Hence $dI_3/dt = O(\varepsilon^2)$. From this estimate and (4.5.9) it follows that

$$\frac{d}{dt} \int_{\Omega_{\varepsilon}(t)} \frac{|u(x,t)|^2}{2} dx = \frac{d}{dt} (I_1 + I_2 + I_3) = 2\varepsilon \frac{d}{dt} \int_{\Gamma(t)} \frac{|v(y,t)|^2}{2} d\mathcal{H}^2(y) + O(\varepsilon^2).$$
(4.5.10)

Let us expand the right-hand side of the energy identity (4.5.1) in ε . By the expansion (4.3.5) of the pressure p, the relation (4.2.6), and the formula (4.2.21),

$$\int_{\partial\Omega_{\varepsilon}(t)} p(x,t) V_{\varepsilon}^{N}(x,t) \, d\mathcal{H}^{2}(x) = J_{1} + \varepsilon J_{2} + O(\varepsilon^{2}), \qquad (4.5.11)$$

where

$$J_1 := \int_{\Gamma(t)} q(y,t) V_{\Gamma}^N(y,t) \{ J(y,t,\varepsilon) - J(y,t,-\varepsilon) \} d\mathcal{H}^2(y),$$

$$J_2 := \int_{\Gamma(t)} q^1(y,t) V_{\Gamma}^N(y,t) \{ J(y,t,\varepsilon) + J(y,t,-\varepsilon) \} d\mathcal{H}^2(y).$$

From (4.2.19) we have

$$\begin{split} J(y,t,\varepsilon) &- J(y,t,-\varepsilon) = -2\varepsilon H(y,t) + O(\varepsilon^2), \\ J(y,t,\varepsilon) &+ J(y,t,-\varepsilon) = 2 + O(\varepsilon^2). \end{split}$$

Hence

$$J_1 = -2\varepsilon \int_{\Gamma(t)} q(y,t)H(y,t)V_{\Gamma}^N(y,t) d\mathcal{H}^2(y)$$
$$J_2 = 2 \int_{\Gamma(t)} q^1(y,t)V_{\Gamma}^N(y,t) d\mathcal{H}^2(y)$$

and applying them to the right-hand side of (4.5.11) we get

$$\int_{\partial\Omega_{\varepsilon}(t)} p(x,t) V_{\varepsilon}^{N}(x,t) d\mathcal{H}^{2}(x)$$

$$= -2\varepsilon \int_{\Gamma(t)} \{q(y,t)H(y,t) - q^{1}(y,t)\} V_{\Gamma}^{N}(y,t) d\mathcal{H}^{2}(y) + O(\varepsilon^{2}). \quad (4.5.12)$$

Finally, we substitute (4.5.10) and (4.5.12) for (4.5.1) and divide both sides by 2ε to obtain

$$\frac{d}{dt} \int_{\Gamma(t)} \frac{|v(y,t)|^2}{2} \, d\mathcal{H}^2(y) = \int_{\Gamma(t)} \{q(y,t)H(y,t) - q^1(y,t)\} V_{\Gamma}^N(y,t) \, d\mathcal{H}^2(y) + O(\varepsilon).$$

Since the left-hand side and the first term on the right-hand side are independent of ε , we conclude that the identity (4.5.5) should be satisfied.

4.5.2 Navier–Stokes equations

Lemma 4.5.4. Let u and p satisfy the Navier–Stokes equations (4.4.1)–(4.4.4) in the moving thin domain $\Omega_{\varepsilon}(t)$. Then we have

$$\frac{d}{dt} \int_{\Omega_{\varepsilon}(t)} \frac{|u|^2}{2} dx = -2\mu_0 \int_{\Omega_{\varepsilon}(t)} |D(u)|^2 dx + \int_{\partial\Omega_{\varepsilon}(t)} (\sigma\nu_{\varepsilon} \cdot \nu_{\varepsilon}) V_{\varepsilon}^N d\mathcal{H}^2.$$
(4.5.13)

Here $\sigma := 2\mu_0 D(u) - pI_3$ denotes the Cauchy stress tensor.

The first term on the right-hand side of (4.5.13) represents the energy dissipation by viscosity and the second term stands for the rate of work done by the normal component of the stress vector $\sigma \nu_{\varepsilon}$ on the moving boundary.

Proof. By the Reynolds transport theorem (see [14]) and the equation (4.4.1),

$$\frac{d}{dt} \int_{\Omega_{\varepsilon}(t)} \frac{|u|^2}{2} dx = \int_{\Omega_{\varepsilon}(t)} u \cdot \partial_t u \, dx + \int_{\partial\Omega_{\varepsilon}(t)} \frac{|u|^2}{2} V_{\varepsilon}^N \, d\mathcal{H}^2
= \int_{\Omega_{\varepsilon}(t)} u \cdot \{-(u \cdot \nabla)u - \nabla p + \mu_0 \Delta u\} \, dx + \int_{\partial\Omega_{\varepsilon}(t)} \frac{|u|^2}{2} V_{\varepsilon}^N \, d\mathcal{H}^2.$$
(4.5.14)

We already computed the integrals of $u \cdot (u \cdot \nabla)u$ and $u \cdot \nabla p$ over $\Omega_{\varepsilon}(t)$ in the proof of Lemma 4.5.1, see (4.5.3) and (4.5.4). Let us calculate the integral of $u \cdot \Delta u$. Since $\Delta u = 2 \operatorname{div} D(u)$ by the divergence-free condition (4.4.2),

$$\int_{\Omega_{\varepsilon}(t)} u \cdot \Delta u \, dx = 2 \int_{\Omega_{\varepsilon}(t)} u \cdot \operatorname{div} D(u) \, dx$$
$$= 2 \int_{\partial\Omega_{\varepsilon}(t)} u \cdot D(u)^{T} \nu_{\varepsilon} \, d\mathcal{H}^{2} - 2 \int_{\Omega_{\varepsilon}(t)} \nabla u : D(u) \, dx,$$

where $F : G := tr[F^T G]$ for square matrices F and G of order three. In the last line we use the symmetry of the strain rate tensor D(u) and the boundary conditions (4.4.3) and (4.4.4) to get

$$u \cdot D(u)^T \nu_{\varepsilon} = (u \cdot \nu_{\varepsilon})(D(u)\nu_{\varepsilon} \cdot \nu_{\varepsilon}) = V_{\varepsilon}^N(D(u)\nu_{\varepsilon} \cdot \nu_{\varepsilon})$$

on $\partial \Omega_{\varepsilon}(t)$. Also, we easily observe that

$$\nabla u : D(u) = (\nabla u)^T : D(u) = |D(u)|^2 \left(= \sum_{i,j=1}^3 [D(u)]_{ij}^2 \right).$$

Here $[D(u)]_{ij}$ is the (i, j)-entry of D(u), i.e. $[D(u)]_{ij} = (\partial_i u_j + \partial_j u_i)/2$. Hence

$$\int_{\Omega_{\varepsilon}(t)} u \cdot \Delta u \, dx = 2 \int_{\partial\Omega_{\varepsilon}(t)} (D(u)\nu_{\varepsilon} \cdot \nu_{\varepsilon}) V_{\varepsilon}^{N} \, d\mathcal{H}^{2} - 2 \int_{\Omega_{\varepsilon}(t)} |D(u)|^{2} \, dx.$$
(4.5.15)

Finally we substitute (4.5.3), (4.5.4), and (4.5.15) for (4.5.14) to obtain (4.5.13).

Lemma 4.5.5. Let v, q, and q^1 satisfy the limit equations (4.4.6) and (4.4.7) of the Navier-Stokes equations. Suppose that the normal component of v is equal to the outward normal velocity of $\Gamma(t)$, i.e. $v \cdot v = V_{\Gamma}^{N}$. Then we have

$$\frac{d}{dt} \int_{\Gamma(t)} \frac{|v|^2}{2} d\mathcal{H}^2 = -2\mu_0 \int_{\Gamma(t)} |P_{\Gamma} D^{tan}(v) P_{\Gamma}|^2 d\mathcal{H}^2 + \int_{\Gamma(t)} (qH - q^1) V_{\Gamma}^N d\mathcal{H}^2.$$
(4.5.16)

The first and second terms on the right-hand side of (4.5.16) correspond to the energy dissipation of the surface fluid by viscosity and the rate of work done by the moving surface, respectively.

Proof. As in the proof of Lemma 4.5.2 we use the Leibniz formula [10, Lemma 2.2] with velocity field v and the equations (4.4.6) and (4.4.7):

$$\frac{d}{dt} \int_{\Gamma(t)} \frac{|v|^2}{2} d\mathcal{H}^2 = \int_{\Gamma(t)} v \cdot \partial_v^{\bullet} v \, d\mathcal{H}^2 + \int_{\Gamma(t)} \frac{|v|^2}{2} \operatorname{div}_{\Gamma} v \, d\mathcal{H}^2
= \int_{\Gamma(t)} v \cdot \{-\nabla_{\Gamma} q - q^1 \nu + 2\mu_0 \operatorname{div}_{\Gamma} (P_{\Gamma} D^{tan}(v) P_{\Gamma})\} \, d\mathcal{H}^2.$$
(4.5.17)

The first two terms in the last line were calculated in the proof of Lemma 4.5.2, see (4.5.7) and (4.5.8). For the viscous term,

$$v \cdot \operatorname{div}_{\Gamma}(P_{\Gamma}D^{tan}(v)P_{\Gamma}) = \operatorname{div}_{\Gamma}(P_{\Gamma}D^{tan}(v)P_{\Gamma}v) - \nabla_{\Gamma}v : P_{\Gamma}D^{tan}(v)P_{\Gamma}.$$

The integral of the first term on the right-hand side over $\Gamma(t)$ vanishes by Stokes' theorem since $\Gamma(t)$ is closed and $P_{\Gamma}D^{tan}(v)P_{\Gamma}v$ is a tangential vector field on $\Gamma(t)$. Also, since the matrix $P_{\Gamma}D^{tan}(v)P_{\Gamma}$ is symmetric,

$$\nabla_{\Gamma} v : P_{\Gamma} D^{tan}(v) P_{\Gamma} = (\nabla_{\Gamma} v)^T : P_{\Gamma} D^{tan}(v) P_{\Gamma} = D^{tan}(v) : P_{\Gamma} D^{tan}(v) P_{\Gamma}.$$

Moreover, by the formulas $P_{\Gamma}^2 = P_{\Gamma}^T = P_{\Gamma}$ and $E : FG = F^TE : G = EG^T : F$ for square matrices E, F, and G of order three we obtain

$$\nabla_{\Gamma} v : P_{\Gamma} D^{tan}(v) P_{\Gamma} = D^{tan}(v) : P_{\Gamma} D^{tan}(v) P_{\Gamma} = |P_{\Gamma} D^{tan}(v) P_{\Gamma}|^{2}.$$

Hence the integral of the inner product of v and $\operatorname{div}_{\Gamma}(P_{\Gamma}D^{tan}(v)P_{\Gamma})$ is

$$\int_{\Gamma(t)} v \cdot \operatorname{div}_{\Gamma}(P_{\Gamma}D^{tan}(v)P_{\Gamma}) \, d\mathcal{H}^2 = -\int_{\Gamma(t)} |P_{\Gamma}D^{tan}(v)P_{\Gamma}|^2 \, d\mathcal{H}^2.$$
(4.5.18)

Applying (4.5.7), (4.5.8), and (4.5.18) to (4.5.17) we obtain (4.5.16).

As in the case of the Euler equations, the energy identity (4.5.16) on the moving surface can be derived as a thin width limit of that in the moving thin domain (4.5.13). Let us expand u and p in powers of d as (4.4.5) and (4.3.5) and determine the zeroth order term in ε of the energy identity (4.5.13).

Theorem 4.5.6. Let u and p satisfy the energy identity (4.5.13). Suppose that the velocity field u satisfies the boundary conditions (4.4.3) and (4.4.4). Then the zeroth order term v in the expansion (4.4.5) and the zeroth and first order terms q and q^1 in the expansion (4.3.5) satisfy the energy identity (4.5.16).

Proof. The remaining part of the proof is to show that

$$\int_{\Omega_{\varepsilon}(t)} |D(u)(x,t)|^2 dx = 2\varepsilon \int_{\Gamma(t)} |(P_{\Gamma}D^{tan}(v)P_{\Gamma})(y,t)|^2 d\mathcal{H}^2(y) + O(\varepsilon^2)$$
(4.5.19)

and

$$\int_{\partial\Omega_{\varepsilon}(t)} [(D(u)\nu_{\varepsilon} \cdot \nu_{\varepsilon})V_{\varepsilon}^{N}](x,t) \, d\mathcal{H}^{2}(x) = O(\varepsilon^{2}) \tag{4.5.20}$$

since we already computed other terms in the proof of Theorem 4.5.3, see (4.5.10) and (4.5.12). By (4.4.18) and (4.4.27) in the proof of Theorem 4.4.1 we have

$$D(u)(x,t) = (P_{\Gamma}D^{tan}(v)P_{\Gamma})(\pi(x,t),t) + d(x,t)S^{1}(\pi(x,t),t) + R(d(x,t)^{2})$$

for $x \in \Omega_{\varepsilon}(t)$ (Note that to get (4.4.27) we only need the boundary conditions (4.4.3) and (4.4.4) for the Navier–Stokes equations. See the proof of Theorem 4.4.1.) Using this expansion and the change of variable formula (4.2.20) we obtain (4.5.19) as

$$\begin{split} \int_{\Omega_{\varepsilon}(t)} |D(u)(x,t)|^2 \, dx &= \int_{\Omega_{\varepsilon}(t)} \{ |(P_{\Gamma}D^{tan}(v)P_{\Gamma})(\pi(x,t),t)|^2 + R(d(x,t)) \} \, dx \\ &= \int_{\Gamma(t)} \int_{-\varepsilon}^{\varepsilon} \{ |(P_{\Gamma}D^{tan}(v)P_{\Gamma})(y,t)|^2 + R(\rho) \} J(y,t,\rho) \, d\rho \, d\mathcal{H}^2(y) \\ &= 2\varepsilon \int_{\Gamma(t)} |(P_{\Gamma}D^{tan}(v)P_{\Gamma})(y,t)|^2 \, d\mathcal{H}^2(y) + O(\varepsilon^2). \end{split}$$

Let us show (4.5.20). By (4.2.5) we have

$$(D(u)\nu_{\varepsilon})(x,t) = \pm (P_{\Gamma}D^{tan}(v)P_{\Gamma}\nu)(\pi(x,t),t) + \varepsilon(S^{1}\nu)(\pi(x,t),t) + O(\varepsilon^{2}).$$

for $x \in \partial \Omega_{\varepsilon}(t)$ according to $d(x,t) = \pm \varepsilon$ (double-sign corresponds). Moreover, the first two terms on the right-hand side vanishes since $P_{\Gamma}\nu = 0$ and $S^{1}\nu = 0$ on $\Gamma(t)$ by (4.4.25). (Note that, similarly to the proof of (4.4.27), only the boundary conditions (4.4.3) and (4.4.4) are necessary to show (4.4.25). See the proof of Theorem 4.4.1.) Hence $D(u)\nu_{\varepsilon} = O(\varepsilon^{2})$ on $\partial \Omega_{\varepsilon}(t)$. Applying this estimate and

$$|\nu_{\varepsilon}(x,t)| = 1, \quad |V_{\varepsilon}^{N}(x,t)| = |V_{\Gamma}^{N}(\pi(x,t),t)| = O(1), \quad x \in \partial \Omega_{\varepsilon}(t),$$

where the second relation follows from (4.2.6) and the fact that V_{Γ}^{N} is independent of ε , to the left-hand side of (4.5.20), and then using the change of variables formula (4.2.21) and $J(y, t, \pm \varepsilon) = O(1)$, we obtain (4.5.20) as

$$\int_{\partial\Omega_{\varepsilon}(t)} [(D(u)\nu_{\varepsilon} \cdot \nu_{\varepsilon})V_{\varepsilon}^{N}](x,t) \, d\mathcal{H}^{2}(x) = \sum_{\rho=\pm\varepsilon} \int_{\Gamma(t)} O(\varepsilon^{2}) J(y,t,\rho) \, d\mathcal{H}^{2}(y) = O(\varepsilon^{2}).$$

Now we substitute (4.5.10), (4.5.12), (4.5.19), and (4.5.20) for the energy identity (4.5.13) and divide both sides by 2ε to obtain

$$\begin{split} \frac{d}{dt} \int_{\Gamma(t)} \frac{|v(y,t)|^2}{2} \, d\mathcal{H}^2(y) &= -2\mu_0 \int_{\Gamma(t)} |(P_{\Gamma}D^{tan}(v)P_{\Gamma})(y,t)|^2 \, d\mathcal{H}^2(y) \\ &+ \int_{\Gamma(t)} (qH-q^1)(y,t) V_{\Gamma}^N(y,t) \, d\mathcal{H}^2(y) + O(\varepsilon). \end{split}$$

Since all terms except for $O(\varepsilon)$ are independent of ε , we conclude that the energy identity (4.5.16) must be satisfied.

Remark 4.5.7. The assumption in Theorem 4.5.6 that the boundary conditions (4.4.3) and (4.4.4) are satisfied is necessary to deal with integrals including the strain rate tensor D(u). Note that, contrary to the case of the Navier–Stokes equations (Theorem 4.5.6), we do not need even the impermeable boundary condition (4.3.3) to derive the thin width limit of the energy identity of the Euler equations in the moving thin domain, see Theorem 4.5.3.

4.A Elementary calculations of various quantities on surfaces

In this appendix we prove elementary facts on various quantities and differential operators on a surface given in Section 4.2. Until the end of the proof of Lemma 4.2.6 we fix and suppress $t \in [0, T]$.

Proof of Lemma 4.2.1. Since $|\nu|^2 = 1$ on Γ , we have

$$0 = \nabla_{\Gamma} |\nu|^2 = 2(\nabla_{\Gamma} \nu)\nu = -2A\nu \quad \text{on} \quad \Gamma,$$

which implies (4.2.7). The formula (4.2.8) is an immediate consequence of (4.2.7). Now let us prove (4.2.9). Let $\tilde{\nu}$ be an extension of ν to N. By (4.2.2) and $\tilde{\nu}|_{\Gamma} = \nu$ we have

$$\nabla^2 d(x) = \nabla \pi(x) (\nabla \tilde{\nu})(\pi(x)), \quad x \in N.$$
(4.A.1)

Moreover, we differentiate both sides of (4.2.1) and apply (4.2.2) to get

$$\nabla \pi(x) = P_{\Gamma}(\pi(x)) - d(x) \nabla \pi(x) (\nabla \tilde{\nu})(\pi(x)), \quad x \in N.$$

In particular, if $x = y \in \Gamma$ then d(x) = 0, $\pi(x) = y$ and thus $\nabla \pi(y) = P_{\Gamma}(y)$. Applying this formula to (4.A.1) with $x = y \in \Gamma$ we obtain (4.2.9).

Proof of Lemma 4.2.2. Let f be a function on Γ and \tilde{f} its extension to N satisfying $\tilde{f}|_{\Gamma} = f$. For j = 1, 2, 3, by (4.2.2) and the definition of the tangential derivative operators we have

$$\partial_j^{tan} f(y) = \sum_{l=1}^3 \{ \delta_{jl} - \partial_j d(y) \partial_l d(y) \} \partial_l \tilde{f}(y), \quad y \in \Gamma.$$

From now on we suppress the argument y. By the above formula we have

$$\partial_i^{tan} \partial_j^{tan} f = \sum_{k,l=1}^3 \{ \delta_{ik} - (\partial_i d)(\partial_k d) \} \partial_k \Big[\{ \delta_{jl} - (\partial_j d)(\partial_l d) \} \partial_l \tilde{f} \Big] = \alpha_1 + \alpha_2 + \alpha_3$$

for i, j = 1, 2, 3, where

$$\begin{aligned} \alpha_1 &:= \sum_{k,l=1}^3 \{\delta_{ik} - (\partial_i d)(\partial_k d)\} \{\delta_{jl} - (\partial_j d)(\partial_l d)\} \partial_k \partial_l \tilde{f}, \\ \alpha_2 &:= -\sum_{k,l=1}^3 \{\delta_{ik} - (\partial_i d)(\partial_k d)\} (\partial_k \partial_j d)(\partial_l d) \partial_l \tilde{f}, \\ \alpha_3 &:= -\sum_{k,l=1}^3 \{\delta_{ik} - (\partial_i d)(\partial_k d)\} (\partial_j d)(\partial_k \partial_l d) \partial_l \tilde{f}. \end{aligned}$$

Similarly, we have $\partial_j^{tan} \partial_i^{tan} f = \beta_1 + \beta_2 + \beta_3$, where

$$\beta_{1} := \sum_{k,l=1}^{3} \{\delta_{jl} - (\partial_{j}d)(\partial_{l}d)\} \{\delta_{ik} - (\partial_{i}d)(\partial_{k}d)\} \partial_{l}\partial_{k}\tilde{f},$$

$$\beta_{2} := -\sum_{k,l=1}^{3} \{\delta_{jl} - (\partial_{j}d)(\partial_{l}d)\} (\partial_{l}\partial_{i}d)(\partial_{k}d)\partial_{k}\tilde{f},$$

$$\beta_{3} := -\sum_{k,l=1}^{3} \{\delta_{jl} - (\partial_{j}d)(\partial_{l}d)\} (\partial_{i}d)(\partial_{l}\partial_{k}d)\partial_{k}\tilde{f}.$$

From $\partial_k \partial_l \tilde{f} = \partial_l \partial_k \tilde{f}$ it immediately follows that $\alpha_1 = \beta_1$. Since $\partial_k \partial_j d = \partial_j \partial_k d$,

$$\begin{aligned} \alpha_2 &= -(\nabla d \cdot \nabla \tilde{f}) \left\{ \partial_i \partial_j d - (\partial_i d) \sum_{k=1}^3 (\partial_k d) (\partial_j \partial_k d) \right\} \\ &= -(\nabla d \cdot \nabla \tilde{f}) \left\{ \partial_i \partial_j d - (\partial_i d) \partial_j \left(\frac{|\nabla d|^2}{2} \right) \right\} \\ &= -(\nabla d \cdot \nabla \tilde{f}) \partial_i \partial_j d. \end{aligned}$$

Here the last equality follows from $|\nabla d|^2 = 1$ on N. By the same calculation we have $\beta_2 = -(\nabla d \cdot \nabla \tilde{f})\partial_j\partial_i d$. Hence $\alpha_2 = \beta_2$ by $\partial_i \partial_j d = \partial_j \partial_i d$. For α_3 and β_3 ,

$$\begin{aligned} \alpha_3 &= -\left[P_{\Gamma}(\nabla^2 d)\nabla \tilde{f}\right]_i \partial_j d = [A\nabla_{\Gamma} f]_i \nu_j, \\ \beta_3 &= -\left[P_{\Gamma}(\nabla^2 d)\nabla \tilde{f}\right]_j \partial_i d = [A\nabla_{\Gamma} f]_j \nu_i \end{aligned}$$

by (4.2.2), (4.2.8), (4.2.9), and the definition of the tangential gradient operator. (Note that we calculate values of functions at $y \in \Gamma$.) Therefore, we obtain

$$\partial_i^{tan}\partial_j^{tan}f - \partial_j^{tan}\partial_i^{tan}f = (\alpha_1 + \alpha_2 + \alpha_3) - (\beta_1 + \beta_2 + \beta_3) = [A\nabla_{\Gamma}f]_i\nu_j - [A\nabla_{\Gamma}f]_j\nu_i,$$

that is, the formula (4.2.11) holds.

Proof of Lemma 4.2.3. Let v be a general vector field on Γ which may have a nonzero normal component. Since $P_{\Gamma}\nabla_{\Gamma}v = \nabla_{\Gamma}v$ and $(\nabla_{\Gamma}v)^{T}P_{\Gamma} = (\nabla_{\Gamma}v)^{T}$ we have

$$2\operatorname{div}_{\Gamma}(P_{\Gamma}D^{tan}(v)P_{\Gamma}) = \operatorname{div}_{\Gamma}((\nabla_{\Gamma}v)P_{\Gamma}) + \operatorname{div}_{\Gamma}(P_{\Gamma}(\nabla_{\Gamma}v)^{T}).$$
(4.A.2)

Let us calculate each term on the right-hand side. For i, j = 1, 2, 3 the (i, j)-entry of $(\nabla_{\Gamma} v)P_{\Gamma}$ is of the form

$$\left[(\nabla_{\Gamma} v) P_{\Gamma} \right]_{ij} = \sum_{k=1}^{3} (\partial_i^{tan} v_k) (\delta_{jk} - \nu_j \nu_k).$$

Thus the *j*-th component of $\operatorname{div}_{\Gamma}((\nabla_{\Gamma} v)P_{\Gamma})$ is

$$\left[\operatorname{div}_{\Gamma}\left((\nabla_{\Gamma}v)P_{\Gamma}\right)\right]_{j} = \sum_{i=1}^{3} \partial_{i}^{tan} \left[(\nabla_{\Gamma}v)P_{\Gamma}\right]_{ij} = \alpha_{1} + \alpha_{2} + \alpha_{3}$$

where

$$\begin{aligned} \alpha_1 &:= \sum_{i,k=1}^3 \{ (\partial_i^{tan})^2 v_k \} (\delta_{jk} - \nu_j \nu_k), \\ \alpha_2 &:= -\sum_{i,k=1}^3 (\partial_i^{tan} v_k) (\partial_i^{tan} \nu_j) \nu_k = \sum_{i,k=1}^3 (\partial_i^{tan} v_k) A_{ij} \nu_k, \\ \alpha_3 &:= -\sum_{i,k=1}^3 (\partial_i^{tan} v_k) \nu_j (\partial_i^{tan} \nu_k) = \sum_{i,k=1}^3 (\partial_i^{tan} v_k) \nu_j A_{ik}. \end{aligned}$$

Here A_{ij} is the (i, j)-entry of the Weingarten map $A = -\nabla_{\Gamma}\nu$. By the definitions of Δ_{Γ} and P_{Γ} we have $\alpha_1 = [P_{\Gamma}(\Delta_{\Gamma}v)]_j$, where Δ_{Γ} applies each component of the vector field v. Also, since A is symmetric,

$$\alpha_2 = \sum_{i,k=1}^3 A_{ji} (\partial_i^{tan} v_k) \nu_k = \left[A(\nabla_{\Gamma} v) \nu \right]_j.$$

Similarly, we have $\alpha_3 = tr[A\nabla_{\Gamma} v]\nu_j$. Therefore, the equality

$$\left[\operatorname{div}_{\Gamma}\left((\nabla_{\Gamma}v)P_{\Gamma}\right)\right]_{j} = \left[P_{\Gamma}(\Delta_{\Gamma}v)\right]_{j} + \left[A(\nabla_{\Gamma}v)\nu\right]_{j} + \operatorname{tr}\left[A\nabla_{\Gamma}v\right]\nu_{j}$$

holds for each j = 1, 2, 3, which means that

$$\operatorname{div}_{\Gamma}((\nabla_{\Gamma}v)P_{\Gamma}) = P_{\Gamma}(\Delta_{\Gamma}v) + A(\nabla_{\Gamma}v)\nu + \operatorname{tr}[A\nabla_{\Gamma}v]\nu.$$
(4.A.3)

Calculations of the second term $\operatorname{div}_{\Gamma}(P_{\Gamma}(\nabla_{\Gamma}v)^T)$ are more complicated. Since

$$\left[P_{\Gamma}(\nabla_{\Gamma}v)^{T}\right]_{ij} = \sum_{k=1}^{3} (\delta_{ik} - \nu_{i}\nu_{k})\partial_{j}^{tan}v_{k},$$

we have
$$\left[\operatorname{div}_{\Gamma}(\mathcal{P}_{\Gamma}(\nabla_{\Gamma}v)^{T})\right]_{j} = \beta_{1} + \beta_{2} + \beta_{3}$$
, where

$$\beta_{1} := -\sum_{i,k=1}^{3} (\partial_{i}^{tan}\nu_{i})\nu_{k}\partial_{j}^{tan}v_{k} = \sum_{i,k=1}^{3} A_{ii}\nu_{k}\partial_{j}^{tan}v_{k},$$

$$\beta_{2} := -\sum_{i,k=1}^{3} \nu_{i}(\partial_{i}^{tan}\nu_{k})\partial_{j}^{tan}v_{k} = \sum_{i,k=1}^{3} \nu_{i}A_{ik}\partial_{j}^{tan}v_{k},$$

$$\beta_{3} := \sum_{i,k=1}^{3} (\delta_{ik} - \nu_{i}\nu_{k})\partial_{i}^{tan}\partial_{j}^{tan}v_{k}.$$

By the definition of the mean curvature,

$$\beta_1 = \operatorname{tr}[A] \sum_{k=1}^3 (\partial_j^{tan} v_k) \nu_k = H \big[(\nabla_{\Gamma} v) \nu \big]_j.$$

Since $A_{ik} = A_{ki}$ and $A\nu = 0$,

$$\beta_2 = \sum_{i,k=1}^3 (\partial_j^{tan} v_k) A_{ki} \nu_i = -\left[(\nabla_{\Gamma} v) A \nu \right]_j = 0.$$

For β_3 we have

$$\beta_3 = \sum_{i=1}^3 \partial_i^{tan} \partial_j^{tan} v_i - \sum_{k=1}^3 \nu_k \{ \nu \cdot \nabla_{\Gamma} (\partial_j^{tan} v_k) \}.$$

The second term on the right-hand side vanishes since $\nu \cdot \nabla_{\Gamma}(\partial_j^{tan}v_k) = 0$ for each j and k. We apply (4.2.11) to the first term to get

$$\beta_3 = \sum_{i=1}^3 \partial_j^{tan} \partial_i^{tan} v_i + \sum_{i=1}^3 [A \nabla_{\Gamma} v_i]_i \nu_j - \sum_{i=1}^3 [A \nabla_{\Gamma} v_i]_j \nu_i$$
$$= \partial_j^{tan} (\operatorname{div}_{\Gamma} v) + \operatorname{tr} [A \nabla_{\Gamma} v] \nu_j - [A (\nabla_{\Gamma} v) \nu]_j.$$

Therefore, it follows that

$$\left[\operatorname{div}_{\Gamma}\left(P_{\Gamma}(\nabla_{\Gamma}v)^{T}\right)\right]_{j} = \partial_{j}^{tan}(\operatorname{div}_{\Gamma}v) + \left[(HI_{3} - A)(\nabla_{\Gamma}v)\nu\right]_{j} + \operatorname{tr}[A\nabla_{\Gamma}v]\nu_{j}$$

for each j = 1, 2, 3 and thus

$$\operatorname{div}_{\Gamma}(P_{\Gamma}(\nabla_{\Gamma}v)^{T}) = \nabla_{\Gamma}(\operatorname{div}_{\Gamma}v) + (HI_{3} - A)(\nabla_{\Gamma}v)\nu + \operatorname{tr}[A\nabla_{\Gamma}v]\nu.$$
(4.A.4)

Substituting (4.A.3) and (4.A.4) for (4.A.2) we obtain the formula (4.2.13). \Box

Proof of Lemma 4.2.6. For $\rho \in [-\varepsilon, \varepsilon]$ let $\Gamma_{\rho} := \{x \in \mathbb{R}^3 \mid d(x) = \rho\}$ be a level-set surface of Γ . Suppose that the change of variables formula

$$\int_{\Gamma_{\rho}} f(z) \, d\mathcal{H}^2(z) = \int_{\Gamma} f(y + \rho\nu(y)) J(y,\rho) \, d\mathcal{H}^2(y) \tag{4.A.5}$$
holds for each $\rho \in [-\varepsilon, \varepsilon]$. Then (4.2.20) and (4.2.21) follow from this formula and

$$\int_{\Omega_{\varepsilon}} f(x) \, dx = \int_{-\varepsilon}^{\varepsilon} \left(\int_{\Gamma_{\rho}} f(z) \, d\mathcal{H}^2(z) \right) d\rho,$$

which is the well-known co-area formula (see e.g. [11, Theorem 2.9]), and

$$\int_{\partial\Omega_{\varepsilon}} f(x) \, d\mathcal{H}^2(x) = \int_{\Gamma_{\varepsilon}} f(z) \, d\mathcal{H}^2(z) + \int_{\Gamma_{-\varepsilon}} f(z) \, d\mathcal{H}^2(z).$$

Let us prove (4.A.5). Since Γ is compact, we may take finitely many open subsets U_k of \mathbb{R}^2 and local parametrizations $\mu_k \colon U_k \to \Gamma$ $(k = 1, \ldots, N)$ such that $\{\mu_k(U_k)\}_{k=1}^N$ is an open covering of Γ . Let $\{\varphi_k\}_{k=1}^N$ be a partition of unity of Γ subordinate to the covering $\{\mu_k(U_k)\}_{k=1}^N$ and for each $\rho \in [-\varepsilon, \varepsilon]$ and $k = 1, \ldots, N$ set

$$\mu_k^{\rho}(s) := \mu_k(s) + \rho \nu(\mu_k(s)), \quad \varphi_k^{\rho}(\mu_k^{\rho}(s)) := \varphi_k(\mu_k(s)), \quad s \in U_k.$$

Then $\mu_k^{\rho}: U_k \to \Gamma_{\rho}$ is a local parametrization of Γ_{ρ} whose domain is the same as that of μ_k and $\{\mu_k^{\rho}(U_k)\}_{k=1}^N$ is an open covering of Γ_{ρ} . Moreover, $\{\varphi_k^{\rho}\}_{k=1}$ is a partition of unity of Γ_{ρ} subordinate to the covering $\{\mu_k^{\rho}(U_k)\}_{k=1}^N$. By these partitions of unity and the definition of integrals over a surface, the proof of (4.A.5) reduces to showing that, for any local parametrization $\mu: U \to \Gamma$ with an open subset U of \mathbb{R}^2 and $\mu^{\rho}: U \to \Gamma_{\rho}$ given by $\mu^{\rho}(s) := \mu(s) + \rho \nu(\mu(s)), s \in U$, the formula

$$\sqrt{\det \theta^{\rho}(s)} = J(\mu(s), \rho)\sqrt{\det \theta(s)}, \quad s \in U$$
(4.A.6)

holds. Here θ is a square matrix of order two given by $\theta := \nabla' \mu (\nabla' \mu)^T$, where

$$\nabla' \mu := \begin{pmatrix} \partial'_1 \mu_1 & \partial'_1 \mu_2 & \partial'_1 \mu_3 \\ \partial'_2 \mu_1 & \partial'_2 \mu_2 & \partial'_2 \mu_3 \end{pmatrix} \quad \left(\partial'_i := \frac{\partial}{\partial s_i} \right),$$

and $\theta^{\rho} := \nabla' \mu^{\rho} (\nabla' \mu^{\rho})^T$. We define square matrices M and M^{ρ} of order three as

$$M(s) := \begin{pmatrix} \nabla' \mu(s) \\ [\nu(\mu(s))]^T \end{pmatrix}, \quad M^{\rho}(s) := \begin{pmatrix} \nabla' \mu^{\rho}(s) \\ [\nu(\mu(s))]^T \end{pmatrix}.$$

Here we see $\nu(\mu(s))$ as a three-dimensional column vector. In the following argument, we sometimes suppress the argument s and abbreviate $\nu(\mu(s))$ to ν . For i = 1, 2 the *i*-th component of $\nabla'\mu(s)\nu(\mu(s)) \in \mathbb{R}^2$ is $\partial_i\mu(s) \cdot \nu(\mu(s)) = 0$ since $\partial_i\mu(s)$ is tangent to Γ at $\mu(s)$. Therefore, $(\nabla'\mu)\nu = 0$ and

$$MM^{T} = \begin{pmatrix} \nabla' \mu (\nabla' \mu)^{T} & (\nabla' \mu)\nu \\ [(\nabla' \mu)\nu]^{T} & |\nu|^{2} \end{pmatrix} = \begin{pmatrix} \theta & 0 \\ 0 & 1 \end{pmatrix},$$

which implies det $\theta = \det(MM^T) = (\det M)^2$. On the other hand, since

$$\mu^{\rho}(s) = \mu(s) + \rho\nu(\mu(s)) = \mu(s) + \rho\nabla d(\mu(s))$$

by (4.2.2) and thus

$$\nabla'\mu^{\rho}(s) = \nabla'\mu(s)\{I_3 + \rho\nabla^2 d(\mu(s))\} = \nabla'\mu(s)\{I_3 - \rho A(\mu(s))\}$$

by (4.2.9), we have $\nabla' \mu^{\rho}(s)\nu(\mu(s)) = 0$ by $\nabla' \mu(s)\nu(\mu(s)) = 0$ and (4.2.7). Hence as in the case of θ and M we have det $\theta^{\rho} = (\det M^{\rho})^2$. Moreover, by (4.2.7) and the symmetry of the matrix $I_3 - \rho A$,

$$M^{\rho} = \begin{pmatrix} (\nabla'\mu)(I_3 - \rho A) \\ \nu^T \end{pmatrix} = \begin{pmatrix} \nabla'\mu \\ \nu^T \end{pmatrix} (I_3 - \rho A) = M(I_3 - \rho A).$$

Hence we get

$$\det \theta^{\rho} = (\det M^{\rho})^{2} = \{\det M \cdot \det(I_{3} - \rho A)\}^{2} = \{\det(I_{3} - \rho A)\}^{2} \det \theta.$$

Finally we observe that the Weingarten map A has the eigenvalues 0, κ_1 , and κ_2 and thus

$$\det\{I_3 - \rho A(\mu(s))\} = 1 \cdot \{1 - \rho \kappa_1(\mu(s))\} \cdot \{1 - \rho \kappa_2(\mu(s))\}$$
$$= J(\mu(s), \rho) \quad (>0 \text{ for sufficiently small } \rho)$$

to obtain the formula (4.A.6).

Now let us return to the moving surface $\Gamma(t)$ and prove Lemmas 4.2.7 and 4.2.8.

Proof of Lemma 4.2.7. As in the proof of Theorem 4.3.1 we use the abbreviations (4.3.13). Let f be a function on S_T and \tilde{f} an arbitrary extension of f to N_T satisfying $\tilde{f}|_{S_T} = f$. For $(x,t) \in Q_{\varepsilon,T}$ we have $f(\pi,t) = \tilde{f}(\pi,t)$ by $\pi = \pi(x,t) \in \Gamma(t)$ and thus

$$\nabla (f(\pi,t)) = \nabla \pi(x,t) \nabla \tilde{f}(\pi,t),$$

$$\partial_t (f(\pi,t)) = \partial_t \tilde{f}(\pi,t) + (\partial_t \pi(x,t) \cdot \nabla) \tilde{f}(\pi,t).$$

Hence it is sufficient for (4.2.25) and (4.2.26) to show that

$$\nabla \pi(x,t) = P_{\Gamma}(\pi,t) + d(x,t)A(\pi,t) + R(d^2), \qquad (4.A.7)$$

$$\partial_t \pi(x,t) = V_{\Gamma}^N(\pi,t)\nu(\pi,t) + d(x,t)\nabla_{\Gamma}V_{\Gamma}^N(\pi,t) + R(d), \qquad (4.A.8)$$

since

$$\begin{split} A\nabla \tilde{f} &= AP_{\Gamma}\nabla \tilde{f} = A\nabla_{\Gamma}f,\\ \partial_t \tilde{f} &+ (V_{\Gamma}^N \nu \cdot \nabla)\tilde{f} = \partial^\circ f, \quad (\nabla_{\Gamma}V_{\Gamma}^N \cdot \nabla)\tilde{f} = (\nabla_{\Gamma}V_{\Gamma}^N \cdot \nabla_{\Gamma})f \end{split}$$

on $\Gamma(t)$ by the definition of the tangential gradient, (4.2.8), and (4.2.2) with $v = V_{\Gamma}^{N} \nu$. By $\pi(x,t) = x - d(x,t) \nabla d(x,t)$ and (4.2.2) we have

$$\nabla \pi(x,t) = I_3 - \nabla d(x,t) \otimes \nabla d(x,t) - d(x,t) \nabla^2 d(x,t)$$
$$= P_{\Gamma}(\pi,t) - d(x,t) \nabla^2 d(x,t).$$

Also, we expand $\nabla^2 d$ in powers of d and apply (4.2.9) to obtain

$$\nabla^2 d(x,t) = \nabla^2 d(\pi,t) + R(d) = -A(\pi,t) + R(d).$$

Hence (4.A.7) follows. Similarly, we differentiate $\pi(x,t) = x - d(x,t)\nabla d(x,t)$ with respect to t and apply (4.2.2) and (4.2.4) to get

$$\partial_t \pi(x,t) = V_{\Gamma}^N(\pi,t)\nu(\pi,t) - d(x,t)\partial_t \nabla d(x,t).$$

Moreover, by $\partial_t \nabla d = \nabla \partial_t d$, (4.2.4), and (4.A.7),

$$\partial_t \nabla d(x,t) = -\nabla \left(V_{\Gamma}^N(\pi,t) \right) = -\nabla \pi(x,t) \nabla \widetilde{V}_{\Gamma}^N(\pi,t) = -\nabla_{\Gamma} V_{\Gamma}^N(\pi,t) + R(d),$$

where $\widetilde{V}_{\Gamma}^{N}$ is an extension of V_{Γ}^{N} to N_{T} with $\widetilde{V}_{\Gamma}^{N}|_{S_{T}} = V_{\Gamma}^{N}$. Applying this to the above equality for $\partial_{t}\pi$ we obtain (4.A.8).

Proof of Lemma 4.2.8. We use the abbreviations (4.3.13). For i, j = 1, 2, 3, let M_{ij} be the (i, j)-entry of a square matrix M of order three. We differentiate both sides of $D_{ij}(x) = S_{ij}(\pi) + d(x, t)S_{ij}^1(\pi) + R(d^2)$ with respect to x_i and apply (4.A.7) to get

$$\partial_i D_{ij}(x) = \partial_i^{tan} S_{ij}(\pi) + S_{ij}^1(\pi) \partial_i d(x, t) + R(d)$$

Therefore, the *j*-th component of $\operatorname{div} D(x)$ is

$$[\operatorname{div} D(x)]_{j} = \sum_{i=1}^{3} \partial_{i} D_{ij}(x) = \sum_{i=1}^{3} \{\partial_{i}^{tan} S_{ij}(\pi) + S_{ij}^{1}(\pi) \partial_{i} d(x,t)\} + R(d(x,t))$$
$$= [\operatorname{div}_{\Gamma} S(\pi)]_{j} + \left[\left(S^{1}(\pi) \right)^{T} \nabla d(x,t) \right]_{j} + R(d)$$

and (4.2.27) follows by (4.2.2).

4.B Comparison of vector Laplacians

The purpose of this appendix is to give a proof of the formula (4.2.15) in Lemma 4.2.4. Main tools for the proof are the Gauss formula (4.2.14) and

$$\Delta_B X = \text{tr}\overline{\nabla}^2 X = \sum_{i=1}^2 \left(\overline{\nabla}_i \overline{\nabla}_i X - \overline{\nabla}_{\overline{\nabla}_i e_i} X\right) \quad \text{on} \quad \Gamma \tag{4.B.1}$$

for any tangential vector field X on Γ , where $\{e_1, e_2\}$ denotes a local orthonormal frame of $T\Gamma$ (i.e. an orthonormal basis of the tangent plane of Γ defined on a relative open subset of Γ) and $\overline{\nabla}_i := \overline{\nabla}_{e_i}$ (for a proof of (4.B.1) see [26, Proposition 34] and [32, Proposition 2.1 in Appendix C]). Hereafter all calculations are carried out on the surface Γ .

We fix coordinates of \mathbb{R}^3 and write x_j (j = 1, 2, 3) for the *j*-th component of a point $x \in \mathbb{R}^3$ under this fixed coordinates. Let $X = (X_1, X_2, X_3)$ be a tangential vector field on Γ and $\{e_1, e_2\}$ be a local orthonormal frame of $T\Gamma$. For i = 1, 2, by the Gauss formula (4.2.16) and the fact that $\overline{\nabla}_i X$ is tangential we have

$$\overline{\nabla}_i X = (e_i \cdot \nabla_{\Gamma}) X - (AX \cdot e_i) \nu = P_{\Gamma} \{ (e_i \cdot \nabla_{\Gamma}) X \}.$$

Here the second equality follows from $P_{\Gamma}\nu = 0$. Hence

$$\overline{\nabla}_i \overline{\nabla}_i X = P_{\Gamma} \big[(e_i \cdot \nabla_{\Gamma}) \{ (e_i \cdot \nabla_{\Gamma}) X - (AX \cdot e_i) \nu \} \big] = P_{\Gamma} \big[(e_i \cdot \nabla_{\Gamma}) \{ (e_i \cdot \nabla_{\Gamma}) X \} \big] - (AX \cdot e_i) P_{\Gamma} \{ (e_i \cdot \nabla_{\Gamma}) \nu \},$$

where we used $P_{\Gamma}\nu = 0$ again in the second equality. By setting $e_i = (e_i^1, e_i^2, e_i^3)$ the *j*-th component of the vector $(e_i \cdot \nabla_{\Gamma})\{(e_i \cdot \nabla_{\Gamma})X\}$ (j = 1, 2, 3) is of the form

$$\sum_{k,l=1}^{3} e_i^k \partial_k^{tan} (e_i^l \partial_l^{tan} X_j) = \sum_{k,l=1}^{3} \{ e_i^k e_i^l \partial_k^{tan} \partial_l^{tan} X_j + e_i^k (\partial_k^{tan} e_i^l) \partial_l^{tan} X_j \}$$
$$= \operatorname{tr} \left[(e_i \otimes e_i) \nabla_{\Gamma}^2 X_j \right] + \{ (e_i \cdot \nabla_{\Gamma}) e_i \} \cdot \nabla_{\Gamma} X_j.$$

Also, by the symmetry of the Weingarten map $A = -\nabla_{\Gamma} \nu$,

$$[(e_i \cdot \nabla_{\Gamma})\nu]_j = \sum_{k=1}^3 e_i^k \partial_k^{tan} \nu_j = -\sum_{k=1}^3 e_i^k A_{kj} = -[Ae_i]_j.$$

By these equalities and (4.2.8) the *j*-th component of $\overline{\nabla}_i \overline{\nabla}_i X$ is

$$\left[\overline{\nabla}_{i}\overline{\nabla}_{i}X\right]_{j} = \sum_{k=1}^{3} [P_{\Gamma}]_{jk} \left(\operatorname{tr}\left[(e_{i} \otimes e_{i})\nabla_{\Gamma}^{2}X_{k} \right] + \{ (e_{i} \cdot \nabla_{\Gamma})e_{i} \} \cdot \nabla_{\Gamma}X_{k} \right) + (AX \cdot e_{i})[Ae_{i}]_{j}. \quad (4.B.2)$$

On the other hand, $\overline{\nabla}_{\overline{\nabla}_i e_i} X$ is of the form

$$\overline{\nabla}_{\overline{\nabla}_i e_i} X = P_{\Gamma} \Big\{ \Big(\overline{\nabla}_i e_i \cdot \nabla_{\Gamma} \Big) X \Big\} = P_{\Gamma} \Big(\big[\{ P_{\Gamma}(e_i \cdot \nabla_{\Gamma}) e_i \} \cdot \nabla_{\Gamma} \big] X \Big)$$

and, since $\{(P_{\Gamma}F) \cdot \nabla_{\Gamma}\}G = (F \cdot \nabla_{\Gamma})G$ holds for (not necessarily tangential) vector fields Fand G on Γ we have

$$\left[\overline{\nabla}_{\overline{\nabla}_i e_i} X\right]_j = \sum_{k=1}^3 [P_{\Gamma}]_{jk} \Big(\{ (e_i \cdot \nabla_{\Gamma}) e_i \} \cdot \nabla_{\Gamma} X_k \Big).$$
(4.B.3)

Applying (4.B.2) and (4.B.3) to (4.B.1) we get

$$[\Delta_B X]_j = \sum_{i=1}^2 \left(\sum_{k=1}^3 [P_\Gamma]_{jk} \operatorname{tr} \left[(e_i \otimes e_i) \nabla_\Gamma^2 X_k \right] + (AX \cdot e_i) [Ae_i]_j \right).$$

Furthermore, since e_1 and e_2 form an orthonormal basis of the tangent plane of Γ it follows that $\sum_{i=1}^{2} (e_i \otimes e_i) = P_{\Gamma}$ and thus

$$\sum_{i=1}^{2} \operatorname{tr} \left[(e_i \otimes e_i) \nabla_{\Gamma}^2 X_k \right] = \operatorname{tr} \left[P_{\Gamma} \nabla_{\Gamma}^2 X_k \right] = \operatorname{tr} \left[\nabla_{\Gamma}^2 X_k \right] = \Delta_{\Gamma} X_k$$

for each k = 1, 2, 3 by $P_{\Gamma} \nabla_{\Gamma} = \nabla_{\Gamma}$, and

$$\sum_{i=1}^{2} (AX \cdot e_i) Ae_i = \sum_{i=1}^{2} A(e_i \otimes e_i) AX = AP_{\Gamma}AX = A^2X,$$

by $(AX \cdot e_i)Ae_i = (Ae_i \otimes e_i)AX = A(e_i \otimes e_i)AX$ and (4.2.8). Therefore,

$$[\Delta_B X]_j = \sum_{k=1}^3 [P_{\Gamma}]_{jk} \Delta_{\Gamma} X_k + [A^2 X]_j = [P_{\Gamma} \Delta_{\Gamma} X]_j + [A^2 X]_j$$

for each j = 1, 2, 3, which yields the formula (4.2.15).

References

- [1] R. Aris, Vectors, tensors, and the basic equations of fluid mechanics, Dover Publications, 1989.
- [2] V. Arnold, Sur la géométrie différentielle des groupes de Lie de dimension infinie et ses applications à l'hydrodynamique des fluides parfaits, Ann. Inst. Fourier (Grenoble) 16 (1966), no. fasc. 1, 319–361 (French).
- [3] V. I. Arnol'd, Mathematical methods of classical mechanics, Graduate Texts in Mathematics, vol. 60, Springer-Verlag, New York, 1989. Translated from the 1974 Russian original by K. Vogtmann and A. Weinstein; Corrected reprint of the second (1989) edition.

- [4] M. Arroyo and A. DeSimone, Relaxation dynamics of fluid membranes, Phys. Rev. E (3) 79 (2009), no. 3, 031915, 17.
- [5] J. W. Barrett, H. Garcke, and R. Nürnberg, Stable numerical approximation of two-phase flow with a Boussinesq-Scriven surface fluid, Commun. Math. Sci. 13 (2015), no. 7, 1829–1874.
- [6] D. Bothe and J. Prüss, On the two-phase Navier-Stokes equations with Boussinesq-Scriven surface fluid, J. Math. Fluid Mech. 12 (2010), no. 1, 133–150.
- [7] M. J. Boussinesq, Sur l'existence d'une viscosité superficielle, dans la mince couche de transition séparant un liquide d'un autre fluide contigu, Ann. Chim. Phys. 29 (1913), 349–357.
- [8] P. Cermelli, E. Fried, and M. E. Gurtin, Transport relations for surface integrals arising in the formulation of balance laws for evolving fluid interfaces, J. Fluid Mech. 544 (2005), 339–351.
- [9] B.-Y. Chen, Total mean curvature and submanifolds of finite type, 2nd ed., Series in Pure Mathematics, vol. 27, World Scientific Publishing Co. Pte. Ltd., Hackensack, NJ, 2015. With a foreword by Leopold Verstraelen.
- [10] G. Dziuk and C. M. Elliott, Finite elements on evolving surfaces, IMA J. Numer. Anal. 27 (2007), no. 2, 262–292.
- [11] G. Dziuk and C. M. Elliott, Finite element methods for surface PDEs, Acta Numer. 22 (2013), 289–396.
- [12] D. G. Ebin and J. Marsden, Groups of diffeomorphisms and the motion of an incompressible fluid., Ann. of Math. (2) 92 (1970), 102–163.
- [13] D. Gilbarg and N. S. Trudinger, *Elliptic partial differential equations of second order*, Classics in Mathematics, Springer-Verlag, Berlin, 2001. Reprint of the 1998 edition.
- [14] M. E. Gurtin, An introduction to continuum mechanics, Mathematics in Science and Engineering, vol. 158, Academic Press, Inc. [Harcourt Brace Jovanovich, Publishers], New York-London, 1981.
- [15] L. T. Hoang and G. R. Sell, Navier-Stokes equations with Navier boundary conditions for an oceanic model, J. Dynam. Differential Equations 22 (2010), no. 3, 563–616.
- [16] D. Iftimie, G. Raugel, and G. R. Sell, Navier-Stokes equations in thin 3D domains with Navier boundary conditions, Indiana Univ. Math. J. 56 (2007), no. 3, 1083–1156.
- [17] T. Jankuhn, M. A. Olshanskii, and A. Reusken, Incompressible fluid problems on embedded surfaces: Modeling and variational formulations, arXiv:1702.02989.
- [18] H. Koba, C. Liu, and Y. Giga, Energetic variational approaches for incompressible fluid systems on an evolving surface, Quart. Appl. Math. 75 (2017), no. 2, 359–389.
- [19] S. Kobayashi and K. Nomizu, Foundations of differential geometry. Vol. II, Wiley Classics Library, John Wiley & Sons, Inc., New York, 1996. Reprint of the 1969 original; A Wiley-Interscience Publication.
- [20] J. M. Lee, Introduction to smooth manifolds, 2nd ed., Graduate Texts in Mathematics, vol. 218, Springer, New York, 2013.
- [21] M. Mitrea and M. Taylor, Navier-Stokes equations on Lipschitz domains in Riemannian manifolds, Math. Ann. 321 (2001), no. 4, 955–987.
- [22] T.-H. Miura, Zero width limit of the heat equation on moving thin domains, Interfaces Free Bound. 19 (2017), no. 1, 31–77.
- [23] T.-H. Miura, Y. Giga, and C. Liu, An energetic variational approach for nonlinear diffusion equations in moving thin domains, Adv. Math. Sci. Appl., to appear.
- [24] L. I. Nicolaescu, Lectures on the geometry of manifolds, 2nd ed., World Scientific Publishing Co. Pte. Ltd., Hackensack, NJ, 2007.
- [25] I. Nitschke, A. Voigt, and J. Wensch, A finite element approach to incompressible two-phase flow on manifolds, J. Fluid Mech. 708 (2012), 418–438.
- [26] P. Petersen, *Riemannian geometry*, 2nd ed., Graduate Texts in Mathematics, vol. 171, Springer, New York, 2006.
- [27] M. Prizzi, M. Rinaldi, and K. P. Rybakowski, Curved thin domains and parabolic equations, Studia Math. 151 (2002), no. 2, 109–140.
- [28] G. Raugel, Dynamics of partial differential equations on thin domains, Dynamical systems (Montecatini Terme, 1994), Lecture Notes in Math., vol. 1609, Springer, Berlin, 1995, pp. 208–315.
- [29] G. Raugel and G. R. Sell, Navier-Stokes equations on thin 3D domains. I. Global attractors and global regularity of solutions, J. Amer. Math. Soc. 6 (1993), no. 3, 503–568.

- [30] L. E. Scriven, Dynamics of a fluid interface Equation of motion for Newtonian surface fluids, Chemical Engineering Science 12 (1960), no. 2, 98–108.
- [31] M. E. Taylor, Analysis on Morrey spaces and applications to Navier-Stokes and other evolution equations, Comm. Partial Differential Equations 17 (1992), no. 9-10, 1407–1456.
- [32] M. E. Taylor, Partial differential equations II. Qualitative studies of linear equations, 2nd ed., Applied Mathematical Sciences, vol. 116, Springer, New York, 2011.
- [33] R. Temam and M. Ziane, Navier-Stokes equations in three-dimensional thin domains with various boundary conditions, Adv. Differential Equations 1 (1996), no. 4, 499–546.
- [34] R. Temam and M. Ziane, Navier-Stokes equations in thin spherical domains, Optimization methods in partial differential equations (South Hadley, MA, 1996), Contemp. Math., vol. 209, Amer. Math. Soc., Providence, RI, 1997, pp. 281–314.

Chapter 5

Navier–Stokes equations in a curved thin domain

5.1 Introduction

We consider the three-dimensional Navier–Stokes equations

$$\partial_t u^{\varepsilon} + (u^{\varepsilon} \cdot \nabla) u^{\varepsilon} - \nu \Delta u^{\varepsilon} + \nabla p^{\varepsilon} = f^{\varepsilon}, \quad \text{div} \, u^{\varepsilon} = 0 \quad \text{in} \quad \Omega_{\varepsilon} \times (0, \infty) \tag{5.1.1}$$

imposed with Navier's slip boundary conditions

$$u^{\varepsilon} \cdot n_{\varepsilon} = 0, \quad [\sigma(u^{\varepsilon}, p^{\varepsilon})n_{\varepsilon}]_{tan} + \gamma_{\varepsilon}u^{\varepsilon} = 0 \quad \text{on} \quad \Gamma_{\varepsilon} \times (0, \infty)$$
 (5.1.2)

and initial condition

$$u^{\varepsilon}|_{t=0} = u_0^{\varepsilon}, \quad \text{in} \quad \Omega_{\varepsilon}.$$
 (5.1.3)

Here Ω_{ε} is a curved thin domain in \mathbb{R}^3 of the form

$$\Omega_{\varepsilon} := \{ y + rn(y) \mid y \in \Gamma, r \in (\varepsilon g_0(y), \varepsilon g_1(y)) \}, \quad \varepsilon \in (0, 1),$$
(5.1.4)

where Γ denotes a two-dimensional closed (i.e. compact and without boundary), connected, and oriented surface in \mathbb{R}^3 with unit outward normal vector field n, and g_0 and g_1 are functions on Γ such that $g := g_1 - g_0$ is bounded from below by a positive constant. We denote by Γ_{ε} and n_{ε} the boundary of Ω_{ε} and its unit outward normal vector field. The boundary Γ_{ε} is the union of the inner and outer boundaries Γ_{ε}^0 and Γ_{ε}^1 given by

$$\Gamma^i_{\varepsilon} := \{ y + \varepsilon g_i(y) n(y) \mid y \in \Gamma \}, \quad i = 0, 1.$$

Also, $\nu > 0$ is the viscosity coefficient independent of ε and $\gamma_{\varepsilon} \ge 0$ is the friction coefficient which takes different values on the inner and outer boundaries, i.e.

$$\gamma_{arepsilon} := \gamma^i_{arepsilon} \quad ext{on} \quad \Gamma^i_{arepsilon}, \ i = 0, 1,$$

where γ_{ε}^{0} and γ_{ε}^{1} are nonnegative constants. We further write

$$\sigma(u^{\varepsilon},p^{\varepsilon}):=2\nu D(u^{\varepsilon})-p^{\varepsilon}I_3,\quad [\sigma(u^{\varepsilon},p^{\varepsilon})n_{\varepsilon}]_{\mathrm{tan}}:=P_{\varepsilon}[\sigma(u^{\varepsilon},p^{\varepsilon})n_{\varepsilon}]$$

for the stress tensor and the tangential component of the stress vector on Γ_{ε} , where

$$D(u^{\varepsilon}) := \frac{\nabla u^{\varepsilon} + (\nabla u^{\varepsilon})^T}{2}, \quad P_{\varepsilon} := I_3 - n_{\varepsilon} \otimes n_{\varepsilon}$$

are the strain rate tensor and the orthogonal projection onto the tangent plane of Γ_{ε} . Note that $[\sigma(u^{\varepsilon}, p^{\varepsilon})n_{\varepsilon}]_{tan} = 2\nu P_{\varepsilon}D(u^{\varepsilon})n_{\varepsilon}$ does not depend on p^{ε} and the slip boundary conditions (5.1.2) can be expressed as

$$u^{\varepsilon} \cdot n_{\varepsilon} = 0, \quad 2\nu P_{\varepsilon} D(u^{\varepsilon}) n_{\varepsilon} + \gamma_{\varepsilon} u^{\varepsilon} = 0 \quad \text{on} \quad \Gamma_{\varepsilon} \times (0, \infty).$$

In what follows, we mainly refer to these conditions as the slip boundary conditions.

Partial differential equations in thin domains appear in many applications in solid mechanics (thin elastic bodies), fluid mechanics (lubrication, meteorology, ocean dynamics), etc. They have been studied for a long time since the pioneering works [16, 17] by Hale and Raugel on damped wave and reaction-diffusion equations. In the study of the threedimensional Navier–Stokes equations in thin domains we expect to show the global-in-time existence of a strong solution for large data according to the smallness of the width of thin domains, since a three-dimensional thin domain with very small width can be considered almost two-dimensional. Raugel and Sell [53] first established the global existence of a strong solution to the Navier–Stokes equations with purely periodic or mixed Dirichlet-periodic boundary conditions in the thin product domain $\Omega_{\varepsilon} = Q_2 \times (0, \varepsilon)$ with a rectangle Q_2 and sufficiently small $\varepsilon > 0$. Later, Temam and Ziane [65] generalized the results in [53] in the case of the thin product domain $\Omega_{\varepsilon} = \omega \times (0, \varepsilon)$ with a bounded domain ω in \mathbb{R}^2 and other boundary conditions which are combinations of the Dirichlet, free, and periodic boundary conditions. We refer to [23–25, 44, 45] and the references cited therein for further generalization and improvement on the results in [53, 65].

The above cited papers deal with thin product domains whose boundaries and limit sets are both flat, but there are various kinds of nonflat thin domains in physical problems (see [54] for examples of nonflat thin domains). A nonflat thin domain was first considered by Temem and Ziane [66], who studied the Navier–Stokes equations with free boundary conditions in a thin spherical shell

$$\Omega_{\varepsilon} = \{ x \in \mathbb{R}^3 \mid a < |x| < a + a\varepsilon \}, \quad a > 0$$

to give a mathematical justification of derivation of the primitive equations for the atmosphere and ocean dynamics (see [38–40]). Another generalization of the shape of a thin domain was made by Iftimie, Raugel, and Sell [26], who studied the Navier–Stokes equations in a flat thin domain with a nonflat top boundary

$$\Omega_{\varepsilon} = \{ x = (x', x_3) \in \mathbb{R}^3 \mid x' \in (0, 1)^2, \ 0 < x_3 < \varepsilon g(x') \}, \quad g \colon (0, 1)^2 \to \mathbb{R}$$

with periodic boundary conditions on the lateral boundaries and Navier's slip boundary conditions on the top and bottom boundaries. Their result was extended by Hoang [20] and Hoang and Sell [21] to a flat thin domain both of whose top and bottom boundaries are not flat (see also [22] for the study of two-phase flows in a flat thin domain with nonflat top and bottom boundaries).

The slip boundary conditions were proposed by Navier [47], which state that the fluid slips on the boundary with velocity proportional to the tangential component of the stress vector. They arise in the study of the atmosphere and ocean dynamics [38–40] and the homogenization of the no-slip boundary condition on a rough boundary [18,27]. We observe in Remark 5.1.7 that the slip boundary conditions give a "proper" viscous term in the sense of [12, 62] in surface fluid equations derived as the thin width limit of the Navier–Stokes equations in a curved thin domain.

In this chapter, we establish the global existence of a strong solution to the Navier–Stokes equations (5.1.1)–(5.1.3) in the curved thin domain Ω_{ε} given by (5.1.4) for large data of order

 $\varepsilon^{-1/2}$. Our result is basically the same as those in [20, 21, 26], but an additional assumption is required (see Assumption 2 and Remark 5.1.5).

Another subject of this chapter is a singular limit problem for the Navier-Stokes equations (5.1.1)–(5.1.3) when the curved thin domain Ω_{ε} degenerates into the closed surface Γ as $\varepsilon \to 0$. We are concerned with derivation of limit equations on Γ and comparison of solutions to the bulk and limit equations. There are several works on the asymptotic behavior of eigenvalues of the Laplacian on a curved thin domain around a hypersurface (see e.g. [29, 34,57], but a singular limit problem for evolution equations in curved thin domains has not been studied well. Temam and Ziane [66] first considered this problem for the Navier–Stokes equations in the thin spherical shell and proved the convergence of the average in the thin direction of a solution towards a unique solution to the Navier–Stokes equations on a sphere in \mathbb{R}^3 . Prizzi, Rinaldi, and Rybakowski [51] studied a reaction-diffusion equations in a curved thin domain degenerating into a lower dimensional manifold and compared the dynamics of the original and limit equations (see also [52]). Later, the present author considered the heat equation in a moving thin domain and derived a limit diffusion equation on its limit evolving surface [42]. In the recent work [43], he also formally derived limit equations of the Navier–Stokes equations in a moving thin domain that is a tubular neighborhood of an evolving closed surface. The purpose of this chapter is to give a mathematical justification (and generalization) of the result in [43] in the case of the stationary curved thin domain of the form (5.1.4).

To state our main results let us give several notations and assumptions. Let \mathbb{P}_{ε} be the Helmholtz–Leray projection from $L^2(\Omega_{\varepsilon})^3$ onto the solenoidal space

$$L^2_{\sigma}(\Omega_{\varepsilon}) = \{ u \in L^2(\Omega_{\varepsilon})^3 \mid \text{div}\, u = 0 \text{ in } \Omega_{\varepsilon}, \, u \cdot n_{\varepsilon} = 0 \text{ on } \Omega_{\varepsilon} \}.$$

We denote by A_{ε} the Stokes operator on $L^2_{\sigma}(\Omega_{\varepsilon})$ associated with slip boundary conditions and write $D(A_{\varepsilon})$ for its domain (see Section 5.5.2). With these notations the problem (5.1.1)–(5.1.3) is formulated as an abstract evolution equation

$$\partial_t u^{\varepsilon} + A_{\varepsilon} u^{\varepsilon} + \mathbb{P}_{\varepsilon} (u^{\varepsilon} \cdot \nabla) u^{\varepsilon} = \mathbb{P}_{\varepsilon} f^{\varepsilon}, \quad u^{\varepsilon}|_{t=0} = u_0^{\varepsilon}.$$

We refer to [8, 10, 61, 64] and the references cited therein for the study of this abstract evolution equation. For a function φ on Ω_{ε} we define its average in the thin direction by

$$M\varphi(y) := \frac{1}{\varepsilon g(y)} \int_{\varepsilon g_0(y)}^{\varepsilon g_1(y)} \varphi(y + rn(y)) \, dr, \quad y \in \Gamma.$$

Also, by $M_{\tau}u := PMu$ we denote the averaged tangential component of a vector field u on Ω_{ε} , where $P := I_3 - n \otimes n$ is the orthogonal projection onto the tangent plane of Γ (see Section 5.6.1). For a vector field v on Γ we define the surface strain rate tensor

$$D_{\Gamma}(v) := P\left(\frac{\nabla_{\Gamma}v + (\nabla_{\Gamma}v)^{T}}{2}\right)P_{\tau}$$

where $\nabla_{\Gamma} := P \nabla$ is the tangential gradient operator on Γ , and set

$$\mathcal{K}(\Gamma) := \{ v \in H^1(\Gamma)^3 \mid v \cdot n = 0 \text{ and } D_{\Gamma}(v) = 0 \text{ on } \Gamma \}.$$
(5.1.5)

A vector field $X \in \mathcal{K}(\Gamma)$ satisfies $\overline{\nabla}_Y X \cdot Z + Y \cdot \overline{\nabla}_Z X = 0$ for all tangential vector fields Y and Z on Γ , where $\overline{\nabla}_Y X := P(Y \cdot \nabla_{\Gamma})X$ is the covariant derivative of X along Y (see Appendix 5.C). Such a vector field generates a one-parameter group of isometries of Γ and

is called a Killing vector field. It is also known that $\mathcal{K}(\Gamma)$ is finite dimensional. For details of Killing vector fields, we refer to [30, 50, 63].

In the proofs of our main results we need the uniform boundedness and coerciveness in ε of the bilinear form corresponding to the Stokes operator A_{ε} on $V_{\varepsilon} := L^2_{\sigma}(\Omega_{\varepsilon}) \cap H^1(\Omega_{\varepsilon})^3$ (see Lemma 5.5.4). To establish them we make the following assumptions on the surface and the friction coefficients.

Assumption 1. There exists a constant c > 0 such that

$$\gamma_{\varepsilon}^{i} \le c\varepsilon, \quad i = 0, 1. \tag{5.1.6}$$

Assumption 2. Either of the following conditions is satisfied:

(A1) There exists a constant c > 0 such that

$$\max_{i=0,1} \gamma_{\varepsilon}^{i} \ge c\varepsilon. \tag{5.1.7}$$

(A2) The function space

$$\mathcal{K}_g(\Gamma) := \{ v \in \mathcal{K}(\Gamma) \mid v \cdot \nabla_{\Gamma} g = 0 \text{ on } \Gamma \}$$
(5.1.8)

contains only a trivial vector field, i.e. $\mathcal{K}_g(\Gamma) = \{0\}.$

Assumptions 1 and 2 are imposed in Section 5.5 except for Section 5.5.1 and Sections 5.7, 5.8, and 5.10.

Now let us give the main results of this chapter. The first result is the global-in-time existence of a strong solution for large data.

Theorem 5.1.1. Let Ω_{ε} be the curved thin domain given by (5.1.4). Suppose that

- the closed surface Γ is of class C^5 ,
- $g_0, g_1 \in C^4(\Gamma)$ satisfy $g = g_1 g_0 \ge c$ on Γ with some constant c > 0, and
- Assumptions 1 and 2 are satisfied.

Then there exist constants $\varepsilon_0, c_0 \in (0, 1)$ such that the following statement holds: for each $\varepsilon \in (0, \varepsilon_0)$ if given data $u_0^{\varepsilon} \in V_{\varepsilon}$ and $f^{\varepsilon} \in L^{\infty}(0, \infty; L^2(\Omega_{\varepsilon})^3)$ satisfy

$$\begin{aligned} \|A_{\varepsilon}^{1/2} u_0^{\varepsilon}\|_{L^2(\Omega_{\varepsilon})}^2 + \|\mathbb{P}_{\varepsilon} f^{\varepsilon}\|_{L^{\infty}(0,\infty;L^2(\Omega_{\varepsilon}))}^2 \\ + \|M_{\tau} u_0^{\varepsilon}\|_{L^2(\Gamma)}^2 + \|M_{\tau} \mathbb{P}_{\varepsilon} f^{\varepsilon}\|_{L^{\infty}(0,T;H^{-1}(\Gamma,T\Gamma))}^2 \le c_0 \varepsilon^{-1}, \quad (5.1.9) \end{aligned}$$

then there exists a global-in-time strong solution

$$u^{\varepsilon} \in C([0,\infty); V_{\varepsilon}) \cap L^2_{loc}([0,\infty); D(A_{\varepsilon})) \cap H^1_{loc}([0,\infty); L^2_{\sigma}(\Omega_{\varepsilon}))$$

to the Navier-Stokes equations (5.1.1)-(5.1.3).

In (5.1.9) we write $H^{-1}(\Gamma, T\Gamma)$ for the dual space of

$$H^1(\Gamma, T\Gamma) := \{ v \in H^1(\Gamma)^3 \mid v \cdot n = 0 \text{ on } \Gamma \}.$$

Note that $V_{\varepsilon} = D(A_{\varepsilon}^{1/2})$ and the $L^2(\Omega_{\varepsilon})$ -norm of $A_{\varepsilon}^{1/2}u$ for $u \in V_{\varepsilon}$ is uniformly equivalent in ε to the canonical $H^1(\Omega_{\varepsilon})$ -norm of u (see Lemma 5.5.5). We also establish several estimates for a strong solution in terms of ε .

Theorem 5.1.2. Let c_1, c_2, α , and β be positive constants. Then, under the same assumptions as in Theorem 5.1.1, there exists $\varepsilon_1 \in (0,1)$ such that the following statement holds: for $\varepsilon \in (0, \varepsilon_1)$ if $u_0^{\varepsilon} \in V_{\varepsilon}$ and $f^{\varepsilon} \in L^2(0, \infty; L^2(\Omega_{\varepsilon})^3)$ satisfy

$$\|A_{\varepsilon}^{1/2}u_{0}^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon})}^{2} + \|\mathbb{P}_{\varepsilon}f^{\varepsilon}\|_{L^{\infty}(0,\infty;L^{2}(\Omega_{\varepsilon}))}^{2} \leq c_{1}\varepsilon^{-1+\alpha},$$

$$\|M_{\tau}u_{0}^{\varepsilon}\|_{L^{2}(\Gamma)}^{2} + \|M_{\tau}\mathbb{P}_{\varepsilon}f^{\varepsilon}\|_{L^{\infty}(0,\infty;H^{-1}(\Gamma,T\Gamma))}^{2} \leq c_{2}\varepsilon^{-1+\beta},$$

(5.1.10)

then there exists a global-in-time strong solution u^{ε} to (5.1.1)–(5.1.3) satisfying

$$\|u^{\varepsilon}(t)\|_{L^{2}(\Omega_{\varepsilon})}^{2} \leq c(\varepsilon^{1+\alpha} + \varepsilon^{\beta}),$$

$$\int_{0}^{t} \|u^{\varepsilon}(s)\|_{H^{1}(\Omega_{\varepsilon})}^{2} ds \leq c(\varepsilon^{1+\alpha} + \varepsilon^{\beta})(1+t),$$
(5.1.11)

and

$$\|u^{\varepsilon}(t)\|_{H^{1}(\Omega_{\varepsilon})}^{2} \leq c(\varepsilon^{-1+\alpha} + \varepsilon^{-1+\beta}),$$

$$\int_{0}^{t} \|u^{\varepsilon}(s)\|_{H^{2}(\Omega_{\varepsilon})}^{2} ds \leq c(\varepsilon^{-1+\alpha} + \varepsilon^{-1+\beta})(1+t)$$
(5.1.12)

for all $t \ge 0$, where c > 0 is a constant independent of ε , u^{ε} , and t.

The proofs of Theorems 5.1.1 and 5.1.2 are given in Section 5.8.

Next we give results on a singular limit problem for the Navier–Stokes equations (5.1.1)–(5.1.3) as the curved thin domain Ω_{ε} degenerates into the closed surface Γ . We define function spaces of tangential vector fields on Γ

$$L^{2}(\Gamma, T\Gamma) := \{ v \in L^{2}(\Gamma)^{3} \mid v \cdot n = 0 \text{ on } \Gamma \},$$

$$L^{2}_{q\sigma}(\Gamma, T\Gamma) := \{ v \in L^{2}(\Gamma, T\Gamma) \mid \operatorname{div}_{\Gamma}(gv) = 0 \text{ on } \Gamma \},$$

and $V_g := L^2_{g\sigma}(\Gamma, T\Gamma) \cap H^1(\Gamma, T\Gamma)$, where $\operatorname{div}_{\Gamma}$ is the surface divergence operator on Γ (see Sections 5.2 and 5.9).

Theorem 5.1.3. For $\varepsilon \in (0,1)$ let $u_0^{\varepsilon} \in V_{\varepsilon}$ and $f^{\varepsilon} \in L^2(0,\infty; L^2(\Omega_{\varepsilon})^3)$. Under the same assumptions as in Theorem 5.1.1, suppose further that the following conditions are satisfied:

(a) There exist c > 0, $\varepsilon_2 \in (0, 1)$, and $\alpha \in (0, 1)$ such that

$$\|A_{\varepsilon}^{1/2}u_0^{\varepsilon}\|_{L^2(\Omega_{\varepsilon})}^2 + \|\mathbb{P}_{\varepsilon}f^{\varepsilon}\|_{L^{\infty}(0,\infty;L^2(\Omega_{\varepsilon}))}^2 \le c\varepsilon^{-1+\epsilon}$$

for all $\varepsilon \in (0, \varepsilon_2)$.

(b) There exist $v_0 \in L^2(\Gamma, T\Gamma)$ and $f \in L^{\infty}(0, \infty; H^{-1}(\Gamma, T\Gamma))$ such that

$$\lim_{\varepsilon \to 0} M_{\tau} u_0^{\varepsilon} = v_0 \quad weakly \ in \qquad L^2(\Gamma, T\Gamma),$$
$$\lim_{\varepsilon \to 0} M_{\tau} \mathbb{P}_{\varepsilon} f^{\varepsilon} = f \quad weakly \text{-} \star \ in \quad L^{\infty}(0, \infty; H^{-1}(\Gamma, T\Gamma)).$$

(c) For i = 0, 1 there exists $\gamma^i \ge 0$ such that $\lim_{\varepsilon \to 0} \varepsilon^{-1} \gamma^i_{\varepsilon} = \gamma^i$.

Then there exists $\varepsilon_3 \in (0,1)$ such that the Navier–Stokes equations (5.1.1)–(5.1.3) admit a global strong solution u^{ε} for each $\varepsilon \in (0, \varepsilon_3)$ and

$$\lim_{\varepsilon \to 0} M u^{\varepsilon} \cdot n = 0 \quad strongly \ in \quad C([0,\infty); L^2(\Gamma)).$$

Moreover, there exists a vector field

$$v \in C([0,\infty); L^2_{g\sigma}(\Gamma, T\Gamma)) \cap L^2_{loc}([0,\infty); V_g) \cap H^1_{loc}([0,\infty); H^{-1}(\Gamma, T\Gamma))$$

such that, for each T > 0,

$$\begin{split} &\lim_{\varepsilon \to 0} M_{\tau} u^{\varepsilon} = v \quad weakly \ in \quad L^2(0,T;H^1(\Gamma,T\Gamma)), \\ &\lim_{\varepsilon \to 0} \partial_t M_{\tau} u^{\varepsilon} = \partial_t v \quad weakly \ in \quad L^2(0,T;H^{-1}(\Gamma,T\Gamma)), \end{split}$$

and v is a unique weak solution to the equations

$$g\left(\partial_t v + \overline{\nabla}_v v\right) - 2\nu \left\{ P \operatorname{div}_{\Gamma}[g D_{\Gamma}(v)] - \frac{1}{g} (\nabla_{\Gamma} g \otimes \nabla_{\Gamma} g) v \right\} + (\gamma^0 + \gamma^1) v + g \nabla_{\Gamma} q = gf \quad on \quad \Gamma \times (0, \infty) \quad (5.1.13)$$

and

$$\operatorname{div}_{\Gamma}(gv) = 0 \quad on \quad \Gamma \times (0, \infty), \quad v|_{t=0} = v_0 \quad on \quad \Gamma$$
(5.1.14)

with an associated pressure q.

We give the definition of a weak solution to (5.1.13)–(5.1.14) and prove Theorem 5.1.3 in Section 5.10.5 (see also Lemma 5.10.21 for construction of an associated pressure). Here $\overline{\nabla}_v v = P(v \cdot \nabla_{\Gamma})v$ is the covariant derivative of v along itself. Note that we do not divide (5.1.13) by g since they correspond to the weighted Helmholtz–Leray decomposition

$$v = v_g + g \nabla_{\Gamma} q, \quad v \in L^2(\Gamma, T\Gamma), \, v_g \in L^2_{q\sigma}(\Gamma, T\Gamma), \, g \nabla_{\Gamma} q \in L^2_{q\sigma}(\Gamma, T\Gamma)^{\perp},$$

which we derive in Section 5.9.3. We also point out that the weak limit v_0 of $M_{\tau} u_0^{\varepsilon}$ actually belongs to $L^2_{a\sigma}(\Gamma, T\Gamma)$, while $M_{\tau} u_0^{\varepsilon}$ does not so in general (see Lemma 5.10.22).

If the weak and weak- \star convergence of $M_{\tau}u_0^{\varepsilon}$ and $M_{\tau}\mathbb{P}_{\varepsilon}f^{\varepsilon}$ are replaced by the strong convergence, then we get the strong convergence of $M_{\tau}u^{\varepsilon}$.

Theorem 5.1.4. For $\varepsilon \in (0,1)$ let $u_0^{\varepsilon} \in V_{\varepsilon}$ and $f^{\varepsilon} \in L^{\infty}(0,\infty; L^2(\Omega_{\varepsilon})^3)$. Suppose that the assumptions in Theorem 5.1.3 are satisfied with the condition (b) replaced by the following condition:

(b) There exist $v_0 \in L^2(\Gamma, T\Gamma)$ and $f \in L^{\infty}(0, \infty; H^{-1}(\Gamma, T\Gamma))$ such that

$$\lim_{\varepsilon \to 0} M_{\tau} u_0^{\varepsilon} = v_0 \quad strongly \ in \quad L^2(\Gamma, T\Gamma),$$
$$\lim_{\varepsilon \to 0} M_{\tau} \mathbb{P}_{\varepsilon} f^{\varepsilon} = f \quad strongly \ in \quad L^{\infty}(0, \infty; H^{-1}(\Gamma, T\Gamma))$$

Then the statements in Theorem 5.1.3 hold. Moreover, for each T > 0 we have

$$\lim_{\varepsilon \to 0} M_{\tau} u^{\varepsilon} = v \quad strongly \ in \quad C([0,T]; L^2(\Gamma, T\Gamma)) \cap L^2(0,T; H^1(\Gamma, T\Gamma)).$$

We give an estimate for the difference between $M_{\tau}u^{\varepsilon}$ and v (see Theorem 5.10.23) and prove Theorem 5.1.4 in Section 5.10.6. Moreover, we estimate the difference between a strong solution u^{ε} to (5.1.1)–(5.1.3) and a weak solution v to (5.1.13)–(5.1.14) in Ω_{ε} (see Theorem 5.10.25). It is worth noting that the normal derivative (with respect to Γ) of u^{ε} is compared with a surface vector field of the form

$$-W(y)v(y,t) + \frac{1}{g(y)} \{ v(y,t) \cdot \nabla_{\Gamma} g(y) \} n(y), \quad (y,t) \in \Gamma \times (0,T).$$

where $W = -\nabla_{\Gamma} n$ is the Weingarten map (or shape operator) of Γ (see Theorem 5.10.26). Therefore, the normal derivative of u^{ε} is not necessarily small even though the curved thin domain Ω_{ε} and the limit surface Γ are stationary.

Let us explain the idea of the proofs of our main results. In the proof of the global existence of a strong solution (Theorem 5.1.1) we follow the arguments in [20,21] to show that the H^1 -norm of a strong solution is bounded uniformly in time by a standard energy method. By the same arguments we also get uniform estimates for a strong solution (Theorem 5.1.2). The main tools for the proof are an extension of a surface vector field to Ω_{ε} that satisfies the impermeable boundary condition (the first equation in (5.1.2)) given in Section 5.3.3 and average operators in the thin direction defined and investigated in Section 5.6. Using them, the slip boundary conditions, and Sobolev type inequalities on Ω_{ε} and Γ , we derive a good estimate for the trilinear term

$$((u \cdot \nabla)u, A_{\varepsilon}u)_{L^2(\Omega_{\varepsilon})}, \quad u \in D(A_{\varepsilon})$$

in Section 5.7. A key idea for the proof is to decompose a vector field on Ω_{ε} into the average part, which is almost two-dimensional, and the residual part, which satisfies the impermeable boundary condition (see Section 5.6.3). Such decomposition enables us to apply a product estimate for a function on Ω_{ε} and that on Γ to the average part (see Corollary 5.6.19) and a good L^{∞} -estimate following from the Agmon and Poincaré inequalities to the residual part (see Lemma 5.6.22).

For the proof of the global existence and uniform estimates of a strong solution, we also require the uniform equivalence of the norms

$$c^{-1} \|u\|_{H^1(\Omega_{\varepsilon})} \le \|A_{\varepsilon}^{1/2}u\|_{L^2(\Omega_{\varepsilon})} \le c \|u\|_{H^1(\Omega_{\varepsilon})}$$

for $u \in V_{\varepsilon} = D(A_{\varepsilon}^{1/2})$ with a constant c > 0 independent of ε (see Lemma 5.5.5). It follows from the uniform boundedness and coerciveness of the bilinear form corresponding to the Stokes problem in Ω_{ε} with slip boundary conditions, for which the uniform Korn inequalities on Ω_{ε} established in Section 5.4.1 and Assumptions 1 and 2 are essential (see Lemma 5.5.4). It is more difficult to show the uniform equivalence of the H^2 -norms

$$c^{-1} \|u\|_{H^2(\Omega_{\varepsilon})} \le \|A_{\varepsilon}u\|_{L^2(\Omega_{\varepsilon})} \le c \|u\|_{H^2(\Omega_{\varepsilon})}$$

for $u \in D(A_{\varepsilon})$ (see Lemma 5.5.11). The right-hand inequality follows from a uniform L^2 estimate for the difference between the Stokes and Laplace operators (see Lemma 5.5.8). To prove the left-hand inequality, we derive a uniform a priori estimate for the vector Laplacian with slip boundary conditions, which involves calculations of covariant derivatives on the boundary Γ_{ε} (see Section 5.5.4).

To prove Theorems 5.1.3 and 5.1.4 on a singular limit problem, we proceed as in [42] to transform a weak formulation for the Navier–Stokes equations (5.1.1)-(5.1.3) into that

for the averaged tangential component of a strong solution to (5.1.1)-(5.1.3) with residual terms (see Section 5.10.2). For this purpose, we approximate the bilinear and trilinear forms in the weak formulation for (5.1.1)-(5.1.3) by those in the weak formulation for (5.1.13)-(5.1.14) by using the slip boundary conditions and the average operators (see Section 5.6.4). Moreover, we need to construct an appropriate test function for (5.1.1)-(5.1.3) from a test function for (5.1.13)-(5.1.14), which is a weighted solenoidal vector field on Γ . To this end, we use the impermeable extension of a surface vector field to Ω_{ε} (see Section 5.3.3) and a uniform H^1 -estimate for the gradient part of the Helmholtz–Leray decomposition on Ω_{ε} (see Lemma 5.5.1).

After transformation of the weak formulation, we derive the energy estimate for the averaged tangential component of a strong solution by using its weak formulation in Section 5.10.3. In derivation of the energy estimate, we cannot substitute the averaged tangential component itself for its weak formulation since it is not weighted solenoidal on Γ in general. To overcome this difficulty, we employ the weighted Helmholtz–Leray projection on Γ (see Section 5.9.3) to replace the averaged tangential component with its weighted solenoidal part. Then we derive the energy estimate for the weighted solenoidal part by substituting it for its weak formulation and apply an estimate for the gradient part of the weighted Helmholtz-Leray decomposition on Γ to obtain the energy estimate for the original averaged tangential component of a strong solution, which enables us to show that the averaged tangential component converges weakly as $\varepsilon \to 0$ and that the limit is a weak solution to the limit equations (5.1.13)-(5.1.14) (see Section 5.10.5). We also derive an estimate for the difference between the averaged tangential component of a strong solution to (5.1.1)-(5.1.3)and a weak solution to (5.1.13)-(5.1.14) by using their weak formulations in Section 5.10.6. Here we again use the weighted Helmholtz–Leray projection on Γ to take the difference of the solutions as a test function in the weak formulations.

Now let us give remarks on Assumption 2 and the limit equations (5.1.13)-(5.1.14).

Remark 5.1.5. In the case of the perfect slip boundary conditions, i.e. the boundary conditions (5.1.2) with $\gamma_{\varepsilon} = 0$, we need to assume that the condition (A2)

$$\mathcal{K}_{q}(\Gamma) = \{ v \in H^{1}(\Gamma)^{3} \mid v \cdot n = 0, \, D_{\Gamma}(v) = 0, \, v \cdot \nabla_{\Gamma} g = 0 \text{ on } \Gamma \} = \{ 0 \}$$

in Assumption 2 is satisfied. This assumption is also made in [20]. On the other hand, the authors of [21, 26] consider the Navier–Stokes equations in

$$\Omega_{\varepsilon} = \{ x = (x', x_3) \in \mathbb{R}^3 \mid x' \in \mathbb{T}^2, \, \varepsilon g_0(x') < x_3 < \varepsilon g_1(x') \}, \quad g_0, g_1 \colon \mathbb{T}^2 \to \mathbb{R}$$

with perfect slip boundary conditions on the top and bottom boundaries without assuming that $\mathcal{K}_g(\mathbb{T}^2) = \{0\}$ (here \mathbb{T}^2 is the flat torus). When $\Gamma = \mathbb{T}^2$, we have

$$\mathcal{K}_g(\mathbb{T}^2) = \{ (a,0) \in \mathbb{R}^3 \mid a \in \mathbb{R}^2, \ a \cdot \nabla_2 g = 0 \text{ in } \mathbb{T}^2 \},\$$

where $g = g_1 - g_0$ and ∇_2 is the gradient operator in $x' \in \mathbb{R}^2$. Assuming that

$$\mathcal{K}_g(\mathbb{T}^2) = \{(a,0) \in \mathbb{R}^3 \mid a \in \mathbb{R}^2, a \cdot \nabla_2 g_0 = a \cdot \nabla_2 g_1 = 0 \text{ in } \mathbb{T}^2\}$$

in [21,26] (in fact $g_0 = 0$ in [26]), the authors get $\mathcal{K}_g(\mathbb{T}^2) \subset L^2_\sigma(\Omega_\varepsilon)$, i.e. any vector in $\mathcal{K}_g(\mathbb{T}^2)$ satisfies the divergence-free condition and the impermeable boundary condition. Based on this fact, they decompose

$$L^{2}(\Omega_{\varepsilon})^{2} = \widehat{L}^{2}_{\sigma}(\Omega_{\varepsilon}) \oplus \mathcal{K}_{g}(\mathbb{T}^{2}) \oplus G^{2}(\Omega_{\varepsilon}),$$

where (we denote by $\mathcal{K}_q(\mathbb{T}^2)^{\perp}$ the orthogonal complement of $\mathcal{K}_q(\mathbb{T}^2)$ in $L^2(\Omega_{\varepsilon})^3$)

$$\widehat{L}^2_{\sigma}(\Omega_{\varepsilon}) = L^2_{\sigma}(\Omega_{\varepsilon}) \cap \mathcal{K}_g(\mathbb{T}^2)^{\perp}, \quad G^2(\Omega_{\varepsilon}) = \{\nabla q \in L^2(\Omega_{\varepsilon}) \mid q \in H^1(\Omega_{\varepsilon})\},$$

and study the abstract evolution equation

$$\partial_t u^{\varepsilon} + \widehat{A}_{\varepsilon} u^{\varepsilon} + \widehat{\mathbb{P}}_{\varepsilon} (u^{\varepsilon} \cdot \nabla) u^{\varepsilon} = \widehat{\mathbb{P}}_{\varepsilon} f^{\varepsilon}, \quad u^{\varepsilon}|_{t=0} = u_0^{\varepsilon}$$
(5.1.15)

in $\widehat{L}^2_{\sigma}(\Omega_{\varepsilon})$, where $\widehat{\mathbb{P}}_{\varepsilon}$ is the orthogonal projection from $L^2(\Omega_{\varepsilon})^3$ onto $\widehat{L}^2_{\sigma}(\Omega_{\varepsilon})$ and $\widehat{A}_{\varepsilon}$ is the Stokes operator on $\widehat{L}^2_{\sigma}(\Omega_{\varepsilon})$ associated with perfect slip boundary conditions. The reason why they consider (5.1.15) in $\widehat{L}^2_{\sigma}(\Omega_{\varepsilon})$ is that they prove the uniform coerciveness in ε of the bilinear form corresponding to $\widehat{A}_{\varepsilon}$ on $\widehat{V}_{\varepsilon} = \widehat{L}^2_{\sigma}(\Omega_{\varepsilon}) \cap H^1(\Omega_{\varepsilon})^3$ (see [21, Proposition 4.12] and [26, Theorem 2.2]).

In this case, however, we need to be careful about recovery of the original Navier–Stokes equations. By (5.1.15) with a strong solution

$$u^{\varepsilon} \in C([0,\infty); \widehat{V}_{\varepsilon}) \cap L^2_{loc}([0,\infty); D(\widehat{A}_{\varepsilon})) \cap H^1_{loc}([0,\infty); \widehat{L}^2_{\sigma}(\Omega_{\varepsilon}))$$

we a priori get

$$\partial_t u^\varepsilon - \nu \Delta u^\varepsilon + (u^\varepsilon \cdot \nabla) u^\varepsilon + \nabla p^\varepsilon + a = f^\varepsilon$$

in $L^2(\Omega_{\varepsilon})^3$ with some function $a = a(t) \in \mathcal{K}_g(\mathbb{T}^2)$, but it vanishes if we assume that $f^{\varepsilon}(t) \in \mathcal{K}_g(\mathbb{T}^2)^{\perp}$ for all $t \geq 0$. Indeed, noting that $\partial_t u^{\varepsilon}, \nabla p^{\varepsilon} \in \mathcal{K}_g(\mathbb{T}^2)^{\perp}$ we take the inner product of the above equation with a to get

$$\int_{\Omega_{\varepsilon}} |a|^2 \, dx = \nu \int_{\Omega_{\varepsilon}} \Delta u^{\varepsilon} \cdot a \, dx - \int_{\Omega_{\varepsilon}} (u^{\varepsilon} \cdot \nabla) u^{\varepsilon} \cdot a \, dx.$$

By integration by parts (see (5.5.13)), the divergence-free and perfect slip boundary conditions on u^{ε} , and D(a) = 0 (note that $a \in \mathcal{K}_g(\mathbb{T}^2)$ is independent of x) we have

$$\int_{\Omega_{\varepsilon}} \Delta u^{\varepsilon} \cdot a \, dx = -2 \int_{\Omega_{\varepsilon}} D(u^{\varepsilon}) : D(a) \, dx = 0.$$
(5.1.16)

Moreover, by integration by parts, the impermeable boundary condition on u^{ε} , and div a = 0 we see that

$$\int_{\Omega_{\varepsilon}} (u^{\varepsilon} \cdot \nabla) u^{\varepsilon} \, dx = -\frac{1}{2} \int_{\Omega_{\varepsilon}} |u^{\varepsilon}|^2 \operatorname{div} a \, dx = 0.$$
(5.1.17)

By these equalities we obtain $||a(t)||^2_{L^2(\Omega_{\varepsilon})} = 0$, i.e. a(t) = 0 for all $t \ge 0$, and the original Navier–Stokes equations are properly recovered.

These arguments are not applicable to our case. To prove the uniform coerciveness in ε of the bilinear form corresponding to the Stokes problem in the curved thin domain Ω_{ε} given by (5.1.4) with perfect slip boundary conditions, we need to work on the space

$$\widehat{L}^2_{\sigma}(\Omega_{\varepsilon}) := \{ u \in L^2_{\sigma}(\Omega_{\varepsilon}) \mid (u, \bar{v})_{L^2(\Omega_{\varepsilon})} = 0 \text{ for all } v \in \mathcal{K}_g(\Gamma) \},\$$

where \bar{v} is the constant extension of $v \colon \Gamma \to \mathbb{R}^3$ in the normal direction of Γ (see Lemmas 5.4.3 and 5.4.5). In this case we have a decomposition

$$L^{2}(\Omega_{\varepsilon}) = \widehat{L}^{2}_{\sigma}(\Omega_{\varepsilon}) \oplus [L^{2}_{\sigma}(\Omega_{\varepsilon}) \cap \overline{\mathcal{K}}_{g}(\Gamma)] \oplus G^{2}(\Omega_{\varepsilon}), \quad \overline{\mathcal{K}}_{g}(\Gamma) := \{ \overline{v} \mid v \in \mathcal{K}_{g}(\Gamma) \}.$$

Here we note that we do not know whether or not $\overline{\mathcal{K}}_g(\Gamma) \subset L^2_\sigma(\Omega_\varepsilon)$. For $\overline{v} \in \overline{\mathcal{K}}_g(\Gamma)$ we have div $\overline{v} = 0$ in Ω_ε (see Lemma 5.B.1), but for $x = y + \varepsilon g_i(y)n(y) \in \Gamma^i_\varepsilon$ with $y \in \Gamma$, i = 0, 1,

$$\bar{v}(x) \cdot n_{\varepsilon}(x) = (-1)^{i} \varepsilon \frac{v(y) \cdot \tau_{\varepsilon}^{i}(y)}{\sqrt{1 + \varepsilon^{2} |\tau_{\varepsilon}^{i}(y)|^{2}}} \quad (\tau_{\varepsilon}^{i}(y) := \{I_{3} - \varepsilon g_{i}(y)W(y)\}^{-1} \nabla_{\Gamma} g_{i}(y))$$

does not vanish in general just by $v \cdot \nabla_{\Gamma} g = 0$ on Γ (here W is the Weingarten map of Γ , see Lemma 5.2.10). If we solve the abstract evolution equation (5.1.15) in $\hat{L}^2_{\sigma}(\Omega_{\varepsilon})$ to obtain a strong solution u^{ε} , then we have

$$\partial_t u^\varepsilon - \nu \Delta u^\varepsilon + (u^\varepsilon \cdot \nabla) u^\varepsilon + \nabla p^\varepsilon + \bar{v} = f^\varepsilon$$

in $L^2(\Omega_{\varepsilon})^3$ with some $\bar{v} = \bar{v}(x,t) \in L^2_{\sigma}(\Omega_{\varepsilon}) \cap \overline{\mathcal{K}}_g(\Gamma)$. However, we cannot get $\bar{v} = 0$ even if we assume that $f^{\varepsilon} \in \overline{\mathcal{K}}_g(\Gamma)^{\perp}$, since the equality (5.1.16) with *a* replaced by \bar{v} is not valid in general. Indeed, by (5.2.12) we have

$$D(\bar{v})(x) = \frac{1}{2} \Big[\{I_3 - rW(y)\}^{-1} \nabla_{\Gamma} v(y) + \{\nabla_{\Gamma} v(y)\}^T \{I_3 - rW(y)\}^{-1} \Big], x = y + rn(y) \in \Omega_{\varepsilon}, \ y \in \Gamma, \ r \in (\varepsilon g_0(y), \varepsilon g_1(y)),$$

which does not vanish just by $D_{\Gamma}(v) = 0$ on Γ . (The equality (5.1.17) is still valid for \bar{v} since div $\bar{v} = 0$ in Ω_{ε} .) To avoid this difficulty in recovery of the original Navier–Stokes equations we assume $\mathcal{K}_g(\Gamma) = \{0\}$ in the case of the perfect slip boundary conditions. This assumption is also essential for derivation of an estimate for the difference between the Stokes and Laplace operators (see Remark 5.5.9).

Remark 5.1.6. As we mentioned in Remark 5.1.5, the assumption $\mathcal{K}_g(\Gamma) = \{0\}$ is necessary for our results in the case of the perfect slip boundary conditions, i.e. the boundary conditions (5.1.2) with $\gamma_{\varepsilon} = 0$. A typical example violating this assumption is the thin spherical shell

$$\Omega_{\varepsilon} = \{ x \in \mathbb{R}^3 \mid 1 < |x| < 1 + \varepsilon \}, \tag{5.1.18}$$

whose limit surface is the unit sphere S^2 in \mathbb{R}^3 . In this case $\nabla_{\Gamma}g = 0$ on S^2 and the restriction on S^2 of $u(x) = e_3 \times x$, $x \in \mathbb{R}^3$ with $e_3 = (0, 0, 1)$ is a Killing vector field on S^2 , and thus $\mathcal{K}_g(S^2) = \mathcal{K}(S^2) \neq \{0\}$. Hence our results do not cover the case of the thin spherical shell with perfect slip boundary conditions.

The Navier–Stokes equations in the thin spherical shell was studied by Temam and Ziane [66]. They imposed the Hodge (or de Rham) boundary conditions

$$u \cdot n_{\varepsilon} = 0$$
, $\operatorname{curl} u \times n_{\varepsilon} = 0$ on $\partial \Omega_{\varepsilon}$,

which is called the free boundary conditions in [66]. It is mentioned in [66] that the Hodge boundary conditions are equivalent to the perfect slip boundary conditions, but it is not true for the thin spherical shell given by (5.1.18). Indeed, for $u(x) = e_3 \times x$ we easily see that D(u) = 0 and curl $u = 2e_3$ in \mathbb{R}^3 . Hence on the inner boundary S^2 with unit outward normal $n_{\varepsilon}(x) = -x$ we have

$$u \cdot n_{\varepsilon} = -(e_3 \times x) \cdot x = 0, \quad P_{\varepsilon} D(u) n_{\varepsilon} = 0, \quad \operatorname{curl} u \times n_{\varepsilon} = -2e_3 \times x$$

for $x \in S^2$ and the last vector does not vanish in general. More generally, the Hodge boundary conditions are equivalent to the perfect slip boundary conditions only when a boundary is

a part of a plane. For example, on the plane $\{(x', 0) \mid x' \in \mathbb{R}^2\}$ these boundary conditions reduce to

$$u_3(x',0) = 0, \quad \partial_3 u_1(x',0) = \partial_3 u_2(x',0) = 0, \quad x' \in \mathbb{R}^2.$$

The difference between the Hodge and perfect slip boundary conditions is due to the curvature of a boundary. It also appears in limit equations on a limit surface of the Navier–Stokes equations in a curved thin domain (see Remark 5.1.7 below).

Remark 5.1.7. If $g \equiv 1$ and $\gamma^0 = \gamma^1 = 0$, the limit equations (5.1.13)–(5.1.14) become

$$\partial_t v + \overline{\nabla}_v v - 2\nu P \operatorname{div}_{\Gamma}[D_{\Gamma}(v)] + \nabla_{\Gamma} q = f, \quad \operatorname{div}_{\Gamma} v = 0 \quad \text{on} \quad \Gamma \times (0, \infty).$$

In [43, Lemma 2.5], it is shown that

$$2P \operatorname{div}_{\Gamma}[D_{\Gamma}(v)] = \Delta_B v + Kv$$
 on Γ

for a tangential vector field v on Γ satisfying $\operatorname{div}_{\Gamma} v = 0$ on Γ , where $\Delta_B = -\overline{\nabla}^* \overline{\nabla}$ is the Bochner Laplacian (see Appendix 5.C) and K is the Gaussian curvature of Γ (see Section 5.2.1). Hence the above equations are of the form

$$\partial_t v + \overline{\nabla}_v v - \nu (\Delta_B v + K v) + \nabla_{\Gamma} q = f, \quad \operatorname{div}_{\Gamma} v = 0 \quad \text{on} \quad \Gamma \times (0, \infty).$$
(5.1.19)

These equations are called the "proper" Navier–Stokes equations on a Riemannian manifold in [12, 62] and were studied by Mitrea and Taylor [41], Nagasawa [46], and Taylor [62]. Note that $Kv = \operatorname{Ric}(v)$ for a tangential vector field v on the embedded surface Γ in \mathbb{R}^3 , where Ric is the Ricci curvature. Also, the Bochner Laplacian is related to the Hodge Laplacian $\Delta_D = -(d_{\Gamma}d_{\Gamma}^* + d_{\Gamma}^*d_{\Gamma})$ with the exterior derivative d_{Γ} by the Weitzenböck formula $\Delta_D = \Delta_B - \operatorname{Ric}$. For details, see e.g. [30, 50].

On the other hand, Temam and Ziane [66] derived the limit equations

$$\partial_t v + \overline{\nabla}_v v - \nu \Delta_2 v + \nabla_{\Gamma} q = f, \quad \operatorname{div}_{\Gamma} v = 0 \quad \text{on} \quad S^2 \times (0, \infty)$$

from the Navier–Stokes equations in the thin spherical shell given by (5.1.18). Here $\Delta_2 v = P\Delta \bar{v}$ is the tangential vector Laplacian of a tangential vector field v on S^2 , where \bar{v} is the constant extension of v in the normal direction of S^2 (see [66, Appendix]). In terms of our notations given in Section 5.2.1 it is expressed as

$$\Delta_2 v = P \Delta \bar{v} = P \Delta_{\Gamma} v = \Delta_B v - W^2 v \quad \text{on} \quad S^2$$

by Lemmas 5.2.3 and 5.C.5. Moreover, when $\Gamma = S^2$ we have W = -P and thus $W^2 v = v$ for the tangential vector field v. Thus, the limit equations in [66] become

$$\partial_t v + \overline{\nabla}_v v - \nu (\Delta_B v - v) + \nabla_\Gamma q = f, \quad \operatorname{div}_\Gamma v = 0 \quad \text{on} \quad S^2 \times (0, \infty).$$
(5.1.20)

Since K = 1 for the unit sphere S^2 our limit equations (5.1.19) are formally of the form

$$\partial_t v + \overline{\nabla}_v v - \nu (\Delta_B v + v) + \nabla_{\Gamma} q = f, \quad \operatorname{div}_{\Gamma} v = 0 \quad \text{on} \quad S^2 \times (0, \infty).$$

(Note that our results do not cover the case of the thin spherical shell given by (5.1.18) with perfect slip boundary conditions.) The sign of v in the viscous term of this system is opposite to that of (5.1.20), which produces different structures of the limit equations such as the stability of a solution. As we mentioned in Remark 5.1.6, this is due to the difference between the Hodge and perfect slip boundary conditions on a curved boundary.

This chapter is organized as follows. In Section 5.2 we introduce notations for surface quantities and function spaces on a closed surface and give the definition and basic properties of a curved thin domain. We present in Section 5.3 fundamental formulas and inequalities for functions on the surface and thin domain. In Section 5.4 we prove the uniform Korn inequalities on the thin domain and the Korn inequalities on the surface. We define the Stokes operator on the thin domain with slip boundary conditions and show its uniform norm equivalence in Section 5.5. Also, we give a uniform estimate for the gradient part of the Helmholtz–Leray decomposition on the thin domain, which is used in the study of a singular limit problem. In Section 5.6 we define average operators in the thin direction and derive several estimates for the average of functions on the thin domain. The main purpose of Section 5.6 is to give decomposition of a vector field on the thin domain into the average and residual parts and to establish useful estimates for them. We also use the average operators to approximate bilinear and trilinear forms for functions on the thin domains by those for functions on the limit surface. In Section 5.7 we prove a good estimate for the trilinear term, i.e. the L^2 -inner product of the inertial term and the Stokes operator. The main ingredients for the proof are the estimates for the Stokes and average operators given in Sections 5.5 and 5.6. Using the estimate for the trilinear term, we establish the global existence and uniform estimates of a strong solution (Theorems 5.1.1 and 5.1.2) in Section 5.8. The last two sections are devoted to the study of a singular limit problem when the curved thin domain degenerates into the closed surface. In Section 5.9 we deal with weighted solenoidal spaces on a closed surface. We give characterization of the annihilator of a weighted solenoidal space and prove the weighted Helmholtz–Leray decomposition on the surface with several estimates for the gradient part. In Section 5.10 we investigate the behavior of the average in the thin direction of a strong solution to the Navier–Stokes equations in the curved thin domain. Our goal is to show the convergence of the average towards a weak solution to the limit equations on the limit surface (Theorems 5.1.3 and 5.1.4). In Appendix 5.A we fix notations on vectors and matrices. We also prove some lemmas by elementary vector calculus. In Appendix 5.B we give the proofs of lemmas in Section 5.2 involving calculations of surface quantities of the surface and the boundary of the thin domain. We introduce the Riemannian connection on a surface and show formulas for the covariant derivative of a tangential vector field in Appendix 5.C.

5.2 Preliminaries

In this section we give notations and formulas on several quantities for a two-dimensional closed surface and a thin domain in \mathbb{R}^3 . Some lemmas in this section are proved just by direct calculations involving differential geometry of surfaces. We give their proofs in Appendix 5.B to avoid making this section too long.

Throughout this chapter we denote by c a general positive constant independent of the parameter ε . Also, we fix a coordinate system of \mathbb{R}^3 and write x_i , i = 1, 2, 3 for the *i*-th component of a point $x \in \mathbb{R}^3$ under the fixed coordinate system. For a vector $a \in \mathbb{R}^3$ we denote by a_i , i = 1, 2, 3 the *i*-th component of a. Sometimes we write a^i or $[a]_i$ instead of a_i . Also, for a matrix $A \in \mathbb{R}^{3\times 3}$ and i, j = 1, 2, 3 we denote by A_{ij} or $[A]_{ij}$ the (i, j)-entry of A. Other notations and basic formulas on vectors and matrices are given in Appendix 5.A.

5.2.1 Closed surface

Let Γ be a two-dimensional closed (i.e. compact and without boundary), connected, and oriented surface in \mathbb{R}^3 . We assume that Γ is of class C^{ℓ} with $\ell \geq 2$. By n and d we denote the unit outward normal vector field of Γ and the signed distance function from Γ increasing in the direction of n. Also, let κ_1 and κ_2 be the principal curvatures of Γ . By the regularity of Γ we have $n \in C^{\ell-1}(\Gamma)^3$ and $\kappa_1, \kappa_2 \in C^{\ell-2}(\Gamma)$. In particular, κ_1 and κ_2 are bounded on the compact set Γ . Hence we can take a tubular neighborhood $N := \{x \in \mathbb{R}^3 \mid \text{dist}(x, \Gamma) < \delta\},$ $\delta > 0$ of Γ such that for each $x \in N$ there exists a unique point $\pi(x) \in \Gamma$ satisfying

$$x = \pi(x) + d(x)n(\pi(x)), \quad \nabla d(x) = n(\pi(x)).$$
(5.2.1)

Moreover, d and π are of class C^{ℓ} and $C^{\ell-1}$ on \overline{N} (see [15, Section 14.6] for details). By the boundedness of κ_1 and κ_2 we also have

$$c^{-1} \le 1 - r\kappa_i(y) \le c$$
 for all $y \in \Gamma, r \in (-\delta, \delta), i = 1, 2$ (5.2.2)

if we take $\delta > 0$ sufficiently small.

Let us define differential operators on the surface Γ . For $y \in \Gamma$ we set

$$P(y) := I_3 - n(y) \otimes n(y), \quad Q(y) := n(y) \otimes n(y).$$

By the definitions and the regularity of Γ we have $P, Q \in C^{\ell-1}(\Gamma)^{3\times 3}$ and

$$I_3 = P + Q, \quad PQ = QP = 0, \quad P^T = P^2 = P, \quad Q^T = Q^2 = Q \quad \text{on} \quad \Gamma$$

The matrices P and Q are the orthogonal projections onto the tangent plane and the normal direction of Γ . In particular, we have |P| = |Q| = 1 on Γ . For $\eta \in C^1(\Gamma)$ we define the tangential gradient and the tangential derivatives of η as

$$\nabla_{\Gamma}\eta(y) := P(y)\nabla\tilde{\eta}(y), \quad \underline{D}_{i}\eta(y) := \sum_{j=1}^{3} P_{ij}(y)\partial_{j}\tilde{\eta}(y), \quad y \in \Gamma, \, i = 1, 2, 3$$

so that $\nabla_{\Gamma}\eta = (\underline{D}_1\eta, \underline{D}_2\eta, \underline{D}_3\eta)$. Here $\tilde{\eta}$ is a C^1 -extension of η to N with $\tilde{\eta}|_{\Gamma} = \eta$. We easily see that

$$P\nabla_{\Gamma}\eta = \nabla_{\Gamma}\eta, \quad n \cdot \nabla_{\Gamma}\eta = 0 \quad \text{on} \quad \Gamma.$$
(5.2.3)

Note that the values of $\nabla_{\Gamma}\eta$ and $\underline{D}_i\eta$ are independent of the choice of an extension $\tilde{\eta}$ (see e.g. [11, Lemma 2.4]). In particular, the constant extension $\bar{\eta} := \eta \circ \pi$ of η in the normal direction of Γ satisfies

$$\nabla \bar{\eta}(y) = \nabla_{\Gamma} \eta(y), \quad \partial_i \bar{\eta}(y) = \underline{D}_i \eta(y), \quad y \in \Gamma, \, i = 1, 2, 3$$
(5.2.4)

since $\nabla \pi(y) = P(y)$ for $y \in \Gamma$ by (5.2.1) and d(y) = 0. In what follows, a function $\bar{\eta}$ with an overline always stands for the constant extension of a function η on Γ in the normal direction of Γ . The tangential Hessian matrix of $\eta \in C^2(\Gamma)$ and the Laplace–Beltrami operator are given by

$$\nabla_{\Gamma}^2 \eta := (\underline{D}_i \underline{D}_j \eta)_{i,j}, \quad \Delta_{\Gamma} \eta := \operatorname{tr}[\nabla_{\Gamma}^2 \eta] = \sum_{i=1}^3 \underline{D}_i^2 \eta \quad \text{on} \quad \Gamma$$

For a (not necessarily tangential) vector field $v = (v_1, v_2, v_3) \in C^1(\Gamma)^3$, we define the tangential gradient matrix and the surface divergence of v by

$$\nabla_{\Gamma} v = \begin{pmatrix} \underline{D}_1 v_1 & \underline{D}_1 v_2 & \underline{D}_1 v_3 \\ \underline{D}_2 v_1 & \underline{D}_2 v_2 & \underline{D}_2 v_3 \\ \underline{D}_3 v_1 & \underline{D}_3 v_2 & \underline{D}_3 v_3 \end{pmatrix}, \quad \operatorname{div}_{\Gamma} v = \operatorname{tr}[\nabla_{\Gamma} v] = \sum_{i=1}^3 \underline{D}_i v_i \quad \text{on} \quad \Gamma.$$

Note that $\nabla_{\Gamma} v = P \nabla \tilde{v}$ on Γ for any C^1 -extension \tilde{v} of v to N with $\tilde{v}|_{\Gamma} = v$. When $v \in C^2(\Gamma)^3$ we write

$$|\nabla_{\Gamma}^2 v|^2 := \sum_{i,j,k=1}^3 |\underline{D}_i \underline{D}_j v_k|^2, \quad \Delta_{\Gamma} v := (\Delta_{\Gamma} v_1, \Delta_{\Gamma} v_2, \Delta_{\Gamma} v_3) \quad \text{on} \quad \Gamma.$$

For a matrix-valued function $A \in C^1(\Gamma)^{3 \times 3}$ of the form

$$A = (A_{ij})_{i,j} = \begin{pmatrix} A_{11} & A_{12} & A_{13} \\ A_{21} & A_{22} & A_{23} \\ A_{31} & A_{32} & A_{33} \end{pmatrix}$$

we define the surface divergence of A as a vector field on Γ with *j*-th component

$$[\operatorname{div}_{\Gamma} A]_j := \sum_{i=1}^2 \underline{D}_i A_{ij}, \quad j = 1, 2, 3.$$

Next we give surface quantities on Γ . We define the Weingarten map W, (twice) the mean curvature H, and the Gaussian curvature K of Γ by

$$W := -\nabla_{\Gamma} n, \quad H := \operatorname{tr}[W] = -\operatorname{div}_{\Gamma} n, \quad K := \kappa_1 \kappa_2 \quad \text{on} \quad \Gamma.$$
 (5.2.5)

Note that W, H, and K are of class $C^{\ell-2}$ by the smoothness of Γ .

Lemma 5.2.1. The Weingarten map W is symmetric and satisfies

$$Wn = 0, \quad PW = WP = W \quad on \quad \Gamma. \tag{5.2.6}$$

We also have

$$\operatorname{div}_{\Gamma} P = Hn \quad on \quad \Gamma. \tag{5.2.7}$$

Proof. For $y \in \Gamma$ we have $W(y) = -\nabla \bar{n}(y) = -\nabla^2 d(y)$ by (5.2.1) and (5.2.4). Hence W is symmetric. Taking the tangential gradient of $|n|^2 = 1$ on Γ we get the first relation of (5.2.6). Also, the second relation immediately follows from the first one. For j = 1, 2, 3 the *j*-th component of $\operatorname{div}_{\Gamma} P$ is of the form

$$[\operatorname{div}_{\Gamma} P]_{j} = \sum_{i=1}^{3} \underline{D}_{i}(\delta_{ij} - n_{i}n_{j}) = \sum_{i=1}^{3} (W_{ii}n_{j} + W_{ij}n_{i}) = Hn_{j} + [W^{T}n]_{j}$$

by (5.2.5) and thus $\operatorname{div}_{\Gamma} P = Hn + W^T n$ on Γ . Applying $W^T = W$ and Wn = 0 to this equality we obtain (5.2.7).

By (5.2.1) the Weingarten map W has the eigenvalue zero associated with eigenvector n. Note that the other eigenvalues of W are the principal curvatures κ_1 and κ_2 (see e.g. [15, Section 14.6] and [33, Section VII.5]) and thus $H = \kappa_1 + \kappa_2$ on Γ .

When we calculate the derivatives of the constant extension of a function on Γ , the inverse matrix of $I_3 - rW(y)$ for $y \in \Gamma$ and $r \in (-\delta, \delta)$ appears.

Lemma 5.2.2. The matrix $I_3 - rW(y)$ is invertible and we have

$$\{I_3 - rW(y)\}^{-1}P(y) = P(y)\{I_3 - rW(y)\}^{-1}$$
(5.2.8)

for all $y \in \Gamma$ and $r \in (-\delta, \delta)$. Moreover, there exists a constant c > 0 such that

$$c^{-1}|a| \le \left| \{I_3 - rW(y)\}^k a \right| \le c|a|, \quad k = \pm 1, \tag{5.2.9}$$

$$\left|I_3 - \{I_3 - rW(y)\}^{-1}\right| \le c|r| \tag{5.2.10}$$

for all $y \in \Gamma$, $r \in (-\delta, \delta)$, and $a \in \mathbb{R}^3$.

Lemma 5.2.3. For all $x \in N$ we have

$$\nabla \pi(x) = \left\{ I_3 - d(x)\overline{W}(x) \right\}^{-1} \overline{P}(x).$$
(5.2.11)

Therefore, the constant extension $\bar{\eta} = \eta \circ \pi$ of $\eta \in C^1(\Gamma)$ satisfies

$$\nabla \bar{\eta}(x) = \left\{ I_3 - d(x)\overline{W}(x) \right\}^{-1} \overline{\nabla_{\Gamma} \eta}(x), \quad x \in N$$
(5.2.12)

and there exists a constant c > 0 independent of η such that

$$c^{-1} \left| \overline{\nabla_{\Gamma} \eta}(x) \right| \le \left| \nabla \overline{\eta}(x) \right| \le c \left| \overline{\nabla_{\Gamma} \eta}(x) \right|, \qquad (5.2.13)$$

$$\left|\nabla\bar{\eta}(x) - \overline{\nabla_{\Gamma}\eta}(x)\right| \le c \left|d(x)\overline{\nabla_{\Gamma}\eta}(x)\right| \tag{5.2.14}$$

for all $x \in N$. If Γ is of class C^3 and $\eta \in C^2(\Gamma)$, then we have

$$|\nabla^2 \eta(x)| \le c \left(\left| \overline{\nabla_{\Gamma} \eta}(x) \right| + \left| \overline{\nabla_{\Gamma}^2 \eta}(x) \right| \right), \quad x \in N.$$
(5.2.15)

Moreover, $\Delta \bar{\eta} = \Delta_{\Gamma} \eta$ on Γ .

We give the proofs of Lemmas 5.2.2 and 5.2.3 in Appendix 5.B. Since $\bar{n} = \nabla d$ in N by (5.2.1), we use (5.2.12) and $W = -\nabla_{\Gamma} n$ on Γ to obtain

$$\nabla \bar{n}(x) = \nabla^2 d(x) = -\left\{I_3 - d(x)\overline{W}(x)\right\}^{-1}\overline{W}(x), \quad x \in N.$$
(5.2.16)

If Γ is of class C^3 , then $n \in C^2(\Gamma)^3$ and thus

$$|\nabla \bar{n}(x)| \le c, \quad |\nabla^2 \bar{n}(x)| \le c, \quad x \in N$$
(5.2.17)

with some constant c > 0 by (5.2.12) and (5.2.15).

Let us define the weak tangential derivatives of a function on Γ and the Sobolev spaces on Γ . For a vector field $v \in C^1(\Gamma)^3$ we have

$$\int_{\Gamma} \operatorname{div}_{\Gamma} v \, d\mathcal{H}^2 = \int_{\Gamma} \operatorname{div}_{\Gamma} (Pv) \, d\mathcal{H}^2 + \int_{\Gamma} \operatorname{div}_{\Gamma} [(v \cdot n)n] \, d\mathcal{H}^2$$

by the decomposition $v = Pv + (v \cdot n)n$, where \mathcal{H}^2 is the two-dimensional Hausdorff measure. The first integral on the right-hand side vanishes by the Stokes theorem, since Pv is tangential on the closed surface Γ . Moreover, to the second integral we apply $\operatorname{div}_{\Gamma}(\xi n) = \nabla_{\Gamma} \xi \cdot n + \xi \operatorname{div}_{\Gamma} n = -\xi H$ for $\xi \in C^1(\Gamma)$. Then we get

$$\int_{\Gamma} \operatorname{div}_{\Gamma} v \, d\mathcal{H}^2 = -\int_{\Gamma} (v \cdot n) H \, d\mathcal{H}^2$$

for all $v \in C^1(\Gamma)^3$. In particular, for $\eta, \xi \in C^1(\Gamma)$ we set $v = \eta \xi e_i$ in the above formula, where $\{e_1, e_2, e_3\}$ is the standard basis of \mathbb{R}^3 , to obtain

$$\int_{\Gamma} (\eta \underline{D}_i \xi + \xi \underline{D}_i \eta) \, d\mathcal{H}^2 = -\int_{\Gamma} \eta \xi H n_i \, d\mathcal{H}^2, \quad i = 1, 2, 3.$$
(5.2.18)

Based on this identity, for $p \in [1, \infty]$ and i = 1, 2, 3 we say that $\eta \in L^p(\Gamma)$ has the *i*-th weak tangential derivative if there exists $\eta_i \in L^p(\Gamma)$ such that

$$\int_{\Gamma} \eta_i \xi \, d\mathcal{H}^2 = -\int_{\Gamma} \eta(\underline{D}_i \xi + \xi H n_i) \, d\mathcal{H}^2 \tag{5.2.19}$$

for all $\xi \in C^1(\Gamma)$. In this case we write $\underline{D}_i \eta = \eta_i$ and define the Sobolev space

$$W^{1,p}(\Gamma) := \{ \eta \in L^p(\Gamma) \mid \underline{D}_i \eta \in L^p(\Gamma) \text{ for all } i = 1, 2, 3 \},$$
$$\|\eta\|_{W^{1,p}(\Gamma)} := \begin{cases} \left(\|\eta\|_{L^p(\Gamma)}^p + \|\nabla_{\Gamma}\eta\|_{L^p(\Gamma)}^p \right)^{1/p} & \text{if } p \in [1, \infty), \\ \|\eta\|_{L^\infty(\Gamma)} + \|\nabla_{\Gamma}\eta\|_{L^\infty(\Gamma)} & \text{if } p = \infty. \end{cases}$$

Note that $W^{1,p}(\Gamma)$ is a Banach space. In particular, $H^1(\Gamma) = W^{1,2}(\Gamma)$ is a Hilbert space equipped with inner product $(\eta, \xi)_{H^1(\Gamma)} := (\eta, \xi)_{L^2(\Gamma)} + (\nabla_{\Gamma} \eta, \nabla_{\Gamma} \xi)_{L^2(\Gamma)}$. We also define the second order Sobolev space

$$W^{2,p}(\Gamma) := \{ \eta \in W^{1,p}(\Gamma) \mid \underline{D}_{i}\underline{D}_{j}\eta \in L^{p}(\Gamma) \text{ for all } i, j = 1, 2, 3 \}, \\ \|\eta\|_{W^{2,p}(\Gamma)} := \begin{cases} \left(\|\eta\|_{W^{1,p}(\Gamma)}^{p} + \|\nabla_{\Gamma}^{2}\eta\|_{L^{p}(\Gamma)}^{p} \right)^{1/p} & \text{if } p \in [1,\infty), \\ \|\eta\|_{W^{1,\infty}(\Gamma)} + \|\nabla_{\Gamma}^{2}\eta\|_{L^{\infty}(\Gamma)} & \text{if } p = \infty. \end{cases}$$

Then $W^{2,p}(\Gamma)$ is again a Banach space and $H^2(\Gamma) = W^{2,2}(\Gamma)$ is a Hilbert space. In what follows, we write $W^{0,p}(\Gamma) = L^p(\Gamma)$ for $p \in [1,\infty]$.

Lemma 5.2.4. Let $p \in [1, \infty)$ and $m = 0, 1, \ldots, \ell$. Then $C^{\ell}(\Gamma)$ is dense in $W^{m,p}(\Gamma)$.

We prove Lemma 5.2.4 in Appendix 5.B by standard localization and mollification arguments. As in the case of a flat domain, Poincaré's inequality holds on Γ .

Lemma 5.2.5. Let $p \in [1, \infty)$. There exists a constant c > 0 such that

$$\|\eta\|_{L^p(\Gamma)} \le c \|\nabla_{\Gamma}\eta\|_{L^p(\Gamma)} \tag{5.2.20}$$

for all $\eta \in W^{1,p}(\Gamma)$ satisfying $\int_{\Gamma} \eta \, d\mathcal{H}^2 = 0$.

We refer to [11, Theorem 2.12] for the proof of Lemma 5.2.5. Note that the proof given there applies to a closed, connected, and oriented hypersurface of class C^2 .

Let $\mathcal{X}(\Gamma)$ be a function space on Γ like $C^m(\Gamma)$, $L^p(\Gamma)$, and $W^{m,p}(\Gamma)$. We define the space of all tangential vector fields on Γ whose components in $\mathcal{X}(\Gamma)$ by

$$\mathcal{X}(\Gamma, T\Gamma) := \{ v \in \mathcal{X}(\Gamma)^3 \mid v \cdot n = 0 \text{ on } \Gamma \}$$

Note that $W^{m,p}(\Gamma, T\Gamma)$ is a closed subspace of $W^{m,p}(\Gamma)^3$, and thus a Banach space. Moreover, an element of $W^{m,p}(\Gamma, T\Gamma)$ with $p \neq \infty$ can be approximated by smooth tangential vector fields on Γ .

Lemma 5.2.6. Let $p \in [1, \infty)$ and $m = 0, 1, \ldots, \ell - 1$. Then $C^{\ell-1}(\Gamma, T\Gamma)$ is dense in $W^{m,p}(\Gamma, T\Gamma)$ with respect to the norm $\|\cdot\|_{W^{m,p}(\Gamma)}$.

Proof. Let $v \in W^{m,p}(\Gamma, T\Gamma) \subset W^{m,p}(\Gamma)^3$. By Lemma 5.2.4 we can take a sequence $\{\tilde{v}_k\}_{k=1}^{\infty}$ in $C^{\ell}(\Gamma)^3$ that converges to v strongly in $W^{m,p}(\Gamma)^3$. For each $k \in \mathbb{N}$ we set $v_k := P\tilde{v}_k \in C^{\ell-1}(\Gamma, T\Gamma)$ (note that P is of class $C^{\ell-1}$ on Γ). Then since v is tangential on Γ , we have $v - v_k = P(v - \tilde{v}_k)$ on Γ . By this equality, $P \in C^{\ell-1}(\Gamma)^{3\times 3}$, and the strong convergence of $\{\tilde{v}_k\}_{k=1}^{\infty}$ to v in $W^{m,p}(\Gamma)^3$ we see that

$$\|v - v_k\|_{W^{m,p}(\Gamma)} \le c \|v - \tilde{v}_k\|_{W^{m,p}(\Gamma)} \to 0 \quad \text{as} \quad k \to \infty.$$

Hence $\{v_k\}_{k=1}^{\infty}$ converges to v strongly in $W^{m,p}(\Gamma, T\Gamma)$ and the claim is valid.

Let $H^{-1}(\Gamma)$ be the dual space of $H^1(\Gamma)$ and $\langle \cdot, \cdot \rangle_{\Gamma}$ be the duality product between $H^{-1}(\Gamma)$ and $H^1(\Gamma)$. We consider $\eta \in L^2(\Gamma)$ as an element of $H^{-1}(\Gamma)$ by setting

$$\langle \eta, \xi \rangle_{\Gamma} := (\eta, \xi)_{L^2(\Gamma)}, \quad \xi \in H^1(\Gamma)$$

$$(5.2.21)$$

to get the compact embeddings $H^1(\Gamma) \hookrightarrow L^2(\Gamma) \hookrightarrow H^{-1}(\Gamma)$. For $\eta \in W^{1,\infty}(\Gamma), \xi \in H^{-1}(\Gamma)$, and $\varphi \in H^1(\Gamma)$ we see that

$$|\langle \xi, \eta \varphi \rangle_{\Gamma}| \le \|\xi\|_{H^{-1}(\Gamma)} \|\eta \varphi\|_{H^{1}(\Gamma)} \le c \|\eta\|_{W^{1,\infty}(\Gamma)} \|\xi\|_{H^{-1}(\Gamma)} \|\varphi\|_{H^{1}(\Gamma)}$$

where c > 0 is independent of η , ξ , and φ . Hence we can define $\eta \xi \in H^{-1}(\Gamma)$ by

$$\langle \eta \xi, \varphi \rangle_{\Gamma} := \langle \xi, \eta \varphi \rangle_{\Gamma}, \quad \varphi \in H^1(\Gamma).$$
 (5.2.22)

Similarly, for $\eta \in L^2(\Gamma)$ we can define $\underline{D}_i \eta \in H^{-1}(\Gamma)$, i = 1, 2, 3 by

$$\langle \underline{D}_i \eta, \xi \rangle_{\Gamma} := -(\eta, \underline{D}_i \xi + \xi H n_i)_{L^2(\Gamma)}, \quad \xi \in H^1(\Gamma)$$
(5.2.23)

since n and H are bounded on Γ . Based on this definition we consider the weak tangential gradient $\nabla_{\Gamma} \eta$ of $\eta \in L^2(\Gamma)$ as an element of $H^{-1}(\Gamma)^3$ satisfying

$$\langle \nabla_{\Gamma} \eta, v \rangle_{\Gamma} = -(\eta, \operatorname{div}_{\Gamma} v + (v \cdot n)H)_{L^{2}(\Gamma)}, \quad v \in H^{1}(\Gamma)^{3}.$$
(5.2.24)

Also, the surface divergence of $v \in L^2(\Gamma)^3$ is given by

$$\langle \operatorname{div}_{\Gamma} v, \eta \rangle_{\Gamma} = -(v, \nabla_{\Gamma} \eta + \eta H n)_{L^{2}(\Gamma)}, \quad \eta \in H^{1}(\Gamma).$$
 (5.2.25)

Let $H^{-1}(\Gamma, T\Gamma)$ be the dual of $H^1(\Gamma, T\Gamma)$ and $[\cdot, \cdot]_{T\Gamma}$ the duality product between $H^{-1}(\Gamma, T\Gamma)$ and $H^1(\Gamma, T\Gamma)$. It is homeomorphic to a quotient space of $H^{-1}(\Gamma)^3$.

Lemma 5.2.7. For $f \in H^{-1}(\Gamma)^3$ we define an equivalence class

$$[f] := \{ \tilde{f} \in H^{-1}(\Gamma)^3 \mid Pf = P\tilde{f} \text{ in } H^{-1}(\Gamma)^3 \}.$$

Then the quotient space $\mathcal{Q} := \{ [f] \mid f \in H^{-1}(\Gamma)^3 \}$ is homeomorphic to $H^{-1}(\Gamma, T\Gamma)$.

Note that \mathcal{Q} is a Banach space equipped with norm $\|[f]\|_{\mathcal{Q}} := \inf_{\tilde{f} \in [f]} \|\tilde{f}\|_{H^{-1}(\Gamma)}$ (see e.g. [56] for details).

Proof. Let $f_1, f_2 \in H^{-1}(\Gamma)^3$. If $Pf_1 = Pf_2$ in $H^{-1}(\Gamma)^3$, then $\langle f_1, v \rangle_{\Gamma} = \langle f_2, v \rangle_{\Gamma}$ for all $v \in H^1(\Gamma, T\Gamma)$ by (5.2.22) and Pv = v. Hence we can define a linear operator L from \mathcal{Q} to $H^{-1}(\Gamma, T\Gamma)$ by $[L[f], v]_{T\Gamma} := \langle \tilde{f}, v \rangle_{\Gamma}$ for $[f] \in \mathcal{Q}$ and $v \in H^1(\Gamma, T\Gamma)$, where \tilde{f} is any element of [f]. By this definition we also see that

$$||L[f]||_{H^{-1}(\Gamma,T\Gamma)} \le \inf_{\tilde{f}\in[f]} ||\tilde{f}||_{H^{-1}(\Gamma)} = ||[f]||_{\mathcal{Q}}.$$

Hence L is bounded. Moreover, if $L[f_1] = L[f_2]$ in $H^{-1}(\Gamma, T\Gamma)$, then

$$\langle Pf_1, v \rangle_{\Gamma} = \langle f_1, Pv \rangle_{\Gamma} = [L[f_1], Pv]_{T\Gamma} = [L[f_2], Pv]_{T\Gamma} = \langle f_2, Pv \rangle_{\Gamma} = \langle Pf_2, v \rangle_{\Gamma}$$

for all $v \in H^1(\Gamma)^3$ and thus $Pf_1 = Pf_2$ in $H^{-1}(\Gamma)^3$, which means that $[f_1] = [f_2]$ and Lis injective. To show its surjectivity, let $F \in H^{-1}(\Gamma, T\Gamma)$. Since $H^1(\Gamma, T\Gamma)$ is a Hilbert space equipped with inner product of $H^1(\Gamma)^3$, by the Riesz representation theorem there exists $v_F \in H^1(\Gamma, T\Gamma)$ such that $[F, v]_{T\Gamma} = (v_F, v)_{H^1(\Gamma)}$ for all $v \in H^1(\Gamma, T\Gamma)$. Then setting $f_F := v_F - \sum_{i=1}^3 \underline{D}_i^2 v_F \in H^{-1}(\Gamma)^3$ we observe by (5.2.21), (5.2.23), and $\sum_{i=1}^3 n_i \underline{D}_i v_F^j = 0$ in $L^2(\Gamma)$ for j = 1, 2, 3 that

$$\begin{split} [F,v]_{T\Gamma} &= \sum_{i,j=1}^{3} \{ (v_F^j, v^j)_{L^2(\Gamma)} + (\underline{D}_i v_F^j, \underline{D}_i v^j)_{L^2(\Gamma)} \} \\ &= \sum_{i,j=1}^{3} \langle v_F^j - \underline{D}_i^2 v_F^j, v^j \rangle_{\Gamma} = \langle f_F, v \rangle_{\Gamma} = [L[f_F], v]_{T\Gamma} \end{split}$$

for all $v \in H^1(\Gamma, T\Gamma)$, i.e. $F = L[f_F]$ in $H^{-1}(\Gamma, T\Gamma)$. Hence $L: \mathcal{Q} \to H^{-1}(\Gamma, T\Gamma)$ is a bounded, injective, and surjective linear operator. Since its inverse is also bounded by the open mapping theorem, \mathcal{Q} is homeomorphic to $H^{-1}(\Gamma, T\Gamma)$.

In what follows, we identify $L[f] \in H^{-1}(\Gamma, T\Gamma)$ in the proof of Lemma 5.2.7 with equivalence class [f] for $f \in H^{-1}(\Gamma)^3$. We further identify [f] with its representative Pf and write $[Pf, v]_{T\Gamma} = \langle f, v \rangle_{\Gamma}$ for $v \in H^1(\Gamma, T\Gamma)$. When Pf = f in $H^{-1}(\Gamma)^3$, we take f as a representative of [f] instead of Pf. For example, if $\eta \in L^2(\Gamma)$, then

$$\langle \nabla_{\Gamma} \eta, v \rangle_{\Gamma} = -(\eta, \operatorname{div}_{\Gamma} v + (v \cdot n)H)_{L^{2}(\Gamma)} = -(\eta, \operatorname{div}_{\Gamma}(Pv))_{L^{2}(\Gamma)} = \langle P \nabla_{\Gamma} \eta, v \rangle_{\Gamma}$$

for all $v \in H^1(\Gamma)^3$ and thus $P \nabla_{\Gamma} \eta = \nabla_{\Gamma} \eta$ in $H^{-1}(\Gamma)^3$. In this case we have

$$[\nabla_{\Gamma}\eta, v]_{T\Gamma} = -(\eta, \operatorname{div}_{\Gamma}v)_{L^{2}(\Gamma)}, \quad \eta \in L^{2}(\Gamma), \ v \in H^{1}(\Gamma, T\Gamma).$$
(5.2.26)

For $\eta \in W^{1,\infty}(\Gamma)$ and $f \in H^{-1}(\Gamma, T\Gamma)$ we can define $\eta f \in H^{-1}(\Gamma, T\Gamma)$ by

$$[\eta f, v]_{T\Gamma} := [f, \eta v]_{T\Gamma}, \quad v \in H^1(\Gamma, T\Gamma)$$
(5.2.27)

since $\eta v \in H^1(\Gamma, T\Gamma)$ and $\|\eta v\|_{H^1(\Gamma)} \leq c \|\eta\|_{W^{1,\infty}(\Gamma)} \|v\|_{H^1(\Gamma)}$. In Section 5.9 we give the characterization of the annihilators in $H^{-1}(\Gamma)^3$ and $H^{-1}(\Gamma, T\Gamma)$ of solenoidal spaces on Γ .

Since Γ is not of class C^{∞} , the space $C^{\infty}(\Gamma)$ does not make sense and we can not consider distributions on Γ . To consider the time derivative of functions with values in function spaces on Γ , we introduce the notion of distributions with values in a Banach space (see [37,61,64] for details). For T > 0 and a Banach space \mathcal{X} we define $\mathcal{D}'(0,T;\mathcal{X})$ as the space of all continuous linear operators from $C_c^{\infty}(0,T)$, the space of all smooth and compactly supported functions on (0,T), into \mathcal{X} . Here we say that $f: C_c^{\infty}(0,T) \to \mathcal{X}$ is continuous if $\{f(\varphi_k)\}_{k=1}^{\infty}$ converges to $f(\varphi)$ strongly in \mathcal{X} when $\varphi_k, \varphi \in C_c^{\infty}(0,T), k \in \mathbb{N}$ are supported in the same closed interval $[a,b] \subset (0,T)$ and $\{\partial_t^l \varphi_k\}_{k=1}^{\infty}$ converges to $\partial_t^l \varphi$ uniformly on [a,b] for all $l \ge 0$. We consider $L^2(0,T;\mathcal{X}) \subset \mathcal{D}'(0,T;\mathcal{X})$ by identifying $f \in L^2(0,T;\mathcal{X})$ with

$$\hat{f}(\varphi) := \int_0^T \varphi(t) f(t) \, dt \in \mathcal{X}, \quad \varphi \in C_c^\infty(0, T).$$

For $f \in \mathcal{D}'(0,T;\mathcal{X})$ we define the time derivative $\partial_t f \in \mathcal{D}'(0,T;\mathcal{X})$ by $\partial_t f(\varphi) := -f(\partial_t \varphi)$ for $\varphi \in C_c^{\infty}(0,T)$. When $f \in L^2(0,T;\mathcal{X})$, we have

$$\partial_t f(\varphi) = -f(\partial_t \varphi) = -\int_0^T \partial_t \varphi(t) f(t) \, dt \in \mathcal{X}, \quad \varphi \in C_c^\infty(0, T).$$
(5.2.28)

If there exists $\xi \in L^2(0,T;\mathcal{X})$ such that

$$\partial_t f(\varphi) = \xi(\varphi), \quad \text{i.e.} \quad -\int_0^T \partial_t \varphi(t) f(t) \, dt = \int_0^T \varphi(t) \xi(t) \, dt \quad (\text{in } \mathcal{X})$$

for all $\varphi \in C_c^{\infty}(0,T)$, then we write $\partial_t f = \xi \in L^2(0,T;\mathcal{X})$ and define

$$H^1(0,T;\mathcal{X}) := \{ f \in L^2(0,T;\mathcal{X}) \mid \partial_t f \in L^2(0,T;\mathcal{X}) \}.$$

When $q \in L^2(0,T;L^2(\Gamma))$, we can consider the time derivative of $\nabla_{\Gamma} q$ as an element of $\mathcal{D}'(0,T;H^{-1}(\Gamma,T\Gamma))$. Let us show that the time derivative commutes with the tangential gradient in an appropriate sense.

Lemma 5.2.8. Let $q \in L^2(0,T;L^2(\Gamma))$. Then

$$\nabla_{\Gamma}[\partial_t q(\varphi)] = [\partial_t (\nabla_{\Gamma} q)](\varphi) \quad in \quad H^{-1}(\Gamma, T\Gamma)$$

for all $\varphi \in C_c^{\infty}(0,T)$.

Proof. For all $v \in H^1(\Gamma, T\Gamma)$ we observe by (5.2.24) and (5.2.28) that

$$\begin{split} [\nabla_{\Gamma}[\partial_{t}q(\varphi)], v]_{T\Gamma} &= (q(\partial_{t}\varphi), \operatorname{div}_{\Gamma}v)_{L^{2}(\Gamma)} = \int_{0}^{T} \partial_{t}\varphi(t)(q(t), \operatorname{div}_{\Gamma}v)_{L^{2}(\Gamma)} dt \\ &= -\int_{0}^{T} \partial_{t}\varphi(t)[\nabla_{\Gamma}q(t), v]_{T\Gamma} dt = \Big[[\partial_{t}(\nabla_{\Gamma}q)](\varphi), v \Big]_{T\Gamma}. \end{split}$$

Hence the claim is valid.

Let $q \in L^2(0,T;L^2(\Gamma))$. Based on Lemma 5.2.8, we consider the tangential gradient of $\partial_t q \in \mathcal{D}'(0,T;L^2(\Gamma))$ as an element of $\mathcal{D}'(0,T;H^{-1}(\Gamma,T\Gamma))$ given by

$$[\nabla_{\Gamma}(\partial_t q)](\varphi) := \nabla_{\Gamma}[(\partial_t q)(\varphi)] = [\partial_t(\nabla_{\Gamma} q)](\varphi) \in H^{-1}(\Gamma, T\Gamma)$$
(5.2.29)

for $\varphi \in C_c^{\infty}(0,T)$. We use this relation in construction of an associated pressure in the limit equations (see Lemma 5.10.21).

5.2.2 Curved thin domain

From now on, we assume that the closed surface Γ is of class C^5 (except for Section 5.9). Let $g_0, g_1 \in C^4(\Gamma)$ such that

$$g(y) := g_1(y) - g_0(y) \ge c \quad \text{for all} \quad y \in \Gamma$$
(5.2.30)

with a constant c > 0. For $\varepsilon \in (0, 1)$ we define a curved thin domain Ω_{ε} in \mathbb{R}^3 as

$$\Omega_{\varepsilon} := \{ y + rn(y) \mid y \in \Gamma, \, \varepsilon g_0(y) < r < \varepsilon g_1(y) \}.$$
(5.2.31)

We write Γ_{ε} for the boundary of Ω_{ε} . It is the union of the inner boundary Γ_{ε}^{0} and the outer boundary Γ_{ε}^{1} given by

$$\Gamma^i_{\varepsilon} := \{ y + \varepsilon g_i(y) n(y) \mid y \in \Gamma \}, \quad i = 0, 1.$$

Since g_0 and g_1 are bounded on Γ , there exists $\tilde{\varepsilon} \in (0, 1)$ such that $\overline{\Omega}_{\varepsilon} \subset N$ for all $\varepsilon \in (0, \tilde{\varepsilon})$. Replacing g_0 and g_1 by $\tilde{\varepsilon}g_0$ and $\tilde{\varepsilon}g_1$ we may assume $\tilde{\varepsilon} = 1$. Note that the boundary Γ_{ε} is of class C^4 since Γ is of class C^5 , $n \in C^4(\Gamma)^3$, and $g_0, g_1 \in C^4(\Gamma)$. We use this fact in the proof of a uniform a priori estimate for the vector Laplacian (see Section 5.5.4).

Let us give surface quantities on Γ_{ε} . We define vector fields τ_{ε}^{i} and n_{ε}^{i} on Γ as

$$\tau_{\varepsilon}^{i}(y) := \{I_{3} - \varepsilon g_{i}(y)W(y)\}^{-1} \nabla_{\Gamma} g_{i}(y), \qquad (5.2.32)$$

$$n_{\varepsilon}^{i}(y) := (-1)^{i+1} \frac{n(y) - \varepsilon \tau_{\varepsilon}^{i}(y)}{\sqrt{1 + \varepsilon^{2} |\tau_{\varepsilon}^{i}(y)|^{2}}}.$$
(5.2.33)

for $y \in \Gamma$ and i = 0, 1. Note that τ_{ε}^{i} is tangential on Γ by $n \cdot \nabla_{\Gamma} g_{i} = 0$ and Wn = 0. Also, τ_{ε}^{i} and n_{ε}^{i} are bounded on Γ uniformly in ε along with their first and second order tangential derivatives.

Lemma 5.2.9. There exists a constant c > 0 independent of ε such that

$$|\tau_{\varepsilon}^{i}(y)| \le c, \quad |\underline{D}_{k}\tau_{\varepsilon}^{i}(y)| \le c, \quad |\underline{D}_{l}\underline{D}_{k}\tau_{\varepsilon}^{i}(y)| \le c, \tag{5.2.34}$$

$$|\tau_{\varepsilon}^{i}(y) - \nabla_{\Gamma} g_{i}(y)| \le c\varepsilon, \quad |\nabla_{\Gamma} \tau_{\varepsilon}^{i}(y) - \nabla_{\Gamma}^{2} g_{i}(y)| \le c\varepsilon$$
(5.2.35)

for all $y \in \Gamma$, i = 0, 1, and k, l = 1, 2, 3. We also have

$$|n_{\varepsilon}^{i}| = 1, \quad |\underline{D}_{k} n_{\varepsilon}^{i}(y)| \le c, \quad |\underline{D}_{l} \underline{D}_{k} n_{\varepsilon}^{i}(y)| \le c, \tag{5.2.36}$$

$$|n_{\varepsilon}^{0}(y) + n_{\varepsilon}^{1}(y)| \le c\varepsilon, \quad |\nabla_{\Gamma} n_{\varepsilon}^{0}(y) + \nabla_{\Gamma} n_{\varepsilon}^{1}(y)| \le c\varepsilon$$
(5.2.37)

for all $y \in \Gamma$, i = 0, 1, and k, l = 1, 2, 3.

Let n_{ε} be the unit outward normal vector field of Γ_{ε} . For i = 0, 1 the direction of n_{ε} on Γ_{ε}^{i} is the same as that of $(-1)^{i+1}\bar{n}$ since the signed distance function d from Γ increases in the direction of n.

Lemma 5.2.10. The unit outward normal vector field n_{ε} of Γ_{ε} is given by

$$n_{\varepsilon}(x) = \bar{n}_{\varepsilon}^{i}(x), \quad x \in \Gamma_{\varepsilon}^{i}, \, i = 0, 1.$$
(5.2.38)

Here $\bar{n}_{\varepsilon}^{i} = n_{\varepsilon}^{i} \circ \pi$ is the constant extension of the vector field n_{ε}^{i} given by (5.2.33).

The proofs of Lemmas 5.2.9 and 5.2.10 are given in Appendix 5.B.

As in the case of the surface Γ , we use n_{ε} to define the orthogonal projections $P_{\varepsilon} := I_3 - n_{\varepsilon} \otimes n_{\varepsilon}$ and $Q_{\varepsilon} := n_{\varepsilon} \otimes n_{\varepsilon}$ onto the tangent plane and the normal direction of Γ_{ε} , and the tangential gradient and the tangential derivatives

$$\nabla_{\Gamma_{\varepsilon}}\varphi := P_{\varepsilon}\nabla\tilde{\varphi}, \quad \underline{D}_{i}^{\varepsilon}\varphi := \sum_{j=1}^{3} [P_{\varepsilon}]_{ij}\partial_{j}\tilde{\varphi} \quad \text{on} \quad \Gamma_{\varepsilon}, \, i = 1, 2, 3$$

for $\varphi \in C^1(\Gamma_{\varepsilon})$, where $\tilde{\varphi}$ is an arbitrary C^1 -extension of φ to an open neighborhood of Γ_{ε} with $\tilde{\varphi}|_{\Gamma_{\varepsilon}} = \varphi$. For $u \in C^1(\Gamma_{\varepsilon})^3$ we define the tangential gradient matrix and the surface divergence of u as

$$\nabla_{\Gamma_{\varepsilon}} u := \begin{pmatrix} \underline{D}_{1}^{\varepsilon} u_{1} & \underline{D}_{1}^{\varepsilon} u_{2} & \underline{D}_{1}^{\varepsilon} u_{3} \\ \underline{D}_{2}^{\varepsilon} u_{1} & \underline{D}_{2}^{\varepsilon} u_{2} & \underline{D}_{2}^{\varepsilon} u_{3} \\ \underline{D}_{3}^{\varepsilon} u_{1} & \underline{D}_{3}^{\varepsilon} u_{2} & \underline{D}_{3}^{\varepsilon} u_{3} \end{pmatrix}, \quad \operatorname{div}_{\Gamma_{\varepsilon}} u := \operatorname{tr}[\nabla_{\Gamma_{\varepsilon}} u] = \sum_{i=1}^{3} \underline{D}_{i}^{\varepsilon} u_{i} \quad \text{on} \quad \Gamma_{\varepsilon}.$$

The Weingarten map W_{ε} and (twice) the mean curvature H_{ε} of Γ_{ε} are given by

$$W_{\varepsilon} := -\nabla_{\Gamma_{\varepsilon}} n_{\varepsilon}, \quad H_{\varepsilon} := -\mathrm{div}_{\Gamma_{\varepsilon}} n_{\varepsilon} = \mathrm{tr}[W_{\varepsilon}] \quad \text{on} \quad \Gamma_{\varepsilon}$$

Note that, as in the case of Γ , the matrices P_{ε} , Q_{ε} , and W_{ε} are symmetric and

$$\nabla_{\Gamma_{\varepsilon}} u = P_{\varepsilon} \nabla \tilde{u}, \quad P_{\varepsilon} W_{\varepsilon} = W_{\varepsilon} P_{\varepsilon} = W_{\varepsilon} \quad \text{on} \quad \Gamma_{\varepsilon}$$
(5.2.39)

for $u \in C^1(\Gamma_{\varepsilon})^3$, where \tilde{u} is an arbitrary C^1 -extension of u to an open neighborhood of Γ_{ε} with $\tilde{u}|_{\Gamma_{\varepsilon}} = u$. We also define the weak tangential derivatives of functions on Γ_{ε} and the Sobolev spaces $W^{m,p}(\Gamma_{\varepsilon})$ for m = 1, 2 and $p \in [1, \infty)$ as in Section 5.2.1.

Since the unit outward normal n_{ε} to Γ_{ε} has the expression (5.2.33), we can compare the surface quantities on Γ_{ε} with those on Γ .

Lemma 5.2.11. There exists a constant c > 0 independent of ε such that

$$\left| n_{\varepsilon}(x) - (-1)^{i+1} \left\{ \bar{n}(x) - \varepsilon \overline{\nabla_{\Gamma} g_i}(x) \right\} \right| \le c\varepsilon^2, \tag{5.2.40}$$

$$|P_{\varepsilon}(x) - \overline{P}(x)| \le c\varepsilon, \quad |Q_{\varepsilon}(x) - \overline{Q}(x)| \le c\varepsilon,$$
(5.2.41)

$$W_{\varepsilon}(x) - (-1)^{i+1}\overline{W}(x) \le c\varepsilon, \quad |H_{\varepsilon}(x) - (-1)^{i+1}\overline{H}(x)| \le c\varepsilon, \quad (5.2.42)$$

$$\left|\underline{D}_{j}^{\varepsilon}W_{\varepsilon}(x) - (-1)^{i+1}\overline{\underline{D}_{j}W}(x)\right| \le c\varepsilon \tag{5.2.43}$$

for all $x \in \Gamma_{\varepsilon}^{i}$, i = 0, 1, and j = 1, 2, 3.

From Lemma 5.2.11 it immediately follows that W_{ε} , H_{ε} , and $\underline{D}_{j}^{\varepsilon}W_{\varepsilon}$, j = 1, 2, 3 are uniformly bounded in ε on Γ_{ε} (note that $|n_{\varepsilon}| = |P_{\varepsilon}| = |Q_{\varepsilon}| = 1$ on Γ_{ε}). Moreover, we can compare the surface quantities on the inner and outer boundaries.

Lemma 5.2.12. There exists a constant c > 0 independent of ε such that

$$|F_{\varepsilon}(y + \varepsilon g_1(y)n(y)) - F_{\varepsilon}(y + \varepsilon g_0(y)n(y))| \le c\varepsilon, \qquad (5.2.44)$$

$$G_{\varepsilon}(y + \varepsilon g_1(y)n(y)) + G_{\varepsilon}(y + \varepsilon g_0(y)n(y))| \le c\varepsilon$$
(5.2.45)

for all $y \in \Gamma$, where $F_{\varepsilon} = P_{\varepsilon}, Q_{\varepsilon}$ and $G_{\varepsilon} = W_{\varepsilon}, H_{\varepsilon}, \underline{D}_{j}^{\varepsilon}W_{\varepsilon}$ with j = 1, 2, 3.

The proofs of Lemmas 5.2.11 and 5.2.12 are given in Appendix 5.B.

Next we give transformation formulas of integrals over Ω_{ε} and Γ_{ε} . For functions φ on Ω_{ε} and η on Γ^{i}_{ε} , i = 0, 1 we use the notations

$$\varphi^{\sharp}(y,r) := \varphi(y + rn(y)), \qquad y \in \Gamma, \, r \in (\varepsilon g_0(y), \varepsilon g_1(y)), \tag{5.2.46}$$

$$\eta_i^{\sharp}(y) := \eta(y + \varepsilon g_i(y)n(y)), \quad y \in \Gamma.$$
(5.2.47)

Let J = J(y, r) be a function given by

$$J(y,r) := \det[I_3 - rW(y)] = \{1 - r\kappa_1(y)\}\{1 - r\kappa_2(y)\}$$
(5.2.48)

for $y \in \Gamma$ and $r \in (-\delta, \delta)$. By (5.2.2) and $\kappa_1, \kappa_2 \in C^3(\Gamma)$ we have

$$c^{-1} \le J(y,r) \le c, \quad |\nabla_{\Gamma} J(y,r)| \le c, \quad \left|\frac{\partial J}{\partial r}(y,r)\right| \le c$$
 (5.2.49)

for all $y \in \Gamma$ and $r \in (-\delta, \delta)$ (here $\nabla_{\Gamma} J$ stands for the tangential gradient of J with respect to $y \in \Gamma$). Also, we easily observe that

$$|J(y,r) - 1| \le c\varepsilon \quad \text{for all} \quad y \in \Gamma, \, r \in [\varepsilon g_0(y), \varepsilon g_1(y)].$$
(5.2.50)

The function J is the Jacobian appearing in the change of variables formula

$$\int_{\Omega_{\varepsilon}} \varphi(x) \, dx = \int_{\Gamma} \int_{\varepsilon g_0(y)}^{\varepsilon g_1(y)} \varphi(y + rn(y)) J(y, r) \, dr \, d\mathcal{H}^2(y) \tag{5.2.51}$$

for a function φ on Ω_{ε} (see e.g. [15, Section 14.6]). The formula (5.2.51) can be seen as a co-area formula. From (5.2.49) and (5.2.51) it immediately follows that

$$c^{-1} \|\varphi\|_{L^p(\Omega_{\varepsilon})}^p \le \int_{\Gamma} \int_{\varepsilon g_0(y)}^{\varepsilon g_1(y)} |\varphi^{\sharp}(y,r)|^p \, dr \, d\mathcal{H}^2(y) \le c \|\varphi\|_{L^p(\Omega_{\varepsilon})}^p \tag{5.2.52}$$

for all $p \in [1, \infty)$ and $\varphi \in L^p(\Omega_{\varepsilon})$, where we used the notation (5.2.46). We frequently use this inequality in the sequel.

Lemma 5.2.13. Let η be a function on Γ and $\bar{\eta} := \eta \circ \pi$ its constant extension in the normal direction of Γ . Then $\eta \in L^p(\Gamma)$, $p \in [1, \infty)$ if and only if $\bar{\eta} \in L^p(\Omega_{\varepsilon})$. Moreover, there exists a constant c > 0 independent of ε and η such that

$$c^{-1}\varepsilon^{1/p} \|\eta\|_{L^{p}(\Gamma)} \leq \|\bar{\eta}\|_{L^{p}(\Omega_{\varepsilon})} \leq c\varepsilon^{1/p} \|\eta\|_{L^{p}(\Gamma)}.$$
(5.2.53)

Also, $\eta \in W^{1,p}(\Gamma)$ if and only if $\bar{\eta} \in W^{1,p}(\Omega_{\varepsilon})$ and we have

$$c^{-1}\varepsilon^{1/p} \|\nabla_{\Gamma}\eta\|_{L^{p}(\Gamma)} \leq \|\nabla\bar{\eta}\|_{L^{p}(\Omega_{\varepsilon})} \leq c\varepsilon^{1/p} \|\nabla_{\Gamma}\eta\|_{L^{p}(\Gamma)}$$
(5.2.54)

and therefore

$$c^{-1}\varepsilon^{1/p} \|\eta\|_{W^{1,p}(\Gamma)} \le \|\bar{\eta}\|_{W^{1,p}(\Omega_{\varepsilon})} \le c\varepsilon^{1/p} \|\eta\|_{W^{1,p}(\Gamma)}.$$
(5.2.55)

Proof. The change of variables formula (5.2.51) implies that

$$\|\bar{\eta}\|_{L^p(\Omega_{\varepsilon})}^p = \int_{\Gamma} |\eta(y)|^p \left(\int_{\varepsilon g_0(y)}^{\varepsilon g_1(y)} J(y,r) \, dr\right) d\mathcal{H}^2(y).$$

Hence the inequality (5.2.53) follows from (5.2.30) and (5.2.49). Similarly, we get (5.2.54) by (5.2.13), (5.2.30), (5.2.49), and (5.2.51).

Lemma 5.2.14. For $p \in [1, \infty)$ let $\eta \in W^{2,p}(\Gamma)$. Then $\bar{\eta} := \eta \circ \pi \in W^{2,p}(\Omega_{\varepsilon})$ and

$$\|\bar{\eta}\|_{W^{2,p}(\Omega_{\varepsilon})} \le c\varepsilon^{1/p} \|\eta\|_{W^{2,p}(\Gamma)}$$
(5.2.56)

with a constant c > 0 independent of ε and η .

Proof. From (5.2.15) and (5.2.53) it follows that

$$\|\nabla^2 \bar{\eta}\|_{L^p(\Omega_{\varepsilon})} \le c \left(\left\|\overline{\nabla_{\Gamma} \eta}\right\|_{L^p(\Omega_{\varepsilon})} + \left\|\overline{\nabla_{\Gamma}^2 \eta}\right\|_{L^p(\Omega_{\varepsilon})} \right) \le c \varepsilon^{1/p} \|\eta\|_{W^{2,p}(\Gamma)}.$$

Combining this inequality with (5.2.55) we obtain (5.2.56).

We also give a change of variables formula for integrals over Γ_{ε} .

Lemma 5.2.15. For $\varphi \in L^1(\Gamma^i_{\varepsilon})$, i = 0, 1 let φ^{\sharp}_i be given by (5.2.47). Then

$$\int_{\Gamma_{\varepsilon}^{i}} \varphi(x) \, d\mathcal{H}^{2}(x) = \int_{\Gamma} \varphi_{i}^{\sharp}(y) J(y, \varepsilon g_{i}(y)) \sqrt{1 + \varepsilon^{2} |\tau_{\varepsilon}^{i}(y)|^{2}} \, d\mathcal{H}^{2}(y), \tag{5.2.57}$$

where τ_{ε}^{i} is given by (5.2.32). Moreover, if $\varphi \in L^{p}(\Gamma_{\varepsilon}^{i})$, $p \in [1, \infty)$ then $\varphi_{i}^{\sharp} \in L^{p}(\Gamma)$ and there exists a constant c > 0 independent of ε such that

$$c^{-1} \|\varphi\|_{L^p(\Gamma^i_{\varepsilon})} \le \|\varphi^{\sharp}_i\|_{L^p(\Gamma)} \le c \|\varphi\|_{L^p(\Gamma^i_{\varepsilon})}.$$
(5.2.58)

Proof. In Lemma 5.B.2 we show a change of variables formula

$$\int_{\Gamma_h} \varphi(x) \, d\mathcal{H}^2(x) = \int_{\Gamma} \varphi^{\sharp}(y) J(y, h(y)) \sqrt{1 + |\tau_h(y)|^2} \, d\mathcal{H}^2(y)$$

for an integrable function φ on a parametrized surface $\Gamma_h := \{y + h(y)n(y) \mid y \in \Gamma\}$, where $h \in C^1(\Gamma)$ satisfies $|h| < \delta$ on Γ and

$$\varphi^{\sharp}(y) := \varphi(y + h(y)n(y)), \quad \tau_h(y) := \{I_3 - h(y)W(y)\}^{-1} \nabla_{\Gamma} h(y), \quad y \in \Gamma.$$

Setting $h = \varepsilon g_i$, i = 0, 1 in the above formula we obtain (5.2.57). Also, (5.2.58) follows from the formula (5.2.57) and the inequalities (5.2.34) and (5.2.49).

5.3 Fundamental tools for analysis

5.3.1 Sobolev inequalities

Let us give several Sobolev inequalities on Γ and Ω_{ε} . First we prove Ladyzhenskaya's inequality on the two-dimensional closed surface Γ .

Lemma 5.3.1. There exists a constant c > 0 such that

$$\|\eta\|_{L^4(\Gamma)} \le c \|\eta\|_{L^2(\Gamma)}^{1/2} \|\eta\|_{H^1(\Gamma)}^{1/2}$$
(5.3.1)

for all $\eta \in H^1(\Gamma)$.

Proof. Since Γ is compact and without boundary, by a standard localization argument with a partition of unity of Γ it is sufficient to prove

$$\|\eta\|_{L^4(\mu(U))} \le c \|\eta\|_{L^2(\mu(U))}^{1/2} \|\eta\|_{H^1(\mu(U))}^{1/2}$$
(5.3.2)

for a bounded open set U in \mathbb{R}^2 , a local parametrization $\mu: U \to \Gamma$, and $\eta \in H^1(\Gamma)$ compactly supported in $\mu(U)$. For such η , the function $\tilde{\eta} := \eta \circ \mu$ is in $H^1(\mathbb{R}^2)$ and compactly supported in U. Hence Ladyzhenskaya's inequality on \mathbb{R}^2 (see [35, Lemma 1 in Chapter 1, Section 1.1]) yields

$$\|\tilde{\eta}\|_{L^4(U)} \le \sqrt{2} \|\tilde{\eta}\|_{L^2(U)}^{1/2} \|\nabla_s \tilde{\eta}\|_{L^2(U)}^{1/2}, \tag{5.3.3}$$

where ∇_s is the gradient operator in $s \in \mathbb{R}^2$. To deduce (5.3.2) from (5.3.3) we set

$$\nabla_s \mu := \begin{pmatrix} \partial_{s_1} \mu_1 & \partial_{s_1} \mu_2 & \partial_{s_1} \mu_3 \\ \partial_{s_2} \mu_1 & \partial_{s_2} \mu_2 & \partial_{s_2} \mu_3 \end{pmatrix}, \quad \theta := \nabla_s \mu (\nabla_s \mu)^T$$

and recall that integrals over the surface are given by

$$\|\eta\|_{L^p(\mu(U))}^p = \int_U |\tilde{\eta}|^p \sqrt{\det\theta} \, ds, \quad p = 2, 4,$$
$$\|\nabla_\Gamma \eta\|_{L^2(\mu(U))}^2 = \int_U |(\nabla_\Gamma \eta) \circ \mu|^2 \sqrt{\det\theta} \, ds.$$

Since μ is of class C^5 (note that Γ is of class C^5), the determinant of θ is continuous and does not vanish on U. In particular, it is bounded from above and below by positive constants on the support of $\tilde{\eta}$ since it is a compact subset of U. Hence

$$c^{-1} \|\eta\|_{L^{p}(\mu(U))} \le \|\tilde{\eta}\|_{L^{p}(U)} \le c \|\eta\|_{L^{p}(\mu(U))}, \quad p = 2, 4$$
(5.3.4)

with a constant c > 0. Also, differentiating both sides of $\tilde{\eta}(s) = \eta(\mu(s)) = \bar{\eta}(\mu(s))$ with respect to s_i , i = 1, 2 and using (5.2.4) (note that $\mu(s) \in \Gamma$) we get

$$\partial_{s_i}\tilde{\eta}(s) = \partial_{s_i}\mu(s) \cdot \nabla \bar{\eta}(\mu(s)) = \partial_{s_i}\mu(s) \cdot \nabla_{\Gamma}\eta(\mu(s)), \quad s \in U.$$

From this equality and the fact that μ is of class C^5 and the determinant of θ is bounded from below by a positive constant on the support of $\tilde{\eta}$ we deduce that

$$\|\nabla_s \tilde{\eta}\|_{L^2(U)}^2 \le c \int_U |(\nabla_\Gamma \eta) \circ \mu|^2 \, ds \le c \|\nabla_\Gamma \eta\|_{L^2(\mu(U))}^2$$

Applying this inequality and (5.3.4) to (5.3.3) we obtain (5.3.2).

Next we give Poincaré type inequalities on the curved thin domain Ω_{ε} . By ∂_n we denote the directional derivative in the normal direction of Γ , i.e. for a function φ on Ω_{ε} and $x \in \Omega_{\varepsilon}$ we set

$$\partial_n \varphi(x) := \left(\bar{n}(x) \cdot \nabla\right) \varphi(x) = \frac{d}{dr} \left(\varphi(y + rn(y)) \right) \Big|_{r=d(x)} \quad (y = \pi(x) \in \Gamma).$$
(5.3.5)

Note that for the constant extension of a function η on Γ we have

$$\partial_n \bar{\eta}(x) = (\bar{n}(x) \cdot \nabla) \bar{\eta}(x) = 0, \quad x \in \Omega_{\varepsilon}.$$
(5.3.6)

Lemma 5.3.2. Let $\varphi \in W^{1,p}(\Omega_{\varepsilon})$ with $p \in [1, \infty)$. There exists a constant c > 0 independent of ε such that

$$\|\varphi\|_{L^{p}(\Omega_{\varepsilon})} \leq c \left(\varepsilon^{1/p} \|\varphi\|_{L^{p}(\Gamma_{\varepsilon}^{i})} + \varepsilon \|\partial_{n}\varphi\|_{L^{p}(\Omega_{\varepsilon})}\right), \quad i = 0, 1,$$
(5.3.7)

$$\|\varphi\|_{L^p(\Gamma^i_{\varepsilon})} \le c \left(\varepsilon^{-1/p} \|\varphi\|_{L^p(\Omega_{\varepsilon})} + \varepsilon^{1-1/p} \|\partial_n \varphi\|_{L^p(\Omega_{\varepsilon})}\right), \quad i = 0, 1.$$
(5.3.8)

Proof. We show (5.3.7) and (5.3.8) for i = 0. Their proofs for i = 1 are the same. We use the notations (5.2.46) and (5.2.47). Let $y \in \Gamma$ and $r \in (\varepsilon g_0(y), \varepsilon g_1(y))$. Since $\partial \varphi^{\sharp} / \partial r = (\partial_n \varphi)^{\sharp}$ by (5.3.5), we have

$$\varphi^{\sharp}(y,r) = \varphi^{\sharp}(y,\varepsilon g_0(y)) + \int_{\varepsilon g_0(y)}^r (\partial_n \varphi)^{\sharp}(y,\tilde{r}) \, d\tilde{r}.$$
(5.3.9)

From (5.3.9) and Hölder's inequality it follows that

$$\begin{aligned} |\varphi^{\sharp}(y,r)| &\leq |\varphi_{0}^{\sharp}(y)| + \int_{\varepsilon g_{0}(y)}^{\varepsilon g_{1}(y)} |(\partial_{n}\varphi)^{\sharp}(y,\tilde{r})| \, d\tilde{r} \\ &\leq |\varphi_{0}^{\sharp}(y)| + c\varepsilon^{1-1/p} \left(\int_{\varepsilon g_{0}(y)}^{\varepsilon g_{1}(y)} |(\partial_{n}\varphi)^{\sharp}(y,\tilde{r})|^{p} \, d\tilde{r} \right)^{1/p} \end{aligned}$$

Here $\varphi_0^{\sharp}(y) = \varphi^{\sharp}(y, \varepsilon g_0(y))$. Noting that the right-hand side is independent of r, we integrate the *p*-th power of both sides of the above inequality with respect to r to get

$$\int_{\varepsilon g_0(y)}^{\varepsilon g_1(y)} |\varphi^{\sharp}(y,r)|^p \, dr \le c \left(\varepsilon |\varphi_0^{\sharp}(y)|^p + \varepsilon^p \int_{\varepsilon g_0(y)}^{\varepsilon g_1(y)} |(\partial_n \varphi)^{\sharp}(y,\tilde{r})|^p \, d\tilde{r} \right). \tag{5.3.10}$$

Hence the inequalities (5.2.52) and (5.3.10) yield that

$$\|\varphi\|_{L^p(\Omega_{\varepsilon})}^p \le c \int_{\Gamma} \int_{\varepsilon g_0(y)}^{\varepsilon g_1(y)} |\varphi^{\sharp}(y,r)|^p \, dr \, d\mathcal{H}^2(y) \le c \left(\varepsilon \|\varphi_0^{\sharp}\|_{L^p(\Gamma)}^p + \varepsilon^p \|\partial_n \varphi\|_{L^p(\Omega_{\varepsilon})}^p\right).$$

Applying (5.2.58) to the first term on the right-hand side we obtain (5.3.7).

Next let us prove (5.3.8). From (5.3.9) we deduce that

$$|\varphi_0^{\sharp}(y)|^p \le c \left(\varepsilon^{-1} \int_{\varepsilon g_0(y)}^{\varepsilon g_1(y)} |\varphi^{\sharp}(y,r)|^p \, dr + \varepsilon^{p-1} \int_{\varepsilon g_0(y)}^{\varepsilon g_1(y)} |(\partial_n \varphi)^{\sharp}(y,\tilde{r})|^p \, d\tilde{r}\right)$$

as in the proof of (5.3.10). This inequality and (5.2.52) imply that

$$\|\varphi_0^{\sharp}\|_{L^p(\Gamma)} \le c \left(\varepsilon^{-1/p} \|\varphi\|_{L^p(\Omega_{\varepsilon})} + \varepsilon^{1-1/p} \|\partial_n \varphi\|_{L^p(\Omega_{\varepsilon})}\right)$$

Hence we apply (5.2.58) to the left-hand side of the above inequality to get (5.3.8).

We also show Agmon's inequality on Ω_{ε} , which gives an estimate for the $L^{\infty}(\Omega_{\varepsilon})$ -norm of a function in $H^2(\Omega_{\varepsilon})$ with explicit dependence on ε of a bound.

Lemma 5.3.3. There exists a constant c > 0 independent of ε such that

$$\begin{aligned} \|\varphi\|_{L^{\infty}(\Omega_{\varepsilon})} &\leq c\varepsilon^{-1/2} \|\varphi\|_{L^{2}(\Omega_{\varepsilon})}^{1/4} \|\varphi\|_{H^{2}(\Omega_{\varepsilon})}^{1/2} \\ &\times \left(\|\varphi\|_{L^{2}(\Omega_{\varepsilon})} + \varepsilon \|\partial_{n}\varphi\|_{L^{2}(\Omega_{\varepsilon})} + \varepsilon^{2} \|\partial_{n}^{2}\varphi\|_{L^{2}(\Omega_{\varepsilon})}\right)^{1/4} \quad (5.3.11)\end{aligned}$$

for all $\varphi \in H^2(\Omega_{\varepsilon})$.

Proof. We use the anisotropic Agmon inequality (see [65, Proposition 2.2])

$$\|\Phi\|_{L^{\infty}(V)} \le c \|\Phi\|_{L^{2}(V)}^{1/4} \prod_{i=1}^{3} \left(\|\Phi\|_{L^{2}(V)} + \|\partial_{i}\Phi\|_{L^{2}(V)} + \|\partial_{i}^{2}\Phi\|_{L^{2}(V)}\right)^{1/4}$$
(5.3.12)

for $\Phi \in H^2(V)$ with $V = (0, 1)^3$. To this end, we localize φ by using a partition of unity on Γ and transform it into a function on V. Since Γ is compact, we can take a finite number of open sets U_k in \mathbb{R}^2 and local parametrizations $\mu_k \colon U_k \to \Gamma, \ k = 1, \ldots, k_0$ such that $\{\mu_k(U_k)\}_{k=1}^{k_0}$ is an open covering of Γ . Then setting

$$\zeta_k(s) = \mu_k(s') + \varepsilon \{ (1 - s_3)g_0(\mu_k(s')) + s_3g_1(\mu_k(s')) \} n(\mu_k(s'))$$

for $s = (s', s_3) \in V_k := U_k \times (0, 1)$ we see that $\{\zeta_k(V_k)\}_{k=1}^{k_0}$ is an open covering of Ω_{ε} . Let $\{\eta_k\}_{k=1}^{k_0}$ be a partition of unity on Γ subordinate to the covering $\{\mu_k(U_k)\}_{k=1}^{k_0}$ and $\bar{\eta}_k := \eta_k \circ \pi$ the constant extension of η_k . Then $\{\bar{\eta}_k\}_{k=1}^{k_0}$ is a partition of unity on Ω_{ε} subordinate to the covering $\{\zeta_k(V_k)\}_{k=1}^{k_0}$. Hence to prove (5.3.11) it is sufficient to show

$$\begin{aligned} \|\varphi_{k}\|_{L^{\infty}(\zeta_{k}(V_{k}))} &\leq c\varepsilon^{-1/2} \|\varphi_{k}\|_{L^{2}(\zeta_{k}(V_{k}))}^{1/4} \|\varphi_{k}\|_{H^{2}(\zeta_{k}(V_{k}))}^{1/2} \\ &\times \left(\|\varphi_{k}\|_{L^{2}(\zeta_{k}(V_{k}))} + \varepsilon\|\partial_{n}\varphi_{k}\|_{L^{2}(\zeta_{k}(V_{k}))} + \varepsilon^{2} \|\partial_{n}^{2}\varphi_{k}\|_{L^{2}(\zeta_{k}(V_{k}))}\right)^{1/4} \quad (5.3.13)\end{aligned}$$

for each $k = 1, \ldots, k_0$, where $\varphi_k := \bar{\eta}_k \varphi$ (note that $\partial_n \varphi_k = \bar{\eta}_k \partial_n \varphi$ by $\partial_n \bar{\eta}_k = 0$). Let us prove (5.3.13). Hereafter we suppress the index k. By taking the open subset U of \mathbb{R}^2 small and scaling it, we may assume $U = (0, 1)^2$ and $V = (0, 1)^3$. Since Γ is of class C^5 , the local parametrization μ is of class C^5 and thus the mapping

$$\zeta(s) = \mu(s') + \varepsilon \{ (1 - s_3)\bar{g}_0(\mu(s')) + s_3\bar{g}_1(\mu(s')) \} \bar{n}(\mu(s')), \quad s = (s', s_3) \in V$$

is of class C^4 by $g_0, g_1 \in C^4(\Gamma)$ and $n \in C^4(\Gamma)^3$ (here $\bar{g}_i = g_i \circ \pi$ and $\bar{n} = n \circ \pi$). We differentiate $\zeta(s)$ and apply (5.2.4) and (5.2.16) with $y = \mu(s') \in \Gamma$ to get

$$\partial_{s_i}\zeta(s) = \left[I_3 - \varepsilon \left\{ (1 - s_3)g_0(\mu(s')) + s_3g_1(\mu(s')) \right\} W(\mu(s')) \right] \partial_{s_i}\mu(s') \\ + \varepsilon \partial_{s_i}\mu(s') \cdot \left\{ (1 - s_3)\nabla_{\Gamma}g_0(\mu(s')) + s_3\nabla_{\Gamma}g_1(\mu(s')) \right\} n(\mu(s')),$$
(5.3.14)
$$\partial_{s_3}\zeta(s) = \varepsilon g(\mu(s'))n(\mu(s'))$$

for $s = (s', s_3) \in V$ and i = 1, 2. From these formulas it follows that

$$\det \nabla_s \zeta(s) = \varepsilon g(\mu(s')) J(\mu(s'), h_\varepsilon(s)) \sqrt{\det \theta(s')}, \quad s = (s', s_3) \in V, \tag{5.3.15}$$

where $\nabla_s \zeta$ is the gradient matrix of ζ in s and

$$h_{\varepsilon}(s) := \varepsilon \{ (1 - s_3)g_0(\mu(s')) + s_3g_1(\mu(s')) \},$$

$$\theta(s') := \nabla_{s'}\mu(s')\nabla_{s'}\mu(s')^T, \quad \nabla_{s'}\mu := \begin{pmatrix} \partial_{s_1}\mu_1 & \partial_{s_1}\mu_2 & \partial_{s_1}\mu_3 \\ \partial_{s_2}\mu_1 & \partial_{s_2}\mu_2 & \partial_{s_2}\mu_3 \end{pmatrix}.$$

(We give detailed calculations for (5.3.15) in Appendix 5.B.) Let $\Phi(s) := \varphi(\zeta(s))$ for $s \in V$. Since φ is localized by the constant extension of a cut-off function on Γ , the function Φ is supported in $\mathcal{K} \times (0, 1)$ with some compact subset \mathcal{K} of U. Then the determinant of θ is bounded from below by a positive constant on \mathcal{K} since it is continuous and does not vanish on U (note that the local parametrization μ is of class C^5). By this fact, (5.2.30), and (5.2.49) we have

$$\det \nabla_s \zeta(s) \ge c\varepsilon, \quad s \in \mathcal{K} \times (0, 1) \tag{5.3.16}$$

with a constant c > 0. Since $\Phi = \varphi \circ \zeta$ is supported in $\mathcal{K} \times (0, 1)$ and ζ is bounded on $\mathcal{K} \times (0, 1)$ along with its first and second order derivatives (note that ζ is of class C^4 on V and depends linearly on s_3), we observe by the change of variables formula

$$\int_{\zeta(V)} \varphi(x) \, dx = \int_V \Phi(s) \det \nabla_s \zeta(s) \, ds, \qquad (5.3.17)$$

the inequality (5.3.16), and $\varphi \in H^2(\Omega_{\varepsilon})$ that $\Phi \in H^2(V)$ and

$$\|\Phi\|_{L^{\infty}(V)} = \|\varphi\|_{L^{\infty}(\zeta(V))}, \quad \|\Phi\|_{L^{2}(V)} \le c\varepsilon^{-1/2} \|\varphi\|_{L^{2}(\zeta(V))}.$$
(5.3.18)

Moreover, by the chain rule of differentiation we have

$$\begin{split} \partial_{s_i} \Phi(s) &= \partial_{s_i} \zeta(s) \cdot \nabla \varphi(\zeta(s)), \\ \partial_{s_i}^2 \Phi(s) &= \partial_{s_i}^2 \zeta(s) \cdot \nabla \varphi(\zeta(s)) + \partial_{s_i} \zeta(s) \cdot \nabla^2 \varphi(\zeta(s)) \partial_{s_i} \zeta(s), \\ \partial_{s_3} \Phi(s) &= \varepsilon g(\mu(s')) \partial_n \varphi(\zeta(s)), \quad \partial_{s_3}^2 \Phi(s) = \varepsilon^2 g(\mu(s'))^2 \partial_n^2 \varphi(\zeta(s)) \end{split}$$

for $s = (s', s_3) \in V$ and i = 1, 2, where the last two equalities follow from (5.3.14) and $\partial_n \varphi = (\bar{n} \cdot \nabla) \varphi$. To the above equalities we apply the boundedness of g on Γ and that of the first and second order derivatives of ζ on $\mathcal{K} \times (0, 1)$ to get

$$\begin{aligned} |\partial_{s_i} \Phi(s)| &\leq c |\nabla \varphi(\zeta(s))|, \quad |\partial_{s_i}^2 \Phi(s)| \leq c (|\nabla \varphi(\zeta(s))| + |\nabla^2 \varphi(\zeta(s))|), \quad i = 1, 2, \\ |\partial_{s_i}^k \Phi(s)| &\leq c \varepsilon^k |\partial_n^k \varphi(\zeta(s))|, \quad k = 1, 2 \end{aligned}$$

for $s \in \mathcal{K} \times (0,1)$. Noting that $\Phi = \varphi \circ \zeta$ is supported in $\mathcal{K} \times (0,1)$, we deduce from the above inequalities, (5.3.16), and (5.3.17) that

$$\|\partial_{s_i}^k \Phi\|_{L^2(V)} \le c\varepsilon^{-1/2} \|\varphi\|_{H^k(\zeta(V))}, \quad \|\partial_{s_3}^k \Phi\|_{L^2(V)} \le c\varepsilon^{k-1/2} \|\partial_n^k \varphi\|_{L^2(\zeta(V))}$$
(5.3.19)

for i, k = 1, 2. Finally, applying the anisotropic Agmon inequality (5.3.12) to $\Phi \in H^2(V)$ and using (5.3.18) and (5.3.19) we obtain (5.3.13).

5.3.2 Consequences of the boundary conditions

In this subsection we derive several properties from the boundary conditions

$$u \cdot n_{\varepsilon} = 0, \tag{5.3.20}$$

$$2\nu P_{\varepsilon} D(u) n_{\varepsilon} + \gamma_{\varepsilon} u = 0, \qquad (5.3.21)$$

where $D(u) =: (\nabla u)_S = {\nabla u + (\nabla u)^T}/2$ is the strain rate tensor. First we consider vector fields satisfying the impermeable boundary condition (5.3.20).

Lemma 5.3.4. For i = 0, 1 let $u \in C(\Gamma_{\varepsilon}^{i})^{3}$ satisfy (5.3.20) on Γ_{ε}^{i} . Then

$$u \cdot \bar{n} = \varepsilon u \cdot \bar{\tau}^i_{\varepsilon}, \quad |u \cdot \bar{n}| \le c\varepsilon |u| \quad on \quad \Gamma^i_{\varepsilon}, \tag{5.3.22}$$

where τ_{ε}^{i} is given by (5.2.32) and c > 0 is a constant independent of ε and u.

Proof. The first equality of (5.3.22) is an immediate consequence of (5.2.33), (5.2.38), and (5.3.20) on Γ_{ε}^{i} . This equality and the first inequality of (5.2.34) implies the second inequality of (5.3.22).

As a consequence of Lemma 5.3.4 we derive Poincaré's inequalities for the normal component (with respect to Γ) of a vector field on Ω_{ε} .

Lemma 5.3.5. Let $p \in [1, \infty)$. There exists c > 0 independent of ε such that

$$\|u \cdot \bar{n}\|_{L^p(\Omega_{\varepsilon})} \le c\varepsilon \|u\|_{W^{1,p}(\Omega_{\varepsilon})}$$
(5.3.23)

for all $u \in W^{1,p}(\Omega_{\varepsilon})^3$ satisfying (5.3.20) on Γ_{ε}^0 or on Γ_{ε}^1 . We also have

$$\left\|\overline{P}\nabla(u\cdot\bar{n})\right\|_{L^{p}(\Omega_{\varepsilon})} \le c\varepsilon \|u\|_{W^{2,p}(\Omega_{\varepsilon})}$$
(5.3.24)

for all $u \in W^{2,p}(\Omega_{\varepsilon})^3$ satisfying (5.3.20) on Γ_{ε}^0 or on Γ_{ε}^1 .

Proof. Let $u \in W^{1,p}(\Omega_{\varepsilon})^3$. We may assume that u satisfies (5.3.20) on Γ_{ε}^0 without loss of generality. By (5.3.6) and (5.3.7) with i = 0,

$$\|u \cdot \bar{n}\|_{L^{p}(\Omega_{\varepsilon})} \leq c \left(\varepsilon^{1/p} \|u \cdot \bar{n}\|_{L^{p}(\Gamma_{\varepsilon}^{0})} + \varepsilon \|\partial_{n}u\|_{L^{p}(\Omega_{\varepsilon})}\right).$$
(5.3.25)

Moreover, we apply the second inequality of (5.3.22) and then use (5.3.8) with i = 0 to the first term on the right-hand side of (5.3.25) to get

$$\|u \cdot \bar{n}\|_{L^p(\Gamma^0_{\varepsilon})} \le c\varepsilon \|u\|_{L^p(\Gamma^0_{\varepsilon})} \le c\varepsilon^{1-1/p} \|u\|_{W^{1,p}(\Omega_{\varepsilon})}.$$
(5.3.26)

Combining (5.3.25) and (5.3.26) we obtain (5.3.23).

Next suppose that $u \in W^{2,p}(\Omega_{\varepsilon})^3$ satisfies (5.3.20) on Γ_{ε}^0 . Noting that

$$\left|\partial_n \left[\overline{P} \nabla (u \cdot \overline{n})\right]\right| \le c(|\nabla u| + |\nabla^2 u|) \quad \text{in} \quad \Omega_{\varepsilon}$$

by (5.2.17) and (5.3.6), we apply (5.3.7) with i = 0 to get

$$\left\|\overline{P}\nabla(u\cdot\bar{n})\right\|_{L^{p}(\Omega_{\varepsilon})} \leq c\left(\varepsilon^{1/p}\left\|\overline{P}\nabla(u\cdot\bar{n})\right\|_{L^{p}(\Gamma_{\varepsilon}^{0})} + \varepsilon\|u\|_{W^{2,p}(\Omega_{\varepsilon})}\right)$$
(5.3.27)

Since the tangential gradient on Γ_{ε} depends only on the value of a function on Γ_{ε} , we see by (5.2.39) and the first equality of (5.3.22) that

$$\overline{P}\nabla(u\cdot\bar{n}) = \nabla_{\Gamma_{\varepsilon}}(u\cdot\bar{n}) + \left(\overline{P} - P_{\varepsilon}\right)\nabla(u\cdot\bar{n}) = \varepsilon\nabla_{\Gamma_{\varepsilon}}(u\cdot\bar{\tau}_{\varepsilon}^{0}) + \left(\overline{P} - P_{\varepsilon}\right)\nabla(u\cdot\bar{n})$$
$$= \varepsilon P_{\varepsilon}\nabla(u\cdot\bar{\tau}_{\varepsilon}^{0}) + \left(\overline{P} - P_{\varepsilon}\right)\nabla(u\cdot\bar{n})$$

on Γ_{ε}^{0} . By this formula, (5.2.13), (5.2.17), (5.2.34), and (5.2.41) we have

$$\left|\overline{P}\nabla(u\cdot\bar{n})\right| \leq c\varepsilon(|u|+|\nabla u|) \quad \text{on} \quad \Gamma^0_{\varepsilon}.$$

From this inequality and (5.3.8) it follows that

$$\left\|\overline{P}\nabla(u\cdot\bar{n})\right\|_{L^{p}(\Gamma^{0}_{\varepsilon})} \leq c\varepsilon\left(\|u\|_{L^{p}(\Gamma^{0}_{\varepsilon})} + \|\nabla u\|_{L^{p}(\Gamma^{0}_{\varepsilon})}\right) \leq c\varepsilon^{1-1/p}\|u\|_{W^{2,p}(\Omega_{\varepsilon})}.$$

Applying this inequality to the right-hand side of (5.3.27) we obtain (5.3.24).

The next lemma gives an expression for the normal component of the directional derivatives on Γ_{ε} for vector fields satisfying the impermeable boundary condition in terms of the Weingarten map of Γ_{ε} .

Lemma 5.3.6. Let i = 0, 1. For $u_1, u_2 \in C^1(\overline{\Omega}_{\varepsilon})^3$ satisfying (5.3.20) on Γ_{ε}^i we have

$$(u_1 \cdot \nabla)u_2 \cdot n_{\varepsilon} = W_{\varepsilon}u_1 \cdot u_2 = u_1 \cdot W_{\varepsilon}u_2 \quad on \quad \Gamma_{\varepsilon}^i.$$
(5.3.28)

Proof. The equality $u_1 \cdot n_{\varepsilon} = 0$ on Γ^i_{ε} means that u_1 is tangential on Γ^i_{ε} . Hence

$$(u_1 \cdot \nabla)u_2 \cdot n_{\varepsilon} = (u_1 \cdot \nabla_{\Gamma_{\varepsilon}})u_2 \cdot n_{\varepsilon} = u_1 \cdot \nabla_{\Gamma_{\varepsilon}}(u_2 \cdot n_{\varepsilon}) - u_2 \cdot (u_1 \cdot \nabla_{\Gamma_{\varepsilon}})n_{\varepsilon}$$

on Γ_{ε}^{i} . The first term on the right-hand side vanishes by $u_{2} \cdot n_{\varepsilon} = 0$ on Γ_{ε}^{i} (note that the tangential gradient on Γ_{ε}^{i} depends only on the values of a function on Γ_{ε}^{i}). Also, by $-\nabla_{\Gamma_{\varepsilon}}n_{\varepsilon} = W_{\varepsilon} = W_{\varepsilon}^{T}$ we have

$$(u_1 \cdot \nabla_{\Gamma_{\varepsilon}})n_{\varepsilon} = -W_{\varepsilon}^T u_1 = -W_{\varepsilon}u_1 \quad \text{on} \quad \Gamma_{\varepsilon}^i.$$

Combining the above two equalities we obtain (5.3.28).

Note that $(u_1 \cdot \nabla)u_2 \cdot n_{\varepsilon}$ can be expressed without using the derivatives of u_1 and u_2 by (5.3.28). We use this fact in the analysis of boundary integrals, see Lemma 5.4.1.

Next we prove formulas for vector fields satisfying the slip boundary conditions (5.3.20)–(5.3.21).

Lemma 5.3.7. Let i = 0, 1. For $u \in C^2(\overline{\Omega}_{\varepsilon})^3$ satisfying (5.3.20) - (5.3.21) on Γ_{ε}^i we have

$$P_{\varepsilon}(n_{\varepsilon} \cdot \nabla)u = -W_{\varepsilon}u - \frac{\gamma_{\varepsilon}}{\nu}u \quad on \quad \Gamma^{i}_{\varepsilon}, \qquad (5.3.29)$$

$$n_{\varepsilon} \times \operatorname{curl} u = -n_{\varepsilon} \times \left\{ n_{\varepsilon} \times \left(2W_{\varepsilon}u + \frac{\gamma_{\varepsilon}}{\nu}u \right) \right\} \quad on \quad \Gamma_{\varepsilon}^{i}.$$
(5.3.30)

Proof. Applying the tangential gradient operator $\nabla_{\Gamma_{\varepsilon}}$ to $u \cdot n_{\varepsilon} = 0$ on Γ_{ε}^{i} we have

$$(\nabla_{\Gamma_{\varepsilon}} u) n_{\varepsilon} = -(\nabla_{\Gamma_{\varepsilon}} n_{\varepsilon}) u = W_{\varepsilon} u \quad \text{on} \quad \Gamma^{i}_{\varepsilon}.$$
(5.3.31)

From this equality, (5.3.21), and

$$2P_{\varepsilon}D(u)n_{\varepsilon} = P_{\varepsilon}\{(\nabla u)n_{\varepsilon} + (\nabla u)^{T}n_{\varepsilon}\} = (\nabla_{\Gamma_{\varepsilon}}u)n_{\varepsilon} + P_{\varepsilon}(n_{\varepsilon}\cdot\nabla)u$$

by (5.2.39) we deduce that

$$P_{\varepsilon}(n_{\varepsilon}\cdot\nabla)u = -(\nabla_{\Gamma_{\varepsilon}}u)n_{\varepsilon} - \frac{\gamma_{\varepsilon}}{\nu}u = -W_{\varepsilon}u - \frac{\gamma_{\varepsilon}}{\nu}u \quad \text{on} \quad \Gamma^i_{\varepsilon}$$

Hence (5.3.29) holds. To prove (5.3.30) we observe that the vector field $n_{\varepsilon} \times \text{curl } u$ is tangential on Γ_{ε}^{i} . By this fact, (5.2.39), (5.3.29), and (5.3.31) we have

$$n_{\varepsilon} \times \operatorname{curl} u = P_{\varepsilon}(n_{\varepsilon} \times \operatorname{curl} u) = P_{\varepsilon}\{(\nabla u)n_{\varepsilon} - (\nabla u)^{T}n_{\varepsilon}\}$$
$$= (\nabla_{\Gamma_{\varepsilon}}u)n_{\varepsilon} - P_{\varepsilon}(n_{\varepsilon} \cdot \nabla)u = 2W_{\varepsilon}u + \frac{\gamma_{\varepsilon}}{u}u$$

on Γ_{ε}^{i} . The equality (5.3.30) follows from this equality and the identity

$$a \times (a \times b) = (a \cdot b)a - |a|^2 b, \quad a, b \in \mathbb{R}^3$$

with $a = n_{\varepsilon}$ and $b = 2W_{\varepsilon}u + \nu^{-1}\gamma_{\varepsilon}u$ since $n_{\varepsilon} \cdot u = 0$, $n_{\varepsilon} \cdot W_{\varepsilon}u = 0$, and $|n_{\varepsilon}|^2 = 1$ on Γ_{ε}^i . \Box

Let us derive an estimate for the $L^p(\Omega_{\varepsilon})$ -norm of the tangential component (with respect to Γ) of the stress vector $D(u)\bar{n}$. It is used in the study of a singular limit problem for (5.1.1)–(5.1.3) as ε tends to zero.

Lemma 5.3.8. Suppose that Assumption 1 is satisfied, i.e. the inequality (5.1.6) holds and let $p \in [1, \infty)$. Then there exists c > 0 independent of ε such that

$$\left\|\overline{P}D(u)\bar{n}\right\|_{L^{p}(\Omega_{\varepsilon})} \le c\varepsilon \|u\|_{W^{2,p}(\Omega_{\varepsilon})}$$
(5.3.32)

for all $u \in W^{2,p}(\Omega_{\varepsilon})^3$ satisfying (5.3.21) on Γ_{ε}^0 or on Γ_{ε}^1 .

Proof. We proceed as in the proof of Lemma 5.3.5. Let $u \in W^{2,p}(\Omega_{\varepsilon})^3$ satisfy (5.3.21) on Γ_{ε}^i for i = 0 or i = 1. By (5.3.6) and the boundedness of n and P we have

$$\left|\partial_n \left[\overline{P}D(u)\overline{n}\right]\right| \le c|\nabla^2 u|$$
 in Ω_{ε}

Hence we use (5.3.7) to get

$$\left\|\overline{P}D(u)\bar{n}\right\|_{L^{p}(\Omega_{\varepsilon})} \leq c\left(\varepsilon^{1/p}\left\|\overline{P}D(u)\bar{n}\right\|_{L^{p}(\Gamma_{\varepsilon}^{i})} + \varepsilon\|u\|_{W^{2,p}(\Omega_{\varepsilon})}\right).$$
(5.3.33)

Moreover, since u satisfies (5.3.21) on Γ^i_{ε} , we have

$$\overline{P}D(u)\bar{n} = (-1)^{i+1}P_{\varepsilon}D(u)n_{\varepsilon} + P_{\varepsilon}D(u)\{\bar{n} - (-1)^{i+1}n_{\varepsilon}\} + \left(\overline{P} - P_{\varepsilon}\right)D(u)\bar{n}$$
$$= (-1)^{i}\frac{\gamma_{\varepsilon}}{2\nu}u + P_{\varepsilon}D(u)\{\bar{n} - (-1)^{i+1}n_{\varepsilon}\} + \left(\overline{P} - P_{\varepsilon}\right)D(u)\bar{n}$$

on Γ_{ε}^{i} . Applying (5.1.6), (5.2.40), and (5.2.41) to the last line of this equality and noting that P_{ε} is bounded uniformly in ε we deduce that

$$\left|\overline{P}D(u)\bar{n}\right| \leq c\varepsilon(|u| + |\nabla u|) \quad \text{on} \quad \Gamma^i_{\varepsilon}$$

By this inequality and (5.3.8) we get

$$\left\|\overline{P}D(u)\bar{n}\right\|_{L^{p}(\Gamma_{\varepsilon}^{i})} \leq c\varepsilon\left(\|u\|_{L^{p}(\Gamma_{\varepsilon}^{i})} + \|\nabla u\|_{L^{p}(\Gamma_{\varepsilon}^{i})}\right) \leq c\varepsilon^{1-1/p}\|u\|_{W^{2,p}(\Omega_{\varepsilon})}.$$

We apply this inequality to (5.3.33) to conclude that (5.3.32) is valid.

Finally, we compare the tangential component (with respect to Γ) of the normal derivative of a vector field u on Ω_{ε} with $-\overline{W}u$.

Lemma 5.3.9. Under Assumption 1, there exists c > 0 independent of ε such that

$$\left\|\overline{P}\partial_n u + \overline{W}u\right\|_{L^p(\Omega_{\varepsilon})} \le c\varepsilon \|u\|_{W^{2,p}(\Omega_{\varepsilon})}$$
(5.3.34)

for all $u \in W^{2,p}(\Omega_{\varepsilon})^3$ with $p \in [1, \infty)$ satisfying the slip boundary conditions (5.3.20)–(5.3.21) on Γ_{ε}^0 or on Γ_{ε}^1 . Here $\partial_n u$ is the normal derivative of u given by (5.3.5).

Proof. For i = 0 or i = 1 let $u \in W^{2,p}(\Omega_{\varepsilon})^3$ satisfy (5.3.20)–(5.3.21) on Γ_{ε}^i . By (5.3.6) and the boundedness of n, P, and W on Γ we have

$$\left|\partial_n \left[\overline{P}\partial_n u + \overline{W}u\right]\right| \le c(|\nabla u| + |\nabla^2 u|) \quad \text{in} \quad \Omega_{\varepsilon}.$$
We apply (5.3.7) and the above inequality to get

$$\left\|\overline{P}\partial_{n}u + \overline{W}u\right\|_{L^{p}(\Omega_{\varepsilon})} \leq c\left(\varepsilon^{1/p}\left\|\overline{P}\partial_{n}u + \overline{W}u\right\|_{L^{p}(\Gamma_{\varepsilon}^{i})} + \varepsilon\|u\|_{W^{2,p}(\Omega_{\varepsilon})}\right).$$
(5.3.35)

Moreover, since $\partial_n u = (\bar{n} \cdot \nabla) u$ and u satisfies (5.3.20)–(5.3.21) on Γ^i_{ε} ,

$$\overline{P}\partial_n u = (-1)^{i+1} P_{\varepsilon}(n_{\varepsilon} \cdot \nabla) u + P_{\varepsilon}[\{\bar{n} - (-1)^{i+1}n_{\varepsilon}\} \cdot \nabla] u + \left(\overline{P} - P_{\varepsilon}\right)(\bar{n} \cdot \nabla) u \\ = (-1)^{i+1} \left(-W_{\varepsilon}u - \frac{\gamma_{\varepsilon}}{\nu}u\right) + P_{\varepsilon}[\{\bar{n} - (-1)^{i+1}n_{\varepsilon}\} \cdot \nabla] u + \left(\overline{P} - P_{\varepsilon}\right)(\bar{n} \cdot \nabla) u$$

on Γ_{ε}^{i} by (5.3.29). Hence by (5.1.6) and (5.2.40)–(5.2.42) we get

$$\left|\overline{P}\partial_{n}u + \overline{W}u\right| \leq \left|\overline{W}u - (-1)^{i+1}W_{\varepsilon}u\right| + c\varepsilon(|u| + |\nabla u|) \leq c\varepsilon(|u| + |\nabla u|)$$

on Γ^i_{ε} , which together with (5.3.8) implies that

$$\left\|\overline{P}\partial_n u + \overline{W}u\right\|_{L^p(\Gamma^i_{\varepsilon})} \le c\varepsilon \left(\|u\|_{L^p(\Gamma^i_{\varepsilon})} + \|\nabla u\|_{L^p(\Gamma^i_{\varepsilon})}\right) \le c\varepsilon^{1-1/p} \|u\|_{W^{2,p}(\Omega_{\varepsilon})}.$$

Applying this inequality to (5.3.35) we obtain (5.3.34).

5.3.3 Impermeable extension of surface vector fields

In the analysis of integrals over Ω_{ε} involving a vector field on Γ it is convenient to consider its extension to Ω_{ε} satisfying the impermeable boundary condition on Γ_{ε} . Let τ_{ε}^{0} and τ_{ε}^{1} be the vector fields on Γ given by (5.2.32). We define a vector field Ψ_{ε} on N by

$$\Psi_{\varepsilon}(x) := \frac{1}{\bar{g}(x)} \left\{ \left(d(x) - \varepsilon \bar{g}_0(x) \right) \bar{\tau}_{\varepsilon}^1(x) + \left(\varepsilon \bar{g}_1(x) - d(x) \right) \bar{\tau}_{\varepsilon}^0(x) \right\}, \quad x \in N.$$
(5.3.36)

By definition, $\Psi_{\varepsilon} = \varepsilon \overline{\tau}_{\varepsilon}^{i}$ on Γ_{ε}^{i} , i = 0, 1. Let us give several estimates for Ψ_{ε} .

Lemma 5.3.10. There exists a constant c > 0 independent of ε such that

$$|\Psi_{\varepsilon}| \le c\varepsilon, \quad |\nabla\Psi_{\varepsilon}| \le c, \quad |\nabla^2\Psi_{\varepsilon}| \le c \quad in \quad \Omega_{\varepsilon}.$$
 (5.3.37)

Moreover, we have

$$\left|\overline{P}\nabla\Psi_{\varepsilon}\right| \le c\varepsilon, \quad \left|\partial_{n}\Psi_{\varepsilon} - \frac{1}{\bar{g}}\overline{\nabla_{\Gamma}g}\right| \le c\varepsilon \quad in \quad \Omega_{\varepsilon}.$$
 (5.3.38)

Proof. Applying (5.2.34) and

$$0 \le d(x) - \varepsilon \bar{g}_0(x) \le \varepsilon \bar{g}(x), \quad 0 \le \varepsilon \bar{g}_1(x) - d(x) \le \varepsilon \bar{g}(x), \quad x \in \Omega_{\varepsilon}$$
(5.3.39)

to (5.3.36) we get the first inequality of (5.3.37). Also, by $\nabla d = \bar{n}$ in N we have

$$\nabla \Psi_{\varepsilon} = \frac{1}{\bar{g}} \{ \bar{n} \otimes (\bar{\tau}_{\varepsilon}^{1} - \bar{\tau}_{\varepsilon}^{0}) + F_{\varepsilon} \} \quad \text{in} \quad N,$$
(5.3.40)

where F_{ε} is a 3 × 3 matrix-valued function on N is given by

$$F_{\varepsilon} := -\nabla \bar{g} \otimes \Psi_{\varepsilon} + \varepsilon (\nabla \bar{g}_1 \otimes \bar{\tau}_{\varepsilon}^0 - \nabla \bar{g}_0 \otimes \bar{\tau}_{\varepsilon}^1) + (d - \varepsilon \bar{g}_0) \nabla \bar{\tau}_{\varepsilon}^1 + (\varepsilon \bar{g}_1 - d) \nabla \bar{\tau}_{\varepsilon}^0.$$

By (5.2.30), (5.2.34), and |n| = 1 on Γ we see that the first term on the right-hand side of (5.3.40) is bounded on N uniformly in ε . Moreover, from (5.2.13), (5.2.30), (5.2.34), the first inequality of (5.3.37), (5.3.39), and $g_0, g_1 \in C^4(\Gamma)$ we deduce that

$$|F_{\varepsilon}| \le c\varepsilon \quad \text{in} \quad \Omega_{\varepsilon}. \tag{5.3.41}$$

Hence the second inequality of (5.3.37) follows. Similarly, differentiating both sides of (5.3.40) and using (5.2.13), (5.2.15), (5.2.34), the first and second inequalities of (5.3.37), and $g_0, g_1 \in C^4(\Gamma)$ we can derive the last inequality of (5.3.37).

Let us prove (5.3.38). First note that

$$P[n \otimes (\tau_{\varepsilon}^{1} - \tau_{\varepsilon}^{0})] = (Pn) \otimes (\tau_{\varepsilon}^{1} - \tau_{\varepsilon}^{0}) = 0,$$

$$[(\tau_{\varepsilon}^{1} - \tau_{\varepsilon}^{0}) \otimes n]n = |n|^{2}(\tau_{\varepsilon}^{1} - \tau_{\varepsilon}^{0}) = \tau_{\varepsilon}^{1} - \tau_{\varepsilon}^{0}$$

on Γ . These equalities and (5.3.5) imply that

$$\overline{P}\nabla\Psi_{\varepsilon} = \frac{1}{\overline{g}}\overline{P}F_{\varepsilon}, \quad \partial_{n}\Psi_{\varepsilon} = (\nabla\Psi_{\varepsilon})^{T}\overline{n} = \frac{1}{\overline{g}}(\tau_{\varepsilon}^{1} - \tau_{\varepsilon}^{0}) + F_{\varepsilon}^{T}\overline{n} \quad \text{in} \quad N.$$

Hence we see by (5.2.30), (5.3.41), and |P| = 1 on Γ that

$$\left|\overline{P}\nabla\Psi_{\varepsilon}\right| \leq c\left|\overline{P}F_{\varepsilon}\right| \leq c|F_{\varepsilon}| \leq c\varepsilon$$
 in Ω_{ε} .

Also, applying (5.2.30), (5.2.35), and (5.3.41) to the equality for $\partial_n \Psi_{\varepsilon}$ we obtain

$$\left|\partial_n \Psi_{\varepsilon} - \frac{1}{\bar{g}} \overline{\nabla_{\Gamma} g}\right| \leq \frac{1}{\bar{g}} \sum_{i=0,1} \left| \bar{\tau}^i_{\varepsilon} - \overline{\nabla_{\Gamma} g_i} \right| + |F_{\varepsilon}| \leq c\varepsilon \quad \text{in} \quad \Omega_{\varepsilon}$$

Hence (5.3.38) is valid.

For a tangential vector field v on Γ (i.e. $v \cdot n = 0$ on Γ) we define

$$E_{\varepsilon}v(x) := \bar{v}(x) + \{\bar{v}(x) \cdot \Psi_{\varepsilon}(x)\}\bar{n}(x), \quad x \in N,$$
(5.3.42)

where \bar{v} and \bar{n} are the constant extensions of v and n. By the definition of Ψ_{ε} we easily see that $E_{\varepsilon}v$ satisfies the impermeable boundary condition on Γ_{ε} .

Lemma 5.3.11. For all $v \in C(\Gamma, T\Gamma)$ we have $E_{\varepsilon}v \cdot n_{\varepsilon} = 0$ on Γ_{ε} .

Proof. For i = 0, 1 we observe by (5.2.33), (5.2.38), and $v \cdot n = 0$ on Γ that

$$\bar{v} \cdot n_{\varepsilon} = (-1)^{i} \frac{\varepsilon \bar{v} \cdot \bar{\tau}_{\varepsilon}^{i}}{\sqrt{1 + \varepsilon^{2} |\bar{\tau}_{\varepsilon}^{i}|^{2}}}, \quad \bar{n} \cdot n_{\varepsilon} = \frac{(-1)^{i+1}}{\sqrt{1 + \varepsilon^{2} |\bar{\tau}_{\varepsilon}^{i}|^{2}}} \quad \text{on} \quad \Gamma_{\varepsilon}^{i}$$

From these equalities and $\Psi_{\varepsilon} = \varepsilon \overline{\tau}_{\varepsilon}^{i}$ on Γ_{ε}^{i} by (5.3.36) we get $E_{\varepsilon} v \cdot n_{\varepsilon} = 0$ on Γ_{ε}^{i} .

Also, it is easy to show that $E_{\varepsilon}v \in W^{m,p}(\Omega_{\varepsilon})$ for $v \in W^{m,p}(\Gamma, T\Gamma)$.

Lemma 5.3.12. There exists a constant c > 0 independent of ε such that

$$||E_{\varepsilon}v||_{W^{m,p}(\Omega_{\varepsilon})} \le c\varepsilon^{1/2} ||v||_{W^{m,p}(\Gamma)}$$
(5.3.43)

for all $v \in W^{m,p}(\Gamma, T\Gamma)$ with $p \in [1, \infty)$ and m = 0, 1, 2.

Proof. By (5.2.13), (5.2.15), and (5.3.37) we have

$$|E_{\varepsilon}v| \le c|\bar{v}|, \quad |\nabla E_{\varepsilon}v| \le c\left(|\bar{v}| + |\overline{\nabla_{\Gamma}v}|\right), \quad |\nabla^2 E_{\varepsilon}v| \le c\left(|\bar{v}| + |\overline{\nabla_{\Gamma}v}| + |\overline{\nabla_{\Gamma}v}|\right)$$

in Ω_{ε} . These inequalities, (5.2.53), (5.2.55), and (5.2.56) imply (5.3.43).

If Ω_{ε} is a flat thin domain of the form

$$\widehat{\Omega}_{\varepsilon} = \{ x = (x', x_3) \in \mathbb{R}^3 \mid x' \in \omega, \, \varepsilon \widetilde{g}_0(x') < x_3 < \varepsilon \widetilde{g}_1(x') \},\$$

where ω is a domain in \mathbb{R}^2 and \tilde{g}_0 and \tilde{g}_1 are functions on ω , then we have

$$\operatorname{div}(E_{\varepsilon}v)(x) = \frac{1}{\tilde{g}(x')}\operatorname{div}(\tilde{g}v)(x'), \quad x = (x', x_3) \in \widetilde{\Omega}_{\varepsilon} \quad (\tilde{g} := \tilde{g}_1 - \tilde{g}_0)$$

for $v: \omega \to \mathbb{R}^2$ (see [21, Lemma 4.24] and [26, Remark 3.1]). This is not the case for the curved thin domain Ω_{ε} given by (5.1.4) because the principal curvatures of the surface Γ do not vanish in general. However, we can show that the difference between $\operatorname{div}(E_{\varepsilon}v)$ and $g^{-1}\operatorname{div}_{\Gamma}(gv)$ is of order ε in Ω_{ε} .

Lemma 5.3.13. There exists a constant c > 0 independent of ε such that

$$\left|\nabla E_{\varepsilon}v - \left\{\overline{\nabla_{\Gamma}v} + \frac{1}{\bar{g}}\left(\bar{v}\cdot\overline{\nabla_{\Gamma}g}\right)\overline{Q}\right\}\right| \le c\varepsilon\left(|\bar{v}| + \left|\overline{\nabla_{\Gamma}v}\right|\right) \quad in \quad \Omega_{\varepsilon}$$
(5.3.44)

for all $v \in C^1(\Gamma, T\Gamma)$. Moreover, we have

$$\left|\operatorname{div}(E_{\varepsilon}v) - \frac{1}{\bar{g}}\overline{\operatorname{div}_{\Gamma}(gv)}\right| \le c\varepsilon \left(\left|\bar{v}\right| + \left|\overline{\nabla_{\Gamma}v}\right|\right) \quad in \quad \Omega_{\varepsilon}.$$
(5.3.45)

Proof. From (5.3.42), $Q = n \otimes n$, and $(n \otimes \nabla_{\Gamma} g)v = (v \cdot \nabla_{\Gamma} g)n$ it follows that

$$\nabla E_{\varepsilon}v = \nabla \bar{v} + \left[(\nabla \bar{v})\Psi_{\varepsilon} + (\nabla \Psi_{\varepsilon})\bar{v} \right] \otimes \bar{n} + (\bar{v} \cdot \Psi_{\varepsilon})\nabla \bar{n}$$
$$\left(\bar{v} \cdot \overline{\nabla_{\Gamma}g} \right) \overline{Q} = \left[\left(\bar{n} \otimes \overline{\nabla_{\Gamma}g} \right) \bar{v} \right] \otimes \bar{n}$$

in N. Hence

$$\left|\nabla E_{\varepsilon}v - \left\{\overline{\nabla_{\Gamma}v} + \frac{1}{\bar{g}}\left(\bar{v}\cdot\overline{\nabla_{\Gamma}g}\right)\overline{Q}\right\}\right| \leq \left|\nabla\bar{v}-\overline{\nabla_{\Gamma}v}\right| + \left|\left\{(\nabla\bar{v})\Psi_{\varepsilon}\right\}\otimes\bar{n}\right| + \left|(\bar{v}\cdot\Psi_{\varepsilon})\nabla\bar{n}\right| + \left|\left[\left(\nabla\Psi_{\varepsilon} - \frac{1}{\bar{g}}\bar{n}\otimes\overline{\nabla_{\Gamma}g}\right)\bar{v}\right]\otimes\bar{n}\right|. \quad (5.3.46)$$

Since $\nabla \Psi_{\varepsilon} = \overline{P} \nabla \Psi_{\varepsilon} + \overline{Q} \nabla \Psi_{\varepsilon} = \overline{P} \nabla \Psi_{\varepsilon} + \overline{n} \otimes \partial_n \Psi_{\varepsilon}$, by (5.3.38) we get

$$\left|\nabla\Psi_{\varepsilon} - \frac{1}{\bar{g}}\bar{n}\otimes\overline{\nabla_{\Gamma}g}\right| \le \left|\overline{P}\nabla\Psi_{\varepsilon}\right| + \left|\bar{n}\otimes\left(\partial_{n}\Psi_{\varepsilon} - \frac{1}{\bar{g}}\overline{\nabla_{\Gamma}g}\right)\right| \le c\varepsilon \quad \text{in} \quad \Omega_{\varepsilon}$$

Applying this inequality, (5.2.13), (5.2.14), (5.2.17), (5.3.37), and $|d| \leq c\varepsilon$ in Ω_{ε} to the right-hand side of (5.3.46) we obtain (5.3.44). Also, since tr[Q] = $n \cdot n = 1$,

$$\operatorname{tr}\left[\nabla_{\Gamma} v + \frac{1}{g}(v \cdot \nabla_{\Gamma} g)Q\right] = \operatorname{div}_{\Gamma} v + \frac{1}{g}(v \cdot \nabla_{\Gamma} g) = \frac{1}{g}\operatorname{div}_{\Gamma}(gv) \quad \text{on} \quad \Gamma.$$

Hence the inequality (5.3.45) follows from (5.3.44).

As a consequence of Lemma 5.3.13 we have the L^p -estimate for div $(E_{\varepsilon}v)$ on Ω_{ε} .

Lemma 5.3.14. Let $p \in [1, \infty)$. There exists c > 0 independent of ε such that

$$\|\operatorname{div}(E_{\varepsilon}v)\|_{L^{p}(\Omega_{\varepsilon})} \leq c\varepsilon^{1/p} \left(\|\operatorname{div}_{\Gamma}(gv)\|_{L^{p}(\Gamma)} + \varepsilon\|v\|_{W^{1,p}(\Gamma)}\right)$$
(5.3.47)

for all $v \in W^{1,p}(\Gamma, T\Gamma)$. In particular, if v satisfies $\operatorname{div}_{\Gamma}(gv) = 0$ on Γ , then

$$\|\operatorname{div}(E_{\varepsilon}v)\|_{L^{p}(\Omega_{\varepsilon})} \leq c\varepsilon^{1+1/p} \|v\|_{W^{1,p}(\Gamma)}.$$
(5.3.48)

Proof. By (5.2.30) and (5.3.45) we have

$$|\operatorname{div}(E_{\varepsilon}v)| \leq c \left\{ \left| \overline{\operatorname{div}_{\Gamma}(gv)} \right| + \varepsilon \left(|\overline{v}| + \left| \overline{\nabla_{\Gamma}v} \right| \right) \right\}$$
 in Ω_{ε} .

The inequality (5.3.47) follows from this inequality and (5.2.53).

5.4 Korn inequalities on a thin domain and a surface

In this section we establish the Korn inequalities on Ω_{ε} and Γ , which play a fundamental role in the study of the Stokes operator on Ω_{ε} associated with slip boundary conditions and the corresponding limit operator on Γ .

5.4.1 Uniform Korn inequalities on a thin domain

For the proof of the global existence of a strong solution to (5.1.1)-(5.1.3), it is essential that the bilinear form corresponding to the Stokes operator on Ω_{ε} with slip boundary conditions is uniformly coercive in ε (see Section 5.5.2). To show the uniform coerciveness of the bilinear form, let us prove the uniform Korn inequalities on Ω_{ε} . First we give an estimate for the L^2 -norm of the gradient matrix of a vector field on Ω_{ε} .

Lemma 5.4.1. There exists a constant $c_{K,1} > 0$ independent of ε such that

$$\|\nabla u\|_{L^{2}(\Omega_{\varepsilon})}^{2} \leq 4\|D(u)\|_{L^{2}(\Omega_{\varepsilon})}^{2} + c_{K,1}\|u\|_{L^{2}(\Omega_{\varepsilon})}^{2}$$
(5.4.1)

for all $\varepsilon \in (0,1)$ and $u \in H^1(\Omega_{\varepsilon})^3$ satisfying (5.3.20) on Γ_{ε} .

Let us prove an auxiliary density result.

Lemma 5.4.2. Let $u \in H^1(\Omega_{\varepsilon})^3$ satisfy (5.3.20) on Γ_{ε} . Then there exists a sequence $\{u_k\}_{k=1}^{\infty}$ in $C^2(\overline{\Omega}_{\varepsilon})^3$ such that u_k satisfies (5.3.20) on Γ_{ε} for each $k \in \mathbb{N}$ and

$$\lim_{k \to \infty} \|u - u_k\|_{H^1(\Omega_{\varepsilon})} = 0.$$

Proof. We follow the idea of the proof of [8, Theorem IV.4.7], but here it is not necessary to localize a vector field on Ω_{ε} . For $x \in N$ we define

$$\tilde{n}(x) := \frac{1}{\varepsilon \bar{g}(x)} \big\{ \big(d(x) - \varepsilon \bar{g}_0(x) \big) \bar{n}_{\varepsilon}^1(x) + \big(\varepsilon \bar{g}_1(x) - d(x) \big) \bar{n}_{\varepsilon}^0(x) \big\},\$$

where n_{ε}^0 and n_{ε}^1 are given by (5.2.33) and $\bar{\eta} = \eta \circ \pi$ denotes the constant extension of a function η on Γ . Then $\tilde{n} \in C^2(N)$ by the regularity of Γ , g_0 , and g_1 . Moreover, $\tilde{n} = n_{\varepsilon}$ on Γ_{ε} by Lemma 5.2.10. Hence if $u \in H^1(\Omega_{\varepsilon})^3$ satisfies (5.3.20) on Γ_{ε} , then we have $u \cdot \tilde{n} \in H_0^1(\Omega_{\varepsilon})$

and $w := u - (u \cdot \tilde{n})\tilde{n} \in H^1(\Omega_{\varepsilon})^3$. Since Γ_{ε} is of class C^1 , there exist sequences $\{\varphi_k\}_{k=1}^{\infty}$ in $C_c^{\infty}(\Omega_{\varepsilon})$ and $\{w_k\}_{k=1}^{\infty}$ in $C^{\infty}(\overline{\Omega_{\varepsilon}})^3$ such that

$$\lim_{k \to \infty} \| u \cdot \tilde{n} - \varphi_k \|_{H^1(\Omega_{\varepsilon})} = \lim_{k \to \infty} \| w - w_k \|_{H^1(\Omega_{\varepsilon})} = 0.$$

Here $C_c^{\infty}(\Omega_{\varepsilon})$ is the space of all smooth and compactly supported functions on Ω_{ε} . Therefore, setting $u_k := \varphi_k \tilde{n} + w_k - (w_k \cdot \tilde{n}) \tilde{n} \in C^2(\overline{\Omega}_{\varepsilon})$ we see that

$$u_k \cdot n_{\varepsilon} = u_k \cdot \tilde{n} = \varphi_k = 0 \quad \text{on} \quad \Gamma_{\varepsilon}$$

for each $k \in \mathbb{N}$ and (note that $u = (u \cdot \tilde{n})\tilde{n} + w$ and $w \cdot \tilde{n} = 0$ in Ω_{ε})

$$\begin{aligned} \|u - u_k\|_{H^1(\Omega_{\varepsilon})} &= \|(u \cdot \tilde{n} - \varphi_k)\tilde{n} + (w - w_k) - \{(w - w_k) \cdot \tilde{n}\}\tilde{n}\|_{H^1(\Omega_{\varepsilon})} \\ &\leq c\left(\|u \cdot \tilde{n} - \varphi_k\|_{H^1(\Omega_{\varepsilon})} + \|w - w_k\|_{H^1(\Omega_{\varepsilon})}\right) \to 0 \end{aligned}$$

as $k \to \infty$.

Proof of Lemma 5.4.1. By Lemma 5.4.2 it is sufficient to show (5.4.1) for all $u \in C^2(\overline{\Omega}_{\varepsilon})^3$ satisfying (5.3.20) on Γ_{ε} . Since $2|D(u)|^2 = |\nabla u|^2 + \nabla u : (\nabla u)^T$,

$$2\|D(u)\|_{L^{2}(\Omega_{\varepsilon})}^{2} = \|\nabla u\|_{L^{2}(\Omega_{\varepsilon})}^{2} + \int_{\Omega_{\varepsilon}} \nabla u : (\nabla u)^{T} dx$$

To the second them on the right-hand side we apply integration by parts twice (note that u is of class C^2 on $\overline{\Omega}_{\varepsilon}$) and then use (5.3.20) on Γ_{ε} to get

$$\int_{\Omega_{\varepsilon}} \nabla u : (\nabla u)^T \, dx = \int_{\Omega_{\varepsilon}} (\operatorname{div} u)^2 \, dx + \int_{\Gamma_{\varepsilon}} (u \cdot \nabla) u \cdot n_{\varepsilon} \, d\mathcal{H}^2.$$

Here the first them on the right-hand side is nonnegative. Therefore,

$$\|\nabla u\|_{L^{2}(\Omega_{\varepsilon})}^{2} \leq 2\|D(u)\|_{L^{2}(\Omega_{\varepsilon})}^{2} + \left|\int_{\Gamma_{\varepsilon}} (u \cdot \nabla)u \cdot n_{\varepsilon} \, d\mathcal{H}^{2}\right|.$$
(5.4.2)

Let us estimate the integral over Γ_{ε} in (5.4.2). Since *u* satisfies (5.3.20) on Γ_{ε} , we can apply (5.3.28) to the integrand of the boundary integral to get

$$\int_{\Gamma_{\varepsilon}} (u \cdot \nabla) u \cdot n_{\varepsilon} \, d\mathcal{H}^2 = \int_{\Gamma_{\varepsilon}} u \cdot W_{\varepsilon} u \, d\mathcal{H}^2 = \sum_{i=0,1} \int_{\Gamma_{\varepsilon}^i} u \cdot W_{\varepsilon} u \, d\mathcal{H}^2.$$
(5.4.3)

To estimate the right-hand side we set

$$F_{i}(y) := \sqrt{1 + \varepsilon^{2} |\tau_{\varepsilon}^{i}(y)|^{2} W_{\varepsilon,i}^{\sharp}(y)}, \quad i = 0, 1,$$

$$F(y,r) := \frac{1}{\varepsilon g(y)} \left\{ \left(r - \varepsilon g_{0}(y) \right) F_{1}(y) - \left(\varepsilon g_{1}(y) - r \right) F_{0}(y) \right\}, \quad (5.4.4)$$

$$\varphi(y,r) := u^{\sharp}(y,r) \cdot F(y,r) u^{\sharp}(y,r) J(y,r)$$

for $y \in \Gamma$ and $r \in [\varepsilon g_0(y), \varepsilon g_1(y)]$. Here and in what follows we use the notations (5.2.46) and (5.2.47), and we sometimes suppress the arguments y and r. By (5.4.4) we observe that

$$[u \cdot W_{\varepsilon}u]_{i}^{\sharp}(y)\sqrt{1+\varepsilon^{2}|\tau_{\varepsilon}^{i}(y)|^{2}}J(y,\varepsilon g_{i}(y)) = (-1)^{i+1}\varphi(y,\varepsilon g_{i}(y)), \quad y \in \Gamma, \ i = 0, 1.$$

From this relation and (5.2.57) we deduce that

$$\sum_{i=0,1} \int_{\Gamma_{\varepsilon}^{i}} [u \cdot W_{\varepsilon} u](x) d\mathcal{H}^{2}(x) = \int_{\Gamma} \{\varphi(y, \varepsilon g_{1}(y)) - \varphi(y, \varepsilon g_{0}(y))\} d\mathcal{H}^{2}(y)$$

$$= \int_{\Gamma} \int_{\varepsilon g_{0}(y)}^{\varepsilon g_{1}(y)} \frac{\partial \varphi}{\partial r}(y, r) dr d\mathcal{H}^{2}(y).$$
(5.4.5)

To estimate the integrand on the last line, we use (5.2.49) to get

$$\left|\frac{\partial\varphi}{\partial r}\right| \le c \left\{ \left(|F| + \left|\frac{\partial F}{\partial r}\right|\right) |u^{\sharp}|^2 + |F||u^{\sharp}||(\nabla u)^{\sharp}| \right\}.$$
(5.4.6)

By (5.2.34) and the uniform boundedness in ε of W_{ε} on Γ_{ε} we observe that F_0 and F_1 are bounded on Γ uniformly in ε . Thus we have

$$|F(y,r)| \le \frac{c}{\varepsilon g(y)} \left\{ \left(r - \varepsilon g_0(y) \right) + \left(\varepsilon g_1(y) - r \right) \right\} = c$$
(5.4.7)

for $y \in \Gamma$ and $r \in [\varepsilon g_0(y), \varepsilon g_1(y)]$. Also, by $\partial F / \partial r = (\varepsilon g)^{-1}(F_1 + F_0)$ and (5.4.4),

$$\left|\frac{\partial F}{\partial r}\right| \le c\varepsilon^{-1} \left(|W_{\varepsilon,1}^{\sharp} + W_{\varepsilon,0}^{\sharp}| + \sum_{i=0,1} \left(\sqrt{1 + \varepsilon^2 |\tau_{\varepsilon}^i|^2} - 1 \right) |W_{\varepsilon,i}^{\sharp}| \right).$$
(5.4.8)

By the mean value theorem for the function $\sqrt{1+s}$, $s \ge 0$ and (5.2.34) we have

$$(0 \le) \sqrt{1 + \varepsilon^2 |\tau_{\varepsilon}^i(y)|^2} - 1 \le \frac{\varepsilon^2}{2} |\tau_{\varepsilon}^i(y)|^2 \le c\varepsilon^2, \quad y \in \Gamma.$$
(5.4.9)

We apply this inequality, (5.2.45) with $G_{\varepsilon} = W_{\varepsilon}$, and the uniform boundedness in ε of W_{ε} to the right-hand side of (5.4.8) to obtain

$$\left|\frac{\partial F}{\partial r}(y,r)\right| \le c \quad \text{for all} \quad y \in \Gamma, \, r \in [\varepsilon g_0(y), \varepsilon g_1(y)]. \tag{5.4.10}$$

From (5.4.6), (5.4.7), and (5.4.10) we deduce that

$$\left|\frac{\partial\varphi}{\partial r}(y,r)\right| \le c \left(|u^{\sharp}(y,r)|^2 + \left[|u^{\sharp}||(\nabla u)^{\sharp}|\right](y,r)\right)$$
(5.4.11)

for all $y \in \Gamma$ and $r \in [\varepsilon g_0(y), \varepsilon g_1(y)]$, where c > 0 is a constant independent of ε . Applying (5.4.5) and (5.4.11) to (5.4.3) and using (5.2.52) and Holdör's inequality we see that

$$\left| \int_{\Gamma_{\varepsilon}} (u \cdot \nabla) u \cdot n_{\varepsilon} \, d\mathcal{H}^2 \right| \leq c \int_{\Gamma} \int_{\varepsilon g_0}^{\varepsilon g_1} \left(|u^{\sharp}|^2 + |u^{\sharp}| |(\nabla u)^{\sharp}| \right) dr \, d\mathcal{H}^2$$

$$\leq c \left(||u||_{L^2(\Omega_{\varepsilon})}^2 + ||u||_{L^2(\Omega_{\varepsilon})} ||\nabla u||_{L^2(\Omega_{\varepsilon})} \right).$$
(5.4.12)

By (5.4.2), (5.4.12), and Young's inequality we obtain

$$\begin{aligned} \|\nabla u\|_{L^{2}(\Omega_{\varepsilon})}^{2} &\leq 2\|D(u)\|_{L^{2}(\Omega_{\varepsilon})}^{2} + c\left(\|u\|_{L^{2}(\Omega_{\varepsilon})}^{2} + \|u\|_{L^{2}(\Omega_{\varepsilon})}\|\nabla u\|_{L^{2}(\Omega_{\varepsilon})}\right) \\ &\leq 2\|D(u)\|_{L^{2}(\Omega_{\varepsilon})}^{2} + c\|u\|_{L^{2}(\Omega_{\varepsilon})}^{2} + \frac{1}{2}\|\nabla u\|_{L^{2}(\Omega_{\varepsilon})}^{2}.\end{aligned}$$

Hence (5.4.1) follows.

Next we show a uniform H^1 -estimate for a vector field u on Ω_{ε} with $u \cdot n_{\varepsilon} = 0$ on Γ_{ε} by the L^2 -norm of the strain rate tensor on Ω_{ε} . To this end, we need to impose another condition on u (see (5.4.14) and Remark 5.4.6).

Lemma 5.4.3. For given $\alpha > 0$ and $\beta \in [0, 1)$ there exist constants

$$\varepsilon_K = \varepsilon_K(\alpha, \beta) \in (0, 1), \quad c_{K,2} = c_{K,2}(\alpha, \beta) > 0$$

independent of ε such that

$$\|u\|_{L^{2}(\Omega_{\varepsilon})}^{2} \leq \alpha \|\nabla u\|_{L^{2}(\Omega_{\varepsilon})}^{2} + c_{K,2} \|D(u)\|_{L^{2}(\Omega_{\varepsilon})}^{2}$$
(5.4.13)

for all $\varepsilon \in (0, \varepsilon_K)$ and $u \in H^1(\Omega_{\varepsilon})^3$ satisfying (5.3.20) on Γ_{ε} and

$$\left| (u, \bar{v})_{L^2(\Omega_{\varepsilon})} \right| \le \beta \|u\|_{L^2(\Omega_{\varepsilon})} \|\bar{v}\|_{L^2(\Omega_{\varepsilon})} \quad for \ all \quad v \in \mathcal{K}_g(\Gamma).$$
(5.4.14)

Here $\mathcal{K}_g(\Gamma)$ is the function space given by (5.1.8) and $\bar{v} = v \circ \pi$.

To prove Lemma 5.4.3 we use a change of variables formula to transform integrals over Ω_{ε} into those over the domain Ω_1 with fixed width (note that we assume $\overline{\Omega}_1 \subset N$ by scaling g_0 and g_1). Define a mapping $\Phi_{\varepsilon} \colon \Omega_1 \to \Omega_{\varepsilon}$ by

$$\Phi_{\varepsilon}(X) := \pi(X) + \varepsilon d(X)\bar{n}(X), \quad X \in \Omega_1.$$
(5.4.15)

We easily see that Φ_{ε} is bijective and its inverse $\Phi_{\varepsilon}^{-1} \colon \Omega_{\varepsilon} \to \Omega_1$ is given by

$$\Phi_{\varepsilon}^{-1}(x) := \pi(x) + \varepsilon^{-1} d(x) \bar{n}(x), \quad x \in \Omega_{\varepsilon}.$$

Also, by (5.2.6), (5.2.11), and (5.2.16) we have

$$\nabla \Phi_{\varepsilon} = \left(I_3 - d\overline{W}\right)^{-1} \left(I_3 - \varepsilon d\overline{W}\right) \overline{P} + \varepsilon \overline{Q} \quad \text{on} \quad \Omega_1.$$

Hence taking an orthonormal basis of \mathbb{R}^3 that consists of n and the other eigenvectors of W we get

$$\det \nabla \Phi_{\varepsilon}(X) = \varepsilon J(\pi(X), d(X))^{-1} J(\pi(X), \varepsilon d(X)), \quad X \in \Omega_1$$

and the change of variables formula

$$\int_{\Omega_{\varepsilon}} \varphi(x) \, dx = \varepsilon \int_{\Omega_1} \varphi(\Phi_{\varepsilon}(X)) J(\pi(X), d(X))^{-1} J(\pi(X), \varepsilon d(X)) \, dX \tag{5.4.16}$$

for a function φ on Ω_{ε} . In particular, by (5.2.49) we have

$$c\varepsilon^{-1} \|\varphi\|_{L^2(\Omega_{\varepsilon})}^2 \le \|\varphi \circ \Phi_{\varepsilon}\|_{L^2(\Omega_1)}^2 \le c'\varepsilon^{-1} \|\varphi\|_{L^2(\Omega_{\varepsilon})}^2$$
(5.4.17)

for $\varphi \in L^2(\Omega_{\varepsilon})$, where c and c' are positive constants independent of ε . By (5.4.17) and direct calculations of matrices we can show the following auxiliary inequalities for the proof of Lemma 5.4.3 (for a proof, see Appendix 5.A).

Lemma 5.4.4. For $u \in H^1(\Omega_{\varepsilon})^3$ let $U := u \circ \Phi_{\varepsilon}$. Then

$$\varepsilon^{-1} \|\nabla u\|_{L^2(\Omega_{\varepsilon})}^2 \ge c \left(\left\| \overline{P} \nabla U \right\|_{L^2(\Omega_1)}^2 + \varepsilon^{-2} \|\partial_n U\|_{L^2(\Omega_1)}^2 \right)$$
(5.4.18)

with a constant c > 0 independent of ε and u, where $\partial_n U = (\bar{n} \cdot \nabla)U$ is the normal derivative (with respect to Γ) of U. We also have

$$\varepsilon^{-1} \|D(u)\|_{L^2(\Omega_{\varepsilon})}^2 \ge c \left(\left\|\overline{P}F_{\varepsilon}(U)_S \overline{P}\right\|_{L^2(\Omega_1)}^2 + \varepsilon^{-2} \|\partial_n(U \cdot \bar{n})\|_{L^2(\Omega_1)}^2 \right),$$
(5.4.19)

where $F_{\varepsilon}(U)_S = \{F_{\varepsilon}(U) + F_{\varepsilon}(U)^T\}/2$ is the symmetric part of the matrix

$$F_{\varepsilon}(U) := \left(I_3 - \varepsilon d\overline{W}\right)^{-1} \left(I_3 - d\overline{W}\right) \nabla U.$$
(5.4.20)

Proof of Lemma 5.4.3. Following the idea of the proof of [21, Lemma 4.14] we prove (5.4.13) by contradiction. Assume to the contrary that there exist a sequence $\{\varepsilon_k\}_{k=1}^{\infty}$ of positive numbers with $\lim_{k\to\infty} \varepsilon_k = 0$ and vector fields $u_k \in H^1(\Omega_{\varepsilon_k})^3$ satisfying (5.3.20) on Γ_{ε_k} , (5.4.14), and

$$\|u_k\|_{L^2(\Omega_{\varepsilon_k})}^2 > \alpha \|\nabla u_k\|_{L^2(\Omega_{\varepsilon_k})}^2 + k \|D(u_k)\|_{L^2(\Omega_{\varepsilon_k})}^2, \quad k \in \mathbb{N}.$$
 (5.4.21)

For each $k \in \mathbb{N}$ let $U_k := u_k \circ \Phi_{\varepsilon_k} \in H^1(\Omega_1)^3$. Dividing both sides of (5.4.21) by ε_k and using (5.4.17), (5.4.18), and (5.4.19) we get

$$\begin{aligned} \|U_k\|_{L^2(\Omega_1)}^2 &> c\alpha \left(\left\|\overline{P}\nabla U_k\right\|_{L^2(\Omega_1)}^2 + \varepsilon_k^{-2} \|\partial_n U_k\|_{L^2(\Omega_1)}^2 \right) \\ &+ ck \left(\left\|\overline{P}F_{\varepsilon_k}(U_k)_S \overline{P}\right\|_{L^2(\Omega_1)}^2 + \varepsilon_k^{-2} \|\partial_n (U_k \cdot \bar{n})\|_{L^2(\Omega_1)}^2 \right), \end{aligned}$$

where $F_{\varepsilon_k}(U_k)_S$ is the symmetric part of the matrix $F_{\varepsilon_k}(U_k)$ given by (5.4.20). By this inequality we have $U_k \neq 0$ and thus we may assume

$$|U_k||_{L^2(\Omega_1)} = 1, \quad k \in \mathbb{N}$$
(5.4.22)

by replacing U_k with $U_k/\|U_k\|_{L^2(\Omega_1)}$. Then we get

$$\left\|\overline{P}\nabla U_k\right\|_{L^2(\Omega_1)}^2 + \varepsilon_k^{-2} \|\partial_n U_k\|_{L^2(\Omega_1)}^2 < c\alpha^{-1}, \tag{5.4.23}$$

$$\left\|\overline{P}F_{\varepsilon_k}(U_k)_S \overline{P}\right\|_{L^2(\Omega_1)}^2 + \varepsilon_k^{-2} \|\partial_n(U_k \cdot \bar{n})\|_{L^2(\Omega_1)}^2 < ck^{-1}.$$
(5.4.24)

From (5.4.22), (5.4.23), and

$$|\nabla U_k|^2 = \left|\overline{P}\nabla U_k\right|^2 + \left|\overline{Q}\nabla U_k\right|^2, \quad \left|\overline{Q}\nabla U_k\right| = \left|\overline{n}\otimes\partial_n U_k\right| = \left|\partial_n U_k\right|$$

it follows that $\{U_k\}_{k=1}^{\infty}$ is bounded in $H^1(\Omega_1)^3$. Hence there exists a subsequence of $\{U_k\}_{k=1}^{\infty}$, which we refer to as $\{U_k\}_{k=1}^{\infty}$ again, that converges to some $U \in H^1(\Omega_1)^3$ weakly in $H^1(\Omega_1)^3$. By the compact embedding $H^1(\Omega_1) \hookrightarrow L^2(\Omega_1)$ we also see that $\{U_k\}_{k=1}^{\infty}$ converges to Ustrongly in $L^2(\Omega_1)^3$ and thus

$$||U||_{L^2(\Omega_1)} = \lim_{k \to \infty} ||U_k||_{L^2(\Omega_1)} = 1$$
(5.4.25)

by (5.4.22). Our goal is to show U = 0 on Ω_1 , which contradicts with (5.4.25). Since

$$\lim_{k \to \infty} \|\partial_n U_k\|_{L^2(\Omega_1)} = 0 \tag{5.4.26}$$

by (5.4.23) and $\{U_k\}_{k=1}^{\infty}$ converges to U weakly in $H^1(\Omega_1)^3$, it follows that $\partial_n U = 0$ on Ω_1 , i.e. U is independent of the normal direction of Γ . Hence setting

$$V(y) := U(y + g_0(y)n(y)), \quad y \in \Gamma$$

we can consider U as the constant extension of V, i.e. $U = \overline{V}$ on Ω_1 . Moreover, from (5.3.8) with $\varepsilon = 1$ and $\partial_n \overline{V} = 0$ on Ω_1 we deduce that

$$\left\|U_k - \overline{V}\right\|_{L^2(\Gamma_1)} \le c \left(\left\|U_k - \overline{V}\right\|_{L^2(\Omega_1)} + \left\|\partial_n U_k\right\|_{L^2(\Omega_1)}\right)$$

Thus, by the strong convergence of $\{U_k\}_{k=1}^{\infty}$ to $\overline{V} = U$ in $L^2(\Omega_1)^3$ and (5.4.26),

$$\lim_{k \to \infty} \left\| U_k - \overline{V} \right\|_{L^2(\Gamma_1)} = 0.$$
 (5.4.27)

Now let us prove that $V \in \mathcal{K}_g(\Gamma)$. Since $\overline{V} = U \in H^1(\Omega_{\varepsilon})^3$, we have $V \in H^1(\Gamma)^3$ (see Lemma 5.2.13). To show that V is tangential on Γ we recall that the vector field u_k satisfies (5.3.20) on Γ_{ε_k} . Hence we can use (5.3.22) to get

$$|u_k \cdot \bar{n}| \le c\varepsilon_k |u_k|$$
 on Γ_{ε_k} , i.e. $|U_k \cdot \bar{n}| \le c\varepsilon_k |U_k|$ on Γ_1 .

From this inequality and (5.3.8) with $\varepsilon = 1$ it follows that

$$\|U_k \cdot \bar{n}\|_{L^2(\Gamma_1)} \le c\varepsilon_k \|U_k\|_{L^2(\Gamma_1)} \le c\varepsilon_k \|U_k\|_{H^1(\Omega_1)} \to 0 \quad \text{as} \quad k \to \infty$$
(5.4.28)

since $\{U_k\}_{k=1}^{\infty}$ is bounded in $H^1(\Omega_1)^3$. Combining this with (5.4.27) we get $\overline{V} \cdot \overline{n} = 0$ on Γ_1 , which implies that $V \cdot n = 0$ on Γ , i.e. V is tangential on Γ .

Next we show that $D_{\Gamma}(V) = 0$ on Γ . To this end we observe that

$$F_{\varepsilon_k}(U_k) = \left(I_3 - \varepsilon_k d\overline{W}\right)^{-1} \left(I_3 - d\overline{W}\right) \nabla U_k$$

converges weakly to $\overline{\nabla_{\Gamma} V}$ in $L^2(\Omega_1)^{3\times 3}$. Indeed, by (5.2.10) and $|d| \leq c$ on Ω_1 ,

$$\left|I_3 - \left\{I_3 - \varepsilon_k d(X)\overline{W}(X)\right\}^{-1}\right| \le c\varepsilon_k |d(X)| \le c\varepsilon_k \to 0 \quad \text{as} \quad k \to \infty$$

for all $X \in \Omega_1$, and thus $(I_3 - \varepsilon_k d\overline{W})^{-1}$ converges uniformly to I_3 on Ω_1 . By this fact, the weak convergence of $\{U_k\}_{k=1}^{\infty}$ to $U = \overline{V}$ in $H^1(\Omega_1)^3$, and (5.2.12) we get

$$\lim_{k \to \infty} F_{\varepsilon_k}(U_k) = \left(I_3 - d\overline{W}\right) \nabla \overline{V} = \overline{\nabla_{\Gamma} V} \quad \text{weakly in} \quad L^2(\Omega_1)^{3 \times 3}.$$

and thus

$$\lim_{\kappa \to \infty} \overline{P} F_{\varepsilon_k}(U_k)_S \overline{P} = \overline{P} \left(\overline{\nabla_{\Gamma} V} \right)_S \overline{P} = \overline{D_{\Gamma}(V)} \quad \text{weakly in} \quad L^2(\Omega_1)^{3 \times 3}.$$

From this fact and

$$\lim_{k \to \infty} \left\| \overline{P} F_{\varepsilon_k}(U_k)_S \overline{P} \right\|_{L^2(\Omega_1)} = 0$$

by (5.4.24) it follows that $\overline{D_{\Gamma}(V)} = 0$ on Ω_1 , i.e. $D_{\Gamma}(V) = 0$ on Γ .

It remains to prove $V \cdot \nabla_{\Gamma} g = 0$ on Γ . In what follows, we use the notations (5.2.46) and (5.2.47) (with $\varepsilon = 1$). Since u_k satisfies (5.3.20) on Γ_{ε_k} , we see by (5.3.22) that

$$u_k \cdot \bar{\tau}^i_{\varepsilon_k} = \varepsilon_k^{-1} u_k \cdot \bar{n}$$
 on $\Gamma^i_{\varepsilon_k}, i = 0, 1$

By this equality we get $U_k \cdot \bar{\tau}^i_{\varepsilon_k} = \varepsilon_k^{-1} U_k \cdot \bar{n}$ on Γ_1^i , i = 0, 1, or equivalently,

$$U_{k,i}^{\sharp}(y) \cdot \tau_{\varepsilon_k}^i(y) = \varepsilon_k^{-1} U_{k,i}^{\sharp}(y) \cdot n(y), \quad y \in \Gamma, \ i = 0, 1.$$

Hence

$$\|U_{k,1}^{\sharp} \cdot \tau_{\varepsilon_k}^1 - U_{k,0}^{\sharp} \cdot \tau_{\varepsilon_k}^0\|_{L^2(\Gamma)} \le \varepsilon_k^{-1} \|U_{k,1}^{\sharp} \cdot n - U_{k,0}^{\sharp} \cdot n\|_{L^2(\Gamma)}.$$
(5.4.29)

Moreover, since $\bar{n}^{\sharp}(y,r) = n(y)$ for $y \in \Gamma$ and $r \in (g_0(y), g_1(y))$, we have

$$(U_{k,1}^{\sharp} \cdot n)(y) - (U_{k,0}^{\sharp} \cdot n)(y) = \int_{g_0(y)}^{g_1(y)} \frac{\partial}{\partial r} \left((U_k \cdot \bar{n})^{\sharp}(y,r) \right) dr = \int_{g_0(y)}^{g_1(y)} [\partial_n (U_k \cdot \bar{n})]^{\sharp}(y,r) dr.$$

Hence by Hölder's inequality, (5.2.52), and (5.4.24),

$$\begin{aligned} \|U_{k,1}^{\sharp} \cdot n - U_{k,0}^{\sharp} \cdot n\|_{L^{2}(\Gamma)}^{2} &= \int_{\Gamma} \left(\int_{g_{0}(y)}^{g_{1}(y)} [\partial_{n}(U_{k} \cdot \bar{n})]^{\sharp}(y,r) \, dr \right)^{2} d\mathcal{H}^{2}(y) \\ &\leq c \|\partial_{n}(U_{k} \cdot \bar{n})\|_{L^{2}(\Omega_{1})}^{2} \leq c \varepsilon_{k}^{2} k^{-1}. \end{aligned}$$

Applying this inequality to the right-hand side of (5.4.29) we get

$$\|U_{k,1}^{\sharp} \cdot \tau_{\varepsilon_k}^1 - U_{k,0}^{\sharp} \cdot \tau_{\varepsilon_k}^0\|_{L^2(\Gamma)} \le ck^{-1/2} \to 0 \quad \text{as} \quad k \to \infty.$$
(5.4.30)

Also, by (5.2.34), (5.2.35), and (5.2.58),

$$\begin{split} \|U_{k,i}^{\sharp} \cdot \tau_{\varepsilon_{k}}^{i} - V \cdot \nabla_{\Gamma} g_{i}\|_{L^{2}(\Gamma)} &\leq \|(U_{k,i}^{\sharp} - V) \cdot \tau_{\varepsilon_{k}}^{i}\|_{L^{2}(\Gamma)} + \|V \cdot (\tau_{\varepsilon_{k}}^{i} - \nabla_{\Gamma} g_{i})\|_{L^{2}(\Gamma)} \\ &\leq c \left(\|U_{k,i}^{\sharp} - V\|_{L^{2}(\Gamma)} + \varepsilon_{k}\|V\|_{L^{2}(\Gamma)}\right) \\ &\leq c \left(\|U_{k} - \overline{V}\|_{L^{2}(\Gamma_{1}^{i})} + \varepsilon_{k}\|V\|_{L^{2}(\Gamma)}\right). \end{split}$$

Since the right-hand side tends to zero as $k \to \infty$ by (5.4.27), we get

$$\lim_{k \to \infty} \|U_{k,i}^{\sharp} \cdot \tau_{\varepsilon_k}^i - V \cdot \nabla_{\Gamma} g_i\|_{L^2(\Gamma)} = 0, \quad i = 0, 1.$$

Combining this equality with (5.4.30) we obtain

$$\|V \cdot \nabla_{\Gamma} g_1 - V \cdot \nabla_{\Gamma} g_0\|_{L^2(\Gamma)} = 0,$$

i.e. $V \cdot \nabla_{\Gamma} g = 0$ on Γ . Therefore, the vector field V is in $\mathcal{K}_g(\Gamma)$. Finally, we recall that $u_k \in H^1(\Omega_{\varepsilon_k})^3$ satisfies (5.4.14) and thus

$$\left| \left(u_k, \overline{V} \right)_{L^2(\Omega_{\varepsilon_k})} \right| \le \beta \| u_k \|_{L^2(\Omega_{\varepsilon_k})} \left\| \overline{V} \right\|_{L^2(\Omega_{\varepsilon_k})}$$
(5.4.31)

with $\beta \in [0, 1)$ by $V \in \mathcal{K}_g(\Gamma)$. Let us express this inequality in terms of U_k and send $k \to \infty$. Noting that $\overline{V}(\Phi_{\varepsilon_k}(X)) = \overline{V}(X)$ for $X \in \Omega_1$ by $\pi(\Phi_{\varepsilon_k}(X)) = \pi(X)$, we use the change of variables formula (5.4.16) to get

$$\left(u_k, \overline{V}\right)_{L^2(\Omega_{\varepsilon_k})} = \varepsilon_k \int_{\Omega_1} U_k \overline{V} \varphi_k \, dX_k$$

where $\varphi_k(X) := J(\pi(X), d(X))^{-1} J(\pi(X), \varepsilon_k d(X))$ for $X \in \Omega_1$. Here U_k converges strongly to $U = \overline{V}$ in $L^2(\Omega_1)^3$ as $k \to \infty$. Moreover, for all $X \in \Omega_1$ we see by (5.2.49) and (5.2.50) that

$$\varphi_k(X) - J(\pi(X), d(X))^{-1} | \le c\varepsilon_k \to 0 \text{ as } k \to \infty$$

and thus φ_k converges uniformly to $J(\pi(\cdot), d(\cdot))^{-1}$ on Ω_1 . Hence we obtain

$$\lim_{k \to \infty} \varepsilon_k^{-1} \left(u_k, \overline{V} \right)_{L^2(\Omega_{\varepsilon_k})} = \lim_{k \to \infty} \int_{\Omega_1} U_k \overline{V} \varphi_k \, dX = \int_{\Omega_1} \left| \overline{V} \right|^2 J(\pi(\cdot), d(\cdot))^{-1} \, dX.$$

By the change of variables formula (5.2.51) the last term is of the form

$$\int_{\Gamma} \int_{g_0(y)}^{g_1(y)} |V(y)|^2 J(y,r)^{-1} J(y,r) \, dr \, d\mathcal{H}^2(y) = \int_{\Gamma} g(y) |V(y)|^2 \, d\mathcal{H}^2(y)$$

Therefore,

$$\lim_{k \to \infty} \varepsilon_k^{-1} \left(u_k, \overline{V} \right)_{L^2(\Omega_{\varepsilon_k})} = \|g^{1/2}V\|_{L^2(\Gamma)}^2.$$
(5.4.32)

By the same arguments we also have

$$\lim_{k \to \infty} \varepsilon_k^{-1} \|u_k\|_{L^2(\Omega_{\varepsilon_k})}^2 = \lim_{k \to \infty} \varepsilon_k^{-1} \|\overline{V}\|_{L^2(\Omega_{\varepsilon_k})}^2 = \|g^{1/2}V\|_{L^2(\Gamma)}^2.$$
(5.4.33)

Now we divide both sides of (5.4.31) by ε_k , send $k \to \infty$, and use (5.4.32) and (5.4.33). Then we obtain

$$\|g^{1/2}V\|_{L^2(\Gamma)}^2 \leq \beta \|g^{1/2}V\|_{L^2(\Gamma)}^2$$

Since $\beta < 1$, we observe by this inequality and (5.2.30) that V = 0 on Γ . This shows $U = \overline{V} = 0$ in Ω_1 , which contradicts with (5.4.25). Hence (5.4.13) is valid.

Lemma 5.4.5. For given $\beta \in [0,1)$ there exist $\varepsilon_{\beta} \in (0,1)$ and $c_{\beta} > 0$ such that

$$\|u\|_{H^1(\Omega_{\varepsilon})}^2 \le c_{\beta} \|D(u)\|_{L^2(\Omega_{\varepsilon})}^2$$
(5.4.34)

for all $\varepsilon \in (0, \varepsilon_{\beta})$ and $u \in H^1(\Omega_{\varepsilon})^3$ satisfying (5.3.20) on Γ_{ε} and (5.4.14).

Proof. Let $c_{K,1} > 0$ be the constant given in Lemma 5.4.1. Also, let $\varepsilon_K \in (0,1)$ and $c_{K,2} > 0$ be the constants given in Lemma 5.4.3 with $\alpha := 1/2c_{K,1}$. For $\varepsilon \in (0, \varepsilon_K)$ let $u \in H^1(\Omega_{\varepsilon})^3$ satisfy (5.3.20) on Γ_{ε} and (5.4.14). By (5.4.1) and (5.4.13) we have

$$\|\nabla u\|_{L^{2}(\Omega_{\varepsilon})}^{2} \leq (4 + c_{K,1}c_{K,2}) \|D(u)\|_{L^{2}(\Omega_{\varepsilon})}^{2} + c_{K,1}\alpha \|\nabla u\|_{L^{2}(\Omega_{\varepsilon})}^{2}.$$

Since $\alpha = 1/2c_{K,1}$, the above inequality implies that

$$\|\nabla u\|_{L^{2}(\Omega_{\varepsilon})}^{2} \leq c_{\beta,1} \|D(u)\|_{L^{2}(\Omega_{\varepsilon})}^{2}, \quad c_{\beta,1} := 2(4 + c_{K,1}c_{K,2}).$$
(5.4.35)

From this inequality and (5.4.13) we further deduce that

$$\|u\|_{L^{2}(\Omega_{\varepsilon})}^{2} \leq c_{\beta,2} \|D(u)\|_{L^{2}(\Omega_{\varepsilon})}^{2}, \quad c_{\beta,2} := 2(2c_{K,1}^{-1} + c_{K,2}).$$
(5.4.36)

By (5.4.35) and (5.4.36) we get (5.4.34) with $\varepsilon_{\beta} := \varepsilon_K$ and $c_{\beta} := c_{\beta,1} + c_{\beta,2}$.

Remark 5.4.6. The uniform Korn inequality (5.4.34) was proved by Lewicka and Müller under a slightly more general assumptions on the curved thin domain Ω_{ε} , see [36, Theorem 2.2]. Here we gave another proof of the same inequality.

Lewicka and Müller also proved that (5.4.34) is not valid if we drop the condition (5.4.14). In the proof, they used a nontrivial Killing vector field in $\mathcal{K}_g(\Gamma)$ to construct a vector field $v_{\varepsilon} \in H^1(\Omega_{\varepsilon})^3$ satisfying $v_{\varepsilon} \cdot n_{\varepsilon} = 0$ on Γ_{ε} but

$$\|\nabla v_{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon})}^{2} \ge c\varepsilon, \quad \|D(v_{\varepsilon})\|_{L^{2}(\Omega_{\varepsilon})}^{2} \le c\varepsilon^{3}.$$

See [36, Section 4] for details.

5.4.2 Korn inequalities on a surface

In this subsection we prove the Korn inequalities on Γ , which are used in the study of a singular limit problem of the Navier–Stokes equations (5.1.1)–(5.1.3) as Ω_{ε} degenerates into Γ .

Lemma 5.4.7. There exists a constant c > 0 such that

$$\|\nabla_{\Gamma} v\|_{L^{2}(\Gamma)}^{2} \leq c \left(\|D_{\Gamma}(v)\|_{L^{2}(\Gamma)}^{2} + \|v\|_{L^{2}(\Gamma)}^{2}\right)$$
(5.4.37)

for all $v \in H^1(\Gamma, T\Gamma)$. Here $D_{\Gamma}(v)$ is the surface strain rate tensor given by

$$D_{\Gamma}(v) := P(\nabla_{\Gamma}v)_{S}P, \quad (\nabla_{\Gamma}v)_{S} = \frac{\nabla_{\Gamma}v + (\nabla_{\Gamma}v)^{T}}{2}.$$
(5.4.38)

Proof. For sufficiently small $\varepsilon > 0$ let $N_{\varepsilon} := \{x \in \mathbb{R}^3 \mid -\varepsilon < d(x) < \varepsilon\}$ be the tubular neighborhood of Γ such that $\overline{N}_{\varepsilon} \subset N$. As in the proof of Lemma 5.4.1 we can show that

$$\|\nabla u\|_{L^{2}(N_{\varepsilon})}^{2} \leq 4\|D(u)\|_{L^{2}(N_{\varepsilon})}^{2} + c\|u\|_{L^{2}(N_{\varepsilon})}^{2}$$
(5.4.39)

for all $u \in H^1(N_{\varepsilon})^3$ satisfying $u \cdot n_{N_{\varepsilon}} = 0$ on ∂N_{ε} , where c > 0 is a constant independent of ε and $n_{N_{\varepsilon}}$ is the unit outward normal vector field of the boundary ∂N_{ε} . Let $\bar{v} = v \circ \pi$ be the constant extension of $v \in H^1(\Gamma, T\Gamma)$. By Lemma 5.2.13 we see that $\bar{v} \in H^1(N_{\varepsilon})^3$. Moreover, since N_{ε} is the tubular neighborhood of Γ , the unit outward normal $n_{N_{\varepsilon}}$ is given by

$$n_{N_{\varepsilon}}(x) = \begin{cases} \bar{n}(x) & \text{if } d(x) = \varepsilon, \\ -\bar{n}(x) & \text{if } d(x) = -\varepsilon \end{cases}$$

for $x \in \partial N_{\varepsilon}$. By this fact and $v \cdot n = 0$ on Γ we see that $\overline{v} \cdot n_{N_{\varepsilon}} = 0$ on ∂N_{ε} . Hence we can apply (5.4.39) to \overline{v} to get

$$\|\nabla \bar{v}\|_{L^2(N_{\varepsilon})}^2 \le 4\|D(\bar{v})\|_{L^2(N_{\varepsilon})}^2 + c\|\bar{v}\|_{L^2(N_{\varepsilon})}^2.$$
(5.4.40)

Let us derive (5.4.37) from (5.4.40). By (5.2.53) and (5.2.54) with Ω_{ε} replaced by N_{ε} ,

$$\|\nabla \bar{v}\|_{L^{2}(N_{\varepsilon})}^{2} \ge c\varepsilon \|\nabla_{\Gamma} v\|_{L^{2}(\Gamma)}^{2}, \quad \|\bar{v}\|_{L^{2}(N_{\varepsilon})}^{2} \le c\varepsilon \|v\|_{L^{2}(\Gamma)}^{2}.$$
(5.4.41)

To estimate the $L^2(N_{\varepsilon})$ -norm of $D(\bar{v})$ we see by (5.2.14) and $|d| \leq \varepsilon$ in N_{ε} that

$$\left| D(\bar{v}) - \left(\overline{\nabla_{\Gamma} v} \right)_{S} \right| \le \left| \nabla \bar{v} - \overline{\nabla_{\Gamma} v} \right| \le c \varepsilon \left| \overline{\nabla_{\Gamma} v} \right| \quad \text{in} \quad N_{\varepsilon}.$$
(5.4.42)

Moreover, by $I_3 = P + Q$, $\nabla_{\Gamma} v = P \nabla_{\Gamma} v$,

$$(\nabla_{\Gamma} v)Q = \{(\nabla_{\Gamma} v)n\} \otimes n = \{\nabla_{\Gamma} (v \cdot n) - (\nabla_{\Gamma} n)v\} \otimes n = (Wv) \otimes n \quad \text{on} \quad \Gamma,$$

which follows from $v \cdot n = 0$ and $-\nabla_{\Gamma} n = W$, and (5.2.6) we have

$$\nabla_{\Gamma} v = P(\nabla_{\Gamma} v)P + P(\nabla_{\Gamma} v)Q = P(\nabla_{\Gamma} v)P + (Wv) \otimes n \quad \text{on} \quad \Gamma$$

Hence $(\nabla_{\Gamma} v)_S = D_{\Gamma}(v) + [(Wv) \otimes n]_S$ on Γ . From this equality, (5.4.42), and

$$|[(Wv) \otimes n]_S| \le |(Wv) \otimes n| = |Wv||n| \le c|v| \quad \text{on} \quad \Gamma$$

by |n| = 1 and the boundedness of W we deduce that

$$|D(\bar{v})| \le \left| \left(\overline{\nabla_{\Gamma} v} \right)_{S} \right| + \left| D(\bar{v}) - \left(\overline{\nabla_{\Gamma} v} \right)_{S} \right| \le \left| \overline{D_{\Gamma}(v)} \right| + c \left(|\bar{v}| + \varepsilon \left| \overline{\nabla_{\Gamma} v} \right| \right)$$

in N_{ε} . This inequality and (5.2.53) show that

$$\|D(\bar{v})\|_{L^{2}(N_{\varepsilon})}^{2} \leq c\varepsilon \left(\|D_{\Gamma}(v)\|_{L^{2}(\Gamma)}^{2} + \|v\|_{L^{2}(\Gamma)}^{2} + \varepsilon^{2}\|\nabla_{\Gamma}v\|_{L^{2}(\Gamma)}^{2}\right).$$
(5.4.43)

Now we apply (5.4.41) and (5.4.43) to (5.4.40) and then divide both sides by ε to get

$$\|\nabla_{\Gamma} v\|_{L^{2}(\Gamma)}^{2} \leq c_{1} \left(\|D_{\Gamma}(v)\|_{L^{2}(\Gamma)}^{2} + \|v\|_{L^{2}(\Gamma)}^{2} + \varepsilon^{2} \|\nabla_{\Gamma} v\|_{L^{2}(\Gamma)}^{2} \right)$$

with some constant $c_1 > 0$ independent of ε . We take $\varepsilon > 0$ so small that $c_1 \varepsilon^2 \le 1/2$ in the above inequality to obtain

$$\|\nabla_{\Gamma} v\|_{L^{2}(\Gamma)}^{2} \leq c_{1} \left(\|D_{\Gamma}(v)\|_{L^{2}(\Gamma)}^{2} + \|v\|_{L^{2}(\Gamma)}^{2} \right) + \frac{1}{2} \|\nabla_{\Gamma} v\|_{L^{2}(\Gamma)}^{2}.$$

Hence the inequality (5.4.37) follows.

Lemma 5.4.8. For given $\beta \in [0, 1)$, there exists $c_{\beta} > 0$ such that

$$\|v\|_{H^{1}(\Gamma)}^{2} \leq c_{\beta} \left(\|D_{\Gamma}(v)\|_{L^{2}(\Gamma)}^{2} + \|v \cdot \nabla_{\Gamma}g\|_{L^{2}(\Gamma)}^{2} \right)$$
(5.4.44)

for all $v \in H^1(\Gamma, T\Gamma)$ satisfying

$$\left| (v,w)_{L^2(\Gamma)} \right| \le \beta \|v\|_{L^2(\Gamma)} \|w\|_{L^2(\Gamma)} \quad for \ all \quad w \in \mathcal{K}_g(\Gamma), \tag{5.4.45}$$

where $\mathcal{K}_{q}(\Gamma)$ is the function space given by (5.1.8).

Proof. We prove the inequality

$$\|v\|_{L^{2}(\Gamma)}^{2} \leq c \left(\|D_{\Gamma}(v)\|_{L^{2}(\Gamma)}^{2} + \|v \cdot \nabla_{\Gamma}g\|_{L^{2}(\Gamma)}^{2}\right).$$
(5.4.46)

Assume to the contrary that for each $k \in \mathbb{N}$ there exists $v_k \in H^1(\Gamma, T\Gamma)$ such that

$$\|v_k\|_{L^2(\Gamma)}^2 > k\left(\|D_{\Gamma}(v_k)\|_{L^2(\Gamma)}^2 + \|v_k \cdot \nabla_{\Gamma}g\|_{L^2(\Gamma)}^2\right)$$

and v_k satisfies (5.4.45). Since $v_k \neq 0$, we may assume $||v_k||_{L^2(\Gamma)} = 1$ by replacing v_k with $v_k/||v_k||_{L^2(\Gamma)}$. Then by the above inequality we have

$$\|D_{\Gamma}(v_k)\|_{L^2(\Gamma)}^2 + \|v_k \cdot \nabla_{\Gamma} g\|_{L^2(\Gamma)}^2 < \frac{1}{k}.$$
(5.4.47)

By $||v_k||_{L^2(\Gamma)} = 1$, (5.4.37), and (5.4.47), the sequence $\{v_k\}_{k=1}^{\infty}$ is bounded in $H^1(\Gamma, T\Gamma)$. Hence, using the compact embedding $H^1(\Gamma, T\Gamma) \hookrightarrow L^2(\Gamma, T\Gamma)$, we can take a subsequence of $\{v_k\}_{k=1}^{\infty}$, which we denote by $\{v_k\}_{k=1}^{\infty}$ again, that converges to some $v \in H^1(\Gamma, T\Gamma)$ weakly in $H^1(\Gamma, T\Gamma)$ and strongly in $L^2(\Gamma, T\Gamma)$, and thus

$$\|v\|_{L^{2}(\Gamma)} = \lim_{k \to \infty} \|v_{k}\|_{L^{2}(\Gamma)} = 1.$$
(5.4.48)

Moreover, since $\{D_{\Gamma}(v_k)\}_{k=1}^{\infty}$ and $\{v_k \cdot \nabla_{\Gamma}g\}_{k=1}^{\infty}$ converge to $D_{\Gamma}(v)$ and $v \cdot \nabla_{\Gamma}g$ weakly in $L^2(\Gamma)^{3\times 3}$ and $L^2(\Gamma)$, we send $k \to \infty$ in (5.4.47) to get

$$D_{\Gamma}(v) = 0, \quad v \cdot \nabla_{\Gamma} g = 0 \quad \text{on} \quad \Gamma, \quad \text{i.e.} \quad v \in \mathcal{K}_q(\Gamma).$$

Now we recall that v_k satisfies (5.4.45). Hence by $||v_k||_{L^2(\Gamma)} = 1$ and (5.4.48) we have

$$\left| (v_k, v)_{L^2(\Gamma)} \right| \le \beta \| v_k \|_{L^2(\Gamma)} \| v \|_{L^2(\Gamma)} = \beta \quad \text{for all} \quad k \in \mathbb{N}.$$

We send $k \to \infty$ in the above equality and then use the strong convergence of $\{v_k\}_{k=1}^{\infty}$ to v in $L^2(\Gamma, T\Gamma)$ and (5.4.48). Then we have $1 \leq \beta$, which contradicts with $\beta \in [0, 1)$. Hence (5.4.46) holds and we get (5.4.44) by combining (5.4.37) and (5.4.46).

5.5 Stokes operator with slip boundary conditions

In this section we prove inequalities for the Helmholtz–Leray projection on Ω_{ε} and the Stokes operator associated with slip boundary conditions. We use the inequality (5.5.1) given in Section 5.5.1 for the study of a singular limit problem and inequalities and formulas in other subsections for the proof of the global existence of a strong solution.

5.5.1 Helmholtz–Leray projection on a thin domain

Let $L^2_{\sigma}(\Omega_{\varepsilon})$ be the closure of the space $C^{\infty}_{c,\sigma}(\Omega_{\varepsilon}) := \{u \in C^{\infty}_{c}(\Omega_{\varepsilon})^3 \mid \operatorname{div} u = 0 \text{ in } \Omega_{\varepsilon}\}$ in $L^2(\Omega_{\varepsilon})^3$. It is well-known (see e.g. [8,14,64]) that $L^2_{\sigma}(\Omega_{\varepsilon})$ is characterized by

$$L^2_{\sigma}(\Omega_{\varepsilon}) = \{ u \in L^2(\Omega_{\varepsilon})^3 \mid \operatorname{div} u = 0 \text{ in } \Omega_{\varepsilon}, \, u \cdot n_{\varepsilon} = 0 \text{ on } \Gamma_{\varepsilon} \}$$

and the Helmholtz–Leray decomposition $L^2(\Omega_{\varepsilon})^3 = L^2_{\sigma}(\Omega_{\varepsilon}) \oplus L^2_{\sigma}(\Omega_{\varepsilon})^{\perp}$ holds with

$$L^2_{\sigma}(\Omega_{\varepsilon})^{\perp} = \{ \nabla p \in L^2(\Omega_{\varepsilon})^3 \mid p \in H^1(\Omega_{\varepsilon}) \}$$

Let \mathbb{P}_{ε} be the Helmholtz–Leray projection from $L^2(\Omega_{\varepsilon})^3$ onto $L^2_{\sigma}(\Omega_{\varepsilon})$. It is given by $\mathbb{P}_{\varepsilon}u = u - \nabla \varphi$ for $u \in L^2(\Omega_{\varepsilon})^3$, where $\varphi \in H^1(\Omega_{\varepsilon})$ is a weak solution to the Neumann problem of Poisson's equation

$$\Delta \varphi = \operatorname{div} u \quad \text{in} \quad \Omega_{\varepsilon}, \quad \frac{\partial \varphi}{\partial n_{\varepsilon}} = u \cdot n_{\varepsilon} \quad \text{on} \quad \Gamma_{\varepsilon}.$$

Moreover, by the elliptic regularity theorem (see [13, 15]) we have $\mathbb{P}_{\varepsilon} u \in H^1(\Omega_{\varepsilon})^3$ when $u \in H^1(\Omega_{\varepsilon})^3$. Our aim is to give a uniform estimate for the $H^1(\Omega_{\varepsilon})$ -norm of the difference $u - \mathbb{P}_{\varepsilon} u$ for $u \in H^1(\Omega_{\varepsilon})^3$ satisfying $u \cdot n_{\varepsilon} = 0$ on Γ_{ε} .

Lemma 5.5.1. There exist constants $\varepsilon_{\sigma} \in (0, 1)$ and c > 0 such that

$$\|u - \mathbb{P}_{\varepsilon} u\|_{H^1(\Omega_{\varepsilon})} \le c \|\operatorname{div} u\|_{L^2(\Omega_{\varepsilon})}$$
(5.5.1)

for all $\varepsilon \in (0, \varepsilon_{\sigma})$ and $u \in H^1(\Omega_{\varepsilon})^3$ satisfying $u \cdot n_{\varepsilon} = 0$ on Γ_{ε} .

To prove Lemma 5.5.1 we first derive the uniform Poincaré inequality for $\varphi \in H^1(\Omega_{\varepsilon})$ whose integral over Ω_{ε} vanishes.

Lemma 5.5.2. There exist constants $\varepsilon_{\sigma} \in (0, 1)$ and c > 0 such that

$$\|\varphi\|_{L^2(\Omega_{\varepsilon})} \le c \|\nabla\varphi\|_{L^2(\Omega_{\varepsilon})} \tag{5.5.2}$$

for all $\varepsilon \in (0, \varepsilon_{\sigma})$ and $\varphi \in H^1(\Omega_{\varepsilon})$ satisfying $\int_{\Omega_{\varepsilon}} \varphi \, dx = 0$.

Proof. As in the proof of Lemma 5.4.4, we prove (5.5.2) by contradiction. Assume to the contrary that there exist a sequence $\{\varepsilon_k\}_{k=1}^{\infty}$ of positive numbers that converges to zero and $\varphi_k \in H^1(\Omega_{\varepsilon_k})$ such that

$$\|\varphi_k\|_{L^2(\Omega_{\varepsilon_k})}^2 > k \|\nabla\varphi_k\|_{L^2(\Omega_{\varepsilon_k})}^2, \quad \int_{\Omega_{\varepsilon_k}} \varphi_k \, dx = 0, \quad k \in \mathbb{N}.$$
(5.5.3)

For each $k \in \mathbb{N}$ let Φ_{ε_k} be the bijection from Ω_1 onto Ω_{ε_k} given by (5.4.15) and define $\xi_k := \varphi_k \circ \Phi_{\varepsilon_k} \in H^1(\Omega_1)$. We divide both sides of the first inequality of (5.5.3) by ε_k and apply (5.4.17) and (5.4.18) to $u = (\varphi_k, 0, 0)$ and $U = (\xi_k, 0, 0)$ to get

$$\|\xi_k\|_{L^2(\Omega_1)}^2 > ck\left(\|\overline{P}\nabla\xi_k\|_{L^2(\Omega_1)}^2 + \varepsilon_k^{-2} \|\partial_n\xi_k\|_{L^2(\Omega_1)}^2\right),\,$$

where $\partial_n \xi_k$ is the normal derivative of ξ_k given by (5.3.5). Since $\|\xi_k\|_{L^2(\Omega_1)} > 0$, we may assume that $\|\xi_k\|_{L^2(\Omega_1)} = 1$ by replacing ξ_k with $\xi_k/\|\xi_k\|_{L^2(\Omega_1)}$. Then by the above inequality we get

$$\|\overline{P}\nabla\xi_k\|_{L^2(\Omega_1)}^2 < ck^{-1}, \quad \|\partial_n\xi_k\|_{L^2(\Omega_1)}^2 < c\varepsilon_k^2 k^{-1}.$$
 (5.5.4)

By these inequalities, $\|\xi_k\|_{L^2(\Omega_1)} = 1$, and

$$|\nabla \xi_k|^2 = \left|\overline{P}\nabla \xi_k\right|^2 + \left|\overline{Q}\nabla \xi_k\right|^2, \quad \left|\overline{Q}\nabla \xi_k\right| = |\overline{n} \otimes \partial_n \xi_k| = |\partial_n \xi_k|.$$

we observe that $\{\xi_k\}_{k=1}^{\infty}$ is bounded in $H^1(\Omega_1)$. From this fact and the compact embedding $H^1(\Omega_1) \hookrightarrow L^2(\Omega_1)$ we deduce that there exists a subsequence of $\{\xi_k\}_{k=1}^{\infty}$, which we denote by $\{\xi_k\}_{k=1}^{\infty}$ again, that converges to some $\xi \in H^1(\Omega_1)$ strongly in $L^2(\Omega_1)$ and weakly in $H^1(\Omega_1)$. Hence

$$\|\xi\|_{L^2(\Omega_1)} = \lim_{k \to \infty} \|\xi_k\|_{L^2(\Omega_1)} = 1.$$
(5.5.5)

By the weak convergence of $\{\xi_k\}_{k=1}^{\infty}$ to ξ in $H^1(\Omega_1)$ and (5.5.4) we have $\overline{P}\nabla\xi = 0$ and $\partial_n\xi = 0$ in Ω_1 . For $y \in \Gamma$ let $\eta(y) := \xi(y + g_0(y)n(y))$. Then by $\partial_n\xi = 0$ we see that $\xi = \overline{\eta}$ is the constant extension of η , and thus $\eta \in H^1(\Gamma)$ by $\xi \in H^1(\Omega_1)$ and Lemma 5.2.13. Moreover, by $\overline{P}\nabla\xi = 0$, (5.2.3), (5.2.8), and (5.2.12) we have

$$\overline{P}\left(I_3 - d\overline{W}\right)^{-1} \overline{\nabla_{\Gamma} \eta} = \left(I_3 - d\overline{W}\right)^{-1} \overline{\nabla_{\Gamma} \eta} = 0 \quad \text{in} \quad \Omega_1,$$

which implies that $\overline{\nabla_{\Gamma}\eta} = 0$ in Ω_1 , i.e. $\nabla_{\Gamma}\eta = 0$ on Γ . Hence setting

$$\hat{\eta} := \eta - \frac{1}{|\Gamma|} \int_{\Gamma} \eta \, d\mathcal{H}^2 \quad \text{on} \quad \Gamma$$

where $|\Gamma|$ is the area of Γ , we can apply Poincaré's inequality (5.2.20) to $\hat{\eta}$ to get

$$\|\hat{\eta}\|_{L^2(\Gamma)} \le c \|\nabla_{\Gamma}\hat{\eta}\|_{L^2(\Gamma)} = c \|\nabla_{\Gamma}\eta\|_{L^2(\Gamma)} = 0$$

and thus $\hat{\eta} = 0$ on Γ , i.e. $\eta = |\Gamma|^{-1} \int_{\Gamma} \eta \, d\mathcal{H}^2$ is constant. Now we return to the second equality of (5.5.3) and use the change of variables formula (5.4.16) to get

$$\int_{\Omega_1} \xi_k(X) J(\pi(X), d(X))^{-1} J(\pi(X), \varepsilon_k d(X)) \, dX = 0.$$

Let $k \to \infty$ in this equality. Then since $J(\pi(\cdot), \varepsilon_k d(\cdot))$ converges to one uniformly in Ω_1 as $k \to \infty$ by (5.2.50) and $\{\xi_k\}_{k=1}^{\infty}$ converges to $\xi = \bar{\eta}$ strongly in $L^2(\Omega_1)$,

$$\int_{\Omega_1} \eta(\pi(X)) J(\pi(X), d(X))^{-1} \, dX = 0$$

Moreover, by the change of variables formula (5.2.51) and the fact that η is constant on Γ , we obtain

$$\eta \int_{\Gamma} \int_{g_0(y)}^{g_1(y)} J(y,r)^{-1} J(y,r) \, dr \, d\mathcal{H}^2(y) = \eta \int_{\Gamma} g(y) \, d\mathcal{H}^2(y) = 0,$$

which together with (5.2.30) yields that $\eta = 0$. Hence

$$\|\xi\|_{L^2(\Omega_1)} = \|\bar{\eta}\|_{L^2(\Omega_1)} = 0,$$

which contradicts with (5.5.5). Therefore, the uniform inequality (5.5.2) is valid.

Next we consider the Neumann problem of Poisson's equation

$$\Delta \varphi = -\xi \quad \text{in} \quad \Omega_{\varepsilon}, \quad \frac{\partial \varphi}{\partial n_{\varepsilon}} = 0 \quad \text{on} \quad \Gamma_{\varepsilon}$$
(5.5.6)

for $\xi \in H^{-1}(\Omega_{\varepsilon})$ satisfying $\langle \xi, 1 \rangle_{\Omega_{\varepsilon}} = 0$, where $\langle \cdot, \cdot \rangle_{\Omega_{\varepsilon}}$ denotes the duality product between $H^{-1}(\Omega_{\varepsilon})$ and $H^{1}(\Omega_{\varepsilon})$. By the Lax–Milgram theory there exists a unique weak solution $\varphi \in H^{1}(\Omega_{\varepsilon})$ satisfying

$$(\nabla\varphi,\nabla\zeta)_{L^2(\Omega_{\varepsilon})} = \langle\xi,\zeta\rangle_{\Omega_{\varepsilon}} \quad \text{for all} \quad \zeta \in H^1(\Omega_{\varepsilon}), \quad \int_{\Omega_{\varepsilon}} \varphi \, dx = 0. \tag{5.5.7}$$

Moreover, by the elliptic regularity theorem, if $\xi \in L^2(\Omega_{\varepsilon})$ then we have $\varphi \in H^2(\Omega_{\varepsilon})$ and there exists a constant $c_{\varepsilon} > 0$ depending on ε such that

$$\|\varphi\|_{H^2(\Omega_{\varepsilon})} \le c_{\varepsilon} \|\xi\|_{L^2(\Omega_{\varepsilon})}.$$
(5.5.8)

In this case, the equation (5.5.6) is satisfied in the strong sense. Let us show that we can take a constant c_{ε} in (5.5.8) independently of ε .

Lemma 5.5.3. Let ε_{σ} be the constant given in Lemma 5.5.2 and $\varepsilon \in (0, \varepsilon_{\sigma})$. Suppose that $\xi \in L^2(\Omega_{\varepsilon})$ satisfies $\int_{\Omega_{\varepsilon}} \xi \, dx = 0$. Then there exists a constant c > 0 independent of ε and ξ such that

$$\|\varphi\|_{H^2(\Omega_{\varepsilon})} \le c \|\xi\|_{L^2(\Omega_{\varepsilon})} \tag{5.5.9}$$

for a unique solution $\varphi \in H^2(\Omega_{\varepsilon})$ to the problem (5.5.6).

Proof. Setting $\zeta = \varphi$ in (5.5.7) and using (5.5.2) we immediately get

$$\|\varphi\|_{H^1(\Omega_{\varepsilon})} \le \|\xi\|_{H^{-1}(\Omega_{\varepsilon})} \le \|\xi\|_{L^2(\Omega_{\varepsilon})}.$$

Hence it is sufficient to show that

$$\|\nabla^2 \varphi\|_{L^2(\Omega_{\varepsilon})} \le c \left(\|\Delta \varphi\|_{L^2(\Omega_{\varepsilon})} + \|\varphi\|_{H^1(\Omega_{\varepsilon})}\right) = c \left(\|\xi\|_{L^2(\Omega_{\varepsilon})} + \|\varphi\|_{H^1(\Omega_{\varepsilon})}\right)$$
(5.5.10)

with some constant c > 0 independent of ε (note that $\Delta \varphi = \xi$ a.e. in Ω_{ε}).

Since $C_c^{\infty}(\Omega_{\varepsilon})$ is dense in $L^2(\Omega_{\varepsilon})$, we can take a sequence $\{\xi_k\}_{k=1}^{\infty}$ of functions in $C_c^{\infty}(\Omega_{\varepsilon})$ that converges strongly to ξ in $L^2(\Omega_{\varepsilon})$. For each $k \in \mathbb{N}$ let

$$\tilde{\xi}_k(x) := \xi_k(x) - \frac{1}{|\Omega_{\varepsilon}|} \int_{\Omega_{\varepsilon}} \xi_k(y) \, dy, \quad x \in \Omega_{\varepsilon}$$

and $\varphi_k \in H^1(\Omega_{\varepsilon})$ be a unique weak solution to (5.5.6) with ξ replaced by $\tilde{\xi}_k$ (note that $\langle \tilde{\xi}_k, 1 \rangle_{\Omega_{\varepsilon}} = (\tilde{\xi}_k, 1)_{L^2(\Omega_{\varepsilon})} = 0$). Since $\tilde{\xi}_k \in C^{\infty}(\overline{\Omega}_{\varepsilon})$ and Γ_{ε} is of class C^4 , the elliptic regularity theorem yields that $\varphi_k \in H^3(\Omega_{\varepsilon})$. Moreover, by the strong convergence of $\{\xi_k\}_{k=1}^{\infty}$ to ξ in $L^2(\Omega_{\varepsilon})$ and

$$\lim_{k \to \infty} \int_{\Omega_{\varepsilon}} \xi_k \, dx = \lim_{k \to \infty} (\xi_k, 1)_{L^2(\Omega_{\varepsilon})} = (\xi, 1)_{L^2(\Omega_{\varepsilon})} = \int_{\Omega_{\varepsilon}} \xi \, dx = 0$$

we observe that

$$\|\xi - \tilde{\xi}_k\|_{L^2(\Omega_{\varepsilon})} \le \|\xi - \xi_k\|_{L^2(\Omega_{\varepsilon})} + \frac{1}{|\Omega_{\varepsilon}|^{1/2}} \left| \int_{\Omega_{\varepsilon}} \xi_k \, dx \right| \to 0 \quad \text{as} \quad k \to \infty$$

Since $\varphi - \varphi_k$ is a unique solution to (5.5.6) for the source term $\xi - \tilde{\xi}_k$, by (5.5.8) and the above convergence we obtain

$$\|\varphi - \varphi_k\|_{H^2(\Omega_{\varepsilon})} \le c_{\varepsilon} \|\xi - \tilde{\xi}_k\|_{L^2(\Omega_{\varepsilon})} \to 0 \quad \text{as} \quad k \to \infty.$$

(Note that the constant c_{ε} does not depend on k.) Hence we can derive (5.5.10) by showing the same inequality for φ_k and sending $k \to \infty$.

From now on, we fix and suppress the subscript k. Hence we suppose that φ is in $H^3(\Omega_{\varepsilon})$ and satisfies (5.5.6) in the strong sense. In particular, we have

$$\nabla \varphi \cdot n_{\varepsilon} = \frac{\partial \varphi}{\partial n_{\varepsilon}} = 0 \quad \text{on} \quad \Gamma_{\varepsilon}.$$
(5.5.11)

By the regularity of φ we can carry out integration by parts twice to get

$$\begin{aligned} |\nabla^{2}\varphi||_{L^{2}(\Omega_{\varepsilon})}^{2} &= \|\Delta\varphi\|_{L^{2}(\Omega_{\varepsilon})}^{2} + \int_{\Gamma_{\varepsilon}} \{(\nabla\varphi\cdot\nabla)\nabla\varphi\cdot n_{\varepsilon} - (\nabla\varphi\cdot n_{\varepsilon})\Delta\varphi\} \, d\mathcal{H}^{2} \\ &= \|\Delta\varphi\|_{L^{2}(\Omega_{\varepsilon})}^{2} + \int_{\Gamma_{\varepsilon}} (\nabla\varphi\cdot\nabla)\nabla\varphi\cdot n_{\varepsilon} \, d\mathcal{H}^{2}. \end{aligned}$$
(5.5.12)

Here we used (5.5.11) in the second equality. Moreover, based on (5.5.11) we can show as in the proof of Lemma 5.4.1 (see (5.4.12)) that

$$\left|\int_{\Gamma_{\varepsilon}} (\nabla \varphi \cdot \nabla) \nabla \varphi \cdot n_{\varepsilon} \, d\mathcal{H}^2\right| \le c \left(\|\nabla \varphi\|_{L^2(\Omega_{\varepsilon})}^2 + \|\nabla \varphi\|_{L^2(\Omega_{\varepsilon})} \|\nabla^2 \varphi\|_{L^2(\Omega_{\varepsilon})} \right).$$

Applying this inequality to (5.5.12) and using Young's inequality we obtain

$$\|\nabla^2 \varphi\|_{L^2(\Omega_{\varepsilon})}^2 \le \|\Delta \varphi\|_{L^2(\Omega_{\varepsilon})}^2 + c\|\nabla \varphi\|_{L^2(\Omega_{\varepsilon})}^2 + \frac{1}{2}\|\nabla^2 \varphi\|_{L^2(\Omega_{\varepsilon})},$$

which yields (5.5.10). Hence the inequality (5.5.9) follows.

Now let us derive the uniform estimate (5.5.1) for the difference $u - \mathbb{P}_{\varepsilon} u$.

Proof of Lemma 5.5.1. Let ε_{σ} be the constant given in Lemma 5.5.2 and $\varepsilon \in (0, \varepsilon_{\sigma})$. Suppose that $u \in H^1(\Omega_{\varepsilon})^3$ satisfies $u \cdot n_{\varepsilon} = 0$ on Γ_{ε} . Then for $\xi := -\operatorname{div} u \in L^2(\Omega_{\varepsilon})$ the divergence theorem implies that

$$\langle \xi, 1 \rangle_{\Omega_{\varepsilon}} = \int_{\Omega_{\varepsilon}} \xi \, dx = -\int_{\Gamma_{\varepsilon}} u \cdot n_{\varepsilon} \, d\mathcal{H}^2 = 0.$$

The Helmholtz-Leray projection of u is given by $\mathbb{P}_{\varepsilon} u = u - \nabla \varphi$, where $\varphi \in H^1(\Omega_{\varepsilon})$ is a unique weak solution to (5.5.6) with $\xi = -\text{div } u$. Since $\xi \in L^2(\Omega_{\varepsilon})$, Lemma 5.5.3 yields

$$\|u - \mathbb{P}_{\varepsilon} u\|_{H^1(\Omega_{\varepsilon})} = \|\nabla \varphi\|_{H^1(\Omega_{\varepsilon})} \le c \|\xi\|_{L^2(\Omega_{\varepsilon})} = c \|\operatorname{div} u\|_{L^2(\Omega_{\varepsilon})}$$

with a constant c > 0 independent of ε . Hence (5.5.1) is valid.

5.5.2 Definition and basic properties of the Stokes operator

Let us define the Stokes operator associated with slip boundary conditions and give its basic properties. For $u_1 \in H^2(\Omega_{\varepsilon})^3$ and $u_2 \in H^1(\Omega_{\varepsilon})^3$ integration by parts yields

$$\int_{\Omega_{\varepsilon}} \{\Delta u_1 + \nabla(\operatorname{div} u_1)\} \cdot u_2 \, dx = -2 \int_{\Omega_{\varepsilon}} D(u_1) : D(u_2) \, dx + 2 \int_{\Gamma_{\varepsilon}} [D(u_1)n_{\varepsilon}] \cdot u_2 \, d\mathcal{H}^2.$$
(5.5.13)

Hence if u_1 and u_2 satisfy div $u_1 = 0$ in Ω_{ε} and

$$u_1 \cdot n_{\varepsilon} = 0, \quad 2\nu P_{\varepsilon} D(u_1) n_{\varepsilon} + \gamma_{\varepsilon} u_1 = 0, \quad u_2 \cdot n_{\varepsilon} = 0 \quad \text{on} \quad \Gamma_{\varepsilon},$$

then from the above identity we have

$$\nu \int_{\Omega_{\varepsilon}} \Delta u_1 \cdot u_2 \, dx = -2\nu \int_{\Omega_{\varepsilon}} D(u_1) : D(u_2) \, dx - \sum_{i=0,1} \gamma_{\varepsilon}^i \int_{\Gamma_{\varepsilon}^i} u_1 \cdot u_2 \, d\mathcal{H}^2.$$

Based on this observation we define a bilinear form

$$a_{\varepsilon}(u_1, u_2) := 2\nu \int_{\Omega_{\varepsilon}} D(u_1) : D(u_2) \, dx + \sum_{i=0,1} \gamma_{\varepsilon}^i \int_{\Gamma_{\varepsilon}^i} u_1 \cdot u_2 \, d\mathcal{H}^2 \tag{5.5.14}$$

for $u_1, u_2 \in H^1(\Omega_{\varepsilon})^3$. By definition, a_{ε} is symmetric on $H^1(\Omega_{\varepsilon})^3$.

Lemma 5.5.4. Under Assumptions 1 and 2, there exist $\varepsilon_0 \in (0,1)$ and c > 0 such that

$$c^{-1} \|u\|_{H^{1}(\Omega_{\varepsilon})}^{2} \leq a_{\varepsilon}(u, u) \leq c \|u\|_{H^{1}(\Omega_{\varepsilon})}^{2}$$
(5.5.15)

for all $\varepsilon \in (0, \varepsilon_0)$ and $u \in H^1(\Omega_{\varepsilon})^3$ satisfying (5.3.20) on Γ_{ε} .

Proof. Let $u \in H^1(\Omega_{\varepsilon})^3$. By (5.1.6) in Assumption 1 and (5.3.8) we have

$$\gamma_{\varepsilon}^{i} \|u\|_{L^{2}(\Gamma_{\varepsilon}^{i})}^{2} \leq c\varepsilon \left(\varepsilon^{-1} \|u\|_{L^{2}(\Omega_{\varepsilon})}^{2} + \varepsilon \|\partial_{n}u\|_{L^{2}(\Omega_{\varepsilon})}^{2}\right) \leq c \|u\|_{H^{1}(\Omega_{\varepsilon})}^{2}$$

for i = 0, 1. Combining this with

$$\|D(u)\|_{L^2(\Omega_{\varepsilon})}^2 = \frac{1}{2} \left(\|\nabla u\|_{L^2(\Omega_{\varepsilon})}^2 + \int_{\Omega_{\varepsilon}} \nabla u : (\nabla u)^T \, dx \right) \le \|\nabla u\|_{L^2(\Omega_{\varepsilon})}^2$$

by Hölder's inequality we get the right-hand inequality of (5.5.15).

Let us prove the left-hand inequality of (5.5.15). First we assume that the condition (A1) in Assumption 2 is satisfied. Without loss of generality, we may assume that $\gamma_{\varepsilon}^{0} = \max_{i=0,1} \gamma_{\varepsilon}^{i}$. For $u \in H^{1}(\Omega_{\varepsilon})^{3}$ we use (5.3.7) with i = 0 to get

$$\|u\|_{L^{2}(\Omega_{\varepsilon})}^{2} \leq c \left(\varepsilon \|u\|_{L^{2}(\Gamma_{\varepsilon}^{0})}^{2} + \varepsilon^{2} \|\nabla u\|_{L^{2}(\Omega_{\varepsilon})}^{2}\right).$$

To the right-hand side we apply (5.1.7) and (5.4.1). Then

$$\begin{aligned} \|u\|_{L^{2}(\Omega_{\varepsilon})}^{2} &\leq c \left(\gamma_{\varepsilon}^{0} \|u\|_{L^{2}(\Gamma_{\varepsilon}^{0})}^{2} + \varepsilon^{2} \|D(u)\|_{L^{2}(\Omega_{\varepsilon})}^{2} + \varepsilon^{2} \|u\|_{L^{2}(\Omega_{\varepsilon})}^{2} \right) \\ &\leq c_{1}a_{\varepsilon}(u, u) + c_{2}\varepsilon^{2} \|u\|_{L^{2}(\Omega_{\varepsilon})}^{2} \end{aligned}$$

with positive constants c_1 and c_2 independent of ε . We set $\varepsilon_1 := 1/\sqrt{2c_2}$ and take $\varepsilon \in (0, \varepsilon_1)$ in the above inequality to get

$$\|u\|_{L^2(\Omega_{\varepsilon})}^2 \le 2c_1 a_{\varepsilon}(u, u)$$

Moreover, from this inequality and (5.4.1) it follows that

$$\|\nabla u\|_{L^2(\Omega_{\varepsilon})}^2 \le ca_{\varepsilon}(u, u).$$

By the above two inequalities we see that the left-hand inequality of (5.5.15) is valid.

Next we suppose that the condition (A2) in Assumption 2 holds. In this case the condition (5.4.14) is automatically satisfied for any $\beta \in [0, 1)$. We fix $\beta \in [0, 1)$ and apply Lemma 5.4.5 to obtain

$$\|u\|_{H^1(\Omega_{\varepsilon})}^2 \le c_{\beta} \|D(u)\|_{L^2(\Omega_{\varepsilon})}^2 \le c_{\beta} a_{\varepsilon}(u, u)$$

for all $\varepsilon \in (0, \varepsilon_{\beta})$ and $u \in H^1(\Omega_{\varepsilon})^3$ satisfying (5.3.20) on Γ_{ε} , where $\varepsilon_{\beta} \in (0, 1)$ and $c_{\beta} > 0$ are the constants given in Lemma 5.4.5. Hence the left-hand inequality of (5.5.15) holds and we conclude that the lemma is valid with $\varepsilon_0 := \min\{\varepsilon_1, \varepsilon_{\beta}\}$.

Throughout the rest of this section (except for Lemmas 5.5.6 and 5.5.7) we suppose that Assumptions 1 and 2 are satisfied and fix the number $\varepsilon_0 \in (0, 1)$ given in Lemma 5.5.4. For $\varepsilon \in (0, \varepsilon_0)$ we define a function space

$$V_{\varepsilon} := L^2_{\sigma}(\Omega_{\varepsilon}) \cap H^1(\Omega_{\varepsilon})^3 = \{ u \in H^1(\Omega_{\varepsilon})^3 \mid \text{div}\, u = 0 \text{ in } \Omega_{\varepsilon}, \, u \cdot n_{\varepsilon} = 0 \text{ on } \Gamma_{\varepsilon} \}.$$

By Lemma 5.5.4 the bilinear form a_{ε} restricted to $V_{\varepsilon} \times V_{\varepsilon}$ is continuous, coercive, and symmetric on the Hilbert space V_{ε} (equipped with $H^1(\Omega_{\varepsilon})$ -inner product). Hence the Lax– Milgram theorem yields that there exists a bounded linear operator A_{ε} from V_{ε} into its dual space V'_{ε} such that

$$V_{\varepsilon} \langle A_{\varepsilon} u_1, u_2 \rangle_{V_{\varepsilon}} = a_{\varepsilon}(u_1, u_2), \quad u_1, u_2 \in V_{\varepsilon},$$

where $_{V_{\varepsilon}'}\langle \cdot, \cdot \rangle_{V_{\varepsilon}}$ stands for the duality product between V_{ε}' and V_{ε} (see e.g. [8]). We consider A_{ε} as an unbounded operator on $L^2_{\sigma}(\Omega_{\varepsilon})$ with its domain

$$D(A_{\varepsilon}) = \{ u \in V_{\varepsilon} \mid A_{\varepsilon}u \in L^2_{\sigma}(\Omega_{\varepsilon}) \}.$$

Then from the Lax–Milgram theory it follows that A_{ε} is a positive self-adjoint operator on $L^2_{\sigma}(\Omega_{\varepsilon})$. Moreover, by a regularity result on the Stokes problem with slip boundary conditions (see [5,60]) we know that

$$D(A_{\varepsilon}) = \{ u \in V_{\varepsilon} \cap H^{2}(\Omega_{\varepsilon})^{3} \mid 2\nu P_{\varepsilon}D(u)n_{\varepsilon} + \gamma_{\varepsilon}u = 0 \text{ on } \Gamma_{\varepsilon} \}, A_{\varepsilon}u = -\nu \mathbb{P}_{\epsilon}\Delta u \quad \text{for} \quad u \in D(A_{\varepsilon}).$$

Here \mathbb{P}_{ε} is the Helmholtz–Leray projection (see Section 5.5.1). Note that $D(A_{\varepsilon}^{1/2}) = V_{\varepsilon}$ and

$$(A_{\varepsilon}u_1, u_2)_{L^2(\Omega_{\varepsilon})} = (A_{\varepsilon}^{1/2}u_1, A_{\varepsilon}^{1/2}u_2)_{L^2(\Omega_{\varepsilon})}$$
(5.5.16)

for all $u_1 \in D(A_{\varepsilon})$ and $u_2 \in V_{\varepsilon}$. Moreover, the identity

$$\|A_{\varepsilon}^{1/2}u\|_{L^{2}(\Omega_{\varepsilon})}^{2} = a_{\varepsilon}(u,u) = 2\nu \|D(u)\|_{L^{2}(\Omega_{\varepsilon})}^{2} + \gamma_{\varepsilon}^{0}\|u\|_{L^{2}(\Gamma_{\varepsilon}^{0})}^{2} + \gamma_{\varepsilon}^{1}\|u\|_{L^{2}(\Gamma_{\varepsilon}^{1})}^{2}$$
(5.5.17)

holds for all $u \in V_{\varepsilon}$.

Lemma 5.5.5. There exists a constant c > 0 independent of ε such that

$$c^{-1} \|u\|_{H^{1}(\Omega_{\varepsilon})} \leq \|A_{\varepsilon}^{1/2}u\|_{L^{2}(\Omega_{\varepsilon})} \leq c \|u\|_{H^{1}(\Omega_{\varepsilon})}$$
(5.5.18)

for all $\varepsilon \in (0, \varepsilon_0)$ and $u \in V_{\varepsilon}$. Moreover, if $u \in D(A_{\varepsilon})$, then we have

$$\|A_{\varepsilon}^{1/2}u\|_{L^2(\Omega_{\varepsilon})} \le c\|A_{\varepsilon}u\|_{L^2(\Omega_{\varepsilon})}.$$
(5.5.19)

Proof. The inequality (5.5.18) is an immediate consequence of (5.5.15) and (5.5.17). To prove (5.5.19) we see by Hölder's inequality that

$$\|A_{\varepsilon}^{1/2}u\|_{L^{2}(\Omega_{\varepsilon})}^{2} = (u, A_{\varepsilon}u)_{L^{2}(\Omega_{\varepsilon})} \leq \|u\|_{L^{2}(\Omega_{\varepsilon})}\|A_{\varepsilon}u\|_{L^{2}(\Omega_{\varepsilon})}$$

for $u \in D(A_{\varepsilon})$. Applying $||u||_{L^2(\Omega_{\varepsilon})} \leq c ||A_{\varepsilon}^{1/2}u||_{L^2(\Omega_{\varepsilon})}$ by (5.5.18) to the right-hand side we obtain (5.5.19).

5.5.3 Uniform regularity estimates for the Stokes operator

In this subsection we estimate the difference between the Stokes and Laplace operators and show the uniform equivalence of the norms $||A_{\varepsilon}u||_{L^2(\Omega_{\varepsilon})}$ and $||u||_{H^2(\Omega_{\varepsilon})}$ for $u \in D(A_{\varepsilon})$.

First we give an integration by parts formula for the curl of a vector field on Ω_{ε} . For $x \in N$ we set

$$\begin{split} \tilde{n}_1(x) &:= \frac{1}{\varepsilon \bar{g}(x)} \left\{ \left(d(x) - \varepsilon \bar{g}_0(x) \right) \bar{n}_{\varepsilon}^1(x) - \left(\varepsilon \bar{g}_1(x) - d(x) \right) \bar{n}_{\varepsilon}^0(x) \right\}, \\ \tilde{n}_2(x) &:= \frac{1}{\varepsilon \bar{g}(x)} \left\{ \left(d(x) - \varepsilon \bar{g}_0(x) \right) \frac{\gamma_{\varepsilon}^1}{\nu} \bar{n}_{\varepsilon}^1(x) + \left(\varepsilon \bar{g}_1(x) - d(x) \right) \frac{\gamma_{\varepsilon}^0}{\nu} \bar{n}_{\varepsilon}^0(x) \right\}, \\ \widetilde{W}(x) &:= \frac{1}{\varepsilon \bar{g}(x)} \left\{ \left(d(x) - \varepsilon \bar{g}_0(x) \right) W_{\varepsilon}^1(x) - \left(\varepsilon \bar{g}_1(x) - d(x) \right) W_{\varepsilon}^0(x) \right\}. \end{split}$$

Where n_{ε}^{i} , i = 0, 1 is given by (5.2.33) and

$$W^i_{\varepsilon}(x) := -\{I_3 - \bar{n}^i_{\varepsilon}(x) \otimes \bar{n}^i_{\varepsilon}(x)\} \nabla \bar{n}^i_{\varepsilon}(x), \quad x \in N, \, i = 0, 1.$$

By definition it immediately follows that

$$\tilde{n}_1 = (-1)^{i+1} n_{\varepsilon}, \quad \tilde{n}_2 = \frac{\gamma_{\varepsilon}}{\nu} n_{\varepsilon}, \quad \widetilde{W} = (-1)^{i+1} W_{\varepsilon} \quad \text{on} \quad \Gamma^i_{\varepsilon}, \, i = 0, 1.$$
 (5.5.20)

Lemma 5.5.6. For a vector field $u: \Omega_{\varepsilon} \to \mathbb{R}^3$ define $G(u) := G_1(u) + G_2(u)$, where

$$G_1(u) := 2\tilde{n}_1 \times \widetilde{W}u, \quad G_2(u) := \tilde{n}_2 \times u \quad in \quad \Omega_{\varepsilon}.$$
(5.5.21)

Then, under Assumption 1 that (5.1.6) is valid, there exists c > 0 independent of ε such that

$$|G(u)| \le c|u|, \quad |\nabla G(u)| \le c(|u| + |\nabla u|) \quad in \quad \Omega_{\varepsilon}.$$
(5.5.22)

Proof. By (5.2.13), (5.2.15), (5.2.36), and $g_0, g_1 \in C^4(\Gamma)$ we see that

$$\left|\partial_x^{\alpha}\bar{n}_{\varepsilon}^i(x)\right| \le c, \quad \left|\partial_x^{\alpha}\bar{g}_i(x)\right| \le c, \quad x \in N, \, i = 0, 1, \, |\alpha| = 0, 1, 2, \tag{5.5.23}$$

where $\partial_x^{\alpha} = \partial_1^{\alpha_1} \partial_2^{\alpha_2} \partial_3^{\alpha_3}$ for $\alpha = (\alpha_1, \alpha_2, \alpha_3) \in \mathbb{Z}^3$ with $\alpha_i \ge 0$, i = 1, 2, 3 and c > 0 is a constant independent of ε . From this it also follows that

$$|W_{\varepsilon}^{i}(x)| \le c, \quad |\partial_{k}W_{\varepsilon}^{i}(x)| \le c, \quad x \in N, \, i = 0, 1, \, k = 1, 2, 3.$$
 (5.5.24)

By (5.1.6), (5.3.39), (5.5.23), and (5.5.24) we have

$$|\tilde{n}_1| \le c, \quad |\tilde{n}_2| \le c\varepsilon, \quad \left|\widetilde{W}\right| \le c \quad \text{in} \quad \Omega_{\varepsilon}.$$
 (5.5.25)

Thus the first inequality of (5.5.22) is valid. To prove the second inequality of (5.5.22) we need to estimate the first order derivatives of \tilde{n}_1 , \tilde{n}_2 , and \widetilde{W} . As in the proof of Lemma 5.3.10, by direct calculations with $\nabla d = \bar{n}$ in N and the inequalities (5.2.30), (5.3.39), (5.5.23), and (5.5.24) we observe that

$$\nabla \tilde{n}_1 = \frac{1}{\varepsilon \bar{g}} \bar{n} \otimes (\bar{n}_{\varepsilon}^0 + \bar{n}_{\varepsilon}^1) + f_1, \quad \nabla \tilde{n}_2 = \frac{1}{\varepsilon \bar{g}} \bar{n} \otimes \left(\frac{\gamma_{\varepsilon}^1}{\nu} \bar{n}_{\varepsilon}^1 - \frac{\gamma_{\varepsilon}^0}{\nu} \bar{n}_{\varepsilon}^0\right) + f_2,$$
$$\partial_k \widetilde{W} = \frac{1}{\varepsilon \bar{g}} \bar{n}_k (W_{\varepsilon}^0 + W_{\varepsilon}^1) + F_k, \quad k = 1, 2, 3$$

in Ω_{ε} , where f_1 , f_2 , and F_k are bounded on Ω_{ε} uniformly in ε . We apply (5.2.37) to $\nabla \tilde{n}_1$, (5.1.6) and (5.2.36) to $\nabla \tilde{n}_2$, and use (5.2.30) to show that

$$|\nabla \tilde{n}_1| \le \frac{1}{\varepsilon \bar{g}} |\bar{n}_{\varepsilon}^0 + \bar{n}_{\varepsilon}^1| + |f_1| \le c, \quad |\nabla \tilde{n}_2| \le \frac{c(\gamma_{\varepsilon}^0 + \gamma_{\varepsilon}^1)}{\varepsilon \bar{g}} + |f_2| \le c$$
(5.5.26)

in Ω_{ε} . To estimate the first order derivatives of \widetilde{W} we see by (5.2.12) that

$$W_{\varepsilon}^{0} + W_{\varepsilon}^{1} = \{ (\bar{n}_{\varepsilon}^{0} + \bar{n}_{\varepsilon}^{1}) \otimes \bar{n}_{\varepsilon}^{0} - \bar{n}_{\varepsilon}^{1} \otimes (\bar{n}_{\varepsilon}^{0} + \bar{n}_{\varepsilon}^{1}) \} \left(I_{3} - d\overline{W} \right)^{-1} \overline{\nabla_{\Gamma} n_{\varepsilon}^{0}} - (I_{3} - \bar{n}_{\varepsilon}^{1} \otimes \bar{n}_{\varepsilon}^{1}) \left(I_{3} - d\overline{W} \right)^{-1} \left(\overline{\nabla_{\Gamma} n_{\varepsilon}^{0}} + \overline{\nabla_{\Gamma} n_{\varepsilon}^{1}} \right)$$

in N. Hence we get $|W_{\varepsilon}^0 + W_{\varepsilon}^1| \leq c\varepsilon$ in N by (5.2.9), (5.2.36), and (5.2.37) and

$$\left|\partial_k \widetilde{W}\right| \le \frac{1}{\varepsilon \overline{g}} |W_{\varepsilon}^0 + W_{\varepsilon}^1| + |F_k| \le c \quad \text{in} \quad \Omega_{\varepsilon}.$$
(5.5.27)

Applying (5.5.25), (5.5.26), and (5.5.27) to the gradients of G_1 and G_2 given by (5.5.21) we obtain the second inequality of (5.5.22).

Lemma 5.5.7. We have the integration by parts formula

$$\int_{\Omega_{\varepsilon}} \operatorname{curl}\operatorname{curl} u \cdot \Phi \, dx = -\int_{\Omega_{\varepsilon}} \operatorname{curl} G(u) \cdot \Phi \, dx + \int_{\Omega_{\varepsilon}} \{\operatorname{curl} u + G(u)\} \cdot \operatorname{curl} \Phi \, dx \qquad (5.5.28)$$

for all $u \in H^2(\Omega_{\varepsilon})^3$ satisfying the slip boundary conditions (5.3.20)–(5.3.21) on Γ_{ε} and $\Phi \in L^2(\Omega_{\varepsilon})^3$ with curl $\Phi \in L^2(\Omega_{\varepsilon})^3$, where G(u) is given in Lemma 5.5.6.

Proof. By standard cut-off, dilatation, and mollification arguments, we can show as in the proof of [64, Chapter 1, Theorem 1.1] that for $\Phi \in L^2(\Omega_{\varepsilon})^3$ with $\operatorname{curl} \Phi \in L^2(\Omega_{\varepsilon})^3$ there exists a sequence $\{\Phi_k\}_{k=1}^{\infty}$ of vector fields in $C^{\infty}(\overline{\Omega}_{\varepsilon})^3$ such that

$$\lim_{k \to \infty} \|\Phi - \Phi_k\|_{L^2(\Omega_{\varepsilon})} = \lim_{k \to \infty} \|\operatorname{curl} \Phi - \operatorname{curl} \Phi_k\|_{L^2(\Omega_{\varepsilon})} = 0.$$

Thus, by a density argument, it is sufficient to prove (5.5.28) for all $\Phi \in C^{\infty}(\overline{\Omega}_{\varepsilon})^3$.

Let $u \in H^2(\Omega_{\varepsilon})^3$ satisfy (5.3.20) and (5.3.21) on Γ_{ε} and $\Phi \in C^{\infty}(\overline{\Omega}_{\varepsilon})^3$. Then

$$\int_{\Omega_{\varepsilon}} \operatorname{curl}\operatorname{curl} u \cdot \Phi \, dx = \int_{\Gamma_{\varepsilon}} (n_{\varepsilon} \times \operatorname{curl} u) \cdot \Phi \, d\mathcal{H}^2 + \int_{\Omega_{\varepsilon}} \operatorname{curl} u \cdot \operatorname{curl} \Phi \, dx \tag{5.5.29}$$

by integration by parts. To the boundary integral we apply

$$n_{\varepsilon} \times \operatorname{curl} u = -n_{\varepsilon} \times \left\{ n_{\varepsilon} \times \left(2W_{\varepsilon}u + \frac{\gamma_{\varepsilon}}{\nu}u \right) \right\}$$
$$= -n_{\varepsilon} \times \left(2\tilde{n}_{1} \times \widetilde{W}u + \tilde{n}_{2} \times u \right) = -n_{\varepsilon} \times G(u)$$

on Γ_{ε} by (5.3.30), (5.5.20), and (5.5.21), and then use integration by parts to get

$$\int_{\Gamma_{\varepsilon}} (n_{\varepsilon} \times \operatorname{curl} u) \cdot \Phi \, d\mathcal{H}^2 = -\int_{\Gamma_{\varepsilon}} \{n_{\varepsilon} \times G(u)\} \cdot \Phi \, d\mathcal{H}^2$$
$$= \int_{\Omega_{\varepsilon}} \{G(u) \cdot \operatorname{curl} \Phi - \operatorname{curl} G(u) \cdot \Phi\} \, dx.$$

Substituting this for (5.5.29) we obtain (5.5.28)

Using the formula (5.5.28), we derive an estimate for the difference between the Stokes and Laplace operators as in [19, Theorem 2.1].

Lemma 5.5.8. There exists a constant c > 0 independent of ε such that

$$\|A_{\varepsilon}u + \nu\Delta u\|_{L^2(\Omega_{\varepsilon})} \le c\|u\|_{H^1(\Omega_{\varepsilon})}$$
(5.5.30)

for all $\varepsilon \in (0, \varepsilon_0)$ and $u \in D(A_{\varepsilon})$.

Note that the $H^1(\Omega_{\varepsilon})$ -norm of u appears in the right-hand side of (5.5.30) instead of its $H^2(\Omega_{\varepsilon})$ -norm.

Proof. Let $u \in D(A_{\varepsilon})$. Since $A_{\varepsilon}u = -\nu \mathbb{P}_{\varepsilon} \Delta u \in L^2_{\sigma}(\Omega_{\varepsilon})$, there exists $q \in H^1(\Omega_{\varepsilon})$ such that $A_{\varepsilon}u + \nu \Delta u = \nabla q$. Then by $A_{\varepsilon}u \in L^2_{\sigma}(\Omega_{\varepsilon})$ and $\nabla q \in L^2_{\sigma}(\Omega_{\varepsilon})^{\perp}$ we get

$$\|A_{\varepsilon}u + \nu\Delta u\|_{L^{2}(\Omega_{\varepsilon})}^{2} = (A_{\varepsilon}u, \nabla q)_{L^{2}(\Omega_{\varepsilon})} + (\nu\Delta u, \nabla q)_{L^{2}(\Omega_{\varepsilon})} = (\nu\Delta u, \nabla q)_{L^{2}(\Omega_{\varepsilon})}.$$

Moreover, noting that

$$\operatorname{curl} \nabla q = 0, \quad \Delta u = -\operatorname{curl} \operatorname{curl} u \quad \text{in} \quad \Omega_{\varepsilon}$$

by div u = 0, we apply (5.5.28) with $\Phi = -\nabla q$ to the last term to get

$$(\nu\Delta u, \nabla q)_{L^{2}(\Omega_{\varepsilon})} = \nu(\operatorname{curl}\operatorname{curl} u, -\nabla q)_{L^{2}(\Omega_{\varepsilon})}$$
$$= \nu(\operatorname{curl} G(u), \nabla q)_{L^{2}(\Omega_{\varepsilon})}$$
$$= \nu(\operatorname{curl} G(u), A_{\varepsilon}u + \nu\Delta u)_{L^{2}(\Omega_{\varepsilon})}$$

where G(u) is given in Lemma 5.5.6. From this equality and (5.5.22) we deduce that

$$\begin{aligned} \|A_{\varepsilon}u + \nu\Delta u\|_{L^{2}(\Omega_{\varepsilon})}^{2} &\leq \nu \|\operatorname{curl} G(u)\|_{L^{2}(\Omega_{\varepsilon})} \|A_{\varepsilon}u + \nu\Delta u\|_{L^{2}(\Omega_{\varepsilon})} \\ &\leq c \|u\|_{H^{1}(\Omega_{\varepsilon})} \|A_{\varepsilon}u + \nu\Delta u\|_{L^{2}(\Omega_{\varepsilon})} \end{aligned}$$

with some constant c > 0 independent of ε . Thus (5.5.30) follows.

Remark 5.5.9. In the proof of Lemma 5.5.8, Assumption 2 is essential since it enables us to consider the Stokes operator A_{ε} on the usual solenoidal space $L^2_{\sigma}(\Omega_{\varepsilon})$ and thus the curl of $A_{\varepsilon}u + \nu\Delta u = \nabla q$ vanishes. If we drop Assumption 2 and consider the Stokes operator \hat{A}_{ε} on

$$\widehat{L}^2_{\sigma}(\Omega_{\varepsilon}) = L^2_{\sigma}(\Omega_{\varepsilon}) \cap \overline{\mathcal{K}}_g(\Gamma)^{\perp}, \quad \overline{\mathcal{K}}_g(\Gamma) = \{ \overline{v} \mid v \in \mathcal{K}_g(\Gamma) \},\$$

where $\mathcal{K}_{q}(\Gamma)$ is given by (5.1.8) and $\bar{v} = v \circ \pi$ is the constant extension of v, then

$$\widehat{A}_{\varepsilon}u + \nu\Delta u = \nabla q + \overline{v}, \quad q \in H^1(\Omega_{\varepsilon}), \ \overline{v} \in L^2_{\sigma}(\Omega_{\varepsilon}) \cap \overline{K}_g(\Gamma)$$

for $u \in D(\widehat{A}_{\varepsilon})$. In this case, however, we cannot prove a similar inequality to (5.5.30) since the curl of \overline{v} does not vanish in general. This difficulty does not occur in the proof of [19, Theorem 2.1] since in that case $\mathcal{K}_g(\Gamma)$ reduces to

$$\mathcal{K}_g(\mathbb{T}^2) = \{ (a,0) \in \mathbb{R}^3 \mid a \in \mathbb{R}^2, \ a \cdot \nabla_2 g = 0 \text{ in } \mathbb{T}^2 \}$$

and the curl of the constant $(a, 0) \in \mathcal{K}_q(\mathbb{T}^2)$ automatically vanishes.

Next we show that for $u \in D(A_{\varepsilon})$ the norm $||A_{\varepsilon}u||_{L^{2}(\Omega_{\varepsilon})}$ is bounded from above and below by the canonical $H^{2}(\Omega_{\varepsilon})$ -norm of u with constants independent of ε .

Lemma 5.5.10. There exists a constant c > 0 independent of ε such that

$$\|u\|_{H^2(\Omega_{\varepsilon})} \le c \left(\|\Delta u\|_{L^2(\Omega_{\varepsilon})} + \|u\|_{H^1(\Omega_{\varepsilon})}\right)$$

$$(5.5.31)$$

for all $\varepsilon \in (0, \varepsilon_0)$ and $u \in H^2(\Omega_{\varepsilon})^3$ satisfying (5.3.20) and (5.3.21) on Γ_{ε} .

The proof of Lemma 5.5.10 is similar to that of Lemma 5.5.3, but we need to carry out calculations a lot and use some formulas for the Riemannian connection on Γ_{ε} . We give it in the next subsection.

Using Lemmas 5.5.8 and 5.5.10 we prove the uniform equivalence of the $L^2(\Omega_{\varepsilon})$ -norm of $A_{\varepsilon}u$ and the $H^2(\Omega_{\varepsilon})$ -norm of $u \in D(A_{\varepsilon})$.

Lemma 5.5.11. There exists a constant c > 0 independent of ε such that

$$c^{-1} \|u\|_{H^2(\Omega_{\varepsilon})} \le \|A_{\varepsilon}u\|_{L^2(\Omega_{\varepsilon})} \le c \|u\|_{H^2(\Omega_{\varepsilon})}$$
(5.5.32)

for all $\varepsilon \in (0, \varepsilon_0)$ and $u \in D(A_{\varepsilon})$.

Proof. By (5.5.30) and (5.5.31) we have

$$\begin{aligned} \|u\|_{H^{2}(\Omega_{\varepsilon})} &\leq c \left(\|\Delta u\|_{L^{2}(\Omega_{\varepsilon})} + \|u\|_{H^{1}(\Omega_{\varepsilon})} \right) \\ &\leq c \left(\|A_{\varepsilon}u\|_{L^{2}(\Omega_{\varepsilon})} + \|A_{\varepsilon}u + \nu\Delta u\|_{L^{2}(\Omega_{\varepsilon})} + \|u\|_{H^{1}(\Omega_{\varepsilon})} \right) \\ &\leq c \left(\|A_{\varepsilon}u\|_{L^{2}(\Omega_{\varepsilon})} + \|u\|_{H^{1}(\Omega_{\varepsilon})} \right). \end{aligned}$$

Applying (5.5.18) and (5.5.19) to the second term on the right-hand side we obtain the left-hand inequality of (5.5.32). Also, from (5.5.30) and $||u||_{H^1(\Omega_{\varepsilon})} \leq ||u||_{H^2(\Omega_{\varepsilon})}$ we deduce that

$$\|A_{\varepsilon}u\|_{L^{2}(\Omega_{\varepsilon})} \leq \|A_{\varepsilon}u + \nu\Delta u\|_{L^{2}(\Omega_{\varepsilon})} + \|\nu\Delta u\|_{L^{2}(\Omega_{\varepsilon})} \leq c\|u\|_{H^{2}(\Omega_{\varepsilon})}$$

Hence the right-hand inequality of (5.5.32) holds.

As a consequence of Lemmas 5.5.5 and 5.5.11, we obtain an interpolation inequality for a vector field in $D(A_{\varepsilon})$.

Lemma 5.5.12. There exists a constant c > 0 independent of ε such that

$$\|u\|_{H^1(\Omega_{\varepsilon})} \le c \|u\|_{L^2(\Omega_{\varepsilon})}^{1/2} \|u\|_{H^2(\Omega_{\varepsilon})}^{1/2}$$
(5.5.33)

for all $\varepsilon \in (0, \varepsilon_0)$ and $u \in D(A_{\varepsilon})$.

Proof. Let $u \in D(A_{\varepsilon})$. From (5.5.16) and (5.5.18) it follows that

$$\|u\|_{H^1(\Omega_{\varepsilon})}^2 \le c\|A_{\varepsilon}^{1/2}u\|_{L^2(\Omega_{\varepsilon})}^2 = c(A_{\varepsilon}u, u)_{L^2(\Omega_{\varepsilon})} \le c\|A_{\varepsilon}u\|_{L^2(\Omega_{\varepsilon})}\|u\|_{L^2(\Omega_{\varepsilon})}.$$

Applying (5.5.32) to the right-hand side of this inequality we get

$$\|u\|_{H^1(\Omega_{\varepsilon})}^2 \le c \|u\|_{L^2(\Omega_{\varepsilon})} \|u\|_{H^2(\Omega_{\varepsilon})}.$$

Hence (5.5.33) is valid.

5.5.4 Uniform a priori estimate for the vector Laplacian

The purpose of this subsection is to give the proof of Lemma 5.5.10. First we give an approximation result of a vector field in $H^2(\Omega_{\varepsilon})^3$ satisfying the slip boundary conditions. Let us consider the equilibrium equations of linear elasticity with slip boundary conditions

$$\begin{cases} \Delta u + \nabla(\operatorname{div} u) = f & \text{in } \Omega_{\varepsilon}, \\ u \cdot n_{\varepsilon} = 0, \quad 2\nu P_{\varepsilon} D(u) n_{\varepsilon} + \gamma_{\varepsilon} u = 0 & \text{on } \Gamma_{\varepsilon}, \end{cases}$$
(5.5.34)

where $f: \Omega_{\varepsilon} \to \mathbb{R}^3$ is an external force (and we artificially impose the slip boundary conditions). By the identity (5.5.13) and the slip boundary conditions we observe that the bilinear form corresponding to the problem (5.5.34) is a_{ε} given by (5.5.14). Thus, under Assumptions 1 and 2, Lemma 5.5.4 and the Lax–Milgram theory show that for each $f \in H^{-1}(\Omega_{\varepsilon})^3$ there exists a unique weak solution $u \in H^1(\Omega_{\varepsilon})^3$ satisfying $u \cdot n_{\varepsilon} = 0$ on Γ_{ε} and

$$a_{\varepsilon}(u, \Phi) = \langle f, \Phi \rangle_{\Omega_{\varepsilon}} \text{ for all } \Phi \in H^1(\Omega_{\varepsilon})^3,$$

where $\langle \cdot, \cdot \rangle_{\Omega_{\varepsilon}}$ stands for the duality product between $H^{-1}(\Omega_{\varepsilon})$ and $H^{1}(\Omega_{\varepsilon})$.

Lemma 5.5.13. Let $u \in H^1(\Omega_{\varepsilon})^3$ be a unique weak solution to the problem (5.5.34). Assume that $f \in L^2(\Omega_{\varepsilon})^3$. Then $u \in H^2(\Omega_{\varepsilon})^3$ and it satisfies (5.5.34) a.e. in Ω_{ε} and on Γ_{ε} . Moreover, there exists a constant $c_{\varepsilon} > 0$ depending on ε such that

$$\|u\|_{H^2(\Omega_{\varepsilon})} \le c_{\varepsilon} \|f\|_{L^2(\Omega_{\varepsilon})}.$$
(5.5.35)

If in addition $f \in H^1(\Omega_{\varepsilon})^3$, then $u \in H^3(\Omega_{\varepsilon})^3$.

Proof. The inequality (5.5.35) and the H^2 -regularity of u are proved by a standard localization argument and a method of the difference quotient. Here we omit their proofs since they are the same as those of [5, Theorem 1.2] and [60, Theorem 2], which establish the H^2 -regularity of a weak solution to the Stokes problem with slip boundary conditions.

Also, the H^3 -regularity of u is proved by induction and a localization argument as in the case of a general second order elliptic equation. For details, see [13, Theorem 5 in Section 6.3]. (Note that the C^4 -regularity of the boundary Γ_{ε} is required for the H^3 -regularity of u, see Section 5.2.2 and the proofs of [5, Theorem 1.2] and [60, Theorem 2].)

Based on Lemma 5.5.13 we show that a vector field in $H^2(\Omega_{\varepsilon})^3$ is approximated by those in $H^3(\Omega_{\varepsilon})^3$ under the slip boundary conditions.

Lemma 5.5.14. Let $u \in H^2(\Omega_{\varepsilon})^3$ satisfy the slip boundary conditions (5.3.20)–(5.3.21) on Γ_{ε} . Then there exists a sequence $\{u_k\}_{k=1}^{\infty}$ in $H^3(\Omega_{\varepsilon})^3$ such that u_k satisfies (5.3.20)–(5.3.21) on Γ_{ε} for each $k \in \mathbb{N}$ and $\lim_{k\to\infty} ||u - u_k||_{H^2(\Omega_{\varepsilon})} = 0$.

Proof. Let $u \in H^2(\Omega_{\varepsilon})^3$ satisfy (5.3.20)–(5.3.21) on Γ_{ε} and $f := \Delta u + \nabla(\operatorname{div} u)$. Since $f \in L^2(\Omega_{\varepsilon})^3$, there exists a sequence $\{f_k\}_{k=1}^{\infty}$ in $C_c^{\infty}(\Omega_{\varepsilon})^3$ that converges to f strongly in $L^2(\Omega_{\varepsilon})^3$. For each $k \in \mathbb{N}$ let u_k be a unique weak solution to the problem (5.5.34) with external force f_k . Then since $f_k \in C_c^{\infty}(\Omega_{\varepsilon})^3$, we see by Lemma 5.5.13 that $u_k \in H^3(\Omega_{\varepsilon})^3$ and it satisfies (5.3.20)–(5.3.21) on Γ_{ε} . Moreover, by (5.5.35) and the fact that $u - u_k$ is a unique weak solution to (5.5.34) with external force $f - f_k$, we have

$$\|u - u_k\|_{H^2(\Omega_{\varepsilon})} \le c_{\varepsilon} \|f - f_k\|_{L^2(\Omega_{\varepsilon})}$$

Letting $k \to \infty$ in this inequality and using the strong convergence of $\{f_k\}_{k=1}^{\infty}$ to f in $L^2(\Omega_{\varepsilon})^3$ we obtain $\lim_{k\to\infty} \|u-u_k\|_{H^2(\Omega_{\varepsilon})} = 0$ (note that the constant c_{ε} does not depend on k). \Box

Now let us show Lemma 5.5.10. As in Section 5.2.1 we denote by

$$H^m(\Gamma_{\varepsilon}, T\Gamma_{\varepsilon}) := \{ u \in H^m(\Gamma_{\varepsilon})^3 \mid u \cdot n_{\varepsilon} = 0 \text{ on } \Gamma_{\varepsilon} \}, \quad m = 0, 1, 2$$

the space of all tangential vector fields on Γ_{ε} of class H^m (here we write $H^0 = L^2$). For $u \in H^1(\Gamma_{\varepsilon}, T\Gamma_{\varepsilon})$ and $v \in L^2(\Gamma_{\varepsilon}, T\Gamma_{\varepsilon})$ we define the covariant derivative

$$\overline{\nabla}_v^{\varepsilon} u := P_{\varepsilon}(v \cdot \nabla) \tilde{u} = P_{\varepsilon}(v \cdot \nabla_{\Gamma_{\varepsilon}}) u \quad \text{on} \quad \Gamma_{\varepsilon},$$

where \tilde{u} is any H^1 -extension of u to an open neighborhood of Γ_{ε} with $\tilde{u}|_{\Gamma_{\varepsilon}} = u$. We use the formulas for the covariant derivatives given in Appendix 5.C.

Proof of Lemma 5.5.10. Let $u \in H^2(\Omega_{\varepsilon})^3$ satisfy (5.3.20)–(5.3.21) on Γ_{ε} . Since

$$||u||_{H^{2}(\Omega_{\varepsilon})}^{2} = ||u||_{H^{1}(\Omega_{\varepsilon})}^{2} + ||\nabla^{2}u||_{L^{2}(\Omega_{\varepsilon})}^{2}$$

it is sufficient for (5.5.31) to show that

$$\|\nabla^2 u\|_{L^2(\Omega_{\varepsilon})}^2 \le c \left(\|\Delta u\|_{L^2(\Omega_{\varepsilon})}^2 + \|u\|_{H^1(\Omega_{\varepsilon})}^2 \right).$$
(5.5.36)

Moreover, by Lemma 5.5.14 we may assume that $u \in H^3(\Omega_{\varepsilon})^3$, and thus its trace on Γ_{ε} is in $H^2(\Gamma_{\varepsilon}, T\Gamma_{\varepsilon})$ (note that u satisfies $u \cdot n_{\varepsilon} = 0$ on Γ_{ε}). For such u we can carry out integration by parts twice to get

$$\|\nabla^2 u\|_{L^2(\Omega_{\varepsilon})}^2 = \|\Delta u\|_{L^2(\Omega_{\varepsilon})}^2 + \int_{\Gamma_{\varepsilon}} \nabla u : \{(n_{\varepsilon} \cdot \nabla)\nabla u - n_{\varepsilon} \otimes \Delta u\} \, d\mathcal{H}^2.$$
(5.5.37)

Here $(n_{\varepsilon} \cdot \nabla) \nabla u$ denotes a 3 × 3 matrix whose (i, j)-entry is given by

$$[(n_{\varepsilon} \cdot \nabla)\nabla u]_{ij} := (n_{\varepsilon} \cdot \nabla)\partial_i u_j, \quad i, j = 1, 2, 3.$$

Let us estimate the boundary integral in (5.5.37). Since u satisfies the slip boundary conditions (5.3.20)–(5.3.21) on Γ_{ε} , we see by Lemma 5.3.7 that

$$(n_{\varepsilon} \cdot \nabla)u = -W_{\varepsilon}u - \tilde{\gamma}_{\varepsilon}u + \xi_{\varepsilon}n_{\varepsilon} \quad \text{on} \quad \Gamma_{\varepsilon}, \qquad (5.5.38)$$

where $\tilde{\gamma}_{\varepsilon} := \gamma_{\varepsilon}/\nu$ and $\xi_{\varepsilon} := (n_{\varepsilon} \cdot \nabla)u \cdot n_{\varepsilon} = \nabla u : Q_{\varepsilon}$ (note that u and $W_{\varepsilon}u$ are tangential on Γ_{ε}). The first step is to show that

$$\int_{\Gamma_{\varepsilon}} \nabla u : \{ (n_{\varepsilon} \cdot \nabla) \nabla u - n_{\varepsilon} \otimes \Delta u \} \, d\mathcal{H}^2 = \sum_{k=1}^4 \int_{\Gamma_{\varepsilon}} \varphi_k \, d\mathcal{H}^2, \tag{5.5.39}$$

where

$$\begin{split} \varphi_{1} &:= -2\{\nabla_{\Gamma_{\varepsilon}}W_{\varepsilon} \cdot u + (\nabla u)W_{\varepsilon} + \tilde{\gamma_{\varepsilon}}\nabla u\} : P_{\varepsilon}(\nabla u)P_{\varepsilon}, \\ \varphi_{2} &:= W_{\varepsilon}\nabla u : (\nabla u)P_{\varepsilon} \\ &- 2(u \cdot \operatorname{div}_{\Gamma_{\varepsilon}}W_{\varepsilon} + 2\nabla u : W_{\varepsilon})(\nabla u : Q_{\varepsilon}) + H_{\varepsilon}(\nabla u : Q_{\varepsilon})^{2}, \\ \varphi_{3} &:= -(W_{\varepsilon}^{3}u - H_{\varepsilon}W_{\varepsilon}^{2}u) \cdot u, \\ \varphi_{4} &:= -\tilde{\gamma_{\varepsilon}}(2W_{\varepsilon}^{2}u - 2H_{\varepsilon}W_{\varepsilon}u - \tilde{\gamma_{\varepsilon}}H_{\varepsilon}u) \cdot u. \end{split}$$

$$(5.5.40)$$

In (5.5.40) we used the notation $\nabla_{\Gamma_{\varepsilon}} W_{\varepsilon} \cdot u$ for the 3 × 3 matrix given by

$$[\nabla_{\Gamma_{\varepsilon}} W_{\varepsilon} \cdot u]_{ij} := \sum_{k=1}^{3} (\underline{D}_{i}^{\varepsilon} [W_{\varepsilon}]_{jk}) u_{k}, \quad i, j = 1, 2, 3.$$
(5.5.41)

Using a partition of unity of Γ_{ε} we may assume that $u|_{\Gamma_{\varepsilon}}$ is compactly supported in a relatively open subset U of Γ_{ε} on which we can take a local orthonormal frame $\{\tau_1, \tau_2\}$ (see Appendix 5.C). Since $\{\tau_1, \tau_2, n_{\varepsilon}\}$ is an orthonormal basis of \mathbb{R}^3 ,

$$\nabla u : \{ (n_{\varepsilon} \cdot \nabla) \nabla u - n_{\varepsilon} \otimes \Delta u \} = (\nabla u)^T : [\{ (n_{\varepsilon} \cdot \nabla) \nabla u \}^T - \Delta u \otimes n_{\varepsilon}] = \sum_{i=1}^3 \eta_i \qquad (5.5.42)$$

on U, where

$$\eta_i := (\nabla u)^T \tau_i \cdot [\{ (n_\varepsilon \cdot \nabla) \nabla u \}^T \tau_i - (\Delta u \otimes n_\varepsilon) \tau_i], \quad i = 1, 2,$$
(5.5.43)

$$\eta_3 := (\nabla u)^T n_{\varepsilon} \cdot [\{(n_{\varepsilon} \cdot \nabla) \nabla u\}^T n_{\varepsilon} - (\Delta u \otimes n_{\varepsilon}) n_{\varepsilon}].$$
(5.5.44)

In what follows, we carry out calculations on U. By (5.C.2) and $\tau_i \cdot n_{\varepsilon} = 0$,

$$(\nabla u)^T \tau_i = (\tau_i \cdot \nabla) u = \overline{\nabla}_i^{\varepsilon} u + (W_{\varepsilon} u \cdot \tau_i) n_{\varepsilon}, (\Delta u \otimes n_{\varepsilon}) \tau_i = (\tau_i \cdot n_{\varepsilon}) \Delta u = 0,$$
(5.5.45)

where $\overline{\nabla}_{i}^{\varepsilon} := \overline{\nabla}_{\tau_{i}}^{\varepsilon}$, i = 1, 2. Writing τ_{i}^{j} and n_{ε}^{j} , j = 1, 2, 3 for the *j*-th component of τ_{i} and n_{ε} , we see that the *j*-th component of $\{(n_{\varepsilon} \cdot \nabla)\nabla u\}^{T}\tau_{i}$ is of the form

$$\begin{split} \sum_{k,l=1}^{3} n_{\varepsilon}^{k} (\partial_{k} \partial_{l} u_{j}) \tau_{i}^{l} &= \sum_{k=1}^{3} n_{\varepsilon}^{k} (\tau_{i} \cdot \nabla) (\partial_{k} u_{j}) = \sum_{k=1}^{3} n_{\varepsilon}^{k} (\tau_{i} \cdot \nabla_{\Gamma_{\varepsilon}}) (\partial_{k} u_{j}) \\ &= \sum_{k=1}^{3} \{ (\tau_{i} \cdot \nabla_{\Gamma_{\varepsilon}}) (n_{\varepsilon}^{k} \partial_{k} u_{j}) - (\tau_{i} \cdot \nabla_{\Gamma_{\varepsilon}} n_{\varepsilon}^{k}) \partial_{k} u_{j} \} \\ &= (\tau_{i} \cdot \nabla_{\Gamma_{\varepsilon}}) \{ (n_{\varepsilon} \cdot \nabla) u_{j} \} - \{ (\tau_{i} \cdot \nabla_{\Gamma_{\varepsilon}}) n_{\varepsilon} \cdot \nabla \} u_{j}. \end{split}$$

(Note that τ_i is tangential on Γ_{ε} and the tangential derivatives depend only on the values of functions on Γ_{ε}). Therefore,

$$\{(n_{\varepsilon}\cdot\nabla)\nabla u\}^{T}\tau_{i}=(\tau_{i}\cdot\nabla_{\Gamma_{\varepsilon}})\{(n_{\varepsilon}\cdot\nabla)u\}-\{(\tau_{i}\cdot\nabla_{\Gamma_{\varepsilon}})n_{\varepsilon}\cdot\nabla\}u$$

By (5.5.38), (5.C.2), $W_{\varepsilon}^T = W_{\varepsilon}$, and

$$(\tau_i \cdot \nabla_{\Gamma_{\varepsilon}}) n_{\varepsilon} = (\nabla_{\Gamma_{\varepsilon}} n_{\varepsilon})^T \tau_i = -W_{\varepsilon}^T \tau_i = -W_{\varepsilon} \tau_i, \qquad (5.5.46)$$

we further observe that

$$\{(n_{\varepsilon} \cdot \nabla)\nabla u\}^{T}\tau_{i} = -\overline{\nabla}_{i}^{\varepsilon}(W_{\varepsilon}u) - \tilde{\gamma}_{\varepsilon}\overline{\nabla}_{i}^{\varepsilon}u + \overline{\nabla}_{W_{\varepsilon}\tau_{i}}^{\varepsilon}u - \xi_{\varepsilon}W_{\varepsilon}\tau_{i} + \{(-\tilde{\gamma}_{\varepsilon}W_{\varepsilon}u + \nabla_{\Gamma_{\varepsilon}}\xi_{\varepsilon})\cdot\tau_{i}\}n_{\varepsilon}.$$
 (5.5.47)

Note that the first four terms on the right-hand side of (5.5.47) are tangential on Γ_{ε} . From (5.5.43), (5.5.45), and (5.5.47) we deduce that

$$\eta_i = -\overline{\nabla}_i^{\varepsilon} u \cdot \left\{ \overline{\nabla}_i^{\varepsilon} (W_{\varepsilon} u) + \tilde{\gamma}_{\varepsilon} \overline{\nabla}_i^{\varepsilon} u - \overline{\nabla}_{W_{\varepsilon} \tau_i}^{\varepsilon} u + \xi_{\varepsilon} W_{\varepsilon} \tau_i \right\} + (W_{\varepsilon} u \cdot \tau_i) \left\{ (-\tilde{\gamma}_{\varepsilon} W_{\varepsilon} u + \nabla_{\Gamma_{\varepsilon}} \xi_{\varepsilon}) \cdot \tau_i \right\}$$

for i = 1, 2. Hence by (5.C.9)–(5.C.13) and the fact that $\{\tau_1, \tau_2\}$ is an orthonormal basis of the tangent plane of Γ_{ε} we obtain

$$\eta_1 + \eta_2 = -\{\nabla_{\Gamma_{\varepsilon}}(W_{\varepsilon}u) + \tilde{\gamma}_{\varepsilon}\nabla_{\Gamma_{\varepsilon}}u - W_{\varepsilon}\nabla_{\Gamma_{\varepsilon}}u\} : (\nabla_{\Gamma_{\varepsilon}}u)P_{\varepsilon} \\ - \xi_{\varepsilon}(\nabla_{\Gamma_{\varepsilon}}u:W_{\varepsilon}) + W_{\varepsilon}u \cdot (-\tilde{\gamma}_{\varepsilon}W_{\varepsilon}u + \nabla_{\Gamma_{\varepsilon}}\xi_{\varepsilon}).$$
(5.5.48)

To calculate η_3 we observe that the *j*-th component of $\{(n_{\varepsilon} \cdot \nabla) \nabla u\}^T n_{\varepsilon}$ is given by

$$\sum_{k,l=1}^{3} n_{\varepsilon}^{k} (\partial_{k} \partial_{l} u_{j}) n_{\varepsilon}^{l} = \operatorname{tr}[Q_{\varepsilon} \nabla^{2} u_{j}] = \operatorname{tr}[\nabla^{2} u_{j}] - \operatorname{tr}[P_{\varepsilon} \nabla^{2} u_{j}]$$
$$= \Delta u_{j} - \sum_{i=1,2} P_{\varepsilon} (\nabla^{2} u_{j}) \tau_{i} \cdot \tau_{i} - P_{\varepsilon} (\nabla^{2} u) n_{\varepsilon} \cdot n_{\varepsilon}$$
$$= \Delta u_{j} - \sum_{i=1,2} (\tau_{i} \cdot \nabla) \nabla u_{j} \cdot \tau_{i}$$

for j = 1, 2, 3. In the last equality we used $P_{\varepsilon}^T = P_{\varepsilon}$, $P_{\varepsilon}\tau_i = \tau_i$, and $P_{\varepsilon}n_{\varepsilon} = 0$. For the second term on the last line we further see that

$$(\tau_i \cdot \nabla) \nabla u_j \cdot \tau_i = (\tau_i \cdot \nabla_{\Gamma_{\varepsilon}}) \nabla u_j \cdot \tau_i = (\tau_i \cdot \nabla_{\Gamma_{\varepsilon}}) \{ (\tau_i \cdot \nabla) u_j \} - \{ (\tau_i \cdot \nabla_{\Gamma_{\varepsilon}}) \tau_i \cdot \nabla \} u_j.$$

From these equalities and $(\Delta u \otimes n_{\varepsilon})n_{\varepsilon} = (n_{\varepsilon} \cdot n_{\varepsilon})\Delta u = \Delta u$ it follows that

$$\{(n_{\varepsilon} \cdot \nabla) \nabla u\}^T n_{\varepsilon} - (\Delta u \otimes n_{\varepsilon}) n_{\varepsilon} \\ = -\sum_{i=1,2} [(\tau_i \cdot \nabla_{\Gamma_{\varepsilon}}) \{(\tau_i \cdot \nabla) u\} - \{(\tau_i \cdot \nabla_{\Gamma_{\varepsilon}}) \tau_i \cdot \nabla\} u]. \quad (5.5.49)$$

By (5.5.38), (5.5.46), and (5.C.2) we have

$$\begin{aligned} (\tau_i \cdot \nabla_{\Gamma_{\varepsilon}})\{(\tau_i \cdot \nabla)u\} &= (\tau_i \cdot \nabla_{\Gamma_{\varepsilon}})\left\{\overline{\nabla}_i^{\varepsilon} u + (W_{\varepsilon} u \cdot \tau_i)n_{\varepsilon}\right\} \\ &= \overline{\nabla}_i^{\varepsilon} \overline{\nabla}_i^{\varepsilon} u - (W_{\varepsilon} u \cdot \tau_i)W_{\varepsilon}\tau_i + \left\{W_{\varepsilon} \overline{\nabla}_i^{\varepsilon} u \cdot \tau_i + \tau_i \cdot \nabla_{\Gamma_{\varepsilon}}(W_{\varepsilon} u \cdot \tau_i)\right\} n_{\varepsilon} \end{aligned}$$

and

$$\{ (\tau_i \cdot \nabla_{\Gamma_{\varepsilon}}) \tau_i \cdot \nabla \} u = \left[\left\{ \overline{\nabla}_i^{\varepsilon} \tau_i + (W_{\varepsilon} \tau_i \cdot \tau_i) n_{\varepsilon} \right\} \cdot \nabla \right] u \\ = \overline{\nabla}_{\overline{\nabla}_i^{\varepsilon} \tau_i}^{\varepsilon} u - (W_{\varepsilon} \tau_i \cdot \tau_i) (W_{\varepsilon} u + \tilde{\gamma}_{\varepsilon} u) + \left(W_{\varepsilon} u \cdot \overline{\nabla}_i^{\varepsilon} \tau_i + \xi_{\varepsilon} W_{\varepsilon} \tau_i \cdot \tau_i \right) n_{\varepsilon}.$$

We substitute these expressions for (5.5.49) and use

$$\sum_{i=1,2} (W_{\varepsilon} u \cdot \tau_i) W_{\varepsilon} \tau_i = \sum_{i=1,2} W_{\varepsilon} (\tau_i \otimes \tau_i) W_{\varepsilon} u = W_{\varepsilon} P_{\varepsilon} W_{\varepsilon} u = W_{\varepsilon}^2 u$$

by $P_{\varepsilon} = \sum_{i=1,2} \tau_i \otimes \tau_i$ and $P_{\varepsilon} W_{\varepsilon} = W_{\varepsilon}$,

$$\sum_{i=1,2} \left\{ \tau_i \cdot \nabla_{\Gamma_{\varepsilon}} (W_{\varepsilon} u \cdot \tau_i) - W_{\varepsilon} u \cdot \overline{\nabla}_i^{\varepsilon} \tau_i \right\} = \sum_{i=1,2} \overline{\nabla}_i^{\varepsilon} (W_{\varepsilon} u) \cdot \tau_i = \operatorname{div}_{\Gamma_{\varepsilon}} (W_{\varepsilon} u)$$

by (5.C.5) and (5.C.9), and the formulas (5.C.8) and (5.C.10) to deduce that

$$\{(n_{\varepsilon} \cdot \nabla)\nabla u\}^{T} n_{\varepsilon} - (\Delta u \otimes n_{\varepsilon}) n_{\varepsilon} = -\sum_{i=1,2} \left(\overline{\nabla}_{i}^{\varepsilon} \overline{\nabla}_{i}^{\varepsilon} u - \overline{\nabla}_{\overline{\nabla}_{i}^{\varepsilon} \tau_{i}}^{\varepsilon} u \right) + W_{\varepsilon}^{2} u - H_{\varepsilon} W_{\varepsilon} u - \tilde{\gamma}_{\varepsilon} H_{\varepsilon} u - \{\nabla_{\Gamma_{\varepsilon}} u : W_{\varepsilon} + \operatorname{div}_{\Gamma_{\varepsilon}} (W_{\varepsilon} u) - \xi_{\varepsilon} H_{\varepsilon} \} n_{\varepsilon}.$$
 (5.5.50)

Hence we take the inner product of (5.5.38) and (5.5.50) to obtain

$$\eta_{3} = \sum_{i=1,2} \left(\overline{\nabla}_{i}^{\varepsilon} \overline{\nabla}_{i}^{\varepsilon} u - \overline{\nabla}_{\overline{\nabla}_{i}^{\varepsilon} \tau_{i}}^{\varepsilon} u \right) \cdot \left(W_{\varepsilon} u + \tilde{\gamma}_{\varepsilon} u \right) - \left(W_{\varepsilon}^{2} u - H_{\varepsilon} W_{\varepsilon} u - \tilde{\gamma}_{\varepsilon} H_{\varepsilon} u \right) \cdot \left(W_{\varepsilon} u + \tilde{\gamma}_{\varepsilon} u \right) \\ - \xi_{\varepsilon} \{ \nabla_{\Gamma_{\varepsilon}} u : W_{\varepsilon} + \operatorname{div}_{\Gamma_{\varepsilon}} (W_{\varepsilon} u) - \xi_{\varepsilon} H_{\varepsilon} \}.$$
(5.5.51)

Now we observe by (5.2.39) and direct calculations that

$$\nabla_{\Gamma_{\varepsilon}} u : (\nabla_{\Gamma_{\varepsilon}} u) P_{\varepsilon} = P_{\varepsilon}(\nabla u) : P_{\varepsilon}(\nabla u) P_{\varepsilon} = \nabla u : P_{\varepsilon}^T P_{\varepsilon}(\nabla u) P_{\varepsilon}.$$

Since $P_{\varepsilon}^T = P_{\varepsilon}^2 = P_{\varepsilon}$, the above equality implies that

$$\nabla_{\Gamma_{\varepsilon}} u : (\nabla_{\Gamma_{\varepsilon}} u) P_{\varepsilon} = \nabla u : P_{\varepsilon}(\nabla u) P_{\varepsilon}.$$
(5.5.52)

By the same calculations and (5.2.39) we have

$$\nabla_{\Gamma_{\varepsilon}}(W_{\varepsilon}u): (\nabla_{\Gamma_{\varepsilon}}u)P_{\varepsilon} = \{\nabla_{\Gamma_{\varepsilon}}W_{\varepsilon} \cdot u + (\nabla_{\Gamma_{\varepsilon}}u)W_{\varepsilon}\}: (\nabla_{\Gamma_{\varepsilon}}u)P_{\varepsilon} \\ = \{\nabla_{\Gamma_{\varepsilon}}W_{\varepsilon} \cdot u + (\nabla u)W_{\varepsilon}\}: P_{\varepsilon}(\nabla u)P_{\varepsilon},$$
(5.5.53)

where the matrix $\nabla_{\Gamma_{\varepsilon}} W_{\varepsilon} \cdot u$ is given by (5.5.41), and

$$W_{\varepsilon}(\nabla_{\Gamma_{\varepsilon}}u): (\nabla_{\Gamma_{\varepsilon}}u)P_{\varepsilon} = W_{\varepsilon}(\nabla u): (\nabla u)P_{\varepsilon}, \quad \nabla_{\Gamma_{\varepsilon}}u: W_{\varepsilon} = \nabla u: W_{\varepsilon}.$$
(5.5.54)

Also, it is easy to see that

$$W_{\varepsilon}u \cdot \nabla_{\Gamma_{\varepsilon}}\xi_{\varepsilon} = \operatorname{div}_{\Gamma_{\varepsilon}}(\xi_{\varepsilon}W_{\varepsilon}u) - \xi_{\varepsilon}\operatorname{div}_{\Gamma_{\varepsilon}}(W_{\varepsilon}u),$$

$$\operatorname{div}_{\Gamma_{\varepsilon}}(W_{\varepsilon}u) = u \cdot \operatorname{div}_{\Gamma_{\varepsilon}}W_{\varepsilon} + \nabla_{\Gamma_{\varepsilon}}u : W_{\varepsilon} = u \cdot \operatorname{div}_{\Gamma_{\varepsilon}}W_{\varepsilon} + \nabla u : W_{\varepsilon}.$$

(5.5.55)

Hence we deduce from (5.5.42), (5.5.48), (5.5.51)–(5.5.55), and $W_{\varepsilon}^{T} = W_{\varepsilon}$ that

$$\begin{split} \int_{\Gamma_{\varepsilon}} \nabla u : \{ (n_{\varepsilon} \cdot \nabla) \nabla u - n_{\varepsilon} \otimes \Delta u \} d\mathcal{H}^2 \\ &= \sum_{i=1,2} \int_{\Gamma_{\varepsilon}} \left(\overline{\nabla}_i^{\varepsilon} \overline{\nabla}_i^{\varepsilon} u - \overline{\nabla}_{\overline{\nabla}_i^{\varepsilon} \tau_i}^{\varepsilon} u \right) \cdot (W_{\varepsilon} u + \tilde{\gamma}_{\varepsilon} u) d\mathcal{H}^2 \\ &+ \int_{\Gamma_{\varepsilon}} \left(\frac{1}{2} \varphi_1 + \sum_{k=2}^4 \varphi_k \right) d\mathcal{H}^2 + \int_{\Gamma_{\varepsilon}} \operatorname{div}_{\Gamma_{\varepsilon}} (\xi_{\varepsilon} W_{\varepsilon} u) d\mathcal{H}^2, \end{split}$$

where $\varphi_1, \ldots, \varphi_4$ are given by (5.5.40). The last integral on the right-hand side vanishes by the Stokes theorem since $\xi_{\varepsilon} W_{\varepsilon} u$ is tangential on Γ_{ε} . Moreover, applying (5.C.15) to the first term and then use (5.C.12), (5.5.52), and (5.5.53) we observe that

$$\sum_{i=1,2} \int_{\Gamma_{\varepsilon}} \left(\overline{\nabla}_{i}^{\varepsilon} \overline{\nabla}_{i}^{\varepsilon} u - \overline{\nabla}_{\overline{\nabla}_{i}^{\varepsilon} \tau_{i}}^{\varepsilon} u \right) \cdot \left(W_{\varepsilon} u + \tilde{\gamma}_{\varepsilon} u \right) d\mathcal{H}^{2} = \frac{1}{2} \int_{\Gamma_{\varepsilon}} \varphi_{1} d\mathcal{H}^{2}.$$

Hence the equality (5.5.39) follows.

The second step is to show that

$$\left| \int_{\Gamma_{\varepsilon}} \varphi_k \, d\mathcal{H}^2 \right| \le c \left(\|u\|_{H^1(\Omega_{\varepsilon})}^2 + \|u\|_{H^1(\Omega_{\varepsilon})} \|\nabla^2 u\|_{L^2(\Omega_{\varepsilon})} \right), \quad k = 1, 2, \tag{5.5.56}$$
$$\left| \int_{\Gamma_{\varepsilon}} \varphi_k \, d\mathcal{H}^2 \right| \le c \|u\|_{H^1(\Omega_{\varepsilon})}^2, \quad k = 3, 4 \tag{5.5.57}$$

with a constant c > 0 independent of ε . The estimate (5.5.57) for k = 4 is an easy consequence of (5.1.6), (5.3.8), and the uniform boundedness in ε of W_{ε} and H_{ε} on Γ_{ε} :

$$\left|\int_{\Gamma_{\varepsilon}}\varphi_4 \, d\mathcal{H}^2\right| \le c\varepsilon \|u\|_{L^2(\Gamma_{\varepsilon})}^2 \le c\varepsilon(\varepsilon^{-1}\|u\|_{L^2(\Omega_{\varepsilon})}^2 + \varepsilon \|\partial_n u\|_{L^2(\Omega_{\varepsilon})}^2) \le c\|u\|_{H^1(\Omega_{\varepsilon})}^2.$$

Let us prove the estimate (5.5.56) for k = 1. We proceed as in the proof of Lemma 5.4.1. In what follows, we use the notations (5.2.46) and (5.2.47) and sometimes suppress the arguments y and r. For $y \in \Gamma$, $r \in [\varepsilon g_0(y), \varepsilon g_1(y)]$, and j, k, l = 1, 2, 3 we set

$$F(y,r) := \frac{1}{\varepsilon g(y)} \{ (r - \varepsilon g_0(y)) W_{\varepsilon,1}^{\sharp}(y) - (\varepsilon g_1(y) - r) W_{\varepsilon,0}^{\sharp}(y) \},$$

$$G_{jk}^l(y,r) := \frac{1}{\varepsilon g(y)} \{ (r - \varepsilon g_0(y)) (\underline{D}_j^{\varepsilon} [W_{\varepsilon}]_{kl})_1^{\sharp}(y) - (\varepsilon g_1(y) - r) (\underline{D}_j^{\varepsilon} [W_{\varepsilon}]_{kl})_0^{\sharp}(y) \},$$

$$\tilde{\gamma}(y,r) := \frac{1}{\varepsilon g(y)} \{ (r - \varepsilon g_0(y)) \tilde{\gamma}_{\varepsilon}^1 - (\varepsilon g_1(y) - r) \tilde{\gamma}_{\varepsilon}^0 \},$$

where $\tilde{\gamma}^i_{\varepsilon} := \gamma^i_{\varepsilon} / \nu, \, i = 0, 1$. Then we have

$$[\nabla_{\Gamma_{\varepsilon}} W_{\varepsilon} \cdot u + (\nabla u) W_{\varepsilon} + \tilde{\gamma}_{\varepsilon} \nabla u]_{i}^{\sharp}(y) = (-1)^{i+1} [G \cdot u^{\sharp} + (\nabla u)^{\sharp} F + \tilde{\gamma} (\nabla u)^{\sharp}](y, \varepsilon g_{i}(y)), \quad y \in \Gamma, \ i = 0, 1, \quad (5.5.58)$$

where $G \cdot u^{\sharp}$ denotes a 3 × 3 matrix whose (j, k)-entry is given by

$$[G \cdot u^{\sharp}]_{jk} := \sum_{l=1}^{3} G_{jk}^{l} u_{l}^{\sharp}, \quad j, k = 1, 2, 3.$$

Moreover, by (5.1.6), (5.2.45) for W_{ε} and $\underline{D}_{j}^{\varepsilon}W_{\varepsilon}$ with j = 1, 2, 3,

$$|r - \varepsilon g_i(y)| \le \varepsilon g(y) \le c\varepsilon, \quad y \in \Gamma, \ r \in [\varepsilon g_0(y), \varepsilon g_1(y)], \ i = 0, 1,$$
(5.5.59)

and the uniform boundedness in ε of W_{ε} and $\underline{D}_{j}^{\varepsilon}W_{\varepsilon}$ on Γ_{ε} we have

$$\left|\eta(y,r)\right| + \left|\frac{\partial\eta}{\partial r}(y,r)\right| \le c, \quad \eta = F, G_{jk}^l, \tilde{\gamma}$$
(5.5.60)

for all $y \in \Gamma$ and $r \in [\varepsilon g_0(y), \varepsilon g_1(y)]$ with a constant c > 0 independent of ε . We also define matrix-valued functions R and S by

$$R(y,r) := \frac{1}{\varepsilon g(y)} \left\{ \left(r - \varepsilon g_0(y) \right) P_{\varepsilon,1}^{\sharp}(y) + \left(\varepsilon g_1(y) - r \right) P_{\varepsilon,0}^{\sharp}(y) \right\}$$

for $y \in \Gamma$ and $r \in [\varepsilon g_0(y), \varepsilon g_1(y)]$, and

$$S_{i}(y) := \sqrt{1 + \varepsilon^{2} |\tau_{\varepsilon}^{i}(y)|^{2}} P_{\varepsilon,i}^{\sharp}(y), \quad i = 0, 1,$$

$$S(y,r) := \frac{1}{\varepsilon g(y)} \{ (r - \varepsilon g_{0}(y)) S_{1}(y) + (\varepsilon g_{1}(y) - r) S_{0}(y) \}.$$

Then we easily observe that

$$\sqrt{1+\varepsilon^2 |\tau^i_{\varepsilon}(y)|^2} \left[P_{\varepsilon}(\nabla u) P_{\varepsilon} \right]^{\sharp}_i(y) = \left[R(\nabla u)^{\sharp} S \right](y, \varepsilon g_i(y)), \quad y \in \Gamma, \ i = 0, 1.$$
(5.5.61)

Moreover, from (5.2.44) for P_{ε} , (5.4.9), and (5.5.59) we deduce that

$$\left|\eta(y,r)\right| + \left|\frac{\partial\eta}{\partial r}(y,r)\right| \le c, \quad \eta = R, S$$
(5.5.62)

for all $y \in \Gamma$ and $r \in [\varepsilon g_0(y), \varepsilon g_1(y)]$ with a constant c > 0 independent of ε . Now let us define a function $\Phi_1 = \Phi_1(y, r)$ for $y \in \Gamma$ and $r \in [\varepsilon g_0(y), \varepsilon g_1(y)]$ by

$$\Phi_1(y,r) := -2[\{G \cdot u^{\sharp} + (\nabla u)^{\sharp}F + \tilde{\gamma}(\nabla u)^{\sharp}\} : R(\nabla u)S](y,r)J(y,r)]$$

Then by the change of variables formula (5.2.57) and the equalities (5.5.58) and (5.5.61) we observe that

$$\begin{split} \int_{\Gamma_{\varepsilon}} \varphi_1(x) \, d\mathcal{H}^2(x) &= \sum_{i=0,1} \int_{\Gamma_{\varepsilon}^i} \varphi_1(x) \, d\mathcal{H}^2(x) = \int_{\Gamma} \{ \Phi_1(y, \varepsilon g_1(y)) - \Phi_1(y, \varepsilon g_0(y)) \} \, d\mathcal{H}^2(y) \\ &= \int_{\Gamma} \int_{\varepsilon g_0(y)}^{\varepsilon g_1(y)} \frac{\partial \Phi_1}{\partial r}(y, r) \, dr \, d\mathcal{H}^2(y). \end{split}$$

Furthermore, the inequalities (5.2.49), (5.5.60), and (5.5.62) imply that

$$\left|\frac{\partial \Phi_1}{\partial r}\right| \le c\{|u^{\sharp}|^2 + |(\nabla u)^{\sharp}|^2 + (|u^{\sharp}| + |(\nabla u)^{\sharp}|)|(\nabla^2 u)^{\sharp}|\}$$

with some constant c > 0 independent of ε (here we also used Young's inequality). From the above relations, (5.2.52), and Hölder's inequality it follows that

$$\left| \int_{\Gamma_{\varepsilon}} \varphi_1 \, d\mathcal{H}^2 \right| \le c \int_{\Gamma} \int_{\varepsilon g_0}^{\varepsilon g_1} \{ |u^{\sharp}|^2 + |(\nabla u)^{\sharp}|^2 + (|u^{\sharp}| + |(\nabla u)^{\sharp}|) |(\nabla^2 u)^{\sharp}| \} \, dr \, d\mathcal{H}^2$$
$$\le c \left(||u||_{H^1(\Omega_{\varepsilon})}^2 + ||u||_{H^1(\Omega_{\varepsilon})} ||\nabla^2 u||_{L^2(\Omega_{\varepsilon})} \right).$$

(Note that $||u||_{L^2(\Omega_{\varepsilon})} \leq ||u||_{H^1(\Omega_{\varepsilon})}$.) Thus, the inequality (5.5.56) for k = 1 holds. By the same arguments we can show (5.5.56) for k = 2 and (5.5.57) for k = 3.

Finally, from (5.5.39), (5.5.56), and (5.5.57) we deduce that

$$\left|\int_{\Gamma_{\varepsilon}} \nabla u : \left\{ (n_{\varepsilon} \cdot \nabla) \nabla u - n_{\varepsilon} \otimes \Delta u \right\} d\mathcal{H}^2 \right| \le c \left(\|u\|_{H^1(\Omega_{\varepsilon})}^2 + \|u\|_{H^1(\Omega_{\varepsilon})} \|\nabla^2 u\|_{L^2(\Omega_{\varepsilon})} \right).$$

We apply this inequality to (5.5.37) and then use Young's inequality to obtain

$$\begin{aligned} \|\nabla^{2}u\|_{L^{2}(\Omega_{\varepsilon})}^{2} &\leq \|\Delta u\|_{L^{2}(\Omega_{\varepsilon})}^{2} + c\left(\|u\|_{H^{1}(\Omega_{\varepsilon})}^{2} + \|u\|_{H^{1}(\Omega_{\varepsilon})}\|\nabla^{2}u\|_{L^{2}(\Omega_{\varepsilon})}\right) \\ &\leq \|\Delta u\|_{L^{2}(\Omega_{\varepsilon})}^{2} + \frac{1}{2}\|\nabla^{2}u\|_{L^{2}(\Omega_{\varepsilon})}^{2} + c\|u\|_{H^{1}(\Omega_{\varepsilon})}^{2}, \end{aligned}$$

which yields (5.5.36). Hence the inequality (5.5.31) is valid.

5.6 Average operators in the thin direction

5.6.1 Definition and basic inequalities of the average operators

In this section we investigate average operators in the thin direction and establish several inequalities related to them, which are useful in the analysis of the Navier–Stokes equations in the curved thin domain Ω_{ε} . Throughout this section we assume $\varepsilon \in (0, 1)$.

Definition 5.6.1. We define the average operator M in the normal direction of Γ as

$$M\varphi(y) := \frac{1}{\varepsilon g(y)} \int_{\varepsilon g_0(y)}^{\varepsilon g_1(y)} \varphi(y + rn(y)) \, dr, \quad y \in \Gamma$$
(5.6.1)

for a function φ on Ω_{ε} . The operator M is also applied to a vector field $u: \Omega_{\varepsilon} \to \mathbb{R}^3$ and we define the averaged tangential component $M_{\tau}u$ of u by

$$M_{\tau}u(y) := P(y)Mu(y) = \frac{1}{\varepsilon g(y)} \int_{\varepsilon g_0(y)}^{\varepsilon g_1(y)} P(y)u(y + rn(y)) \, dr, \quad y \in \Gamma.$$
(5.6.2)

For the sake of simplicity, we denote the tangential and normal components (with respect to the surface Γ) of a vector field $u: \Omega_{\varepsilon} \to \mathbb{R}^3$ by

$$u_{\tau}(x) := \overline{P}(x)u(x), \quad u_n(x) := \{u(x) \cdot \overline{n}(x)\}\overline{n}(x), \quad x \in \Omega_{\varepsilon},$$
(5.6.3)

where \overline{P} and \overline{n} are the constant extensions of P and n in the normal direction of Γ , so that $u = u_{\tau} + u_n$ and $u_{\tau} \cdot u_n = 0$ (note that u_n is a vector field). Also, we use the notations (5.2.46) and (5.2.47) and sometimes suppress the arguments of functions. For example, we write

$$M\varphi = \frac{1}{\varepsilon g} \int_{\varepsilon g_0}^{\varepsilon g_1} \varphi^{\sharp} dr, \quad M_{\tau} u = \frac{1}{\varepsilon g} \int_{\varepsilon g_0}^{\varepsilon g_1} u_{\tau}^{\sharp} dr.$$

Let us derive basic inequalities for the average operators M and M_{τ} . First note that for a vector field u on Ω_{ε} we have $M_{\tau}u = Mu_{\tau}$ by (5.6.2) and (5.6.3). Hence the following inequalities for M are also valid for M_{τ} .

Lemma 5.6.2. Let $p \in [1, \infty)$. There exists c > 0 independent of ε such that

$$\|M\varphi\|_{L^p(\Gamma)} \le c\varepsilon^{-1/p} \|\varphi\|_{L^p(\Omega_\varepsilon)},\tag{5.6.4}$$

$$\left\|\overline{M\varphi}\right\|_{L^{p}(\Omega_{\varepsilon})} \le c \|\varphi\|_{L^{p}(\Omega_{\varepsilon})} \tag{5.6.5}$$

for all $\varphi \in L^p(\Omega_{\varepsilon})$. Here $\overline{M\varphi} := (M\varphi) \circ \pi$ is the constant extension of $M\varphi$.

Proof. By Hölder's inequality and (5.2.30),

$$|M\varphi(y)|^p = \left|\frac{1}{\varepsilon g(y)} \int_{\varepsilon g_0(y)}^{\varepsilon g_1(y)} \varphi^{\sharp}(y,r) \, dr\right|^p \le c\varepsilon^{-1} \int_{\varepsilon g_0(y)}^{\varepsilon g_1(y)} |\varphi^{\sharp}(y,r)|^p \, dr$$

for all $y \in \Gamma$. Integrating both sides of the above inequality over Γ and using (5.2.52) we obtain (5.6.4). The inequality (5.6.5) follows from (5.2.53) and (5.6.4).

Lemma 5.6.3. Let $p \in [1, \infty)$. There exists c > 0 independent of ε such that

$$\left\|\varphi - \overline{M\varphi}\right\|_{L^{p}(\Omega_{\varepsilon})} \le c\varepsilon \left\|\partial_{n}\varphi\right\|_{L^{p}(\Omega_{\varepsilon})},\tag{5.6.6}$$

$$\left\|\varphi - \overline{M\varphi}\right\|_{L^{p}(\Gamma^{i}_{\varepsilon})} \le c\varepsilon^{1-1/p} \|\partial_{n}\varphi\|_{L^{p}(\Omega_{\varepsilon})}, \quad i = 0, 1$$
(5.6.7)

for all $\varphi \in W^{1,p}(\Omega_{\varepsilon})$, where $\partial_n \varphi$ is the normal derivative of φ given by (5.3.5).

Proof. For $y \in \Gamma$ and $r \in (\varepsilon g_0(y), \varepsilon g_1(y))$ we have

$$\varphi^{\sharp}(y,r) - M\varphi(y) = \frac{1}{\varepsilon g(y)} \int_{\varepsilon g_0(y)}^{\varepsilon g_1(y)} \{\varphi^{\sharp}(y,r) - \varphi^{\sharp}(y,r_1)\} dr_1.$$
(5.6.8)

Since $\partial \varphi^{\sharp} / \partial r = (\partial_n \varphi)^{\sharp}$ by (5.2.46) and (5.3.5),

$$\left|\varphi^{\sharp}(y,r) - \varphi^{\sharp}(y,r_1)\right| = \left|\int_{r_1}^r \frac{\partial}{\partial r_2} \left(\varphi^{\sharp}(y,r_2)\right) dr_2\right| \le \int_{\varepsilon g_0(y)}^{\varepsilon g_1(y)} \left|\left(\partial_n \varphi\right)^{\sharp}(y,r_2)\right| dr_2.$$

Noting that the right-hand side is independent of r_1 , we apply the above inequality to the right-hand side of (5.6.8) to get

$$|\varphi^{\sharp}(y,r) - M\varphi(y)| \le \int_{\varepsilon g_0(y)}^{\varepsilon g_1(y)} |(\partial_n \varphi)^{\sharp}(y,r_2)| \, dr_2 \tag{5.6.9}$$

for all $y \in \Gamma$ and $r \in (\varepsilon g_0(y), \varepsilon g_1(y))$. We use Hölder's inequality to (5.6.9) to get

$$|\varphi^{\sharp}(y,r) - M\varphi(y)| \le c\varepsilon^{1-1/p} \left(\int_{\varepsilon g_0(y)}^{\varepsilon g_1(y)} |(\partial_n \varphi)^{\sharp}(y,r_2)|^p \, dr_2 \right)^{1/p}$$

for all $y \in \Gamma$ and $r \in (\varepsilon g_0(y), \varepsilon g_1(y))$. Note that the right-hand side is independent of r. The above inequality and (5.2.52) imply that

$$\begin{aligned} \left\|\varphi - \overline{M\varphi}\right\|_{L^{p}(\Omega_{\varepsilon})}^{p} &\leq c \int_{\Gamma} \int_{\varepsilon g_{0}(y)}^{\varepsilon g_{1}(y)} |\varphi^{\sharp}(y,r) - M\varphi(y)|^{p} dr d\mathcal{H}^{2}(y) \\ &\leq c \int_{\Gamma} \varepsilon g(y) \left(c\varepsilon^{p-1} \int_{\varepsilon g_{0}(y)}^{\varepsilon g_{1}(y)} |(\partial_{n}\varphi)^{\sharp}(y,r_{2})|^{p} dr_{2} \right) d\mathcal{H}^{2}(y) \\ &\leq c\varepsilon^{p} \|\partial_{n}\varphi\|_{L^{p}(\Omega_{\varepsilon})}^{p}. \end{aligned}$$

Hence (5.6.6) holds. We also have (5.6.7) by applying (5.3.8) to $\varphi - \overline{M}\overline{\varphi}$ and using (5.3.6) and (5.6.6).

Lemma 5.6.4. Let $p \in [1, \infty)$. There exists c > 0 independent of ε such that

$$\|Mu \cdot n\|_{L^{p}(\Gamma)} \le c\varepsilon^{1-1/p} \|u\|_{W^{1,p}(\Omega_{\varepsilon})}$$
(5.6.10)

for all $u \in W^{1,p}(\Omega_{\varepsilon})^3$ satisfying $u \cdot n_{\varepsilon} = 0$ on Γ_{ε}^0 or on Γ_{ε}^1 .

Proof. Applying (5.6.4) to $Mu \cdot n = M(u \cdot \bar{n})$ and using (5.3.23) we obtain (5.6.10).

Unlike the case of a flat thin domain, the constant extension of the average operator on $L^2(\Omega_{\varepsilon})$ is not symmetric because the Jacobian J(y, r) appears in the change of variables formula (5.2.51). However, its skew-symmetric part is small of order ε .

Lemma 5.6.5. There exists a constant c > 0 independent of ε such that

$$\left| \left(\overline{M\varphi}, \xi \right)_{L^2(\Omega_{\varepsilon})} - \left(\varphi, \overline{M\xi} \right)_{L^2(\Omega_{\varepsilon})} \right| \le c \varepsilon \|\varphi\|_{L^2(\Omega_{\varepsilon})} \|\xi\|_{L^2(\Omega_{\varepsilon})}$$
(5.6.11)

for all $\varphi, \xi \in L^2(\Omega_{\varepsilon})$.

Proof. By (5.2.51) and (5.6.1) we have

$$\left(\overline{M\varphi},\xi\right)_{L^2(\Omega_{\varepsilon})} = \int_{\Gamma} M\varphi \left(\int_{\varepsilon g_0}^{\varepsilon g_1} \xi^{\sharp} J \, dr\right) d\mathcal{H}^2$$
$$= \varepsilon (M\varphi,gM\xi)_{L^2(\Gamma)} + \int_{\Gamma} M\varphi \left(\int_{\varepsilon g_0}^{\varepsilon g_1} \xi^{\sharp} (J-1) \, dr\right) d\mathcal{H}^2$$

and

$$\left(\varphi, \overline{M\xi}\right)_{L^2(\Omega_{\varepsilon})} = \varepsilon (gM\varphi, M\xi)_{L^2(\Gamma)} + \int_{\Gamma} \left(\int_{\varepsilon g_0}^{\varepsilon g_1} \varphi^{\sharp} (J-1) \, dr\right) M\xi \, d\mathcal{H}^2.$$

Since $(M\varphi, gM\xi)_{L^2(\Gamma)} = (gM\varphi, M\xi)_{L^2(\Gamma)}$, we see by (5.2.50) that

$$\left| \left(\overline{M\varphi}, \xi \right)_{L^{2}(\Omega_{\varepsilon})} - \left(\varphi, \overline{M\xi} \right)_{L^{2}(\Omega_{\varepsilon})} \right|$$

$$\leq c\varepsilon \left\{ \int_{\Gamma} |M\varphi| \left(\int_{\varepsilon g_{0}}^{\varepsilon g_{1}} |\xi^{\sharp}| dr \right) d\mathcal{H}^{2} + \int_{\Gamma} \left(\int_{\varepsilon g_{0}}^{\varepsilon g_{1}} |\varphi^{\sharp}| dr \right) |M\xi| d\mathcal{H}^{2} \right\}.$$
(5.6.12)

Moreover, applying Hölder's inequality twice and using (5.2.52) and (5.6.4) we get

$$\int_{\Gamma} |M\varphi| \left(\int_{\varepsilon g_0}^{\varepsilon g_1} |\xi^{\sharp}| \, dr \right) d\mathcal{H}^2 \leq ||M\varphi||_{L^2(\Gamma)} \left\{ \int_{\Gamma} \varepsilon g \left(\int_{\varepsilon g_0}^{\varepsilon g_1} |\xi^{\sharp}|^2 \, dr \right) d\mathcal{H}^2 \right\}^{1/2} \\ \leq c ||\varphi||_{L^2(\Omega_{\varepsilon})} ||\xi||_{L^2(\Omega_{\varepsilon})}$$

and a similar inequality for the second term on the right-hand side of (5.6.12). Hence (5.6.11) follows. $\hfill \Box$

Now let us consider the time derivative of the average operator.

Lemma 5.6.6. Let $\varphi \in H^1(0,T; L^2(\Omega_{\varepsilon})), T > 0$. Then

$$\partial_t M \varphi = M(\partial_t \varphi) \in L^2(0,T;L^2(\Gamma))$$

and there exists a constant c > 0 independent of ε and φ such that

$$\|\partial_t M\varphi\|_{L^2(0,T;L^2(\Gamma))} \le c\varepsilon^{-1/2} \|\partial_t \varphi\|_{L^2(0,T;L^2(\Omega_\varepsilon))}.$$
(5.6.13)

Proof. First note that $M(\partial_t \varphi) \in L^2(0, T; L^2(\Gamma))$ by $\partial_t \varphi \in L^2(0, T; L^2(\Omega_{\varepsilon}))$ and (5.6.4). The relation $\partial_t M \varphi = M(\partial_t \varphi)$ is formally trivial since g_0, g_1 , and the surface quantities on Γ are independent of time. To prove it rigorously we show that

$$\int_0^T \partial_t \xi(t) M\varphi(t) \, dt = -\int_0^T \xi(t) [M(\partial_t \varphi)](t) \, dt \quad \text{in} \quad L^2(\Gamma)$$

for all $\xi \in C_c^{\infty}(0,T)$. Since $L^2(\Gamma)$ is a Hilbert space, this is equivalent to

$$\int_0^T \partial_t \xi(t) (M\varphi(t), \eta)_{L^2(\Gamma)} dt = -\int_0^T \xi(t) ([M(\partial_t \varphi)](t), \eta)_{L^2(\Gamma)} dt$$
(5.6.14)

for all $\xi \in C_c^{\infty}(0,T)$ and $\eta \in L^2(\Gamma)$. We define a function $\tilde{\eta}$ on Ω_{ε} by

$$\tilde{\eta}(x) := \frac{\eta(\pi(x))}{\varepsilon g(\pi(x)) J(\pi(x), d(x))}, \quad x \in \Omega_{\varepsilon}$$

Then by (5.2.30), (5.2.49), and (5.2.53) we see that $\tilde{\eta} \in L^2(\Omega_{\varepsilon})$. Also, by the change of variables formula (5.2.51) and the definition (5.6.1) of M we have

$$(\varphi(t),\tilde{\eta})_{L^2(\Omega_{\varepsilon})} = \int_{\Gamma} \left(\frac{1}{\varepsilon g} \int_{\varepsilon g_0}^{\varepsilon g_1} \varphi^{\sharp}(t) \, dr \right) \eta \, d\mathcal{H}^2 = (M\varphi(t),\eta)_{L^2(\Gamma)}$$
(5.6.15)

for all $t \in (0, T)$. Hence

$$\int_0^T \partial_t \xi(t) (M\varphi(t), \eta)_{L^2(\Gamma)} dt = \int_0^T \partial_t \xi(t) (\varphi(t), \tilde{\eta})_{L^2(\Omega_\varepsilon)} dt.$$
(5.6.16)

Moreover, since $\varphi \in H^1(0,T; L^2(\Omega_{\varepsilon}))$ and $\tilde{\eta} \in L^2(\Omega_{\varepsilon})$ is independent of time,

$$\int_0^T \partial_t \xi(t)(\varphi(t),\tilde{\eta})_{L^2(\Omega_\varepsilon)} dt = -\int_0^T \xi(t)(\partial_t \varphi(t),\tilde{\eta})_{L^2(\Omega_\varepsilon)} dt.$$

Applying this equality to the right-hand side of (5.6.16) and using (5.6.15) with $\varphi(t)$ replaced by $\partial_t \varphi(t)$ we obtain (5.6.14). Hence the relation $\partial_t M \varphi = M(\partial_t \varphi)$ is valid and the inequality (5.6.13) follows from (5.6.4).

5.6.2 Tangential derivatives of the average operators

Let us give several formulas and inequalities for the tangential derivatives of the average operators.

Lemma 5.6.7. For $\varphi \in C^1(\Omega_{\varepsilon})$ we have

$$\nabla_{\Gamma} M \varphi = M(B \nabla \varphi) + M((\partial_n \varphi) \psi_{\varepsilon}) \quad on \quad \Gamma, \tag{5.6.17}$$

where the matrix-valued function B and the vector field ψ_{ε} are given by

$$B(x) := \left\{ I_3 - d(x)\overline{W}(x) \right\} \overline{P}(x),$$

$$\psi_{\varepsilon}(x) := \frac{1}{\overline{g}(x)} \left\{ \left(d(x) - \varepsilon \overline{g}_0(x) \right) \overline{\nabla_{\Gamma} g_1}(x) + \left(\varepsilon \overline{g}_1(x) - d(x) \right) \overline{\nabla_{\Gamma} g_0}(x) \right\}$$
(5.6.18)

for $x \in N$. Here the functions on the right-hand sides of (5.6.18) except for d(x) are the constant extensions of the functions on Γ .

Proof. The constant extension of $M\varphi$ is given by

$$\overline{M\varphi}(x) = \frac{1}{\varepsilon \overline{g}(x)} \int_{\varepsilon \overline{g}_0(x)}^{\varepsilon \overline{g}_1(x)} \varphi(\pi(x) + r\overline{n}(x)) \, dr, \quad x \in N.$$

We differentiate both sides of this equality with respect to $x \in N$ and set $x = y \in \Gamma$. Then by (5.2.4), (5.2.6), (5.2.11), and (5.2.16) with d(y) = 0 we get

$$\nabla_{\Gamma} M\varphi(y) = \frac{I(y)}{\varepsilon g(y)} + \frac{1}{\varepsilon g(y)} \int_{\varepsilon g_0(y)}^{\varepsilon g_1(y)} \{I_3 - rW(y)\} P(y) (\nabla\varphi)^{\sharp}(y,r) \, dr \tag{5.6.19}$$

for $y \in \Gamma$. Here and hereafter we use the notations (5.2.46) and (5.2.47) and set

$$I(y) := -\frac{\nabla_{\Gamma} g(y)}{g(y)} \int_{\varepsilon g_0(y)}^{\varepsilon g_1(y)} \varphi^{\sharp}(y, r) \, dr + \varepsilon \varphi_1^{\sharp}(y) \nabla_{\Gamma} g_1(y) - \varepsilon \varphi_0^{\sharp}(y) \nabla_{\Gamma} g_0(y) \, dr$$

To the right-hand side we apply

$$\varepsilon\varphi_1^{\sharp}(y)\nabla_{\Gamma}g_1(y) - \varepsilon\varphi_0^{\sharp}(y)\nabla_{\Gamma}g_0(y) = \left[(\varphi\psi_{\varepsilon})^{\sharp}(y,r) \right]_{r=\varepsilon g_0(y)}^{\varepsilon g_1(y)} = \int_{\varepsilon g_0(y)}^{\varepsilon g_1(y)} \frac{\partial}{\partial r} \left((\varphi\psi_{\varepsilon})^{\sharp}(y,r) \right) dr$$

and the equalities $\partial \varphi^{\sharp} / \partial r = (\partial_n \varphi)^{\sharp}$ and $\partial \psi_{\varepsilon}^{\sharp} / \partial r = \nabla_{\Gamma} g / g$ by

$$\psi_{\varepsilon}^{\sharp}(y,r) = \frac{1}{g(y)} \big\{ \big(r - \varepsilon g_0(y)\big) \nabla_{\Gamma} g_1(y) + \big(\varepsilon g_1(y) - r\big) \nabla_{\Gamma} g_0(y) \big\}.$$

Then we have

$$I(y) = \int_{\varepsilon g_0(y)}^{\varepsilon g_1(y)} \left((\partial_n \varphi) \psi_{\varepsilon} \right)^{\sharp} (y, r) \, dr = \varepsilon g(y) \left[M \left((\partial_n \varphi) \psi_{\varepsilon} \right) \right] (y)$$

for all $y \in \Gamma$. Applying this and $\{I_3 - rW(y)\}P(y) = B^{\sharp}(y,r)$ to the right-hand side of (5.6.19) we obtain (5.6.17).

Remark 5.6.8. By (5.3.39) and (5.6.18) there exists c > 0 independent of ε such that

$$|B| \le c, \quad |\psi_{\varepsilon}| \le c\varepsilon \quad \text{in} \quad \Omega_{\varepsilon}. \tag{5.6.20}$$

Also, from (5.2.13), (5.2.30), and $\nabla d = \bar{n}$ in N it follows that

$$|\nabla B| \le c, \quad |\nabla \psi_{\varepsilon}| \le c, \quad \left|\nabla \psi_{\varepsilon} - \frac{1}{\bar{g}}\bar{n} \otimes \overline{\nabla_{\Gamma}g}\right| \le c\varepsilon \quad \text{in} \quad \Omega_{\varepsilon}.$$
 (5.6.21)

Lemma 5.6.9. Let m = 1, 2 and $p \in [1, \infty)$. For $\varphi \in W^{m,p}(\Omega_{\varepsilon})$ we have

$$\|M\varphi\|_{W^{m,p}(\Gamma)} \le c\varepsilon^{-1/p} \|\varphi\|_{W^{m,p}(\Omega_{\varepsilon})},\tag{5.6.22}$$

$$\left\|\overline{M\varphi}\right\|_{W^{m,p}(\Omega_{\varepsilon})} \le c \|\varphi\|_{W^{m,p}(\Omega_{\varepsilon})} \tag{5.6.23}$$

with some constant c > 0 independent of ε and φ .

Proof. Let $\varphi \in W^{1,p}(\Omega_{\varepsilon})$. From (5.6.4) and (5.6.17) it follows that

$$\begin{aligned} \|\nabla_{\Gamma} M\varphi\|_{L^{p}(\Gamma)} &\leq c \left(\|M(B\nabla\varphi)\|_{L^{p}(\Gamma)} + \|M((\partial_{n}\varphi)\psi_{\varepsilon})\|_{L^{p}(\Gamma)} \right) \\ &\leq c\varepsilon^{-1/p} \left(\|B\nabla\varphi\|_{L^{p}(\Omega_{\varepsilon})} + \|(\partial_{n}\varphi)\psi_{\varepsilon}\|_{L^{p}(\Omega_{\varepsilon})} \right). \end{aligned}$$

Here B and ψ_{ε} are bounded on Ω_{ε} uniformly in ε (see Remark 5.6.8). Hence

$$\|\nabla_{\Gamma} M\varphi\|_{L^{p}(\Gamma)} \leq c\varepsilon^{-1/p} \|\nabla\varphi\|_{L^{p}(\Omega_{\varepsilon})} \leq c\varepsilon^{-1/p} \|\varphi\|_{W^{1,p}(\Omega_{\varepsilon})}$$
(5.6.24)

Combining (5.6.24) with (5.6.4) we obtain (5.6.22) with m = 1. When $\varphi \in W^{2,p}(\Omega_{\varepsilon})$, we apply (5.6.24) with φ replaced by $B\nabla\varphi$ and $(\partial_n\varphi)\psi_{\varepsilon}$. Then by (5.6.20)–(5.6.21) we see that

$$\|\nabla_{\Gamma} M(B\nabla\varphi)\|_{L^{p}(\Gamma)} + \|\nabla_{\Gamma} M((\partial_{n}\varphi)\psi_{\varepsilon})\|_{L^{p}(\Gamma)} \le c\varepsilon^{-1/p}\|\varphi\|_{W^{2,p}(\Omega_{\varepsilon})}.$$

Therefore, applying ∇_{Γ} to (5.6.17) and using the above inequality we get

 $\|\nabla_{\Gamma}^2 M\varphi\|_{L^p(\Gamma)} \le c\varepsilon^{-1/p} \|\varphi\|_{W^{2,p}(\Omega_{\varepsilon})}$

and (5.6.22) with m = 2 follows from this inequality, (5.6.4), and (5.6.24). The inequality (5.6.23) is an immediate consequence of (5.2.55), (5.2.56), and (5.6.22).

Lemma 5.6.10. There exists a constant c > 0 independent of ε such that

$$\left\|\overline{P}\nabla\varphi - \overline{\nabla_{\Gamma}M\varphi}\right\|_{L^{p}(\Omega_{\varepsilon})} \le c\varepsilon \|\varphi\|_{W^{2,p}(\Omega_{\varepsilon})}$$
(5.6.25)

$$\left\|\overline{P}\nabla\varphi - \overline{\nabla_{\Gamma}M\varphi}\right\|_{L^{p}(\Gamma_{\varepsilon}^{i})} \le c\varepsilon^{1-1/p} \|\varphi\|_{W^{2,p}(\Omega_{\varepsilon})}, \quad i = 0, 1$$
(5.6.26)

for all $\varphi \in W^{2,p}(\Omega_{\varepsilon})$ with $p \in [1,\infty)$.

Proof. By (5.6.17) and (5.6.18) we have $\overline{P}\nabla\varphi - \overline{\nabla_{\Gamma}M\varphi} = u + \overline{v}$ in Ω_{ε} , where

$$u(x) := \overline{P}(x)\nabla\varphi(x) - \left[M\left(\overline{P}\nabla\varphi\right)\right](\pi(x)), \quad x \in \Omega_{\varepsilon},$$
$$v(y) := \left[M\left(d\overline{W}\nabla\varphi\right)\right](y) - \left[M\left((\partial_{n}\varphi)\psi_{\varepsilon}\right)\right](y), \quad y \in \Gamma.$$

We apply (5.6.6) to u and use (5.3.6) to get

$$\|u\|_{L^{p}(\Omega_{\varepsilon})} \leq c\varepsilon \left\|\partial_{n}\left(\overline{P}\nabla\varphi\right)\right\|_{L^{p}(\Omega_{\varepsilon})} \leq c\varepsilon \|\varphi\|_{W^{2,p}(\Omega_{\varepsilon})}.$$
(5.6.27)
Also, from (5.6.5) and

$$d\overline{W}\nabla\varphi\big| \le c\varepsilon |\nabla\varphi|, \quad |(\partial_n\varphi)\psi_\varepsilon| \le c\varepsilon |\nabla\varphi| \quad \text{in} \quad \Omega_\varepsilon$$

by (5.6.20) and $|d| \leq c\varepsilon$ in Ω_{ε} it follows that

$$\|\bar{v}\|_{L^{p}(\Omega_{\varepsilon})} \leq c \left(\left\| d\overline{W} \nabla \varphi \right\|_{L^{p}(\Omega_{\varepsilon})} + \| (\partial_{n} \varphi) \psi_{\varepsilon} \|_{L^{p}(\Omega_{\varepsilon})} \right) \leq c \varepsilon \| \nabla \varphi \|_{L^{p}(\Omega_{\varepsilon})}.$$
(5.6.28)

Combining (5.6.27) and (5.6.28) we obtain

$$\left\|\overline{P}\nabla\varphi - \overline{\nabla_{\Gamma}M\varphi}\right\|_{L^{p}(\Omega_{\varepsilon})} \leq \|u\|_{L^{p}(\Omega_{\varepsilon})} + \|\bar{v}\|_{L^{p}(\Omega_{\varepsilon})} \leq c\varepsilon \|\varphi\|_{W^{2,p}(\Omega_{\varepsilon})}.$$

Hence (5.6.25) holds. Also, (5.6.26) follows from (5.3.6), (5.3.8), and (5.6.25).

Lemma 5.6.11. There exists a constant c > 0 independent of ε such that

$$\|Mu \cdot n\|_{W^{1,p}(\Gamma)} \le c\varepsilon^{1-1/p} \|u\|_{W^{2,p}(\Omega_{\varepsilon})}$$
(5.6.29)

for all $u \in W^{2,p}(\Omega_{\varepsilon})^3$, $p \in [1,\infty)$ satisfying $u \cdot n_{\varepsilon} = 0$ on Γ_{ε}^0 or on Γ_{ε}^1 .

Proof. The estimate for the $L^p(\Gamma)$ -norm of $Mu \cdot n$ is given in Lemma 5.6.4. Let us consider the tangential gradient of $Mu \cdot n = M(u \cdot \bar{n})$. By (5.6.17),

$$\nabla_{\Gamma}(Mu \cdot n) = \nabla_{\Gamma}M(u \cdot \bar{n}) = M(B\nabla(u \cdot \bar{n})) + M(\partial_n(u \cdot \bar{n})\psi_{\varepsilon}) \quad \text{on} \quad \Gamma.$$

To the right-hand side we apply (5.6.4) and

$$|B\nabla(u\cdot\bar{n})| \le c \left| \overline{P}\nabla(u\cdot\bar{n}) \right|, \quad |\partial_n(u\cdot\bar{n})\psi_{\varepsilon}| \le c\varepsilon |\nabla u| \quad \text{in} \quad \Omega_{\varepsilon}$$

by (5.2.9), (5.3.6), and (5.6.20) to get

$$\begin{aligned} \|\nabla_{\Gamma}(Mu \cdot n)\|_{L^{p}(\Gamma)} &\leq c \left(\left\|M\left(B\nabla(u \cdot \bar{n})\right)\right\|_{L^{p}(\Gamma)} + \|M(\partial_{n}(u \cdot \bar{n})\psi_{\varepsilon})\|_{L^{p}(\Gamma)} \right) \\ &\leq c\varepsilon^{-1/p} \left(\|B\nabla(u \cdot \bar{n})\|_{L^{p}(\Omega_{\varepsilon})} + \|\partial_{n}(u \cdot \bar{n})\psi_{\varepsilon}\|_{L^{p}(\Omega_{\varepsilon})} \right) \\ &\leq c \left(\varepsilon^{-1/p} \left\|\overline{P}\nabla(u \cdot \bar{n})\right\|_{L^{p}(\Omega_{\varepsilon})} + \varepsilon^{1-1/p} \|\nabla u\|_{L^{p}(\Omega_{\varepsilon})} \right). \end{aligned}$$

Applying (5.3.24) to the first term on the last line we obtain

$$\|\nabla_{\Gamma}(Mu \cdot n)\|_{L^{p}(\Gamma)} \le c\varepsilon^{1-1/p} \|u\|_{W^{2,p}(\Omega_{\varepsilon})}.$$

Hence (5.6.29) follows.

Next we establish estimates for the weighted surface divergence of the average of a solenoidal vector field on Ω_{ε} . They are useful for the proof of the global existence of a strong solution as well as the study of a singular limit problem for (5.1.1)–(5.1.3).

Lemma 5.6.12. Let $p \in [1, \infty)$. There exists c > 0 independent of ε such that

$$\|\operatorname{div}_{\Gamma}(gMu)\|_{L^{p}(\Gamma)} \leq c\varepsilon^{1-1/p} \|u\|_{W^{1,p}(\Omega_{\varepsilon})}$$
(5.6.30)

for all $u \in W^{1,p}(\Omega_{\varepsilon})^3$ satisfying div u = 0 in Ω_{ε} and $u \cdot n_{\varepsilon} = 0$ on Γ_{ε} . If in addition $u \in W^{2,p}(\Omega_{\varepsilon})^3$, then we have

$$\|\operatorname{div}_{\Gamma}(gMu)\|_{W^{1,p}(\Gamma)} \le c\varepsilon^{1-1/p} \|u\|_{W^{2,p}(\Omega_{\varepsilon})}.$$
(5.6.31)

Proof. As in the proofs of the previous lemmas we use the notations (5.2.46) and (5.2.47). Let $u \in W^{1,p}(\Omega_{\varepsilon})^3$ satisfy div u = 0 in Ω_{ε} and $u \cdot n_{\varepsilon} = 0$ on Γ_{ε} . First we show that

$$\operatorname{div}_{\Gamma}(gMu) = \sum_{j=1}^{4} \eta_j \quad \text{on} \quad \Gamma,$$
(5.6.32)

where η_1, \ldots, η_4 are functions on Γ defined by

$$\eta_1 := \sum_{i=0,1} (-1)^i (u_i^{\sharp} - Mu) \cdot \tau_{\varepsilon}^i, \quad \eta_2 := \sum_{i=0,1} (-1)^i Mu \cdot (\tau_{\varepsilon}^i - \nabla_{\Gamma} g_i),$$

$$\eta_3 := -gM \left(d \operatorname{tr} \left[\overline{W} \nabla u \right] \right), \quad \eta_4 := gM(\partial_n u \cdot \psi_{\varepsilon})$$
(5.6.33)

with τ_{ε}^{i} , i = 0, 1 given by (5.2.32). By (5.6.17) and (5.6.18) we have

$$g \operatorname{div}_{\Gamma}(Mu) = g \operatorname{tr}[\nabla_{\Gamma} Mu] = g M(\operatorname{tr}[B\nabla u]) + g M(\operatorname{tr}[\psi_{\varepsilon} \otimes \partial_{n} u])$$
$$= g M\left(\operatorname{tr}\left[\overline{P}\nabla u\right]\right) + \eta_{3} + \eta_{4}$$

on Γ . Moreover, since div u = 0 in Ω_{ε} and $(\bar{n} \otimes \bar{n}) \nabla u = \bar{n} \otimes \partial_n u$,

$$\operatorname{tr}\left[\overline{P}\nabla u\right] = \operatorname{div} u - \operatorname{tr}\left[(\bar{n}\otimes\bar{n})\nabla u\right] = -\bar{n}\cdot\partial_{n}u \quad \text{in} \quad \Omega_{\varepsilon}$$

By these equalities and $\operatorname{div}_{\Gamma}(gMu) = \nabla_{\Gamma}g \cdot Mu + g \operatorname{div}_{\Gamma}(Mu)$ we get

$$\operatorname{div}_{\Gamma}(gMu) = \nabla_{\Gamma}g \cdot Mu - gM(\partial_n u \cdot \bar{n}) + \eta_3 + \eta_4 \quad \text{on} \quad \Gamma.$$
 (5.6.34)

Let us calculate the second term on the right-hand side. Since

$$(\partial_n u \cdot \bar{n})^{\sharp}(y, r) = \frac{\partial}{\partial r} \Big((u \cdot \bar{n})^{\sharp}(y, r) \Big), \quad y \in \Gamma, \, r \in (\varepsilon g_0(y), \varepsilon g_1(y))$$

by $\partial_n \bar{n} = 0$ in Ω_{ε} , we have

$$g(y)[M(\partial_n u \cdot \bar{n})](y) = \frac{1}{\varepsilon} \int_{\varepsilon g_0(y)}^{\varepsilon g_1(y)} \frac{\partial}{\partial r} \Big((u \cdot \bar{n})^{\sharp}(y, r) \Big) dr$$
$$= \frac{1}{\varepsilon} \{ (u \cdot \bar{n})^{\sharp}(y, \varepsilon g_1(y)) - (u \cdot \bar{n})^{\sharp}(y, \varepsilon g_0(y)) \}$$

for $y \in \Gamma$. Moreover, since $u \cdot n_{\varepsilon} = 0$ on Γ_{ε} , we use (5.3.22) to get

$$(u \cdot \bar{n})^{\sharp}(y, \varepsilon g_i(y)) = \varepsilon(u \cdot \bar{\tau}_{\varepsilon}^i)^{\sharp}(y, \varepsilon g_i(y)) = \varepsilon(u_i^{\sharp} \cdot \tau_{\varepsilon}^i)(y), \quad y \in \Gamma.$$

Hence

$$gM(\partial_n u \cdot \bar{n}) = u_1^{\sharp} \cdot \tau_{\varepsilon}^1 - u_0^{\sharp} \cdot \tau_{\varepsilon}^0 = \nabla_{\Gamma} g \cdot M u - \eta_1 - \eta_2 \quad \text{on} \quad \Gamma.$$
(5.6.35)

Combining (5.6.34) and (5.6.35) we obtain (5.6.32). Let us estimate the L^p -norms of the functions η_1, \ldots, η_4 on Γ . By (5.2.34), (5.2.58), and (5.6.7),

$$\|\eta_1\|_{L^p(\Gamma)} \le c \sum_{i=0,1} \|u - \overline{Mu}\|_{L^p(\Gamma^i_{\varepsilon})} \le c\varepsilon^{1-1/p} \|u\|_{W^{1,p}(\Omega_{\varepsilon})}.$$
 (5.6.36)

The first inequality of (5.2.35) and (5.6.4) imply that

$$\|\eta_2\|_{L^p(\Gamma)} \le c\varepsilon \|Mu\|_{L^p(\Gamma)} \le c\varepsilon^{1-1/p} \|u\|_{L^p(\Omega_\varepsilon)}.$$
(5.6.37)

To η_3 and η_4 we apply (5.6.4) and

$$\left| d \operatorname{tr} \left[\overline{W} \nabla u \right] \right| \le c \varepsilon |\nabla u|, \quad |\partial_n u \cdot \psi_{\varepsilon}| \le c \varepsilon |\nabla u| \quad \text{in} \quad \Omega_{\varepsilon}$$

by (5.6.20), $|d| \leq c\varepsilon$ in Ω_{ε} , and the boundedness of W to get

$$\|\eta_3\|_{L^p(\Gamma)} \le c\varepsilon^{-1/p} \left\| d\operatorname{tr}\left[\overline{W}\nabla u\right] \right\|_{L^p(\Omega_{\varepsilon})} \le c\varepsilon^{1-1/p} \|\nabla u\|_{L^p(\Omega_{\varepsilon})},$$

$$\|\eta_4\|_{L^p(\Gamma)} \le c\varepsilon^{-1/p} \|\partial_n u \cdot \psi_{\varepsilon}\|_{L^p(\Omega_{\varepsilon})} \le c\varepsilon^{1-1/p} \|\nabla u\|_{L^p(\Omega_{\varepsilon})}.$$
(5.6.38)

Applying (5.6.36), (5.6.37), and (5.6.38) to (5.6.32) we obtain (5.6.30).

Now we suppose $u \in W^{2,p}(\Omega_{\varepsilon})^3$ and estimate the L^p -norms of $\nabla_{\Gamma}\eta_1, \ldots, \nabla_{\Gamma}\eta_4$ on Γ . By (5.2.47) and

$$\nabla_{\Gamma} (y + \varepsilon g_i(y)n(y)) = P(y) + \varepsilon \{ \nabla_{\Gamma} g_i(y) \otimes n(y) - g_i(y)W(y) \}, \quad y \in \Gamma$$

we have $\nabla_{\Gamma} u_i^{\sharp} = (P + \varepsilon G_i) (\nabla u)_i^{\sharp}$ on Γ , where

$$G_i(y) := \nabla_{\Gamma} g_i(y) \otimes n(y) - g_i(y) W(y), \quad y \in \Gamma, \ i = 0, 1.$$

Hence

$$\nabla_{\Gamma}\eta_{1} = \sum_{i=0,1} (-1)^{i} \{ (\nabla_{\Gamma}u_{i}^{\sharp} - \nabla_{\Gamma}Mu)\tau_{\varepsilon}^{i} + (\nabla_{\Gamma}\tau_{\varepsilon}^{i})(u_{i}^{\sharp} - Mu) \}$$
$$= \sum_{i=0,1} (-1)^{i} \Big[\{ (P(\nabla u)_{i}^{\sharp} - \nabla_{\Gamma}Mu) + \varepsilon G_{i}(\nabla u)_{i}^{\sharp} \} \tau_{\varepsilon}^{i} + (\nabla_{\Gamma}\tau_{\varepsilon}^{i})(u_{i}^{\sharp} - Mu) \Big]$$

on Γ . Since G_0 and G_1 are bounded on Γ , we see by (5.2.34) that

$$|\nabla_{\Gamma}\eta_1| \le c \sum_{i=0,1} (|P(\nabla u)_i^{\sharp} - \nabla_{\Gamma} M u| + \varepsilon |(\nabla u)_i^{\sharp}| + |u_i^{\sharp} - M u|) \quad \text{on} \quad \Gamma.$$

From this inequality and (5.2.58) we deduce that

$$\begin{split} |\nabla_{\Gamma}\eta_{1}\|_{L^{p}(\Gamma)} &\leq c \sum_{i=0,1} \left(\|P(\nabla u)_{i}^{\sharp} - \nabla_{\Gamma}Mu\|_{L^{p}(\Gamma)} + \varepsilon \|(\nabla u)_{i}^{\sharp}\|_{L^{p}(\Gamma)} + \|u_{i}^{\sharp} - Mu\|_{L^{p}(\Gamma)} \right) \\ &\leq c \sum_{i=0,1} \left(\left\|\overline{P}\nabla u - \overline{\nabla_{\Gamma}Mu}\right\|_{L^{p}(\Gamma_{\varepsilon}^{i})} + \varepsilon \|\nabla u\|_{L^{p}(\Gamma_{\varepsilon}^{i})} + \left\|u - \overline{Mu}\right\|_{L^{p}(\Gamma_{\varepsilon}^{i})} \right). \end{split}$$

To the right-hand side we apply (5.3.8), (5.6.7), and (5.6.26) to obtain

$$\|\nabla_{\Gamma}\eta_1\|_{L^p(\Gamma)} \le c\varepsilon^{1-1/p} \|u\|_{W^{2,p}(\Omega_{\varepsilon})}.$$
(5.6.39)

Next we estimate the L^p -norm of $\nabla_{\Gamma}\eta_2$ on Γ . Since

$$\nabla_{\Gamma}\eta_2 = \sum_{i=0,1} (-1)^i \{ (\nabla_{\Gamma}Mu)(\tau_{\varepsilon}^i - \nabla_{\Gamma}g_i) + (\nabla_{\Gamma}\tau_{\varepsilon}^i - \nabla_{\Gamma}^2g_i)Mu \} \quad \text{on} \quad \Gamma_i$$

it follows from (5.2.35) that $|\nabla_{\Gamma}\eta_2| \leq c\varepsilon(|Mu| + |\nabla_{\Gamma}Mu|)$ on Γ . By this inequality and (5.6.22) with m = 1 we get

$$\|\nabla_{\Gamma}\eta_2\|_{L^p(\Gamma)} \le c\varepsilon \|Mu\|_{W^{1,p}(\Gamma)} \le c\varepsilon^{1-1/p} \|u\|_{W^{1,p}(\Omega_{\varepsilon})}.$$
(5.6.40)

Let us consider the tangential gradient of $\eta_3 = -gM(d\phi)$, where $\phi := tr[\overline{W}\nabla u]$. By (5.6.17) we see that

$$\nabla_{\Gamma}\eta_3 = -M(d\phi)\nabla_{\Gamma}g - g\left\{M(\phi B\nabla d) + M(dB\nabla\phi) + M\left(\partial_n(d\phi)\psi_{\varepsilon}\right)\right\}$$

on Γ . Moreover, since $B\nabla d = 0$ in Ω_{ε} by $\nabla d = \bar{n}$ in Ω_{ε} and Pn = 0 on Γ , the second term on the right-hand side vanishes. These facts and (5.6.4) imply that

$$\begin{aligned} \|\nabla_{\Gamma}\eta_{3}\|_{L^{p}(\Gamma)} &\leq c\left(\|M(d\phi)\|_{L^{p}(\Gamma)} + \|M(dB\nabla\phi)\|_{L^{p}(\Gamma)} + \|M(\partial_{n}(d\phi)\psi_{\varepsilon})\|_{L^{p}(\Gamma)}\right) \\ &\leq c\varepsilon^{-1/p}\left(\|d\phi\|_{L^{p}(\Omega_{\varepsilon})} + \|dB\nabla\phi\|_{L^{p}(\Omega_{\varepsilon})} + \|\partial_{n}(d\phi)\psi_{\varepsilon}\|_{L^{p}(\Omega_{\varepsilon})}\right). \end{aligned}$$

To the second line we apply

$$|d\phi| \le c\varepsilon |\nabla u|, \quad |dB\nabla\phi| \le c\varepsilon (|\nabla u| + |\nabla^2 u|), \quad |\partial_n (d\phi)\psi_\varepsilon| \le c\varepsilon (|\nabla u| + |\nabla^2 u|)$$

in Ω_{ε} by (5.6.20), $|d| \leq c\varepsilon$ in Ω_{ε} , and $\partial_n d = \bar{n} \cdot \nabla d = |\bar{n}|^2 = 1$ in N to obtain

$$\|\nabla_{\Gamma}\eta_3\|_{L^p(\Gamma)} \le c\varepsilon^{1-1/p} \|u\|_{W^{2,p}(\Omega_{\varepsilon})}.$$
 (5.6.41)

Let us estimate the $L^p(\Gamma)$ -norm of $\nabla_{\Gamma}\eta_4$. Setting $\xi := \partial_n u \cdot \psi_{\varepsilon}$ we have

$$\nabla_{\Gamma}\eta_4 = (M\xi)\nabla_{\Gamma}g + g\left\{M(B\nabla\xi) + M\left((\partial_n\xi)\psi_{\varepsilon}\right)\right\} \quad \text{on} \quad \mathbf{I}$$

by (5.6.17). From this equality and (5.6.4) we deduce that

$$\begin{aligned} \|\nabla_{\Gamma}\eta_{4}\|_{L^{p}(\Gamma)} &\leq c \left(\|M\xi\|_{L^{p}(\Gamma)} + \|M(B\nabla\xi)\|_{L^{p}(\Gamma)} + \|M((\partial_{n}\xi)\psi_{\varepsilon})\|_{L^{p}(\Gamma)} \right) \\ &\leq c\varepsilon^{-1/p} \left(\|\xi\|_{L^{p}(\Omega_{\varepsilon})} + \|B\nabla\xi\|_{L^{p}(\Omega_{\varepsilon})} + \|(\partial_{n}\xi)\psi_{\varepsilon}\|_{L^{p}(\Omega_{\varepsilon})} \right). \end{aligned}$$
(5.6.42)

We apply (5.6.20) and (5.6.21) to $\xi = \partial_n u \cdot \psi_{\varepsilon}$ and $(\partial_n \xi) \psi_{\varepsilon}$ to obtain

$$\|\xi\|_{L^{p}(\Omega_{\varepsilon})} \leq c\varepsilon \|\nabla u\|_{L^{p}(\Omega_{\varepsilon})},$$

$$\|(\partial_{n}\xi)\psi_{\varepsilon}\|_{L^{p}(\Omega_{\varepsilon})} \leq c\varepsilon \left(\|\nabla u\|_{L^{p}(\Omega_{\varepsilon})} + \|\nabla^{2}u\|_{L^{p}(\Omega_{\varepsilon})}\right).$$

(5.6.43)

Moreover, by (5.6.21) and $P(n \otimes \nabla_{\Gamma} g) = (Pn) \otimes \nabla_{\Gamma} g = 0$ on Γ we see that

$$\overline{P}\nabla\psi_{\varepsilon}\big|=\left|\overline{P}\left(\nabla\psi_{\varepsilon}-\frac{1}{\bar{g}}\bar{n}\otimes\overline{\nabla_{\Gamma}g}\right)\right|\leq c\varepsilon\quad\text{in}\quad\Omega_{\varepsilon}.$$

Using this inequality and (5.6.20) to $B\nabla\xi = B\{\nabla(\partial_n u)\}\psi_{\varepsilon} + B(\nabla\psi_{\varepsilon})\partial_n u$ we get

$$\|B\nabla\xi\|_{L^p(\Omega_{\varepsilon})} \le c\varepsilon \left(\|\nabla u\|_{L^p(\Omega_{\varepsilon})} + \|\nabla^2 u\|_{L^p(\Omega_{\varepsilon})}\right).$$
(5.6.44)

From (5.6.42), (5.6.43), and (5.6.44) it follows that

$$\|\nabla_{\Gamma}\eta_4\|_{L^p(\Omega_{\varepsilon})} \le c\varepsilon^{1-1/p} \|u\|_{W^{2,p}(\Omega_{\varepsilon})}.$$
(5.6.45)

Finally, from (5.6.32), (5.6.39), (5.6.40), (5.6.41), and (5.6.45) we deduce that

$$\left\|\nabla_{\Gamma}\left(\operatorname{div}_{\Gamma}(gMu)\right)\right\|_{L^{p}(\Gamma)} \leq \sum_{j=1}^{4} \|\nabla_{\Gamma}\eta_{j}\|_{L^{p}(\Gamma)} \leq c\varepsilon^{1-1/p} \|u\|_{W^{2,p}(\Omega_{\varepsilon})}$$

and conclude that (5.6.31) is valid.

Lemma 5.6.13. There exists a constant c > 0 independent of ε such that

$$\|\operatorname{div}_{\Gamma}(gM_{\tau}u)\|_{L^{p}(\Gamma)} \leq c\varepsilon^{1-1/p} \|u\|_{W^{1,p}(\Omega_{\varepsilon})}$$
(5.6.46)

for all $u \in W^{1,p}(\Omega_{\varepsilon})^3$, $p \in [1,\infty)$ satisfying div u = 0 in Ω_{ε} and $u \cdot n_{\varepsilon} = 0$ on Γ_{ε} . If in addition $u \in W^{2,p}(\Omega_{\varepsilon})^3$, then we have

$$\|\operatorname{div}_{\Gamma}(gM_{\tau}u)\|_{W^{1,p}(\Gamma)} \le c\varepsilon^{1-1/p} \|u\|_{W^{2,p}(\Omega_{\varepsilon})}.$$
(5.6.47)

Proof. Let $\varphi := g(Mu \cdot n)$. Then

$$\operatorname{div}_{\Gamma}(gM_{\tau}u) = \operatorname{div}_{\Gamma}(gMu) - \operatorname{div}_{\Gamma}(\varphi n) = \operatorname{div}_{\Gamma}(gMu) + \varphi H$$

by $\nabla_{\Gamma} \varphi \cdot n = 0$ and $\operatorname{div}_{\Gamma} n = -H$. Hence

$$\begin{aligned} |\operatorname{div}_{\Gamma}(gM_{\tau}u)| &\leq c(|\operatorname{div}_{\Gamma}(gMu)| + |Mu \cdot n|), \\ \left|\nabla_{\Gamma}\left(\operatorname{div}_{\Gamma}(gM_{\tau}u)\right)\right| &\leq c\left(\left|\nabla_{\Gamma}\left(\operatorname{div}_{\Gamma}(gMu)\right)\right| + |Mu \cdot n| + |\nabla_{\Gamma}(Mu \cdot n)|\right)\end{aligned}$$

on Γ . These inequalities, (5.6.10), and (5.6.29)–(5.6.31) imply (5.6.46) and (5.6.47).

As a consequence of Lemmas 5.6.10 and 5.6.12, we get a relation for the normal derivative (with respect to Γ) of a vector field on Ω_{ε} and its averaged tangential component.

Lemma 5.6.14. Let $p \in [1, \infty)$. There exists c > 0 independent of ε such that

$$\left\| \partial_n u \cdot \bar{n} - \frac{1}{\bar{g}} \overline{M_\tau u} \cdot \overline{\nabla_\Gamma g} \right\|_{L^p(\Omega_\varepsilon)} \le c\varepsilon \|u\|_{W^{2,p}(\Omega_\varepsilon)}$$
(5.6.48)

for all $u \in W^{2,p}(\Omega_{\varepsilon})^3$ satisfying div u = 0 in Ω_{ε} and $u \cdot n_{\varepsilon} = 0$ on Γ_{ε} . Here $\partial_n u$ is the normal derivative of u given by (5.3.5).

Proof. Since $Q = n \otimes n$, $I_3 = P + Q$ on Γ and $\partial_n u = (\bar{n} \cdot \nabla) u = (\nabla u)^T \bar{n}$ in Ω_{ε} ,

$$\partial_n u \cdot \bar{n} = \operatorname{tr}[\bar{n} \otimes \partial_n u] = \operatorname{tr}\left[\overline{Q}\nabla u\right] = \operatorname{div} u - \operatorname{tr}\left[\overline{P}\nabla u\right] \quad \text{in} \quad \Omega_{\varepsilon}.$$

Also, since $\nabla_{\Gamma} g$ is tangential on Γ ,

$$\frac{1}{g}M_{\tau}u\cdot\nabla_{\Gamma}g = \frac{1}{g}Mu\cdot\nabla_{\Gamma}g = \frac{1}{g}\operatorname{div}_{\Gamma}(gMu) - \operatorname{div}_{\Gamma}(Mu) = \frac{1}{g}\operatorname{div}_{\Gamma}(gMu) - \operatorname{tr}[\nabla_{\Gamma}Mu]$$

on Γ . From these equalities, div u = 0 in Ω_{ε} , and (5.2.30) we deduce that

$$\left\|\partial_n u \cdot \bar{n} - \frac{1}{\bar{g}} \overline{M_\tau u} \cdot \overline{\nabla_\Gamma g}\right\|_{L^p(\Omega_\varepsilon)} \le \left\|\overline{P} \nabla u - \overline{\nabla_\Gamma M u}\right\|_{L^p(\Omega_\varepsilon)} + c \left\|\overline{\operatorname{div}_\Gamma(gMu)}\right\|_{L^p(\Omega_\varepsilon)}.$$

Applying (5.2.53), (5.6.25), and (5.6.30) to the right-hand side we obtain (5.6.48).

When $u \in L^2(\Omega_{\varepsilon})^3$, we can consider the weighted surface divergence of $M_{\tau}u$ as an element of $H^{-1}(\Gamma)$. In particular, if $u \in L^2_{\sigma}(\Omega_{\varepsilon})$ then we have an estimate for its $H^{-1}(\Gamma)$ -norm similar to (5.6.46).

Lemma 5.6.15. There exists a constant c > 0 in dependent of ε such that

$$\|\operatorname{div}_{\Gamma}(gM_{\tau}u)\|_{H^{-1}(\Gamma)} \le c\varepsilon^{1/2} \|u\|_{L^{2}(\Omega_{\varepsilon})}$$
(5.6.49)

for all $u \in L^2_{\sigma}(\Omega_{\varepsilon})$.

Proof. We use the notation (5.2.46) and suppress the arguments of functions. Let η be an arbitrary function in $H^1(\Gamma)$. By (5.2.25) we have

$$\langle \operatorname{div}_{\Gamma}(gM_{\tau}u),\eta\rangle_{\Gamma} = -\int_{\Gamma} gM_{\tau}u \cdot \nabla_{\Gamma}\eta \, d\mathcal{H}^2 - \int_{\Gamma} g(M_{\tau}u \cdot n)\eta H \, d\mathcal{H}^2.$$

The second term on the right-hand side vanishes by $M_{\tau}u \cdot n = 0$ on Γ . To estimate the first term, we observe by $M_{\tau}u \cdot \nabla_{\Gamma}\eta = Mu \cdot \nabla_{\Gamma}\eta$ and (5.6.1) that

$$\int_{\Gamma} g M_{\tau} u \cdot \nabla_{\Gamma} \eta \, d\mathcal{H}^2 = \varepsilon^{-1} \int_{\Gamma} \int_{\varepsilon g_0}^{\varepsilon g_1} u^{\sharp} \cdot \nabla_{\Gamma} \eta \, dr \, d\mathcal{H}^2.$$

From this equality and the change of variables formula (5.2.51) it follows that

$$\left|\varepsilon^{-1}\int_{\Omega_{\varepsilon}} u \cdot \nabla \bar{\eta} \, dx - \int_{\Gamma} g M_{\tau} u \cdot \nabla_{\Gamma} \eta \, d\mathcal{H}^2\right| \leq \varepsilon^{-1} (I_1 + I_2),$$

where $\bar{\eta} = \eta \circ \pi$ is the constant extension of η and

$$I_1 := \left| \int_{\Omega_{\varepsilon}} u \cdot \left(\nabla \bar{\eta} - \overline{\nabla_{\Gamma} \eta} \right) dx \right|, \quad I_2 := \left| \int_{\Gamma} \int_{\varepsilon g_0}^{\varepsilon g_1} (u^{\sharp} \cdot \nabla_{\Gamma} \eta) (J-1) \, dr \, d\mathcal{H}^2 \right|.$$

To I_1 we apply (5.2.14) with $|d| \leq c\varepsilon$ in Ω_{ε} , Hölder's inequality, and (5.2.53) to get

$$I_1 \le c\varepsilon \|u\|_{L^2(\Omega_{\varepsilon})} \|\overline{\nabla_{\Gamma}\eta}\|_{L^2(\Omega_{\varepsilon})} \le c\varepsilon^{3/2} \|u\|_{L^2(\Omega_{\varepsilon})} \|\nabla_{\Gamma}\eta\|_{L^2(\Gamma)}$$

We also have the same inequality for I_2 by (5.2.50), (5.2.52), and (5.2.53). Hence

$$\left|\varepsilon^{-1}\int_{\Omega_{\varepsilon}} u \cdot \nabla \bar{\eta} \, dx - \int_{\Gamma} g M_{\tau} u \cdot \nabla_{\Gamma} \eta \, d\mathcal{H}^2\right| \leq \varepsilon^{-1} (I_1 + I_2) \leq c\varepsilon^{1/2} \|u\|_{L^2(\Omega_{\varepsilon})} \|\nabla_{\Gamma} \eta\|_{L^2(\Gamma)}.$$

Here $\int_{\Omega_{\varepsilon}} u \cdot \nabla \bar{\eta} \, dx = 0$ by $u \in L^2_{\sigma}(\Omega_{\varepsilon})$ and $\nabla \bar{\eta} \in L^2_{\sigma}(\Omega_{\varepsilon})^{\perp}$. Therefore,

$$\left|\langle \operatorname{div}_{\Gamma}(gM_{\tau}u),\eta\rangle_{\Gamma}\right| = \left|\int_{\Gamma} gM_{\tau}u \cdot \nabla_{\Gamma}\eta \, d\mathcal{H}^{2}\right| \le c\varepsilon^{1/2} \|u\|_{L^{2}(\Omega_{\varepsilon})} \|\nabla_{\Gamma}\eta\|_{L^{2}(\Gamma)}.$$

Since this inequality holds for all $\eta \in H^1(\Gamma)$, the inequality (5.6.49) is valid.

5.6.3 Decomposition of vector fields into the average and residual parts

In the study of the Navier–Stokes equations in a thin domain it is convenient to decompose a three-dimensional vector field into an almost two-dimensional vector field and its residual term and analyze them separately. Moreover, we can derive a good L^{∞} -estimate for the residual term if it satisfies the impermeable boundary condition. To give such a decomposition we use the impermeable extension operator E_{ε} given by (5.3.42) in Section 5.3.3.

Definition 5.6.16. For a vector field u on Ω_{ε} we define

$$u^{a}(x) := E_{\varepsilon} M_{\tau} u(x) = \overline{M_{\tau} u}(x) + \left\{ \overline{M_{\tau} u}(x) \cdot \Psi_{\varepsilon}(x) \right\} \bar{n}(x), \quad x \in N,$$
(5.6.50)

where Ψ_{ε} is the vector field given by (5.3.36) and $M_{\tau}u$ is the averaged tangential component of u given by (5.6.2). Also, we denote by $u^r := u - u^a$ the residual part of u.

From Lemmas 5.3.12, 5.6.2, and 5.6.9 we observe that if $u \in H^m(\Omega_{\varepsilon})^3$, m = 0, 1, 2, then u^a and u^r are in the same space (here we write $H^0 = L^2$).

Lemma 5.6.17. For $u \in H^m(\Omega_{\varepsilon})^3$ with m = 0, 1, 2 we have $u^a, u^r \in H^m(\Omega_{\varepsilon})^3$ and

$$\|u^a\|_{H^m(\Omega_{\varepsilon})} \le c\|u\|_{H^m(\Omega_{\varepsilon})}, \quad \|u^r\|_{H^m(\Omega_{\varepsilon})} \le c\|u\|_{H^m(\Omega_{\varepsilon})}$$
(5.6.51)

with a constant c > 0 independent of ε and u.

Since the average part u^a can be seen as almost two-dimensional, we expect to have a good L^2 -estimate for the product of u^a and a function on Ω_{ε} . Indeed, we can apply the following product estimate of functions on Γ and Ω_{ε} to u^a .

Lemma 5.6.18. There exists a constant c > 0 independent of ε such that

$$\|\bar{\eta}\varphi\|_{L^{2}(\Omega_{\varepsilon})} \leq c \|\eta\|_{L^{2}(\Gamma)}^{1/2} \|\eta\|_{H^{1}(\Gamma)}^{1/2} \|\varphi\|_{L^{2}(\Omega_{\varepsilon})}^{1/2} \|\varphi\|_{H^{1}(\Omega_{\varepsilon})}^{1/2}$$
(5.6.52)

for all $\eta \in H^1(\Gamma)$ and $\varphi \in H^1(\Omega_{\varepsilon})$, where $\bar{\eta} := \eta \circ \pi$ is the constant extension of η .

Proof. Throughout the proof, we use the notation (5.2.46) and suppress the arguments of functions. By (5.2.52) and (5.6.1) we have

$$\|\bar{\eta}\varphi\|_{L^2(\Omega_{\varepsilon})}^2 \le c \int_{\Gamma} |\eta|^2 \left(\int_{\varepsilon g_0}^{\varepsilon g_1} |\varphi^{\sharp}|^2 \, dr \right) d\mathcal{H}^2 = c\varepsilon \int_{\Gamma} g|\eta|^2 M(|\varphi|^2) \, d\mathcal{H}^2.$$

Noting that g is bounded on Γ , we apply Hölder's inequality to the last integral to get

$$\|\bar{\eta}\varphi\|_{L^{2}(\Omega_{\varepsilon})}^{2} \leq c\varepsilon \|\eta\|_{L^{4}(\Gamma)}^{2} \|M(|\varphi|^{2})\|_{L^{2}(\Gamma)}.$$
(5.6.53)

Here the $L^4(\Gamma)$ -norm of η is estimated by Ladyzhenskaya's inequality (5.3.1). To estimate the $L^2(\Gamma)$ -norm of $M(|\varphi|^2)$ we see that $M(|\varphi|^2) \in W^{1,1}(\Gamma)$ for $\varphi \in H^1(\Omega_{\varepsilon})$. Indeed, by (5.2.30) and (5.2.52) we have

$$\|M(|\varphi|^2)\|_{L^1(\Gamma)} = \int_{\Gamma} \frac{1}{\varepsilon g} \left(\int_{\varepsilon g_0}^{\varepsilon g_1} |\varphi^{\sharp}|^2 \, dr \right) d\mathcal{H}^2 \le c\varepsilon^{-1} \|\varphi\|_{L^2(\Omega_{\varepsilon})}^2.$$
(5.6.54)

Also, by (5.6.17), (5.6.20), and $\nabla(|\varphi|^2) = 2\varphi\nabla\varphi$ we have

$$|\nabla_{\Gamma} M(|\varphi|^2)| \le |M(B\nabla(|\varphi|^2))| + |M(\partial_n(|\varphi|^2)\psi_{\varepsilon})| \le cM(|\varphi\nabla\varphi|) \quad \text{on} \quad \Gamma.$$

Hence from (5.6.4) and Hölder's inequality we deduce that

$$\|\nabla_{\Gamma} M(|\varphi|^{2})\|_{L^{1}(\Gamma)} \leq c \|M(|\varphi \nabla \varphi|)\|_{L^{1}(\Gamma)} \leq c \varepsilon^{-1} \|\varphi \nabla \varphi\|_{L^{1}(\Omega_{\varepsilon})}$$

$$\leq c \varepsilon^{-1} \|\varphi\|_{L^{2}(\Omega_{\varepsilon})} \|\nabla \varphi\|_{L^{2}(\Omega_{\varepsilon})}.$$

$$(5.6.55)$$

Now we observe that the Sobolev embedding $W^{1,1}(\Gamma) \hookrightarrow L^2(\Gamma)$ is valid since $\Gamma \subset \mathbb{R}^3$ is a two-dimensional compact surface without boundary (see e.g. [3, Theorem 2.20]). By this fact, (5.6.54), (5.6.55), and $\|\varphi\|_{L^2(\Omega_{\varepsilon})} \leq \|\varphi\|_{H^1(\Omega_{\varepsilon})}$ we obtain

$$\|M(|\varphi|^2)\|_{L^2(\Gamma)} \le c \|M(|\varphi|^2)\|_{W^{1,1}(\Gamma)} \le c\varepsilon^{-1} \|\varphi\|_{L^2(\Omega_{\varepsilon})} \|\varphi\|_{H^1(\Omega_{\varepsilon})}.$$

Finally, we apply the above inequality and (5.3.1) to (5.6.53) to get

$$\|\bar{\eta}\varphi\|_{L^2(\Omega_{\varepsilon})}^2 \le c\|\eta\|_{L^2(\Gamma)}\|\eta\|_{H^1(\Gamma)}\|\varphi\|_{L^2(\Omega_{\varepsilon})}\|\varphi\|_{H^1(\Omega_{\varepsilon})},$$

which shows (5.6.52).

Corollary 5.6.19. For $\varphi \in H^1(\Omega_{\varepsilon})$, $u \in H^1(\Omega_{\varepsilon})^3$, and u^a given by (5.6.50) we have

$$\| |u^{a}| \varphi \|_{L^{2}(\Omega_{\varepsilon})} \leq c \varepsilon^{-1/2} \| \varphi \|_{L^{2}(\Omega_{\varepsilon})}^{1/2} \| \varphi \|_{H^{1}(\Omega_{\varepsilon})}^{1/2} \| u \|_{L^{2}(\Omega_{\varepsilon})}^{1/2} \| u \|_{H^{1}(\Omega_{\varepsilon})}^{1/2}$$
(5.6.56)

with a constant c > 0 independent of ε , φ , and u. If in addition $u \in H^2(\Omega_{\varepsilon})^3$, then

$$\left\| \nabla u^{a} \right\| \varphi \right\|_{L^{2}(\Omega_{\varepsilon})} \leq c \varepsilon^{-1/2} \|\varphi\|_{L^{2}(\Omega_{\varepsilon})}^{1/2} \|\varphi\|_{H^{1}(\Omega_{\varepsilon})}^{1/2} \|u\|_{H^{1}(\Omega_{\varepsilon})}^{1/2} \|u\|_{H^{2}(\Omega_{\varepsilon})}^{1/2}.$$
(5.6.57)

Proof. By (5.3.37) and $|M_{\tau}u| = |PMu| \le |Mu|$ on Γ we have

 $|u^a| \le (1+|\Psi_{\varepsilon}|) \left| \overline{M_{\tau}u} \right| \le c \left| \overline{Mu} \right| \quad \text{in} \quad \Omega_{\varepsilon}.$

We apply this inequality and (5.6.52) with $\eta = |Mu|$ to the L^2 -norm of $|u^a| \varphi$ on Ω_{ε} and then use (5.6.4) and (5.6.22) to obtain (5.6.56).

Let us prove (5.6.57). We differentiate both sides of (5.6.50) to get

$$\nabla u^a = \nabla \left(\overline{M_\tau u} \right) + \left[\left\{ \nabla \left(\overline{M_\tau u} \right) \right\} \Psi_\varepsilon + (\nabla \Psi_\varepsilon) \overline{M_\tau u} \right] \otimes \overline{n} + \left(\overline{M_\tau u} \cdot \Psi_\varepsilon \right) \nabla \overline{n}$$

in Ω_{ε} . To the right-hand side we apply (5.2.13), (5.2.17), and (5.3.37) to get

$$|\nabla u^{a}| \leq c \left(\left| \overline{M_{\tau} u} \right| + \left| \overline{\nabla_{\Gamma} M_{\tau} u} \right| \right) \leq c \left(\left| \overline{M u} \right| + \left| \overline{\nabla_{\Gamma} M u} \right| \right) \quad \text{in} \quad \Omega_{\varepsilon}.$$

Here the second inequality follows from $M_{\tau}u = PMu$ and $P \in C^4(\Gamma)^{3\times 3}$. Applying the above inequality and (5.6.52) with $\eta = |Mu|$ and $\eta = |\nabla_{\Gamma}Mu|$ to the L^2 -norm of $|\nabla u^a| \varphi$ on Ω_{ε} and then using (5.6.4) and (5.6.22) we obtain (5.6.57).

Next we show Poincaré type inequalities for the residual part $u^r = u - u^a$. For any vector field u on Ω_{ε} we have $u^a \cdot n_{\varepsilon} = 0$ on Γ_{ε} by Lemma 5.3.11. Hence $u^r \cdot n_{\varepsilon} = 0$ on Γ_{ε} if uitself satisfies the same impermeable boundary condition, which is essential for derivation of Poincaré type inequalities.

Lemma 5.6.20. Let $u \in H^1(\Omega_{\varepsilon})^3$ satisfy $u \cdot n_{\varepsilon} = 0$ on Γ_{ε} . Then there exists a constant c > 0 independent of ε and u such that

$$\|u^r\|_{L^2(\Omega_{\varepsilon})} \le c\varepsilon \|\partial_n u^r\|_{L^2(\Omega_{\varepsilon})}$$
(5.6.58)

for $u^r = u - u^a$, where $\partial_n u^r$ is the normal derivative of u^r given by (5.3.5).

Proof. We use the notation (5.6.3) for the tangential and normal components (with respect to Γ) of a vector field on Ω_{ε} . Since $u_{\tau}^{r} = u_{\tau} - \overline{M_{\tau}u}$ in Ω_{ε} by (5.6.50) and $M_{\tau}u = Mu_{\tau}$ on Γ , we use (5.6.6) to get

$$\|u_{\tau}^{r}\|_{L^{2}(\Omega_{\varepsilon})} = \|u_{\tau} - \overline{Mu_{\tau}}\|_{L^{2}(\Omega_{\varepsilon})} \le c\varepsilon \|\partial_{n}u_{\tau}\|_{L^{2}(\Omega_{\varepsilon})} = c\varepsilon \|\partial_{n}u_{\tau}^{r}\|_{L^{2}(\Omega_{\varepsilon})}.$$

Here the last equality follows from (5.3.6) with $\eta = M_{\tau} u$. From this inequality and

$$\|\partial_n u_\tau^r\|_{L^2(\Omega_\varepsilon)} = \|\overline{P}(\partial_n u^r)\|_{L^2(\Omega_\varepsilon)} \le \|\partial_n u^r\|_{L^2(\Omega_\varepsilon)}$$

by $u_{\tau}^{r} = \overline{P}u^{r}$ and $\partial_{n}\overline{P} = 0$ in Ω_{ε} we deduce that

$$\|u_{\tau}^{r}\|_{L^{2}(\Omega_{\varepsilon})} \le c\varepsilon \|\partial_{n}u^{r}\|_{L^{2}(\Omega_{\varepsilon})}.$$
(5.6.59)

To estimate the $L^2(\Omega_{\varepsilon})$ -norm of u_n^r , we see that u^r satisfies $u^r \cdot n_{\varepsilon} = 0$ on Γ_{ε} since u and u^a satisfy the same boundary condition. Hence we can apply (5.3.23) to get

$$\|u_n^r\|_{L^2(\Omega_{\varepsilon})} = \|u^r \cdot \bar{n}\|_{L^2(\Omega_{\varepsilon})} \le c\varepsilon \|\partial_n u^r\|_{L^2(\Omega_{\varepsilon})}.$$

Combining (5.6.59) and the above inequality we obtain (5.6.58).

Lemma 5.6.21. Suppose that Assumption 1 is satisfied, i.e. the inequality (5.1.6) is valid. Let $u \in H^2(\Omega_{\varepsilon})^3$ satisfy div u = 0 in Ω_{ε} and the slip boundary conditions (5.3.20)–(5.3.21) on Γ_{ε} . Then $u^r = u - u^a$ satisfies

$$\|\nabla u^r\|_{L^2(\Omega_{\varepsilon})} \le c\left(\varepsilon \|u\|_{H^2(\Omega_{\varepsilon})} + \|u\|_{L^2(\Omega_{\varepsilon})}\right).$$
(5.6.60)

Here c > 0 is a constant independent of ε and u.

Proof. As in the proof of Lemma 5.6.20, we use the notation (5.6.3). Based on the equalities $u^r = u^r_{\tau} + u^r_n$ and $I_3 = \overline{P} + \overline{Q}$ we split the gradient matrix of u^r into

$$\nabla u^r = \overline{P} \nabla u^r_\tau + \overline{Q} \nabla u^r_\tau + \overline{P} \nabla u^r_n + \overline{Q} \nabla u^r_n \quad \text{in} \quad \Omega_{\varepsilon}$$
(5.6.61)

and estimate the $L^2(\Omega_{\varepsilon})$ -norm of each term on the right-hand side.

First we derive an estimate for $\overline{P}\nabla u_{\tau}^{r}$. We differentiate $u_{\tau}^{r} = u_{\tau} - \overline{M}u_{\tau}$ and use (5.2.8), (5.2.12), and $P\nabla_{\Gamma} = \nabla_{\Gamma}$ to get

$$\overline{P}\nabla u_{\tau}^{r} = \left(\overline{P}\nabla u_{\tau} - \overline{\nabla_{\Gamma}Mu_{\tau}}\right) - \left\{I_{3} - \left(I_{3} - d\overline{W}\right)^{-1}\right\}\overline{\nabla_{\Gamma}Mu_{\tau}} \quad \text{in} \quad \Omega_{\varepsilon}$$

Hence by (5.2.10) with $|d| \le c\varepsilon$ in Ω_{ε} , (5.2.53), (5.6.22), and (5.6.25) we see that

$$\begin{aligned} \left\| \overline{P} \nabla u_{\tau}^{r} \right\|_{L^{2}(\Omega_{\varepsilon})} &\leq \left\| \overline{P} \nabla u_{\tau} - \overline{\nabla_{\Gamma} M u_{\tau}} \right\|_{L^{2}(\Omega_{\varepsilon})} + c\varepsilon \left\| \overline{\nabla_{\Gamma} M u_{\tau}} \right\|_{L^{2}(\Omega_{\varepsilon})} \\ &\leq c\varepsilon \| u_{\tau} \|_{H^{2}(\Omega_{\varepsilon})} \leq c\varepsilon \| u \|_{H^{2}(\Omega_{\varepsilon})}. \end{aligned}$$
(5.6.62)

Next we deal with $\overline{Q}\nabla u_{\tau}^{r}$. By $u_{\tau}^{r} = \overline{P}u - \overline{M_{\tau}u}$ and (5.3.6),

$$\left|\overline{Q}\nabla u_{\tau}^{r}\right| = \left|\overline{n}\otimes\partial_{n}u_{\tau}^{r}\right| = \left|\partial_{n}u_{\tau}^{r}\right| = \left|\overline{P}\partial_{n}u\right|$$
 in Ω_{ε} .

Since we suppose that Assumption 1 is satisfied and that u satisfies (5.3.20)–(5.3.21) on Γ_{ε} , we can use (5.3.34) to get

$$\begin{aligned} \left\| \overline{Q} \nabla u_{\tau}^{r} \right\|_{L^{2}(\Omega_{\varepsilon})} &= \left\| \overline{P} \partial_{n} u \right\|_{L^{2}(\Omega_{\varepsilon})} \leq \left\| \overline{P} \partial_{n} u + \overline{W} u \right\|_{L^{2}(\Omega_{\varepsilon})} + \left\| \overline{W} u \right\|_{L^{2}(\Omega_{\varepsilon})} \\ &\leq c \left(\varepsilon \| u \|_{H^{2}(\Omega_{\varepsilon})} + \| u \|_{L^{2}(\Omega_{\varepsilon})} \right). \end{aligned}$$

$$(5.6.63)$$

Let us estimate the $L^2(\Omega_{\varepsilon})$ -norm of $\overline{P}\nabla u_n^r$. Since $u_n^r = (u^r \cdot \overline{n})\overline{n}$ we have

$$\overline{P}\nabla u_n^r = \left[\overline{P}\nabla(u^r\cdot\bar{n})\right]\otimes\bar{n} - (u^r\cdot\bar{n})\overline{P}\nabla\bar{n} \quad \text{in} \quad \Omega_{\varepsilon}$$

By this formula, (5.2.13), (5.2.17), and |n| = |P| = 1 on Γ ,

$$\left\|\overline{P}\nabla u_n^r\right\|_{L^2(\Omega_{\varepsilon})} \le c\left(\left\|\overline{P}\nabla(u^r\cdot\bar{n})\right\|_{L^2(\Omega_{\varepsilon})} + \|u^r\|_{L^2(\Omega_{\varepsilon})}\right).$$

Here u^r satisfies $u^r \cdot n_{\varepsilon} = 0$ on Γ_{ε} since u satisfies the same boundary condition. Thus, we can apply (5.3.24) and (5.6.58) to the above inequality to get

$$\left\|\overline{P}\nabla u_n^r\right\|_{L^2(\Omega_{\varepsilon})} \le c\varepsilon \|u\|_{H^2(\Omega_{\varepsilon})}.$$
(5.6.64)

Now let us consider $\overline{Q}\nabla u_n^r = \overline{n} \otimes \partial_n u_n^r$. Since

$$u_n^r = (u^r \cdot \bar{n})\bar{n} = \left(u \cdot \bar{n} - \overline{M_\tau u} \cdot \Psi_\varepsilon\right)\bar{n}, \quad \partial_n u_n^r = \left(\partial_n u \cdot \bar{n} - \overline{M_\tau u} \cdot \partial_n \Psi_\varepsilon\right)\bar{n}$$

by (5.3.6) and (5.6.50), we have

$$\begin{aligned} \overline{Q}\nabla u_n^r \Big| &= \left|\partial_n u_n^r\right| = \left|\partial_n u \cdot \bar{n} - \overline{M_\tau u} \cdot \partial_n \Psi_\varepsilon\right| \\ &\leq \left|\partial_n u \cdot \bar{n} - \frac{1}{\bar{g}}\overline{M_\tau u} \cdot \overline{\nabla_\Gamma g}\right| + \left|\overline{M_\tau u}\right| \left|\partial_n \Psi_\varepsilon - \frac{1}{\bar{g}}\overline{\nabla_\Gamma g}\right| \end{aligned}$$

in Ω_{ε} . By this inequality, $|M_{\tau}u| \leq |Mu|$ on Γ , (5.3.38), (5.6.5), and (5.6.48) we see that

$$\left\|\overline{Q}\nabla u_n^r\right\|_{L^2(\Omega_{\varepsilon})} \le \left\|\partial_n u \cdot \bar{n} - \frac{1}{\bar{g}}\overline{M_{\tau}u} \cdot \overline{\nabla_{\Gamma}g}\right\|_{L^2(\Omega_{\varepsilon})} + c\varepsilon \left\|\overline{Mu}\right\|_{L^2(\Omega_{\varepsilon})} \le c\varepsilon \|u\|_{H^2(\Omega_{\varepsilon})}.$$
 (5.6.65)

Here we used the fact that u satisfies div u = 0 in Ω_{ε} and (5.3.20) on Γ_{ε} to apply (5.6.48). Finally, applying (5.6.62)–(5.6.65) to the right-hand side of (5.6.61) we obtain (5.6.60).

As a consequence of Lemmas 5.6.20 and 5.6.21, we obtain a good L^{∞} -estimate for the residual term u^r on Ω_{ε} .

Lemma 5.6.22. Under Assumption 1, let $u \in H^2(\Omega_{\varepsilon})^3$ satisfy div u = 0 in Ω_{ε} and (5.3.20) - (5.3.21) on Γ_{ε} . Then there exists a constant c > 0 independent of ε and u such that

$$\|u^{r}\|_{L^{\infty}(\Omega_{\varepsilon})} \leq c \left(\varepsilon^{1/2} \|u\|_{H^{2}(\Omega_{\varepsilon})} + \|u\|_{L^{2}(\Omega_{\varepsilon})}^{1/2} \|u\|_{H^{2}(\Omega_{\varepsilon})}^{1/2}\right)$$
(5.6.66)

for $u^r = u - u^a$, where u^a is given by (5.6.50).

Proof. Since $u^r \in H^2(\Omega_{\varepsilon})^3$, we can use Agmon's inequality (5.3.11) to get

$$\begin{aligned} \|u^r\|_{L^{\infty}(\Omega_{\varepsilon})} &\leq c\varepsilon^{-1/2} \|u^r\|_{L^2(\Omega_{\varepsilon})}^{1/4} \|u^r\|_{H^2(\Omega_{\varepsilon})}^{1/2} \\ &\times \left(\|u^r\|_{L^2(\Omega_{\varepsilon})} + \varepsilon \|\partial_n u^r\|_{L^2(\Omega_{\varepsilon})} + \varepsilon^2 \|\partial_n^2 u^r\|_{L^2(\Omega_{\varepsilon})}\right)^{1/4} \end{aligned}$$

To the right-hand side of this inequality we apply

$$\|\partial_n^2 u^r\|_{L^2(\Omega_{\varepsilon})} \le c \|u^r\|_{H^2(\Omega_{\varepsilon})} \le c \|u\|_{H^2(\Omega_{\varepsilon})},$$

$$\|u^r\|_{L^2(\Omega_{\varepsilon})} \le c\varepsilon \|\partial_n u^r\|_{L^2(\Omega_{\varepsilon})} \le c \left(\varepsilon^2 \|u\|_{H^2(\Omega_{\varepsilon})} + \varepsilon \|u\|_{L^2(\Omega_{\varepsilon})}\right)$$

by (5.6.51), (5.6.58), and (5.6.60) to get

$$\|u^{r}\|_{L^{\infty}(\Omega_{\varepsilon})} \leq c\varepsilon^{-1/2} \left(\varepsilon^{2} \|u\|_{H^{2}(\Omega_{\varepsilon})} + \varepsilon \|u\|_{L^{2}(\Omega_{\varepsilon})}\right)^{1/2} \|u\|_{H^{2}(\Omega_{\varepsilon})}^{1/2} = c \left(\varepsilon \|u\|_{H^{2}(\Omega_{\varepsilon})} + \|u\|_{L^{2}(\Omega_{\varepsilon})}\right)^{1/2} \|u\|_{H^{2}(\Omega_{\varepsilon})}^{1/2}.$$

Using $(a+b)^{1/2} \le a^{1/2} + b^{1/2}$ for $a, b \ge 0$ to this inequality we obtain (5.6.66).

Finally, let us estimate the $L^2(\Omega_{\varepsilon})$ -norm of $u \otimes u$ and $(u \cdot \nabla)u$ by using the product estimate for the average part and the L^{∞} -estimate for the residual part.

Lemma 5.6.23. Under Assumption 1, let $u \in H^2(\Omega_{\varepsilon})^3$ satisfy div u = 0 in Ω_{ε} and (5.3.20) - (5.3.21) on Γ_{ε} . Then there exists a constant c > 0 independent of ε and u such that

$$\|u \otimes u\|_{L^{2}(\Omega_{\varepsilon})} \leq c \left(\varepsilon^{-1/2} \|u\|_{L^{2}(\Omega_{\varepsilon})} \|u\|_{H^{1}(\Omega_{\varepsilon})} + \varepsilon^{1/2} \|u\|_{L^{2}(\Omega_{\varepsilon})} \|u\|_{H^{2}(\Omega_{\varepsilon})} + \|u\|_{L^{2}(\Omega_{\varepsilon})}^{3/2} \|u\|_{H^{2}(\Omega_{\varepsilon})}^{1/2} \right)$$
(5.6.67)

and

$$\|(u\cdot\nabla)u\|_{L^2(\Omega_{\varepsilon})} \le c\left(\varepsilon^{-1/2}\|u\|_{L^2(\Omega_{\varepsilon})} + \varepsilon^{1/2}\|u\|_{H^1(\Omega_{\varepsilon})}\right)\|u\|_{H^2(\Omega_{\varepsilon})}.$$
(5.6.68)

Proof. Let u^a be the average part of u given by (5.6.50) and $u^r = u - u^a$ the residual part. Since $u \in H^2(\Omega_{\varepsilon})^3$ satisfies div u = 0 in Ω_{ε} and (5.3.20)–(5.3.21) on Γ_{ε} we have

$$\begin{aligned} \|u^{a} \otimes u\|_{L^{2}(\Omega_{\varepsilon})} &\leq c\varepsilon^{-1/2} \|u\|_{L^{2}(\Omega_{\varepsilon})} \|u\|_{H^{1}(\Omega_{\varepsilon})}, \\ \|u^{r} \otimes u\|_{L^{2}(\Omega_{\varepsilon})} &\leq \|u^{r}\|_{L^{\infty}(\Omega_{\varepsilon})} \|u\|_{L^{2}(\Omega_{\varepsilon})} \leq c \left(\varepsilon^{1/2} \|u\|_{H^{2}(\Omega_{\varepsilon})} + \|u\|_{L^{2}(\Omega_{\varepsilon})}^{1/2} \|u\|_{H^{2}(\Omega_{\varepsilon})}^{1/2}\right) \|u\|_{L^{2}(\Omega_{\varepsilon})} \end{aligned}$$

by (5.6.56) and (5.6.66). Applying these inequalities to the right-hand side of

$$\|u \otimes u\|_{L^{2}(\Omega_{\varepsilon})} \leq \|u^{a} \otimes u\|_{L^{2}(\Omega_{\varepsilon})} + \|u^{r} \otimes u\|_{L^{2}(\Omega_{\varepsilon})}$$

we obtain (5.6.67). To prove (5.6.68) we observe by (5.5.33) and (5.6.56) that

$$\|(u^{a} \cdot \nabla)u\|_{L^{2}(\Omega_{\varepsilon})} \leq c\varepsilon^{-1/2} \|u\|_{L^{2}(\Omega_{\varepsilon})}^{1/2} \|u\|_{H^{1}(\Omega_{\varepsilon})} \|u\|_{H^{2}(\Omega_{\varepsilon})}^{1/2} \leq c\varepsilon^{-1/2} \|u\|_{L^{2}(\Omega_{\varepsilon})} \|u\|_{H^{2}(\Omega_{\varepsilon})}.$$

Also, by (5.5.33) and (5.6.66) we get

$$\begin{split} \| (u^{r} \cdot \nabla) u \|_{L^{2}(\Omega_{\varepsilon})} &\leq c \| u^{r} \|_{L^{\infty}(\Omega_{\varepsilon})} \| \nabla u \|_{L^{2}(\Omega_{\varepsilon})} \\ &\leq c \left(\varepsilon^{1/2} \| u \|_{H^{2}(\Omega_{\varepsilon})} + \| u \|_{L^{2}(\Omega_{\varepsilon})}^{1/2} \| u \|_{H^{2}(\Omega_{\varepsilon})} \right) \| u \|_{H^{1}(\Omega_{\varepsilon})} \\ &= c \left(\| u \|_{L^{2}(\Omega_{\varepsilon})}^{1/2} \| u \|_{H^{1}(\Omega_{\varepsilon})} \| u \|_{H^{2}(\Omega_{\varepsilon})}^{1/2} + \varepsilon^{1/2} \| u \|_{H^{1}(\Omega_{\varepsilon})} \| u \|_{H^{2}(\Omega_{\varepsilon})} \right) \\ &\leq c \left(\| u \|_{L^{2}(\Omega_{\varepsilon})} + \varepsilon^{1/2} \| u \|_{H^{1}(\Omega_{\varepsilon})} \right) \| u \|_{H^{2}(\Omega_{\varepsilon})}. \end{split}$$

We use these two estimates to the right-hand side of

$$\|(u \cdot \nabla)u\|_{L^2(\Omega_{\varepsilon})} \le \|(u^a \cdot \nabla)u\|_{L^2(\Omega_{\varepsilon})} + \|(u^r \cdot \nabla)u\|_{L^2(\Omega_{\varepsilon})}$$

and note that $1 \le \varepsilon^{-1/2}$ by $\varepsilon < 1$ to obtain (5.6.68).

5.6.4 Average of bilinear and trilinear forms

We consider approximation of bilinear and trilinear forms for functions on Ω_{ε} by those for functions on Γ and the average operators. The results in this subsection are fundamental for the study of a singular limit problem for the Navier–Stokes equations (5.1.1)–(5.1.3). Throughout this subsection, we denote by $\bar{\eta} = \eta \circ \pi$ the constant extension of a function η on Γ in the normal direction of Γ . We also use the notations (5.2.46) and (5.2.47) and suppress the arguments of functions.

First we consider the L^2 -inner products on Ω_{ε} and Γ_{ε}^i , i = 0, 1.

Lemma 5.6.24. There exists a constant c > 0 independent of ε such that

$$\left| \int_{\Omega_{\varepsilon}} \varphi \bar{\eta} \, dx - \varepsilon \int_{\Gamma} g(M\varphi) \eta \, d\mathcal{H}^2 \right| \le c \varepsilon^{3/2} \|\varphi\|_{L^2(\Omega_{\varepsilon})} \|\eta\|_{L^2(\Gamma)} \tag{5.6.69}$$

for all $\varphi \in L^2(\Omega_{\varepsilon})$ and $\eta \in L^2(\Gamma)$.

Proof. By the change of variables formula (5.2.51) and the definition (5.6.1) of the average operator,

$$\int_{\Omega_{\varepsilon}} \varphi \bar{\eta} \, dx - \varepsilon \int_{\Gamma} g(M\varphi) \eta \, d\mathcal{H}^2 = \int_{\Gamma} \int_{\varepsilon g_0}^{\varepsilon g_1} \varphi^{\sharp} \eta(J-1) \, dr \, d\mathcal{H}^2$$

We apply (5.2.50), (5.2.52), Hölder's inequality, and (5.2.53) to the right-hand side to get

$$\left| \int_{\Gamma} \int_{\varepsilon g_0}^{\varepsilon g_1} \varphi^{\sharp} \eta(J-1) \, dr \, d\mathcal{H}^2 \right| \le c \varepsilon \|\varphi\|_{L^2(\Omega_{\varepsilon})} \|\bar{\eta}\|_{L^2(\Omega_{\varepsilon})} \le c \varepsilon^{3/2} \|\varphi\|_{L^2(\Omega_{\varepsilon})} \|\eta\|_{L^2(\Gamma)}.$$

Hence we obtain (5.6.69).

Lemma 5.6.25. There exists a constant c > 0 independent of ε such that

$$\left| \int_{\Gamma_{\varepsilon}^{i}} \varphi \bar{\eta} \, d\mathcal{H}^{2} - \int_{\Gamma} (M\varphi) \eta \, d\mathcal{H}^{2} \right| \leq \varepsilon^{1/2} \|\varphi\|_{H^{1}(\Omega_{\varepsilon})} \|\eta\|_{L^{2}(\Gamma)}, \quad i = 0, 1$$
(5.6.70)

for all $\varphi \in H^1(\Omega_{\varepsilon})$ and $\eta \in L^2(\Gamma)$.

Proof. Using the change of variables formula (5.2.57) we have

$$\int_{\Gamma_{\varepsilon}^{i}} \varphi \bar{\eta} \, d\mathcal{H}^{2} - \int_{\Gamma} (M\varphi) \eta \, d\mathcal{H}^{2} = I_{1} + I_{2}, \qquad (5.6.71)$$

where

$$I_1 := \int_{\Gamma} \varphi_i^{\sharp} \eta \left(J_i^{\sharp} \sqrt{1 + \varepsilon^2 |\tau_{\varepsilon}^i|^2} - 1 \right) d\mathcal{H}^2, \quad I_2 := \int_{\Gamma} (\varphi_i^{\sharp} - M\varphi) \eta \, d\mathcal{H}^2.$$

Here we use the notation (5.2.47) and $J_i^{\sharp}(y) := J(y, \varepsilon g_i(y))$ for $y \in \Gamma$. By (5.2.34), (5.2.50), and (5.4.9) we observe that

$$\left|J_i^{\sharp}\sqrt{1+\varepsilon^2|\tau_{\varepsilon}^i|^2}-1\right| \le |J_i^{\sharp}-1|\sqrt{1+\varepsilon^2|\tau_{\varepsilon}^i|^2} + \left(\sqrt{1+\varepsilon^2|\tau_{\varepsilon}^i|^2}-1\right) \le c\varepsilon$$

on Γ . From this inequality, (5.2.58), and (5.3.8) it follows that

$$|I_1| \le c\varepsilon \|\varphi_i^{\sharp}\|_{L^2(\Gamma)} \|\eta\|_{L^2(\Gamma)} \le c\varepsilon \|\varphi\|_{L^2(\Gamma_{\varepsilon}^i)} \|\eta\|_{L^2(\Gamma)} \le c\varepsilon^{1/2} \|\varphi\|_{H^1(\Omega_{\varepsilon})} \|\eta\|_{L^2(\Gamma)}.$$

Also, by (5.2.58) and (5.6.7) we have

$$|I_2| \le \|\varphi_i^{\sharp} - M\varphi\|_{L^2(\Gamma)} \|\eta\|_{L^2(\Gamma)} \le c \left\|\varphi - \overline{M\varphi}\right\|_{L^2(\Gamma_{\varepsilon}^i)} \|\eta\|_{L^2(\Gamma)} \le c\varepsilon^{1/2} \|\varphi\|_{H^1(\Omega_{\varepsilon})} \|\eta\|_{L^2(\Gamma)}.$$

Applying these two estimates to (5.6.71) we obtain (5.6.70).

Next we deal with bilinear forms including the strain rate tensor

$$D(u) = (\nabla u)_S = \frac{\nabla u + (\nabla u)^T}{2}$$

for a vector field u on Ω_{ε} .

Lemma 5.6.26. There exists a constant c > 0 independent of ε such that

$$\left| \int_{\Omega_{\varepsilon}} D(u) : \overline{A} \, dx - \varepsilon \int_{\Gamma} g D_{\Gamma}(M_{\tau} u) : A \, d\mathcal{H}^2 \right| \le c \varepsilon^{3/2} \|u\|_{H^1(\Omega_{\varepsilon})} \|A\|_{L^2(\Gamma)} \tag{5.6.72}$$

for all $u \in H^1(\Omega_{\varepsilon})^3$ and $A \in L^2(\Gamma)^{3 \times 3}$ satisfying

$$u \cdot n_{\varepsilon} = 0$$
 on Γ_{ε} , $PA = AP = A$ on Γ .

Here $D_{\Gamma}(M_{\tau}u)$ is the surface strain rate tensor given by (5.4.38).

Proof. Since PA = AP = A and $P^T = P$ on Γ , we have

$$PM(D(u))P : A = M(D(u)) : A \text{ on } \Gamma.$$

From this equality, (5.6.69), and $||D(u)||_{L^2(\Omega_{\varepsilon})} \leq c ||u||_{H^1(\Omega_{\varepsilon})}$ it follows that

$$\int_{\Omega_{\varepsilon}} D(u) : \overline{A} \, dx - \varepsilon \int_{\Gamma} gPM(D(u))P : A \, d\mathcal{H}^2 \bigg| \le c\varepsilon^{3/2} \|u\|_{H^1(\Omega_{\varepsilon})} \|A\|_{L^2(\Gamma)}.$$
(5.6.73)

By (5.6.17), (5.6.18), (5.6.20), and $|d| \leq c \varepsilon$ in Ω_{ε} we have

$$\left|\nabla_{\Gamma} M u - P M(\nabla u)\right| \le \left|M\left(d\overline{W}\nabla u\right)\right| + \left|M(\psi_{\varepsilon} \otimes \partial_{n} u)\right| \le c\varepsilon M(|\nabla u|) \quad \text{on} \quad \Gamma.$$
(5.6.74)

Noting that $P\nabla_{\Gamma}Mu = \nabla_{\Gamma}Mu$ and P is bounded on Γ , we deduce from (5.6.74) that

$$|D_{\Gamma}(Mu) - PM(D(u))P| \le c|\{\nabla_{\Gamma}Mu - PM(\nabla u)\}P| \le c\varepsilon M(|\nabla u|) \text{ on } \Gamma.$$

By this inequality, the boundedness of g on Γ , and (5.6.4) we see that

$$\left| \int_{\Gamma} gD_{\Gamma}(Mu) : A \, d\mathcal{H}^2 - \int_{\Gamma} gPM(D(u))P : A \, d\mathcal{H}^2 \right| \le c\varepsilon \|M(|\nabla u|)\|_{L^2(\Gamma)} \|A\|_{L^2(\Gamma)}$$

$$\le c\varepsilon^{1/2} \|u\|_{H^1(\Omega_{\varepsilon})} \|A\|_{L^2(\Gamma)}.$$
(5.6.75)

Moreover, by the decomposition $Mu = (Mu \cdot n)n + M_{\tau}u$ and $-\nabla_{\Gamma}n = W$ we get

$$\nabla_{\Gamma} M u = \nabla_{\Gamma} (M u \cdot n) \otimes n - (M u \cdot n) W + \nabla_{\Gamma} M_{\tau} u \quad \text{on} \quad \Gamma$$

Since $(a \otimes n)P = a \otimes (Pn) = 0$ for any $a \in \mathbb{R}^3$ by $P^T = P$ and Pn = 0,

$$P(\nabla_{\Gamma} M u)P - P(\nabla_{\Gamma} M_{\tau} u)P = -(Mu \cdot n)PWP = -(Mu \cdot n)W \quad \text{on} \quad \Gamma,$$

where we used (5.2.6) in the last equality. By this equality and the boundedness of W on Γ ,

$$|D_{\Gamma}(Mu) - D_{\Gamma}(M_{\tau}u)| \le |P(\nabla_{\Gamma}Mu)P - P(\nabla_{\Gamma}M_{\tau}u)P| \le c|Mu \cdot n| \quad \text{on} \quad \Gamma.$$

Hence by (5.6.10) we have

$$\left| \int_{\Gamma} g D_{\Gamma}(Mu) : A \, d\mathcal{H}^2 - \int_{\Gamma} g D_{\Gamma}(M_{\tau}u) : A \, d\mathcal{H}^2 \right| \le c \|Mu \cdot n\|_{L^2(\Gamma)} \|A\|_{L^2(\Gamma)}$$
$$\le c \varepsilon^{1/2} \|u\|_{H^1(\Omega_{\varepsilon})} \|A\|_{L^2(\Gamma)}.$$

Combining this inequality, (5.6.73), and (5.6.75) we obtain (5.6.72).

Lemma 5.6.27. There exists a constant c > 0 independent of ε such that

$$\left| \int_{\Omega_{\varepsilon}} \left(D(u) : \overline{Q} \right) \bar{\eta} \, dx - \varepsilon \int_{\Gamma} (M_{\tau} u \cdot \nabla_{\Gamma} g) \eta \, d\mathcal{H}^2 \right| \le c \varepsilon^{3/2} \|u\|_{H^1(\Omega_{\varepsilon})} \|\eta\|_{L^2(\Gamma)} \tag{5.6.76}$$

for all $u \in H^1(\Omega_{\varepsilon})^3$ satisfying $u \cdot n_{\varepsilon} = 0$ on Γ_{ε} and $\eta \in L^2(\Gamma)$.

Proof. First note that, by the change of variables formula (5.2.51), the inequalities (5.2.50), (5.2.52), and (5.2.53), and |Q| = |n| = 1 we have

$$\left| \int_{\Omega_{\varepsilon}} \left(D(u) : \overline{Q} \right) \bar{\eta} \, dx - \int_{\Gamma} \left(\int_{\varepsilon g_0}^{\varepsilon g_1} D(u)^{\sharp} : Q \, dr \right) \eta \, d\mathcal{H}^2 \right| \le c \varepsilon^{3/2} \|u\|_{H^1(\Omega_{\varepsilon})} \|\eta\|_{L^2(\Gamma)}.$$
(5.6.77)

Let us calculate the integral of $D(u)^{\sharp}(y,r) : Q(y)$ with respect to r. Since $Q = n \otimes n$ is symmetric, we have

$$D(u)(y+rn(y)):Q(y) = \nabla u(y+rn(y)):n(y) \otimes n(y) = [(n(y) \cdot \nabla)u](y+rn(y)) \cdot n(y)$$
$$= \frac{\partial}{\partial r} \Big(u(y+rn(y)) \Big) \cdot n(y)$$

for all $y \in \Gamma$ and $r \in (\varepsilon g_0(y), \varepsilon g_1(y))$, i.e. $D(u)^{\sharp} : Q = (\partial u^{\sharp} / \partial r) \cdot n$. Hence

$$\int_{\varepsilon g_0}^{\varepsilon g_1} D(u)^{\sharp} : Q \, dr = \left(\int_{\varepsilon g_0}^{\varepsilon g_1} \frac{\partial u^{\sharp}}{\partial r} \, dr \right) \cdot n = u_1^{\sharp} \cdot n - u_0^{\sharp} \cdot n \quad \text{on} \quad \Gamma.$$
(5.6.78)

Here and hereafter we use the notation (5.2.47). Since u satisfies $u \cdot n_{\varepsilon} = 0$ on Γ_{ε} ,

$$(-1)^{i+1}u_i^{\sharp} \cdot n = u_i^{\sharp} \cdot \{(-1)^{i+1}(n - \varepsilon \nabla_{\Gamma} g_i) - n_{\varepsilon,i}^{\sharp}\} + \varepsilon (-1)^{i+1}u_i^{\sharp} \cdot \nabla_{\Gamma} g_i$$

on Γ for i = 0, 1. From this equality and $g = g_1 - g_0$ it follows that

$$\begin{split} u_1^{\sharp} \cdot n - u_0^{\sharp} \cdot n &= \sum_{i=0,1} (-1)^{i+1} u_i^{\sharp} \cdot n \\ &= \sum_{i=0,1} u_i^{\sharp} \cdot \{ (-1)^{i+1} (n - \varepsilon \nabla_{\Gamma} g_i) - n_{\varepsilon,i}^{\sharp} \} \\ &+ \varepsilon \sum_{i=0,1} (-1)^{i+1} (u_i^{\sharp} - M u) \cdot \nabla_{\Gamma} g_i + \varepsilon M u \cdot \nabla_{\Gamma} g \end{split}$$

on Γ . Applying (5.2.40) to the second line we get

$$|(u_1^{\sharp} \cdot n - u_0^{\sharp} \cdot n) - \varepsilon M u \cdot \nabla_{\Gamma} g| \le c \varepsilon \sum_{i=0,1} (\varepsilon |u_i^{\sharp}| + |u_i^{\sharp} - M u|) \quad \text{on} \quad \Gamma.$$
(5.6.79)

Combining (5.6.78) and (5.6.79) and using Hölder's inequality we see that

$$\begin{aligned} \left| \int_{\Gamma} \left(\int_{\varepsilon g_0}^{\varepsilon g_1} D(u)^{\sharp} : Q \, dr \right) \eta \, d\mathcal{H}^2 - \varepsilon \int_{\Gamma} (Mu \cdot \nabla_{\Gamma} g) \eta \, d\mathcal{H}^2 \right| \\ & \leq c\varepsilon \sum_{i=0,1} \left(\varepsilon \|u_i^{\sharp}\|_{L^2(\Gamma)} + \|u_i^{\sharp} - Mu\|_{L^2(\Gamma)} \right) \|\eta\|_{L^2(\Gamma)}. \end{aligned}$$

Moreover, by (5.2.58), (5.3.8), and (5.6.7),

$$\varepsilon \|u_i^{\sharp}\|_{L^2(\Gamma)} + \|u_i^{\sharp} - Mu\|_{L^2(\Gamma)} \le c \left(\varepsilon \|u\|_{L^2(\Gamma_{\varepsilon}^i)} + \|u - \overline{Mu}\|_{L^2(\Gamma_{\varepsilon}^i)}\right) \le c\varepsilon^{1/2} \|u\|_{H^1(\Omega_{\varepsilon})}$$

From the above two estimates we deduce that

$$\left| \int_{\Gamma} \left(\int_{\varepsilon g_0}^{\varepsilon g_1} D(u)^{\sharp} : Q \, dr \right) \eta \, d\mathcal{H}^2 - \varepsilon \int_{\Gamma} (Mu \cdot \nabla_{\Gamma} g) \eta \, d\mathcal{H}^2 \right| \\ \leq c \varepsilon^{3/2} \|u\|_{H^1(\Omega_{\varepsilon})} \|\eta\|_{L^2(\Gamma)}. \quad (5.6.80)$$

Finally, we combine (5.6.77) and (5.6.80) and note that $Mu \cdot \nabla_{\Gamma}g = M_{\tau}u \cdot \nabla_{\Gamma}g$ by the fact that $\nabla_{\Gamma}g$ is tangential on Γ to obtain the inequality (5.6.76).

Lemma 5.6.28. Suppose that Assumption 1 is satisfied, i.e. the inequality (5.1.6) holds. Let $u^{\varepsilon} \in H^2(\Omega_{\varepsilon})^3$ satisfy (5.3.21) on Γ^0_{ε} or on Γ^1_{ε} and $v \in L^2(\Gamma, T\Gamma)$. Then we have

$$\left| \int_{\Omega_{\varepsilon}} D(u) : \bar{v} \otimes \bar{n} \, dx \right| \le c \varepsilon^{3/2} \|u\|_{H^2(\Omega_{\varepsilon})} \|v\|_{L^2(\Gamma)}, \tag{5.6.81}$$

where c > 0 is a constant independent of ε , u, and v.

Proof. Since v is tangential on Γ ,

$$D(u): \bar{v} \otimes \bar{n} = \operatorname{tr}[D(u)^T (\bar{v} \otimes \bar{n})] = (D(u)^T \bar{v}) \cdot \bar{n} = \bar{v} \cdot (D(u)\bar{n}) = \bar{v} \cdot \overline{P}D(u)\bar{n}$$

in Ω_{ε} . Hence by (5.2.53) and (5.3.32) we see that

$$\left|\int_{\Omega_{\varepsilon}} D(u): \bar{v} \otimes \bar{n} \, dx\right| \le c \left\|\overline{P}D(u)\bar{n}\right\|_{L^{2}(\Omega_{\varepsilon})} \|\bar{v}\|_{L^{2}(\Omega_{\varepsilon})} \le c\varepsilon^{3/2} \|u\|_{H^{2}(\Omega_{\varepsilon})} \|v\|_{L^{2}(\Gamma)}.$$

Here we used Assumption 1 and the condition on u to apply (5.3.32).

Now let us derive estimates for trilinear forms. The main tools for the estimates are the product estimate (5.6.52) for functions on Γ and Ω_{ε} and the L^{∞} -estimate (5.6.66) for the residual part of a vector field on Ω_{ε} .

Lemma 5.6.29. Let $u_1 \in H^2(\Omega_{\varepsilon})^3$, $u_2 \in H^1(\Omega_{\varepsilon})^3$, and $A \in L^2(\Gamma)^{3\times 3}$. Under Assumption 1, suppose further that u_1 satisfies div $u_1 = 0$ in Ω_{ε} and (5.3.20)–(5.3.21) on Γ_{ε} and that A satisfies PA = AP = A on Γ . Then

$$\left| \int_{\Omega_{\varepsilon}} u_1 \otimes u_2 : \overline{A} \, dx - \varepsilon \int_{\Gamma} g(M_{\tau} u_1) \otimes (M_{\tau} u_2) : A \, d\mathcal{H}^2 \right| \le cR_{\varepsilon}(u_1, u_2) \|A\|_{L^2(\Gamma)}, \quad (5.6.82)$$

where c > 0 is a constant independent of ε , u_1 , u_2 , and A and

$$R_{\varepsilon}(u_{1}, u_{2}) := \varepsilon \|u_{1}\|_{H^{1}(\Omega_{\varepsilon})} \|u_{2}\|_{H^{1}(\Omega_{\varepsilon})} + \left(\varepsilon \|u_{1}\|_{H^{2}(\Omega_{\varepsilon})} + \varepsilon^{1/2} \|u_{1}\|_{L^{2}(\Omega_{\varepsilon})}^{1/2} \|u_{1}\|_{H^{2}(\Omega_{\varepsilon})}^{1/2} \right) \|u_{2}\|_{L^{2}(\Omega_{\varepsilon})}.$$
 (5.6.83)

Proof. We use the notations (5.6.3) for the tangential and normal components (with respect to Γ) of a vector field on Ω_{ε} . Since PA = AP = A and $P^T = P$ on Γ ,

$$u_1 \otimes u_2 : \overline{A} = \overline{P}(u_1 \otimes u_2)\overline{P} : \overline{A} = u_{1,\tau} \otimes u_{2,\tau} : \overline{A} \quad \text{in} \quad \Omega_{\varepsilon}.$$

Using this equality we decompose the difference

$$\int_{\Omega_{\varepsilon}} u_1 \otimes u_2 : \overline{A} \, dx - \varepsilon \int_{\Gamma} g(M_{\tau} u_1) \otimes (M_{\tau} u_2) : A \, d\mathcal{H}^2 = I_1 + I_2 \tag{5.6.84}$$

into

$$I_{1} := \int_{\Omega_{\varepsilon}} u_{1,\tau} \otimes u_{2,\tau} : \overline{A} \, dx - \int_{\Omega_{\varepsilon}} \left(\overline{M_{\tau} u_{1}} \right) \otimes u_{2,\tau} : \overline{A} \, dx,$$

$$I_{2} := \int_{\Omega_{\varepsilon}} \left(\overline{M_{\tau} u_{1}} \right) \otimes u_{2,\tau} : \overline{A} \, dx - \varepsilon \int_{\Gamma} g(M_{\tau} u_{1}) \otimes (M_{\tau} u_{2}) : A \, d\mathcal{H}^{2}.$$

Let u_1^a be the average part of u_1 given by (5.6.50) and $u_1^r := u_1 - u_1^a$. Since

$$u_{1,\tau} - \overline{M_{\tau}u_1} = \overline{P}u_1 - \overline{P}u_1^a = \overline{P}u_1^r, \quad u_{2,\tau} = \overline{P}u_2 \quad \text{in} \quad \Omega_{\varepsilon}, \quad |P| = 1 \quad \text{on} \quad \Gamma,$$

where the first equality follows from (5.6.50), we have

$$|I_1| = \left| \int_{\Omega_{\varepsilon}} \left(\overline{P} u_1^r \right) \otimes u_{2,\tau} : \overline{A} \, dx \right| \le c \|u_1^r\|_{L^{\infty}(\Omega_{\varepsilon})} \|u_2\|_{L^2(\Omega_{\varepsilon})} \left\| \overline{A} \right\|_{L^2(\Omega_{\varepsilon})}.$$

We apply (5.2.53) and (5.6.66) to the right-hand side to obtain

$$|I_1| \le c \left(\varepsilon \|u_1\|_{H^2(\Omega_{\varepsilon})} + \varepsilon^{1/2} \|u_1\|_{L^2(\Omega_{\varepsilon})}^{1/2} \|u_1\|_{H^2(\Omega_{\varepsilon})}^{1/2} \right) \|u_2\|_{L^2(\Omega_{\varepsilon})} \|A\|_{L^2(\Gamma)}.$$
(5.6.85)

Here we used Assumption 1 and the conditions on u_1 to apply (5.6.66).

Let us estimate I_2 . Noting that $M_{\tau}u_2 = Mu_{2,\tau}$ on Γ , we use the change of variables formula (5.2.51) and the inequalities (5.2.50), (5.2.52), and

$$|M_{\tau}u_1| = |PMu_1| \le |Mu_1| \quad \text{on} \quad \Gamma, \quad |u_{2,\tau}| \le |u_2| \quad \text{in} \quad \Omega_{\varepsilon}$$

to deduce that

$$|I_2| = \left| \int_{\Gamma} (M_{\tau} u_1) \otimes \left(\int_{\varepsilon g_0}^{\varepsilon g_1} u_{2,\tau}^{\sharp} (J-1) \, dr \right) : A \, d\mathcal{H}^2 \right| \le c\varepsilon \int_{\Omega_{\varepsilon}} \left| \overline{M_{\tau} u_1} \right| |u_{2,\tau}| \left| \overline{A} \right| \, dx$$
$$\le c\varepsilon \left\| \left| \overline{M u_1} \right| |u_2| \right\|_{L^2(\Omega_{\varepsilon})} \left\| \overline{A} \right\|_{L^2(\Omega_{\varepsilon})}.$$

Moreover, from (5.6.4), (5.6.22), and (5.6.52) it follows that

$$\begin{aligned} \left\| \left\| \overline{Mu_{1}} \right\| \left\| u_{2} \right\| \right\|_{L^{2}(\Omega_{\varepsilon})} &\leq c \| Mu_{1} \|_{L^{2}(\Gamma)}^{1/2} \| Mu_{1} \|_{H^{1}(\Gamma)}^{1/2} \| u_{2} \|_{L^{2}(\Omega_{\varepsilon})}^{1/2} \| u_{2} \|_{H^{1}(\Omega_{\varepsilon})}^{1/2} \\ &\leq c \varepsilon^{-1/2} \| u_{1} \|_{L^{2}(\Omega_{\varepsilon})}^{1/2} \| u_{1} \|_{H^{1}(\Omega_{\varepsilon})}^{1/2} \| u_{2} \|_{L^{2}(\Omega_{\varepsilon})}^{1/2} \| u_{2} \|_{H^{1}(\Omega_{\varepsilon})}^{1/2}. \end{aligned}$$

We apply this inequality and (5.2.53) to the above estimate for I_2 to get

$$|I_{2}| \leq c\varepsilon ||u_{1}||_{L^{2}(\Omega_{\varepsilon})}^{1/2} ||u_{1}||_{H^{1}(\Omega_{\varepsilon})}^{1/2} ||u_{2}||_{L^{2}(\Omega_{\varepsilon})}^{1/2} ||u_{2}||_{H^{1}(\Omega_{\varepsilon})}^{1/2} ||A||_{L^{2}(\Gamma)}.$$

$$\leq c\varepsilon ||u_{1}||_{H^{1}(\Omega_{\varepsilon})} ||u_{2}||_{H^{1}(\Omega_{\varepsilon})} ||A||_{L^{2}(\Gamma)}.$$
(5.6.86)

By (5.6.84), (5.6.85), and (5.6.86) we conclude that the inequality (5.6.82) is valid with $R_{\varepsilon}(u_1, u_2)$ given by (5.6.83).

Lemma 5.6.30. Let $u_1, u_2 \in H^1(\Omega_{\varepsilon})^3$ and $v \in H^1(\Gamma)^3$. Suppose that u_2 satisfies $u_2 \cdot n_{\varepsilon} = 0$ on Γ_{ε}^0 or on Γ_{ε}^1 . Then there exists a constant c > 0 independent of ε , u_1 , u_2 , and v such that

$$\left| \int_{\Omega_{\varepsilon}} u_1 \otimes u_2 : \bar{v} \otimes \bar{n} \, dx \right| \le c \varepsilon \|u_1\|_{H^1(\Omega_{\varepsilon})} \|u_2\|_{H^1(\Omega_{\varepsilon})} \|v\|_{H^1(\Gamma)}.$$
(5.6.87)

Proof. Since $u_1 \otimes u_2 : \bar{v} \otimes \bar{n} = (u_1 \cdot \bar{v})(u_2 \cdot \bar{n})$, we use (5.3.23) and (5.6.52) to get

$$\begin{split} \left| \int_{\Omega_{\varepsilon}} u_{1} \otimes u_{2} : \bar{v} \otimes \bar{n} \, dx \right| &\leq \| u_{1} \cdot \bar{v} \|_{L^{2}(\Omega_{\varepsilon})} \| u_{2} \cdot \bar{n} \|_{L^{2}(\Omega_{\varepsilon})} \\ &\leq c \varepsilon \| u_{1} \|_{L^{2}(\Omega_{\varepsilon})}^{1/2} \| u_{1} \|_{H^{1}(\Omega_{\varepsilon})}^{1/2} \| v \|_{L^{2}(\Gamma)}^{1/2} \| v \|_{H^{1}(\Gamma)}^{1/2} \| u_{2} \|_{H^{1}(\Omega_{\varepsilon})} \\ &\leq c \varepsilon \| u_{1} \|_{H^{1}(\Omega_{\varepsilon})} \| u_{2} \|_{H^{1}(\Omega_{\varepsilon})} \| v \|_{H^{1}(\Gamma)}^{1/2} . \end{split}$$

Thus, the inequality (5.6.87) holds.

5.7 Estimate for the trilinear term

The purpose of this section is to give an estimate for the trilinear term, which is essential for our proof of the global existence of a strong solution. Throughout this section we impose Assumptions 1 and 2 and let ε_0 be the positive constant given in Lemma 5.5.4.

Lemma 5.7.1. For any $\alpha > 0$ there exist $c_{\alpha}^1, c_{\alpha}^2 > 0$ independent of ε such that

$$\left| \left((u \cdot \nabla) u, A_{\varepsilon} u \right)_{L^{2}(\Omega_{\varepsilon})} \right| \leq \left(\alpha + c_{\alpha}^{1} \varepsilon^{1/2} \|u\|_{H^{1}(\Omega_{\varepsilon})} \right) \|u\|_{H^{2}(\Omega_{\varepsilon})}^{2} + c_{\alpha}^{2} \left(\|u\|_{L^{2}(\Omega_{\varepsilon})}^{2} \|u\|_{H^{1}(\Omega_{\varepsilon})}^{4} + \varepsilon^{-1} \|u\|_{L^{2}(\Omega_{\varepsilon})}^{2} \|u\|_{H^{1}(\Omega_{\varepsilon})}^{2} \right)$$
(5.7.1)

for all $\varepsilon \in (0, \varepsilon_0)$ and $u \in D(A_{\varepsilon})$. (In fact, c_{α}^1 does not depend on α .)

The main tools for the proof of Lemma 5.7.1 are the estimates given in Sections 5.5.3 and 5.6.3. We also use the following inequality for the tangential component (with respect to Γ) of the curl of the average part u^a (for a proof, see Appendix 5.A).

Lemma 5.7.2. For $u \in C^1(\Omega_{\varepsilon})^3$ let $u^a := E_{\varepsilon}M_{\tau}u$ be given by (5.6.50). Then

$$\left|\overline{P}\operatorname{curl} u^{a}\right| \leq c\left(\left|\overline{Mu}\right| + \varepsilon \left|\overline{\nabla_{\Gamma}Mu}\right|\right) \quad in \quad \Omega_{\varepsilon},$$
(5.7.2)

where c > 0 is a constant independent of ε and u.

Proof of Lemma 5.7.1. For $u \in D(A_{\varepsilon})$ let u^a be given by (5.6.50), $u^r := u - u^a$, and $\omega := \operatorname{curl} u$. Since $(u \cdot \nabla)u = \omega \times u + \nabla(|u|^2)/2$ and $(\nabla(|u|^2), A_{\varepsilon}u)_{L^2(\Omega_{\varepsilon})} = 0$ by $A_{\varepsilon}u \in L^2_{\sigma}(\Omega_{\varepsilon})$ and $\nabla(|u|^2) \in L^2_{\sigma}(\Omega_{\varepsilon})^{\perp}$, we have

$$\left((u\cdot\nabla)u, A_{\varepsilon}u\right)_{L^{2}(\Omega_{\varepsilon})} = (\omega \times u, A_{\varepsilon}u)_{L^{2}(\Omega_{\varepsilon})} = I_{1} + I_{2} + I_{3},$$

where

$$I_1 := (\omega \times u^r, A_{\varepsilon} u)_{L^2(\Omega_{\varepsilon})},$$

$$I_2 := (\omega \times u^a, A_{\varepsilon} u + \nu \Delta u)_{L^2(\Omega_{\varepsilon})}, \quad I_3 := (\omega \times u^a, -\nu \Delta u)_{L^2(\Omega_{\varepsilon})}.$$

Let us estimate I_1 , I_2 , and I_3 separately. By (5.5.32) and (5.6.66),

$$\begin{split} I_{1} &| \leq \|u^{r}\|_{L^{\infty}(\Omega_{\varepsilon})} \|\omega\|_{L^{2}(\Omega_{\varepsilon})} \|A_{\varepsilon}u\|_{L^{2}(\Omega_{\varepsilon})} \\ &\leq c \left(\varepsilon^{1/2} \|u\|_{H^{2}(\Omega_{\varepsilon})} + \|u\|_{L^{2}(\Omega_{\varepsilon})}^{1/2} \|u\|_{H^{2}(\Omega_{\varepsilon})}^{1/2} \right) \|u\|_{H^{1}(\Omega_{\varepsilon})} \|u\|_{H^{2}(\Omega_{\varepsilon})} \\ &= c \varepsilon^{1/2} \|u\|_{H^{1}(\Omega_{\varepsilon})} \|u\|_{H^{2}(\Omega_{\varepsilon})}^{2} + c \|u\|_{L^{2}(\Omega_{\varepsilon})}^{1/2} \|u\|_{H^{1}(\Omega_{\varepsilon})} \|u\|_{H^{2}(\Omega_{\varepsilon})}^{3/2}. \end{split}$$

To the second term we apply Young's inequality $ab \leq \alpha a^{4/3} + c_{\alpha} b^4$ to get

$$|I_1| \le \left(\alpha + c\varepsilon^{1/2} \|u\|_{H^1(\Omega_{\varepsilon})}\right) \|u\|_{H^2(\Omega_{\varepsilon})}^2 + c_\alpha \|u\|_{L^2(\Omega_{\varepsilon})}^2 \|u\|_{H^1(\Omega_{\varepsilon})}^4.$$
(5.7.3)

(Note that the constant c in the above inequality does not depend on α .)

Next we deal with I_2 . By (5.6.56) we have

$$\begin{aligned} \|\omega \times u^{a}\|_{L^{2}(\Omega_{\varepsilon})} &\leq c\varepsilon^{-1/2} \|\omega\|_{L^{2}(\Omega_{\varepsilon})}^{1/2} \|\omega\|_{H^{1}(\Omega_{\varepsilon})}^{1/2} \|u\|_{L^{2}(\Omega_{\varepsilon})}^{1/2} \|u\|_{H^{1}(\Omega_{\varepsilon})}^{1/2} \\ &\leq c\varepsilon^{-1/2} \|u\|_{L^{2}(\Omega_{\varepsilon})}^{1/2} \|u\|_{H^{1}(\Omega_{\varepsilon})}^{1/2} \|u\|_{H^{2}(\Omega_{\varepsilon})}^{1/2}. \end{aligned}$$
(5.7.4)

From this inequality, (5.5.30), and (5.5.33) it follows that

$$I_{2}| \leq \|\omega \times u^{a}\|_{L^{2}(\Omega_{\varepsilon})} \|A_{\varepsilon}u + \nu\Delta u\|_{L^{2}(\Omega_{\varepsilon})}$$
$$\leq c\varepsilon^{-1/2} \|u\|_{L^{2}(\Omega_{\varepsilon})}^{1/2} \|u\|_{H^{1}(\Omega_{\varepsilon})}^{2} \|u\|_{H^{2}(\Omega_{\varepsilon})}^{1/2}$$
$$\leq c\varepsilon^{-1/2} \|u\|_{L^{2}(\Omega_{\varepsilon})} \|u\|_{H^{1}(\Omega_{\varepsilon})} \|u\|_{H^{2}(\Omega_{\varepsilon})}.$$

Applying Young's inequality $ab \leq \alpha a^2 + c_{\alpha}b^2$ to the last line we further get

$$|I_2| \le \alpha ||u||^2_{H^2(\Omega_{\varepsilon})} + c_{\alpha} \varepsilon^{-1} ||u||^2_{L^2(\Omega_{\varepsilon})} ||u||^2_{H^1(\Omega_{\varepsilon})}.$$
(5.7.5)

The estimate for I_3 is more complicated. Setting $\Phi := \omega \times u^a$ we have

$$I_3 = -\nu(\Delta u, \Phi)_{L^2(\Omega_{\varepsilon})} = \nu(\operatorname{curl}\omega, \Phi)_{L^2(\Omega_{\varepsilon})}$$

by div u = 0 in Ω_{ε} . Here Φ is in $H^1(\Omega_{\varepsilon})^3$ since $\omega \in H^1(\Omega_{\varepsilon})^3$, $u^a \in H^2(\Omega_{\varepsilon})^3$, and the Sobolev embeddings $H^1(\Omega_{\varepsilon}) \hookrightarrow L^4(\Omega_{\varepsilon})$ and $H^2(\Omega_{\varepsilon}) \hookrightarrow L^{\infty}(\Omega_{\varepsilon})$ hold (see [1]). Hence we can apply the identity (5.5.28) to get

$$I_3 = -\nu(\operatorname{curl} G(u), \Phi)_{L^2(\Omega_{\varepsilon})} + \nu(\omega + G(u), \operatorname{curl} \Phi)_{L^2(\Omega_{\varepsilon})} = J_1 + J_2 + J_3,$$

where G(u) is given in Lemma 5.5.6 and

$$J_1 := -\nu(\operatorname{curl} G(u), \Phi)_{L^2(\Omega_{\varepsilon})},$$

$$J_2 := \nu(G(u), \operatorname{curl} \Phi)_{L^2(\Omega_{\varepsilon})}, \quad J_3 := \nu(\omega, \operatorname{curl} \Phi)_{L^2(\Omega_{\varepsilon})}.$$

From (5.7.4) and (5.5.22) we deduce that

$$|J_1| \le c \|\nabla G(u)\|_{L^2(\Omega_{\varepsilon})} \|\Phi\|_{L^2(\Omega_{\varepsilon})} \le c\varepsilon^{-1/2} \|u\|_{L^2(\Omega_{\varepsilon})}^{1/2} \|u\|_{H^1(\Omega_{\varepsilon})}^2 \|u\|_{H^2(\Omega_{\varepsilon})}^{1/2}.$$

Then using (5.5.33) and Young's inequality $ab \leq \alpha a^2 + c_{\alpha}b^2$ we get

$$|J_{1}| \leq c\varepsilon^{-1/2} ||u||_{L^{2}(\Omega_{\varepsilon})} ||u||_{H^{1}(\Omega_{\varepsilon})} ||u||_{H^{2}(\Omega_{\varepsilon})} \leq \alpha ||u||^{2}_{H^{2}(\Omega_{\varepsilon})} + c_{\alpha}\varepsilon^{-1} ||u||^{2}_{L^{2}(\Omega_{\varepsilon})} ||u||^{2}_{H^{1}(\Omega_{\varepsilon})}.$$
(5.7.6)

Let us estimate J_2 . By (5.5.22),

$$|\operatorname{curl} \Phi| \le c(|\nabla \omega| |u^a| + |\omega| |\nabla u^a|) \le c(|u^a| |\nabla^2 u| + |\nabla u^a| |\nabla u|) \quad \text{in} \quad \Omega_{\varepsilon},$$

and Hölder's inequality we have

$$\begin{aligned} |J_2| &\leq c \int_{\Omega_{\varepsilon}} |u| (|u^a| |\nabla^2 u| + |\nabla u^a| |\nabla u|) \, dx \\ &\leq c \left(\| |u^a| |u| \|_{L^2(\Omega_{\varepsilon})} \|\nabla^2 u\|_{L^2(\Omega_{\varepsilon})} + \| |\nabla u^a| |u| \|_{L^2(\Omega_{\varepsilon})} \|\nabla u\|_{L^2(\Omega_{\varepsilon})} \right). \end{aligned}$$

To the last line we apply (5.6.56) and (5.6.57) to obtain

$$|J_{2}| \leq c\varepsilon^{-1/2} \left(\|u\|_{L^{2}(\Omega_{\varepsilon})} \|u\|_{H^{1}(\Omega_{\varepsilon})} \|u\|_{H^{2}(\Omega_{\varepsilon})} + \|u\|_{L^{2}(\Omega_{\varepsilon})}^{1/2} \|u\|_{H^{1}(\Omega_{\varepsilon})}^{2} \|u\|_{H^{2}(\Omega_{\varepsilon})}^{1/2} \right)$$

$$\leq c\varepsilon^{-1/2} \|u\|_{L^{2}(\Omega_{\varepsilon})} \|u\|_{H^{1}(\Omega_{\varepsilon})} \|u\|_{H^{2}(\Omega_{\varepsilon})},$$

where the second inequality follows from (5.5.33). Hence Young's inequality yields

$$|J_2| \le \alpha ||u||_{H^2(\Omega_{\varepsilon})}^2 + c_{\alpha} \varepsilon^{-1} ||u||_{L^2(\Omega_{\varepsilon})}^2 ||u||_{H^1(\Omega_{\varepsilon})}^2.$$
(5.7.7)

To derive an estimate for $J_3 = \nu(\omega, \operatorname{curl} \Phi)_{L^2(\Omega_{\varepsilon})}$ we observe that

$$\operatorname{curl} \Phi = (u^a \cdot \nabla)\omega - (\omega \cdot \nabla)u^a + (\operatorname{div} u^a)\omega - (\operatorname{div} \omega)u^a.$$

Moreover, $\operatorname{div} \omega = \operatorname{div} \operatorname{curl} u = 0$ and

$$\int_{\Omega_{\varepsilon}} \omega \cdot (u^a \cdot \nabla) \omega \, dx = -\frac{1}{2} \int_{\Omega_{\varepsilon}} (\operatorname{div} u^a) |\omega|^2 \, dx$$

by integration by parts and $u^a \cdot n_{\varepsilon} = 0$ on Γ_{ε} (see Section 5.6.3). Therefore,

$$J_{3} = \nu(\omega, (u^{a} \cdot \nabla)\omega - (\omega \cdot \nabla)u^{a} + (\operatorname{div} u^{a})\omega)_{L^{2}(\Omega_{\varepsilon})}$$

$$= \frac{\nu}{2} (\operatorname{div} u^{a}, |\omega|^{2})_{L^{2}(\Omega_{\varepsilon})} - \nu(\omega, (\omega \cdot \nabla)u^{a})_{L^{2}(\Omega_{\varepsilon})}.$$
 (5.7.8)

Noting that $u^a = E_{\varepsilon} M_{\tau} u$ is given by (5.6.50), we split the first term into

$$(\operatorname{div} u^{a}, |\omega|^{2})_{L^{2}(\Omega_{\varepsilon})} = \int_{\Omega_{\varepsilon}} \frac{1}{\bar{g}} \left(\overline{\operatorname{div}_{\Gamma}(gM_{\tau}u)} \right) |\omega|^{2} dx + \int_{\Omega_{\varepsilon}} \left(\operatorname{div}(E_{\varepsilon}M_{\tau}u) - \frac{1}{\bar{g}} \overline{\operatorname{div}_{\Gamma}(gM_{\tau}u)} \right) |\omega|^{2} dx.$$

Applying (5.2.30), (5.3.45), and Hölder's inequality to the right-hand side we have

$$|(\operatorname{div} u^a, |\omega|^2)_{L^2(\Omega_{\varepsilon})}| \le c(K_1 + \varepsilon K_2 + \varepsilon K_3) \|\omega\|_{L^2(\Omega_{\varepsilon})},$$

where

$$K_{1} := \left\| \left| \overline{\operatorname{div}_{\Gamma}(gM_{\tau}u)} \right| |\omega| \right\|_{L^{2}(\Omega_{\varepsilon})},$$
$$K_{2} := \left\| \left| \overline{M_{\tau}u} \right| |\omega| \right\|_{L^{2}(\Omega_{\varepsilon})}, \quad K_{3} := \left\| \left| \overline{\nabla_{\Gamma}M_{\tau}u} \right| |\omega| \right\|_{L^{2}(\Omega_{\varepsilon})}.$$

To K_1 we apply (5.6.52) and then use (5.6.46) and (5.6.47) to get

$$K_{1} \leq c \|\operatorname{div}_{\Gamma}(gM_{\tau}u)\|_{L^{2}(\Gamma)}^{1/2} \|\operatorname{div}_{\Gamma}(gM_{\tau}u)\|_{H^{1}(\Gamma)}^{1/2} \|\omega\|_{L^{2}(\Omega_{\varepsilon})}^{1/2} \|\omega\|_{H^{1}(\Omega_{\varepsilon})}^{1/2} \\ \leq c\varepsilon^{1/2} \|u\|_{H^{1}(\Omega_{\varepsilon})} \|u\|_{H^{2}(\Omega_{\varepsilon})} \leq c\varepsilon^{1/2} \|u\|_{H^{2}(\Omega_{\varepsilon})}^{2}.$$

Similarly, applying (5.6.52) to K_2 and K_3 and noting that

$$\begin{split} \|M_{\tau}u\|_{H^{k}(\Gamma)} &\leq c \|Mu\|_{H^{k}(\Gamma)} \leq c\varepsilon^{-1/2} \|u\|_{H^{k}(\Omega_{\varepsilon})}, \\ \|\nabla_{\Gamma}M_{\tau}u\|_{H^{k}(\Gamma)} &\leq c \|Mu\|_{H^{k+1}(\Gamma)} \leq c\varepsilon^{-1/2} \|u\|_{H^{k+1}(\Omega_{\varepsilon})} \end{split}$$

for k = 0, 1 (with $H^0 = L^2$) by $M_\tau u = PMu$ on Γ , $P \in C^4(\Gamma)^{3 \times 3}$, (5.6.4), and (5.6.22) we obtain

$$K_{2} \leq c\varepsilon^{-1/2} \|u\|_{L^{2}(\Omega_{\varepsilon})}^{1/2} \|u\|_{H^{1}(\Omega_{\varepsilon})} \|u\|_{H^{2}(\Omega_{\varepsilon})}^{1/2} \leq c\varepsilon^{-1/2} \|u\|_{H^{2}(\Omega_{\varepsilon})}^{2},$$

$$K_{3} \leq c\varepsilon^{-1/2} \|u\|_{H^{1}(\Omega_{\varepsilon})} \|u\|_{H^{2}(\Omega_{\varepsilon})} \leq c\varepsilon^{-1/2} \|u\|_{H^{2}(\Omega_{\varepsilon})}^{2}.$$

From these inequalities and $\|\omega\|_{L^2(\Omega_{\varepsilon})} \leq c \|u\|_{H^1(\Omega_{\varepsilon})}$ we deduce that

$$|(\operatorname{div} u^{a}, |\omega|^{2})_{L^{2}(\Omega_{\varepsilon})}| \leq c(K_{1} + \varepsilon K_{2} + \varepsilon K_{3}) \|\omega\|_{L^{2}(\Omega_{\varepsilon})} \leq c\varepsilon^{1/2} \|u\|_{H^{1}(\Omega_{\varepsilon})} \|u\|_{H^{2}(\Omega_{\varepsilon})}^{2}.$$
 (5.7.9)

Let us estimate $(\omega, (\omega \cdot \nabla)u^a)_{L^2(\Omega_{\varepsilon})}$. Using $\omega = \operatorname{curl} u^r + \operatorname{curl} u^a$ we have

$$(\omega, (\omega \cdot \nabla)u^a)_{L^2(\Omega_{\varepsilon})} = (\omega, (\operatorname{curl} u^r \cdot \nabla)u^a)_{L^2(\Omega_{\varepsilon})} + (\omega, (\operatorname{curl} u^a \cdot \nabla)u^a)_{L^2(\Omega_{\varepsilon})}.$$

The first term on the right-hand side is bounded by

$$|(\omega, (\operatorname{curl} u^r \cdot \nabla) u^a)_{L^2(\Omega_{\varepsilon})}| \le c \|\nabla u^r\|_{L^2(\Omega_{\varepsilon})} \| |\nabla u^a| |\omega| \|_{L^2(\Omega_{\varepsilon})}.$$

To the right-hand side we apply (5.6.60) and

$$\| |\nabla u^{a}| |\omega| \|_{L^{2}(\Omega_{\varepsilon})} \leq c\varepsilon^{-1/2} \|\omega\|_{L^{2}(\Omega_{\varepsilon})}^{1/2} \|\omega\|_{H^{1}(\Omega_{\varepsilon})}^{1/2} \|u\|_{H^{1}(\Omega_{\varepsilon})}^{1/2} \|u\|_{H^{2}(\Omega_{\varepsilon})}^{1/2}$$

$$\leq c\varepsilon^{-1/2} \|u\|_{H^{1}(\Omega_{\varepsilon})} \|u\|_{H^{2}(\Omega_{\varepsilon})}$$
(5.7.10)

by (5.6.57). Then we get

$$\begin{aligned} |(\omega, (\operatorname{curl} u^{r} \cdot \nabla) u^{a})_{L^{2}(\Omega_{\varepsilon})}| \\ &\leq c \left(\varepsilon^{1/2} \|u\|_{H^{1}(\Omega_{\varepsilon})} \|u\|_{H^{2}(\Omega_{\varepsilon})}^{2} + \varepsilon^{-1/2} \|u\|_{L^{2}(\Omega_{\varepsilon})} \|u\|_{H^{1}(\Omega_{\varepsilon})} \|u\|_{H^{2}(\Omega_{\varepsilon})} \right). \end{aligned}$$
(5.7.11)

Also, we decompose $(\omega, (\operatorname{curl} u^a \cdot \nabla) u^a)_{L^2(\Omega_{\varepsilon})}$ into the sum of

$$L_1 := \left(\omega, \left((\overline{P}\operatorname{curl} u^a) \cdot \nabla\right) u^a\right)_{L^2(\Omega_{\varepsilon})}, \quad L_2 := \left(\omega, (\operatorname{curl} u^a \cdot \overline{n}) \partial_n u^a\right)_{L^2(\Omega_{\varepsilon})}.$$

To L_1 we apply (5.7.2) and Hölder's inequality to get

$$|L_1| \le c \int_{\Omega_{\varepsilon}} |\omega| \left(\left| \overline{Mu} \right| + \varepsilon \left| \overline{\nabla_{\Gamma} Mu} \right| \right) |\nabla u^a| \, dx$$

$$\le c \| \left| \nabla u^a \right| |\omega| \|_{L^2(\Omega_{\varepsilon})} \left(\left\| \overline{Mu} \right\|_{L^2(\Omega_{\varepsilon})} + \varepsilon \left\| \overline{\nabla_{\Gamma} Mu} \right\|_{L^2(\Omega_{\varepsilon})} \right).$$

Hence from (5.6.5), (5.6.23), (5.7.10), and $||u||_{H^1(\Omega_{\varepsilon})} \leq ||u||_{H^2(\Omega_{\varepsilon})}$ it follows that

$$\begin{aligned} |L_1| &\leq c\varepsilon^{-1/2} \|u\|_{H^1(\Omega_{\varepsilon})} \|u\|_{H^2(\Omega_{\varepsilon})} \left(\|u\|_{L^2(\Omega_{\varepsilon})} + \varepsilon \|u\|_{H^1(\Omega_{\varepsilon})} \right) \\ &\leq c \left(\varepsilon^{1/2} \|u\|_{H^1(\Omega_{\varepsilon})} \|u\|_{H^2(\Omega_{\varepsilon})}^2 + \varepsilon^{-1/2} \|u\|_{L^2(\Omega_{\varepsilon})} \|u\|_{H^1(\Omega_{\varepsilon})} \|u\|_{H^2(\Omega_{\varepsilon})} \right). \end{aligned}$$

To estimate L_2 we see by the definition (5.6.50) of u^a , (5.3.6), and (5.3.37) that

$$|\partial_n u^a| = \left|\overline{M_\tau u} \cdot \partial_n \Psi_\varepsilon\right| \le c \left|\overline{M_\tau u}\right| = c \left|\overline{PMu}\right| \le c \left|\overline{Mu}\right| \quad \text{in} \quad \Omega_\varepsilon.$$

We apply this inequality and $|\operatorname{curl} u^a \cdot \bar{n}| \leq c |\nabla u^a|$ to L_2 and use (5.6.5) and (5.7.10). Then we have

$$|L_{2}| \leq c \int_{\Omega_{\varepsilon}} |\omega| |\nabla u^{a}| \left| \overline{Mu} \right| \, dx \leq c \| |\nabla u^{a}| \, |\omega| \, \|_{L^{2}(\Omega_{\varepsilon})} \, \left\| \overline{Mu} \right\|_{L^{2}(\Omega_{\varepsilon})}$$
$$\leq c \varepsilon^{-1/2} \|u\|_{L^{2}(\Omega_{\varepsilon})} \|u\|_{H^{1}(\Omega_{\varepsilon})} \|u\|_{H^{2}(\Omega_{\varepsilon})}.$$

Applying the above inequalities to $(\omega, (\operatorname{curl} u^a \cdot \nabla) u^a)_{L^2(\Omega_{\varepsilon})} = L_1 + L_2$ we obtain

 $\begin{aligned} |(\omega, (\operatorname{curl} u^{a} \cdot \nabla) u^{a})_{L^{2}(\Omega_{\varepsilon})}| \\ &\leq c \left(\varepsilon^{1/2} \|u\|_{H^{1}(\Omega_{\varepsilon})} \|u\|_{H^{2}(\Omega_{\varepsilon})}^{2} + \varepsilon^{-1/2} \|u\|_{L^{2}(\Omega_{\varepsilon})} \|u\|_{H^{1}(\Omega_{\varepsilon})} \|u\|_{H^{2}(\Omega_{\varepsilon})} \right). \end{aligned}$

From this inequality and (5.7.11) we deduce that

$$\begin{aligned} |(\omega, (\omega \cdot \nabla)u^{a})_{L^{2}(\Omega_{\varepsilon})}| &\leq |(\omega, (\operatorname{curl} u^{r} \cdot \nabla)u^{a})_{L^{2}(\Omega_{\varepsilon})}| + |(\omega, (\operatorname{curl} u^{a} \cdot \nabla)u^{a})_{L^{2}(\Omega_{\varepsilon})}| \\ &\leq c \left(\varepsilon^{1/2} \|u\|_{H^{1}(\Omega_{\varepsilon})} \|u\|_{H^{2}(\Omega_{\varepsilon})}^{2} + \varepsilon^{-1/2} \|u\|_{L^{2}(\Omega_{\varepsilon})} \|u\|_{H^{1}(\Omega_{\varepsilon})} \|u\|_{H^{2}(\Omega_{\varepsilon})} \right). \end{aligned}$$

Since Young's inequality implies that

 $\varepsilon^{-1/2} \|u\|_{L^2(\Omega_{\varepsilon})} \|u\|_{H^1(\Omega_{\varepsilon})} \|u\|_{H^2(\Omega_{\varepsilon})} \le \alpha \|u\|_{H^2(\Omega_{\varepsilon})}^2 + c_{\alpha} \varepsilon^{-1} \|u\|_{L^2(\Omega_{\varepsilon})}^2 \|u\|_{H^1(\Omega_{\varepsilon})}^2,$ we further get

$$\left| (\omega, (\operatorname{curl} u^{a} \cdot \nabla) u^{a})_{L^{2}(\Omega_{\varepsilon})} \right| \leq \left(\alpha + c \varepsilon^{1/2} \|u\|_{H^{1}(\Omega_{\varepsilon})} \right) \|u\|_{H^{2}(\Omega_{\varepsilon})}^{2} + c_{\alpha} \varepsilon^{-1} \|u\|_{L^{2}(\Omega_{\varepsilon})}^{2} \|u\|_{H^{1}(\Omega_{\varepsilon})}^{2}.$$

We apply this inequality and (5.7.9) to (5.7.8) to show that

$$|J_3| \le c \left(|(\operatorname{div} u^a, |\omega|^2)_{L^2(\Omega_{\varepsilon})}| + |(\omega, (\operatorname{curl} u^a \cdot \nabla) u^a)_{L^2(\Omega_{\varepsilon})}| \right) \\\le c \left(\alpha + \varepsilon^{1/2} \|u\|_{H^1(\Omega_{\varepsilon})} \right) \|u\|_{H^2(\Omega_{\varepsilon})}^2 + c_\alpha \varepsilon^{-1} \|u\|_{L^2(\Omega_{\varepsilon})}^2 \|u\|_{H^1(\Omega_{\varepsilon})}^2.$$

$$(5.7.12)$$

Since $I_3 = J_1 + J_2 + J_3$, we see by (5.7.6), (5.7.7), and (5.7.12) that

$$|I_3| \le c \left(\alpha + \varepsilon^{1/2} \|u\|_{H^1(\Omega_{\varepsilon})}\right) \|u\|_{H^2(\Omega_{\varepsilon})}^2 + c_{\alpha} \varepsilon^{-1} \|u\|_{L^2(\Omega_{\varepsilon})}^2 \|u\|_{H^1(\Omega_{\varepsilon})}^2$$

and this inequality combined with (5.7.3) and (5.7.5) yields

$$\begin{split} \left| \left((u \cdot \nabla) u, A_{\varepsilon} u \right)_{L^{2}(\Omega_{\varepsilon})} \right| &\leq |I_{1}| + |I_{2}| + |I_{3}| \\ &\leq \left(c_{1} \alpha + c_{2} \varepsilon^{1/2} \|u\|_{H^{1}(\Omega_{\varepsilon})} \right) \|u\|_{H^{2}(\Omega_{\varepsilon})}^{2} \\ &+ c_{\alpha} \left(\|u\|_{L^{2}(\Omega_{\varepsilon})}^{2} \|u\|_{H^{1}(\Omega_{\varepsilon})}^{4} + \varepsilon^{-1} \|u\|_{L^{2}(\Omega_{\varepsilon})}^{2} \|u\|_{H^{1}(\Omega_{\varepsilon})}^{2} \right) \end{split}$$

with positive constants c_1 , c_2 , and c_{α} independent of ε . Replacing $c_1 \alpha$ by α in the above inequality we obtain (5.7.1).

Finally, we fix α and write (5.7.1) in terms of the Stokes operator A_{ε} . Corollary 5.7.3. There exist $d_1, d_2 > 0$ independent of ε such that

Coronary 5.1.5. There exist
$$u_1, u_2 > 0$$
 thus period in the line in the line in the line is $u_1, u_2 > 0$ the line in the line is $u_1, u_2 > 0$.

$$\left| \left((u \cdot \nabla) u, A_{\varepsilon} u \right)_{L^{2}(\Omega_{\varepsilon})} \right| \leq \left(\frac{1}{4} + d_{1} \varepsilon^{1/2} \| A_{\varepsilon}^{1/2} u \|_{L^{2}(\Omega_{\varepsilon})} \right) \| A_{\varepsilon} u \|_{L^{2}(\Omega_{\varepsilon})}^{2} + d_{2} \left(\| u \|_{L^{2}(\Omega_{\varepsilon})}^{2} \| A_{\varepsilon}^{1/2} u \|_{L^{2}(\Omega_{\varepsilon})}^{4} + \varepsilon^{-1} \| u \|_{L^{2}(\Omega_{\varepsilon})}^{2} \| A_{\varepsilon}^{1/2} u \|_{L^{2}(\Omega_{\varepsilon})}^{2} \right)$$
(5.7.13)

for all $\varepsilon \in (0, \varepsilon_0)$ and $u \in D(A_{\varepsilon})$.

Proof. Applying (5.5.18) and (5.5.32) to the right-hand side of (5.7.1) we get

$$\begin{aligned} \left| \left((u \cdot \nabla) u, A_{\varepsilon} u \right)_{L^{2}(\Omega_{\varepsilon})} \right| &\leq \left(c\alpha + d_{\alpha}^{1} \varepsilon^{1/2} \| A_{\varepsilon}^{1/2} u \|_{L^{2}(\Omega_{\varepsilon})} \right) \| A_{\varepsilon} u \|_{L^{2}(\Omega_{\varepsilon})}^{2} \\ &+ d_{\alpha}^{2} \left(\| u \|_{L^{2}(\Omega_{\varepsilon})}^{2} \| A_{\varepsilon}^{1/2} u \|_{L^{2}(\Omega_{\varepsilon})}^{4} + \varepsilon^{-1} \| u \|_{L^{2}(\Omega_{\varepsilon})}^{2} \| A_{\varepsilon}^{1/2} u \|_{L^{2}(\Omega_{\varepsilon})}^{2} \right) \end{aligned}$$

with positive constants c, d_{α}^1 , and d_{α}^2 independent of ε . We take $\alpha = 1/4c$ in the above inequality to obtain (5.7.13).

5.8 Global existence and uniform estimates of a strong solution

Based on the estimate (5.7.13) for the trilinear term and the inequalities given in the previous sections we prove Theorems 5.1.1 and 5.1.2. As in Section 5.7 we suppose that Assumptions 1 and 2 are satisfied and let ε_0 be the positive constant given in Lemma 5.5.4.

First we recall the well-known result on the local-in-time existence of a strong solution (see e.g. [8, 10, 61, 64]).

Theorem 5.8.1. For given $u_0^{\varepsilon} \in V_{\varepsilon}$ and $f^{\varepsilon} \in L^{\infty}(0, \infty; L^2(\Omega_{\varepsilon})^3)$ there exist $T_0 > 0$ depending on Ω_{ε} , ν , u_0^{ε} , and f^{ε} and a strong solution u^{ε} to the Navier–Stokes equations (5.1.1)–(5.1.3) on $[0, T_0)$ satisfying

 $u^{\varepsilon} \in C([0,T]; V_{\varepsilon}) \cap L^2(0,T; D(A_{\varepsilon})) \cap H^1(0,T; L^2_{\sigma}(\Omega_{\varepsilon})) \text{ for all } T \in (0,T_0).$

If u^{ε} is maximally defined on the time interval $[0, T_{\max})$ and T_{\max} is finite, then

$$\lim_{t \to T_{\max}^-} \|A_{\varepsilon}^{1/2} u^{\varepsilon}(t)\|_{L^2(\Omega_{\varepsilon})} = \infty.$$

To establish the global-in-time existence of a strong solution u^{ε} we show that the $L^2(\Omega_{\varepsilon})$ norm of $A_{\varepsilon}^{1/2}u^{\varepsilon}(t)$ is bounded uniformly in t. We argue by a standard energy method and use the uniform Gronwall inequality (see [59, Lemma D.3]).

Lemma 5.8.2 (Uniform Gronwall inequality). Let z, ξ , and ζ be nonnegative functions in $L^1_{loc}([0,T);\mathbb{R}), T \in (0,\infty]$. Suppose that $z \in C(0,T;\mathbb{R})$ and

$$\frac{dz}{dt}(t) \le \xi(t)z(t) + \zeta(t) \quad for \ a.a. \quad t \in (0,T).$$

Then $z \in L^{\infty}_{loc}(0,T;\mathbb{R})$ and

$$z(t_2) \le \left(\frac{1}{t_2 - t_1} \int_{t_1}^{t_2} z(s) \, ds + \int_{t_1}^{t_2} \zeta(s) \, ds\right) \exp\left(\int_{t_1}^{t_2} \xi(s) \, ds\right)$$

for all $t_1, t_2 \in (0, T)$ with $t_1 < t_2$.

We also use an estimate for the duality product between a vector field on Ω_{ε} and the constant extension of a tangential vector field on Γ .

Lemma 5.8.3. There exists a constant c > 0 independent of ε such that

$$\left| (\bar{v}, u)_{L^2(\Omega_{\varepsilon})} \right| \le c \varepsilon^{1/2} \|v\|_{H^{-1}(\Gamma, T\Gamma)} \|u\|_{H^1(\Omega_{\varepsilon})}$$

$$(5.8.1)$$

for all $v \in L^2(\Gamma, T\Gamma)$ and $u \in H^1(\Omega_{\varepsilon})^3$, where $\bar{v} := v \circ \pi$ is the constant extension of v in the normal direction of Γ .

Proof. For the sake of simplicity, we use the notations (5.2.46) and (5.2.47) and suppress the arguments of functions. By the change of variables formula (5.2.51),

$$(\bar{v}, u)_{L^2(\Omega_{\varepsilon})} = \int_{\Gamma} v \left(\int_{\varepsilon g_0}^{\varepsilon g_1} u^{\sharp} J \, dr \right) d\mathcal{H}^2 = (v, \eta)_{L^2(\Gamma)}, \tag{5.8.2}$$

where the vector field $\eta \colon \Gamma \to \mathbb{R}^3$ is given by

$$\eta(y) := \int_{\varepsilon g_0(y)}^{\varepsilon g_1(y)} u^{\sharp}(y, r) J(y, r) \, dr, \quad y \in \Gamma.$$

Let us estimate the $H^1(\Gamma)$ -norm of η . By (5.2.49), Hölder's inequality, and (5.2.52),

$$\|\eta\|_{L^{2}(\Gamma)}^{2} \leq \int_{\Gamma} \varepsilon g\left(\int_{\varepsilon g_{0}}^{\varepsilon g_{1}} |u^{\sharp}|^{2} dr\right) d\mathcal{H}^{2} \leq c\varepsilon \|u\|_{L^{2}(\Omega_{\varepsilon})}^{2}.$$
(5.8.3)

Moreover, by the same calculations as in the proof of Lemma 5.6.7 we see that

$$\nabla_{\Gamma}\eta = \int_{\varepsilon g_0}^{\varepsilon g_1} \left\{ \frac{\partial}{\partial r} \left(J\psi_{\varepsilon}^{\sharp} \otimes u^{\sharp} \right) + J(B\nabla u)^{\sharp} + \nabla_{\Gamma}J \otimes u^{\sharp} \right\} dr$$

on Γ , where B and ψ_{ε} are given by (5.6.18). To the right-hand side we apply the inequalities (5.2.49), (5.6.20), and (5.6.21) to obtain

$$|\nabla_{\Gamma}\eta| \le c \int_{\varepsilon g_0}^{\varepsilon g_1} (|u^{\sharp}| + |(\nabla u)^{\sharp}|) dr$$
 on Γ .

Hence Hölder's inequality and (5.2.52) imply that

$$\|\nabla_{\Gamma}\eta\|_{L^{2}(\Gamma)}^{2} \leq c \int_{\Gamma} \varepsilon g\left(\int_{\varepsilon g_{0}}^{\varepsilon g_{1}} (|u^{\sharp}|^{2} + |(\nabla u)^{\sharp}|^{2}) dr\right) d\mathcal{H}^{2} \leq c\varepsilon \|u\|_{H^{1}(\Omega_{\varepsilon})}^{2}.$$
(5.8.4)

By (5.8.3) and (5.8.4) we get $\eta \in H^1(\Gamma)^3$ and thus $P\eta \in H^1(\Gamma, T\Gamma)$. Moreover,

$$\|P\eta\|_{H^{1}(\Gamma)} \le c\|\eta\|_{H^{1}(\Gamma)} \le c\varepsilon^{1/2}\|u\|_{H^{1}(\Omega_{\varepsilon})}.$$
(5.8.5)

Now we observe that

$$(\bar{v}, u)_{L^2(\Omega_{\varepsilon})} = (v, \eta)_{L^2(\Gamma)} = (v, P\eta)_{L^2(\Gamma)} = [v, P\eta]_{T\Gamma}$$

by (5.8.2), the fact that v is tangential on Γ , and $P\eta \in H^1(\Gamma, T\Gamma)$. Hence

$$\left| (\bar{v}, u)_{L^2(\Omega_{\varepsilon})} \right| = \left| [v, P\eta]_{T\Gamma} \right| \le \|v\|_{H^{-1}(\Gamma, T\Gamma)} \|P\eta\|_{H^1(\Gamma)}$$

and we obtain (5.8.1) by applying (5.8.5) to the right-hand side of this inequality.

Now we are ready to establish the global-in-time existence of a strong solution to the Navier–Stokes equations (5.1.1)–(5.1.3).

Proof of Theorem 5.1.1. We follow the idea of the proofs of [20, Theorem 7.4] and [21, Theorem 3.1]. In what follows, we use the notation (5.6.3) for the tangential and normal components (with respect to Γ) of a vector field on Ω_{ε} . We also write c for a general positive constant independent of ε , c_0 , and T_{max} .

Let $u_0^{\varepsilon} \in V_{\varepsilon}$ and $f \in L^{\infty}(0, \infty; L^2(\Omega_{\varepsilon})^3)$ satisfy (5.1.9) with

$$c_0 := \min\left\{1, \frac{d_3^2}{4}, \frac{d_3^2}{4d_4}\right\}, \quad d_3 := \frac{1}{4d_1}, \tag{5.8.6}$$

where d_1 is the positive constant given in Corollary 5.7.3 and d_4 is a positive constant given later. Noting that $M_{\tau}u_0^{\varepsilon} = Mu_{0,\tau}^{\varepsilon}$ on Γ and

$$u_0^{\varepsilon} = u_{0,n}^{\varepsilon} + \left(u_{0,\tau}^{\varepsilon} - \overline{M} u_{0,\tau}^{\varepsilon} \right) + \overline{M_{\tau} u_0^{\varepsilon}}, \quad |u_{0,n}^{\varepsilon}| = |u_0^{\varepsilon} \cdot \overline{n}| \quad \text{in} \quad \Omega_{\varepsilon},$$

we apply (5.2.53), (5.3.23), and (5.6.6) to the right-hand side of

$$\|u_0^{\varepsilon}\|_{L^2(\Omega_{\varepsilon})}^2 \le c \left(\|u_0^{\varepsilon} \cdot \bar{n}\|_{L^2(\Omega_{\varepsilon})}^2 + \left\|u_{0,\tau}^{\varepsilon} - \overline{M}u_{0,\tau}^{\varepsilon}\right\|_{L^2(\Omega_{\varepsilon})}^2 + \left\|\overline{M}_{\tau}u_0^{\varepsilon}\right\|_{L^2(\Omega_{\varepsilon})}^2 \right)$$

and then use $\|u_{0,\tau}^{\varepsilon}\|_{H^1(\Omega_{\varepsilon})} \leq c \|u_0^{\varepsilon}\|_{H^1(\Omega_{\varepsilon})}$ and (5.5.18) to get

$$\|u_0^{\varepsilon}\|_{L^2(\Omega_{\varepsilon})}^2 \le c \left(\varepsilon^2 \|A_{\varepsilon}^{1/2} u_0^{\varepsilon}\|_{L^2(\Omega_{\varepsilon})}^2 + \varepsilon \|M_{\tau} u_0^{\varepsilon}\|_{L^2(\Gamma)}^2\right).$$
(5.8.7)

Hence from (5.1.9) and $c_0 \leq 1$ it follows that

$$\|u_0^{\varepsilon}\|_{L^2(\Omega_{\varepsilon})}^2 \le cc_0 \le c. \tag{5.8.8}$$

Let u^{ε} be a strong solution to the Navier–Stokes equations (5.1.1)–(5.1.3) defined on the maximal time interval $[0, T_{\text{max}})$. First we derive estimates for

$$\|u^{\varepsilon}(t)\|_{L^{2}(\Omega_{\varepsilon})}^{2}, \quad \int_{t}^{\min\{t+1, T_{\max}\}} \|A_{\varepsilon}^{1/2}u^{\varepsilon}(s)\|_{L^{2}(\Omega_{\varepsilon})}^{2} ds, \quad t \in [0, T_{\max})$$

with explicit dependence of constants on ε . Taking the $L^2(\Omega_{\varepsilon})$ -inner product of

$$\partial_t u^{\varepsilon} + A_{\varepsilon} u^{\varepsilon} + \mathbb{P}_{\varepsilon} (u^{\varepsilon} \cdot \nabla) u^{\varepsilon} = \mathbb{P}_{\varepsilon} f^{\varepsilon} \quad \text{on} \quad (0, T_{\max})$$
(5.8.9)

with u^{ε} and using

$$\left(\mathbb{P}_{\varepsilon}(u^{\varepsilon}\cdot\nabla)u^{\varepsilon},u^{\varepsilon}\right)_{L^{2}(\Omega_{\varepsilon})}=\left((u^{\varepsilon}\cdot\nabla)u^{\varepsilon},u^{\varepsilon}\right)_{L^{2}(\Omega_{\varepsilon})}=0$$

by integration by parts, div $u^{\varepsilon} = 0$ in Ω_{ε} , and $u^{\varepsilon} \cdot n_{\varepsilon} = 0$ on Γ_{ε} we get

$$\frac{1}{2}\frac{d}{dt}\|u^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon})}^{2}+\|A_{\varepsilon}^{1/2}u^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon})}^{2}=(\mathbb{P}_{\varepsilon}f^{\varepsilon},u^{\varepsilon})_{L^{2}(\Omega_{\varepsilon})}\quad\text{on}\quad(0,T_{\max}).$$
(5.8.10)

We split the right-hand side into the sum of

$$I_1 := \left(\mathbb{P}_{\varepsilon} f^{\varepsilon}, u^{\varepsilon} - \overline{M_{\tau} u^{\varepsilon}} \right)_{L^2(\Omega_{\varepsilon})}, \quad I_2 := \left(\mathbb{P}_{\varepsilon} f^{\varepsilon}, \overline{M_{\tau} u^{\varepsilon}} \right)_{L^2(\Omega_{\varepsilon})}.$$

Applying (5.3.23) and (5.6.6) to $u^{\varepsilon} - \overline{M_{\tau}u^{\varepsilon}} = u_n^{\varepsilon} + (u_{\tau}^{\varepsilon} - \overline{Mu_{\tau}^{\varepsilon}})$ we get

$$\left\| u^{\varepsilon} - \overline{M_{\tau} u^{\varepsilon}} \right\|_{L^{2}(\Omega_{\varepsilon})} \le \left\| u^{\varepsilon} \cdot \overline{n} \right\|_{L^{2}(\Omega_{\varepsilon})} + \left\| u^{\varepsilon}_{\tau} - \overline{M u^{\varepsilon}_{\tau}} \right\|_{L^{2}(\Omega_{\varepsilon})} \le c\varepsilon \| u^{\varepsilon} \|_{H^{1}(\Omega_{\varepsilon})}$$

(Note that $\|u_{\tau}^{\varepsilon}\|_{H^{1}(\Omega_{\varepsilon})} \leq c \|u^{\varepsilon}\|_{H^{1}(\Omega_{\varepsilon})}$.) This inequality and (5.5.18) show that

$$|I_1| \le \|\mathbb{P}_{\varepsilon} f^{\varepsilon}\|_{L^2(\Omega_{\varepsilon})} \|u^{\varepsilon} - \overline{M_{\tau} u^{\varepsilon}}\|_{L^2(\Omega_{\varepsilon})} \le c\varepsilon \|\mathbb{P}_{\varepsilon} f^{\varepsilon}\|_{L^2(\Omega_{\varepsilon})} \|A_{\varepsilon}^{1/2} u^{\varepsilon}\|_{L^2(\Omega_{\varepsilon})}.$$

To I_2 we use (5.6.11) and (5.8.1) to get

$$\begin{aligned} |I_2| &\leq \left| \left(\overline{M_\tau \mathbb{P}_\varepsilon f^\varepsilon}, u^\varepsilon \right)_{L^2(\Omega_\varepsilon)} \right| + \left| \left(\mathbb{P}_\varepsilon f^\varepsilon, \overline{M_\tau u^\varepsilon} \right)_{L^2(\Omega_\varepsilon)} - \left(\overline{M_\tau \mathbb{P}_\varepsilon f^\varepsilon}, u^\varepsilon \right)_{L^2(\Omega_\varepsilon)} \right| \\ &\leq c \left(\varepsilon^{1/2} \| M_\tau \mathbb{P}_\varepsilon f^\varepsilon \|_{H^{-1}(\Gamma, T\Gamma)} \| u^\varepsilon \|_{H^1(\Omega_\varepsilon)} + \varepsilon \| \mathbb{P}_\varepsilon f^\varepsilon \|_{L^2(\Omega_\varepsilon)} \| u^\varepsilon \|_{L^2(\Omega_\varepsilon)} \right). \end{aligned}$$

Applying these estimates to $(\mathbb{P}_{\varepsilon}f^{\varepsilon}, u^{\varepsilon})_{L^{2}(\Omega_{\varepsilon})} = I_{1} + I_{2}$ and using the inequality

$$\|u^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon})} \leq \|u^{\varepsilon}\|_{H^{1}(\Omega_{\varepsilon})} \leq c\|A_{\varepsilon}^{1/2}u^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon})}$$
(5.8.11)

by (5.5.18) and Young's inequality we obtain

$$|(\mathbb{P}_{\varepsilon}f^{\varepsilon}, u^{\varepsilon})_{L^{2}(\Omega_{\varepsilon})}| \leq \frac{1}{2} ||A_{\varepsilon}^{1/2}u^{\varepsilon}||_{L^{2}(\Omega_{\varepsilon})}^{2} + c\left(\varepsilon^{2} ||\mathbb{P}_{\varepsilon}f^{\varepsilon}||_{L^{2}(\Omega_{\varepsilon})}^{2} + \varepsilon ||M_{\tau}\mathbb{P}_{\varepsilon}f^{\varepsilon}||_{H^{-1}(\Gamma, T\Gamma)}^{2}\right).$$

From this inequality and (5.8.10) we deduce that

$$\frac{d}{dt} \| u^{\varepsilon} \|_{L^{2}(\Omega_{\varepsilon})}^{2} + \| A_{\varepsilon}^{1/2} u^{\varepsilon} \|_{L^{2}(\Omega_{\varepsilon})}^{2} \le c \left(\varepsilon^{2} \| \mathbb{P}_{\varepsilon} f^{\varepsilon} \|_{L^{2}(\Omega_{\varepsilon})}^{2} + \varepsilon \| M_{\tau} \mathbb{P}_{\varepsilon} f^{\varepsilon} \|_{H^{-1}(\Gamma, T\Gamma)}^{2} \right)$$
(5.8.12)

on $(0, T_{\text{max}})$. By (5.8.11) we further get

$$\frac{d}{dt} \|u^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon})}^{2} + \frac{1}{a_{1}} \|u^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon})}^{2} \leq c \left(\varepsilon^{2} \|\mathbb{P}_{\varepsilon}f^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon})}^{2} + \varepsilon \|M_{\tau}\mathbb{P}_{\varepsilon}f^{\varepsilon}\|_{H^{-1}(\Gamma,T\Gamma)}^{2}\right)$$
(5.8.13)

on $(0, T_{\text{max}})$ with some constant $a_1 > 0$ independent of ε , c_0 , and T_{max} . For each $t \in (0, T_{\text{max}})$ we multiply both sides of (5.8.13) at $s \in (0, t)$ by $e^{(s-t)/a_1}$ and integrate them over (0, t). Then we have

$$\begin{aligned} \|u^{\varepsilon}(t)\|_{L^{2}(\Omega_{\varepsilon})}^{2} &\leq e^{-t/a_{1}} \|u_{0}^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon})}^{2} \\ &+ ca_{1}(1 - e^{-t/a_{1}}) \left(\varepsilon^{2} \|\mathbb{P}_{\varepsilon}f^{\varepsilon}\|_{L^{\infty}(0,\infty;L^{2}(\Omega_{\varepsilon}))}^{2} + \varepsilon \|M_{\tau}\mathbb{P}_{\varepsilon}f^{\varepsilon}\|_{L^{\infty}(0,\infty;H^{-1}(\Gamma,T\Gamma))}^{2}\right). \end{aligned}$$
(5.8.14)

Also, integrating (5.8.12) over (t, t_*) with $t_* := \min\{t+1, T_{\max}\}$ we deduce that

$$\int_{t}^{t_{*}} \|A_{\varepsilon}^{1/2} u^{\varepsilon}(s)\|_{L^{2}(\Omega_{\varepsilon})}^{2} ds \leq \|u_{0}^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon})}^{2} \\
+ c \left(\varepsilon^{2} \|\mathbb{P}_{\varepsilon}f^{\varepsilon}\|_{L^{\infty}(0,\infty;L^{2}(\Omega_{\varepsilon}))}^{2} + \varepsilon \|M_{\tau}\mathbb{P}_{\varepsilon}f^{\varepsilon}\|_{L^{\infty}(0,\infty;H^{-1}(\Gamma,T\Gamma))}^{2}\right). \quad (5.8.15)$$

Hence we apply (5.1.9) and (5.8.8) to the right-hand sides of (5.8.14) and (5.8.15) to get

$$\|u^{\varepsilon}(t)\|_{L^{2}(\Omega_{\varepsilon})}^{2} + \int_{t}^{t_{*}} \|A_{\varepsilon}^{1/2}u^{\varepsilon}(s)\|_{L^{2}(\Omega_{\varepsilon})}^{2} ds \le cc_{0} \quad \text{for all} \quad t \in [0, T_{\max}).$$
(5.8.16)

Next we prove the uniform boundedness of $||A_{\varepsilon}^{1/2}u^{\varepsilon}(t)||_{L^{2}(\Omega_{\varepsilon})}$ in $t \in [0, T_{\max})$. Our goal is to show that

$$\varepsilon^{1/2} \|A_{\varepsilon}^{1/2} u^{\varepsilon}(t)\|_{L^2(\Omega_{\varepsilon})} < d_3 = \frac{1}{4d_1} \quad \text{for all} \quad t \in [0, T_{\max}).$$
 (5.8.17)

If (5.8.17) is valid, then Theorem 5.8.1 implies that $T_{\max} = \infty$, i.e. the strong solution u^{ε} exists on the whole time interval $[0, \infty)$. First note that (5.8.17) is valid at t = 0 by (5.1.9) and (5.8.6). Let us prove (5.8.17) for all $t \in (0, T_{\max})$ by contradiction. Assume to the contrary that there exists $T \in (0, T_{\max})$ such that

$$\varepsilon^{1/2} \|A_{\varepsilon}^{1/2} u^{\varepsilon}(t)\|_{L^2(\Omega_{\varepsilon})} < d_3 \quad \text{for all} \quad t \in [0, T),$$
(5.8.18)

$$\varepsilon^{1/2} \|A_{\varepsilon}^{1/2} u^{\varepsilon}(T)\|_{L^2(\Omega_{\varepsilon})} = d_3.$$
(5.8.19)

(Note that $u^{\varepsilon} \in C([0, T_{\max}); V_{\varepsilon})$ and thus $\|A_{\varepsilon}^{1/2} u^{\varepsilon}(t)\|_{L^{2}(\Omega_{\varepsilon})}$ is continuous in t.) We consider (5.8.9) on (0, T] and take its $L^{2}(\Omega_{\varepsilon})$ -inner product with $A_{\varepsilon} u^{\varepsilon}$ to get

$$\frac{1}{2} \frac{d}{dt} \|A_{\varepsilon}^{1/2} u^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon})}^{2} + \|A_{\varepsilon} u^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon})}^{2} \\
\leq \left| \left((u^{\varepsilon} \cdot \nabla) u^{\varepsilon}, A_{\varepsilon} u^{\varepsilon} \right)_{L^{2}(\Omega_{\varepsilon})} \right| + |(\mathbb{P}_{\varepsilon} f^{\varepsilon}, A_{\varepsilon} u^{\varepsilon})_{L^{2}(\Omega_{\varepsilon})}|. \quad (5.8.20)$$

on (0, T]. To the first term on the right-hand side we apply (5.7.13) and (5.8.18)–(5.8.19). Then noting that $d_3 = 1/4d_1$ we see that

$$\begin{split} \left| \left((u^{\varepsilon} \cdot \nabla) u^{\varepsilon}, A_{\varepsilon} u^{\varepsilon} \right)_{L^{2}(\Omega_{\varepsilon})} \right| &\leq \frac{1}{2} \| A_{\varepsilon} u^{\varepsilon} \|_{L^{2}(\Omega_{\varepsilon})}^{2} \\ &+ d_{2} \left(\| u^{\varepsilon} \|_{L^{2}(\Omega_{\varepsilon})}^{2} \| A_{\varepsilon}^{1/2} u^{\varepsilon} \|_{L^{2}(\Omega_{\varepsilon})}^{4} + \varepsilon^{-1} \| u^{\varepsilon} \|_{L^{2}(\Omega_{\varepsilon})}^{2} \| A_{\varepsilon}^{1/2} u^{\varepsilon} \|_{L^{2}(\Omega_{\varepsilon})}^{2} \right) \end{split}$$

on (0,T]. Also, Young's inequality yields that

$$|(\mathbb{P}_{\varepsilon}f^{\varepsilon}, A_{\varepsilon}u^{\varepsilon})_{L^{2}(\Omega_{\varepsilon})}| \leq \frac{1}{4} ||A_{\varepsilon}u^{\varepsilon}||^{2}_{L^{2}(\Omega_{\varepsilon})} + ||\mathbb{P}_{\varepsilon}f^{\varepsilon}||^{2}_{L^{2}(\Omega_{\varepsilon})}$$

Applying these inequalities to the right-hand side of (5.8.20) we obtain

$$\frac{d}{dt} \|A_{\varepsilon}^{1/2} u^{\varepsilon}\|_{L^2(\Omega_{\varepsilon})}^2 + \frac{1}{2} \|A_{\varepsilon} u^{\varepsilon}\|_{L^2(\Omega_{\varepsilon})}^2 \le \xi \|A_{\varepsilon}^{1/2} u^{\varepsilon}\|_{L^2(\Omega_{\varepsilon})}^2 + \zeta \quad \text{on} \quad (0,T],$$
(5.8.21)

where the functions ξ and ζ are given by

$$\begin{aligned} \xi(t) &:= 2d_2 \|u^{\varepsilon}(t)\|_{L^2(\Omega_{\varepsilon})}^2 \|A_{\varepsilon}^{1/2} u^{\varepsilon}(t)\|_{L^2(\Omega_{\varepsilon})}^2, \\ \zeta(t) &:= 2\left(d_2 \varepsilon^{-1} \|u^{\varepsilon}(t)\|_{L^2(\Omega_{\varepsilon})}^2 \|A_{\varepsilon}^{1/2} u^{\varepsilon}(t)\|_{L^2(\Omega_{\varepsilon})}^2 + \|\mathbb{P}_{\varepsilon} f^{\varepsilon}(t)\|_{L^2(\Omega_{\varepsilon})}^2\right) \end{aligned}$$
(5.8.22)

for $t \in (0, T]$. By (5.8.16), (5.8.18), and (5.8.19) we see that

$$\xi \le cc_0 \varepsilon^{-1}, \quad \zeta \le c \left(c_0 \varepsilon^{-1} \| A_{\varepsilon}^{1/2} u^{\varepsilon} \|_{L^2(\Omega_{\varepsilon})}^2 + \| \mathbb{P}_{\varepsilon} f^{\varepsilon} \|_{L^2(\Omega_{\varepsilon})}^2 \right) \quad \text{on} \quad (0, T].$$

Applying these inequalities to (5.8.21) we have

$$\frac{d}{dt} \|A_{\varepsilon}^{1/2} u^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon})}^{2} + \frac{1}{2} \|A_{\varepsilon} u^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon})}^{2} \le c \left(c_{0} \varepsilon^{-1} \|A_{\varepsilon}^{1/2} u^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon})}^{2} + \|\mathbb{P}_{\varepsilon} f^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon})}^{2}\right)$$
(5.8.23)

on (0,T]. From (5.5.19) and (5.8.23) we further deduce that

$$\frac{d}{dt} \|A_{\varepsilon}^{1/2} u^{\varepsilon}\|_{L^2(\Omega_{\varepsilon})}^2 + \frac{1}{a_2} \|A_{\varepsilon}^{1/2} u^{\varepsilon}\|_{L^2(\Omega_{\varepsilon})}^2 \le c \left(c_0 \varepsilon^{-1} \|A_{\varepsilon}^{1/2} u^{\varepsilon}\|_{L^2(\Omega_{\varepsilon})}^2 + \|\mathbb{P}_{\varepsilon} f^{\varepsilon}\|_{L^2(\Omega_{\varepsilon})}^2\right)$$

with a constant $a_2 > 0$ independent of ε , c_0 , and T_{\max} . For $t \in (0, T]$ we multiply both sides of the above inequality at $s \in (0, t)$ by $e^{(s-t)/a_2}$ and integrate them over (0, t) to get

$$\begin{aligned} \|A_{\varepsilon}^{1/2}u^{\varepsilon}(t)\|_{L^{2}(\Omega_{\varepsilon})}^{2} &\leq e^{-t/a_{2}}\|A_{\varepsilon}^{1/2}u_{0}^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon})}^{2} + cc_{0}\varepsilon^{-1}\int_{0}^{t}e^{(s-t)/a_{2}}\|A_{\varepsilon}^{1/2}u^{\varepsilon}(s)\|_{L^{2}(\Omega_{\varepsilon})}^{2}\,ds \\ &+ ca_{2}(1-e^{-t/a_{2}})\|\mathbb{P}_{\varepsilon}f^{\varepsilon}\|_{L^{\infty}(0,\infty;L^{2}(\Omega_{\varepsilon}))}^{2}. \end{aligned}$$
(5.8.24)

When $t \leq T_* := \min\{1, T\}$, we apply (5.1.9) and (5.8.16) to the right-hand side of (5.8.24) and use $c_0 \leq 1$ by (5.8.6) to deduce that

$$\|A_{\varepsilon}^{1/2}u^{\varepsilon}(t)\|_{L^{2}(\Omega_{\varepsilon})}^{2} \leq cc_{0}(1+c_{0})\varepsilon^{-1} \leq cc_{0}\varepsilon^{-1} \quad \text{for all} \quad t \in (0, T_{*}].$$

$$(5.8.25)$$

Now we suppose that $T \ge 1$ and estimate $||A_{\varepsilon}^{1/2}u^{\varepsilon}(t)||_{L^{2}(\Omega_{\varepsilon})}$ for $t \in [1, T]$. Since

$$\frac{d}{dt} \|A_{\varepsilon}^{1/2} u^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon})}^{2} \leq \xi \|A_{\varepsilon}^{1/2} u^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon})}^{2} + \zeta \quad \text{on} \quad (0,T]$$

by (5.8.21), we can use Lemma 5.8.2 with $z(t) = \|A_{\varepsilon}^{1/2} u^{\varepsilon}(t)\|_{L^{2}(\Omega_{\varepsilon})}^{2}$ to obtain

$$\|A_{\varepsilon}^{1/2}u^{\varepsilon}(t)\|_{L^{2}(\Omega_{\varepsilon})}^{2} \leq \left(\int_{t-1}^{t} \|A_{\varepsilon}^{1/2}u^{\varepsilon}(s)\|_{L^{2}(\Omega_{\varepsilon})}^{2} \, ds + \int_{t-1}^{t} \zeta(s) \, ds\right) \exp\left(\int_{t-1}^{t} \xi(s) \, ds\right) \tag{5.8.26}$$

for all $t \in [1, T]$. Moreover, the functions ξ and ζ given by (5.8.22) satisfy

$$\int_{t-1}^{t} \xi(s) \, ds \le cc_0 \int_{t-1}^{t} \|A_{\varepsilon}^{1/2} u^{\varepsilon}(s)\|_{L^2(\Omega_{\varepsilon})}^2 \, ds \le c,$$
$$\int_{t-1}^{t} \zeta(s) \, ds \le c \left(c_0 \varepsilon^{-1} \int_{t-1}^{t} \|A_{\varepsilon}^{1/2} u^{\varepsilon}(s)\|_{L^2(\Omega_{\varepsilon})}^2 \, ds + \|\mathbb{P}_{\varepsilon} f^{\varepsilon}\|_{L^{\infty}(0,\infty;L^2(\Omega_{\varepsilon}))}^2\right) \le cc_0 \varepsilon^{-1}$$

by (5.1.9), (5.8.16), and $c_0 \leq 1$. Using these inequalities and (5.8.16) to (5.8.26) we have

$$\|A_{\varepsilon}^{1/2}u^{\varepsilon}(t)\|_{L^{2}(\Omega_{\varepsilon})}^{2} \leq cc_{0}\varepsilon^{-1} \quad \text{for all} \quad t \in [1, T].$$
(5.8.27)

Now we combine (5.8.25) and (5.8.27) to observe that

$$\|A_{\varepsilon}^{1/2}u^{\varepsilon}(t)\|_{L^{2}(\Omega_{\varepsilon})}^{2} \leq d_{4}c_{0}\varepsilon^{-1} \quad \text{for all} \quad t \in (0,T]$$

with some constant $d_4 > 0$ independent of ε , c_0 , and T. Hence if we define c_0 by (5.8.6), then by setting t = T in the above inequality we get

$$\|A_{\varepsilon}^{1/2}u^{\varepsilon}(T)\|_{L^{2}(\Omega_{\varepsilon})}^{2} \leq \frac{d_{3}^{2}\varepsilon^{-1}}{4}, \quad \text{i.e.} \quad \varepsilon^{1/2}\|A_{\varepsilon}^{1/2}u^{\varepsilon}(T)\|_{L^{2}(\Omega_{\varepsilon})} \leq \frac{d_{3}}{2} < d_{3},$$

which contradicts with (5.8.19). Therefore, the inequality (5.8.17) is valid and we conclude by Theorem 5.8.1 that $T_{\text{max}} = \infty$, i.e. the strong solution u^{ε} to (5.1.1)–(5.1.3) exists on the whole time interval $[0, \infty)$.

Using the inequalities given in the proof of Theorem 5.1.1, we can also show the uniform estimates (5.1.11) and (5.1.12) for a strong solution.

Proof of Theorem 5.1.2. Let $\varepsilon_0, c_0 \in (0, 1)$ be the constants given in Theorem 5.1.1. Since α and β are positive, we can take $\varepsilon_1 \in (0, \varepsilon_0)$ such that

$$c_1 \varepsilon^{\alpha} \le c_0, \quad c_2 \varepsilon^{\beta} \le c_0 \quad \text{for all} \quad \varepsilon \in (0, \varepsilon_1).$$

Hence, for $\varepsilon \in (0, \varepsilon_1)$ if the initial velocity u_0^{ε} and the external force f^{ε} satisfy (5.1.10), then they also satisfy (5.1.9) and Theorem 5.1.1 implies that there exists a global strong solution u^{ε} to (5.1.1)–(5.1.3). Let us derive the estimates (5.1.11) and (5.1.12) for the strong solution u^{ε} . Hereafter we denote by c a general positive constant independent of ε . First note that

$$\|u_0^{\varepsilon}\|_{L^2(\Omega_{\varepsilon})}^2 \le c(\varepsilon^{1+\alpha} + \varepsilon^{\beta}) \tag{5.8.28}$$

by (5.1.10) and (5.8.7). We apply this inequality and (5.1.10) to (5.8.14) to get

$$\|u^{\varepsilon}(t)\|_{L^{2}(\Omega_{\varepsilon})}^{2} \leq c(\varepsilon^{1+\alpha} + \varepsilon^{\beta}) \quad \text{for all} \quad t \geq 0.$$
(5.8.29)

Also, integrating (5.8.12) over [0, t] and using (5.1.10) and (5.8.28) we have

$$\int_0^t \|A_{\varepsilon}^{1/2} u^{\varepsilon}(s)\|_{L^2(\Omega_{\varepsilon})}^2 ds \le c(\varepsilon^{1+\alpha} + \varepsilon^{\beta})(1+t) \quad \text{for all} \quad t \ge 0.$$
(5.8.30)

Combining (5.8.29) and (5.8.30) with (5.5.18) we obtain (5.1.11).

Next let us prove (5.1.12). We use (5.1.10) and (5.8.28) to (5.8.15) to deduce that

$$\int_{t}^{t+1} \|A_{\varepsilon}^{1/2} u^{\varepsilon}(s)\|_{L^{2}(\Omega_{\varepsilon})}^{2} ds \le c(\varepsilon^{1+\alpha} + \varepsilon^{\beta}) \quad \text{for all} \quad t \ge 0.$$
(5.8.31)

(Note that $t_* = \min\{t+1, T_{\max}\} = t+1$ in (5.8.15) since $T_{\max} = \infty$.) Since (5.8.17) and (5.8.29) are valid for all $t \ge 0$, as in the proof of Theorem 5.1.1 we can show that (5.8.24) holds for all $t \ge 0$. When $t \in [0, 1]$, we apply (5.1.10) and (5.8.30) to the right-hand side of (5.8.24) to obtain

$$\|A_{\varepsilon}^{1/2}u^{\varepsilon}(t)\|_{L^{2}(\Omega_{\varepsilon})}^{2} \leq c(\varepsilon^{-1+\alpha} + \varepsilon^{-1+\beta}) \quad \text{for all} \quad t \in [0,1].$$
(5.8.32)

Let $t \ge 1$. In (5.8.26) the functions ξ and ζ given by (5.8.22) satisfy

$$\begin{split} \int_{t-1}^{t} \xi(s) \, ds &\leq c \int_{t-1}^{t} \|A_{\varepsilon}^{1/2} u^{\varepsilon}(s)\|_{L^{2}(\Omega_{\varepsilon})}^{2} \, ds \leq c, \\ \int_{t-1}^{t} \zeta(s) \, ds &\leq c \left(\varepsilon^{-1} \int_{t-1}^{t} \|A_{\varepsilon}^{1/2} u^{\varepsilon}(s)\|_{L^{2}(\Omega_{\varepsilon})}^{2} \, ds + \|\mathbb{P}_{\varepsilon} f^{\varepsilon}\|_{L^{\infty}(0,\infty;L^{2}(\Omega_{\varepsilon}))}^{2} \right) \\ &\leq c(\varepsilon^{-1+\alpha} + \varepsilon^{-1+\beta}) \end{split}$$

by (5.1.10), (5.8.16), and (5.8.31). Applying these inequalities and (5.8.31) to (5.8.26) we get

$$\|A_{\varepsilon}^{1/2}u^{\varepsilon}(t)\|_{L^{2}(\Omega_{\varepsilon})}^{2} \leq c(\varepsilon^{-1+\alpha} + \varepsilon^{-1+\beta}) \quad \text{for all} \quad t \geq 1.$$
(5.8.33)

By (5.5.18), (5.8.32), and (5.8.33) we obtain the first inequality of (5.1.12). To derive the second inequality we observe that (5.8.23) is valid for all $s \ge 0$ since (5.8.16) and (5.8.17) hold on the whole time interval $[0, \infty)$. Hence for each $t \ge 0$ we integrate (5.8.23) over [0, t] and use (5.1.10) and (5.8.30) to get

$$\int_0^t \|A_{\varepsilon} u^{\varepsilon}(s)\|_{L^2(\Omega_{\varepsilon})}^2 \, ds \le c(\varepsilon^{-1+\alpha} + \varepsilon^{-1+\beta})(1+t).$$

This inequality combined with (5.5.32) yields the second inequality of (5.1.12).

For the study of a singular limit problem in Section 5.10 we need estimates for the tensor product and the time derivative of a strong solution with $\beta = 1$. Let us derive them by using the inequalities given in Lemma 5.6.23.

Theorem 5.8.4. Let c_1 and c_2 be positive constants, $\alpha \in (0, 1]$, and $\beta = 1$. Under the same assumptions as in Theorem 5.1.1, let ε_1 be the constant given in Theorem 5.1.2. For $\varepsilon \in (0, \varepsilon_1)$ suppose that $u_0^{\varepsilon} \in V_{\varepsilon}$ and $f^{\varepsilon} \in L^2(0, \infty; L^2(\Omega_{\varepsilon})^3)$ satisfy (5.1.10). Then there exists a global strong solution u^{ε} to (5.1.1)–(5.1.3) such that

$$\|u^{\varepsilon}(t)\|_{L^{2}(\Omega_{\varepsilon})}^{2} \leq c\varepsilon, \quad \int_{0}^{t} \|u^{\varepsilon}(s)\|_{H^{1}(\Omega_{\varepsilon})}^{2} ds \leq c\varepsilon(1+t),$$

$$\|u^{\varepsilon}(t)\|_{H^{1}(\Omega_{\varepsilon})}^{2} \leq c\varepsilon^{-1+\alpha}, \quad \int_{0}^{t} \|u^{\varepsilon}(s)\|_{H^{2}(\Omega_{\varepsilon})}^{2} ds \leq c\varepsilon^{-1+\alpha}(1+t)$$
(5.8.34)

for all $t \geq 0$, and

$$\int_0^t \|[u^{\varepsilon} \otimes u^{\varepsilon}](s)\|_{L^2(\Omega_{\varepsilon})}^2 ds \le c\varepsilon(1+t), \tag{5.8.35}$$

$$\int_0^t \|\partial_t u^{\varepsilon}(s)\|_{L^2(\Omega_{\varepsilon})}^2 \, ds \le c\varepsilon^{-1+\alpha}(1+t) \tag{5.8.36}$$

for all $t \ge 0$, where c > 0 is a constant independent of ε , u^{ε} , and t.

Proof. A global strong solution u^{ε} exists by Theorem 5.1.2. Also, we have (5.8.34) by (5.1.11) and (5.1.12) since $\beta = 1$ and $\varepsilon \leq \varepsilon^{\alpha}$ by $\alpha \leq 1$ and $\varepsilon < 1$.

Let us derive (5.8.35). Hereafter we suppress the argument s of integrands. Noting that $u^{\varepsilon} \in L^2_{loc}([0,\infty); D(A_{\varepsilon}))$ satisfies the conditions in Lemma 5.6.23, we use (5.6.67) to get

$$\int_{0}^{t} \|u^{\varepsilon} \otimes u^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon})}^{2} ds \leq c \left(\varepsilon^{-1} \int_{0}^{t} \|u^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon})}^{2} \|u\|_{H^{1}(\Omega_{\varepsilon})}^{2} ds + \varepsilon \int_{0}^{t} \|u^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon})}^{2} \|u^{\varepsilon}\|_{H^{2}(\Omega_{\varepsilon})}^{2} ds + \int_{0}^{t} \|u^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon})}^{3} \|u^{\varepsilon}\|_{H^{2}(\Omega_{\varepsilon})}^{2} ds\right)$$
(5.8.37)

for all $t \ge 0$. By (5.8.34) we have

$$\int_0^t \|u^\varepsilon\|_{L^2(\Omega_\varepsilon)}^2 \|u^\varepsilon\|_{H^1(\Omega_\varepsilon)}^2 \, ds \le c\varepsilon^2(1+t),\tag{5.8.38}$$

$$\int_0^t \|u^{\varepsilon}\|_{L^2(\Omega_{\varepsilon})}^2 \|u^{\varepsilon}\|_{H^2(\Omega_{\varepsilon})}^2 ds \le c\varepsilon^{\alpha}(1+t).$$
(5.8.39)

Also, Hölder's inequality and (5.8.34) imply that

$$\int_{0}^{t} \|u^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon})}^{3} \|u^{\varepsilon}\|_{H^{2}(\Omega_{\varepsilon})} ds \le c\varepsilon^{3/2} t^{1/2} \left(\int_{0}^{t} \|u^{\varepsilon}\|_{H^{2}(\Omega_{\varepsilon})}^{2} ds\right)^{1/2} \le c\varepsilon^{1+\alpha/2}(1+t). \quad (5.8.40)$$

Applying (5.8.38)–(5.8.40) to (5.8.37) and noting that $\varepsilon^{\alpha}, \varepsilon^{\alpha/2} \leq 1$ we obtain (5.8.35).

Next we prove (5.8.36). We take the $L^2(\Omega_{\varepsilon})$ -inner product of $\partial_t u^{\varepsilon}$ with

$$\partial_t u^{\varepsilon} + A_{\varepsilon} u^{\varepsilon} + \mathbb{P}_{\varepsilon}[(u^{\varepsilon} \cdot \nabla) u^{\varepsilon}] = \mathbb{P}_{\varepsilon} f^{\varepsilon} \quad \text{on} \quad (0, \infty)$$

and then integrate it over (0, t) with t > 0 and use Young's inequality to get

$$\int_{0}^{t} \|\partial_{t}u^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon})}^{2} ds + \|A_{\varepsilon}^{1/2}u^{\varepsilon}(t)\|_{L^{2}(\Omega_{\varepsilon})}^{2} \\
\leq \|A_{\varepsilon}^{1/2}u_{0}^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon})}^{2} + c \int_{0}^{t} \left(\|(u^{\varepsilon} \cdot \nabla)u^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon})}^{2} + \|\mathbb{P}_{\varepsilon}f^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon})}^{2}\right) ds. \quad (5.8.41)$$

Let us estimate the integral of $(u^{\varepsilon} \cdot \nabla)u^{\varepsilon}$. By (5.6.68) we have

$$\int_0^t \|(u^{\varepsilon} \cdot \nabla)u^{\varepsilon}\|_{L^2(\Omega_{\varepsilon})}^2 \, ds \le c \int_0^t \left(\varepsilon^{-1} \|u^{\varepsilon}\|_{L^2(\Omega_{\varepsilon})}^2 + \varepsilon \|u^{\varepsilon}\|_{H^1(\Omega_{\varepsilon})}^2\right) \|u^{\varepsilon}\|_{H^2(\Omega_{\varepsilon})}^2 \, ds.$$

To the right-hand side we apply (5.8.39) and

$$\int_0^r \|u^{\varepsilon}\|_{H^1(\Omega_{\varepsilon})}^2 \|u^{\varepsilon}\|_{H^2(\Omega_{\varepsilon})}^2 \, ds \le c\varepsilon^{-2+2\alpha}(1+t) \tag{5.8.42}$$

by (5.8.34), and then use $\varepsilon^{\alpha} \leq 1$ by $\varepsilon < 1$ to deduce that

$$\int_0^t \|(u^{\varepsilon} \cdot \nabla)u^{\varepsilon}\|_{L^2(\Omega_{\varepsilon})}^2 \, ds \le c\varepsilon^{-1+\alpha}(1+t).$$

Applying this inequality and (5.1.10) to (5.8.41) we obtain (5.8.36).

5.9 Weighted solenoidal spaces on a surface

In the study of a singular limit problem for the Navier–Stokes equations (5.1.1)–(5.1.3) we deal with a weighted solenoidal space of the form

$$H^1_{q\sigma}(\Gamma, T\Gamma) = \{ v \in H^1(\Gamma, T\Gamma) \mid \operatorname{div}_{\Gamma}(gv) = 0 \text{ on } \Gamma \}.$$

The purpose of this section is to establish several properties of weighted solenoidal spaces on a surface. Throughout this section we assume that Γ is a two-dimensional closed, connected, and oriented surface in \mathbb{R}^3 of class C^2 . We use the notations for the surface quantities on Γ given in Section 5.2.1.

5.9.1 Nečas inequality on a surface

Let $q \in L^2(\Gamma)$. We consider q and its weak tangential gradient as elements in $H^{-1}(\Gamma)$ and $H^{-1}(\Gamma, T\Gamma)$ given by (5.2.21) and (5.2.26). By these equalities we easily get

 $||q||_{H^{-1}(\Gamma)} + ||\nabla_{\Gamma} q||_{H^{-1}(\Gamma,T\Gamma)} \le c ||q||_{L^{2}(\Gamma)}.$

For bounded Lipschitz domains in \mathbb{R}^m , $m \in \mathbb{N}$ the inverse inequality is also valid and known as the Nečas inequality (see [48, Chapter 3, Lemma 7.1]). Let us show the Nečas inequality on the surface Γ .

Lemma 5.9.1. There exists a constant c > 0 such that

$$\|q\|_{L^{2}(\Gamma)} \leq c \left(\|q\|_{H^{-1}(\Gamma)} + \|\nabla_{\Gamma} q\|_{H^{-1}(\Gamma, T\Gamma)}\right)$$
(5.9.1)

for all $q \in L^2(\Gamma)$.

In the proof of Lemma 5.9.1 we use the Načas inequality on the whole space

$$\|\tilde{q}\|_{L^{2}(\mathbb{R}^{2})} \leq c \left(\|\tilde{q}\|_{H^{-1}(\mathbb{R}^{2})} + \|\nabla_{s}\tilde{q}\|_{H^{-1}(\mathbb{R}^{2})} \right)$$
(5.9.2)

for $\tilde{q} \in L^2(\mathbb{R}^2)$, where $H^{-1}(\mathbb{R}^2)$ is the dual space of $H^1(\mathbb{R}^2)$ (via the $L^2(\mathbb{R}^2)$ -inner product). Also, $\tilde{q} \in H^{-1}(\mathbb{R}^2)$ and $\nabla_s \tilde{q} \in H^{-1}(\mathbb{R}^2)^2$ are given by

$$\langle \tilde{q}, \xi \rangle_{\mathbb{R}^2} := (\tilde{q}, \xi)_{L^2(\mathbb{R}^2)}, \quad \langle \nabla_s \tilde{q}, \varphi \rangle_{\mathbb{R}^2} := -(q, \partial_{s_1} \varphi_1 + \partial_{s_2} \varphi_2)_{L^2(\mathbb{R}^2)}$$

for $\xi \in H^1(\mathbb{R}^2)$ and $\varphi = (\varphi_1, \varphi_2) \in H^1(\mathbb{R}^2)^2$, where $\langle \cdot, \cdot \rangle_{\mathbb{R}^2}$ is the duality product between $H^{-1}(\mathbb{R}^2)$ and $H^1(\mathbb{R}^2)$. The inequality (5.9.2) follows from the characterization of the L^2 -Sobolev spaces on \mathbb{R}^2 by the Fourier transform. See the proof of [48, Chapter 3, Lemma 7.1] for details.

Proof. First we note that it is sufficient to show (5.9.1) when q is compactly supported in a relatively open subset of Γ on which we can take a local coordinate system. To see this, let $q \in L^2(\Gamma)$ and $\eta \in C^2(\Gamma)$. For $\xi \in H^1(\Gamma)$ we have

$$|\langle \eta q, \xi \rangle_{\Gamma}| = |\langle q, \eta \xi \rangle_{\Gamma}| \le ||q||_{H^{-1}(\Gamma)} ||\eta \xi||_{H^{1}(\Gamma)} \le c ||\eta||_{W^{1,\infty}(\Gamma)} ||q||_{H^{-1}(\Gamma)} ||\xi||_{H^{1}(\Gamma)},$$

where c > 0 is a constant independent of q, η , and ξ . Also,

$$\begin{aligned} [\nabla_{\Gamma}(\eta q), v]_{T\Gamma} &= -(\eta q, \operatorname{div}_{\Gamma} v)_{L^{2}(\Gamma)} = -\left(q, \operatorname{div}_{\Gamma}(\eta v)\right)_{L^{2}(\Gamma)} + (q, \nabla_{\Gamma} \eta \cdot v)_{L^{2}(\Gamma)} \\ &= [\nabla_{\Gamma} q, \eta v]_{T\Gamma} + \langle q, \nabla_{\Gamma} \eta \cdot v \rangle_{\Gamma} \end{aligned}$$

for all $v \in H^1(\Gamma, T\Gamma)$ by (5.2.26) (note that $\eta v \in H^1(\Gamma, T\Gamma)$) and thus

$$\begin{aligned} |[\nabla_{\Gamma}(\eta q), v]_{T\Gamma}| &\leq \|\nabla_{\Gamma} q\|_{H^{-1}(\Gamma, T\Gamma)} \|\eta v\|_{H^{1}(\Gamma)} + \|q\|_{H^{-1}(\Gamma)} \|\nabla_{\Gamma} \eta \cdot v\|_{H^{1}(\Gamma)} \\ &\leq c \|\eta\|_{W^{2,\infty}(\Gamma)} \left(\|q\|_{H^{-1}(\Gamma)} + \|\nabla_{\Gamma} q\|_{H^{-1}(\Gamma, T\Gamma)} \right) \|v\|_{H^{1}(\Gamma)}. \end{aligned}$$

From the above inequalities it follows that

$$\begin{aligned} \|\eta q\|_{H^{-1}(\Gamma)} &\leq c \|\eta\|_{W^{1,\infty}(\Gamma)} \|q\|_{H^{-1}(\Gamma)}, \\ \|\nabla_{\Gamma}(\eta q)\|_{H^{-1}(\Gamma,T\Gamma)} &\leq c \|\eta\|_{W^{2,\infty}(\Gamma)} \left(\|q\|_{H^{-1}(\Gamma)} + \|\nabla_{\Gamma} q\|_{H^{-1}(\Gamma,T\Gamma)}\right). \end{aligned}$$

By these inequalities we can get (5.9.1) for q if we localize it by a partition of unity of Γ consisting of functions in $C^2(\Gamma)$ (we can take such functions since Γ is of class C^2) and prove (5.9.1) for each localized function.

From now on, we assume that $q \in L^2(\Gamma)$ is compactly supported in a relatively open subset $\mu(U)$ of Γ , where U is an open subset of \mathbb{R}^2 and $\mu: U \to \mathbb{R}^2$ is a local parametrization of Γ . (Note that μ is of class C^2 on U since Γ is of class C^2 .) Let $\tilde{q} := q \circ \mu$ on U and \mathcal{K} be the support of \tilde{q} , which is a compact subset of U. The Riemannian metric $\theta = (\theta_{ij})_{i,j}$ of Γ is locally defined by

$$\theta_{ij}(s) := \partial_{s_i} \mu(s) \cdot \partial_{s_j} \mu(s), \quad s \in U, \, i, j = 1, 2.$$

We write $\theta^{-1} = (\theta^{ij})_{i,j}$ for the inverse matrix of θ . Since μ is of class C^2 , there exists a constant c > 0 such that

$$|\partial_s^{\alpha}\mu(s)| \le c, \quad s \in \mathcal{K}, \ |\alpha| = 1, 2, \tag{5.9.3}$$

where $\partial_s^{\alpha} = \partial_{s_1}^{\alpha_1} \partial_{s_2}^{\alpha_2}$ for $\alpha = (\alpha_1, \alpha_2) \in \mathbb{Z}^2$ with $\alpha_i \ge 0$, i = 1, 2. By this inequality and the formula $\partial_{s_i} \theta^{-1} = -\theta^{-1} (\partial_{s_i} \theta) \theta^{-1}$ we also have

$$|\theta(s)| \le c, \quad |\theta^{-1}(s)| \le c, \quad |\partial_{s_i}\theta(s)| \le c, \quad |\partial_{s_i}\theta^{-1}(s)| \le c, \quad s \in \mathcal{K}, \ i = 1, 2.$$
(5.9.4)

Moreover, since the determinant of θ is continuous and does not vanish on U, there exists a constant c > 0 such that

$$c^{-1} \le \det \theta(s) \le c, \quad s \in \mathcal{K}.$$
 (5.9.5)

We extend \tilde{q} to \mathbb{R}^2 by setting zero outside of U. Then by (5.9.5) we get

$$\|q\|_{L^{2}(\Gamma)}^{2} = \int_{\mathcal{K}} |\tilde{q}|^{2} \sqrt{\det \theta} \, ds \le c \|\tilde{q}\|_{L^{2}(\mathcal{K})}^{2} = c \|\tilde{q}\|_{L^{2}(\mathbb{R}^{2})}^{2}$$

and we apply the inequality (5.9.2) to the last term to obtain

$$\|q\|_{L^{2}(\Gamma)} \leq c \left(\|\tilde{q}\|_{H^{-1}(\mathbb{R}^{2})} + \|\nabla_{s}\tilde{q}\|_{H^{-1}(\mathbb{R}^{2})}\right).$$
(5.9.6)

Let us estimate the right-hand side of (5.9.6) by that of (5.9.1). To this end, we first consider the duality product $\langle \tilde{q}, \xi \rangle_{\mathbb{R}^2} = (\tilde{q}, \xi)_{L^2(\mathbb{R}^2)}$ for an arbitrary $\xi \in H^1(\mathbb{R}^2)$. Since \tilde{q} is supported in \mathcal{K} , we may assume that ξ is also supported in \mathcal{K} . We define a function η on $\mu(\mathcal{K}) \subset \Gamma$ by

$$\eta(\mu(s)) := \frac{\xi(s)}{\sqrt{\det \theta(s)}}, \quad s \in \mathcal{K}$$
(5.9.7)

and extend it to Γ by setting zero outside of $\mu(\mathcal{K})$. Then we have

$$\langle \tilde{q}, \xi \rangle_{\mathbb{R}^2} = (\tilde{q}, \xi)_{L^2(\mathbb{R}^2)} = \int_{\mathcal{K}} \tilde{q} \left(\frac{\xi}{\sqrt{\det \theta}}\right) \sqrt{\det \theta} \, ds = \int_{\mu(\mathcal{K})} q\eta \, d\mathcal{H}^2. \tag{5.9.8}$$

Let us show $\eta \in H^1(\Gamma)$. Since η is supported in $\mu(\mathcal{K})$, we see by (5.9.5) that

$$\|\eta\|_{L^{2}(\Gamma)}^{2} = \int_{\mathcal{K}} |\eta \circ \mu|^{2} \sqrt{\det \theta} \, ds \le c \|\xi\|_{L^{2}(\mathcal{K})}^{2} = c \|\xi\|_{L^{2}(\mathbb{R}^{2})}^{2}.$$
(5.9.9)

To estimate the $L^2(\Gamma)$ -norm of the tangential gradient of η , we differentiate both sides of (5.9.7) with respect to s_i , i = 1, 2 and use (5.9.4) and (5.9.5) to get

$$|\partial_{s_i}(\eta \circ \mu)(s)| \le c(|\xi(s)| + |\partial_{s_i}\xi(s)|), \quad s \in \mathcal{K}.$$

Since the norm of $(\nabla_{\Gamma}\eta) \circ \mu = \sum_{i,j=1}^{2} \theta^{ij} \partial_{s_i}(\eta \circ \mu) \partial_{s_j}\mu$ is given by

$$|(\nabla_{\Gamma}\eta)\circ\mu|^2 = \sum_{i,j=1}^2 \theta^{ij}\partial_{s_i}(\eta\circ\mu)\partial_{s_j}(\eta\circ\mu) \quad \text{in} \quad \mathcal{K},$$

we apply the above inequality and (5.9.4) to the right-hand side to get

$$|(\nabla_{\Gamma}\eta)\circ\mu|^2 \le c(|\xi|^2+|\nabla_s\xi|^2)$$
 in \mathcal{K} ,

where ∇_s is the gradient operator in $s \in \mathbb{R}^2$. Noting that η and ξ are supported in $\mu(\mathcal{K})$ and \mathcal{K} , respectively, we use this inequality and (5.9.5) to obtain

$$\|\nabla_{\Gamma}\eta\|_{L^{2}(\Gamma)}^{2} = \int_{\mathcal{K}} |(\nabla_{\Gamma}\eta)\circ\mu|^{2}\sqrt{\det\theta}\,ds \le c\|\xi\|_{H^{1}(\mathcal{K})}^{2} = c\|\xi\|_{H^{1}(\mathbb{R}^{2})}^{2}.$$

By this inequality and (5.9.9) we have $\eta \in H^1(\Gamma)$ and $\|\eta\|_{H^1(\Gamma)} \leq c \|\xi\|_{H^1(\mathbb{R}^2)}$. Hence

$$\langle \tilde{q}, \xi \rangle_{\mathbb{R}^2} = \int_{\mu(\mathcal{K})} q\eta \, d\mathcal{H}^2 = (q, \eta)_{L^2(\Gamma)} = \langle q, \eta \rangle_{\Gamma}$$

by (5.9.8) and the above inequality implies that

$$\left| \langle \tilde{q}, \xi \rangle_{\mathbb{R}^2} \right| = \left| \langle q, \eta \rangle_{\Gamma} \right| \le \|q\|_{H^{-1}(\Gamma)} \|\eta\|_{H^1(\Gamma)} \le c \|q\|_{H^{-1}(\Gamma)} \|\xi\|_{H^1(\mathbb{R}^2)}$$

for all $\xi \in H^1(\mathbb{R}^2)$. Therefore,

$$\|\tilde{q}\|_{H^{-1}(\mathbb{R}^2)} \le c \|q\|_{H^{-1}(\Gamma)}.$$
(5.9.10)

Next let $\varphi = (\varphi_1, \varphi_2) \in H^1(\mathbb{R}^2)^2$ and we consider the duality product

$$\langle \nabla_s \tilde{q}, \varphi \rangle_{\mathbb{R}^2} = -(\tilde{q}, \partial_{s_1} \varphi_1 + \partial_{s_2} \varphi_2)_{L^2(\mathbb{R}^2)}$$

We may assume again that φ is supported in \mathcal{K} since \tilde{q} is so. To express this product in terms of an integral over Γ , we recall that the surface divergence of a tangential vector field $X(\mu(s)) = \sum_{i=1}^{2} X_i(s) \partial_{s_i} \mu(s), s \in \mathcal{K}$ is given by

$$\operatorname{div}_{\Gamma} X(\mu(s)) = \frac{1}{\sqrt{\det \theta(s)}} \sum_{i=1}^{2} \frac{\partial}{\partial s_i} \Big(X_i(s) \sqrt{\det \theta(s)} \Big), \quad s \in \mathcal{K}.$$

Based on this formula, we define a tangential vector field Φ on $\mu(\mathcal{K})$ by

$$\Phi(\mu(s)) := \frac{1}{\sqrt{\det \theta(s)}} \sum_{i=1}^{2} \varphi_i(s) \partial_{s_i} \mu(s), \quad s \in \mathcal{K}$$
(5.9.11)

and extend it to Γ by setting zero outside of $\mu(\mathcal{K})$. Then we have

$$\operatorname{div}_{\Gamma}\Phi(\mu(s)) = \frac{\partial_{s_1}\varphi_1(s) + \partial_{s_2}\varphi_2(s)}{\sqrt{\det \theta(s)}}, \quad s \in \mathcal{K}$$

and thus we get (note that $\tilde{q} = q \circ \mu$ and φ are supported in \mathcal{K})

$$\langle \nabla_s \tilde{q}, \varphi \rangle_{\mathbb{R}^2} = -(\tilde{q}, \partial_{s_1} \varphi_1 + \partial_{s_2} \varphi_2)_{L^2(\mathbb{R}^2)} = -\int_{\mathcal{K}} \tilde{q} \left(\frac{\partial_{s_1} \varphi_1 + \partial_{s_2} \varphi_2}{\sqrt{\det \theta}} \right) \sqrt{\det \theta} \, ds = -\int_{\mu(\mathcal{K})} q \operatorname{div}_{\Gamma} \Phi \, d\mathcal{H}^2.$$

$$(5.9.12)$$

Let us estimate the $H^1(\Gamma)$ -norm of Φ . By (5.9.4), (5.9.5), and (5.9.11) we have

$$|\Phi \circ \mu|^2 = \frac{1}{\det \theta} \sum_{i,j=1}^2 \theta_{ij} \varphi_i \varphi_j \le c |\varphi|^2 \quad \text{in} \quad \mathcal{K}.$$

Since Φ is supported in $\mu(\mathcal{K})$, the above inequality and (5.9.5) imply

$$\|\Phi\|_{L^{2}(\Gamma)}^{2} = \int_{\mathcal{K}} |\Phi \circ \mu|^{2} \sqrt{\det \theta} \, ds \le c \|\varphi\|_{L^{2}(\mathcal{K})}^{2} = c \|\varphi\|_{L^{2}(\mathbb{R}^{2})}^{2}, \tag{5.9.13}$$

where the last equality follows from the fact that φ is supported in \mathcal{K} . To estimate the tangential derivatives of Φ , let $\{e_1, e_2, e_3\}$ be the standard basis of \mathbb{R}^3 and $\Phi_k := \Phi \cdot e_k$ for k = 1, 2, 3. Then Φ_k is supported in $\mu(\mathcal{K})$ and

$$\Phi_k(\mu(s)) = \frac{1}{\sqrt{\det \theta(s)}} \sum_{j=1}^2 \varphi_j(s) \partial_{s_j} \mu(s) \cdot e_k, \quad s \in \mathcal{K}.$$

We differentiate both sides with respect to s_i and use (5.9.3), (5.9.4), and (5.9.5) to get

$$|\partial_{s_i}(\Phi_k \circ \mu)(s)| \le c \left(|\varphi(s)| + |\nabla_s \varphi(s)|\right), \quad s \in \mathcal{K}, \ i = 1, 2.$$

From this inequality and (5.9.4) we deduce that

$$|(\nabla_{\Gamma}\Phi_k)\circ\mu|^2 = \sum_{i,j=1}^2 \theta^{ij}\partial_{s_i}(\Phi_k\circ\mu)\partial_{s_j}(\Phi_k\circ\mu) \le c(|\varphi|^2 + |\nabla_s\varphi|^2)$$

in \mathcal{K} for k = 1, 2, 3, and thus $|(\nabla_{\Gamma} \Phi) \circ \mu|^2 \leq c(|\varphi|^2 + |\nabla_s \varphi|^2)$ in \mathcal{K} . Noting that Φ and φ are supported in $\mu(\mathcal{K})$ and \mathcal{K} , respectively, we use this inequality and (5.9.5) to observe that

$$\|\nabla_{\Gamma}\Phi\|_{L^{2}(\Gamma)}^{2} = \int_{\mathcal{K}} |(\nabla_{\Gamma}\Phi)\circ\mu|^{2}\sqrt{\det\theta}\,ds \le c\|\varphi\|_{H^{1}(\mathcal{K})}^{2} = c\|\varphi\|_{H^{1}(\mathbb{R}^{2})}^{2}.$$

From this inequality and (5.9.13) it follows that $\Phi \in H^1(\Gamma, T\Gamma)$ and

$$\|\Phi\|_{H^1(\Gamma)} \le c \|\varphi\|_{H^1(\mathbb{R}^2)}.$$
(5.9.14)

Now we return to (5.9.12) and use (5.2.26) to get

$$\langle \nabla_s \tilde{q}, \varphi \rangle_{\mathbb{R}^2} = -\int_{\mu(\mathcal{K})} q \operatorname{div}_{\Gamma} \Phi \, d\mathcal{H}^2 = -(q, \operatorname{div}_{\Gamma} \Phi)_{L^2(\Gamma)} = [\nabla_{\Gamma} q, \Phi]_{T\Gamma}.$$

Hence by (5.9.14) we obtain

$$\left| \langle \nabla_s \tilde{q}, \varphi \rangle_{\mathbb{R}^2} \right| = \left| [\nabla_{\Gamma} q, \Phi]_{T\Gamma} \right| \le \|\nabla_{\Gamma} q\|_{H^{-1}(\Gamma, T\Gamma)} \|\Phi\|_{H^1(\Gamma)} \le c \|\nabla_{\Gamma} q\|_{H^{-1}(\Gamma, T\Gamma)} \|\varphi\|_{H^1(\mathbb{R}^2)}$$

for all $\varphi \in H^1(\mathbb{R}^2)^2$, which implies that

$$\|\nabla_s \tilde{q}\|_{H^{-1}(\mathbb{R}^2)} \le c \|\nabla_\Gamma q\|_{H^{-1}(\Gamma, T\Gamma)}$$

Finally, we apply this inequality and (5.9.10) to (5.9.6) to conclude that the inequality (5.9.1) is valid.

Next we prove Poincaré's inequality for $q \in L^2(\Gamma)$ based on the Nečas inequality (5.9.1). We first show that the tangential gradient of q vanishes in $H^{-1}(\Gamma, T\Gamma)$ if and only if q is constant on Γ .

Lemma 5.9.2. Let $q \in L^2(\Gamma)$. Then

$$\nabla_{\Gamma} q = 0$$
 in $H^{-1}(\Gamma, T\Gamma)$

if and only if q is constant on Γ .

Proof. Suppose first that q is constant on Γ . Then for all $v \in H^1(\Gamma, T\Gamma)$ we have

$$[\nabla_{\Gamma} q, v]_{T\Gamma} = -q \int_{\Gamma} \operatorname{div}_{\Gamma} v \, d\mathcal{H}^2 = 0$$

by (5.2.26) and the Stokes theorem. Hence $\nabla_{\Gamma} q = 0$ in $H^{-1}(\Gamma, T\Gamma)$.

Conversely, assume that $\nabla_{\Gamma}q = 0$ in $H^{-1}(\Gamma, T\Gamma)$. We first prove $q \in H^{1}(\Gamma)$. For $\eta \in C^{1}(\Gamma)$ and i = 1, 2, 3 we set $v := \eta Pe_{i}$, where $\{e_{1}, e_{2}, e_{3}\}$ is the standard basis of \mathbb{R}^{3} . Then $v \in H^{1}(\Gamma, T\Gamma)$ since P is of class C^{1} . Moreover,

$$\operatorname{div}_{\Gamma} v = \nabla_{\Gamma} \eta \cdot P e_i + \eta (\operatorname{div}_{\Gamma} P \cdot e_i) = \underline{D}_i \eta + \eta H n_i \quad \text{on} \quad \Gamma$$

by $P^T = P$, (5.2.3), and (5.2.7). From this equality we deduce that

$$0 = [\nabla_{\Gamma} q, v]_{T\Gamma} = -(q, \operatorname{div}_{\Gamma} v)_{L^2(\Gamma)} = -(q, \underline{D}_i \eta + \eta H n_i)_{L^2(\Gamma)}$$

for all $\eta \in C^1(\Gamma)$. Hence by the definition of the weak tangential derivative in $L^2(\Gamma)$ (see (5.2.19)) we get $\underline{D}_i q = 0$ in $L^2(\Gamma)$ for i = 1, 2, 3, which shows that $q \in H^1(\Gamma)$ and $\nabla_{\Gamma} q = 0$ in $L^2(\Gamma)^3$. Now we set

$$\hat{q} := q - \frac{1}{|\Gamma|} \int_{\Gamma} q \, d\mathcal{H}^2 \quad \text{on} \quad \Gamma,$$

where $|\Gamma|$ is the area of Γ . Then we have $\hat{q} \in H^1(\Gamma)$ and $\int_{\Gamma} \hat{q} d\mathcal{H}^2 = 0$. Hence we can apply Poincaré's inequality (5.2.20) to \hat{q} to get

$$\|\hat{q}\|_{L^2(\Gamma)} \le c \|\nabla_{\Gamma} \hat{q}\|_{L^2(\Gamma)} = c \|\nabla_{\Gamma} q\|_{L^2(\Gamma)} = 0,$$

i.e. $\hat{q} = 0$ on Γ , which implies that $q = |\Gamma|^{-1} \int_{\Gamma} q \, d\mathcal{H}^2$ is constant on Γ .

Next we estimate $q \in L^2(\Gamma)$ in $H^{-1}(\Gamma)$ by its tangential gradient.

Lemma 5.9.3. There exists a constant c > 0 such that

$$\|q\|_{H^{-1}(\Gamma)} \le c \left(\left| \int_{\Gamma} q \, d\mathcal{H}^2 \right| + \|\nabla_{\Gamma} q\|_{H^{-1}(\Gamma, T\Gamma)} \right)$$
(5.9.15)

for all $q \in L^2(\Gamma)$.

Proof. We prove (5.9.15) by contradiction. Assume to the contrary that for each $k \in \mathbb{N}$ there exists $q_k \in L^2(\Gamma)$ such that

$$\|q_k\|_{H^{-1}(\Gamma)} > k\left(\left|\int_{\Gamma} q_k \, d\mathcal{H}^2\right| + \|\nabla_{\Gamma} q_k\|_{H^{-1}(\Gamma, T\Gamma)}\right).$$

Replacing q_k with $q_k/||q_k||_{H^{-1}(\Gamma)}$ we may assume that

$$||q_k||_{H^{-1}(\Gamma)} = 1, \quad \left| \int_{\Gamma} q_k \, d\mathcal{H}^2 \right| + ||\nabla_{\Gamma} q_k||_{H^{-1}(\Gamma, T\Gamma)} < k^{-1}.$$
 (5.9.16)

From the second inequality it follows that

$$\lim_{k \to \infty} \int_{\Gamma} q_k \, d\mathcal{H}^2 = \lim_{k \to \infty} (q_k, 1)_{L^2(\Gamma)} = 0, \quad \lim_{k \to \infty} \|\nabla_{\Gamma} q_k\|_{H^{-1}(\Gamma, T\Gamma)} = 0.$$
(5.9.17)

By (5.9.1) and (5.9.16) the sequence $\{q_k\}_{k=1}^{\infty}$ is bounded in $L^2(\Gamma)$. Hence there exists a subsequence of $\{q_k\}_{k=1}^{\infty}$, which is referred to as $\{q_k\}_{k=1}^{\infty}$ again, that converges to some q weakly in $L^2(\Gamma)$. Since the embedding $L^2(\Gamma) \hookrightarrow H^{-1}(\Gamma)$ is compact, by taking a subsequence we may assume that $\{q_k\}_{k=1}^{\infty}$ converges to q strongly in $H^{-1}(\Gamma)$. Hence by the first equality of (5.9.16) we have

$$\|q\|_{H^{-1}(\Gamma)} = \lim_{k \to \infty} \|q_k\|_{H^{-1}(\Gamma)} = 1.$$
(5.9.18)

By (5.2.26) and the weak convergence of $\{q_k\}_{k=1}^{\infty}$ to q in $L^2(\Gamma)$ we see that $\{\nabla_{\Gamma} q_k\}_{k=1}^{\infty}$ converges to $\nabla_{\Gamma} q$ weakly in $H^{-1}(\Gamma, T\Gamma)$. Hence by (5.9.17) we get

$$\int_{\Gamma} q \, d\mathcal{H}^2 = (q, 1)_{L^2(\Gamma)} = 0, \quad \|\nabla_{\Gamma} q\|_{H^{-1}(\Gamma, T\Gamma)} = 0.$$

By these equalities and Lemma 5.9.2 we find that q = 0 on Γ and thus $||q||_{H^{-1}(\Gamma)} = 0$, which contradicts with (5.9.18). Hence (5.9.15) is valid.

Combining (5.9.1) and (5.9.15) we obtain Poincaré's inequality for $q \in L^2(\Gamma)$.

Lemma 5.9.4. There exists a constant c > 0 such that

$$\|q\|_{L^{2}(\Gamma)} \leq c \left(\left| \int_{\Gamma} q \, d\mathcal{H}^{2} \right| + \|\nabla_{\Gamma} q\|_{H^{-1}(\Gamma, T\Gamma)} \right)$$
(5.9.19)

for all $q \in L^2(\Gamma)$.

5.9.2 Characterization of the annihilator of a weighted solenoidal space

Let $g \in C^1(\Gamma)$ satisfy $g \ge c$ on Γ with some constant c > 0. We define a weighted solenoidal space of tangential vector fields

$$H^1_{g\sigma}(\Gamma, T\Gamma) := \{ v \in H^1(\Gamma, T\Gamma) \mid \operatorname{div}_{\Gamma}(gv) = 0 \text{ on } \Gamma \},\$$

If $q \in L^2(\Gamma)$, then (5.2.26) and (5.2.27) imply that

$$[g\nabla_{\Gamma}q, v]_{T\Gamma} = -(q, \operatorname{div}_{\Gamma}(gv))_{L^{2}(\Gamma)} = 0 \quad \text{for all} \quad v \in H^{1}_{g\sigma}(\Gamma, T\Gamma).$$

Let us prove the converse of this statement for an element of $H^{-1}(\Gamma, T\Gamma)$, which is a weighted version of de Rham's theorem.

Theorem 5.9.5. Suppose that $f \in H^{-1}(\Gamma, T\Gamma)$ satisfies

$$[f, v]_{T\Gamma} = 0$$
 for all $v \in H^1_{a\sigma}(\Gamma, T\Gamma)$.

Then there exists a unique $q \in L^2(\Gamma)$ such that

$$f = g \nabla_{\Gamma} q$$
 in $H^{-1}(\Gamma, T\Gamma)$, $\int_{\Gamma} q \, d\mathcal{H}^2 = 0$.

Moreover, there exists a constant c > 0 independent of f such that

$$\|q\|_{L^{2}(\Gamma)} \leq c \|f\|_{H^{-1}(\Gamma, T\Gamma)}.$$
(5.9.20)

We give auxiliary lemmas for Theorem 5.9.5.

Lemma 5.9.6. There exists c > 0 such that

$$c^{-1} \|\nabla_{\Gamma} q\|_{H^{-1}(\Gamma, T\Gamma)} \le \|g\nabla_{\Gamma} q\|_{H^{-1}(\Gamma, T\Gamma)} \le c \|\nabla_{\Gamma} q\|_{H^{-1}(\Gamma, T\Gamma)}$$
(5.9.21)

for all $q \in L^2(\Gamma)$.

Proof. Since $g \in C^1(\Gamma)$ is bounded from below by a positive constant, we have

$$\begin{aligned} |[\nabla_{\Gamma}q, v]_{T\Gamma}| &= \left| [g\nabla_{\Gamma}q, g^{-1}v]_{T\Gamma} \right| \le \|g\nabla_{\Gamma}q\|_{H^{-1}(\Gamma, T\Gamma)} \|g^{-1}v\|_{H^{1}(\Gamma)} \\ &\le c \|g\nabla_{\Gamma}q\|_{H^{-1}(\Gamma, T\Gamma)} \|v\|_{H^{1}(\Gamma)} \end{aligned}$$

for all $v \in H^1(\Gamma, T\Gamma)$. Hence the left-hand inequality of (5.9.21) holds. Similarly, we can show the right-hand inequality of (5.9.21).

Lemma 5.9.7. The subspace

$$\mathcal{X} := \{ g \nabla_{\Gamma} q \in H^{-1}(\Gamma, T\Gamma) \mid q \in L^2(\Gamma) \}$$
(5.9.22)

is closed in $H^{-1}(\Gamma, T\Gamma)$.
Proof. Let $\{q_k\}_{k=1}^{\infty}$ be a sequence in $L^2(\Gamma)$ such that $\{g\nabla_{\Gamma}q_k\}_{k=1}^{\infty}$ converges to some f strongly in $H^{-1}(\Gamma, T\Gamma)$. For each $k \in \mathbb{N}$ we subtract the average of q_k over Γ from q_k to assume $\int_{\Gamma} q_k d\mathcal{H}^2 = 0$ without changing $g\nabla_{\Gamma}q_k$ (see Lemma 5.9.2). Then by (5.9.19) and (5.9.21) we see that

$$||q_k - q_l||_{L^2(\Gamma)} \le c ||g \nabla_{\Gamma} q_k - g \nabla_{\Gamma} q_l||_{H^{-1}(\Gamma, T\Gamma)} \to 0 \quad \text{as} \quad k, l \to \infty$$

Hence $\{q_k\}_{k=1}^{\infty}$ is a Cauchy sequence in $L^2(\Gamma)$ and converges to some q strongly in $L^2(\Gamma)$. Then by (5.2.26) and (5.2.27) we easily see that

$$\|g\nabla_{\Gamma}q - g\nabla_{\Gamma}q_k\|_{H^{-1}(\Gamma,T\Gamma)} \le c\|q - q_k\|_{L^2(\Gamma)} \to 0 \quad \text{as} \quad k \to \infty$$

Since $\{g\nabla_{\Gamma}q_k\}_{k=1}^{\infty}$ converges to f strongly in $H^{-1}(\Gamma, T\Gamma)$, we conclude by the above convergence that $f = g\nabla_{\Gamma}q \in \mathcal{X}$. Therefore, \mathcal{X} is closed in $H^{-1}(\Gamma, T\Gamma)$.

To prove Theorem 5.9.5 we use basic results of functional analysis. Let \mathcal{B} be a Banach space. For a subset \mathcal{X} of \mathcal{B} we define the annihilator of \mathcal{X} by

$$\mathcal{X}^{\perp} := \{ f \in \mathcal{B}' \mid {}_{\mathcal{B}'} \langle f, v \rangle_{\mathcal{B}} = 0 \text{ for all } v \in \mathcal{X} \},\$$

where \mathcal{B}' is the dual of \mathcal{B} and $_{\mathcal{B}'}\langle \cdot, \cdot \rangle_{\mathcal{B}}$ is the duality product between \mathcal{B}' and \mathcal{B} .

Lemma 5.9.8. Let \mathcal{X} and \mathcal{Y} be subsets of \mathcal{B} . If $\mathcal{X} \subset \mathcal{Y}$ in \mathcal{B} , then $\mathcal{Y}^{\perp} \subset \mathcal{X}^{\perp}$ in \mathcal{B}' .

Lemma 5.9.9. If \mathcal{B} is reflexive and \mathcal{X} is a closed subspace of \mathcal{B} , then $(\mathcal{X}^{\perp})^{\perp} = \mathcal{X}$.

Lemma 5.9.8 is an immediate consequence of the definition of the annihilator. Also, Lemma 5.9.9 follows from the Hahn–Banach theorem, see e.g. [56, Theorem 4.7].

Proof of Theorem 5.9.5. Since $H^1(\Gamma, T\Gamma)$ is a Hilbert space, its dual $H^{-1}(\Gamma, T\Gamma)$ is also a Hilbert space and thus reflexive. Let \mathcal{X} be the subspace of $H^{-1}(\Gamma, T\Gamma)$ given by (5.9.22) and $v \in \mathcal{X}^{\perp} \subset H^1(\Gamma, T\Gamma)$. Then for all $q \in L^2(\Gamma)$ we have

$$0 = [g\nabla_{\Gamma}q, v]_{T\Gamma} = -(q, \operatorname{div}_{\Gamma}(gv))_{L^{2}(\Gamma)}$$

by $g\nabla_{\Gamma}q \in \mathcal{X}$, (5.2.26), and (5.2.27) and thus $\operatorname{div}_{\Gamma}(gv) = 0$ on Γ , i.e. $v \in H^1_{g\sigma}(\Gamma, T\Gamma)$. Hence $\mathcal{X}^{\perp} \subset H^1_{a\sigma}(\Gamma, T\Gamma)$ in $H^1(\Gamma, T\Gamma)$ and by Lemma 5.9.8 we have

$$H^1_{g\sigma}(\Gamma, T\Gamma)^{\perp} = \{ f \in H^{-1}(\Gamma, T\Gamma) \mid [f, v]_{T\Gamma} = 0 \text{ for all } v \in H^1_{g\sigma}(\Gamma, T\Gamma) \} \subset (\mathcal{X}^{\perp})^{\perp}.$$

Since \mathcal{X} is closed in $H^{-1}(\Gamma, T\Gamma)$ by Lemma 5.9.7, we have $(\mathcal{X}^{\perp})^{\perp} = \mathcal{X}$ by Lemma 5.9.9. Hence $H^1_{g\sigma}(\Gamma, T\Gamma)^{\perp} \subset \mathcal{X}$, i.e. for $f \in H^1_{g\sigma}(\Gamma, T\Gamma)^{\perp}$ there exists $q \in L^2(\Gamma)$ such that $f = g\nabla_{\Gamma}q$ in $H^{-1}(\Gamma, T\Gamma)$. Moreover, subtracting the average of q over Γ from q we may assume $\int_{\Gamma} q \, d\mathcal{H}^2 = 0$ without changing $g\nabla_{\Gamma}q$ (see Lemma 5.9.2). Therefore, the existence part of the theorem is valid. To prove the uniqueness, suppose that $q_1, q_2 \in L^2(\Gamma)$ satisfy

$$g\nabla_{\Gamma}q_1 = g\nabla_{\Gamma}q_2$$
 in $H^{-1}(\Gamma, T\Gamma)$, $\int_{\Gamma}q_1 d\mathcal{H}^2 = \int_{\Gamma}q_2 d\mathcal{H}^2 = 0$

Then $\nabla_{\Gamma}(q_1 - q_2) = 0$ in $H^{-1}(\Gamma, T\Gamma)$ by (5.9.21) and thus $q_1 - q_2$ is constant on Γ by Lemma 5.9.2. Since $\int_{\Gamma}(q_1 - q_2) d\mathcal{H}^2 = 0$, the constant $q_1 - q_2$ is equal to zero, i.e. $q_1 = q_2$ on Γ . Hence the uniqueness is also valid. Finally, the estimate (5.9.20) follows from (5.9.19) with $\int_{\Gamma} q d\mathcal{H}^2 = 0$ and (5.9.21).

5.9.3 Weighted Helmholtz-Leray decomposition of tangential vector fields

The aim of this subsection is to establish the weighted Helmholtz–Leray decomposition of $L^2(\Gamma, T\Gamma)$ and give estimates for the gradient part of the decomposition. For $v \in L^2(\Gamma)^3$ we consider $\operatorname{div}_{\Gamma}(gv)$ as an element of $H^{-1}(\Gamma)$ given by (5.2.25). We define a subspace of $L^2(\Gamma, T\Gamma)$ by

$$L^2_{q\sigma}(\Gamma, T\Gamma) := \{ v \in L^2(\Gamma, T\Gamma) \mid \operatorname{div}_{\Gamma}(gv) = 0 \text{ in } H^{-1}(\Gamma) \}.$$

By (5.2.25) and $g \in C^1(\Gamma)$ we easily deduce that

$$\|\operatorname{div}_{\Gamma}(gv)\|_{H^{-1}(\Gamma)} \le c \|v\|_{L^{2}(\Gamma)} \quad \text{for all} \quad v \in L^{2}(\Gamma)^{3}.$$

Hence $L^2_{g\sigma}(\Gamma, T\Gamma)$ is closed in $L^2(\Gamma, T\Gamma)$ (note that $L^2(\Gamma, T\Gamma)$ is closed in $L^2(\Gamma)$). Let $L^2_{a\sigma}(\Gamma, T\Gamma)^{\perp}$ be the orthogonal complement of $L^2_{a\sigma}(\Gamma, T\Gamma)$ in $L^2(\Gamma, T\Gamma)$.

Lemma 5.9.10. The orthogonal complement of $L^2_{a\sigma}(\Gamma, T\Gamma)$ is of the form

$$L^2_{g\sigma}(\Gamma, T\Gamma)^{\perp} = \{g\nabla_{\Gamma}q \in L^2(\Gamma, T\Gamma) \mid q \in H^1(\Gamma)\}$$

Proof. Let $\mathcal{X} := \{ g \nabla_{\Gamma} q \in L^2(\Gamma, T\Gamma) \mid q \in H^1(\Gamma) \}$. By (5.2.25) we have

$$(v, g\nabla_{\Gamma} q)_{L^{2}(\Gamma)} = (gv, \nabla_{\Gamma} q)_{L^{2}(\Gamma)} = -\langle \operatorname{div}_{\Gamma}(gv), q \rangle_{\Gamma} = 0$$

for all $v \in L^2_{g\sigma}(\Gamma, T\Gamma)$ and $q \in H^1(\Gamma)$, which shows $\mathcal{X} \subset L^2_{g\sigma}(\Gamma, T\Gamma)^{\perp}$. Conversely, let $f \in L^2_{g\sigma}(\Gamma, T\Gamma)^{\perp}$. We consider f = Pf in $H^{-1}(\Gamma, T\Gamma)$ (see Section 5.2.1) to get

$$[f,v]_{T\Gamma} = (f,v)_{L^2(\Gamma)} = 0 \quad \text{for all} \quad v \in H^1_{g\sigma}(\Gamma,T\Gamma) \subset L^2_{g\sigma}(\Gamma,T\Gamma).$$

Hence by Theorem 5.9.5 there exists $q \in L^2(\Gamma)$ such that $f = g \nabla_{\Gamma} q$ in $H^{-1}(\Gamma, T\Gamma)$. To prove $q \in H^1(\Gamma)$ let $\{e_1, e_2, e_3\}$ be the standard basis of \mathbb{R}^3 and $v := g^{-1}\eta Pe_i$ for $\eta \in C^1(\Gamma)$ and i = 1, 2, 3. Then $v \in H^1(\Gamma, T\Gamma)$ since P and g are of class C^1 and $g \ge c > 0$ on Γ . Moreover,

$$\operatorname{div}_{\Gamma}(gv) = \nabla_{\Gamma}\eta \cdot Pe_i + \eta(\operatorname{div}_{\Gamma}P \cdot e_i) = \underline{D}_i\eta + \eta Hn_i \quad \text{on} \quad \Gamma$$

by the symmetry of P, (5.2.3), and (5.2.7). Hence we get

$$-(q,\underline{D}_i\eta + \eta Hn_i)_{L^2(\Gamma)} = -(q,\operatorname{div}_{\Gamma}(gv))_{L^2(\Gamma)} = [g\nabla_{\Gamma}q,v]_{T\Gamma}$$
$$= [f,v]_{T\Gamma} = (f,v)_{L^2(\Gamma)} = (g^{-1}f_i,\eta)_{L^2(\Gamma)}$$

for all $\eta \in C^1(\Gamma)$, where f_i is the *i*-th component of f (note that f is tangential on Γ). From this equality and the definition of the weak tangential derivative in $L^2(\Gamma)$ (see (5.2.19)) it follows that $\underline{D}_i q = g^{-1} f_i \in L^2(\Gamma)$ for i = 1, 2, 3. Hence we get $q \in H^1(\Gamma)$ and $f = g \nabla_{\Gamma} q \in \mathcal{X}$, and the inclusion $L^2_{q\sigma}(\Gamma, \Gamma\Gamma)^{\perp} \subset \mathcal{X}$ holds.

For $q_1, q_2 \in H^1(\Gamma)$ we have $g\nabla_{\Gamma}q_1 = g\nabla_{\Gamma}q_2$ on Γ if and only if $q_1 - q_2$ is constant on Γ by (5.2.30) and Lemma 5.9.2. By this fact and Lemma 5.9.10 we obtain the weighted Helmholtz-Leray decomposition of tangential vector fields on Γ .

Theorem 5.9.11. For each $v \in L^2(\Gamma, T\Gamma)$ we have the orthogonal decomposition

$$v = v_q + g \nabla_{\Gamma} q, \quad v_q \in L^2_{a\sigma}(\Gamma, T\Gamma), \ g \nabla_{\Gamma} q \in L^2_{a\sigma}(\Gamma, T\Gamma)^{\perp}.$$

Here $q \in H^1(\Gamma)$ is uniquely determined up to a constant.

Next we consider approximation of surface solenoidal vector fields. In general, for $v \in L^2(\Gamma, T\Gamma)$ satisfying $\operatorname{div}_{\Gamma} v = 0$ in $H^{-1}(\Gamma)$ and $\eta \in C^1(\Gamma)$, the surface divergence $\operatorname{div}_{\Gamma}(\eta v) = \nabla_{\Gamma} \eta \cdot v$ does not vanish in $H^{-1}(\Gamma)$. Hence standard localization and mollification arguments with a partition of unity of Γ do not work on approximation of surface solenoidal vector fields by smooth ones. Instead, we use a solution to Poisson's equation on Γ .

Lemma 5.9.12. For each $\eta \in H^{-1}(\Gamma)$ satisfying $\langle \eta, 1 \rangle_{\Gamma} = 0$ there exists a unique weak solution $q \in H^{1}(\Gamma)$ to Poisson's equation

$$\Delta_{\Gamma}q = -\eta \quad on \quad \Gamma, \quad \int_{\Gamma} q \, d\mathcal{H}^2 = 0 \tag{5.9.23}$$

in the sense that

$$(\nabla_{\Gamma}q, \nabla_{\Gamma}\xi)_{L^{2}(\Gamma)} = \langle \eta, \xi \rangle_{\Gamma} \quad for \ all \quad \xi \in H^{1}(\Gamma).$$
(5.9.24)

Moreover, there exists a constant c > 0 such that

$$\|q\|_{H^1(\Gamma)} \le c \|\eta\|_{H^{-1}(\Gamma)}.$$
(5.9.25)

If in addition $\eta \in L^2(\Gamma)$, then $q \in H^2(\Gamma)$ and

$$\|q\|_{H^2(\Gamma)} \le c \|\eta\|_{L^2(\Gamma)}.$$
(5.9.26)

The existence and uniqueness of a weak solution to (5.9.23) and the estimate (5.9.25) follow from Poincaré's inequality (5.2.20) and the Lax–Milgram theorem. Also, the H^2 -regularity and (5.9.26) are proved by a standard localization argument and the elliptic regularity theorem. For details, see [11, Theorems 3.1 and 3.3].

Lemma 5.9.13. The space $H^1_{q\sigma}(\Gamma, T\Gamma)$ is dense in $L^2_{q\sigma}(\Gamma, T\Gamma)$.

Proof. Let $v \in L^2_{g\sigma}(\Gamma, T\Gamma)$. By Lemma 5.2.6 we can take a sequence $\{\tilde{v}_k\}_{k=1}^{\infty}$ in $C^1(\Gamma, T\Gamma)$ that converges to v strongly in $L^2(\Gamma, T\Gamma)$. For each $k \in \mathbb{N}$ we have

$$|\operatorname{div}_{\Gamma}(g\tilde{v}_{k})||_{H^{-1}(\Gamma)} = \|\operatorname{div}_{\Gamma}[g(\tilde{v}_{k} - v)]\|_{H^{-1}(\Gamma)} \le c\|\tilde{v}_{k} - v\|_{L^{2}(\Gamma)}$$
(5.9.27)

by $\operatorname{div}_{\Gamma}(gv) = 0$ in $H^{-1}(\Gamma)$ and (5.2.25). We consider Poisson's equation (5.9.23) with source term $\eta_k := -\operatorname{div}_{\Gamma}(g\tilde{v}_k) \in L^2(\Gamma)$. By Lemma 5.9.12 there exists a unique solution $q_k \in H^2(\Gamma)$ to (5.9.23). Moreover, by (5.9.25) and (5.9.27) we have

$$\|q_k\|_{H^1(\Gamma)} \le c \|\eta_k\|_{H^{-1}(\Gamma)} = c \|\operatorname{div}_{\Gamma}(g\tilde{v}_k)\|_{H^{-1}(\Gamma)} \le c \|v - \tilde{v}_k\|_{L^2(\Gamma)}.$$

Hence $v_k := \tilde{v}_k - g^{-1} \nabla_{\Gamma} q_k \in H^1_{g\sigma}(\Gamma, T\Gamma)$ and

$$\|v - v_k\|_{L^2(\Gamma)} \le \|v - \tilde{v}_k\|_{L^2(\Gamma)} + c\|q_k\|_{H^1(\Gamma)} \le c\|v - \tilde{v}_k\|_{L^2(\Gamma)} \to 0$$

as $k \to \infty$ by $g \ge c > 0$ on Γ and the strong convergence of $\{\tilde{v}_k\}_{k=1}^{\infty}$ to v in $L^2(\Gamma, T\Gamma)$, which shows that $H^1_{q\sigma}(\Gamma, T\Gamma)$ is dense in $L^2_{q\sigma}(\Gamma, T\Gamma)$. Let \mathbb{P}_g be the orthogonal projection from $L^2(\Gamma, T\Gamma)$ onto $L^2_{g\sigma}(\Gamma, T\Gamma)$. We call it the weighted Helmholtz-Leray projection. In the study of a singular limit problem for (5.1.1)–(5.1.3) we need to estimate the difference $v - \mathbb{P}_g v$ for $v \in L^2(\Gamma, T\Gamma)$.

Lemma 5.9.14. There exists a constant c > 0 such that

$$\|v - \mathbb{P}_{g}v\|_{L^{2}(\Gamma)} \le c \|\operatorname{div}_{\Gamma}(gv)\|_{H^{-1}(\Gamma)}$$
(5.9.28)

for all $v \in L^2(\Gamma, T\Gamma)$. If in addition $v \in H^1(\Gamma, T\Gamma)$, then $\mathbb{P}_g v \in H^1_{q\sigma}(\Gamma, T\Gamma)$ and

$$\|v - \mathbb{P}_{g}v\|_{H^{1}(\Gamma)} \le c \|\operatorname{div}_{\Gamma}(gv)\|_{L^{2}(\Gamma)}.$$
(5.9.29)

Proof. Let $v \in L^2(\Gamma, T\Gamma)$ and $\eta := -\operatorname{div}_{\Gamma}(gv) \in H^{-1}(\Gamma)$. Since v is tangential on Γ , we see by (5.2.25) that

$$\langle \eta, 1 \rangle_{\Gamma} = (gv, Hn)_{L^2(\Gamma)} = \int_{\Gamma} g(v \cdot n) H \, d\mathcal{H}^2 = 0.$$

Hence by Lemma 5.9.12 there exists a unique weak solution $q \in H^1(\Gamma)$ to (5.9.23) with $\eta = -\text{div}_{\Gamma}(gv)$. Then since $v - g^{-1}\nabla_{\Gamma}q \in L^2_{g\sigma}(\Gamma, T\Gamma)$, we have $\mathbb{P}_g v = v - g^{-1}\nabla_{\Gamma}q$ by the uniqueness of the weighted Helmholtz-Leray decomposition. Moreover, since

$$\|q\|_{H^{1}(\Gamma)} \leq c \|\eta\|_{H^{-1}(\Gamma)} = c \|\operatorname{div}_{\Gamma}(gv)\|_{H^{-1}(\Gamma)}$$

by (5.9.25) and $g \in C^1(\Gamma)$ is bounded from below by a positive constant, we obtain

$$\|v - \mathbb{P}_g v\|_{L^2(\Gamma)} = \|g^{-1} \nabla_{\Gamma} q\|_{L^2(\Gamma)} \le c \|q\|_{H^1(\Gamma)} \le c \|\operatorname{div}_{\Gamma}(gv)\|_{H^{-1}(\Gamma)}$$

Hence the inequality (5.9.28) holds.

Next suppose that $v \in H^1(\Gamma, T\Gamma)$. Then since $\eta = -\operatorname{div}_{\Gamma}(gv) \in L^2(\Gamma)$ Lemma 5.9.12 implies that that $q \in H^2(\Gamma)$ and thus $\mathbb{P}_g v = v - g^{-1} \nabla_{\Gamma} q \in H^1_{q\sigma}(\Gamma, T\Gamma)$. Moreover, since

$$||q||_{H^2(\Gamma)} \le c ||\eta||_{L^2(\Gamma)} = c ||\operatorname{div}_{\Gamma}(gv)||_{L^2(\Gamma)}$$

by (5.9.26) and $g \in C^1(\Gamma)$ is bounded from below by a positive constant,

$$\|v - \mathbb{P}_g v\|_{H^1(\Gamma)} = \|g^{-1} \nabla_{\Gamma} q\|_{H^1(\Gamma)} \le c \|q\|_{H^2(\Gamma)} \le c \|\operatorname{div}_{\Gamma}(gv)\|_{L^2(\Gamma)},$$

i.e. the inequality (5.9.29) is valid.

Corollary 5.9.15. There exists a constant c > 0 such that

$$\|\mathbb{P}_{g}v\|_{H^{k}(\Gamma)} \le c\|v\|_{H^{k}(\Gamma)} \tag{5.9.30}$$

for all $v \in H^k(\Gamma, T\Gamma)$, k = 0, 1 (note that $H^0 = L^2$).

Proof. If k = 0, then (5.9.30) holds with c = 1 since \mathbb{P}_g is the orthogonal projection from $L^2(\Gamma, T\Gamma)$ onto $L^2_{g\sigma}(\Gamma, T\Gamma)$. Moreover, the inequality (5.9.30) for k = 1 follows from (5.9.29) and $\|\operatorname{div}_{\Gamma}(gv)\|_{L^2(\Gamma)} \leq c \|v\|_{H^1(\Gamma)}$.

Next we consider the time derivative of $v - \mathbb{P}_g v$. We derive an estimate for the time derivative of a weak solution to Poisson's equation (5.9.23).

Lemma 5.9.16. Let T > 0. Suppose that $\eta \in H^1(0,T; H^{-1}(\Gamma))$ satisfies

$$\langle \eta(t), 1 \rangle_{\Gamma} = 0 \quad for \ all \quad t \in [0, T].$$

For each $t \in [0,T]$ let $q(t) \in H^1(\Gamma)$ be a unique weak solution to (5.9.23) with source term $\eta(t)$. Then $q \in H^1(0,T; H^1(\Gamma))$ and there exists a constant c > 0 such that

$$\|\partial_t q\|_{L^2(0,T;H^1(\Gamma))} \le c \|\partial_t \eta\|_{L^2(0,T;H^{-1}(\Gamma))}.$$
(5.9.31)

Moreover, for a.a. $t \in (0,T)$ the time derivative $\partial_t q(t) \in H^1(\Gamma)$ is a unique weak solution to (5.9.23) with source term $\partial_t \eta(t)$.

Note that, when $\eta \in H^1(0,T; H^{-1}(\Gamma))$, $\eta(t) \in H^{-1}(\Gamma)$ is well-defined for each $t \in [0,T]$ since $H^1(0,T; H^{-1}(\Gamma))$ is continuously embedded into $C([0,T]; H^{-1}(\Gamma))$.

Proof. First note that $q \in L^2(0,T; H^1(\Gamma))$ by (5.9.25) and $\eta \in L^2(0,T; H^{-1}(\Gamma))$. Let us prove $\partial_t q \in L^2(0,T; H^1(\Gamma))$ by means of the difference quotient. Fix $\delta \in (0,T/2)$ and $h \in \mathbb{R} \setminus \{0\}$ with $|h| < \delta/2$. For $t \in (\delta, T - \delta)$ we define

$$D_h q(t) := \frac{q(t+h) - q(t)}{h} \in H^1(\Gamma), \quad D_h \eta(t) := \frac{\eta(t+h) - \eta(t)}{h} \in H^{-1}(\Gamma).$$

Note that these definitions make sense since $t + h \in (\delta/2, T - \delta/2)$. Moreover,

$$\int_{\Gamma} D_h q(t) \, d\mathcal{H}^2 = 0, \quad \langle D_h \eta(t), 1 \rangle_{\Gamma} = 0, \quad t \in (\delta, T - \delta)$$

since q(t) and $\eta(t)$ satisfy the same equalities for all $t \in [0, T]$. For $\xi \in H^1(\Gamma)$ we subtract (5.9.24) for q(t) from that for q(t+h) and divide both sides by h to get

$$(\nabla_{\Gamma} D_h q(t), \nabla_{\Gamma} \xi)_{L^2(\Gamma)} = \langle D_h \eta(t), \xi \rangle_{\Gamma}.$$
(5.9.32)

Since this equality holds for all $\xi \in H^1(\Gamma)$, the function $D_h q(t)$ is a unique weak solution to (5.9.23) with source term $D_h \eta(t)$. Hence by (5.9.25) we have

$$||D_h q(t)||_{H^1(\Gamma)} \le c ||D_h \eta(t)||_{H^{-1}(\Gamma)}, \quad t \in (\delta, T - \delta).$$

Note that the constant c > 0 in this inequality does not depend on t, δ , and h. From this inequality it immediately follows that

$$\|D_h q\|_{L^2(\delta, T-\delta; H^1(\Gamma))} \le c \|D_h \eta\|_{L^2(\delta, T-\delta; H^{-1}(\Gamma))}$$

Moreover, since $\eta \in H^1(0,T; H^{-1}(\Gamma))$, we have

$$\|D_h\eta\|_{L^2(\delta,T-\delta;H^{-1}(\Gamma))} \le c\|\partial_t\eta\|_{L^2(0,T;H^{-1}(\Gamma))}$$

with a constant c > 0 independent of h and δ (see [13, Section 5.8, Theorem 3 (i)]). Combining the above two estimates we obtain

$$\|D_h q\|_{L^2(\delta, T-\delta; H^1(\Gamma))} \le c \|\partial_t \eta\|_{L^2(0,T; H^{-1}(\Gamma))}$$

for all $h \in \mathbb{R} \setminus \{0\}$ with $|h| < \delta/2$. Since the right-hand side of this inequality is independent of h, it follows that $\partial_t q \in L^2(\delta, T - \delta; H^1(\Gamma))$ and

$$\|\partial_t q\|_{L^2(\delta, T-\delta; H^1(\Gamma))} \le c \|\partial_t \eta\|_{L^2(0,T; H^{-1}(\Gamma))}$$

for all $\delta \in (0, T/2)$ (see [13, Section 5.8, Theorem 3 (ii)]). In particular, we have $\partial_t q(t) \in H^1(\Gamma)$ for a.a. $t \in (0, T)$. Since the right-hand side of the above inequality is independent of δ , the monotone convergence theorem yields

$$\|\partial_t q\|_{L^2(0,T;H^1(\Gamma))} = \lim_{\delta \to 0} \|\partial_t q\|_{L^2(\delta, T-\delta;H^1(\Gamma))} \le c \|\partial_t \eta\|_{L^2(0,T;H^{-1}(\Gamma))}$$

and thus $\partial_t q \in L^2(0,T; H^1(\Gamma))$ and (5.9.31) is valid.

Next we show that $\partial_t q(t)$ is a unique weak solution to (5.9.23) with $\partial_t \eta(t)$ for a.a. $t \in (0,T)$. Let $\xi \in H^1(\Gamma)$ and $\varphi \in C_c^{\infty}(0,T)$. Suppose that φ is supported in $(\delta, T - \delta)$ with $\delta \in (0,T/2)$. We extend φ to \mathbb{R} by setting zero outside of (0,T). For $h \in \mathbb{R} \setminus \{0\}, |h| < \delta/2$ we multiply both sides of (5.9.32) by $\varphi(t)$, integrate them over $(\delta, T - \delta)$, and make the change of a variable

$$\int_{\delta}^{T-\delta} \psi(t+h)\varphi(t) \, dt = \int_{\delta+h}^{T-\delta+h} \psi(s)\varphi(s-h) \, ds$$

for $\psi(t) = (\nabla_{\Gamma} q(t), \nabla_{\Gamma} \xi)_{L^2(\Gamma)}, \langle \eta(t), \xi \rangle_{\Gamma}$ to get

$$-\int_0^T (\nabla_{\Gamma} q(t), \nabla_{\Gamma} \xi)_{L^2(\Gamma)} D_{-h} \varphi(t) \, dt = -\int_0^T \langle \eta(t), \xi \rangle_{\Gamma} D_{-h} \varphi(t) \, dt,$$

where $D_{-h}\varphi(t) := \{\varphi(t-h) - \varphi(t)\}/(-h)$ (note that φ is supported in $(\delta, T-\delta)$). Let $h \to 0$ in this equality. Then since $D_{-h}\varphi$ converges to $\partial_t\varphi$ uniformly on (0,T),

$$-\int_0^T (\nabla_{\Gamma} q(t), \nabla_{\Gamma} \xi)_{L^2(\Gamma)} \partial_t \varphi(t) \, dt = -\int_0^T \langle \eta(t), \xi \rangle_{\Gamma} \partial_t \varphi(t) \, dt$$

for all $\varphi \in C_c^{\infty}(0,T)$. By this equality, $q \in H^1(0,T;H^1(\Gamma))$, and $\eta \in H^1(0,T;H^{-1}(\Gamma))$,

$$([\nabla_{\Gamma}(\partial_t q)](t), \nabla_{\Gamma}\xi)_{L^2(\Gamma)} = \langle \partial_t \eta(t), \xi \rangle_{\Gamma}$$

for all $\xi \in H^1(\Gamma)$ and a.a. $t \in (0,T)$. Here we note that $\partial_t(\nabla_{\Gamma} q) = \nabla_{\Gamma}(\partial_t q)$ a.e. on $\Gamma \times (0,T)$ by $q \in H^1(0,T; H^1(\Gamma))$. In the same way we can show $\int_{\Gamma} \partial_t q(t) d\mathcal{H}^2 = 0$ for a.a. $t \in (0,T)$ since q(t) satisfies the same equality for all $t \in [0,T]$. Hence $\partial_t q(t)$ is a unique weak solution to (5.9.23) with $\partial_t \eta(t)$ for a.a. $t \in (0,T)$.

Based on Lemma 5.9.16 we give an estimate for the time derivative of $v - \mathbb{P}_g v$.

Lemma 5.9.17. Let $v \in H^1(0,T;L^2(\Gamma,T\Gamma)), T > 0$. Then

$$\mathbb{P}_g v \in H^1(0,T; L^2_{g\sigma}(\Gamma,T\Gamma))$$

and there exists a constant c > 0 such that

$$\|\partial_t v - \partial_t \mathbb{P}_g v\|_{L^2(0,T;L^2(\Gamma))} \le c \|\operatorname{div}_{\Gamma}(g\partial_t v)\|_{L^2(0,T;H^{-1}(\Gamma))}.$$
(5.9.33)

Proof. Let $\eta := -\operatorname{div}_{\Gamma}(gv)$. Since $v \in H^1(0,T; L^2(\Gamma,T\Gamma))$, we have

$$\eta \in H^1(0,T; H^{-1}(\Gamma)), \quad \partial_t \eta = -\operatorname{div}_{\Gamma}(g\partial_t v) \in L^2(0,T; H^{-1}(\Gamma)).$$

Here the second relation is due to the fact that g and P are independent of time (note that P appears in the definition of the tangential derivatives). For each $t \in [0, T]$ let $q(t) \in H^1(\Gamma)$

be a unique weak solution to (5.9.23) with $\eta(t) = -\text{div}_{\Gamma}[gv(t)]$, which satisfies $\langle \eta(t), 1 \rangle_{\Gamma} = 0$ as in the proof of Lemma 5.9.14. Then Lemma 5.9.16 implies that $q \in H^1(0, T; H^1(\Gamma))$ and

$$\|\partial_t q\|_{L^2(0,T;H^1(\Gamma))} \le c \|\partial_t \eta\|_{L^2(0,T;H^{-1}(\Gamma))} = c \|\operatorname{div}_{\Gamma}(g\partial_t v)\|_{L^2(0,T;H^{-1}(\Gamma))}.$$

Moreover, for a.a. $t \in (0, T)$ the time derivative $\partial_t q(t) \in H^1(\Gamma)$ is a unique weak solution to (5.9.23) with $\partial_t \eta(t) = -\text{div}_{\Gamma}[g\partial_t v(t)]$. By these facts and

$$\mathbb{P}_g v = v - g^{-1} \nabla_{\Gamma} q, \quad \partial_t \mathbb{P}_g v = \partial_t v - g^{-1} \nabla_{\Gamma} (\partial_t q) \quad \text{a.e. on} \quad \Gamma \times (0, T)$$

we observe that $\mathbb{P}_{g}v \in H^{1}(0,T; L^{2}_{q\sigma}(\Gamma,T\Gamma))$ and

$$\begin{aligned} \|\partial_t v - \partial_t \mathbb{P}_g v\|_{L^2(0,T;L^2(\Gamma))} &= \|g^{-1} \nabla_{\Gamma}(\partial_t q)\|_{L^2(0,T;L^2(\Gamma))} \le c \|\partial_t q\|_{L^2(0,T;H^1(\Gamma))} \\ &\le c \|\operatorname{div}_{\Gamma}(g\partial_t v)\|_{L^2(0,T;H^{-1}(\Gamma))}, \end{aligned}$$

where we also used the inequality $g \ge c > 0$ on Γ . Hence the lemma is valid.

5.9.4 Solenoidal spaces of general vector fields

In this subsection we briefly investigate solenoidal spaces of general (not necessarily tangential) vector fields on Γ . Although the results of this subsection are not used in the sequel, we believe that they are useful for the future study of surface fluid equations including fluid equations on an evolving surface (see e.g. [28, 31, 32, 43]). For the sake of simplicity, we only consider the case $g \equiv 1$ and give a remark on the case of general g at the end of this subsection.

Let $q \in L^2(\Gamma)$. By (5.2.21) and (5.2.24) we have

$$\langle \nabla_{\Gamma} q + qHn, v \rangle_{\Gamma} = -(q, \operatorname{div}_{\Gamma} v)_{L^{2}(\Gamma)}$$
(5.9.34)

for all $v \in H^1(\Gamma)^3$. Hence $\langle \nabla_{\Gamma} q + qHn, v \rangle_{\Gamma} = 0$ for all v in the solenoidal space

$$H^1_{\sigma}(\Gamma) := \{ v \in H^1(\Gamma)^3 \mid \operatorname{div}_{\Gamma} v = 0 \text{ on } \Gamma \}.$$

Our goal is to prove that each element of the annihilator of $H^1_{\sigma}(\Gamma)$ is of the form $\nabla_{\Gamma} q + qHn$. To this end, we give two properties of a functional of this form.

Lemma 5.9.18. Let $q \in L^2(\Gamma)$. Then

$$\nabla_{\Gamma} q + qHn = 0 \quad in \quad H^{-1}(\Gamma)^3$$

if and only if q = 0 on Γ .

Proof. We first note that for all $q \in L^2(\Gamma)$ we have

$$\|\nabla_{\Gamma} q\|_{H^{-1}(\Gamma, T\Gamma)} \le \|\nabla_{\Gamma} q + q H n\|_{H^{-1}(\Gamma)}$$
(5.9.35)

since $[\nabla_{\Gamma}q, v]_{T\Gamma} = \langle \nabla_{\Gamma}q + qHn, v \rangle_{\Gamma}$ for all $v \in H^1(\Gamma, T\Gamma)$ (see Section 5.2.1).

Suppose that $\nabla_{\Gamma}q + qHn = 0$ in $H^{-1}(\Gamma)^3$. Then $\nabla_{\Gamma}q = 0$ in $H^{-1}(\Gamma, T\Gamma)$ by (5.9.35) and thus q is constant on Γ by Lemma 5.9.2. To prove q = 0, we set $v := \xi n$ in (5.9.34) for an arbitrary $\xi \in H^1(\Gamma)$ (note that n is of class C^1 on Γ) to get

$$0 = \langle \nabla_{\Gamma} q + qHn, v \rangle_{\Gamma} = q \int_{\Gamma} \xi H \, d\mathcal{H}^2.$$

Since $H \in C(\Gamma) \subset L^2(\Gamma)$ and $H^1(\Gamma)$ is dense in $L^2(\Gamma)$ (see Lemma 5.2.4), we observe by the above equality and a density argument that

$$q \int_{\Gamma} H^2 \, d\mathcal{H}^2 = 0.$$

Moreover, for the compact surface Γ in \mathbb{R}^3 it is known (see e.g. (16.32) in [15]) that

$$\frac{1}{4} \int_{\Gamma} H^2 \, d\mathcal{H}^2 \ge 4\pi.$$

(Note that in our definition H is not divided by the dimension of Γ .) Hence q = 0.

Conversely, if q = 0 on Γ , then $\nabla_{\Gamma} q + qHn = 0$ in $H^{-1}(\Gamma)^3$ by (5.9.34).

Lemma 5.9.19. There exists a constant c > 0 such that

$$\|q\|_{L^{2}(\Gamma)} \leq c \|\nabla_{\Gamma} q + qHn\|_{H^{-1}(\Gamma)}$$
(5.9.36)

for all $q \in L^2(\Gamma)$.

Proof. By the Nečas inequality (5.9.1) and (5.9.35) it is sufficient to show that

$$\|q\|_{H^{-1}(\Gamma)} \le c \|\nabla_{\Gamma} q + q H n\|_{H^{-1}(\Gamma)}$$
(5.9.37)

for all $q \in L^2(\Gamma)$. Assume to the contrary that there exists $q_k \in L^2(\Gamma)$ such that

$$\|q_k\|_{H^{-1}(\Gamma)} > k\|\nabla_{\Gamma} q_k + q_k H n\|_{H^{-1}(\Gamma)}$$
(5.9.38)

for each $k \in \mathbb{N}$. Since $||q_k||_{H^{-1}(\Gamma)} \neq 0$, we may assume that $||q_k||_{H^{-1}(\Gamma)} = 1$ for all $k \in \mathbb{N}$ by replacing q_k with $q_k/||q_k||_{H^{-1}(\Gamma)}$. Then we observe by (5.9.1), (5.9.35), and (5.9.38) that $\{q_k\}_{k=1}^{\infty}$ is bounded in $L^2(\Gamma)$. By this fact and the compact embedding $L^2(\Gamma) \hookrightarrow H^{-1}(\Gamma)$ we can take a subsequence of $\{q_k\}_{k=1}^{\infty}$, which we denote by $\{q_k\}_{k=1}^{\infty}$ again, that converges to some $q \in L^2(\Gamma)$ weakly in $L^2(\Gamma)$ and strongly in $H^{-1}(\Gamma)$. Moreover, the weak convergence of $\{q_k\}_{k=1}^{\infty}$ to q in $L^2(\Gamma)$ and (5.9.34) imply that

$$\lim_{k \to \infty} (\nabla_{\Gamma} q_k + q_k H n) = \nabla_{\Gamma} q + q H n \quad \text{weakly in} \quad H^{-1}(\Gamma)^3.$$

By this fact, (5.9.38), and $||q_k||_{H^{-1}(\Gamma)} = 1$ we have

$$\|\nabla_{\Gamma} q + qHn\|_{H^{-1}(\Gamma)} \le \liminf_{k \to \infty} \|\nabla_{\Gamma} q_k + q_k Hn\|_{H^{-1}(\Gamma)} = 0.$$

Hence $\nabla_{\Gamma} q + qHn = 0$ in $H^{-1}(\Gamma)$ and q = 0 on Γ by Lemma 5.9.18. However, the strong convergence of $\{q_k\}_{k=1}^{\infty}$ to q in $H^{-1}(\Gamma)$ implies that

$$\|q\|_{H^{-1}(\Gamma)} = \lim_{k \to \infty} \|q_k\|_{H^{-1}(\Gamma)} = 1,$$
(5.9.39)

which contradicts with q = 0. Therefore, the inequality (5.9.37) is valid.

Now we prove de Rham's theorem for the annihilator of $H^1_{\sigma}(\Gamma)$.

Theorem 5.9.20. Suppose that $f \in H^{-1}(\Gamma)^3$ satisfies

$$\langle f, v \rangle_{\Gamma} = 0 \quad for \ all \quad v \in H^1_{\sigma}(\Gamma).$$

Then there exists a unique $q \in L^2(\Gamma)$ such that

$$f = \nabla_{\Gamma} q + qHn$$
 in $H^{-1}(\Gamma)^3$, $\|q\|_{L^2(\Gamma)} \le c \|f\|_{H^{-1}(\Gamma)}$

with a constant c > 0 independent of f.

Proof. Using (5.9.36) we can show as in the proof of Lemma 5.9.7 that the subspace

$$\mathcal{X} := \{ \nabla_{\Gamma} q + qHn \in H^{-1}(\Gamma)^3 \mid q \in L^2(\Gamma) \}$$

is closed in $H^{-1}(\Gamma)^3$. Moreover, by (5.9.34) we easily observe that $\mathcal{X}^{\perp} \subset H^1_{\sigma}(\Gamma)$ in $H^1(\Gamma)^3$. Noting that the dual $H^{-1}(\Gamma)^3$ of the Hilbert space $H^1(\Gamma)^3$ is reflexive, we use Lemmas 5.9.8 and 5.9.9 to obtain

$$H^1_{\sigma}(\Gamma)^{\perp} = \{ f \in H^{-1}(\Gamma)^3 \mid \langle f, v \rangle_{\Gamma} = 0 \text{ for all } v \in H^1_{\sigma}(\Gamma) \} \subset (\mathcal{X}^{\perp})^{\perp} = \mathcal{X}$$

in $H^{-1}(\Gamma)^3$. Hence the existence part of the theorem is valid. Also, the uniqueness and the estimate immediately follow from Lemma 5.9.19.

Next we derive the Helmholtz–Leray decomposition of general vector fields on Γ . We define a subspace of $L^2(\Gamma)^3$ by

$$L^2_{\sigma}(\Gamma) := \{ v \in L^2(\Gamma)^3 \mid \operatorname{div}_{\Gamma} v = 0 \text{ in } H^{-1}(\Gamma) \}.$$

By (5.2.25) we have $\|\operatorname{div}_{\Gamma} v\|_{H^{-1}(\Gamma)} \leq c \|v\|_{L^2(\Gamma)}$ for all $v \in L^2(\Gamma)^3$. Hence $L^2_{\sigma}(\Gamma)$ is closed in $L^2(\Gamma)^3$. Let us give the characterization of the orthogonal complement of $L^2_{\sigma}(\Gamma)$ in $L^2(\Gamma)^3$.

Lemma 5.9.21. The orthogonal complement of $L^2_{\sigma}(\Gamma)$ in $L^2(\Gamma)^3$ is of the form

$$L^2_{\sigma}(\Gamma)^{\perp} = \{ \nabla_{\Gamma} q + qHn \in L^2(\Gamma)^3 \mid q \in H^1(\Gamma) \}.$$

Proof. The proof is similar to that of Lemma 5.9.10. By (5.2.25) we immediately get $\nabla_{\Gamma}q + qHn \in L^2_{\sigma}(\Gamma)^{\perp}$ for all $q \in H^1(\Gamma)$. Conversely, let $f \in L^2_{\sigma}(\Gamma)^{\perp}$. Then we have $\langle f, v \rangle_{\Gamma} = 0$ for all $v \in H^1_{\sigma}(\Gamma) \subset L^2_{\sigma}(\Gamma)$ by (5.2.21). Hence by Theorem 5.9.20 there exists a unique $q \in L^2(\Gamma)$ such that $f = \nabla_{\Gamma}q + qHn$ in $H^{-1}(\Gamma)^3$. To prove $q \in H^1(\Gamma)$ let $v := \eta e_i$ for $\eta \in C^1(\Gamma)$ and i = 1, 2, 3, where $\{e_1, e_2, e_3\}$ is the standard basis of \mathbb{R}^3 . Then since $v \in H^1(\Gamma)^3$ and $\operatorname{div}_{\Gamma}v = \underline{D}_i\eta$, we have

$$-(q,\underline{D}_{i}\eta)_{L^{2}(\Gamma)} = -(q,\operatorname{div}_{\Gamma}v)_{L^{2}} = \langle \nabla_{\Gamma}q + qHn, v \rangle_{\Gamma}$$
$$= \langle f, v \rangle_{\Gamma} = (f,v)_{L^{2}(\Gamma)} = (f_{i},\eta)_{L^{2}(\Gamma)},$$

where f_i is the *i*-th component of f. From this equality we deduce that

$$-(q,\underline{D}_i\eta + \eta Hn_i)_{L^2(\Gamma)} = (f_i - qHn_i,\eta)_{L^2(\Gamma)} \text{ for all } \eta \in C^1(\Gamma).$$

This means that $\underline{D}_i q = f_i - qHn_i \in L^2(\Gamma)$ by the definition of the weak tangential derivative in $L^2(\Gamma)$ (see (5.2.19)). Hence we obtain $q \in H^1(\Gamma)$ and $f = \nabla_{\Gamma} q + qHn$ in $L^2(\Gamma)^3$.

The result of Lemma 5.9.21 was given in [31, Lemma 2.7] (see also [32, Theorem 1.1]). Here we gave another proof of it. By Lemmas 5.9.18 and 5.9.21 we obtain the Helmholtz– Leray decomposition of vector fields in $L^2(\Gamma)^3$ with uniqueness of the gradient part.

Theorem 5.9.22. For each $v \in L^2(\Gamma)^3$ we have the orthogonal decomposition

$$v = v_{\sigma} + \nabla_{\Gamma} q + qHn, \quad v_{\sigma} \in L^2_{\sigma}(\Gamma), \, \nabla_{\Gamma} q + qHn \in L^2_{\sigma}(\Gamma)^{\perp}$$

Here $q \in H^1(\Gamma)$ is uniquely determined.

The Helmholtz–Leray decomposition in Theorem 5.9.22 was already stated in [32] without an explicit formulation (see a remark after [32, Theorem 1.1]).

Remark 5.9.23. Theorem 5.9.22 applied to a tangential vector field on Γ does not imply the tangential Helmholtz–Leray decomposition (with $g \equiv 1$) given in Theorem 5.9.11 in general. To see this, suppose that Γ is strictly convex and thus the mean curvature H of Γ does not vanish on the whole surface. Let $v \in L^2(\Gamma, T\Gamma)$ be a tangential vector field on Γ such that $\operatorname{div}_{\Gamma} v \neq 0$ in $H^{-1}(\Gamma)$. By Theorem 5.9.22 we get the orthogonal decomposition

$$v = v_{\sigma} + \nabla_{\Gamma} q + qHn, \quad v_{\sigma} \in L^2_{\sigma}(\Gamma), \, \nabla_{\Gamma} q + qHn \in L^2_{\sigma}(\Gamma)^{\perp}$$

with a unique $q \in H^1(\Gamma)$. Since v is tangential on Γ ,

$$0 = v_{\sigma} \cdot n + qH, \quad \text{i.e.} \quad v_{\sigma} \cdot n = -qH \quad \text{on} \quad \Gamma.$$

Moreover, $q \neq 0$ in $H^1(\Gamma)$ by $\operatorname{div}_{\Gamma} v \neq 0$ in $H^{-1}(\Gamma)$. By this property and the fact that H does not vanish on the whole surface Γ by the strict convexity of Γ we see that $v_{\sigma} \cdot n = -qH \neq 0$ in $L^2(\Gamma)$. Hence the solenoidal part v_{σ} of v given by Theorem 5.9.22 is not tangential on Γ , while the solenoidal part v_g (with $g \equiv 1$) of the same v given by Theorem 5.9.11 must be tangential on Γ .

The vector field $\nabla_{\Gamma}q + qHn$ appears in the interface equations of two-phase flows [4,6,49] as well as the Navier–Stokes equations on an evolving surface [28,31]. By (5.2.7) we observe that the surface divergence of qP is of this form:

$$\operatorname{div}_{\Gamma}(qP) = P\nabla_{\Gamma}q + q\operatorname{div}_{\Gamma}P = \nabla_{\Gamma}q + qHn.$$

The tensor qP is a part of the Boussinesq–Scriven surface stress tensor [2, 7, 58]

$$S_{\Gamma} = \{q + (\lambda_s - \mu_s) \operatorname{div}_{\Gamma} v\} P + 2\mu_s D_{\Gamma}(v).$$

Here q is the surface tension, λ_s is the surface dilatational viscosity, μ_s is the surface shear viscosity, v is the total velocity of surface flow, and $D_{\Gamma}(v)$ is the surface strain rate tensor given by (5.4.38).

Finally, we give a remark on the case of general g. Let $g \in C^1(\Gamma)$ be bounded from below by a positive constant. We define weighted solenoidal spaces

$$L^{2}_{g\sigma}(\Gamma) := \{ v \in L^{2}(\Gamma)^{3} \mid \operatorname{div}_{\Gamma}(gv) = 0 \text{ in } H^{-1}(\Gamma) \},\$$

$$H^{1}_{g\sigma}(\Gamma) := \{ v \in H^{1}(\Gamma)^{3} \mid \operatorname{div}_{\Gamma}(gv) = 0 \text{ on } \Gamma \}.$$

By (5.2.22), (5.2.24), and (5.2.25) we have

$$\langle g(\nabla_{\Gamma}q + qHn), v \rangle_{\Gamma} = -(q, \operatorname{div}_{\Gamma}(gv))_{L^{2}(\Gamma)}, \qquad q \in L^{2}(\Gamma), \ v \in H^{1}(\Gamma)^{3}, \\ \langle \operatorname{div}_{\Gamma}(gw), \eta \rangle_{\Gamma} = -(w, g(\nabla_{\Gamma}\eta + \eta Hn))_{L^{2}(\Gamma)}, \quad w \in L^{2}(\Gamma)^{3}, \ \eta \in H^{1}(\Gamma).$$

Using these formulas and applying Theorem 5.9.20 or Lemma 5.9.21 to $g^{-1}f$ for f in $H^1_{g\sigma}(\Gamma)^{\perp}$ or $L^2_{g\sigma}(\Gamma)^{\perp}$, we can show the following weighted version of the main results in this subsection.

Theorem 5.9.24. Suppose that $f \in H^{-1}(\Gamma)^3$ satisfies

 $\langle f, v \rangle_{\Gamma} = 0 \quad for \ all \quad v \in H^1_{g\sigma}(\Gamma).$

Then there exists a unique $q \in L^2(\Gamma)$ such that

$$f = g(\nabla_{\Gamma}q + qHn)$$
 in $H^{-1}(\Gamma)^3$, $||q||_{L^2(\Gamma)} \le c||f||_{H^{-1}(\Gamma)}$

with a constant c > 0 independent of f.

Theorem 5.9.25. For each $v \in L^2(\Gamma)^3$ we have the orthogonal decomposition

$$v = v_g + g(\nabla_{\Gamma}q + qHn), \quad v_g \in L^2_{g\sigma}(\Gamma), \ g(\nabla_{\Gamma}q + qHn) \in L^2_{g\sigma}(\Gamma)^{\perp}.$$

Here $q \in H^1(\Gamma)$ is uniquely determined.

5.10 Singular limit problem on degeneration of a thin domain

In this section we study a singular limit problem for the Navier–Stokes equations (5.1.1)– (5.1.3) as the curved thin domain Ω_{ε} degenerates into the closed surface Γ . Our goal is to derive the limit system on Γ of (5.1.1)–(5.1.3) and compare it with the bulk system.

Throughout this section we impose Assumptions 1 and 2 and let ε_1 and ε_{σ} be the positive constants given in Theorem 5.1.2 and Lemma 5.5.1, respectively. We assume that $\varepsilon \in (0, \varepsilon'_1)$ with $\varepsilon'_1 := \min\{\varepsilon_1, \varepsilon_{\sigma}\}$ and the assumptions in Theorem 5.8.4 are satisfied. Also, we denote by $\bar{\eta} = \eta \circ \pi$ the constant extension of a function η on Γ to the normal direction of Γ .

5.10.1 Weak formulation for the bulk system

Our ansatz is a weak formulation for (5.1.1)–(5.1.3) satisfied by a strong solution. Let

$$u^{\varepsilon} \in C([0,\infty); V_{\varepsilon}) \cap L^2_{loc}([0,\infty); D(A_{\varepsilon})) \cap H^1_{loc}([0,\infty); L^2_{\sigma}(\Omega_{\varepsilon}))$$

be the global strong solution to (5.1.1)–(5.1.3) given by Theorem 5.8.4. It satisfies

$$\int_{0}^{T} \{ (\partial_{t} u^{\varepsilon}, \varphi)_{L^{2}(\Omega_{\varepsilon})} + a_{\varepsilon}(u^{\varepsilon}, \varphi) + b_{\varepsilon}(u^{\varepsilon}, u^{\varepsilon}, \varphi) \} dt = \int_{0}^{T} (\mathbb{P}_{\varepsilon} f^{\varepsilon}, \varphi)_{L^{2}(\Omega_{\varepsilon})} dt$$
(5.10.1)

for all T > 0 and $\varphi \in L^2(0, T; V_{\varepsilon})$, and $u^{\varepsilon}|_{t=0} = u_0^{\varepsilon}$ in V_{ε} . Here a_{ε} is the bilinear form given by (5.5.14), i.e.

$$a_{\varepsilon}(u_1, u_2) = 2\nu \int_{\Omega_{\varepsilon}} D(u_1) : D(u_2) \, dx + \sum_{i=0,1} \gamma_{\varepsilon}^i \int_{\Gamma_{\varepsilon}^i} u_1 \cdot u_2 \, d\mathcal{H}^2$$

for $u_1, u_2 \in H^1(\Omega_{\varepsilon})^3$ and b_{ε} is a trilinear form defined by

$$b_{\varepsilon}(u_1, u_2, u_3) := -\int_{\Omega_{\varepsilon}} u_1 \otimes u_2 : \nabla u_3 \, dx \tag{5.10.2}$$

for $u_1, u_2, u_3 \in H^1(\Omega_{\varepsilon})^3$. Note that, if $u_1 \in V_{\varepsilon}$, then we have

$$\int_{\Omega_{\varepsilon}} (u_1 \cdot \nabla) u_2 \cdot u_3 \, dx = b(u_1, u_2, u_3)$$

by the integration by parts formula

$$\int_{\Omega_{\varepsilon}} (u_1 \cdot \nabla) u_2 \cdot u_3 \, dx = \int_{\Gamma_{\varepsilon}} (u_1 \cdot n_{\varepsilon}) (u_2 \cdot u_3) \, d\mathcal{H}^2 - \int_{\Omega_{\varepsilon}} \{ (\operatorname{div} u_1) (u_2 \cdot u_3) + u_1 \otimes u_2 : \nabla u_3 \} \, dx$$

and the conditions div $u_1 = 0$ in Ω_{ε} and $u_1 \cdot n_{\varepsilon} = 0$ on Γ_{ε} .

Our goal is to derive the limit of the weak formulation (5.10.1) as well as to show the convergence of the average of the strong solution u^{ε} as $\varepsilon \to 0$.

5.10.2 Average of the weak formulation

The first step is to derive a weak formulation satisfied by the averaged tangential component of the strong solution u^{ε} , in which we take a test function from the weighted solenoidal space

$$V_g := H^1_{g\sigma}(\Gamma, T\Gamma) = \{ v \in H^1(\Gamma, T\Gamma) \mid \operatorname{div}_{\Gamma}(gv) = 0 \text{ on } \Gamma \}.$$

Since the constant extension of a vector field in V_g is not in V_{ε} , we need to construct an appropriate test function in V_{ε} from a weighted solenoidal vector field on Γ . To this end, we use the impermeable extension operator E_{ε} given by (5.3.42) and the Helmholtz–Leray projection \mathbb{P}_{ε} onto $L^2_{\sigma}(\Omega_{\varepsilon})$. **Lemma 5.10.1.** For $\eta \in V_g$ let $\eta_{\varepsilon} := \mathbb{P}_{\varepsilon} E_{\varepsilon} \eta$. Then $\eta_{\varepsilon} \in V_{\varepsilon}$ and

$$\|\eta_{\varepsilon} - \bar{\eta}\|_{L^{2}(\Omega_{\varepsilon})} + \left\|\nabla\eta_{\varepsilon} - \overline{F(\eta)}\right\|_{L^{2}(\Omega_{\varepsilon})} \le c\varepsilon^{3/2} \|\eta\|_{H^{1}(\Gamma)},$$
(5.10.3)

$$\|\eta_{\varepsilon} - \bar{\eta}\|_{L^{2}(\Gamma_{\varepsilon})} \le c\varepsilon \|\eta\|_{H^{1}(\Gamma)}, \qquad (5.10.4)$$

where c > 0 is a constant independent of ε and η , and

$$F(\eta) := \nabla_{\Gamma} \eta + \frac{1}{g} (\eta \cdot \nabla_{\Gamma} g) Q \quad on \quad \Gamma.$$
(5.10.5)

Proof. Since $\eta \in H^1(\Gamma)^3$, we have $E_{\varepsilon}\eta \in H^1(\Omega_{\varepsilon})^3$ by Lemma 5.3.12 and thus $\eta_{\varepsilon} \in V_{\varepsilon}$. Let us derive the estimates (5.10.3) and (5.10.4). By the definition (5.3.42) of the extension $E_{\varepsilon}\eta$ and the inequalities (5.2.53) and (5.3.37) we have

$$||E_{\varepsilon}\eta - \bar{\eta}||_{L^{2}(\Omega_{\varepsilon})} \le c\varepsilon ||\bar{\eta}||_{L^{2}(\Omega_{\varepsilon})} \le c\varepsilon^{3/2} ||\eta||_{L^{2}(\Gamma)}.$$
(5.10.6)

Also, from (5.2.53) and (5.3.44) it follows that

$$\left\|\nabla E_{\varepsilon}\eta - \overline{F(\eta)}\right\|_{L^{2}(\Omega_{\varepsilon})} \le c\varepsilon \left(\|\bar{\eta}\|_{L^{2}(\Omega_{\varepsilon})} + \left\|\overline{\nabla_{\Gamma}\eta}\right\|_{L^{2}(\Omega_{\varepsilon})}\right) \le c\varepsilon^{3/2} \|\eta\|_{H^{1}(\Gamma)}.$$
(5.10.7)

Since $E_{\varepsilon}\eta \in H^1(\Omega_{\varepsilon})^3$ satisfies $E_{\varepsilon}\eta \cdot n_{\varepsilon} = 0$ on Γ_{ε} (see Lemma 5.3.11), we can apply (5.5.1) to $u = E_{\varepsilon}\eta$ and $\mathbb{P}_{\varepsilon}u = \eta_{\varepsilon}$ to get

$$\|\eta_{\varepsilon} - E_{\varepsilon}\eta\|_{H^1(\Omega_{\varepsilon})} \le c \|\operatorname{div}(E_{\varepsilon}\eta)\|_{L^2(\Omega_{\varepsilon})}$$

Moreover, noting that $\eta \in V_g$ satisfies $\operatorname{div}_{\Gamma}(gv) = 0$ on Γ , we use (5.3.48) to the right-hand side of the above inequality to observe that

$$\|\eta_{\varepsilon} - E_{\varepsilon}\eta\|_{H^1(\Omega_{\varepsilon})} \le c\varepsilon^{3/2} \|\eta\|_{H^1(\Gamma)}.$$
(5.10.8)

By (5.10.6)–(5.10.8) we obtain (5.10.3). To prove (5.10.4) we use (5.3.8), (5.10.3), and $\partial_n \bar{\eta} = 0$ in Ω_{ε} to get

$$\begin{aligned} \|\eta_{\varepsilon} - \bar{\eta}\|_{L^{2}(\Gamma_{\varepsilon})} &\leq c \left(\varepsilon^{-1/2} \|\eta_{\varepsilon} - \bar{\eta}\|_{L^{2}(\Omega_{\varepsilon})} + \varepsilon^{1/2} \|\partial_{n}\eta_{\varepsilon} - \partial_{n}\bar{\eta}\|_{L^{2}(\Omega_{\varepsilon})}\right) \\ &\leq c \left(\varepsilon \|\eta\|_{H^{1}(\Gamma)} + \varepsilon^{1/2} \|\eta_{\varepsilon}\|_{H^{1}(\Omega_{\varepsilon})}\right). \end{aligned}$$

(Recall that $\partial_n = \bar{n} \cdot \nabla$ is the directional derivative in the normal direction of Γ). Moreover, from (5.3.43) and (5.10.8) we deduce that

$$\|\eta_{\varepsilon}\|_{H^{1}(\Omega_{\varepsilon})} \leq \|E_{\varepsilon}\eta\|_{H^{1}(\Omega_{\varepsilon})} + \|\eta_{\varepsilon} - E_{\varepsilon}\eta\|_{H^{1}(\Omega_{\varepsilon})} \leq c\varepsilon^{1/2}\|\eta\|_{H^{1}(\Gamma)}.$$

Therefore, the inequality (5.10.4) follows.

Next we approximate the bilinear and trilinear forms a_{ε} and b_{ε} by bilinear and trilinear forms for tangential vector fields on Γ . Let γ^0 and γ^1 be nonnegative constants. For $v_1, v_2 \in H^1(\Gamma, T\Gamma)$ we define

$$a_{g}(v_{1}, v_{2}) := 2\nu \int_{\Gamma} \left\{ gD_{\Gamma}(v_{1}) : D_{\Gamma}(v_{2}) + \frac{1}{g}(v_{1} \cdot \nabla_{\Gamma}g)(v_{2} \cdot \nabla_{\Gamma}g) \right\} d\mathcal{H}^{2} + (\gamma^{0} + \gamma^{1}) \int_{\Gamma} v_{1} \cdot v_{2} \, d\mathcal{H}^{2}, \quad (5.10.9)$$

where $D_{\Gamma}(v_1)$ is the surface strain rate tensor given by (5.4.38). Also, we set

$$b_g(v_1, v_2, v_3) := -\int_{\Gamma} g(v_1 \otimes v_2) : \nabla_{\Gamma} v_3 \, d\mathcal{H}^2$$
(5.10.10)

for $v_1, v_2, v_3 \in H^1(\Gamma, T\Gamma)$. Let us give their basic properties.

Lemma 5.10.2. There exists a constant c > 0 such that

$$\|\nabla_{\Gamma} v\|_{L^{2}(\Gamma)}^{2} \leq c \left\{ a_{g}(v,v) + \|v\|_{L^{2}(\Gamma)}^{2} \right\}$$
(5.10.11)

for all $v \in H^1(\Gamma, T\Gamma)$.

Proof. By (5.2.30) and the Korn inequality (5.4.37) we have

$$\begin{aligned} \|\nabla_{\Gamma} v\|_{L^{2}(\Gamma)}^{2} &\leq c \left(\|D_{\Gamma}(v)\|_{L^{2}(\Gamma)}^{2} + \|v\|_{L^{2}(\Gamma)}^{2} \right) \leq c \left(2\nu \|g^{1/2} D_{\Gamma}(v)\|_{L^{2}(\Gamma)}^{2} + \|v\|_{L^{2}(\Gamma)}^{2} \right) \\ &\leq c \left\{ a_{g}(v,v) + \|v\|_{L^{2}(\Gamma)}^{2} \right\} \end{aligned}$$

for all $v \in H^1(\Gamma, T\Gamma)$. Hence (5.10.11) holds.

Lemma 5.10.3. There exists a constant c > 0 such that

$$|b_g(v_1, v_2, v_3)| \le c ||v_1||_{L^2(\Gamma)}^{1/2} ||v_1||_{H^1(\Gamma)}^{1/2} ||v_2||_{L^2(\Gamma)}^{1/2} ||v_2||_{H^1(\Gamma)}^{1/2} ||v_3||_{H^1(\Gamma)}$$
(5.10.12)

for all $v_1, v_2, v_3 \in H^1(\Gamma, T\Gamma)$. Moreover,

$$b_g(v_1, v_2, v_3) = -b_g(v_1, v_3, v_2), \quad b_g(v_1, v_2, v_2) = 0$$
 (5.10.13)

for all $v_1 \in V_g$ and $v_2, v_3 \in H^1(\Gamma, T\Gamma)$.

Proof. The inequality (5.10.12) follows from Hölder's inequality and Ladyzhenskaya's inequality (5.3.1). Let $v_1 \in V_g$ and $v_2, v_3 \in H^1(\Gamma, T\Gamma)$. For $a \in \mathbb{R}^3$ and i = 1, 2, 3 we denote by a^i the *i*-th component of *a*. Since

$$g(v_1 \otimes v_2) : \nabla_{\Gamma} v_3 = \sum_{i,j=1}^3 g v_1^i v_2^j \underline{D}_i v_3^j = \sum_{i,j=1}^3 \{ \underline{D}_i (g v_1^i v_2^j v_3^j) - v_2^j v_3^j \underline{D}_i (g v_1^i) - g v_1^i v_3^j \underline{D}_i v_2^j \}$$

= div_{\Gamma} [g(v_2 \cdot v_3)v_1] - (v_2 \cdot v_3) div_{\Gamma} (g v_1) - v_1 \otimes v_3 : \nabla_{\Gamma} v_2

on Γ and $\operatorname{div}_{\Gamma}(gv_1) = 0$ by $v_1 \in V_g$, we have

$$b_g(v_1, v_2, v_3) = \int_{\Gamma} \operatorname{div}_{\Gamma}[g(v_2 \cdot v_3)v_1] d\mathcal{H}^2 - b_g(v_1, v_3, v_2).$$

Here the first term on the right-hand side vanishes by the Stokes theorem, since $g(v_2 \cdot v_3)v_1$ is tangential on the closed surface Γ . Hence the first equality (5.10.13) follows. We also get the second equality by setting $v_2 = v_3$ in the first one.

Now let us approximate the bilinear and trilinear forms a_{ε} and b_{ε} by a_g and b_g .

Lemma 5.10.4. Let $u \in H^2(\Omega_{\varepsilon})^3$ satisfy the slip boundary conditions (5.3.20)–(5.3.21) on Γ_{ε} . Also, let $\eta \in V_g$ and $\eta_{\varepsilon} := \mathbb{P}_{\varepsilon} E_{\varepsilon} \eta$. Then

$$|a_{\varepsilon}(u,\eta_{\varepsilon}) - \varepsilon a_g(M_{\tau}u,\eta)| \le cR_{\varepsilon}^a(u) \|\eta\|_{H^1(\Gamma)}, \qquad (5.10.14)$$

where c > 0 is a constant independent of ε , u, and η , and

$$R^{a}_{\varepsilon}(u) := \varepsilon^{3/2} \|u\|_{H^{2}(\Omega_{\varepsilon})} + \varepsilon^{1/2} \|u\|_{L^{2}(\Omega_{\varepsilon})} \sum_{i=0,1} |\varepsilon^{-1}\gamma^{i}_{\varepsilon} - \gamma^{i}|.$$
(5.10.15)

Proof. Let $F(\eta)$ be the matrix given by (5.10.5) and

$$J_{1} := \left(D(u), D(\eta_{\varepsilon})\right)_{L^{2}(\Omega_{\varepsilon})} - \left(D(u), \overline{F(\eta)}\right)_{L^{2}(\Omega_{\varepsilon})},$$

$$J_{2} := \left(D(u), \overline{F(\eta)}\right)_{L^{2}(\Omega_{\varepsilon})} - \varepsilon \left\{ \left(gD_{\Gamma}(M_{\tau}u), D_{\Gamma}(\eta)\right)_{L^{2}(\Gamma)} + \left(M_{\tau}u \cdot \nabla_{\Gamma}g, g^{-1}\eta \cdot \nabla_{\Gamma}g\right)_{L^{2}(\Gamma)} \right\}.$$

We also define

$$K_{1} := \sum_{i=0,1} \gamma_{\varepsilon}^{i} \{ (u,\eta_{\varepsilon})_{L^{2}(\Gamma_{\varepsilon}^{i})} - (u,\bar{\eta})_{L^{2}(\Gamma_{\varepsilon}^{i})} \},$$

$$K_{2} := \sum_{i=0,1} \gamma_{\varepsilon}^{i} \{ (u,\bar{\eta})_{L^{2}(\Gamma_{\varepsilon}^{i})} - (M_{\tau}u,\eta)_{L^{2}(\Gamma)} \}, \quad K_{3} := \sum_{i=0,1} (\gamma_{\varepsilon}^{i} - \varepsilon \gamma^{i})(M_{\tau}u,\eta)_{L^{2}(\Gamma)}$$

so that

$$a_{\varepsilon}(u,\eta_{\varepsilon}) - \varepsilon a_g(M_{\tau}u,\eta) = 2\nu(J_1 + J_2) + K_1 + K_2 + K_3.$$

Let us estimate each term on the right-hand side. Since D(u) is symmetric,

$$D(u): D(\eta_{\varepsilon}) = D(u): \nabla \eta_{\varepsilon}$$
 in Ω_{ε} .

By this equality and (5.10.3) we have

$$|J_1| \le \|D(u)\|_{L^2(\Omega_{\varepsilon})} \left\|\nabla \eta_{\varepsilon} - \overline{F(\eta)}\right\|_{L^2(\Omega_{\varepsilon})} \le c\varepsilon^{3/2} \|u\|_{H^1(\Omega_{\varepsilon})} \|\eta\|_{H^1(\Gamma)}.$$
(5.10.16)

Next we deal with J_2 . Since $I_3 = P + Q$, $\nabla_{\Gamma} \eta = P \nabla_{\Gamma} \eta$, and $\eta \cdot n = 0$,

$$\nabla_{\Gamma}\eta = (\nabla_{\Gamma}\eta)P + (\nabla_{\Gamma}\eta)Q = P(\nabla_{\Gamma}\eta)P + \{(\nabla_{\Gamma}\eta)n\} \otimes n, (\nabla_{\Gamma}\eta)n = \nabla_{\Gamma}(\eta \cdot n) - (\nabla_{\Gamma}n)\eta = W\eta$$

on Γ . From these relations we deduce that $F(\eta) = A + v \otimes n + \xi Q$ on Γ , where

$$A := P(\nabla_{\Gamma} \eta) P, \quad v := W\eta, \quad \xi := g^{-1} \eta \cdot \nabla_{\Gamma} g.$$

Moreover, by the symmetry of $D_{\Gamma}(M_{\tau}u)$ we see that

$$D_{\Gamma}(M_{\tau}u): D_{\Gamma}(\eta) = D_{\Gamma}(M_{\tau}u): P(\nabla_{\Gamma}\eta)P = D_{\Gamma}(M_{\tau}u): A$$

on Γ . Hence we have $J_2 = J_2^1 + J_2^2 + J_2^3$, where

$$J_2^1 := \left(D(u), \overline{A} \right)_{L^2(\Omega_{\varepsilon})} - \varepsilon (g D_{\Gamma}(M_{\tau} u), A)_{L^2(\Gamma)},$$

$$J_2^2 := \left(D(u), \overline{\xi Q} \right)_{L^2(\Omega_{\varepsilon})} - \varepsilon (M_{\tau} u \cdot \nabla_{\Gamma} g, \xi)_{L^2(\Gamma)}, \quad J_2^3 := (D(u), \overline{v} \otimes \overline{n})_{L^2(\Omega_{\varepsilon})}.$$

Since u and A satisfies the conditions in Lemma 5.6.26, we can apply (5.6.72) to J_2^1 to get

$$|J_2^1| \le c\varepsilon^{3/2} \|u\|_{H^1(\Omega_{\varepsilon})} \|A\|_{L^2(\Gamma)} \le c\varepsilon^{3/2} \|u\|_{H^1(\Omega_{\varepsilon})} \|\eta\|_{H^1(\Gamma)}.$$

Also, since u satisfies $u \cdot n_{\varepsilon} = 0$ on Γ_{ε} , we use (5.6.76) to deduce that

$$|J_2^2| \le c\varepsilon^{3/2} ||u||_{H^1(\Omega_{\varepsilon})} ||\xi||_{L^2(\Gamma)} \le c\varepsilon^{3/2} ||u||_{H^1(\Omega_{\varepsilon})} ||\eta||_{L^2(\Gamma)}.$$

To estimate J_2^3 , we note that $v = W\eta \in H^1(\Gamma, T\Gamma)$ and u satisfies the boundary condition (5.3.21) on Γ_{ε} . Hence the inequality (5.6.81) implies that

$$|J_2^3| \le c\varepsilon^{3/2} ||u||_{H^2(\Omega_{\varepsilon})} ||v||_{L^2(\Gamma)} \le c\varepsilon^{3/2} ||u||_{H^2(\Omega_{\varepsilon})} ||\eta||_{L^2(\Gamma)}.$$

From the above three estimates it follows that

$$|J_2| \le |J_2^1| + |J_2^2| + |J_2^3| \le c\varepsilon^{3/2} ||u||_{H^2(\Omega_{\varepsilon})} ||\eta||_{H^1(\Gamma)}.$$
(5.10.17)

Now let us estimate K_1 , K_2 , and K_3 . To K_1 we apply (5.1.6), (5.3.8), and (5.10.4) to get

$$|K_1| \le c\varepsilon ||u||_{L^2(\Gamma_\varepsilon)} ||\eta_\varepsilon - \bar{\eta}||_{L^2(\Gamma_\varepsilon)} \le c\varepsilon^{3/2} ||u||_{H^1(\Omega_\varepsilon)} ||\eta||_{H^1(\Gamma)}.$$
(5.10.18)

Also, since η is tangential on Γ , we have $M_{\tau}u \cdot \eta = Mu \cdot \eta$ on Γ and thus

$$|K_2| \le c\varepsilon \sum_{i=0,1} \left| (u,\bar{\eta})_{L^2(\Gamma_{\varepsilon}^i)} - (Mu,\eta)_{L^2(\Gamma)} \right| \le c\varepsilon^{3/2} ||u||_{H^1(\Omega_{\varepsilon})} ||\eta||_{L^2(\Gamma)}$$
(5.10.19)

by (5.1.6) and (5.6.70). To K_3 we just use (5.6.4) to obtain

$$|K_{3}| \leq c\varepsilon^{-1/2} \|u\|_{L^{2}(\Omega_{\varepsilon})} \|\eta\|_{L^{2}(\Gamma)} \sum_{i=0,1} |\gamma_{\varepsilon}^{i} - \varepsilon\gamma^{i}|$$

$$= c\varepsilon^{1/2} \|u\|_{L^{2}(\Omega_{\varepsilon})} \|\eta\|_{L^{2}(\Gamma)} \sum_{i=0,1} |\varepsilon^{-1}\gamma_{\varepsilon}^{i} - \gamma^{i}|.$$
(5.10.20)

Finally, we deduce from (5.10.16) - (5.10.20) that

$$|a_{\varepsilon}(u,\eta_{\varepsilon}) - \varepsilon a_g(M_{\tau}u,\eta)| \le c(|J_1| + |J_2| + |K_1| + |K_2| + |K_3|) \le cR_{\varepsilon}^a(u) \|\eta\|_{H^1(\Gamma)},$$

where $R^a_{\varepsilon}(u)$ is given by (5.10.15). Hence (5.10.14) is valid.

Lemma 5.10.5. Let $u_1 \in H^2(\Omega_{\varepsilon})^3$, $u_2 \in H^1(\Omega_{\varepsilon})^3$, $\eta \in V_g$, and $\eta_{\varepsilon} := \mathbb{P}_{\varepsilon} E_{\varepsilon} \eta$. Suppose that u_1 satisfies div $u_1 = 0$ in Ω_{ε} and (5.3.20)–(5.3.21) on Γ_{ε} and that u_2 satisfies (5.3.20) on Γ_{ε}^0 or on Γ_{ε}^1 . Then

$$|b_{\varepsilon}(u_1, u_2, \eta_{\varepsilon}) - \varepsilon b_g(M_{\tau} u_1, M_{\tau} u_2, \eta)| \le c R_{\varepsilon}^b(u_1, u_2) \|\eta\|_{H^1(\Gamma)}.$$
 (5.10.21)

Here c > 0 is a constant independent of ε , u_1 , u_2 , and η , and

$$R^{b}_{\varepsilon}(u_{1}, u_{2}) := \varepsilon^{3/2} \|u_{1} \otimes u_{2}\|_{L^{2}(\Omega_{\varepsilon})} + \varepsilon \|u_{1}\|_{H^{1}(\Omega_{\varepsilon})} \|u_{2}\|_{H^{1}(\Omega_{\varepsilon})} + \left(\varepsilon \|u_{1}\|_{H^{2}(\Omega_{\varepsilon})} + \varepsilon^{1/2} \|u_{1}\|_{L^{2}(\Omega_{\varepsilon})}^{1/2} \|u_{1}\|_{H^{2}(\Omega_{\varepsilon})}^{1/2} \right) \|u_{2}\|_{L^{2}(\Omega_{\varepsilon})}.$$
(5.10.22)

Proof. Let $F(\eta)$ be the matrix given by (5.10.5). By (5.10.3) we see that

$$\begin{aligned} \left| b_{\varepsilon}(u_{1}, u_{2}, \eta_{\varepsilon}) - \left(u_{1} \otimes u_{2}, \overline{F(\eta)} \right)_{L^{2}(\Omega_{\varepsilon})} \right| &\leq \| u_{1} \otimes u_{2} \|_{L^{2}(\Omega_{\varepsilon})} \left\| \nabla \eta_{\varepsilon} - \overline{F(\eta)} \right\|_{L^{2}(\Omega_{\varepsilon})} \\ &\leq c \varepsilon^{3/2} \| u_{1} \otimes u_{2} \|_{L^{2}(\Omega_{\varepsilon})} \| \eta \|_{H^{1}(\Gamma)}. \end{aligned}$$
(5.10.23)

Since η is tangential on Γ , we have

$$\nabla_{\Gamma}\eta = (\nabla_{\Gamma}\eta)P + \{(\nabla_{\Gamma}\eta)n\} \otimes n = P(\nabla_{\Gamma}\eta)P + (W\eta) \otimes n \quad \text{on} \quad \Gamma$$

as in the proof of Lemma 5.10.4. Based on this formula, we decompose

$$F(\eta) = A + v \otimes n, \quad A := P(\nabla_{\Gamma} \eta)P, \quad v := W\eta + g^{-1}(\eta \cdot \nabla_{\Gamma} g)n \quad \text{on} \quad \Gamma.$$

Then since u_1 and A satisfies the conditions in Lemma 5.6.29, we can use (5.6.82) to get

$$\left| \left(u_1 \otimes u_2, \overline{A} \right)_{L^2(\Omega_{\varepsilon})} - \varepsilon(g(M_{\tau}u_1) \otimes (M_{\tau}u_2), A)_{L^2(\Gamma)} \right| \le cR_{\varepsilon}(u_1, u_2) \|A\|_{L^2(\Gamma)}$$

$$\le cR_{\varepsilon}(u_1, u_2) \|\eta\|_{H^1(\Gamma)},$$
(5.10.24)

where $R_{\varepsilon}(u_1, u_2)$ is given by (5.6.83). Also, since $v \in H^1(\Gamma)^3$ and u_2 satisfies (5.3.20) on Γ_{ε}^0 or on Γ_{ε}^1 , the inequality (5.6.87) yields that

$$\begin{aligned} |(u_1 \otimes u_2, \bar{v} \otimes \bar{n})_{L^2(\Omega_{\varepsilon})}| &\leq c\varepsilon ||u_1||_{H^1(\Omega_{\varepsilon})} ||u_2||_{H^1(\Omega_{\varepsilon})} ||v||_{H^1(\Gamma)} \\ &\leq c\varepsilon ||u_1||_{H^1(\Omega_{\varepsilon})} ||u_2||_{H^1(\Omega_{\varepsilon})} ||\eta||_{H^1(\Gamma)}. \end{aligned}$$
(5.10.25)

Noting that $F(\eta) = A + v \otimes n$ on Γ and $R_{\varepsilon}(u_1, u_2)$ is of the form (5.6.83), we combine (5.10.23), (5.10.24), and (5.10.25) to obtain

$$\left|b_{\varepsilon}(u_1, u_2, \eta_{\varepsilon}) - \varepsilon(g(M_{\tau}u_1) \otimes (M_{\tau}u_2), A)_{L^2(\Gamma)}\right| \le cR_{\varepsilon}^b(u_1, u_2) \|\eta\|_{H^1(\Gamma)}, \tag{5.10.26}$$

where $R^b_{\varepsilon}(u_1, u_2)$ is given by (5.10.22). Finally, we observe that

$$(M_{\tau}u_1) \otimes (M_{\tau}u_2) : A = (M_{\tau}u_1) \otimes (M_{\tau}u_2) : \nabla_{\Gamma}\eta$$
 on I

by $A = P(\nabla_{\Gamma} \eta)P$ and the fact that $M_{\tau}u_1$ and $M_{\tau}u_2$ are tangential on Γ . Therefore,

$$(g(M_{\tau}u_1) \otimes (M_{\tau}u_2), A)_{L^2(\Gamma)} = b_g(M_{\tau}u_1, M_{\tau}u_2, \eta)$$

and the inequality (5.10.21) follows from (5.10.26).

Now let us derive a weak formulation for the averaged tangential component of u^{ε} from (5.10.1).

Lemma 5.10.6. Suppose that the assumptions in Theorem 5.8.4 hold. For $\varepsilon \in (0, \varepsilon'_1)$ let u^{ε} be the global strong solution to (5.1.1)–(5.1.3) given by Theorem 5.8.4. Then

$$M_{\tau}u^{\varepsilon} \in C([0,\infty); H^1(\Gamma, T\Gamma)) \cap H^1_{loc}([0,\infty); L^2(\Gamma, T\Gamma))$$

and for all $\eta \in L^2(0,T;V_g)$, T > 0 we have

$$\int_0^T \{ (g\partial_t M_\tau u^\varepsilon, \eta)_{L^2(\Gamma)} + a_g(M_\tau u, \eta) + b_g(M_\tau u, M_\tau u, \eta) \} dt$$
$$= \int_0^T (gM_\tau \mathbb{P}_\varepsilon f^\varepsilon, \eta)_{L^2(\Gamma)} dt + R_\varepsilon^1(\eta). \quad (5.10.27)$$

Here the residual term $R^1_{\varepsilon}(\eta)$ satisfies

$$|R_{\varepsilon}^{1}(\eta)| \leq c \left(\varepsilon^{\alpha/4} + \sum_{i=0,1} |\varepsilon^{-1} \gamma_{\varepsilon}^{i} - \gamma^{i}| \right) (1+T)^{1/2} \|\eta\|_{L^{2}(0,T;H^{1}(\Gamma))}$$
(5.10.28)

with a constant c > 0 independent of ε , u^{ε} , η , and T.

Proof. The space-time regularity of $M_{\tau}u^{\varepsilon}$ follows from that of u^{ε} and Lemmas 5.6.2, 5.6.6, and 5.6.9. Let us show that $M_{\tau}u^{\varepsilon}$ satisfies (5.10.27). For $\eta \in L^2(0,T;V_g)$ we set $\eta_{\varepsilon} := \mathbb{P}_{\varepsilon}E_{\varepsilon}\eta$. Then $\eta_{\varepsilon} \in L^2(0,T;V_{\varepsilon})$ by Lemma 5.10.1 and thus we can substitute it for φ in (5.10.1):

$$\int_0^T \{ (\partial_t u^\varepsilon, \eta_\varepsilon)_{L^2(\Omega_\varepsilon)} + a_\varepsilon (u^\varepsilon, \eta_\varepsilon) + b_\varepsilon (u^\varepsilon, u^\varepsilon, \eta_\varepsilon) \} \, dt = \int_0^T (\mathbb{P}_\varepsilon f^\varepsilon, \eta_\varepsilon)_{L^2(\Omega_\varepsilon)} \, dt.$$

We divide both sides of this equality by ε and replace each term by the corresponding term of (5.10.27). Then we get (5.10.27) with $R_{\varepsilon}^{1}(\eta) := \varepsilon^{-1}(I_1 + I_2 + I_3 + I_4)$, where

$$\begin{split} I_1 &:= \int_0^T (\partial_t u^{\varepsilon}, \eta_{\varepsilon})_{L^2(\Omega_{\varepsilon})} \, dt - \varepsilon \int_0^T (g \partial_t M_{\tau} u^{\varepsilon}, \eta)_{L^2(\Gamma)} \, dt, \\ I_2 &:= \int_0^T a_{\varepsilon}(u^{\varepsilon}, \eta_{\varepsilon}) \, dt - \varepsilon \int_0^T a_g(M_{\tau} u, \eta) \, dt, \\ I_3 &:= \int_0^T b_{\varepsilon}(u^{\varepsilon}, u^{\varepsilon}, \eta_{\varepsilon}) \, dt - \varepsilon \int_0^T b_g(M_{\tau} u^{\varepsilon}, M_{\tau} u^{\varepsilon}, \eta) \, dt, \\ I_4 &:= \int_0^T (\mathbb{P}_{\varepsilon} f^{\varepsilon}, \eta_{\varepsilon})_{L^2(\Omega_{\varepsilon})} \, dt - \varepsilon \int_0^T (g M_{\tau} \mathbb{P}_{\varepsilon} f^{\varepsilon}, \eta)_{L^2(\Gamma)} \, dt. \end{split}$$

Let us estimate these differences. First note that

$$(g\partial_t M_\tau u^\varepsilon, \eta)_{L^2(\Gamma)} = (gM_\tau(\partial_t u^\varepsilon), \eta)_{L^2(\Gamma)} = (gM(\partial_t u^\varepsilon), \eta)_{L^2(\Gamma)}$$

by Lemma 5.6.6 and the fact that η is tangential on Γ . Thus, by (5.6.69) and (5.10.4),

$$\begin{aligned} |(\partial_t u^{\varepsilon}, \eta_{\varepsilon})_{L^2(\Omega_{\varepsilon})} &- \varepsilon (g \partial_t M_{\tau} u^{\varepsilon}, \eta)_{L^2(\Gamma)}| \\ &\leq |(\partial_t u^{\varepsilon}, \bar{\eta})_{L^2(\Omega_{\varepsilon})} - \varepsilon (g M(\partial_t u^{\varepsilon}), \eta)_{L^2(\Gamma)}| + \|\partial_t u^{\varepsilon}\|_{L^2(\Omega_{\varepsilon})} \|\eta_{\varepsilon} - \bar{\eta}\|_{L^2(\Omega_{\varepsilon})} \\ &\leq c \varepsilon^{3/2} \|\partial_t u^{\varepsilon}\|_{L^2(\Omega_{\varepsilon})} \|\eta\|_{L^2(\Gamma)}. \end{aligned}$$

From this inequality, Hölder's inequality, and (5.8.36) it follows that

$$|I_1| \le c\varepsilon^{3/2} \|\partial_t u^\varepsilon\|_{L^2(0,T;L^2(\Omega_\varepsilon))} \|\eta\|_{L^2(0,T;L^2(\Gamma))} \le c\varepsilon^{1+\alpha/2} (1+T)^{1/2} \|\eta\|_{L^2(0,T;L^2(\Gamma))}.$$
(5.10.29)

In the same way, we apply (5.6.69) and (5.10.4) to I_4 and then use (5.1.10) to get

$$|I_4| \le c\varepsilon^{1+\alpha/2} T^{1/2} ||\eta||_{L^2(0,T;L^2(\Gamma))}.$$
(5.10.30)

Next we deal with I_2 . By (5.10.14) we see that

$$|I_2| \le c \left(\int_0^T R_{\varepsilon}^a (u^{\varepsilon})^2 \, dt \right)^{1/2} \|\eta\|_{L^2(0,T;H^1(\Gamma))},$$

where $R^a_{\varepsilon}(u^{\varepsilon})$ is given by (5.10.15). Moreover, by (5.8.34) we have

$$\begin{split} \int_0^T R^a_{\varepsilon}(u^{\varepsilon})^2 \, dt &\leq c \left(\varepsilon^3 \int_0^T \|u^{\varepsilon}\|^2_{H^2(\Omega_{\varepsilon})} \, dt + \varepsilon \gamma(\varepsilon)^2 \int_0^T \|u^{\varepsilon}\|^2_{L^2(\Omega_{\varepsilon})} \, dt \right) \\ &\leq c \varepsilon^2 \{ \varepsilon^{\alpha} + \gamma(\varepsilon)^2 \} (1+T), \end{split}$$

where $\gamma(\varepsilon) := \sum_{i=0,1} |\varepsilon^{-1} \gamma^i_{\varepsilon} - \gamma^i|$. Therefore,

$$|I_2| \le c\varepsilon \{\varepsilon^{\alpha/2} + \gamma(\varepsilon)\} (1+T)^{1/2} \|\eta\|_{L^2(0,T;H^1(\Gamma))}.$$
(5.10.31)

Now let us estimate I_3 . By (5.10.21) we have

$$|I_3| \le c \left(\int_0^T R^b_{\varepsilon}(u, u)^2 \, dt \right)^{1/2} \|\eta\|_{L^2(0,T; H^1(\Gamma))},$$

where $R^b_{\varepsilon}(u, u)$ is given by (5.10.22). To estimate the right-hand side, we see that

$$\int_0^T \|u\|_{H^1(\Omega_{\varepsilon})}^4 dt \le \left(\sup_{t \in (0,T)} \|u^{\varepsilon}(t)\|_{H^1(\Omega_{\varepsilon})}^2\right) \int_0^T \|u^{\varepsilon}\|_{H^1(\Omega_{\varepsilon})}^2 dt \le c\varepsilon^{\alpha}(1+T)$$

by (5.8.34). Using this inequality, (5.8.35), (5.8.39), and (5.8.40) we deduce that

$$\begin{split} \int_0^T R^b_{\varepsilon}(u,u)^2 \, dt &\leq c \left(\varepsilon^3 \int_0^T \|u^{\varepsilon} \otimes u^{\varepsilon}\|^2_{L^2(\Omega_{\varepsilon})} \, dt + \varepsilon^2 \int_0^T \|u^{\varepsilon}\|^4_{H^1(\Omega_{\varepsilon})} \, dt \\ &+ \varepsilon^2 \int_0^T \|u^{\varepsilon}\|^2_{L^2(\Omega_{\varepsilon})} \|u^{\varepsilon}\|^2_{H^2(\Omega_{\varepsilon})} \, dt + \varepsilon \int_0^T \|u^{\varepsilon}\|^3_{L^2(\Omega_{\varepsilon})} \|u^{\varepsilon}\|_{H^2(\Omega_{\varepsilon})} \, dt \right) \\ &\leq c \varepsilon^2 (\varepsilon^2 + \varepsilon^{\alpha} + \varepsilon^{\alpha/2}) (1+T) \leq c \varepsilon^{2+\alpha/2} (1+T). \end{split}$$

Hence we obtain

$$|I_3| \le c\varepsilon^{1+\alpha/4} (1+T)^{1/2} \|\eta\|_{L^2(0,T;H^1(\Gamma))}.$$
(5.10.32)

Finally, by (5.10.29) - (5.10.32) we observe that

$$|R_{\varepsilon}^{1}(\eta)| \leq \varepsilon^{-1} \sum_{j=1}^{4} |I_{j}| \leq c \{\varepsilon^{\alpha/4} + \varepsilon^{\alpha/2} + \gamma(\varepsilon)\} (1+T)^{1/2} \|\eta\|_{L^{2}(0,T;H^{1}(\Gamma))}$$

and thus (5.10.28) holds by $\gamma(\varepsilon) = \sum_{i=0,1} |\varepsilon^{-1} \gamma_{\varepsilon}^i - \gamma^i|$ and $\varepsilon^{\alpha/2} \le \varepsilon^{\alpha/4}$.

Remark 5.10.7. By $u \in L^2_{loc}([0,\infty); D(A_{\varepsilon}))$ and Lemma 5.6.9 we also have

$$M_{\tau}u^{\varepsilon} \in L^2_{loc}([0,\infty); H^2(\Gamma, T\Gamma)).$$

We do not use this property in the sequel.

5.10.3 Energy estimate for the average of a solution

Next we prove the energy estimate for the averaged tangential component of the strong solution u^{ε} . We would like to take the averaged tangential component $M_{\tau}u^{\varepsilon}$ as a test function of the weak formulation (5.10.27), but it is not allowed since $M_{\tau}u^{\varepsilon}$ is not in V_g , i.e. the surface divergence of $gM_{\tau}u^{\varepsilon}$ does not vanish in general. To overcome this difficulty, we use the weighted Helmholtz–Leray projection \mathbb{P}_g onto $L^2_{g\sigma}(\Gamma, T\Gamma)$ given in Section 5.9.3 and replace $M_{\tau}u^{\varepsilon}$ in (5.10.27) with $\mathbb{P}_qM_{\tau}u^{\varepsilon}$.

Lemma 5.10.8. Let $u \in L^2_{\sigma}(\Omega_{\varepsilon})$. Then $\mathbb{P}_g M_{\tau} u \in L^2_{q\sigma}(\Gamma, T\Gamma)$ and

$$\|M_{\tau}u - \mathbb{P}_{g}M_{\tau}u\|_{L^{2}(\Gamma)} \le c\varepsilon^{1/2} \|u\|_{L^{2}(\Omega_{\varepsilon})},$$
(5.10.33)

where c > 0 is a constant independent of ε . Also, if $u \in V_{\varepsilon}$, then $\mathbb{P}_{g}M_{\tau}u \in V_{g}$ and

$$\|M_{\tau}u - \mathbb{P}_{g}M_{\tau}u\|_{H^{1}(\Gamma)} \le c\varepsilon^{1/2}\|u\|_{H^{1}(\Omega_{\varepsilon})}.$$
(5.10.34)

Proof. Let $u \in L^2_{\sigma}(\Omega_{\varepsilon})$. Since $M_{\tau}u \in L^2(\Gamma, T\Gamma)$ by Lemma 5.6.2, we have $\mathbb{P}_g M_{\tau}u \in L^2_{q\sigma}(\Gamma, T\Gamma)$. Moreover, from (5.6.49) and (5.9.28) it follows that

$$\|M_{\tau}u - \mathbb{P}_g M_{\tau}u\|_{L^2(\Gamma)} \le c \|\operatorname{div}_{\Gamma}(gM_{\tau}u)\|_{H^{-1}(\Gamma)} \le c\varepsilon^{1/2} \|u\|_{L^2(\Omega_{\varepsilon})}$$

Hence (5.10.33) is valid. If $u \in V_{\varepsilon}$, then $M_{\tau}u \in H^1(\Gamma, T\Gamma)$ by Lemma 5.6.9 and thus Lemma 5.9.14 shows that $\mathbb{P}_g M_{\tau}u \in V_g$. Also, we deduce from (5.6.46) and (5.9.29) that

$$\|M_{\tau}u - \mathbb{P}_g M_{\tau}u\|_{H^1(\Gamma)} \le c \|\operatorname{div}_{\Gamma}(gM_{\tau}u)\|_{L^2(\Gamma)} \le c\varepsilon^{1/2} \|u\|_{H^1(\Omega_{\varepsilon})}.$$

Thus, the inequality (5.10.34) holds.

Lemma 5.10.9. For T > 0, let $u \in H^1(0, T; L^2_{\sigma}(\Omega_{\varepsilon}))$. Then

$$\mathbb{P}_g M_\tau u \in H^1(0,T; L^2_{q\sigma}(\Gamma,T\Gamma))$$

and there exists a constant c > 0 independent of ε and u such that

$$\|\partial_t M_\tau u - \partial_t \mathbb{P}_g M_\tau u\|_{L^2(0,T;L^2(\Gamma))} \le c\varepsilon^{1/2} \|\partial_t u\|_{L^2(0,T;L^2(\Omega_\varepsilon))}.$$
(5.10.35)

Proof. By the space-time regularity of u and Lemmas 5.6.2 and 5.6.6 we see that $M_{\tau}u$ is in $H^1(0,T; L^2(\Gamma,T\Gamma))$. Hence $\mathbb{P}_g M_{\tau}u \in H^1(0,T; L^2_{q\sigma}(\Gamma,T\Gamma))$ and

$$\|\partial_t M_\tau u - \partial_t \mathbb{P}_g M_\tau u\|_{L^2(0,T;L^2(\Gamma))} \le c \|\operatorname{div}_{\Gamma}(g\partial_t M_\tau u)\|_{L^2(0,T;H^{-1}(\Gamma))}$$

by Lemma 5.9.17. Since $\partial_t M_\tau u = M_\tau(\partial_t u)$ in $L^2(0,T;L^2(\Gamma,T\Gamma))$ by Lemma 5.6.6, we further observe by $\partial_t u \in L^2(0,T;L^2_\sigma(\Omega_\varepsilon))$ and (5.6.49) that

$$\|\operatorname{div}_{\Gamma}(g\partial_{t}M_{\tau}u)\|_{L^{2}(0,T;H^{-1}(\Gamma))} = \|\operatorname{div}_{\Gamma}[gM_{\tau}(\partial_{t}u)]\|_{L^{2}(0,T;H^{-1}(\Gamma))} \le c\varepsilon^{1/2}\|\partial_{t}u\|_{L^{2}(0,T;L^{2}(\Omega_{\varepsilon}))}.$$

Combining the above two inequalities we obtain (5.10.35).

Using Lemmas 5.10.8 and 5.10.9 we derive a weak formulation for $\mathbb{P}_{q}M_{\tau}u$.

Lemma 5.10.10. Let u^{ε} be as in Lemma 5.10.6. Then

$$v^{\varepsilon} := \mathbb{P}_{g} M_{\tau} u^{\varepsilon} \in C([0,\infty); V_{g}) \cap H^{1}_{loc}([0,\infty), L^{2}_{g\sigma}(\Gamma, T\Gamma))$$

and there exists a constant c > 0 independent of ε and u^{ε} such that

$$\|M_{\tau}u^{\varepsilon}(t) - v^{\varepsilon}(t)\|_{L^{2}(\Gamma)}^{2} \le c\varepsilon^{2}, \quad \int_{0}^{t} \|M_{\tau}u^{\varepsilon}(s) - v^{\varepsilon}(s)\|_{H^{1}(\Gamma)}^{2} ds \le c\varepsilon^{2}(1+t)$$
(5.10.36)

for all $t \geq 0$. Moreover, for all $\eta \in L^2(0,T;V_g)$, T > 0 we have

$$\int_{0}^{T} \{ (g\partial_{t}v^{\varepsilon}, \eta)_{L^{2}(\Gamma)} + a_{g}(v^{\varepsilon}, \eta) + b_{g}(v^{\varepsilon}, v^{\varepsilon}, \eta) \} dt$$
$$= \int_{0}^{T} (gM_{\tau}\mathbb{P}_{\varepsilon}f^{\varepsilon}, \eta)_{L^{2}(\Gamma)} dt + R_{\varepsilon}^{1}(\eta) + R_{\varepsilon}^{2}(\eta), \quad (5.10.37)$$

where $R^1_{\varepsilon}(\eta)$ is given in Lemma 5.10.6 and $R^2_{\varepsilon}(\eta)$ satisfies

$$|R_{\varepsilon}^{2}(\eta)| \leq c\varepsilon^{\alpha/2} (1+T)^{1/2} \|\eta\|_{L^{2}(0,T;H^{1}(\Gamma))}$$
(5.10.38)

with a constant c > 0 independent of ε , v^{ε} , η , and T.

Proof. By Lemmas 5.10.6, 5.10.8, and 5.10.9 we obtain the space-time regularity of v^{ε} . Also, the inequalities (5.10.36) immediately follow from (5.8.34), (5.10.33), and (5.10.34). For $\eta \in L^2(0,T;V_g)$ let

$$I_{1} := \int_{0}^{T} (g\partial_{t}M_{\tau}u^{\varepsilon}, \eta)_{L^{2}(\Gamma)} dt - \int_{0}^{T} (g\partial_{t}v^{\varepsilon}, \eta)_{L^{2}(\Gamma)} dt,$$

$$I_{2} := \int_{0}^{T} a_{g}(M_{\tau}u^{\varepsilon}, \eta) dt - \int_{0}^{T} a_{g}(v^{\varepsilon}, \eta) dt,$$

$$I_{3} := \int_{0}^{T} b_{g}(M_{\tau}u^{\varepsilon}, M_{\tau}u^{\varepsilon}, \eta) dt - \int_{0}^{T} b_{g}(v^{\varepsilon}, v^{\varepsilon}, \eta) dt.$$

Then by (5.10.27) we obtain (5.10.37) with $R_{\varepsilon}^2(\eta) := I_1 + I_2 + I_3$. Let us estimate I_1, I_2 , and I_3 . By (5.8.36) and (5.10.35) we have

$$|I_{1}| \leq c \|\partial_{t} M_{\tau} u^{\varepsilon} - \partial_{t} v^{\varepsilon}\|_{L^{2}(0,T;L^{2}(\Gamma))} \|\eta\|_{L^{2}(0,T;L^{2}(\Gamma))}$$

$$\leq c \varepsilon^{1/2} \|\partial_{t} u^{\varepsilon}\|_{L^{2}(0,T;L^{2}(\Omega_{\varepsilon}))} \|\eta\|_{L^{2}(0,T;L^{2}(\Gamma))}$$

$$\leq c (1+T)^{1/2} \varepsilon^{\alpha/2} \|\eta\|_{L^{2}(0,T;L^{2}(\Gamma))}.$$

(5.10.39)

Also, we use (5.2.30) and (5.10.36) to get

$$|I_2| \le c \|M_\tau u^\varepsilon - v^\varepsilon\|_{L^2(0,T;H^1(\Gamma))} \|\eta\|_{L^2(0,T;H^1(\Gamma))} \le c(1+T)^{1/2} \varepsilon \|\eta\|_{L^2(0,T;H^1(\Gamma))}.$$
 (5.10.40)

Now let us consider I_3 . Using Hölder's inequality twice we have

$$\begin{aligned} |b_g(M_\tau u^\varepsilon, M_\tau u^\varepsilon, \eta) - b_g(v^\varepsilon, v^\varepsilon, \eta)| \\ &\leq \|M_\tau u^\varepsilon - v^\varepsilon\|_{L^4(\Gamma)} \left(\|M_\tau u^\varepsilon\|_{L^4(\Gamma)} + \|v^\varepsilon\|_{L^4(\Gamma)} \right) \|\nabla_\Gamma \eta\|_{L^2(\Gamma)}. \end{aligned}$$

Moreover, by (5.3.1), (5.6.4), (5.6.22), (5.9.30), (5.10.33), and (5.10.34) we observe that

$$\|w\|_{L^4(\Gamma)} \le c\varepsilon^{-1/2} \|u^{\varepsilon}\|_{L^2(\Omega_{\varepsilon})}^{1/2} \|u^{\varepsilon}\|_{H^1(\Omega_{\varepsilon})}^{1/2}, \quad w = M_{\tau}u^{\varepsilon}, v^{\varepsilon},$$
$$\|M_{\tau}u^{\varepsilon} - v^{\varepsilon}\|_{L^4(\Gamma)} \le c\varepsilon^{1/2} \|u^{\varepsilon}\|_{L^2(\Omega_{\varepsilon})}^{1/2} \|u^{\varepsilon}\|_{H^1(\Omega_{\varepsilon})}^{1/2}.$$

Hence

$$|b_g(M_\tau u^\varepsilon, M_\tau u^\varepsilon, \eta) - b_g(v^\varepsilon, v^\varepsilon, \eta)| \le c ||u^\varepsilon||_{L^2(\Omega_\varepsilon)} ||u^\varepsilon||_{H^1(\Omega_\varepsilon)} ||\eta||_{H^1(\Gamma)}$$

and we use (5.8.38) to get

$$|I_{3}| \leq c \left(\int_{0}^{T} \|u^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon})}^{2} \|u^{\varepsilon}\|_{H^{1}(\Omega_{\varepsilon})}^{2} dt \right)^{1/2} \|\eta\|_{L^{2}(0,T;H^{1}(\Gamma))}$$

$$\leq c\varepsilon (1+T)^{1/2} \|\eta\|_{L^{2}(0,T;H^{1}(\Gamma))}.$$
(5.10.41)

By (5.10.39), (5.10.40), (5.10.41), and $\varepsilon \leq \varepsilon^{\alpha/2}$ we obtain (5.10.38).

Based on (5.10.37) we prove the energy estimate for $v^{\varepsilon} = \mathbb{P}_{g} M_{\tau} u^{\varepsilon}$.

Lemma 5.10.11. Let u^{ε} be as in Lemma 5.10.6 and $v^{\varepsilon} = \mathbb{P}_{g}M_{\tau}u^{\varepsilon}$. Then

$$\max_{t \in [0,T]} \|v^{\varepsilon}(t)\|_{L^{2}(\Gamma)}^{2} + \int_{0}^{T} \|\nabla_{\Gamma} v^{\varepsilon}(t)\|_{L^{2}(\Gamma)}^{2} dt \le c_{T}$$
(5.10.42)

for all $T \ge 0$, where $c_T > 0$ is a constant depending on T and independent of ε .

Proof. For $t \in [0, T]$ let $1_{[0,t]} \colon \mathbb{R} \to \mathbb{R}$ be the characteristic function of the time interval [0, t]. Since $v^{\varepsilon} \in C([0, \infty); V_g)$, we can take $\eta := 1_{[0,t]}v^{\varepsilon}$ as a test function in (5.10.37). Then using (5.10.13) we obtain

$$\int_{0}^{t} \{ (g\partial_{s}v^{\varepsilon}, v^{\varepsilon})_{L^{2}(\Gamma)} + a_{g}(v^{\varepsilon}, v^{\varepsilon}) \} ds$$
$$= \int_{0}^{t} (gM_{\tau}\mathbb{P}_{\varepsilon}f^{\varepsilon}, v^{\varepsilon})_{L^{2}(\Gamma)} ds + R_{\varepsilon}^{1}(v^{\varepsilon}) + R_{\varepsilon}^{2}(v^{\varepsilon}), \quad (5.10.43)$$

where $R_{\varepsilon}^1(v^{\varepsilon})$ and $R_{\varepsilon}^2(v^{\varepsilon})$ satisfy (5.10.28) and (5.10.38), respectively. Since g is nonnegative (see (5.2.30)) and independent of time,

$$\int_{0}^{t} (g\partial_{s}v^{\varepsilon}, v^{\varepsilon})_{L^{2}(\Gamma)} ds = \frac{1}{2} \int_{0}^{t} \frac{d}{ds} \|g^{1/2}v^{\varepsilon}\|_{L^{2}(\Gamma)}^{2} ds$$

$$= \frac{1}{2} \|g^{1/2}v^{\varepsilon}(t)\|_{L^{2}(\Gamma)}^{2} - \frac{1}{2} \|g^{1/2}v^{\varepsilon}(0)\|_{L^{2}(\Gamma)}^{2}.$$
 (5.10.44)

Also, we see by (5.10.11) that

$$\int_0^t \|\nabla_{\Gamma} v^{\varepsilon}\|_{L^2(\Gamma)}^2 ds \le c \int_0^t \left\{ a_g(v^{\varepsilon}, v^{\varepsilon}) + \|v^{\varepsilon}\|_{L^2(\Gamma)}^2 \right\} ds.$$
(5.10.45)

For the right-hand side of (5.10.43), we consider $M_{\tau}\mathbb{P}_{\varepsilon}f^{\varepsilon} = PM_{\tau}\mathbb{P}_{\varepsilon}f^{\varepsilon}$ as an element of $H^{-1}(\Gamma, T\Gamma)$ (see Section 5.2.1). Then we have

$$\int_{0}^{t} (gM_{\tau}\mathbb{P}_{\varepsilon}f^{\varepsilon}, v^{\varepsilon})_{L^{2}(\Gamma)} ds = \int_{0}^{t} [M_{\tau}\mathbb{P}_{\varepsilon}f^{\varepsilon}, gv^{\varepsilon}]_{T\Gamma} ds$$

$$\leq c \int_{0}^{t} \|M_{\tau}\mathbb{P}_{\varepsilon}f^{\varepsilon}\|_{H^{-1}(\Gamma, T\Gamma)} \|gv^{\varepsilon}\|_{H^{1}(\Gamma)} ds \qquad (5.10.46)$$

$$\leq c \int_{0}^{t} \|M_{\tau}\mathbb{P}_{\varepsilon}f^{\varepsilon}\|_{H^{-1}(\Gamma, T\Gamma)} \|v^{\varepsilon}\|_{H^{1}(\Gamma)} ds.$$

To estimate the residual terms, we note that $\varepsilon^{-1}\gamma_{\varepsilon}^{0}$ and $\varepsilon^{-1}\gamma_{\varepsilon}^{1}$ are bounded by (5.1.6). Hence we see by (5.10.28) and (5.10.38) (with T replaced by t) that

$$|R_{\varepsilon}^{1}(v^{\varepsilon})| + |R_{\varepsilon}^{2}(v^{\varepsilon})| \le c(1+t)^{1/2} \left(\int_{0}^{t} \|v^{\varepsilon}\|_{H^{1}(\Gamma)}^{2} \, ds \right)^{1/2}.$$
 (5.10.47)

Now we deduce from (5.10.43) - (5.10.47) that

$$\begin{split} \|g^{1/2}v^{\varepsilon}(t)\|_{L^{2}(\Gamma)}^{2} + \int_{0}^{t} \|\nabla_{\Gamma}v^{\varepsilon}\|_{L^{2}(\Gamma)}^{2} \, ds \\ &\leq c \left\{ \|g^{1/2}v^{\varepsilon}(0)\|_{L^{2}(\Gamma)}^{2} + \int_{0}^{t} \left(\|v^{\varepsilon}\|_{L^{2}(\Gamma)}^{2} + \|M_{\tau}\mathbb{P}_{\varepsilon}f^{\varepsilon}\|_{H^{-1}(\Gamma,T\Gamma)} \|v^{\varepsilon}\|_{H^{1}(\Gamma)} \right) ds \right\} \\ &+ c(1+t)^{1/2} \left(\int_{0}^{t} \|v^{\varepsilon}\|_{H^{1}(\Gamma)}^{2} \, ds \right)^{1/2}. \end{split}$$

Noting that $\|v^{\varepsilon}\|_{H^1(\Gamma)}^2 = \|v^{\varepsilon}\|_{L^2(\Gamma)}^2 + \|\nabla_{\Gamma}v^{\varepsilon}\|_{L^2(\Gamma)}^2$, we apply Young's inequality to the last two terms of the above inequality to get

$$\begin{split} \|g^{1/2}v^{\varepsilon}(t)\|_{L^{2}(\Gamma)}^{2} &+ \int_{0}^{t} \|\nabla_{\Gamma}v^{\varepsilon}\|_{L^{2}(\Gamma)}^{2} ds \\ &\leq c \left\{ \|g^{1/2}v^{\varepsilon}(0)\|_{L^{2}(\Gamma)}^{2} + \int_{0}^{t} \left(\|v^{\varepsilon}\|_{L^{2}(\Gamma)}^{2} + \|M_{\tau}\mathbb{P}_{\varepsilon}f^{\varepsilon}\|_{H^{-1}(\Gamma,T\Gamma)}^{2} \right) ds + 1 + t \right\} \\ &+ \frac{1}{2} \int_{0}^{t} \|\nabla_{\Gamma}v^{\varepsilon}\|_{L^{2}(\Gamma)}^{2} ds. \end{split}$$

Then we make the integral of $\|\nabla_{\Gamma} v^{\varepsilon}\|_{L^{2}(\Gamma)}^{2}$ on the right-hand side absorbed into the left-hand side and use the inequalities (5.1.10) with $\beta = 1$, (5.2.30), and

$$\|g^{1/2}v^{\varepsilon}(0)\|_{L^{2}(\Gamma)} \leq c\|v^{\varepsilon}(0)\|_{L^{2}(\Gamma)} \leq c\|M_{\tau}u^{\varepsilon}(0)\|_{L^{2}(\Gamma)} = c\|M_{\tau}u^{\varepsilon}_{0}\|_{L^{2}(\Gamma)}$$

by (5.9.30) with k = 0 (note that $v^{\varepsilon} = \mathbb{P}_g M_{\tau} u^{\varepsilon}$) and $u^{\varepsilon}(0) = u_0^{\varepsilon}$ in V_{ε} to obtain

$$\|v^{\varepsilon}(t)\|_{L^{2}(\Gamma)}^{2} + \int_{0}^{t} \|\nabla_{\Gamma} v^{\varepsilon}\|_{L^{2}(\Gamma)}^{2} ds \le c \left(1 + t + \int_{0}^{t} \|v^{\varepsilon}\|_{L^{2}(\Gamma)}^{2} ds\right)$$
(5.10.48)

for all $t \in [0, T]$. From this inequality we deduce that

$$\|v^{\varepsilon}(t)\|_{L^{2}(\Gamma)}^{2} + 1 \le c \left\{ 1 + \int_{0}^{t} \left(\|v^{\varepsilon}\|_{L^{2}(\Gamma)}^{2} + 1 \right) ds \right\}, \quad t \in [0, T]$$

and thus Gronwall's inequality yields

$$\|v^{\varepsilon}(t)\|_{L^{2}(\Gamma)}^{2} + 1 \le ce^{ct} \le ce^{cT}, \quad t \in [0, T].$$

Applying this inequality to (5.10.48) with t = T we also get

$$\int_0^T \|\nabla_{\Gamma} v^{\varepsilon}\|_{L^2(\Gamma)}^2 dt \le c(1+T+e^{cT}).$$

Hence we conclude that (5.10.42) holds with $c_T := c(1 + T + e^{cT})$, where c > 0 is a constant independent of ε and T.

As a consequence of (5.10.36) and (5.10.42) we obtain the energy estimate for $M_{\tau} u^{\varepsilon}$.

Corollary 5.10.12. Let u^{ε} be as in Lemma 5.10.6. Then

$$\max_{t \in [0,T]} \|M_{\tau} u^{\varepsilon}(t)\|_{L^{2}(\Gamma)}^{2} + \int_{0}^{T} \|\nabla_{\Gamma} M_{\tau} u^{\varepsilon}(t)\|_{L^{2}(\Gamma)}^{2} dt \le c_{T}$$
(5.10.49)

for all $T \ge 0$, where $c_T > 0$ is a constant depending on T and independent of ε .

5.10.4 Estimate for the time derivative of the average

By the energy estimate (5.10.49) we observe that (a subsequence of) $M_{\tau}u^{\varepsilon}$ converges weakly in an appropriate function space. However, to show the convergence of the trilinear term in (5.10.27) we also require the strong convergence of $M_{\tau}u^{\varepsilon}$, which is proved by the Aubin–Lions lemma. For this purpose, let us estimate the time derivative of $M_{\tau}u^{\varepsilon}$. We first construct an appropriate test function.

Lemma 5.10.13. For each $w \in H^1(\Gamma, T\Gamma)$ there exist $\eta \in V_g$ and $q \in H^2(\Gamma)$ such that $w = g\eta + g\nabla_{\Gamma}q$ and

$$\|\eta\|_{H^1(\Gamma)} \le c \|w\|_{H^1(\Gamma)},\tag{5.10.50}$$

where c > 0 is a constant independent of w.

Proof. Let $w \in H^1(\Gamma, T\Gamma)$ and $\xi := -\operatorname{div}_{\Gamma} w \in L^2(\Gamma)$. Since w is tangential on the closed surface Γ , the integral of ξ over Γ vanishes by the Stokes theorem. Also,

$$\|q\|_{L^{2}(\Gamma)} \leq c \|\nabla_{\Gamma} q\|_{L^{2}(\Gamma)} \leq c \|g^{1/2} \nabla_{\Gamma} q\|_{L^{2}(\Gamma)}$$
(5.10.51)

for all $q \in H^1(\Gamma)$ with $\int_{\Gamma} q \, d\mathcal{H}^2 = 0$ by Poincaré's inequality (5.2.20) and (5.2.30). Hence the Lax–Milgram theorem shows that the problem

$$\operatorname{div}_{\Gamma}(g\nabla_{\Gamma}q) = -\xi \quad \text{on} \quad \Gamma, \quad \int_{\Gamma} q \, d\mathcal{H}^2 = 0$$

admits a unique weak solution $q \in H^1(\Gamma)$ in the sense that

$$(g\nabla_{\Gamma}q, \nabla_{\Gamma}\varphi)_{L^{2}(\Gamma)} = (\xi, \varphi)_{L^{2}(\Gamma)} \quad \text{for all} \quad \varphi \in H^{1}(\Gamma).$$
(5.10.52)

From this equality with $\varphi = q$ and (5.10.51) we deduce that

$$\|q\|_{H^{1}(\Gamma)} \leq c \|\xi\|_{L^{2}(\Gamma)} = c \|\operatorname{div}_{\Gamma} w\|_{L^{2}(\Gamma)} \leq c \|w\|_{H^{1}(\Gamma)}.$$
(5.10.53)

Moreover, replacing φ by $g^{-1}\varphi$ in (5.10.52) we get

$$(\nabla_{\Gamma} q, \nabla_{\Gamma} \varphi)_{L^{2}(\Gamma)} = (g^{-1}(\xi + \nabla_{\Gamma} g \cdot \nabla_{\Gamma} q), \varphi)_{L^{2}(\Gamma)} \quad \text{for all} \quad \varphi \in H^{1}(\Gamma),$$

which combined with (5.2.20) shows that q is a unique weak solution to

$$\Delta_{\Gamma}\psi = -g^{-1}(\xi + \nabla_{\Gamma}g \cdot \nabla_{\Gamma}q) \in L^{2}(\Gamma), \quad \int_{\Gamma} \psi \, d\mathcal{H}^{2} = 0.$$

(Note that the integral of the source term over Γ vanishes by (5.10.52).) Hence Lemma 5.9.12 and the inequalities (5.2.30) and (5.10.53) imply that $q \in H^2(\Gamma)$ and

$$\|q\|_{H^{2}(\Gamma)} \leq c\|g^{-1}(\xi + \nabla_{\Gamma}g \cdot \nabla_{\Gamma}q)\|_{L^{2}(\Gamma)} \leq c\|w\|_{H^{1}(\Gamma)}.$$
(5.10.54)

Now let $\eta := g^{-1}w - \nabla_{\Gamma}q$ on Γ . Then by $q \in H^2(\Gamma)$ and $\operatorname{div}_{\Gamma}(g\nabla_{\Gamma}q) = \operatorname{div}_{\Gamma}w$ on Γ we have $\eta \in V_q$. Moreover, from (5.2.30) and (5.10.54) it follows that

$$\|\eta\|_{H^{1}(\Gamma)} \leq c \left(\|w\|_{H^{1}(\Gamma)} + \|\nabla_{\Gamma}q\|_{H^{1}(\Gamma)}\right) \leq c \|w\|_{H^{1}(\Gamma)}$$

Hence we obtain $w = g\eta + g\nabla_{\Gamma}q$ and (5.10.50).

As in the previous section, we estimate the time derivative of v^{ε} and then derive an estimate for the time derivative of $M_{\tau}u^{\varepsilon}$ by using a difference estimate.

Lemma 5.10.14. Let u^{ε} be as in Lemma 5.10.6 and $v^{\varepsilon} = \mathbb{P}_{q}M_{\tau}u^{\varepsilon}$. Then

$$\|\partial_t v^{\varepsilon}\|_{L^2(0,T;H^{-1}(\Gamma,T\Gamma))} \le c_T \tag{5.10.55}$$

for all T > 0, where $c_T > 0$ is a constant depending on T and independent of ε .

Proof. Let $w \in L^2(0,T; H^1(\Gamma,T\Gamma))$. By Lemma 5.10.13 we can take $\eta \in L^2(0,T; V_g)$ and $q \in L^2(0,T; H^2(\Gamma))$ such that $w = g\eta + g\nabla_{\Gamma}q$. Since $\partial_t v^{\varepsilon}(t) \in L^2_{g\sigma}(\Gamma,T\Gamma)$ and $g\nabla_{\Gamma}q(t) \in L^2_{g\sigma}(\Gamma,T\Gamma)^{\perp}$ for a.a. $t \in (0,T)$ by Lemmas 5.9.10 and 5.10.10,

$$\int_0^T (\partial_t v^\varepsilon, g \nabla_\Gamma q)_{L^2(\Gamma)} \, dt = 0.$$

By this equality and $g\eta = w - g\nabla_{\Gamma}q$ we have

$$\int_0^T (g\partial_t v^\varepsilon, \eta)_{L^2(\Gamma)} dt = \int_0^T (\partial_t v^\varepsilon, g\eta)_{L^2(\Gamma)} dt = \int_0^T (\partial_t v^\varepsilon, w)_{L^2(\Gamma)} dt.$$

We substitute $\eta = g^{-1}w - \nabla_{\Gamma}q$ for (5.10.37) and use this equality. Then

$$\int_{0}^{T} (\partial_{t} v^{\varepsilon}, w)_{L^{2}(\Gamma)} dt = -\int_{0}^{T} a_{g}(v^{\varepsilon}, \eta) dt - \int_{0}^{T} b_{g}(v^{\varepsilon}, v^{\varepsilon}, \eta) dt + \int_{0}^{T} (gM_{\tau} \mathbb{P}_{\varepsilon} f^{\varepsilon}, \eta)_{L^{2}(\Gamma)} dt + R_{\varepsilon}^{1}(\eta) + R_{\varepsilon}^{2}(\eta), \quad (5.10.56)$$

where $R_{\varepsilon}^{1}(\eta)$ and $R_{\varepsilon}^{2}(\eta)$ are given in Lemmas 5.10.6 and 5.10.10. To the first term on the right-hand side we apply (5.2.30), (5.10.42), and (5.10.50) to get

$$\left| \int_{0}^{T} a_{g}(v^{\varepsilon}, \eta) \, dt \right| \leq c \|v^{\varepsilon}\|_{L^{2}(0,T;H^{1}(\Gamma))} \|\eta\|_{L^{2}(0,T;H^{1}(\Gamma))} \leq c_{T} \|w\|_{L^{2}(0,T;H^{1}(\Gamma))}.$$

Here and in what follows we denote by c_T a general positive constant depending on T and independent of ε . Also, by (5.10.12), (5.10.42), and (5.10.50) we see that

$$\begin{aligned} \left| \int_{0}^{T} b_{g}(v^{\varepsilon}, v^{\varepsilon}, \eta) \, dt \right| &\leq c \int_{0}^{T} \|v^{\varepsilon}\|_{L^{2}(\Gamma)} \|v^{\varepsilon}\|_{H^{1}(\Gamma)} \|\eta\|_{H^{1}(\Gamma)} \, dt \\ &\leq c \|v^{\varepsilon}\|_{L^{\infty}(0,T;L^{2}(\Gamma))} \|v^{\varepsilon}\|_{L^{2}(0,T;H^{1}(\Gamma))} \|\eta\|_{L^{2}(0,T;H^{1}(\Gamma))} \\ &\leq c_{T} \|w\|_{L^{2}(0,T;H^{1}(\Gamma))}. \end{aligned}$$

For the other terms we proceed as in the proof of Lemma 5.10.11 (see (5.10.46) and (5.10.47)) and use (5.1.10) with $\beta = 1$ and (5.10.50). Then we get

$$\left| \int_{0}^{T} (gM_{\tau} \mathbb{P}_{\varepsilon} f^{\varepsilon}, \eta)_{L^{2}(\Gamma)} dt \right| \leq c \int_{0}^{T} \|M_{\tau} \mathbb{P}_{\varepsilon} f^{\varepsilon}\|_{H^{-1}(\Gamma, T\Gamma)} \|\eta\|_{H^{1}(\Gamma)} dt$$
$$\leq cT^{1/2} \|\eta\|_{L^{2}(0, T; H^{1}(\Gamma))} \leq cT^{1/2} \|w\|_{L^{2}(0, T; H^{1}(\Gamma))}$$

and

$$|R_{\varepsilon}^{1}(\eta)| + |R_{\varepsilon}^{2}(\eta)| \le c(1+T)^{1/2} \|\eta\|_{L^{2}(0,T;H^{1}(\Gamma))} \le c(1+T)^{1/2} \|w\|_{L^{2}(0,T;H^{1}(\Gamma))}.$$

Applying these inequalities to the right-hand side of (5.10.56) we obtain

$$\left|\int_0^T (\partial_t v^{\varepsilon}, w)_{L^2(\Gamma)} dt\right| \le c_T \|w\|_{L^2(0,T;H^1(\Gamma))}$$

for all $w \in L^2(0,T; H^1(\Gamma,T\Gamma))$ (note that w is not necessarily a weighted solenoidal vector field). Hence (5.10.55) holds.

Corollary 5.10.15. Let u^{ε} be as in Lemma 5.10.6. Then

$$\|\partial_t M_\tau u^\varepsilon\|_{L^2(0,T;H^{-1}(\Gamma,T\Gamma))} \le c_T \tag{5.10.57}$$

for all T > 0, where $c_T > 0$ is a constant depending on T and independent of ε .

Proof. Let $v^{\varepsilon} = \mathbb{P}_{g} M_{\tau} u^{\varepsilon}$. Noting that $\|v\|_{H^{-1}(\Gamma, T\Gamma)} \leq \|v\|_{L^{2}(\Gamma)}$ for $v \in L^{2}(\Gamma, T\Gamma)$, we see by (5.8.36) and (5.10.35) that

$$\begin{aligned} \|\partial_t M_\tau u^\varepsilon - \partial_t v^\varepsilon\|_{L^2(0,T;H^{-1}(\Gamma,T\Gamma))} &\leq \varepsilon^{1/2} \|\partial_t u^\varepsilon\|_{L^2(0,T;L^2(\Omega_\varepsilon))} \\ &\leq c\varepsilon^{\alpha/2} (1+T)^{1/2} \leq c(1+T)^{1/2}. \end{aligned}$$

Combining this inequality and (5.10.55) we obtain (5.10.57).

Remark 5.10.16. In construction of a weak solution to the Navier–Stokes equations, we usually estimate the time derivative of an approximate solution in the dual of a solenoidal space. However, in Lemma 5.10.14 we proved an estimate for $\partial_t v^{\varepsilon}$ in $H^{-1}(\Gamma, T\Gamma)$, not in the dual V'_g of V_g . This is because we multiply $\partial_t v^{\varepsilon}$ by g in (5.10.37). When $f \in V'_g$, we can not define a functional $gf: v \mapsto_{V'_g} \langle f, gv \rangle_{V_g}$ for $v \in V_g$ since gv is not in V_g in general (here $V'_g \langle \cdot, \cdot \rangle_{V_g}$ stands for the duality product between V'_g and V_g). Since this issue does not occur for a functional in $H^{-1}(\Gamma, T\Gamma)$ (see (5.2.27)), we consider $\partial_t v^{\varepsilon}$ and $\partial_t M_\tau u^{\varepsilon}$ in this space.

5.10.5 Weak convergence of the average and characterization of the limit

The goal of this subsection is to prove Theorem 5.1.3. We proceed as in the case of the two-dimensional Navier–Stokes equations (see e.g. [8, 10, 61, 64]). First we give a definition of a weak solution to the limit equations (5.1.13)-(5.1.14) based on (5.10.27).

Definition 5.10.17. Let T > 0, $v_0 \in L^2_{g\sigma}(\Gamma, T\Gamma)$, and $f \in L^2(0, T; H^{-1}(\Gamma, T\Gamma))$. We say that a vector field

$$v \in L^{\infty}(0,T;L^2_{g\sigma}(\Gamma,T\Gamma)) \cap L^2(0,T;V_g) \quad \text{with} \quad \partial_t v \in L^1(0,T;H^{-1}(\Gamma,T\Gamma))$$

is a weak solution to the equations (5.1.13)–(5.1.14) on [0, T) if it satisfies

$$\int_{0}^{T} \{ [g\partial_{t}v,\eta]_{T\Gamma} + a_{g}(v,\eta) + b_{g}(v,v,\eta) \} dt = \int_{0}^{T} [gf,\eta]_{T\Gamma} dt$$
(5.10.58)

for all $\eta \in C_c(0,T;V_g)$ and $v|_{t=0} = v_0$ in $H^{-1}(\Gamma,T\Gamma)$.

Definition 5.10.18. Let $v_0 \in L^2_{g\sigma}(\Gamma, T\Gamma)$ and $f \in L^2_{loc}([0, \infty); H^{-1}(\Gamma, T\Gamma))$. We say that v is a weak solution to (5.1.13)–(5.1.14) on $[0, \infty)$ if it is a weak solution to (5.1.13)–(5.1.14) on [0, T) for all T > 0.

For T > 0, a weak solution to (5.1.13)-(5.1.14) on [0, T) is continuous on [0, T] with values in $H^{-1}(\Gamma, T\Gamma)$ and thus the initial condition makes sense. In fact, it becomes a continuous function with values in $L^2(\Gamma, T\Gamma)$.

Lemma 5.10.19. Let T > 0 and $f \in L^2(0,T; H^{-1}(\Gamma,T\Gamma))$. Suppose that

$$v \in L^{\infty}(0,T; L^2_{q\sigma}(\Gamma,T\Gamma)) \cap L^2(0,T; V_g) \quad with \quad \partial_t v \in L^1(0,T; H^{-1}(\Gamma,T\Gamma))$$

satisfies (5.10.58) for all $\eta \in C_c(0,T;V_q)$. Then

$$v \in C([0,T]; L^2_{q\sigma}(\Gamma,T\Gamma)), \quad \partial_t v \in L^2(0,T; H^{-1}(\Gamma,T\Gamma))$$

and (5.10.58) is valid for all $\eta \in L^2(0,T;V_q)$.

Note that here the initial condition $v|_{t=0} = v_0$ in $H^{-1}(\Gamma, T\Gamma)$ is not imposed.

Proof. We estimate $\partial_t v$ as in the proof of Lemma 5.10.14, where we used the fact that $\partial_t v^{\varepsilon}(t) \in L^2_{q\sigma}(\Gamma, T\Gamma)$ for a.a. $t \in (0, T)$. This is not valid for v, but we have

$$[\partial_t v(t), g\nabla_{\Gamma} q]_{T\Gamma} = 0 \quad \text{for all } q \in H^2(\Gamma) \text{ and a.a. } t \in (0, T).$$
(5.10.59)

Indeed, for all $\xi \in C_c^{\infty}(0,T)$ we have

$$\int_0^T \xi(t) [\partial_t v(t), g \nabla_{\Gamma} q]_{T\Gamma} dt = -\int_0^T \partial_t \xi(t) [v(t), g \nabla_{\Gamma} q]_{T\Gamma} dt$$
$$= -\int_0^T \partial_t \xi(t) (v(t), g \nabla_{\Gamma} q)_{L^2(\Gamma)} dt = 0$$

by $v(t) \in L^2_{g\sigma}(\Gamma, T\Gamma)$ for a.a. $t \in (0, T)$ and $g\nabla_{\Gamma}q \in L^2_{g\sigma}(\Gamma, T\Gamma)^{\perp}$ (see Lemma 5.9.10). Hence (5.10.59) is valid. Now let $w \in C_c(0, T; H^1(\Gamma, T\Gamma))$. By Lemma 5.10.13 we can take $\eta \in C_c(0, T; V_g)$ and $q \in C_c(0, T; H^2(\Gamma))$ such that $w = g\eta + g\nabla_{\Gamma}q$. Moreover,

$$\int_0^T [\partial_t v, w]_{T\Gamma} dt = \int_0^T \left([\partial_t v, g\eta]_{T\Gamma} + [\partial_t v, g\nabla_{\Gamma} q]_{T\Gamma} \right) dt = \int_0^T [g\partial_t v, \eta]_{T\Gamma} dt$$

by (5.10.59). We substitute η for (5.10.58). Then using the above equality, (5.10.50), and

$$\left| \int_{0}^{T} b_{g}(v, v, \eta) \, dt \right| \leq c \|v\|_{L^{\infty}(0,T;L^{2}(\Gamma))} \|v\|_{L^{2}(0,T;H^{1}(\Gamma))} \|\eta\|_{L^{2}(0,T;H^{1}(\Gamma))}$$
(5.10.60)

by (5.10.12) we calculate as in the proof of Lemma 5.10.14 to get

$$\left| \int_0^T [\partial_t v, w]_{T\Gamma} dt \right| \le c \|w\|_{L^2(0,T;H^1(\Gamma))} \quad \text{for all} \quad w \in C_c(0,T;H^1(\Gamma,T\Gamma)).$$

Since $C_c(0,T; H^1(\Gamma,T\Gamma))$ is dense in $L^2(0,T; H^1(\Gamma,T\Gamma))$, this inequality implies

$$\partial_t v \in L^2(0, T; H^{-1}(\Gamma, T\Gamma)).$$
 (5.10.61)

Combining this property with

$$v \in L^2(0,T;V_g) \subset L^2(0,T;H^1(\Gamma,T\Gamma))$$

we apply the interpolation result of Lions–Magenes [37, Chapter 1, Theorem 3.1] (see also [64, Chapter III, Lemma 1.2]) to v to get

$$v \in C([0,T]; L^2(\Gamma, T\Gamma)).$$

Moreover, since $v \in L^{\infty}(0,T; L^2_{g\sigma}(\Gamma,T\Gamma))$, the vector field v(t) is in $L^2_{g\sigma}(\Gamma,T\Gamma)$ for a.a. $t \in (0,T)$ and, in particular, for all t in a dense subset of [0,T]. Thus, by the continuity of v(t) on [0,T] in $L^2(\Gamma,T\Gamma)$ and the fact that $L^2_{g\sigma}(\Gamma,T\Gamma)$ is closed in $L^2(\Gamma,T\Gamma)$, we observe that $v(t) \in L^2_{q\sigma}(\Gamma,T\Gamma)$ for all $t \in [0,T]$ and

$$v \in C([0,T]; L^2_{a\sigma}(\Gamma,T\Gamma)).$$

Finally, since $C_c(0, T; V_g)$ is dense in $L^2(0, T; V_g)$ and both sides of (5.10.58) are linear and continuous for $\eta \in L^2(0, T; V_g)$ by (5.10.60) and (5.10.61), the equality (5.10.58) is also valid for all $\eta \in L^2(0, T; V_g)$.

By Lemma 5.10.19 the value of a weak solution to (5.1.13)-(5.1.14) at t = 0 is well-defined as a vector field on Γ . Hence we can consider the initial condition $v|_{t=0} = v_0$ as an equality for vector fields on Γ . Let us show the uniqueness of a weak solution to (5.1.13)-(5.1.14) and the existence of an associated pressure.

Lemma 5.10.20. For given $v_0 \in L^2_{g\sigma}(\Gamma, T\Gamma)$ and $f \in L^2(0, T; H^{-1}(\Gamma, T\Gamma))$, T > 0 there exists at most one weak solution to (5.1.13)–(5.1.14) on [0, T).

Proof. Let v_1 and v_2 be weak solutions to (5.1.13)–(5.1.14) and $w := v_1 - v_2$. Then

$$w \in C([0,T]; L^2(\Gamma, T\Gamma)), \quad \partial_t w \in L^2(0,T; H^{-1}(\Gamma, T\Gamma))$$
 (5.10.62)

by Lemma 5.10.19 and $w|_{t=0} = 0$ on Γ . Moreover, subtracting the weak formulation (5.10.58) for v_2 from that for v_1 we get

$$\int_0^T \{ [g\partial_t w, \eta]_{T\Gamma} + a_g(w, \eta) + b_g(w, v_1, \eta) + b_g(v_2, w, \eta) \} \, ds = 0 \tag{5.10.63}$$

for all $\eta \in L^2(0,T;V_g)$. For each $t \in [0,T]$ let $\eta(s) := 1_{[0,t]}(s)w(s)$, $s \in [0,T]$, where $1_{[0,t]}$ is the characteristic function of the time interval [0,t]. Since $\eta \in L^2(0,T;V_g)$ we can substitute it for (5.10.63). Then we use

$$\int_{0}^{t} [g\partial_{s}w,w]_{T\Gamma} ds = \frac{1}{2} \int_{0}^{t} \frac{d}{ds} \|g^{1/2}w\|_{L^{2}(\Gamma)}^{2} ds$$

$$= \frac{1}{2} \|g^{1/2}w(t)\|_{L^{2}(\Gamma)}^{2} - \frac{1}{2} \|g^{1/2}w(0)\|_{L^{2}(\Gamma)}^{2}$$

$$\ge c \left(\|w(t)\|_{L^{2}(\Gamma)}^{2} - \|w(0)\|_{L^{2}(\Gamma)}^{2}\right)$$
(5.10.64)

by (5.10.62) and the fact that g is bounded on Γ from above and below by positive constants (see (5.2.30)), the inequality (5.10.11), and

$$|b_g(w, v_1, w)| = |b_g(w, w, v_1)| \le c ||w||_{L^2(\Gamma)} ||w||_{H^1(\Gamma)} ||v_1||_{H^1(\Gamma)}$$

and $b_g(v_2, w, w) = 0$ by (5.10.12) and (5.10.13) to obtain

$$\begin{aligned} \|w(t)\|_{L^{2}(\Gamma)}^{2} + \int_{0}^{t} \|\nabla_{\Gamma}w\|_{L^{2}(\Gamma)}^{2} ds \\ &\leq c \left\{ \|w(0)\|_{L^{2}(\Gamma)}^{2} + \int_{0}^{t} \left(\|w\|_{L^{2}(\Gamma)}^{2} + \|w\|_{L^{2}(\Gamma)} \|w\|_{H^{1}(\Gamma)} \|v_{1}\|_{H^{1}(\Gamma)} \right) ds \right\}. \end{aligned}$$

We further apply Young's inequality to the last term to get

$$\begin{split} \|w(t)\|_{L^{2}(\Gamma)}^{2} &+ \int_{0}^{t} \|\nabla_{\Gamma} w\|_{L^{2}(\Gamma)}^{2} \, ds \\ &\leq c \left\{ \|w(0)\|_{L^{2}(\Gamma)}^{2} + \int_{0}^{t} \left(1 + \|v_{1}\|_{H^{1}(\Gamma)}^{2}\right) \|w\|_{L^{2}(\Gamma)}^{2} \, ds \right\} + \frac{1}{2} \int_{0}^{t} \|\nabla_{\Gamma} w\|_{L^{2}(\Gamma)}^{2} \, ds. \end{split}$$

Then we subtract the half of the integral of $\|\nabla_{\Gamma} w\|_{L^2(\Gamma)}^2$ from both sides and use $w|_{t=0} = 0$ on Γ to find that

$$\|w(t)\|_{L^{2}(\Gamma)}^{2} \leq c \int_{0}^{t} \left(1 + \|v_{1}\|_{H^{1}(\Gamma)}^{2}\right) \|w\|_{L^{2}(\Gamma)}^{2} ds \quad \text{for all} \quad t \in [0, T].$$

(Here we omit the integral of $\|\nabla_{\Gamma} w\|_{L^{2}(\Gamma)}^{2}$ on the left-hand side.) Since $1 + \|v_{1}\|_{H^{1}(\Gamma)}^{2}$ is integrable on (0,T), we can use Gronwall's inequality to the above inequality to get $\|w(t)\|_{L^{2}(\Gamma)}^{2} = 0$ for all $t \in [0,T]$. Hence $v_{1} = v_{2}$.

Lemma 5.10.21. Let v be a weak solution to (5.1.13)-(5.1.14) on [0,T), T > 0. Then there exists a unique $\hat{q} \in C([0,T]; L^2(\Gamma))$ such that $\int_{\Gamma} \hat{q}(t) d\mathcal{H}^2 = 0$ for all $t \in [0,T]$ and

$$g\left(\partial_t v + \overline{\nabla}_v v\right) - 2\nu \left\{ P \operatorname{div}_{\Gamma}[gD_{\Gamma}(v)] - \frac{1}{g} (\nabla_{\Gamma} g \otimes \nabla_{\Gamma} g) v \right\}$$
$$+ (\gamma^0 + \gamma^1) v + g \nabla_{\Gamma} q = gf \quad in \quad \mathcal{D}'(0, T; H^{-1}(\Gamma, T\Gamma)) \quad (5.10.65)$$

with $q := \partial_t \hat{q} \in \mathcal{D}'(0, T; L^2(\Gamma))$ (see Section 5.2.1).

Here $\overline{\nabla}_v v = P(v \cdot \nabla_{\Gamma})v$ is the covariant derivative of v along itself (see Appendix 5.C). Also, recall that we identity $H^{-1}(\Gamma, T\Gamma)$ with quotient space

$$\mathcal{Q} = \{ [f] \mid f \in H^{-1}(\Gamma)^3 \}, \quad [f] = \{ \tilde{f} \in H^{-1}(\Gamma)^3 \mid Pf = P\tilde{f} \text{ in } H^{-1}(\Gamma)^3 \}$$

and take Pf (or f when Pf = f in $H^{-1}(\Gamma)^3$) as a representative of the equivalence class [f] to write $[Pf, v]_{T\Gamma} = \langle f, v \rangle_{\Gamma}$ for $v \in H^1(\Gamma, T\Gamma)$ (see Section 5.2.1).

Proof. Let $A \in L^2(\Gamma)^{3\times 3}$ and $\eta \in H^1(\Gamma, T\Gamma)$. If $A^T = A$ and PA = AP = A on Γ , then by (5.2.23) and An = APn = 0 we see that

$$\begin{split} \left(gA, D_{\Gamma}(\eta)\right)_{L^{2}(\Gamma)} &= (gA, \nabla_{\Gamma}\eta)_{L^{2}(\Gamma)} = \sum_{i,j=1}^{3} (gA_{ij}, \underline{D}_{i}\eta_{j})_{L^{2}(\Gamma)} \\ &= -\sum_{i,j=1}^{3} \left\{ \langle \underline{D}_{i}(gA_{ij}), \eta_{j} \rangle_{\Gamma} + (gA_{ij}Hn_{i}, \eta_{j})_{L^{2}(\Gamma)} \right\} \\ &= -\left\{ \langle \operatorname{div}_{\Gamma}(gA), \eta \rangle_{\Gamma} + (gHAn, \eta)_{L^{2}(\Gamma)} \right\} = -[P\operatorname{div}_{\Gamma}(gA), \eta]_{T\Gamma}. \end{split}$$

Also, for $v \in V_g$ and $\eta \in H^1(\Gamma, T\Gamma)$ we use (5.10.13) and $v \otimes \eta : \nabla_{\Gamma} v = (v \cdot \nabla_{\Gamma}) v \cdot \eta$ to get

$$b_g(v,v,\eta) = -b_g(v,\eta,v) = \left(g(v\cdot\nabla_{\Gamma})v,\eta\right)_{L^2(\Gamma)} = \left(gP(v\cdot\nabla_{\Gamma})v,\eta\right)_{L^2(\Gamma)} = \left(g\overline{\nabla}_v v,\eta\right)_{L^2(\Gamma)} = \left(g\overline{\nabla}_v v,\eta\right$$

Let v be a weak solution to (5.1.13)–(5.1.14) on [0, T). We apply the above equalities with $A = D_{\Gamma}(v)$ and $(v \cdot \nabla_{\Gamma} g)(\eta \cdot \nabla_{\Gamma} g) = (\nabla_{\Gamma} g \otimes \nabla_{\Gamma} g)v \cdot \eta$ to (5.10.58) to obtain

$$\int_{0}^{T} \left([g\partial_{t}v,\eta]_{T\Gamma} + [A_{g}v,\eta]_{T\Gamma} + [B_{g}(v,v),\eta]_{T\Gamma} \right) ds = \int_{0}^{T} [gf,\eta]_{T\Gamma} ds$$
(5.10.66)

for all $\eta \in L^2(0,T;V_g)$ (see Lemma 5.10.19), where $B_g(v,v) := g \overline{\nabla}_v v$ and

$$A_g v := -2\nu \left\{ P \operatorname{div}_{\Gamma}[g D_{\Gamma}(v)] - \frac{1}{g} (\nabla_{\Gamma} g \otimes \nabla_{\Gamma} g) v \right\} + (\gamma^0 + \gamma^1) v.$$
 (5.10.67)

Since $A_g v, B_g(v, v), f \in L^2(0, T; H^{-1}(\Gamma, T\Gamma))$ by Definition 5.10.17, the functions

$$\begin{split} \widehat{A}_g v(t) &:= \int_0^t A_g v(s) \, ds, \quad \widehat{B}_g(v, v)(t) := \int_0^t B_g(v(s), v(s)) \, ds, \\ \widehat{f}(t) &:= \int_0^t f(s) \, ds, \quad t \in [0, T] \end{split}$$

are continuous with values in $H^{-1}(\Gamma, T\Gamma)$. For each $t \in [0, T]$ and $\xi \in V_g$ we take a test function $\eta(s) := 1_{[0,t]}(s)\xi$, $s \in [0, T]$ in (5.10.66), where $1_{[0,t]} \colon \mathbb{R} \to \mathbb{R}$ is the characteristic function of [0, t]. Then since ξ is independent of time,

$$v \in H^1(0,T;H^{-1}(\Gamma,T\Gamma)) \subset C([0,T];H^{-1}(\Gamma,T\Gamma)),$$

and $v|_{t=0} = v_0$ in $H^{-1}(\Gamma, T\Gamma)$, we have $[F(t), \xi]_{T\Gamma} = 0$ for all $\xi \in V_g$, where

$$F := gv - gv_0 + \widehat{A}_g v + \widehat{B}_g(v, v) - g\widehat{f} \in C([0, T]; H^{-1}(\Gamma, T\Gamma)).$$

Hence by Theorem 5.9.5 there exists a unique $\hat{q}(t) \in L^2(\Gamma)$ such that

$$F(t) = -g\nabla_{\Gamma}\hat{q}(t)$$
 in $H^{-1}(\Gamma, T\Gamma)$, $\int_{\Gamma}\hat{q}(t) d\mathcal{H}^2 = 0$

for all $t \in [0,T]$. Moreover, by (5.9.20) and $F \in C([0,T]; H^{-1}(\Gamma, T\Gamma))$ we see that

$$\hat{q} \in C([0,T]; L^2(\Gamma)) \subset L^2(0,T; L^2(\Gamma))$$

and thus $q := \partial_t \hat{q} \in \mathcal{D}'(0,T;L^2(\Gamma))$ is well-defined. Now we have

$$0 = -\int_0^T \partial_t \varphi(t) \{ F(t) + g \nabla_{\Gamma} \hat{q}(t) \} dt = \int_0^T \varphi(t) \{ \partial_t F(t) + g \partial_t (\nabla_{\Gamma} \hat{q})(t) \} dt$$

in $H^{-1}(\Gamma, T\Gamma)$ for all $\varphi \in C_c^{\infty}(0, T)$, which means that

$$\partial_t F(t) + g \partial_t (\nabla_{\Gamma} \hat{q})(t) = 0 \quad \text{in} \quad \mathcal{D}'(0, T; H^{-1}(\Gamma, T\Gamma)).$$

Moreover, $\partial_t(\nabla_{\Gamma}\hat{q}) = \nabla_{\Gamma}q$ by (5.2.29) and

$$\partial_t F = g\partial_t v + A_g v + B_g(v, v) - gf = g\left(\partial_t v + \overline{\nabla}_v v\right) + A_g v - gf$$

with $A_a v$ given by (5.10.67). Hence (5.10.65) is valid.

Before starting the proof of Theorem 5.1.3, we give an auxiliary statement on the weak limit of the averaged tangential component of a vector field in $L^2_{\sigma}(\Omega_{\varepsilon})$ as $\varepsilon \to 0$.

Lemma 5.10.22. For $\varepsilon \in (0,1)$ let $u^{\varepsilon} \in L^2_{\sigma}(\Omega_{\varepsilon})$. Also, let $v \in L^2(\Gamma, T\Gamma)$. Suppose that $M_{\tau}u^{\varepsilon}$ converges to v weakly in $L^2(\Gamma, T\Gamma)$ as $\varepsilon \to 0$ and

$$\|u^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon})}^{2} \leq c\varepsilon^{-1+\alpha} \quad \text{for sufficiently small} \quad \varepsilon \in (0,1)$$

with some $c, \alpha > 0$. Then $v \in L^2_{q\sigma}(\Gamma, T\Gamma)$.

Proof. By (5.2.25) and the weak convergence of $\{M_{\tau}u^{\varepsilon}\}_{\varepsilon}$ to v in $L^{2}(\Gamma, T\Gamma)$ we see that $\operatorname{div}_{\Gamma}(gM_{\tau}u^{\varepsilon})$ converges to $\operatorname{div}_{\Gamma}(gv)$ weakly in $H^{-1}(\Gamma)$ as $\varepsilon \to 0$. Moreover, by (5.6.49) and the assumption on the $L^2(\Omega_{\varepsilon})$ -norm of u^{ε} we have

$$\|\operatorname{div}_{\Gamma}(gM_{\tau}u^{\varepsilon})\|_{H^{-1}(\Gamma)} \le c\varepsilon^{1/2} \|u^{\varepsilon}\|_{L^{2}(\Omega_{\varepsilon})} \le c\varepsilon^{\alpha/2}$$

for sufficiently small $\varepsilon \in (0, 1)$ with $\alpha > 0$. Hence

$$\|\operatorname{div}_{\Gamma}(gv)\|_{H^{-1}(\Gamma)} \leq \liminf_{\varepsilon \to 0} \|\operatorname{div}_{\Gamma}(gM_{\tau}u^{\varepsilon})\|_{H^{-1}(\Gamma)} = 0$$

and $\operatorname{div}_{\Gamma}(gv) = 0$ in $H^{-1}(\Gamma)$, which means that $v \in L^2_{q\sigma}(\Gamma, T\Gamma)$.

Now we are ready to prove Theorem 5.1.3.

Proof of Theorem 5.1.3. For $\varepsilon \in (0,1)$ suppose that the initial velocity u_0^{ε} and the external force f^{ε} satisfy the assumptions of Theorem 5.1.3. Then $\{M_{\tau}u_0^{\varepsilon}\}_{\varepsilon}$ and $\{M_{\tau}\mathbb{P}_{\varepsilon}f^{\varepsilon}\}_{\varepsilon}$ are bounded in $L^2(\Gamma, T\Gamma)$ and $L^{\infty}(0, \infty; H^{-1}(\Gamma, T\Gamma))$, respectively, by the condition (b). By this fact and the condition (a) we see that the inequalities (5.1.10) hold with $\beta = 1$ for $\varepsilon \in (0, \varepsilon_2)$. Hence Theorem 5.8.4 implies that there exists a global strong solution u^{ε} to (5.1.1)-(5.1.3) satisfying (5.8.34)-(5.8.36) for each $\varepsilon \in (0,\varepsilon_3)$ with $\varepsilon_3 := \min\{\varepsilon'_1,\varepsilon_2\}$, where $\varepsilon'_1 = \min\{\varepsilon_1, \varepsilon_\sigma\}$ with ε_1 and ε_σ given in Theorem 5.1.2 and Lemma 5.5.1. Moreover, by (5.6.10) and (5.8.34) we have

$$\sup_{\varepsilon \in [0,\infty)} \|Mu^{\varepsilon}(t) \cdot n\|_{L^{2}(\Gamma)} \le c\varepsilon^{1/2} \sup_{t \in [0,\infty)} \|u^{\varepsilon}(t)\|_{H^{1}(\Omega_{\varepsilon})} \le c\varepsilon^{\alpha/2} \to 0$$

as $\varepsilon \to 0$. Hence $\{Mu^{\varepsilon} \cdot n\}_{\varepsilon}$ converges to zero strongly in $C([0,\infty); L^2(\Gamma))$.

Now let us consider the averaged tangential component $M_{\tau}u^{\varepsilon}$. First note that the weak limit v_0 of $\{M_{\tau}u_0^{\varepsilon}\}_{\varepsilon}$ is in $L^2_{q\sigma}(\Gamma, T\Gamma)$ by the condition (a), the inequality (5.5.18), and Lemma 5.10.22. Since all results in the previous subsections apply to u^{ε} , for fixed T > 0we see by (5.10.49) and (5.10.57) that

- $\{M_{\tau}u^{\varepsilon}\}_{\varepsilon}$ is bounded in $L^{\infty}(0,T;L^2(\Gamma,T\Gamma)) \cap L^2(0,T;H^1(\Gamma,T\Gamma)),$
- $\{\partial_t M_\tau u^\varepsilon\}_\varepsilon$ is bounded in $L^2(0,T; H^{-1}(\Gamma,T\Gamma)).$

Hence there exist $\varepsilon_k \in (0, \varepsilon_3), k \in \mathbb{N}$ and

$$v \in L^{\infty}(0,T; L^2(\Gamma,T\Gamma)) \cap L^2(0,T; H^1(\Gamma,T\Gamma)) \text{ with } \partial_t v \in L^2(0,T; H^{-1}(\Gamma,T\Gamma))$$

such that $\lim_{k\to\infty} \varepsilon_k = 0$ and

$$\lim_{k \to \infty} M_{\tau} u^{\varepsilon_k} = v \quad \text{weakly-}\star \text{ in } L^{\infty}(0,T;L^2(\Gamma,T\Gamma)),$$

$$\lim_{k \to \infty} M_{\tau} u^{\varepsilon_k} = v \quad \text{weakly in } L^2(0,T;H^1(\Gamma,T\Gamma)),$$

$$\lim_{k \to \infty} \partial_t M_{\tau} u^{\varepsilon_k} = \partial_t v \quad \text{weakly in } L^2(0,T;H^{-1}(\Gamma,T\Gamma)).$$
(5.10.68)

Moreover, by the Aubin–Lions lemma (see e.g. [8, Theorem II.5.16]) there exists a subsequence of $\{M_{\tau}u^{\varepsilon_k}\}_{k=1}^{\infty}$, which we denote by $\{M_{\tau}u^{\varepsilon_k}\}_{k=1}^{\infty}$ again, such that

$$\lim_{k \to \infty} M_{\tau} u^{\varepsilon_k} = v \quad \text{strongly in} \quad L^2(0, T; L^2(\Gamma, T\Gamma)). \tag{5.10.69}$$

Then $v(t) \in L^2_{q\sigma}(\Gamma, T\Gamma)$ for a.a. $t \in (0, T)$ by (5.8.34) and Lemma 5.10.22. Hence

$$v \in L^{\infty}(0,T; L^2_{g\sigma}(\Gamma,T\Gamma)) \cap L^2(0,T;V_g).$$

Let us show that v satisfies (5.10.58) for all $\eta \in C_c(0,T;V_g)$. In what follows, we write c for a general positive constant that may depend on v and η but is independent of ε_k and u^{ε_k} . We consider the weak formulation (5.10.27) for $M_\tau u^{\varepsilon_k}$:

$$\int_{0}^{T} \{ [g\partial_{t}M_{\tau}u^{\varepsilon_{k}},\eta]_{T\Gamma} + a_{g}(M_{\tau}u^{\varepsilon_{k}},\eta) + b_{g}(M_{\tau}u^{\varepsilon_{k}},M_{\tau}u^{\varepsilon_{k}},\eta) \} dt$$
$$= \int_{0}^{T} [gM_{\tau}\mathbb{P}_{\varepsilon_{k}}f^{\varepsilon_{k}},\eta]_{T\Gamma} dt + R^{1}_{\varepsilon_{k}}(\eta). \quad (5.10.70)$$

Here $\partial_t M_\tau u^{\varepsilon_k}$ and $M_\tau \mathbb{P}_{\varepsilon_k} f^{\varepsilon_k}$ are considered in $H^{-1}(\Gamma, T\Gamma)$ (see Section 5.2.1). Let $k \to \infty$ in this equality. Noting that $[gF, w]_{T\Gamma} = [F, gw]_{T\Gamma}$ for $F \in H^{-1}(\Gamma, T\Gamma)$ and $w \in H^1(\Gamma, T\Gamma)$ (see Section 5.2.1), we deduce from the assumption (b) and (5.10.68) that

$$\lim_{k \to \infty} \int_0^T [g\partial_t M_\tau u^{\varepsilon_k}, \eta]_{T\Gamma} dt = \int_0^T [g\partial_t v, \eta]_{T\Gamma} dt,$$
$$\lim_{k \to \infty} \int_0^T a_g(M_\tau u^{\varepsilon_k}, \eta) dt = \int_0^T a_g(v, \eta) dt,$$
$$(5.10.71)$$
$$\lim_{k \to \infty} \int_0^T [gM_\tau \mathbb{P}_{\varepsilon_k} f^{\varepsilon_k}, \eta]_{T\Gamma} dt = \int_0^T [gf, \eta]_{T\Gamma} dt.$$

Also, by (5.10.28), the assumption (c), and $\alpha > 0$ we have

$$|R_{\varepsilon_k}^1(\eta)| \le c \left(\varepsilon_k^{\alpha/4} + \sum_{i=0,1} |\varepsilon_k^{-1} \gamma_{\varepsilon_k}^i - \gamma^i|\right) (1+T)^{1/2} \|\eta\|_{L^2(0,T;H^1(\Gamma))} \to 0$$
(5.10.72)

as $k \to \infty$. To show the convergence of the trilinear term, we set

$$I_1^k := \int_0^T b_g(M_\tau u^{\varepsilon_k}, M_\tau u^{\varepsilon_k}, \eta) \, dt - \int_0^T b_g(v, M_\tau u^{\varepsilon_k}, \eta) \, dt,$$
$$I_2^k := \int_0^T b_g(v, M_\tau u^{\varepsilon_k}, \eta) \, dt - \int_0^T b_g(v, v, \eta) \, dt.$$

Since $\|\eta(t)\|_{H^1(\Gamma)}$ is bounded on [0,T] by $\eta \in C_c(0,T;V_g)$, we see by (5.10.12) that

$$\begin{aligned} |I_1^k| &\leq c \int_0^T \|M_\tau u^{\varepsilon_k} - v\|_{L^2(\Gamma)}^{1/2} \|M_\tau u^{\varepsilon_k} - v\|_{H^1(\Gamma)}^{1/2} \|M_\tau u^{\varepsilon_k}\|_{H^1(\Gamma)} \|\eta\|_{H^1(\Gamma)} dt \\ &\leq c \|M_\tau u^{\varepsilon_k} - v\|_{L^2(0,T;L^2(\Gamma))}^{1/2} \|M_\tau u^{\varepsilon_k} - v\|_{L^2(0,T;H^1(\Gamma))}^{1/2} \|M_\tau u^{\varepsilon_k}\|_{L^2(0,T;H^1(\Gamma))}. \end{aligned}$$

Applying (5.10.49) and (5.10.69) to the last line we obtain

$$|I_1^k| \le c \|M_\tau u^{\varepsilon_k} - v\|_{L^2(0,T;L^2(\Gamma))}^{1/2} \to 0 \quad \text{as} \quad k \to \infty.$$
(5.10.73)

For I_2^k , we consider the linear functional

$$\Phi(\xi) := \int_0^T b_g(v,\xi,\eta) \, dt, \quad \xi \in L^2(0,T;H^1(\Gamma,T\Gamma)).$$

By (5.10.12) and the boundedness of $\|\eta(t)\|_{H^1(\Gamma)}$ on [0,T] we get

$$|\Phi(\xi)| \le c \|\eta\|_{L^{\infty}(0,T;H^{1}(\Gamma))} \|v\|_{L^{2}(0,T;H^{1}(\Gamma))} \|\xi\|_{L^{2}(0,T;H^{1}(\Gamma))}$$

for all $\xi \in L^2(0,T; H^1(\Gamma,T\Gamma))$. Hence Φ is bounded on $L^2(0,T; H^1(\Gamma,T\Gamma))$ and the weak convergence (5.10.68) in $L^2(0,T; H^1(\Gamma,T\Gamma))$ implies that

$$\lim_{k \to \infty} I_2^k = \lim_{k \to \infty} \{ \Phi(M_\tau u^{\varepsilon_k}) - \Phi(v) \} = 0.$$

Combining this equality with (5.10.73) we obtain

$$\lim_{k \to \infty} \int_0^T b_g(M_\tau u^{\varepsilon_k}, M_\tau u^{\varepsilon_k}, \eta) \, dt = \int_0^T b_g(v, v, \eta) \, dt.$$
(5.10.74)

We send $k \to \infty$ in (5.10.70) and apply (5.10.71), (5.10.72), and (5.10.74) to show that v satisfies (5.10.58) for all $\eta \in C_c(0,T;V_q)$. Moreover, by Lemma 5.10.19 we see that

$$v \in C([0,T], L^2_{g\sigma}(\Gamma, T\Gamma)), \quad \partial_t v \in L^2(0,T; H^{-1}(\Gamma, T\Gamma))$$

and (5.10.58) is valid for all $\eta \in L^2(0,T;V_g)$.

To show that v is a weak solution to (5.1.13)-(5.1.14) on [0,T) we also need to verify the initial condition. Let $\xi \in V_g$ and φ be a smooth function on [0,T] such that $\varphi(0) = 1$ and $\varphi(T) = 0$. We define $\eta \in L^2(0,T;V_g)$ by $\eta(t) := \varphi(t)\xi$ for $t \in [0,T]$ and substitute it for (5.10.58) and (5.10.70). Then we carry out integration by parts for $\partial_t v$ and $\partial_t M_\tau u^\varepsilon$ and use $\varphi(0) = 1$ and $\varphi(T) = 0$ to get

$$(gv(0),\xi)_{L^2(\Gamma)} = J_{\infty}, \quad (gM_{\tau}u_0^{\varepsilon_k},\xi)_{L^2(\Gamma)} = J_k,$$
(5.10.75)

where

$$J_{\infty} := -\int_{0}^{T} \partial_{t} \varphi(gv,\xi)_{L^{2}(\Gamma)} dt + \int_{0}^{T} \{a_{g}(v,\eta) + b_{g}(v,v,\eta)\} dt - \int_{0}^{T} [gf,\eta]_{T\Gamma} dt$$

and

$$\begin{aligned} J_k &:= -\int_0^T \partial_t \varphi(gM_\tau u^{\varepsilon_k}, \xi)_{L^2(\Gamma)} \, dt + \int_0^T \{a_g(M_\tau u^{\varepsilon_k}, \eta) + b_g(M_\tau u^{\varepsilon_k}, M_\tau u^{\varepsilon_k}, \eta)\} \, dt \\ &- \int_0^T [gM_\tau \mathbb{P}_{\varepsilon_k} f^{\varepsilon_k}, \eta]_{T\Gamma} \, dt - R^1_{\varepsilon_k}(\eta). \end{aligned}$$

We send $k \to \infty$ in the second equality of (5.10.75). Then the left-hand side converges to $(gv_0,\xi)_{L^2(\Gamma)}$ by the assumption (b). Also, we use (5.10.71), (5.10.72), (5.10.74), and

$$\lim_{k \to \infty} \int_0^T \partial_t \varphi(gM_\tau u^{\varepsilon_k}, \xi)_{L^2(\Gamma)} dt = \int_0^T \partial_t \varphi(gv, \xi)_{L^2(\Gamma)} dt$$

by the strong convergence (5.10.69) to find that $\lim_{k\to\infty} J_k = J_{\infty}$ (note that in the proof of (5.10.74) we only used the boundedness of $\|\eta(t)\|_{H^1(\Gamma)}$ on [0,T]). Hence

$$(gv(0),\xi)_{L^2(\Gamma)} = J_\infty = (gv_0,\xi)_{L^2(\Gamma)}$$
 for all $\xi \in V_g$.

Since V_g is dense in $L^2_{g\sigma}(\Gamma, T\Gamma)$ (see Lemma 5.9.13), the above equality is also valid for all $\xi \in L^2_{g\sigma}(\Gamma, T\Gamma)$. Thus, setting $\xi := v(0) - v_0$ we get

$$(g\{v(0) - v_0\}, v(0) - v_0)_{L^2(\Gamma)} = \|g^{1/2}\{v(0) - v_0\}\|_{L^2(\Gamma)}^2 = 0,$$

which combined with (5.2.30) shows $v|_{t=0} = v_0$ on Γ . Therefore, v is a unique weak solution to (5.1.13)–(5.1.14) on [0, T) (here the uniqueness follows from Lemma 5.10.20).

Let us prove the convergence of the full sequence

$$\lim_{\varepsilon \to 0} M_{\tau} u^{\varepsilon} = v \quad \text{weakly in} \quad L^{2}(0, T; H^{1}(\Gamma, T\Gamma)),$$

$$\lim_{\varepsilon \to 0} \partial_{t} M_{\tau} u^{\varepsilon} = \partial_{t} v \quad \text{weakly in} \quad L^{2}(0, T; H^{-1}(\Gamma, T\Gamma)).$$
(5.10.76)

By the boundedness of $\{M_{\tau}u^{\varepsilon}\}_{\varepsilon}$ and $\{\partial_{t}M_{\tau}u^{\varepsilon}\}_{\varepsilon}$ (see (5.10.49) and (5.10.57)) and the uniqueness of a weak solution to (5.1.13)–(5.1.14) (see Lemma 5.10.20) we can show as above that for any sequence $\{\varepsilon_{l}\}_{l=1}^{\infty}$ of positive numbers convergent to zero there exists its subsequence $\{\varepsilon_{k}\}_{k=1}^{\infty}$ such that $\{M_{\tau}u^{\varepsilon_{k}}\}_{k=1}^{\infty}$ converges to v in the sense of (5.10.68) and (5.10.69). This proves (5.10.76).

Since the strong solution u^{ε} to (5.1.1)–(5.1.3) exists globally in time for $\varepsilon \in (0, \varepsilon_3)$, by the above arguments we get a unique weak solution

$$v_T \in C([0,T]; L^2_{q\sigma}(\Gamma,T\Gamma)) \cap L^2(0,T;V_q) \cap H^1(0,T;H^{-1}(\Gamma,T\Gamma))$$

to (5.1.13)–(5.1.14) on [0, T) satisfying (5.10.76) for all T > 0. Moreover, if T < T' then $v_T = v_{T'}$ on [0, T] by the uniqueness of a weak solution. Hence we can define

$$v \in C([0,\infty); L^2_{g\sigma}(\Gamma,T\Gamma)) \cap L^2_{loc}([0,\infty); V_g) \cap H^1_{loc}([0,\infty); H^{-1}(\Gamma,T\Gamma))$$

by $v := v_T$ on [0, T] for each T > 0, which is a unique weak solution to (5.1.13)–(5.1.14) on $[0, \infty)$ and satisfies (5.10.76) for all T > 0.

As a consequence of Theorem 5.1.3, we obtain the existence of a weak solution to (5.1.13)-(5.1.14) for the initial velocity v_0 and the external force f given by the weak and weak- \star limit of $M_{\tau}u_0^{\varepsilon}$ and $M_{\tau}\mathbb{P}_{\varepsilon}f^{\varepsilon}$, respectively. For general $v_0 \in L^2_{g\sigma}(\Gamma, T\Gamma)$ and $f \in L^2_{loc}([0,\infty); H^{-1}(\Gamma, T\Gamma))$ we can construct a global weak solution to (5.1.13)-(5.1.14) by the Galerkin method as in the case of the Navier–Stokes equations in a two-dimensional bounded domain (see e.g. [8, 10, 64]). Here we just give the outline of construction of a weak solution by the Galerkin method.

Countable basis of V_g . If Assumption 2 and the condition (c) in Theorem 5.1.3 are satisfied, we have $\mathcal{K}_g(\Gamma) = \{0\}$ or $\max_{i=0,1} \gamma^i > 0$. By this fact and the Korn inequalities (5.4.37) and (5.4.44) the bilinear form a_g given by (5.10.9) is coercive (and bounded) on V_g and thus induces a linear homeomorphism A_g from V_g onto V'_g by the Lax–Milgram theorem. We consider A_g as an unbounded operator on $L^2_{q\sigma}(\Gamma, T\Gamma)$ equipped with inner product

$$(v_1, v_2)_{L^2_g(\Gamma)} := (g^{1/2}v_1, g^{1/2}v_2)_{L^2(\Gamma)}, \quad v_1, v_2 \in L^2_{g\sigma}(\Gamma, T\Gamma),$$

which is equivalent to the canonical $L^2(\Gamma)$ -inner product by (5.2.30). Then we can show as in the case of the Stokes operator on a bounded domain (see [8, Theorem IV.5.5]) that there exists a sequence $\{w_k\}_{k=1}^{\infty}$ of eigenvectors of A_g that is an orthonormal basis of $L^2_{g\sigma}(\Gamma, T\Gamma)$ equipped with inner product $(\cdot, \cdot)_{L^2_g(\Gamma)}$ as well as an orthogonal basis of V_g equipped with inner product $a_g(\cdot, \cdot)$. In particular,

$$(gw_i, w_j)_{L^2(\Gamma)} = (w_i, w_j)_{L^2_g(\Gamma)} = \begin{cases} 1 & \text{if } i = j, \\ 0 & \text{if } i \neq j. \end{cases}$$
(5.10.77)

Note that, even if $\mathcal{K}_g(\Gamma) \neq \{0\}$ and $\gamma^0 = \gamma^1 = 0$, we can take such a sequence by replacing a_g with $\tilde{a}_g(v_1, v_2) := a_g(v_1, v_2) + (v_1, v_2)_{L^2(\Gamma)}$, which is coercive and bounded on V_g (see Lemma 5.10.2).

Approximate problem. For $k \in \mathbb{N}$ we seek for an approximate solution

$$v_k(y,t) = \sum_{i=1}^k \xi_i(t) w_i(y), \quad y \in \Gamma, \ t \in [0,T] \quad (T > 0)$$

that satisfies

$$(g\partial_t v_k(t), \eta_k)_{L^2(\Gamma)} + a_g(v_k(t), \eta_k) + b_g(v_k(t), v_k(t), \eta_k) = [gf(t), \eta_k]_{T\Gamma}$$
(5.10.78)

for all $\eta_k \in V_g^k$ and $t \in (0, T)$ (with initial condition), where V_g^k is the linear span of $\{w_i\}_{i=1}^k$. (In fact, we need to approximate f(t) by a continuous function.) This problem is equivalent to a system of ordinary differential equations of the form

$$\sum_{i=1}^{k} (gw_i, w_j)_{L^2(\Gamma)} \frac{d\xi_i}{dt}(t) = \mathcal{P}_j(\xi(t)) + [gf(t), w_j]_{T\Gamma}, \quad j = 1, \dots, k$$

with polynomials $\mathcal{P}_1, \ldots, \mathcal{P}_k$ of $\xi = (\xi_1, \ldots, \xi_k)$. Applying (5.10.77) to the left-hand side we see that this system reduces to $d\xi_j/dt = \mathcal{P}_j(\xi) + [gf, w_j]_{T\Gamma}$, which we can solve locally by the Cauchy–Lipschitz theorem. Using (5.10.77) we can also derive the energy estimate for v_k and show its global existence.

Estimate for the time derivative of the approximate solution. As in Lemma 5.10.14 (see also Remark 5.10.16), we estimate the time derivative of v_k in $H^{-1}(\Gamma, T\Gamma)$. To this end, we take $w \in H^1(\Gamma, T\Gamma)$ and apply Lemma 5.10.13 to get $w = g\eta + g\nabla_{\Gamma}q$ with $\eta \in V_g$ satisfying (5.10.50) and $q \in H^2(\Gamma)$. Since $\{w_k\}_{k=1}^{\infty}$ is an orthogonal basis of V_g equipped with inner product $a_q(\cdot, \cdot)$, which is equivalent to the canonical $H^1(\Gamma)$ -inner product,

$$\eta = \sum_{i=1}^{\infty} a_g(\eta, \tilde{w}_i) \tilde{w}_i \quad \text{in} \quad V_g, \quad \tilde{w}_i := \frac{w_i}{a_g(w_i, w_i)^{1/2}}$$

Then we set $\eta_k := \sum_{i=1}^k a_g(\eta, \tilde{w}_k) \tilde{w}_k \in V_g^k$ to get

$$\|\eta_k\|_{H^1(\Gamma)} \le c \|\eta\|_{H^1(\Gamma)} \le c \|w\|_{H^1(\Gamma)}$$

by (5.10.50) and

$$(g\partial_t v_k, \eta_k)_{L^2(\Gamma)} = (g\partial_t v_k, \eta)_{L^2(\Gamma)} = (\partial_t v_k, g\eta)_{L^2(\Gamma)}$$
$$= (\partial_t v_k, w - g\nabla_{\Gamma} q)_{L^2(\Gamma)} = (\partial_t v_k, w)_{L^2(\Gamma)}$$

where we used (5.10.77) in the first equality and $\partial_t v_k \in L^2_{g\sigma}(\Gamma, T\Gamma)$ and $g\nabla_{\Gamma} q \in L^2_{g\sigma}(\Gamma, T\Gamma)^{\perp}$ in the last equality (note that here we take the canonical $L^2(\Gamma)$ -inner product). Hence, substituting η_k for (5.10.78) and using these relations, we can show as in the proof of Lemma 5.10.14 that

$$\|\partial_t v_k(t)\|_{H^{-1}(\Gamma,T\Gamma)} \le c\left\{ \left(1 + \|v_k(t)\|_{L^2(\Gamma)}\right) \|v_k(t)\|_{H^1(\Gamma)} + \|f(t)\|_{H^{-1}(\Gamma,T\Gamma)} \right\}$$

for all $t \in [0, T]$. By this inequality and the energy estimate for v_k we obtain the boundedness of $\{\partial_t v_k\}_{k=1}^{\infty}$ in $L^2(0, T; H^{-1}(\Gamma, T\Gamma))$ and we can prove the convergence of $\{v_k\}_{k=1}^{\infty}$ to a weak solution to (5.1.13)–(5.1.14) as in the proof of Theorem 5.1.3.

5.10.6 Strong convergence of the average and error estimates

In this subsection we show the strong convergence of the averaged tangential component of a strong solution to the Navier–Stokes equations (5.1.1)-(5.1.3) towards a weak solution to the limit equations (5.1.13)-(5.1.14). We also estimate the difference between a strong solution to (5.1.1)-(5.1.3) and a weak solution to (5.1.13)-(5.1.14).

Theorem 5.10.23. Suppose that the assumptions in Theorem 5.8.4 are satisfied. For $\varepsilon \in (0, \varepsilon_1)$ let u^{ε} be the global strong solution to (5.1.1)-(5.1.3) given by Theorem 5.8.4. Also, let $v_0 \in L^2_{g\sigma}(\Gamma, T\Gamma)$, $f \in L^{\infty}(0, \infty; H^{-1}(\Gamma, T\Gamma))$, and v be a weak solution to (5.1.13)-(5.1.14) on $[0, \infty)$. Then for all T > 0 we have

$$\max_{t\in[0,T]} \|M_{\tau}u^{\varepsilon}(t) - v(t)\|_{L^{2}(\Gamma)}^{2} + \int_{0}^{T} \|\nabla_{\Gamma}M_{\tau}u^{\varepsilon}(t) - \nabla_{\Gamma}v(t)\|_{L^{2}(\Gamma)}^{2} dt$$
$$\leq c_{T} \left\{ \delta(\varepsilon)^{2} + \|M_{\tau}u_{0}^{\varepsilon} - v_{0}\|_{L^{2}(\Gamma)}^{2} + \|M_{\tau}\mathbb{P}_{\varepsilon}f^{\varepsilon} - f\|_{L^{\infty}(0,\infty;H^{-1}(\Gamma,T\Gamma))}^{2} \right\}, \quad (5.10.79)$$

where $c_T > 0$ is a constant depending only on T and

$$\delta(\varepsilon) := \varepsilon^{\alpha/4} + \sum_{i=0,1} |\varepsilon^{-1} \gamma^i_{\varepsilon} - \gamma^i|.$$
(5.10.80)

As in Section 5.10.3, we first compare the auxiliary vector field $v^{\varepsilon} = \mathbb{P}_g M_{\tau} u^{\varepsilon}$ with a weak solution to (5.1.13)–(5.1.14) and then derive (5.10.79) by using the estimate for the difference between v^{ε} and $M_{\tau} u^{\varepsilon}$.

Lemma 5.10.24. Under the same assumptions as in Theorem 5.10.23, we have

$$\max_{t\in[0,T]} \|v^{\varepsilon}(t) - v(t)\|_{L^{2}(\Gamma)}^{2} + \int_{0}^{T} \|\nabla_{\Gamma}v^{\varepsilon}(t) - \nabla_{\Gamma}v(t)\|_{L^{2}(\Gamma)}^{2} dt$$
$$\leq c_{T} \left\{ \delta(\varepsilon)^{2} + \|v^{\varepsilon}(0) - v_{0}\|_{L^{2}(\Gamma)}^{2} + \|M_{\tau}\mathbb{P}_{\varepsilon}f^{\varepsilon} - f\|_{L^{\infty}(0,\infty;H^{-1}(\Gamma,T\Gamma))}^{2} \right\} (5.10.81)$$

for all T > 0, where $v^{\varepsilon} = \mathbb{P}_{g}M_{\tau}u^{\varepsilon}$ is given in Lemma 5.10.10, $c_{T} > 0$ is a constant depending only on T, and $\delta(\varepsilon)$ is given by (5.10.80).

Proof. For the sake of simplicity, we set

$$w^{\varepsilon} := v^{\varepsilon} - v, \quad w_0^{\varepsilon} := v^{\varepsilon}(0) - v_0 = \mathbb{P}_g M_{\tau} u_0^{\varepsilon} - v_0, \quad F^{\varepsilon} := M_{\tau} \mathbb{P}_{\varepsilon} f^{\varepsilon} - f.$$

Let T > 0. We subtract both sides of (5.10.58) from those of (5.10.37) to get

$$\begin{split} \int_0^T \{ [g\partial_s w^{\varepsilon}, \eta]_{T\Gamma} + a_g(w^{\varepsilon}, \eta) + b_g(w^{\varepsilon}, v^{\varepsilon}, \eta) + b_g(v, w^{\varepsilon}, \eta) \} \, ds \\ &= \int_0^T [gF^{\varepsilon}, \eta]_{T\Gamma} \, ds + R^1_{\varepsilon}(\eta) + R^2_{\varepsilon}(\eta) \end{split}$$

for all $\eta \in L^2(0,T;V_g)$ (see also Lemma 5.10.19), where $R^1_{\varepsilon}(\eta)$ and $R^2_{\varepsilon}(\eta)$ are given in Lemmas 5.10.6 and 5.10.10. For each $t \in [0,T]$, let $1_{[0,t]} \colon \mathbb{R} \to \mathbb{R}$ be the characteristic function of [0,t]. We substitute $\eta = 1_{[0,t]}w^{\varepsilon}$ for the above equality and calculate as in the proofs of Lemmas 5.10.11 and 5.10.20 by using (5.10.11)–(5.10.13), (5.10.28), (5.10.38), (5.10.64), and Young's inequality. Then we get

$$\begin{split} \|w^{\varepsilon}(t)\|_{L^{2}(\Gamma)}^{2} + \int_{0}^{t} \|\nabla_{\Gamma}w^{\varepsilon}\|_{L^{2}(\Gamma)}^{2} ds \\ &\leq c \left\{ \|w_{0}^{\varepsilon}\|_{L^{2}(\Gamma)}^{2} + \int_{0}^{t} \left(1 + \|v^{\varepsilon}\|_{H^{1}(\Gamma)}^{2}\right) \|w^{\varepsilon}\|_{L^{2}(\Gamma)}^{2} ds \\ &\quad + \int_{0}^{t} \|F^{\varepsilon}\|_{H^{-1}(\Gamma,T\Gamma)}^{2} ds + \delta(\varepsilon)^{2}(1+t) \right\} \quad (5.10.82) \end{split}$$

for all $t \in [0, T]$. Here $\delta(\varepsilon)$ given by (5.10.80) comes from (5.10.28) and (5.10.38) (note that $\varepsilon^{\alpha/2} \leq \varepsilon^{\alpha/4}$). This inequality implies that

$$\xi(t) \le c \left\{ \xi(0) + \int_0^t \left(\varphi(s)\xi(s) + \|F^{\varepsilon}(s)\|_{H^{-1}(\Gamma,T\Gamma)}^2 \right) ds \right\} \quad \text{for all} \quad t \in [0,T],$$

where $\xi(t) := \delta(\varepsilon)^2 + \|w^{\varepsilon}(t)\|_{L^2(\Gamma)}^2$ and $\varphi(t) := 1 + \|v^{\varepsilon}(t)\|_{H^1(\Gamma)}^2$. Hence by Gronwall's inequality we have

$$\xi(t) \le c \left(\xi(0) + \int_0^t \|F^{\varepsilon}(s)\|_{H^{-1}(\Gamma, T\Gamma)}^2 \, ds \right) \exp\left(\int_0^t \varphi(s) \, ds \right), \quad t \in [0, T].$$

From this inequality and the estimate (5.10.42) for $||v^{\varepsilon}||^2_{H^1(\Gamma)}$ we deduce that

$$\|w^{\varepsilon}(t)\|_{L^{2}(\Gamma)}^{2} \leq c_{T} \left\{ \delta(\varepsilon)^{2} + \|w_{0}^{\varepsilon}\|_{L^{2}(\Gamma)}^{2} + \|F^{\varepsilon}\|_{L^{\infty}(0,\infty;H^{-1}(\Gamma,T\Gamma))}^{2} \right\}$$

for all $t \in [0, T]$, where $c_T > 0$ is a constant depending only on T. Applying this inequality and (5.10.42) to (5.10.82) we also get the same estimate for the time integral of $\|\nabla_{\Gamma} w^{\varepsilon}\|_{L^2(\Gamma)}^2$. Therefore, the inequality (5.10.81) is valid.
Proof of Theorem 5.10.23. Let $v^{\varepsilon} = \mathbb{P}_g M_{\tau} u^{\varepsilon}$. Since $v_0 \in L^2_{g\sigma}(\Gamma, T\Gamma)$ and \mathbb{P}_g is the orthogonal projection from $L^2(\Gamma, T\Gamma)$ onto $L^2_{g\sigma}(\Gamma, T\Gamma)$,

$$\|v^{\varepsilon}(0) - v_0\|_{L^2(\Gamma)} = \|\mathbb{P}_g(M_{\tau}u_0^{\varepsilon} - v_0)\|_{L^2(\Gamma)} \le \|M_{\tau}u_0^{\varepsilon} - v_0\|_{L^2(\Gamma)}.$$

By this inequality, (5.10.36), and (5.10.81) we obtain (5.10.79) (note that $\varepsilon^2 \leq \delta(\varepsilon)^2$). \Box

Now we see that Theorem 5.1.4 is an immediate consequence of Theorem 5.10.23.

Proof of Theorem 5.1.4. The condition (b') implies the condition (b) in Theorem 5.1.3. Thus, the statements in Theorem 5.1.3 are valid. Also, for each T > 0 the right-hand side of (5.10.79) converges to zero as $\varepsilon \to 0$ by (5.10.80), $\alpha > 0$, and the conditions (b') and (c). Hence $M_{\tau}u^{\varepsilon}$ converges to v strongly in $C([0,T]; L^2(\Gamma, T\Gamma))$ and $L^2(0,T; H^1(\Gamma, T\Gamma))$ as $\varepsilon \to 0$.

Next let us estimate the difference between a strong solution to (5.1.1)–(5.1.3) and a weak solution to (5.1.13)–(5.1.14) in Ω_{ε} . Recall that we denote by $\bar{\eta} = \eta \circ \pi$ the constant extension of a function η on Γ in the normal direction of Γ .

Theorem 5.10.25. Under the same assumptions as in Theorem 5.10.23, we have

$$\max_{t\in[0,T]} \|u^{\varepsilon}(t) - \bar{v}(t)\|_{L^{2}(\Omega_{\varepsilon})}^{2} + \int_{0}^{T} \|\overline{P}\nabla u^{\varepsilon}(t) - \overline{\nabla_{\Gamma}v}(t)\|_{L^{2}(\Omega_{\varepsilon})}^{2} dt$$
$$\leq c_{T}\varepsilon \left\{ \delta(\varepsilon)^{2} + \|M_{\tau}u_{0}^{\varepsilon} - v_{0}\|_{L^{2}(\Gamma)}^{2} + \|M_{\tau}\mathbb{P}_{\varepsilon}f^{\varepsilon} - f\|_{L^{\infty}(0,\infty;H^{-1}(\Gamma,T\Gamma))}^{2} \right\} (5.10.83)$$

for all T > 0, where $c_T > 0$ is a constant depending only on T. In particular,

$$\lim_{\varepsilon \to 0} \varepsilon^{-1} \| u^{\varepsilon} - \bar{v} \|_{C([0,T];L^2(\Omega_{\varepsilon}))}^2 = \lim_{\varepsilon \to 0} \varepsilon^{-1} \| \overline{P} \nabla u^{\varepsilon} - \overline{\nabla_{\Gamma} v} \|_{L^2(0,T;L^2(\Omega_{\varepsilon}))}^2 = 0$$

for all T > 0 provided that $\lim_{\varepsilon \to 0} \varepsilon^{-1} \gamma^i_{\varepsilon} = \gamma^i$ for i = 0, 1 and

$$\lim_{\varepsilon \to 0} \|M_{\tau} u_0^{\varepsilon} - v_0\|_{L^2(\Gamma)}^2 = \lim_{\varepsilon \to 0} \|M_{\tau} \mathbb{P}_{\varepsilon} f^{\varepsilon} - f\|_{L^{\infty}(0,\infty;H^{-1}(\Gamma,T\Gamma))}^2 = 0.$$

Proof. For the sake of simplicity, we denote by I_{ε} the right-hand side of (5.10.79). Also, we fix T > 0 and suppress $t \in [0, T]$. Let us estimate $u^{\varepsilon} - \overline{v}$. Since

$$u^{\varepsilon} - \bar{v} = \left(u^{\varepsilon} - \overline{Mu^{\varepsilon}}\right) + \left(\overline{Mu^{\varepsilon}} \cdot \bar{n}\right)\bar{n} + \left(\overline{M_{\tau}u^{\varepsilon}} - \bar{v}\right) \quad \text{in} \quad \Omega_{\varepsilon},$$

we apply (5.2.53), (5.6.6), and (5.6.10) to the right-hand side to get

$$\|u^{\varepsilon} - \bar{v}\|_{L^{2}(\Omega_{\varepsilon})}^{2} \leq c\varepsilon \left(\varepsilon \|u^{\varepsilon}\|_{H^{1}(\Omega_{\varepsilon})}^{2} + \|M_{\tau}u^{\varepsilon} - v\|_{L^{2}(\Gamma)}^{2}\right).$$

Hence we see by (5.8.34), (5.10.79), and (5.10.80) that

$$\|u^{\varepsilon}(t) - \bar{v}(t)\|_{L^{2}(\Omega_{\varepsilon})}^{2} \le c\varepsilon(\varepsilon^{\alpha} + I_{\varepsilon}) \le c\varepsilon I_{\varepsilon} \quad \text{for all} \quad t \in [0, T].$$
(5.10.84)

Next we consider the second term on the left-hand side of (5.10.83). Let

$$J_1 := \left\| \overline{P} \nabla u^{\varepsilon} - \overline{\nabla_{\Gamma} M u^{\varepsilon}} \right\|_{L^2(\Omega_{\varepsilon})}, \quad J_2 := \left\| \overline{\nabla_{\Gamma} [(M u^{\varepsilon} \cdot n)n]} \right\|_{L^2(\Omega_{\varepsilon})},$$
$$J_3 := \left\| \overline{\nabla_{\Gamma} M_{\tau} u^{\varepsilon}} - \overline{\nabla_{\Gamma} v} \right\|_{L^2(\Omega_{\varepsilon})}.$$

By (5.2.53) and (5.6.25) we have

$$J_1 \le c\varepsilon \|u^\varepsilon\|_{H^2(\Omega_\varepsilon)}, \quad J_3 \le c\varepsilon^{1/2} \|\nabla_\Gamma M_\tau u^\varepsilon - \nabla_\Gamma v\|_{L^2(\Gamma)}.$$

Also, the inequalities (5.2.53) and (5.6.29) imply that

$$J_2 \le c\varepsilon^{1/2} \|Mu^{\varepsilon} \cdot n\|_{H^1(\Gamma)} \le c\varepsilon \|u^{\varepsilon}\|_{H^2(\Omega_{\varepsilon})}.$$

From these inequalities we deduce that

$$\left\|\overline{P}\nabla u^{\varepsilon} - \overline{\nabla_{\Gamma}v}\right\|_{L^{2}(\Omega_{\varepsilon})} \leq J_{1} + J_{2} + J_{3} \leq c\varepsilon^{1/2} \left(\varepsilon^{1/2} \|u^{\varepsilon}\|_{H^{2}(\Omega_{\varepsilon})} + \|\nabla_{\Gamma}M_{\tau}u^{\varepsilon} - \nabla_{\Gamma}v\|_{L^{2}(\Gamma)}\right)$$

Thus, by (5.8.34) and (5.10.79) we have

$$\int_{0}^{T} \left\| \overline{P} \nabla u^{\varepsilon} - \overline{\nabla_{\Gamma} v} \right\|_{L^{2}(\Omega_{\varepsilon})}^{2} dt \leq c \varepsilon \int_{0}^{T} \left(\varepsilon \| u^{\varepsilon} \|_{H^{2}(\Omega_{\varepsilon})}^{2} + \| \nabla_{\Gamma} M_{\tau} u^{\varepsilon} - \nabla_{\Gamma} v \|_{L^{2}(\Gamma)}^{2} \right) dt$$
$$\leq c \varepsilon \{ \varepsilon^{\alpha} (1+T) + I_{\varepsilon} \} \leq c \varepsilon (1+T) I_{\varepsilon}.$$

Combining this inequality and (5.10.84) we obtain (5.10.83).

We can also compare the normal derivative (with respect to Γ) of a strong solution to (5.1.1)-(5.1.3) with a weak solution to (5.1.13)-(5.1.14).

Theorem 5.10.26. Under the same assumptions as in Theorem 5.10.23, we have

$$\int_{0}^{T} \left(\left\| \overline{P} \partial_{n} u^{\varepsilon}(t) + \overline{Wv}(t) \right\|_{L^{2}(\Omega_{\varepsilon})}^{2} + \left\| \partial_{n} u^{\varepsilon}(t) \cdot \overline{n} - \frac{1}{\overline{g}} \overline{v}(t) \cdot \overline{\nabla_{\Gamma} g} \right\|_{L^{2}(\Omega_{\varepsilon})}^{2} \right) dt \\
\leq c_{T} \varepsilon \left\{ \delta(\varepsilon)^{2} + \left\| M_{\tau} u_{0}^{\varepsilon} - v_{0} \right\|_{L^{2}(\Gamma)}^{2} + \left\| M_{\tau} \mathbb{P}_{\varepsilon} f^{\varepsilon} - f \right\|_{L^{\infty}(0,\infty;H^{-1}(\Gamma,T\Gamma))}^{2} \right\} \quad (5.10.85)$$

for all T > 0, where $\partial_n u = (\bar{n} \cdot \nabla)u$ is the normal derivative of u given by (5.3.5) and $c_T > 0$ is a constant depending only on T. Hence, setting

$$V := -Wv + \frac{1}{g}(v \cdot \nabla_{\Gamma}g)n \quad on \quad \Gamma \times (0, \infty)$$

we have (note that Wv is tangential on Γ)

$$\lim_{\varepsilon \to 0} \varepsilon^{-1} \left\| \partial_n u^{\varepsilon} - \overline{V} \right\|_{L^2(0,T;L^2(\Omega_{\varepsilon}))}^2 = 0$$

for all T > 0 provided that $\lim_{\varepsilon \to 0} \varepsilon^{-1} \gamma^i_{\varepsilon} = \gamma^i$ for i = 0, 1 and

$$\lim_{\varepsilon \to 0} \|M_{\tau} u_0^{\varepsilon} - v_0\|_{L^2(\Gamma)}^2 = \lim_{\varepsilon \to 0} \|M_{\tau} \mathbb{P}_{\varepsilon} f^{\varepsilon} - f\|_{L^{\infty}(0,\infty; H^{-1}(\Gamma, T\Gamma))}^2 = 0.$$

Proof. We fix T > 0 and suppress $t \in [0, T]$. Let

$$J_1 := \left\| \overline{P} \partial_n u^{\varepsilon} + \overline{W} u^{\varepsilon} \right\|_{L^2(\Omega_{\varepsilon})}, \quad J_2 := \left\| \overline{WMu^{\varepsilon}} - \overline{W} u^{\varepsilon} \right\|_{L^2(\Omega_{\varepsilon})},$$
$$J_3 := \left\| \overline{Wv} - \overline{WMu^{\varepsilon}} \right\|_{L^2(\Omega_{\varepsilon})}.$$

We apply (5.3.34) to J_1 and (5.6.6) to J_2 to get

$$J_1 \le c\varepsilon \|u^\varepsilon\|_{H^2(\Omega_\varepsilon)}, \quad J_2 \le c\varepsilon \|u^\varepsilon\|_{H^1(\Omega_\varepsilon)}.$$

Also, noting that $WMu^{\varepsilon} = WM_{\tau}u^{\varepsilon}$ on Γ by (5.2.6), we use (5.2.53) to get

$$J_3 \le c\varepsilon^{1/2} \|Wv - WM_\tau u^\varepsilon\|_{L^2(\Gamma)} \le c\varepsilon^{1/2} \|v - M_\tau u^\varepsilon\|_{L^2(\Gamma)}$$

From these inequalities we deduce that

$$\left\|\overline{P}\partial_{n}u^{\varepsilon}+\overline{Wv}\right\|_{L^{2}(\Omega_{\varepsilon})}\leq J_{1}+J_{2}+J_{3}\leq c\varepsilon^{1/2}\left(\varepsilon^{1/2}\|u^{\varepsilon}\|_{H^{2}(\Omega_{\varepsilon})}+\|M_{\tau}u^{\varepsilon}-v\|_{L^{2}(\Gamma)}\right).$$

We also observe by (5.2.30), (5.2.53), and (5.6.48) that

$$\begin{split} \left\| \partial_n u^{\varepsilon} \cdot \bar{n} - \frac{1}{\bar{g}} \bar{v} \cdot \overline{\nabla_{\Gamma} g} \right\|_{L^2(\Omega_{\varepsilon})} \\ & \leq \left\| \partial_n u^{\varepsilon} \cdot \bar{n} - \frac{1}{\bar{g}} \overline{M_{\tau} u^{\varepsilon}} \cdot \overline{\nabla_{\Gamma} g} \right\|_{L^2(\Omega_{\varepsilon})} + \left\| \frac{1}{\bar{g}} \left(\overline{M_{\tau} u^{\varepsilon}} - \bar{v} \right) \cdot \overline{\nabla_{\Gamma} g} \right\|_{L^2(\Omega_{\varepsilon})} \\ & \leq c \varepsilon^{1/2} \left(\varepsilon^{1/2} \| u^{\varepsilon} \|_{H^2(\Omega_{\varepsilon})} + \| M_{\tau} u^{\varepsilon} - v \|_{L^2(\Gamma)} \right). \end{split}$$

Hence, as in the proof of Theorem 5.10.25, we integrate the square of the above inequalities over (0, T) and then use (5.8.34) and (5.10.79) to obtain (5.10.85).

Remark 5.10.27. In the estimate (5.10.85) the Weingarten map W represents the curvatures of the limit surface Γ . On the other hand, the functions g_0 and g_1 with $g = g_1 - g_0$ are used to define the inner and outer boundaries of the curved thin domain Ω_{ε} . Therefore, roughly speaking, the tangential component (with respect to Γ) of the normal derivative $\partial_n u^{\varepsilon}$ of the bulk velocity depends only on the shape of Γ , while the geometry of the boundaries of Ω_{ε} affects only the normal component of $\partial_n u^{\varepsilon}$.

5.A Notations and basic formulas on vectors and matrices

We give notations and basic formulas on vectors and matrices, and use them to prove Lemmas 5.4.4 and 5.7.2.

For a matrix $A \in \mathbb{R}^{3\times 3}$ we denote by A^T and $A_S := (A + A^T)/2$ the transpose and the symmetric part of A, respectively. We define the tensor product of vectors $a \in \mathbb{R}^l$ and $b \in \mathbb{R}^m$ with $l, m \in \mathbb{N}$ as

$$a \otimes b := \begin{pmatrix} a_1 b_1 & \cdots & a_1 b_m \\ \vdots & & \vdots \\ a_l b_1 & \cdots & a_l b_m \end{pmatrix}, \quad a = (a_1, \dots, a_l), \ b = (b_1, \dots, b_m).$$

Also, for a vector field $u = (u_1, u_2, u_3)$ on an open subset of \mathbb{R}^3 we write

$$\nabla u := \begin{pmatrix} \partial_1 u_1 & \partial_1 u_2 & \partial_1 u_3 \\ \partial_2 u_1 & \partial_2 u_2 & \partial_2 u_3 \\ \partial_3 u_1 & \partial_3 u_2 & \partial_3 u_3 \end{pmatrix}, \quad |\nabla^2 u|^2 := \sum_{i,j,k=1}^3 |\partial_i \partial_j u_k|^2 \quad \left(\partial_i := \frac{\partial}{\partial x_i}\right).$$

We define the inner product of matrices $A, B \in \mathbb{R}^{3 \times 3}$ and the norm of A as

$$A: B := \operatorname{tr}[A^T B] = \sum_{i=1}^{3} Ae_i \cdot Be_i, \quad |A| := \sqrt{A: A},$$

where $\{e_1, e_2, e_3\}$ is an orthonormal basis of \mathbb{R}^3 . Note that A : B does not depend on a choice of an orthonormal basis of \mathbb{R}^3 . In particular, taking the standard basis of \mathbb{R}^3 we see that

$$A: B = \sum_{i,j=1}^{3} A_{ij} B_{ij} = B: A = A^{T}: B^{T}, \quad AB: C = A: CB^{T} = B: A^{T}C$$

for $A, B, C \in \mathbb{R}^{3 \times 3}$. Also, for $a, b \in \mathbb{R}^3$ we have $|a \otimes b| = |a||b|$.

Lemma 5.A.1. Let $n_0 \in \mathbb{R}^3$ and $A_1 \in \mathbb{R}^{3 \times 3}$ satisfy $|n_0| = 1$ and $A_1 b \cdot n_0 = 0$ for all $b \in \mathbb{R}^3$. Then for a matrix $B := A_1 + n_0 \otimes a$ with $a \in \mathbb{R}^3$ we have

$$|B|^2 = |A_1|^2 + |a|^2.$$
(5.A.1)

Also, let $\tau_1, \tau_2 \in \mathbb{R}^3$ and $A_2 \in \mathbb{R}^{3 \times 3}$ satisfy

$$n_0 \cdot \tau_1 = n_0 \cdot \tau_2 = 0, \quad A_2 n_0 = 0, \quad A_2 b \cdot n_0 = 0 \quad for \ all \quad b \in \mathbb{R}^3.$$

Then for a matrix $C := A_2 + \tau_1 \otimes n_0 + n_0 \otimes \tau_2 + cn_0 \otimes n_0$ with $c \in \mathbb{R}$ we have

$$|C|^{2} = |A_{2}|^{2} + |\tau_{1}|^{2} + |\tau_{2}|^{2} + |c|^{2}.$$
(5.A.2)

Proof. We take an orthonormal basis $\{e_1, e_2, e_3\}$ of \mathbb{R}^3 with $e_3 = n_0$. Then since $(n_0 \otimes a)b = (a \cdot b)n_0$, $A_1b \cdot n_0 = 0$ for any $b \in \mathbb{R}^3$ and $|n_0| = 1$,

$$|Be_i|^2 = |A_1e_i + (a \cdot e_i)n_0|^2 = |A_1e_i|^2 + (a \cdot e_i)^2, \quad i = 1, 2,$$

$$|Bn_0|^2 = |A_1n_0 + (a \cdot n_0)n_0|^2 = |A_1n_0|^2 + (a \cdot n_0)^2.$$

Applying these equalities to $|B|^2 = |Be_1|^2 + |Be_2|^2 + |Bn_0|^2$ we get

$$|B|^{2} = (|A_{1}e_{1}|^{2} + |A_{1}e_{2}|^{2} + |A_{1}n_{0}|^{2}) + \{(a \cdot e_{1})^{2} + (a \cdot e_{2})^{2} + (a \cdot n_{0})^{2}\} = |A_{1}|^{2} + |a|^{2}.$$

Thus (5.A.1) is valid. Next we prove (5.A.2). Since $e_i \cdot n_0 = 0$ and $A_2 e_i \cdot n_0 = 0$,

$$|Ce_i|^2 = |A_2e_i + (\tau_2 \cdot e_i)n_0|^2 = |A_2e_i|^2 + (\tau_2 \cdot e_i)^2, \quad i = 1, 2.$$

Also, by $A_2n_0 = 0$, $|n_0| = 1$, and $\tau_1 \cdot n_0 = 0$,

$$|Cn_0|^2 = |\tau_1 + cn_0|^2 = |\tau_1|^2 + |c|^2.$$

Hence $|C|^2 = |Ce_1|^2 + |Ce_2|^2 + |Cn_0|^2$ is of the form

$$|C|^{2} = \sum_{i=1,2} \left(|A_{2}e_{i}|^{2} + (\tau_{2} \cdot e_{i})^{2} \right) + |\tau_{1}|^{2} + |c|^{2}.$$

Here the first two terms on the right-hand side are equal to $|A_2|^2$ and $|\tau_2|^2$ since $A_2n_0 = 0$ and $\tau_2 \cdot n_0 = 0$. Hence (5.A.2) follows.

Based on the formulas (5.A.1) and (5.A.2) we prove Lemma 5.4.4.

Proof of Lemma 5.4.4. Let $u \in H^1(\Omega_{\varepsilon})^3$ and $U := u \circ \Phi_{\varepsilon}$, where $\Phi_{\varepsilon} \colon \Omega_1 \to \Omega_{\varepsilon}$ is given by (5.4.15). The inverse map of Φ_{ε} is of the form

$$\Phi_{\varepsilon}^{-1}(x) := \pi(x) + \varepsilon^{-1} d(x) \bar{n}(x), \quad x \in \Omega_{\varepsilon}.$$

Then by (5.2.6), (5.2.11), (5.2.16), and $\nabla d = \bar{n}$ we have

$$\nabla \Phi_{\varepsilon}^{-1}(x) = \left\{ I_3 - d(x)\overline{W}(x) \right\}^{-1} \left\{ I_3 - \varepsilon^{-1}d(x)\overline{W}(x) \right\} \overline{P}(x) + \varepsilon^{-1}\overline{Q}(x)$$

for $x \in \Omega_{\varepsilon}$. We set $x = \Phi_{\varepsilon}(X)$ with $X \in \Omega_1$ in this equality and use

 $d(\Phi_{\varepsilon}(X)) = \varepsilon d(X), \quad \pi(\Phi_{\varepsilon}(X)) = \pi(X)$

to get $\nabla \Phi_{\varepsilon}^{-1}(\Phi_{\varepsilon}(X)) = \Lambda_{\varepsilon}(X)\overline{P}(X) + \varepsilon^{-1}\overline{Q}(X)$, where

$$\Lambda_{\varepsilon}(X) := \left\{ I_3 - \varepsilon d(X)\overline{W}(X) \right\}^{-1} \left\{ I_3 - d(X)\overline{W}(X) \right\}, \quad X \in \Omega_1.$$

From this formula and

$$\nabla u(\Phi_{\varepsilon}(X)) = \nabla \Phi_{\varepsilon}^{-1}(\Phi_{\varepsilon}(X)) \nabla U(X), \quad \overline{Q}(X) \nabla U(X) = \overline{n}(X) \otimes \partial_{n} U(X)$$

it follows that

$$\nabla u(\Phi_{\varepsilon}(X)) = \Lambda_{\varepsilon}(X)\overline{P}(X)\nabla U(X) + \varepsilon^{-1}\overline{n}(X) \otimes \partial_{n}U(X), \quad X \in \Omega_{1}$$

Hereafter we refer to this equality as

$$(\nabla u) \circ \Phi_{\varepsilon} = \Lambda_{\varepsilon} \overline{P} \nabla U + \varepsilon^{-1} \overline{n} \otimes \partial_n U \quad \text{in} \quad \Omega_1.$$
(5.A.3)

Since $\Lambda_{\varepsilon}\overline{P} = \overline{P}\Lambda_{\varepsilon}$ by (5.2.6) and (5.2.8), we have $(\Lambda_{\varepsilon}\overline{P}\nabla U)b\cdot\overline{n} = 0$ for any $b \in \mathbb{R}^3$. Hence we can use (5.A.1) with $n_0 = \overline{n}$, $A_1 = \Lambda_{\varepsilon}\overline{P}\nabla U$, and $a = \varepsilon^{-1}\partial_n U$ to get

$$|(\nabla u) \circ \Phi_{\varepsilon}|^{2} = |\Lambda_{\varepsilon} \overline{P} \nabla U|^{2} + \varepsilon^{-2} |\partial_{n} U|^{2} \ge c \left(|\overline{P} \nabla U|^{2} + \varepsilon^{-2} |\partial_{n} U|^{2} \right) \quad \text{in} \quad \Omega_{1},$$

where the second inequality follows from (5.2.9). By this inequality and (5.4.17),

$$\varepsilon^{-1} \|\nabla u\|_{L^2(\Omega_{\varepsilon})}^2 \ge c \|(\nabla u) \circ \Phi_{\varepsilon}\|_{L^2(\Omega_1)}^2 \ge c \left(\left\|\overline{P}\nabla U\right\|_{L^2(\Omega_1)}^2 + \varepsilon^{-2} \|\partial_n U\|_{L^2(\Omega_1)}^2 \right).$$

Hence (5.4.18) is valid. To prove (5.4.19) we observe by $I_3 = P + Q$ and (5.3.6) that

$$\overline{P}\nabla U = \overline{P}(\nabla U)\overline{P} + \overline{P}(\nabla U)\overline{Q} = \overline{P}(\nabla U)\overline{P} + \left[\overline{P}(\nabla U)\bar{n}\right] \otimes \bar{n},$$
$$\partial_n U = \partial_n \left[\overline{P}U + (U \cdot \bar{n})\bar{n}\right] = \overline{P}\partial_n U + \{\partial_n (U \cdot \bar{n})\}\bar{n}.$$

We apply these equalities to (5.A.3) and use $\Lambda_{\varepsilon}\overline{P} = \overline{P}\Lambda_{\varepsilon}$ by (5.2.6) and (5.2.8) to get

$$\begin{aligned} (\nabla u) \circ \Phi_{\varepsilon} &= \overline{P} \Lambda_{\varepsilon} (\nabla U) \overline{P} + \left[\overline{P} \Lambda_{\varepsilon} (\nabla U) \overline{n} \right] \otimes \overline{n} \\ &+ \varepsilon^{-1} \overline{n} \otimes \left(\overline{P} \partial_n U \right) + \varepsilon^{-1} \{ \partial_n (U \cdot \overline{n}) \} \overline{n} \otimes \overline{n} \quad \text{in} \quad \Omega_1. \end{aligned}$$

Hence $D(u) \circ \Phi_{\varepsilon} = \{ (\nabla u) \circ \Phi_{\varepsilon} + (\nabla u)^T \circ \Phi_{\varepsilon} \}/2$ is of the form

$$D(u) \circ \Phi_{\varepsilon} = A_2 + \tau \otimes \bar{n} + \bar{n} \otimes \tau + \varepsilon^{-1} \{\partial_n (U \cdot \bar{n})\} \bar{n} \otimes \bar{n} \quad \text{in} \quad \Omega_1.$$

where $F_{\varepsilon}(U) := \Lambda_{\varepsilon} \nabla U$ is the matrix (5.4.20) and

$$A_2 := \overline{P} F_{\varepsilon}(U)_S \overline{P}, \quad \tau := \frac{1}{2} \overline{P} \{ \Lambda_{\varepsilon}(\nabla U) \bar{n} + \varepsilon^{-1} \partial_n U \}.$$

(Note that $F_{\varepsilon}(U)_S$ is the symmetric part of $F_{\varepsilon}(U)$.) By the above definitions it is obvious that $\tau \cdot \bar{n} = 0$, $A_2\bar{n} = 0$, and $A_2b \cdot \bar{n} = 0$ for all $b \in \mathbb{R}^3$. Therefore, we can apply (5.A.2) to $C = D(u) \circ \Phi_{\varepsilon}$ to obtain

$$|D(u) \circ \Phi_{\varepsilon}|^{2} = |A_{2}|^{2} + 2|\tau|^{2} + \varepsilon^{-2}|\partial_{n}(U \cdot \bar{n})|^{2} \ge |A_{2}|^{2} + \varepsilon^{-2}|\partial_{n}(U \cdot \bar{n})|^{2}$$

in Ω_1 . From this inequality and (5.4.17) we deduce that

$$\varepsilon^{-1} \|D(u)\|_{L^{2}(\Omega_{\varepsilon})}^{2} \ge c \|D(u) \circ \Phi_{\varepsilon}\|_{L^{2}(\Omega_{1})}^{2} \ge c \left(\|A_{2}\|_{L^{2}(\Omega_{1})}^{2} + \varepsilon^{-2} \|\partial_{n}(U \cdot \bar{n})\|_{L^{2}(\Omega_{1})}^{2}\right).$$

Hence (5.4.19) is valid (note that $A_2 = \overline{P}F_{\varepsilon}(U)_S\overline{P}$).

Next we give a formula on the curl of a vector field and show Lemma 5.7.2.

Lemma 5.A.2. Let U be an open set in \mathbb{R}^3 and E_1 , E_2 , and E_3 vector fields on U such that $\{E_1(x), E_2(x), E_3(x)\}$ is an orthonormal basis of \mathbb{R}^3 for each $x \in \mathbb{R}^3$ and

$$E_1 \times E_2 = E_3, \quad E_2 \times E_3 = E_1, \quad E_3 \times E_1 = E_2 \quad in \quad U$$

Then for $u \in C^1(U)^3$ we have

$$\operatorname{curl} u = \{ (E_2 \cdot \nabla) u \cdot E_3 - (E_3 \cdot \nabla) u \cdot E_2 \} E_1 + \{ (E_3 \cdot \nabla) u \cdot E_1 - (E_1 \cdot \nabla) u \cdot E_3 \} E_2 + \{ (E_1 \cdot \nabla) u \cdot E_2 - (E_2 \cdot \nabla) u \cdot E_1 \} E_3 \quad in \quad U. \quad (5.A.4)$$

Proof. By the assumption, $\operatorname{curl} u = \sum_{i=1}^{3} (\operatorname{curl} u \cdot E_i) E_i$. Since $E_1 = E_2 \times E_3$,

$$\operatorname{curl} u \cdot E_1 = \operatorname{curl} u \cdot (E_2 \times E_3) = E_2 \cdot (E_3 \times \operatorname{curl} u)$$
$$= E_2 \cdot \{ (\nabla u) E_3 - (\nabla u)^T E_3 \}$$
$$= (\nabla u)^T E_2 \cdot E_3 - (\nabla u)^T E_3 \cdot E_2$$
$$= (E_2 \cdot \nabla) u \cdot E_3 - (E_3 \cdot \nabla) u \cdot E_2.$$

Calculating curl $u \cdot E_i$, i = 2, 3 in the same way we observe that (5.A.4) holds.

Proof of Lemma 5.7.2. Let $u \in C^1(\Omega_{\varepsilon})^3$ and $u^a := E_{\varepsilon}M_{\tau}u$ be given by (5.6.50). Since the surface Γ is compact, we can take finite relative open sets O_k of Γ and pairs of tangential vector fields $\{\tau_1^k, \tau_2^k\}$ on O_k , $k = 1, \ldots, k_0$ such that $\Gamma = \bigcup_{k=1}^{k_0} O_k$, the triplet $\{\tau_1^k, \tau_2^k, n\}$ forms an orthonormal basis of \mathbb{R}^3 on O_k , and

$$\tau_1^k \times \tau_2^k = n, \quad \tau_2^k \times n = \tau_1^k, \quad n \times \tau_1^k = \tau_2^k \quad \text{on} \quad O_k$$

for each $k = 1, ..., k_0$. Then since $\Omega_{\varepsilon} = \bigcup_{k=1}^{k_0} U_k$, where

$$U_k := \{ y + rn(y) \mid y \in O_k, r \in (\varepsilon g_0(y), \varepsilon g_1(y)) \}, \quad k = 1, \dots, k_0,$$

it is sufficient to show (5.7.2) in U_k for each $k = 1, ..., k_0$. From now on, we fix and suppress k. We use (5.A.4) to u^a with $E_1 := \overline{\tau}_1$, $E_2 := \overline{\tau}_2$, and $E_3 := \overline{n}$. Then since $P\tau_i = \tau_i$, i = 1, 2 and Pn = 0,

$$\overline{P}\operatorname{curl} u^a = \{(\overline{\tau}_2 \cdot \nabla) u^a \cdot \overline{n} - (\overline{n} \cdot \nabla) u^a \cdot \overline{\tau}_2\} \overline{\tau}_1 + \{(\overline{n} \cdot \nabla) u^a \cdot \overline{\tau}_1 - (\overline{\tau}_1 \cdot \nabla) u^a \cdot \overline{n}\} \overline{\tau}_2 \quad \text{in} \quad U.$$

From this equality, $(\bar{n} \cdot \nabla)u^a = \partial_n u^a$, and $|\tau_1| = |\tau_2| = |n| = 1$ we deduce that

$$\left|\overline{P}\operatorname{curl} u^{a}\right| \leq c\left(\left|\partial_{n} u^{a}\right| + \left|(\bar{\tau}_{1} \cdot \nabla) u^{a} \cdot \bar{n}\right| + \left|(\bar{\tau}_{2} \cdot \nabla) u^{a} \cdot \bar{n}\right|\right) \quad \text{in} \quad U.$$
(5.A.5)

Let us estimate each term on the right-hand side. By (5.3.6), (5.3.37), and (5.6.50),

$$\left|\partial_{n}u^{a}\right| = \left|\overline{M_{\tau}u} \cdot \partial_{n}\Psi_{\varepsilon}\right| \le c\left|\overline{M_{\tau}u}\right| = c\left|\overline{PMu}\right| \le c\left|\overline{Mu}\right| \quad \text{in} \quad U.$$
(5.A.6)

To estimate the other terms we set

$$u_{\tau}^{a} := \overline{P}u^{a} = \overline{M_{\tau}u}, \quad u_{n}^{a} := (u^{a} \cdot \overline{n})\overline{n} = \left(\overline{M_{\tau}u} \cdot \Psi_{\varepsilon}\right)\overline{n}$$

so that $u^a = u^a_{\tau} + u^a_n$. Let i = 1, 2. Since $u^a_{\tau} \cdot \bar{n} = 0$ in U, we have

$$(\bar{\tau}_i \cdot \nabla) u^a_\tau \cdot \bar{n} = (\bar{\tau}_i \cdot \nabla) (u^a_\tau \cdot \bar{n}) - u^a_\tau \cdot (\bar{\tau}_i \cdot \nabla) \bar{n} = -u^a_\tau \cdot (\bar{\tau}_i \cdot \nabla) \bar{n}.$$

Hence by (5.2.17) and $|\tau_i| = 1$ we get

$$\left| (\bar{\tau}_i \cdot \nabla) u^a_{\tau} \cdot \bar{n} \right| \le c \left| u^a_{\tau} \right| \le c \left| \overline{Mu} \right| \quad \text{in} \quad U, \, i = 1, 2.$$
(5.A.7)

Also, by $\tau_i = P\tau_i$, $P = P^T$, and $|\tau_i| = 1$ we see that

$$\left| (\bar{\tau}_i \cdot \nabla) u_n^a \right| = \left| (\nabla u_n^a)^T \overline{P} \bar{\tau}_i \right| = \left| \left[\overline{P} (\nabla u_n^a) \right]^T \bar{\tau}_i \right| \le \left| \overline{P} (\nabla u_n^a) \right| \quad \text{in} \quad U.$$

Moreover, by $u_n^a = (\overline{M_\tau u} \cdot \Psi_\varepsilon) \overline{n}$ we have

$$\overline{P}(\nabla u_n^a) = \left[\left\{ \overline{P} \nabla \left(\overline{M_\tau u} \right) \right\} \Psi_\varepsilon + \left(\overline{P} \nabla \Psi_\varepsilon \right) \overline{M_\tau u} \right] \otimes \overline{n} + \left(\overline{M_\tau u} \cdot \Psi_\varepsilon \right) \overline{P} \nabla \overline{n}$$

and thus the inequalities (5.2.13), (5.2.17), and (5.3.37) imply that

$$\left|\overline{P}(\nabla u_n^a)\right| \le c\varepsilon \left(\left|\overline{M_\tau u}\right| + \left|\overline{\nabla_\Gamma M_\tau u}\right|\right) \le c\varepsilon \left(\left|\overline{Mu}\right| + \left|\overline{\nabla_\Gamma Mu}\right|\right) \quad \text{in} \quad U.$$

Here the last inequality follows from $M_{\tau}u = PMu$ and $P \in C^4(\Gamma)^{3\times 3}$. Therefore,

$$|(\bar{\tau}_i \cdot \nabla) u_n^a \cdot \bar{n}| \le c\varepsilon \left(\left| \overline{Mu} \right| + \left| \overline{\nabla_{\Gamma} Mu} \right| \right) \quad \text{in} \quad U, \, i = 1, 2.$$
(5.A.8)

Noting that $u^a = u^a_{\tau} + u^a_n$, we apply (5.A.6), (5.A.7), and (5.A.8) to the right-hand side of (5.A.5) to conclude that the inequality (5.7.2) is valid in U.

5.B Calculations involving differential geometry of surfaces

The purpose of this appendix is to give the proofs of the lemmas in Section 5.2 and related results, which involve calculations of the surface quantities on Γ , Γ_{ε}^{0} , and Γ_{ε}^{1} . We also show the formula (5.3.15) in the proof of Lemma 5.3.3.

First we assume that the closed surface Γ is of class C^{ℓ} with $\ell \geq 2$ and prove the lemmas given in Section 5.2.1 and a few results on Γ .

Proof of Lemma 5.2.2. Since W has the eigenvalues zero, κ_1 , and κ_2 ,

$$\det[I_3 - rW(y)] = \{1 - r\kappa_1(y)\}\{1 - r\kappa_2(y)\} > 0, \quad y \in \Gamma, \ r \in (-\delta, \delta)$$

by (5.2.2). Hence $I_3 - rW(y)$ is invertible. Also, the equality (5.2.8) follows from (5.2.6).

Let us prove (5.2.9) and (5.2.10). For the sake of simplicity, we fix and suppress the argument $y \in \Gamma$. Since W is real and symmetric by Lemma 5.2.1 and has the eigenvalues κ_1, κ_2 , and zero with Wn = 0, we can take an orthonormal basis $\{\tau_1, \tau_2, n\}$ of \mathbb{R}^3 such that $W\tau_i = \kappa_i \tau_i, i = 1, 2$. Then we have

$$(I_3 - rW)\tau_i = (1 - r\kappa_i)\tau_i, \quad (I_3 - rW)n = n$$

for $r \in (-\delta, \delta)$ and i = 1, 2, and thus

$$(I_3 - rW)^{-1}\tau_i = (1 - r\kappa_i)^{-1}\tau_i, \quad (I_3 - rW)^{-1}n = n.$$
 (5.B.1)

Since $\{\tau_1, \tau_2, n\}$ is an orthonormal basis of \mathbb{R}^3 , these formulas imply that

$$(I_3 - rW)^k a = \sum_{i=1,2} (a \cdot \tau_i) (I_3 - rW)^k \tau_i + (a \cdot n) (I_3 - rW)^k n$$
$$= \sum_{i=1,2} (a \cdot \tau_i) (1 - r\kappa_i)^k \tau_i + (a \cdot n) n$$

for all $a \in \mathbb{R}^3$ and $k = \pm 1$. Hence

$$|(I_3 - rW)^k a|^2 = \sum_{i=1,2} (a \cdot \tau_i)^2 (1 - r\kappa_i)^{2k} + (a \cdot n)^2$$

and (5.2.9) follows from (5.2.2) and $|a|^2 = (a \cdot \tau_1)^2 + (a \cdot \tau_2)^2 + (a \cdot n)^2$. Also, from (5.B.1), $|\tau_1| = |\tau_2| = 1$, and $|1 - (1 - r\kappa_i)^{-1}| \le |r\kappa_i(1 - r\kappa_i)^{-1}| \le c|r|$ by (5.2.2) we deduce that

$$|I_3 - (I_3 - rW)^{-1}|^2 = \sum_{i=1,2} |1 - (1 - r\kappa_i)^{-1}|^2 \le c|r|^2$$

Hence (5.2.10) is valid.

Proof of Lemma 5.2.3. By (5.2.1) and $n(\pi(x)) = \overline{n}(\pi(x))$ we have

$$\pi(x) = x - d(x)\bar{n}(\pi(x)), \quad x \in N.$$

We differentiate both sides in x and then use $\nabla d = \bar{n}$ and

$$-\nabla \bar{n}(\pi(x)) = -\overline{\nabla_{\Gamma} n}(\pi(x)) = \overline{W}(\pi(x)) = \overline{W}(x)$$

by (5.2.4) (note that $\pi(x) \in \Gamma$) to get

$$\nabla \pi(x) = I_3 - \bar{n}(x) \otimes \bar{n}(x) - d(x) \nabla \pi(x) \nabla \bar{n}(\pi(x)) = \overline{P}(x) + d(x) \nabla \pi(x) \overline{W}(x)$$

for $x \in N$, which implies that

$$\pi(x)\left\{I_3 - d(x)\overline{W}(x)\right\} = \overline{P}(x), \quad x \in N.$$

Since $I_3 - d(x)\overline{W}(x)$ is invertible for $x \in N$ by Lemma 5.2.2, the equality (5.2.11) follows from this equality and (5.2.8). Also, by (5.2.11) and $P\nabla_{\Gamma}\eta = \nabla_{\Gamma}\eta$ on Γ we get (5.2.12). The inequalities (5.2.13) and (5.2.14) follows from (5.2.9), (5.2.10), and (5.2.12).

Now suppose that Γ is of class C^3 and let us prove (5.2.15). For $\eta \in C^2(\Gamma)$ and i = 1, 2, 3we differentiate both sides of (5.2.12) with respect to x_i to get

$$\partial_i \nabla \bar{\eta} = \left\{ \partial_i \left(I_3 - d\overline{W} \right)^{-1} \right\} \overline{\nabla_{\Gamma} \eta} + \left(I_3 - d\overline{W} \right)^{-1} \partial_i \left(\overline{\nabla_{\Gamma} \eta} \right) \quad \text{in} \quad N.$$
(5.B.2)

To estimate the right-hand side we differentiate both sides of

$$\left\{I_3 - d(x)\overline{W}(x)\right\}^{-1}\left\{I_3 - d(x)\overline{W}(x)\right\} = I_3, \quad x \in \mathbb{N}$$

with respect to x_i and use $\nabla d = \bar{n}$ to get

$$\partial_i \left(I_3 - d\overline{W} \right)^{-1} = \left(I_3 - d\overline{W} \right)^{-1} \left(\bar{n}_i \overline{W} + d\partial_i \overline{W} \right) \left(I_3 - d\overline{W} \right)^{-1} \quad \text{in} \quad N.$$
(5.B.3)

Here the right-hand side is bounded on N by (5.2.9), (5.2.12), and the C^1 -regularity of W on Γ (note that Γ is of class C^3). Using this fact and (5.2.9) to (5.B.2) we obtain

$$|\partial_i \nabla \bar{\eta}| \le c \left(\left| \overline{\nabla_\Gamma \eta} \right| + \left| \overline{\nabla_\Gamma^2 \eta} \right| \right) \quad \text{in} \quad N, \, i = 1, 2, 3,$$

which shows (5.2.15). Moreover, by (5.2.4), (5.B.2), (5.B.3), and d = 0 on Γ ,

$$\partial_i \nabla \bar{\eta} = n_i W \nabla_\Gamma \eta + \underline{D}_i (\nabla_\Gamma \eta), \quad \text{i.e.} \quad \partial_i \partial_j \bar{\eta} = n_i \sum_{k=1}^3 W_{jk} \underline{D}_k \eta + \underline{D}_i \underline{D}_j \eta$$

on Γ for i, j = 1, 2, 3, which implies that

$$\Delta \bar{\eta} = \sum_{i=1}^{3} \partial_i^2 \bar{\eta} = \sum_{i,k=1}^{3} (n_i W_{ik} \underline{D}_k \eta + \underline{D}_i^2 \eta) = W^T n \cdot \nabla_{\Gamma} \eta + \Delta_{\Gamma} \eta.$$

Since $W^T n = W n = 0$ by Lemma 5.2.1, we obtain $\Delta \bar{\eta} = \Delta_{\Gamma} \eta$ on Γ .

Using the orthonormal frame $\{\tau_1, \tau_2, n\}$ given in the proof of Lemma 5.2.2 and the formula (5.2.12), we can express the divergence of the constant extension of a surface vector field in terms of the surface quantities on Γ .

Lemma 5.B.1. For $v \in C^1(\Gamma)^3$ let $\bar{v} = v \circ \pi$ be its constant extension. Then

$$\operatorname{div} \bar{v} = \frac{1}{(1 - d\bar{\kappa}_1)(1 - d\bar{\kappa}_2)} \left\{ \left(1 - d\overline{H} \right) \overline{\operatorname{div}_{\Gamma} v} + d \left(\overline{\nabla_{\Gamma} v} : \overline{W} \right) \right\} \quad in \quad N.$$
(5.B.4)

Moreover, div $\bar{v} = 0$ in N provided that $D_{\Gamma}(v) = 0$ on Γ , where $D_{\Gamma}(v)$ is the surface strain rate tensor given by (5.4.38).

Proof. Let $x = y + rn(y) \in N$ with $y = \pi(x) \in \Gamma$ and $r = d(x) \in (-\delta, \delta)$. In what follows, we suppress the argument y for functions on Γ . As in the proof of Lemma 5.2.2, we take

an orthonormal basis $\{\tau_1, \tau_2, n\}$ of \mathbb{R}^3 such that $W\tau_i = \kappa_i \tau_i$ for i = 1, 2. Then by (5.2.12), (5.B.1), and the symmetry of W we have

$$\operatorname{div} \bar{v}(x) = \operatorname{tr}[\nabla \bar{v}(x)] = \operatorname{tr}[(I_3 - rW)^{-1} \nabla_{\Gamma} v]$$
$$= \sum_{i=1,2} (I_3 - rW)^{-1} (\nabla_{\Gamma} v) \tau_i \cdot \tau_i + (I_3 - rW)^{-1} (\nabla_{\Gamma} v) n \cdot n$$
$$= \sum_{i=1,2} (\nabla_{\Gamma} v) \tau_i \cdot (1 - r\kappa_i)^{-1} \tau_i + (\nabla_{\Gamma} v) n \cdot n.$$

Here $(\nabla_{\Gamma} v)n \cdot n = P(\nabla_{\Gamma} v)n \cdot n = 0$. Hence

$$\operatorname{div} \bar{v}(x) = \frac{(1 - r\kappa_2)(\nabla_{\Gamma} v)\tau_1 \cdot \tau_1 + (1 - r\kappa_1)(\nabla_{\Gamma} v)\tau_2 \cdot \tau_2}{(1 - r\kappa_1)(1 - r\kappa_2)} = \frac{\{1 - r(\kappa_1 + \kappa_2)\}I_1 + rI_2}{(1 - r\kappa_1)(1 - r\kappa_2)},$$
(5.B.5)

where

$$I_1 := \sum_{i=1,2} (\nabla_{\Gamma} v) \tau_i \cdot \tau_i, \quad I_2 := \sum_{i=1,2} (\nabla_{\Gamma} v) \tau_i \cdot \kappa_i \tau_i$$

Since $(\nabla_{\Gamma} v)n \cdot n = 0$ and $\{\tau_1, \tau_2, n\}$ is an orthonormal basis of \mathbb{R}^3 ,

$$I_1 = \sum_{i=1,2} (\nabla_{\Gamma} v) \tau_i \cdot \tau_i + (\nabla_{\Gamma} v) n \cdot n = \operatorname{tr}[\nabla_{\Gamma} v] = \operatorname{div}_{\Gamma} v.$$
(5.B.6)

Also, since $W\tau_i = \kappa_i \tau_i$ and Wn = 0,

$$I_2 = \sum_{i=1,2} (\nabla_{\Gamma} v) \tau_i \cdot W \tau_i + (\nabla_{\Gamma} v) n \cdot W n = \nabla_{\Gamma} v : W.$$

We apply these equalities, $\kappa_1 + \kappa_2 = H$, and r = d(x) to (5.B.5) to get (5.B.4).

Now let v satisfy $D_{\Gamma}(v) = 0$ on Γ . Since $\{\tau_1, \tau_2, n\}$ is an orthonormal basis of \mathbb{R}^3 , τ_1 and τ_2 are tangent to Γ (at y). Hence $P\tau_i = \tau_i$ for i = 1, 2 and

$$0 = \operatorname{tr}[D_{\Gamma}(v)] = \sum_{i=1,2} P(\nabla_{\Gamma}v) P\tau_i \cdot \tau_i + P(\nabla_{\Gamma}v) Pn \cdot n = \sum_{i=1,2} (\nabla_{\Gamma}v)\tau_i \cdot \tau_i = \operatorname{div}_{\Gamma}v$$

by $P^T = P$, Pn = 0, and (5.B.6). Moreover, by (5.2.6) and the symmetry of W,

$$\nabla_{\Gamma} v : W = P(\nabla_{\Gamma} v)P : W = D_{\Gamma}(v) : W = 0.$$

Applying these equalities to (5.B.4) we observe that div $\bar{v} = 0$ in N.

Let us prove the density result on the Sobolev space on Γ .

Proof of Lemma 5.2.4. Here we only show the density of $C^{\ell}(\Gamma)$ in $W^{m,p}(\Gamma)$ in the case $\ell = m = 2$ and $p \in [1, \infty)$. The assertion in other cases are proved similarly.

Let $\eta \in W^{2,p}(\Gamma)$. By a standard localization argument with a partition of unity of Γ (note that Γ is compact), we may assume that there exist an open set U in \mathbb{R}^2 and a local parametrization $\mu: U \to \Gamma$ such that η is supported in $\mu(\mathcal{K})$, where \mathcal{K} is a compact subset

of U. Since Γ is of class C^2 , the local parametrization μ is of class C^2 on U and thus there exists a constant c > 0 such that

$$|\partial_{s_i}\mu(s)| \le c, \quad |\partial_{s_i}\partial_{s_j}\mu(s)| \le c, \quad s \in \mathcal{K}, \, i, j = 1, 2.$$
(5.B.7)

We denote by $\theta = (\theta_{ij})_{i,j}$ the Riemannian metric of Γ . It is locally given by

$$\theta_{ij}(s) := \partial_{s_i} \mu(s) \cdot \partial_{s_j} \mu(s), \quad s \in U, \, i, j = 1, 2.$$

Since the determinant of θ is continuous and does not vanish on U, it is bounded from above and below by a positive constant on \mathcal{K} :

$$c^{-1} \le \det \theta(s) \le c, \quad s \in \mathcal{K}.$$
 (5.B.8)

Let $\theta^{-1} = (\theta^{ij})_{i,j}$ be the inverse matrix of θ . Since θ_{ij} and its first order derivatives are bounded on \mathcal{K} by (5.B.7) and $\partial_{s_k} \theta^{-1} = -\theta^{-1} (\partial_{s_k} \theta) \theta^{-1}$, we have

$$|\theta^{ij}(s)| \le c, \quad |\partial_{s_k}\theta^{ij}(s)| \le c, \quad s \in \mathcal{K}, \, i, j, k = 1, 2.$$
(5.B.9)

Moreover, there exists c > 0 such that

$$c^{-1}|a|^2 \le \theta^{-1}(s)a \cdot a \le c|a|^2, \quad s \in \mathcal{K}, \, a \in \mathbb{R}^2.$$
 (5.B.10)

To see this, let $X(s,a) := \sum_{i,j=1}^{2} \theta^{ij}(s) a_i \partial_{s_j} \mu(s)$ for $s \in U$ and $a = (a_1, a_2) \in \mathbb{R}^2$. Since $\partial_{s_1} \mu(s)$ and $\partial_{s_2} \mu(s)$ are linearly independent, X(s, a) vanishes if and only if

$$\sum_{i=1,2} \theta^{ij}(s)a_i = \sum_{i=1,2} \theta^{ji}(s)a_i = 0 \quad \text{for} \quad j = 1, 2, \quad \text{i.e.} \quad \theta^{-1}(s)a = 0$$

(Note that θ^{-1} is symmetric since θ is so.) By this fact we observe that X(s, a) = 0 if and only if a = 0 and the function

$$|X(s,a)|^{2} = \sum_{i,j=1}^{2} \theta^{ij}(s)a_{i}a_{j} = \theta^{-1}(s)a \cdot a_{i}a_{j}$$

is continuous for $(s, a) \in U \times \mathbb{R}^2$ and does not vanish for $a \neq 0$. In particular, it is bounded from above and below by positive constants on the compact set $\mathcal{K} \times S^1$, where S^1 is the unit circle in \mathbb{R}^2 . Hence (5.B.10) follows.

For $s \in U$ let $\tilde{\eta}(s) := \eta(\mu(s))$. We show that there exists c > 0 such that

$$c^{-1} \|\tilde{\eta}\|_{W^{2,p}(U)} \le \|\eta\|_{W^{2,p}(\mu(U))} \le c \|\tilde{\eta}\|_{W^{2,p}(U)}.$$
(5.B.11)

By the definition of an integral over a surface,

$$\int_{\mu(U)} |\eta(y)|^p \, d\mathcal{H}^2(y) = \int_U |\tilde{\eta}(s)|^p \sqrt{\det \theta(s)} \, ds.$$

Noting that η is supported in $\mu(\mathcal{K})$, we use (5.B.8) to the this equality to get

$$c^{-1} \|\tilde{\eta}\|_{L^{p}(U)} \le \|\eta\|_{L^{p}(\mu(U))} \le c \|\tilde{\eta}\|_{L^{p}(U)}.$$
(5.B.12)

Next we compare the first order derivatives of η and $\tilde{\eta}$. Since

$$\nabla_{\Gamma}\eta(\mu(s)) = \sum_{i,j=1}^{2} \theta^{ij}(s)\partial_{s_{i}}\tilde{\eta}(s)\partial_{s_{j}}\mu(s),$$

$$|\nabla_{\Gamma}\eta(\mu(s))|^{2} = \sum_{i,j=1}^{2} \theta^{ij}(s)\partial_{s_{i}}\tilde{\eta}(s)\partial_{s_{j}}\tilde{\eta}(s) = \theta^{-1}(s)\nabla_{s}\tilde{\eta}(s) \cdot \nabla_{s}\tilde{\eta}(s)$$
(5.B.13)

for $s \in U$, where ∇_s is the gradient operator in $s \in \mathbb{R}^2$, by (5.B.10) we have

$$c^{-1}|\nabla_s \tilde{\eta}(s)| \le |\nabla_\Gamma \eta(\mu(s))| \le c|\nabla_s \tilde{\eta}(s)|, \quad s \in \mathcal{K}.$$

We apply this inequality and (5.B.8) to

$$\int_{\mu(U)} |\nabla_{\Gamma} \eta(y)|^p \, d\mathcal{H}^2(y) = \int_U |\nabla_{\Gamma} \eta(\mu(s))|^p \sqrt{\det \theta(s)} \, ds$$

(note that η is supported in $\mu(\mathcal{K})$) to obtain

$$c^{-1} \|\nabla_{s} \tilde{\eta}\|_{L^{p}(U)} \leq \|\nabla_{\Gamma} \eta\|_{L^{p}(\mu(U))} \leq c \|\nabla_{s} \tilde{\eta}\|_{L^{p}(U)}.$$
(5.B.14)

Let us consider the second order derivatives of η and $\tilde{\eta}$. For $s \in U$ and k = 1, 2, 3 we set $\tilde{\eta}_k(s) := \underline{D}_k \eta(\mu(s))$. Then by the right-hand inequality of (5.B.14) we see that

$$\|\nabla_{\Gamma}\underline{D}_k\eta\|_{L^p(\mu(U))} \le c\|\nabla_s\tilde{\eta}_k\|_{L^p(U)}.$$

Moreover, since $\underline{D}_k \eta = \nabla_{\Gamma} \eta \cdot e_k$, where $\{e_1, e_2, e_3\}$ is the standard basis of \mathbb{R}^3 ,

$$\tilde{\eta}_k(s) = \nabla_{\Gamma} \eta(\mu(s)) \cdot e_k = \sum_{i,j=1}^2 \theta^{ij}(s) \partial_{s_i} \tilde{\eta}(s) \partial_{s_j} \mu(s) \cdot e_k, \quad s \in U$$

by (5.B.13). Thus, applying ∇_s to both sides and using (5.B.7) and (5.B.9) we get

$$|\nabla_s \tilde{\eta}_k(s)| \le c(|\nabla_s \tilde{\eta}(s)| + |\nabla_s^2 \tilde{\eta}(s)|), \quad s \in \mathcal{K}.$$

Since η is supported in $\mu(\mathcal{K})$, we find by the above inequalities that

$$\|\nabla_{\Gamma}\underline{D}_k\eta\|_{L^p(\mu(U))} \le c\|\nabla_s\tilde{\eta}_k\|_{L^p(U)} \le c\|\nabla_s\tilde{\eta}\|_{W^{1,p}(U)}$$

for k = 1, 2, 3. Therefore,

$$\|\nabla_{\Gamma}^2 \eta\|_{L^p(\mu(U))} \le c \|\tilde{\eta}\|_{W^{2,p}(U)}.$$
(5.B.15)

To estimate the $L^p(U)$ -norms of the second order derivatives of $\tilde{\eta}$, we take the inner product of the first equality of (5.B.13) with $\partial_{s_k}\mu(s)$, k = 1, 2, 3 to get

$$\nabla_{\Gamma}\eta(\mu(s)) \cdot \partial_{s_k}\mu(s) = \sum_{i,j=1}^{2} \theta^{ij}(s)\theta_{jk}(s)\partial_{s_i}\tilde{\eta}(s) = \partial_{s_k}\tilde{\eta}(s), \quad s \in U.$$

Since $\mu(s) \in \Gamma$, we have $\nabla_{\Gamma} \eta(\mu(s)) = \overline{\nabla_{\Gamma} \eta}(\mu(s))$, where the right-hand side stands for the constant extension of $\nabla_{\Gamma} \eta$ in the normal direction of Γ , and thus

$$\partial_{s_l} \Big(\nabla_{\Gamma} \eta(\mu(s)) \Big) = \partial_{s_l} \Big(\overline{\nabla_{\Gamma} \eta}(\mu(s)) \Big) = \Big[\nabla \Big(\overline{\nabla_{\Gamma} \eta} \Big) \Big] (\mu(s)) \partial_{s_l} \mu(s) = \nabla_{\Gamma}^2 \eta(\mu(s)) \partial_{s_l} \mu(s), \quad l = 1, 2, 3$$

by (5.2.4). Hence for $s \in U$ and k, l = 1, 2, 3 we have

$$\partial_{s_l}\partial_{s_k}\tilde{\eta}(s) = \partial_{s_l} \Big(\nabla_{\Gamma}\eta(\mu(s)) \cdot \partial_{s_k}\mu(s) \Big) \\ = \nabla_{\Gamma}^2\eta(\mu(s))\partial_{s_l}\mu(s) \cdot \partial_{s_k}\mu(s) + \nabla_{\Gamma}\eta(\mu(s)) \cdot \partial_{s_l}\partial_{s_k}\mu(s)$$

and by applying (5.B.7) to the last line we deduce that

$$|\nabla_s^2 \tilde{\eta}(s)| \le c(|\nabla_\Gamma \eta(\mu(s))| + |\nabla_\Gamma^2 \eta(\mu(s))|), \quad s \in \mathcal{K}.$$

Noting that η is supported in $\mu(\mathcal{K})$ and

$$\int_{\mu(U)} |\nabla_{\Gamma}^k \eta(y)|^p \, d\mathcal{H}^2(y) = \int_U |\nabla_{\Gamma}^k \eta(\mu(s))|^p \sqrt{\det \theta(s)} \, ds, \quad k = 1, 2,$$

we use the above inequality and (5.B.8) to the $L^p(U)$ -norm of $\nabla_s^2 \tilde{\eta}$ to get

$$\|\nabla_s^2 \tilde{\eta}\|_{L^p(U)} \le c \|\eta\|_{W^{2,p}(\mu(U))}.$$
(5.B.16)

By (5.B.12), (5.B.14), (5.B.15), and (5.B.16) we obtain (5.B.11).

Now we observe by $\eta \in W^{2,p}(\Gamma)$ and the left-hand inequality of (5.B.11) that $\tilde{\eta}$ is in $W^{2,p}(U)$. Since it is also supported in the compact subset \mathcal{K} of U, by a standard mollification argument (see e.g. [1, Lemma 3.16]) we can take a sequence $\{\tilde{\eta}_k\}_{k=1}^{\infty}$ in $C^{\infty}(U)$ that converges to $\tilde{\eta}$ strongly in $W^{2,p}(U)$. Hence the right-hand inequality of (5.B.11) yields that

$$\|\eta - \eta_k\|_{W^{2,p}(\mu(U))} \le c \|\tilde{\eta} - \tilde{\eta}_k\|_{W^{2,p}(U)} \to 0 \quad \text{as} \quad k \to \infty,$$

where $\eta_k(y) := \tilde{\eta}_k(\mu^{-1}(y))$ for $k \in \mathbb{N}$ and $y \in \Gamma$. Since the local parametrization $\mu : U \to \Gamma$ is of class C^2 , we have $\eta_k \in C^2(\mu(U))$ for each $k \in \mathbb{N}$. Hence $C^2(\Gamma)$ is dense in $W^{2,p}(\Gamma)$ (recall that we localized η by using a partition of unity of Γ).

Next we give a change of variables formula for an integral over a parametrized surface. Let $h \in C^1(\Gamma)$ satisfy $|h| < \delta$ on Γ . We define a parametrized surface

$$\Gamma_h := \{ y + h(y)n(y) \mid y \in \Gamma \} \subset \mathbb{R}^3.$$
(5.B.17)

Note that $\Gamma_h \subset N$ by $|h| < \delta$ on Γ (see Section 5.2.1). For $y \in \Gamma$ we set

$$\tau_h(y) := \{I_3 - h(y)W(y)\}^{-1} \nabla_{\Gamma} h(y), \quad n_h(y) := \frac{n(y) - \tau_h(y)}{\sqrt{1 + |\tau_h(y)|^2}}.$$
(5.B.18)

Here the vector field τ_h is tangential on Γ . We assume that the orientation of Γ_h is the same as that of Γ . Then as in the proof of Lemma 5.2.10 below we can show that the constant extension $\bar{n}_h := n_h \circ \pi$ gives the unit outward normal vector field of Γ_h .

For $\varphi \in C^1(\Gamma_h)$ we define the tangential gradient $\nabla_{\Gamma_h} \varphi$ as

$$\nabla_{\Gamma_h}\varphi(x) := \{I_3 - \bar{n}_h(x) \otimes \bar{n}_h(x)\} \nabla \tilde{\varphi}(x), \quad x \in \Gamma_h,$$

where $\tilde{\varphi}$ is an arbitrary extension of φ to N satisfying $\tilde{\varphi}|_{\Gamma_h} = \varphi$. Let us give a change of variables formula for integrals over Γ_h .

Lemma 5.B.2. Suppose that Γ is of class C^2 and $h \in C^1(\Gamma)$ satisfies $|h| < \delta$ on Γ . Let Γ_h be the parametrized surface given by (5.B.17). For $\varphi \in L^1(\Gamma_h)$ we have

$$\int_{\Gamma_h} \varphi(x) \, d\mathcal{H}^2(x) = \int_{\Gamma} \varphi^{\sharp}(y) J(y, h(y)) \sqrt{1 + |\tau_h(y)|^2} \, d\mathcal{H}^2(y), \tag{5.B.19}$$

where $\varphi^{\sharp}(y) := \varphi(y + h(y)n(y))$ for $y \in \Gamma$ and J and τ_h are given by (5.2.48) and (5.B.18).

Before starting to prove Lemma 5.B.2 we give a remark on a partition of unity of Γ_h . Since the surface Γ is compact, we can take finite open subsets U_k of \mathbb{R}^2 and local parametrizations $\mu^k \colon U_k \to \Gamma, \ k = 1, \ldots, k_0$ such that $\{\mu^k(U_k)\}_{k=1}^{k_0}$ is an open covering of Γ . Let $\{\eta^k\}_{k=1}^{k_0}$ be a partition of unity of Γ subordinate to the covering $\{\mu^k(U_k)\}_{k=1}^{k_0}$. Then setting

$$\mu_h^k(s) := \mu_k(s) + h(\mu_k(s))n(\mu_k(s)), \quad \eta_h^k(\mu_h^k(s)) := \eta^k(\mu^k(s)), \quad s \in U_k$$

we observe that $\mu_h^k \colon U_k \to \Gamma_h$, $k = 1, \ldots, k_0$ are local parametrizations of Γ_h such that $\{\mu_h^k(U_k)\}_{k=1}^{k_0}$ is an open covering of Γ_h , and that $\{\eta_h^k\}_{k=1}^{k_0}$ is a partition of unity of Γ_h subordinate to the covering $\{\mu_h^k(U_k)\}_{k=1}^{k_0}$. Using these local parametrizations and partitions of unity we can localize integrals over Γ and Γ_h and express them as integrals over the same domains U_k , $k = 1, \ldots, k_0$.

Proof. Let U be an open subset of \mathbb{R}^2 and $\mu: U \to \Gamma$ be a local parametrization of Γ . The Riemannian metric of Γ is locally given by $\theta = \nabla_s \mu (\nabla_s \mu)^T$ on U, where

$$\nabla_s \mu := \begin{pmatrix} \partial_{s_1} \mu_1 & \partial_{s_1} \mu_2 & \partial_{s_1} \mu_3 \\ \partial_{s_2} \mu_1 & \partial_{s_2} \mu_2 & \partial_{s_2} \mu_3 \end{pmatrix}.$$

Note that $\nabla_s \mu(s) n(\mu(s)) = 0$ for $s \in U$ since $\partial_{s_1} \mu(s)$ and $\partial_{s_2} \mu(s)$ are tangent to Γ at $\mu(s)$. Using μ we give a local parametrization $\mu_h \colon U \to \Gamma_h$ and the Riemannian metric θ_h of Γ_h by

$$\mu_h(s) := \mu(s) + h(\mu(s))n(\mu(s)), \quad \theta_h(s) := \nabla_s \mu_h(s) \{\nabla_s \mu_h(s)\}^T, \quad s \in U.$$

Hereafter we use the notation $\eta^{\sharp}(s) := \eta(\mu(s)), s \in U$ for a function η on Γ and suppress the argument $s \in U$. Then by a localization argument with partitions of unity of Γ and Γ_h mentioned above, the proof of (5.B.19) reduces to show

$$\sqrt{\det \theta_h} = J(\mu, h^{\sharp}) \sqrt{(1 + |\tau_h^{\sharp}|^2) \det \theta} \quad \text{on} \quad U.$$
(5.B.20)

We prove (5.B.20) in two steps. First we show the equality

$$(1 - |(\nabla_{\Gamma_h} \bar{h}) \circ \mu_h|^2) \det \theta_h = J(\mu, h^{\sharp})^2 \det \theta \quad \text{on} \quad U,$$
(5.B.21)

where $\bar{h} := h \circ \pi$ is the constant extension of h in the normal direction of Γ . We differentiate both sides of $\mu_h = \mu + h^{\sharp} n^{\sharp}$ in s and use

$$\nabla_s n^{\sharp}(s) = \nabla_s \big(\bar{n}(\mu(s)) \big) = \nabla_s \mu(s) \nabla \bar{n}(\mu(s)) = -\nabla_s \mu(s) W(\mu(s))$$

by (5.2.4) (note that $\mu(s) \in \Gamma$). Then

$$\nabla_s \mu_h = \nabla_s \mu (I_3 - h^{\sharp} W^{\sharp}) + \nabla_s h^{\sharp} \otimes n^{\sharp} \quad \text{on} \quad U.$$

Also, noting that W^{\sharp} is a symmetric matrix we get

$$(\nabla_s \mu_h)^T = (I_3 - h^{\sharp} W^{\sharp})(\nabla_s \mu) + n^{\sharp} \otimes \nabla_s h^{\sharp}$$
 on U .

From these equalities, $(\nabla_s \mu) n^{\sharp} = 0$, $W^{\sharp} n^{\sharp} = 0$, and

$$(\nabla_s h^{\sharp} \otimes n^{\sharp})(n^{\sharp} \otimes \nabla_s h^{\sharp}) = |n^{\sharp}|^2 \nabla_s h^{\sharp} \otimes \nabla_s h^{\sharp} = \nabla_s h^{\sharp} \otimes \nabla_s h^{\sharp}$$

we deduce that

$$\theta_h = \nabla_s \mu_h (\nabla_s \mu_h)^T = \nabla_s \mu (I_3 - h^{\sharp} W^{\sharp})^2 (\nabla_s \mu)^T + \nabla_s h^{\sharp} \otimes \nabla_s h^{\sharp}$$

and thus

$$\det(\theta_h - \nabla_s h^{\sharp} \otimes \nabla_s h^{\sharp}) = \det[\nabla_s \mu (I_3 - h^{\sharp} W^{\sharp})^2 (\nabla_s \mu)^T] \quad \text{on} \quad U.$$
(5.B.22)

The formula $\det(I_2 + a \otimes b) = 1 + a \cdot b$ for $a, b \in \mathbb{R}^2$ yields

$$\det(\theta_h - \nabla_s h^{\sharp} \otimes \nabla_s h^{\sharp}) = \det[I_3 - (\theta_h^{-1} \nabla_s h^{\sharp}) \otimes \nabla_s h^{\sharp}] \det \theta_h$$
$$= \{1 - (\theta_h^{-1} \nabla_s h^{\sharp}) \cdot \nabla_s h^{\sharp}\} \det \theta_h.$$

To calculate the right-hand side of (5.B.22) we set

$$A := \begin{pmatrix} \nabla_s \mu \\ (n^{\sharp})^T \end{pmatrix}, \quad A_h := \begin{pmatrix} \nabla_s \mu (I_3 - h^{\sharp} W^{\sharp}) \\ (n^{\sharp})^T \end{pmatrix}.$$

Here we see $n^{\sharp} \in \mathbb{R}^3$ as a column vector. Then by $(\nabla_s \mu)n^{\sharp} = 0$, $W^{\sharp}n^{\sharp} = 0$, and the symmetry of the matrix W^{\sharp} we have $A_h = A(I_3 - h^{\sharp}W^{\sharp})$ and

$$AA^{T} = \begin{pmatrix} \theta & 0 \\ 0 & 1 \end{pmatrix}, \quad A_{h}A_{h}^{T} = \begin{pmatrix} \nabla_{s}\mu(I_{3} - h^{\sharp}W^{\sharp})(\nabla_{s}\mu)^{T} & 0 \\ 0 & 1 \end{pmatrix}.$$

From these equalities and $\det(I_3 - h^{\sharp}W^{\sharp}) = J(\mu, h^{\sharp})$ it follows that

$$det[\nabla_s \mu (I_3 - h^{\sharp} W^{\sharp})^2 (\nabla_s \mu)^T] = det[A_h A_h^T] = det[A(I_3 - h^{\sharp} W^{\sharp})^2 A^T]$$
$$= det[(I_3 - h^{\sharp} W^{\sharp})^2] det[AA^T]$$
$$= J(\mu, h^{\sharp})^2 det \theta.$$
(5.B.23)

(Note that A and $I_3 - h^{\sharp}W^{\sharp}$ are 3×3 matrices.) Hence (5.B.22) reads

$$\{1 - (\theta_h^{-1} \nabla_s h^{\sharp}) \cdot \nabla_s h^{\sharp}\} \det \theta_h = J(\mu, h^{\sharp})^2 \det \theta \quad \text{on} \quad U.$$
(5.B.24)

Now we recall that the tangential gradient of \bar{h} on Γ_h is locally expressed as

$$\nabla_{\Gamma_h}\bar{h}(\mu_h(s)) = \sum_{i,j=1}^2 \theta_h^{ij}(s) \frac{\partial(\bar{h} \circ \mu_h)}{\partial s_i}(s) \partial_{s_j} \mu_h(s), \quad s \in U.$$

Here θ_h^{ij} is the (i, j)-entry of θ_h^{-1} . Since $\bar{h} = h \circ \pi$ is the constant extension of h and $\pi(\mu_h(s)) = \mu(s)$, we have $\bar{h}(\mu_h(s)) = h(\mu(s)) = h^{\sharp}(s)$ and thus

$$(\nabla_{\Gamma_h}\bar{h}) \circ \mu_h = \sum_{i,j=1}^2 \theta_h^{ij} (\partial_{s_i} h^{\sharp}) \partial_{s_j} \mu,$$
$$\left| (\nabla_{\Gamma_h}\bar{h}) \circ \mu_h \right|^2 = \sum_{i,j=1}^2 \theta_h^{ij} (\partial_{s_i} h^{\sharp}) (\partial_{s_j} h^{\sharp}) = (\theta_h^{-1} \nabla_s h^{\sharp}) \cdot \nabla_s h^{\sharp}.$$

Applying this equality to the left-hand side of (5.B.24) we obtain (5.B.21).

Next we show that

$$1 - \left| \nabla_{\Gamma_h} \bar{h}(y + h(y)n(y)) \right|^2 = \frac{1}{1 + |\tau_h(y)|^2}, \quad y \in \Gamma,$$
(5.B.25)

where τ_h is the tangential vector field on Γ given by (5.B.18). For the sake of simplicity, we set $\alpha := \sqrt{1 + |\tau_h|^2}$ and sometimes suppress the argument $y \in \Gamma$ of functions on Γ . By (5.2.12) and d(y + h(y)n(y)) = h(y) we have

$$\nabla \bar{h}(x)|_{x=y+h(y)n(y)} = \{I_3 - h(y)W(y)\}^{-1} \nabla_{\Gamma} h(y) = \tau_h(y).$$

Since $\nabla_{\Gamma_h} \bar{h} = (I_3 - \bar{n}_h \otimes \bar{n}_h) \nabla \bar{h}$ on Γ_h and $n_h = \alpha^{-1} (n - \tau_h)$ on Γ , we see by the above equality that

$$\nabla_{\Gamma_h} \bar{h}(y+h(y)n(y)) = \{I_3 - \alpha^{-2}(n-\tau_h) \otimes (n-\tau_h)\} \tau_h = \alpha^{-2} \{|\tau_h|^2 n + (\alpha^2 - |\tau_h|^2) \tau_h\}.$$

From this equality, $n \cdot \tau_h = 0$, and $\alpha^2 = 1 + |\tau_h|^2$ we deduce that

$$1 - \left| \nabla_{\Gamma_h} \bar{h}(y + h(y)n(y)) \right|^2 = 1 - \alpha^{-4} (|\tau_h|^4 + |\tau_h|^2) = 1 - \alpha^{-2} |\tau_h|^2.$$

Since $\alpha^{-2} = (1 + |\tau_h|^2)^{-1}$, the above equality implies (5.B.25). Finally, we conclude by (5.B.21) and (5.B.25) that (5.B.20) is valid.

Now we assume that Γ is of class C^5 and prove the formulas and inequalities in Section 5.2.2 for the surface quantities on the boundary Γ_{ε} of the curved thin domain.

Proof of Lemma 5.2.9. First note that, since Γ is of class C^5 , the Weingarten map $W \in C^3(\Gamma)^{3\times 3}$ and the functions $g_0, g_1 \in C^4(\Gamma)$ are bounded on Γ along with their first and second order tangential derivatives.

Let τ_{ε}^{i} and n_{ε}^{i} , i = 0, 1 be the vector fields on Γ given by (5.2.32) and (5.2.33). Then the first inequalities of (5.2.34) and (5.2.35) immediately follow from (5.2.9) and (5.2.10). To show the second inequalities of (5.2.34) and (5.2.35) we set

$$R^i_{\varepsilon}(y) := \{I_3 - \varepsilon g_i(y)W(y)\}^{-1}, \quad y \in \Gamma$$

and apply \underline{D}_k , k = 1, 2, 3 to both sides of $R^i_{\varepsilon}(I_3 - \varepsilon g_i W) = I_3$ on Γ to get

$$\underline{D}_k R^i_{\varepsilon} = \varepsilon R^i_{\varepsilon} \{ (\underline{D}_k g_i) W + g_i \underline{D}_k W \} R^i_{\varepsilon} \quad \text{on} \quad \Gamma.$$
(5.B.26)

Then by (5.2.9) we see that there exists a constant c > 0 independent of ε such that

$$|\underline{D}_k R^i_{\varepsilon}| \le c\varepsilon \quad \text{on} \quad \Gamma. \tag{5.B.27}$$

Applying (5.2.9), (5.2.10), and (5.B.27) to $\underline{D}_k \tau_{\varepsilon}^i = (\underline{D}_k R_{\varepsilon}^i) \nabla_{\Gamma} g_i + R_{\varepsilon}^i (\underline{D}_k \nabla_{\Gamma} g)$ we obtain

$$|\underline{D}_k \tau_{\varepsilon}^i| \leq c, \quad |\underline{D}_k \tau_{\varepsilon}^i - \underline{D}_k \nabla_{\Gamma} g| \leq |(\underline{D}_k R_{\varepsilon}^i) \nabla_{\Gamma} g_i| + |(R_{\varepsilon}^i - I_3)(\underline{D}_k \nabla_{\Gamma} g)| \leq c\varepsilon$$

on Γ for all k = 1, 2, 3. Hence the second inequalities of (5.2.34) and (5.2.35) follows. We further apply \underline{D}_l , l = 1, 2, 3 to both sides of (5.B.26) and use (5.2.9) and (5.B.27) to get $|\underline{D}_l \underline{D}_k R_{\varepsilon}^i| \leq c\varepsilon$ on Γ . From this inequality, (5.2.9), (5.B.27) we deduce that

$$\begin{split} |\underline{D}_{l}\underline{D}_{k}\tau_{\varepsilon}^{i}| &\leq |(\underline{D}_{l}\underline{D}_{k}R_{\varepsilon}^{i})\nabla_{\Gamma}g_{i}| + |(\underline{D}_{k}R_{\varepsilon}^{i})(\underline{D}_{l}\nabla_{\Gamma}g_{i})| \\ &+ |(\underline{D}_{l}R_{\varepsilon}^{i})(\underline{D}_{k}\nabla_{\Gamma}g_{i})| + |R_{\varepsilon}^{i}(\underline{D}_{l}\underline{D}_{k}\nabla_{\Gamma}g_{i})| \leq c. \end{split}$$

Hence the third inequality of (5.2.34) is valid.

Now let us prove the inequalities for n_{ε}^i . The first equality of (5.2.36) is a direct consequence of (5.2.33). Also, the other inequalities of (5.2.36) follow from (5.2.34). To prove (5.2.37) we see that $n_{\varepsilon}^0 + n_{\varepsilon}^1 = \varphi_{\varepsilon}n + \varepsilon\tau_{\varepsilon}$, where

$$\varphi_{\varepsilon} := \frac{1}{\sqrt{1 + \varepsilon^2 |\tau_{\varepsilon}^1|^2}} - \frac{1}{\sqrt{1 + \varepsilon^2 |\tau_{\varepsilon}^0|^2}}, \quad \tau_{\varepsilon} := -\frac{\tau_{\varepsilon}^1}{\sqrt{1 + \varepsilon^2 |\tau_{\varepsilon}^1|^2}} + \frac{\tau_{\varepsilon}^0}{\sqrt{1 + \varepsilon^2 |\tau_{\varepsilon}^0|^2}}.$$

From (5.2.34) it follows that

$$|\tau_{\varepsilon}| \le c, \quad |\nabla_{\Gamma} \tau_{\varepsilon}| \le c \quad \text{on} \quad \Gamma$$
 (5.B.28)

with a positive constant c independent of ε . Let us estimate φ_{ε} and its tangential gradient. By the mean value theorem for the function $(1+s)^{-1/2}$, $s \ge 0$ and (5.2.34).

$$|\varphi_{\varepsilon}| \le \frac{\varepsilon^2}{2} \left| |\tau_{\varepsilon}^1|^2 - |\tau_{\varepsilon}^0|^2 \right| \le c\varepsilon^2 \quad \text{on} \quad \Gamma.$$
(5.B.29)

Also, since

$$\nabla_{\Gamma}\left(\frac{1}{\sqrt{1+\varepsilon^2|\tau_{\varepsilon}^i|^2}}\right) = -\frac{\varepsilon^2(\nabla_{\Gamma}\tau_{\varepsilon}^i)\tau_{\varepsilon}^i}{(1+\varepsilon^2|\tau_{\varepsilon}^i|^2)^{3/2}}, \quad i=0,1,$$

we use (5.2.34) to get

$$|\nabla_{\Gamma}\varphi_{\varepsilon}| \le c\varepsilon^2 \quad \text{on} \quad \Gamma. \tag{5.B.30}$$

Applying (5.B.28), (5.B.29), and (5.B.30) to $n_{\varepsilon}^{0} + n_{\varepsilon}^{1} = \varphi_{\varepsilon}n + \varepsilon\tau_{\varepsilon}$ and its tangential gradient matrix we obtain (5.2.37).

Proof of Lemma 5.2.10. We take an open subset U of \mathbb{R}^2 and a local parametrization $\mu: U \to \Gamma$. Then for i = 0, 1 the mapping

$$\mu_{\varepsilon}^{i}(s) := \mu(s) + \varepsilon g_{i}(\mu(s))n(\mu(s)), \quad s \in U$$

defines a local parametrization $\mu_{\varepsilon}^i \colon U \to \Gamma_{\varepsilon}^i$ and the pair $\{\partial_{s_1} \mu_{\varepsilon}^i(s), \partial_{s_2} \mu_{\varepsilon}^i(s)\}$ is a basis of the tangent plane of Γ_{ε}^i at $\mu_{\varepsilon}^i(s)$. Hence if we show

$$\bar{n}^i_{\varepsilon}(\mu^i_{\varepsilon}(s)) \cdot \partial_{s_k} \mu^i_{\varepsilon}(s) = 0, \quad s \in U, \, k = 1, 2,$$

then $\bar{n}_{\varepsilon}^{i}$ is normal to Γ_{ε}^{i} and thus $n_{\varepsilon} = \bar{n}_{\varepsilon}^{i}$ on Γ_{ε}^{i} , since $|n_{\varepsilon}^{i}| = 1$ and both n_{ε} and $\bar{n}_{\varepsilon}^{i}$ have the direction of $(-1)^{i+1}\bar{n}$. Moreover, since $\bar{n}_{\varepsilon}^{i}(\mu_{\varepsilon}^{i}(s)) = n_{\varepsilon}^{i}(\mu(s))$ is given by (5.2.33) (note that $\pi(\mu_{\varepsilon}^{i}(s)) = \mu(s)$), the above equality reduces to

$$n(\mu(s)) \cdot \partial_{s_k} \mu_{\varepsilon}^i(s) = \varepsilon \tau_{\varepsilon}^i(\mu(s)) \cdot \partial_{s_k} \mu_{\varepsilon}^i(s), \quad s \in U, \, k = 1, 2.$$
(5.B.31)

To prove (5.B.31) we differentiate both sides of

$$\mu_{\varepsilon}^{i}(s) = \mu(s) + \varepsilon \bar{g}_{i}(\mu(s))\bar{n}(\mu(s)), \quad s \in U$$

with respect to s_k , k = 1, 2 and use (5.2.4) (note that $\mu(s) \in \Gamma$) to get

$$\partial_{s_k} \mu_{\varepsilon}^i(s) = \{ I_3 - \varepsilon g_i(\mu(s)) W(\mu(s)) \} \partial_{s_k} \mu(s) + \varepsilon \partial_{s_k} \mu(s) \cdot \nabla_{\Gamma} g_i(\mu(s)) n(\mu(s)).$$

Here the first term on the right-hand side is tangent to Γ at $\mu(s)$ since μ is a local parametrization of Γ and W = PW on Γ . Hence

$$n(\mu(s)) \cdot \partial_{s_k} \mu_{\varepsilon}^i(s) = \varepsilon \partial_{s_k} \mu(s) \cdot \nabla_{\Gamma} g_i(\mu(s))$$

Also, since $\tau_{\varepsilon}^{i} = (I_{3} - \varepsilon g_{i}W)^{-1} \nabla_{\Gamma} g_{i}$ is tangential on Γ and W is symmetric,

$$\varepsilon \tau_{\varepsilon}^{i}(\mu(s)) \cdot \partial_{s_{k}} \mu_{\varepsilon}^{i}(s) = \varepsilon (I_{3} - \varepsilon g_{i}W)^{-1} \nabla_{\Gamma} g_{i}(\mu(s)) \cdot (I_{3} - \varepsilon g_{i}W) \partial_{s_{k}} \mu(s)$$
$$= \varepsilon \partial_{s_{k}} \mu(s) \cdot \nabla_{\Gamma} g_{i}(\mu(s)).$$

Here we suppressed the argument $\mu(s)$ of the function $I_3 - \varepsilon g_i W$. Therefore, we obtain (5.B.31) and conclude that the assertion $n_{\varepsilon} = \bar{n}_{\varepsilon}^i$ on Γ_{ε}^i , i = 0, 1 is valid.

Proof of Lemma 5.2.11. For i = 0, 1 let τ_{ε}^{i} and n_{ε}^{i} be given by (5.2.32) and (5.2.33) and

$$\varphi_{\varepsilon}^{i}(x) := \frac{1}{\sqrt{1 + \varepsilon^{2} |\bar{\tau}_{\varepsilon}^{i}(x)|^{2}}} - 1, \quad x \in N.$$

By direct calculations and the inequalities (5.2.13), (5.2.15), and (5.2.34) we see that

$$|\partial_x^{\alpha} \varphi_{\varepsilon}^i(x)| \le c\varepsilon^2 \quad \text{for all} \quad x \in N, \, |\alpha| = 0, 1, 2, \tag{5.B.32}$$

where $\partial_x^{\alpha} = \partial_1^{\alpha_1} \partial_2^{\alpha_2} \partial_3^{\alpha_3}$ for $\alpha = (\alpha_1, \alpha_2, \alpha_3) \in \mathbb{Z}^3$ with $\alpha_j \ge 0, j = 1, 2, 3$ and c > 0 is a constant independent of ε . Since $n_{\varepsilon} = \bar{n}_{\varepsilon}^i$ on Γ_{ε}^i by Lemma 5.2.10,

$$n_{\varepsilon} - (-1)^{i+1} \left(\bar{n} - \overline{\nabla_{\Gamma} g_i} \right) = (-1)^{i+1} \varphi_{\varepsilon}^i (\bar{n} - \varepsilon \bar{\tau}_{\varepsilon}^i) - (-1)^{i+1} \varepsilon \left(\bar{\tau}_{\varepsilon} - \overline{\nabla_{\Gamma} g_i} \right)$$

on Γ_{ε}^{i} . Applying (5.2.34), (5.2.35), and (5.B.32) to the right-hand side we obtain (5.2.40). Also, the inequalities (5.2.41) immediately follow from (5.2.40).

Let us prove (5.2.42). For $x \in N$ we set

$$\Phi^i_{\varepsilon}(x) := (-1)^{i+1} \left\{ \varphi^i_{\varepsilon}(x)\bar{n}(x) - \frac{\varepsilon \bar{\tau}^i_{\varepsilon}(x)}{\sqrt{1 + \varepsilon^2 |\bar{\tau}^i_{\varepsilon}(x)|^2}} \right\}.$$

Then we observe by direct calculations, (5.2.13), (5.2.15), (5.2.34), and (5.B.32) that

$$|\partial_x^{\alpha} \Phi_{\varepsilon}^i(x)| \le c\varepsilon \quad \text{for all} \quad x \in N, \ |\alpha| = 0, 1, 2.$$
(5.B.33)

We differentiate $\bar{n}_{\varepsilon}^{i}(x) = (-1)^{i+1}\bar{n}(x) + \Phi_{\varepsilon}^{i}(x), x \in N$ and apply (5.2.16) to get

$$\nabla \bar{n}^i_{\varepsilon}(x) = (-1)^i \{ I_3 - d(x) \overline{W}(x) \}^{-1} \overline{W}(x) + \nabla \Phi^i_{\varepsilon}(x), \quad x \in N.$$

Since $\bar{n}_{\varepsilon}^{i}$ is an extension of n_{ε} (on Γ_{ε}^{i}) to N, the Weingarten map of Γ_{ε}^{i} is given by $W_{\varepsilon} = -P_{\varepsilon}\nabla \bar{n}_{\varepsilon}^{i}$ on Γ_{ε} . Thus, the above equality yields that

$$W_{\varepsilon}(x) = P_{\varepsilon}(x) \left\{ (-1)^{i+1} \overline{R}^{i}_{\varepsilon}(x) \overline{W}(x) - \nabla \Phi^{i}_{\varepsilon}(x) \right\}, \quad x \in \Gamma^{i}_{\varepsilon},$$
(5.B.34)

where $R_{\varepsilon}^{i} := (I_{3} - \varepsilon g_{i}W)^{-1}$ on Γ . By this equality and (5.2.8) we observe that

$$\left|W_{\varepsilon} - (-1)^{i+1}\overline{W}\right| \leq \left|\left(P_{\varepsilon} - \overline{P}\right)\overline{R}_{\varepsilon}^{i}\overline{W}\right| + \left|\left(\overline{R}_{\varepsilon}^{i} - I_{3}\right)\overline{W}\right| + \left|P_{\varepsilon}\nabla\Phi_{\varepsilon}^{i}\right| \quad \text{on} \quad \Gamma_{\varepsilon}^{i}$$

and thus we obtain the first inequality of (5.2.42) by applying (5.2.9), (5.2.10), (5.2.41), and (5.B.33) to the right-hand side of the above inequality. The second inequality of (5.2.42) follows from the first one since H = tr[W] and $H_{\varepsilon} = tr[W_{\varepsilon}]$.

Finally, let us show the inequality (5.2.43). Based on (5.B.34) we define an extension of the Weingarten map W_{ε} (on Γ_{ε}^{i}) to N by

$$\widetilde{W}^i_{\varepsilon}(x) := \overline{P}^i_{\varepsilon}(x) \left\{ (-1)^{i+1} \overline{R}^i_{\varepsilon}(x) \overline{W}(x) - \nabla \Phi^i_{\varepsilon}(x) \right\}, \quad x \in N$$

where $P_{\varepsilon}^{i} := I_{3} - n_{\varepsilon}^{i} \otimes n_{\varepsilon}^{i}$ on Γ . For $x \in N$ let

$$E_{\varepsilon}^{i}(x) := (-1)^{i+1} \left\{ \overline{P}_{\varepsilon}^{i}(x) - \overline{P}(x) \right\} \overline{R}_{\varepsilon}^{i}(x) \overline{W}(x),$$
$$F_{\varepsilon}^{i}(x) := (-1)^{i+1} \overline{P}(x) \left\{ \overline{R}_{\varepsilon}^{i}(x) - I_{3} \right\} \overline{W}(x), \quad G_{\varepsilon}^{i}(x) := -\overline{P}_{\varepsilon}^{i}(x) \nabla \Phi_{\varepsilon}^{i}(x)$$

so that $\widetilde{W}^i_{\varepsilon} = (-1)^{i+1}\overline{W} + E^i_{\varepsilon} + F^i_{\varepsilon} + G^i_{\varepsilon}$ in N by (5.2.6). Hence by (5.2.12) we get

$$\partial_{j}\widetilde{W}_{\varepsilon}^{i} = \sum_{k=1}^{3} (-1)^{i+1} \left[\left(I_{3} - d\overline{W} \right)^{-1} \right]_{jk} \overline{\underline{D}_{k}W} + \partial_{j}E_{\varepsilon}^{i} + \partial_{j}F_{\varepsilon}^{i} + \partial_{j}G_{\varepsilon}^{i}$$
(5.B.35)

in N for j = 1, 2, 3. To estimate the derivatives of E_{ε}^i , F_{ε}^i , and G_{ε}^i , we observe by the equality $\bar{n}_{\varepsilon}^i = (-1)^{i+1}\bar{n} + \Phi_{\varepsilon}^i$ on N and the estimate (5.B.33) that

$$\left|\overline{P}^{i}_{\varepsilon} - \overline{P}\right| \leq c\varepsilon, \quad \left|\partial_{j}\overline{P}^{i}_{\varepsilon} - \partial_{j}\overline{P}\right| \leq c\varepsilon \quad \text{in} \quad N.$$

From these inequalities, (5.2.10), (5.2.13), (5.B.27), and (5.B.33) we deduce that

$$|\partial_j E^i_{\varepsilon}| \le c\varepsilon, \quad |\partial_j F^i_{\varepsilon}| \le c\varepsilon, \quad |\partial_j G^i_{\varepsilon}| \le c\varepsilon \quad \text{in} \quad N.$$

Applying (5.2.10) and the above inequalities to (5.B.35) we get

$$\left|\partial_{j}\widetilde{W}^{i}_{\varepsilon}(x) - (-1)^{i+1}\overline{\underline{D}_{j}W}(x)\right| \leq c(|d(x)| + \varepsilon), \quad x \in N.$$
(5.B.36)

Now we recall that $\widetilde{W}^i_{\varepsilon}$ is an extension of W_{ε} (on Γ^i_{ε}) to N. Hence

$$\underline{D}_{j}^{\varepsilon}W_{\varepsilon}(x) = \sum_{k=1}^{3} [P_{\varepsilon}(x)]_{jk} \partial_{k} \widetilde{W}_{\varepsilon}^{i}(x), \quad x \in \Gamma_{\varepsilon}^{i}$$

and from $P\nabla_{\Gamma} = \nabla_{\Gamma}$ it follows that

$$\left|\underline{D}_{j}^{\varepsilon}W_{\varepsilon} - \overline{\underline{D}_{j}W}\right| \leq \sum_{k=1}^{3} \left(\left| \left[P_{\varepsilon} - \overline{P} \right]_{jk} \partial_{k} \widetilde{W}_{\varepsilon}^{i} \right| + \left| \overline{P}_{jk} \left\{ \partial_{k} \widetilde{W}_{\varepsilon}^{i} - (-1)^{i+1} \overline{\underline{D}_{k}W} \right\} \right| \right)$$

on Γ_{ε} . Applying (5.2.41) and (5.B.36) with $|d| = \varepsilon |\bar{g}_i| \le c\varepsilon$ on Γ_{ε}^i to the right-hand side we conclude that (5.2.43) is valid.

Proof of Lemma 5.2.12. Since

$$\overline{P}(y + \varepsilon g_0(y)n(y)) = \overline{P}(y + \varepsilon g_1(y)n(y)) = P(y), \quad y \in \Gamma,$$

we observe that

$$\left|P_{\varepsilon}(y+\varepsilon g_{1}(y)n(y))-P_{\varepsilon}(y+\varepsilon g_{0}(y)n(y))\right| \leq \sum_{i=0,1}\left|\left[P_{\varepsilon}-\overline{P}\right](y+\varepsilon g_{i}(y)n(y))\right|.$$

Applying (5.2.41) to the right-hand side we obtain the inequality (5.2.44) for $F_{\varepsilon} = P_{\varepsilon}$. The other inequalities are proved by (5.2.41)–(5.2.43) in the same way.

Finally, let us prove the formula (5.3.15) in the proof of Lemma 5.3.3.

Proof of (5.3.15). In the proof of Lemma 5.3.3 we consider the mapping

$$\zeta(s) = \mu(s') + \varepsilon \{ (1 - s_3)g_0(\mu(s')) + s_3g_1(\mu(s')) \} n(\mu(s))$$

for $s = (s', s_3) \in (0, 1), s' \in (0, 1)^2$, where $\mu : (0, 1)^2 \to \Gamma$ is a local parametrization of Γ . Differentiating ζ in s we get (5.3.14), i.e.

$$\partial_{s_i}\zeta(s) = \{I_3 - h_{\varepsilon}(s)W(\mu(s'))\}\partial_{s_i}\mu(s') + \eta^i_{\varepsilon}(s)n(\mu(s')), \quad i = 1, 2, \\
\partial_{s_3}\zeta(s) = \varepsilon g(\mu(s'))n(\mu(s'))$$
(5.B.37)

for $s \in (0, 1)^3$, where

$$\begin{split} h_{\varepsilon}(s) &:= \varepsilon \{ (1 - s_3) g_0(\mu(s')) + s_3 g_1(\mu(s')) \}, \\ \eta_{\varepsilon}^i(s) &:= \varepsilon \partial_{s_i} \mu(s') \cdot \{ (1 - s_3) \nabla_{\Gamma} g_0(\mu(s')) + s_3 \nabla_{\Gamma} g_1(\mu(s')) \}, \quad i = 1, 2. \end{split}$$

For the sake of simplicity, we use the notation $\xi^{\sharp}(s') := \xi(\mu(s')), s' \in (0, 1)^2$ for a function ξ on Γ and suppress the arguments s and s'. Hence we refer to (5.B.37) as

$$\partial_{s_i}\zeta = (I_3 - h_\varepsilon W^{\sharp})\partial_{s_i}\mu + \eta_\varepsilon^i n^{\sharp}, \quad \partial_{s_3}\zeta = \varepsilon g^{\sharp} n^{\sharp} \quad \text{on} \quad (0,1)^3, \, i = 1, 2.$$
(5.B.38)

By (5.B.38) we observe that the gradient matrix of ζ is of the form

$$\nabla_s \zeta = \begin{pmatrix} \partial_{s_1} \zeta_1 & \partial_{s_1} \zeta_2 & \partial_{s_1} \zeta_3 \\ \partial_{s_2} \zeta_1 & \partial_{s_2} \zeta_2 & \partial_{s_2} \zeta_3 \\ \partial_{s_3} \zeta_1 & \partial_{s_3} \zeta_2 & \partial_{s_3} \zeta_3 \end{pmatrix} = \begin{pmatrix} \nabla_s \mu (I_3 - h_\varepsilon W^\sharp) + \eta_\varepsilon \otimes n^\sharp \\ \varepsilon g^\sharp (n^\sharp)^T \end{pmatrix}.$$

Here we consider $n^{\sharp} \in \mathbb{R}^3$ as a column vector and define

$$\nabla_s \mu := \begin{pmatrix} \partial_{s_1} \mu_1 & \partial_{s_1} \mu_2 & \partial_{s_1} \mu_3 \\ \partial_{s_2} \mu_1 & \partial_{s_2} \mu_2 & \partial_{s_2} \mu_3 \end{pmatrix}, \quad \eta_{\varepsilon} := \begin{pmatrix} \eta_{\varepsilon}^1 \\ \eta_{\varepsilon}^2 \end{pmatrix}.$$

Since μ is a local parametrization of Γ , the vectors $\partial_{s_1}\mu$ and $\partial_{s_2}\mu$ are tangent to Γ and thus $(\nabla_s\mu)n^{\sharp} = 0$. Also, we have $W^{\sharp}n^{\sharp} = 0$, $(\eta_{\varepsilon} \otimes n^{\sharp})n^{\sharp} = |n^{\sharp}|^2\eta_{\varepsilon} = \eta_{\varepsilon}$, and

$$(\eta_{\varepsilon} \otimes n^{\sharp})(n^{\sharp} \otimes \eta_{\varepsilon}) = |n^{\sharp}|^2 \eta_{\varepsilon} \otimes \eta_{\varepsilon} = \eta_{\varepsilon} \otimes \eta_{\varepsilon}.$$

From these equalities and the symmetry of the matrix W^{\sharp} it follows that

$$\nabla_s \zeta (\nabla_s \zeta)^T = \begin{pmatrix} \nabla_s \mu (I_3 - h_\varepsilon W^{\sharp})^2 (\nabla_s \mu)^T + \eta_\varepsilon \otimes \eta_\varepsilon & \varepsilon g^{\sharp} \eta_\varepsilon \\ \varepsilon g^{\sharp} \eta_\varepsilon^T & \varepsilon^2 (g^{\sharp})^2 \end{pmatrix}.$$

Therefore, by elementary row operations we have

$$det[\nabla_s \zeta(\nabla_s \zeta)^T] = det \begin{pmatrix} \nabla_s \mu (I_3 - h_{\varepsilon} W^{\sharp})^2 (\nabla_s \mu)^T + \eta_{\varepsilon} \otimes \eta_{\varepsilon} & \varepsilon g^{\sharp} \eta_{\varepsilon} \\ \varepsilon g^{\sharp} \eta_{\varepsilon}^T & \varepsilon^2 (g^{\sharp})^2 \end{pmatrix}$$
$$= det \begin{pmatrix} \nabla_s \mu (I_3 - h_{\varepsilon} W^{\sharp})^2 (\nabla_s \mu)^T & 0 \\ \varepsilon g^{\sharp} \eta_{\varepsilon}^T & \varepsilon^2 (g^{\sharp})^2 \end{pmatrix}$$
$$= \varepsilon^2 (g^{\sharp})^2 det[\nabla_s \mu (I_3 - h_{\varepsilon} W^{\sharp})^2 (\nabla_s \mu)^T]$$
$$= \varepsilon^2 (g^{\sharp})^2 J(\mu, h_{\varepsilon})^2 det \theta,$$

where $\theta := \nabla_s \mu (\nabla_s \mu)^T$ and the last equality follows from the same calculations as in (5.B.23). Since det $\nabla_s \zeta = \sqrt{\det[\nabla_s \zeta(\nabla_s \zeta)^T]}$, we conclude by the above equality that the formula (5.3.15) is valid.

5.C Riemannian connection on a surface

In this appendix we introduce the notion of the Riemannian (or Levi-Civita) connection on a surface in \mathbb{R}^3 and give several formulas for the covariant derivatives.

Let Γ be a two-dimensional closed, connected, oriented, and sufficiently smooth (at least of class C^3) surface in \mathbb{R}^3 . We use the same notations on the differential operators and the surface quantities on Γ as in Section 5.2.1. For $X \in C^1(\Gamma, T\Gamma)$ and $Y \in C(\Gamma, T\Gamma)$ we define the covariant derivative of X along Y as

$$\overline{\nabla}_Y X := P\left\{ (Y \cdot \nabla) \widetilde{X} \right\} \quad \text{on} \quad \Gamma.$$
(5.C.1)

Here \widetilde{X} is a C^1 -extension of X to an open neighborhood of Γ with $\widetilde{X}|_{\Gamma} = X$. Note that, since Y is tangential on Γ , we have $(Y \cdot \nabla)\widetilde{X} = (Y \cdot \nabla_{\Gamma})X$ and thus the value of $\overline{\nabla}_Y X$ does not depend on the choice of an extension of X. The directional derivative $(Y \cdot \nabla_{\Gamma})X$ is expressed in terms of the covariant derivative and the Weingarten map.

Lemma 5.C.1. For $X \in C^1(\Gamma, T\Gamma)$ and $Y \in C(\Gamma, T\Gamma)$ we have

$$(Y \cdot \nabla)\widetilde{X} = (Y \cdot \nabla_{\Gamma})X = \overline{\nabla}_{Y}X + (WX \cdot Y)n \quad on \quad \Gamma,$$
(5.C.2)

where \widetilde{X} is any C^1 -extension of X to an open neighborhood of Γ satisfying $\widetilde{X}|_{\Gamma} = X$.

Proof. Since $X \cdot n = 0$ and $-\nabla_{\Gamma} n = W$ on Γ ,

$$(Y \cdot \nabla_{\Gamma})X \cdot n = Y \cdot \nabla_{\Gamma}(X \cdot n) - X \cdot (Y \cdot \nabla_{\Gamma})n = X \cdot (-\nabla_{\Gamma}n)^{T}Y = X \cdot W^{T}Y = WX \cdot Y$$

on Γ . Combining this with (5.C.1) we obtain (5.C.2).

The formula (5.C.2) is called the Gauss formula (see e.g. [9, Section 4.2] and [33, Section VII.3]). Let us prove fundamental properties of $\overline{\nabla}$.

Lemma 5.C.2. The following equalities hold:

• For $X \in C^1(\Gamma, T\Gamma)$, $Y, Z \in C(\Gamma, T\Gamma)$, and $\eta, \xi \in C(\Gamma)$,

$$\overline{\nabla}_{\eta Y+\xi Z}X = \eta \overline{\nabla}_Y X + \xi \overline{\nabla}_Z X \quad on \quad \Gamma.$$
(5.C.3)

• For $X \in C^1(\Gamma, T\Gamma)$, $Y \in C(\Gamma, T\Gamma)$, and $\eta \in C^1(\Gamma)$,

$$\overline{\nabla}_Y(\eta X) = (Y \cdot \nabla_\Gamma \eta) X + \eta \overline{\nabla}_Y X \quad on \quad \Gamma.$$
(5.C.4)

• For $X, Y \in C^1(\Gamma, T\Gamma)$ and $Z \in C(\Gamma, T\Gamma)$,

$$Z \cdot \nabla_{\Gamma} (X \cdot Y) = \overline{\nabla}_Z X \cdot Y + X \cdot \overline{\nabla}_Z Y \quad on \quad \Gamma.$$
(5.C.5)

• For $X, Y \in C^1(\Gamma, T\Gamma)$ and $\eta \in C^2(\Gamma)$,

$$X \cdot \nabla_{\Gamma} (Y \cdot \nabla_{\Gamma} \eta) - Y \cdot \nabla_{\Gamma} (X \cdot \nabla_{\Gamma} \eta) = \left(\overline{\nabla}_X Y - \overline{\nabla}_Y X\right) \cdot \nabla_{\Gamma} \eta.$$
(5.C.6)

Proof. The identities (5.C.3) and (5.C.4) immediately follow from the definition (5.C.1) of the covariant derivative. Also, applying (5.C.2) and $X \cdot n = Y \cdot n = 0$ on Γ to the right-hand side of

$$Z \cdot \nabla_{\Gamma} (X \cdot Y) = (Z \cdot \nabla_{\Gamma}) X \cdot Y + X \cdot (Z \cdot \nabla_{\Gamma}) Y$$

we get (5.C.5). To prove (5.C.6) we use the formula (see e.g. [43, Lemma 2.2])

$$\underline{D}_i \underline{D}_j \eta - \underline{D}_j \underline{D}_i \eta = [W \nabla_{\Gamma} \eta]_i n_j - [W \nabla_{\Gamma} \eta]_j n_i \quad \text{on} \quad \Gamma, \, i, j = 1, 2, 3.$$
(5.C.7)

The left-hand side of (5.C.6) is of the form

$$\sum_{i,j=1}^{3} \{X_i \underline{D}_i (Y_j \underline{D}_j \eta) - Y_i \underline{D}_i (X_j \underline{D}_j \eta)\}$$
$$= \sum_{i,j=1}^{3} (X_i \underline{D}_i Y_j - Y_i \underline{D}_i X_j) \underline{D}_j \eta + \sum_{i,j=1}^{3} (X_i Y_j - X_j Y_i) \underline{D}_i \underline{D}_j \eta.$$

By (5.C.2) and $\nabla_{\Gamma} \eta \cdot n = 0$ on Γ , we have

$$\sum_{i,j=1}^{3} (X_i \underline{D}_i Y_j - Y_i \underline{D}_i X_j) \underline{D}_j \eta = \{ (X \cdot \nabla_{\Gamma}) Y - (X \cdot \nabla_{\Gamma}) X \} \cdot \nabla_{\Gamma} \eta$$
$$= \left(\overline{\nabla}_X Y - \overline{\nabla}_Y X \right) \cdot \nabla_{\Gamma} \eta.$$

Also, using (5.C.7) and noting that $X \cdot n = Y \cdot n = 0$ on Γ we observe that

$$\sum_{i,j=1}^{3} (X_i Y_j - X_j Y_i) \underline{D}_i \underline{D}_j \eta = \sum_{i,j=1}^{3} X_i Y_j (\underline{D}_i \underline{D}_j \eta - \underline{D}_j \underline{D}_i \eta)$$
$$= \sum_{i,j=1}^{3} X_i Y_j ([W \nabla \eta]_i n_j - [W \nabla_{\Gamma} \eta]_j n_i)$$
$$= (X \cdot W \nabla_{\Gamma} \eta) (Y \cdot n) - (X \cdot n) (Y \cdot W \nabla_{\Gamma} \eta) = 0.$$

Combining the above three equalities we obtain (5.C.6).

By Lemma 5.C.2 we observe that the assignment $\overline{\nabla} \colon (X,Y) \mapsto \overline{\nabla}_Y X$ defines the Riemannian (or Levi-Civita) connection on Γ (see e.g. [9, 30, 50] for the definition of the Riemannian connection). Note that the formula (5.C.6) represents the torsion-free condition $[X,Y] = \overline{\nabla}_X Y - \overline{\nabla}_Y X$ for $X, Y \in C^1(\Gamma, T\Gamma)$, where [X,Y] = XY - YX is the Lie bracket of X and Y (see e.g. [63]).

Let U be a relatively open subset of Γ and τ_1 and τ_2 be C^1 tangential vector fields defined on U such that the pair $\{\tau_1(y), \tau_2(y)\}$ is an orthonormal basis of the tangent plane of Γ at each $y \in U$. (Since Γ is at least of class C^3 , we can take such vector fields for sufficiently small U.) We call the pair $\{\tau_1, \tau_2\}$ a local orthonormal frame of the tangent plane of Γ on U. Note that

$$H = \operatorname{tr}[W] = \sum_{i=1,2} W \tau_i \cdot \tau_i \quad \text{on} \quad U$$
(5.C.8)

since $\{\tau_1, \tau_2, n\}$ is an orthonormal basis of \mathbb{R}^3 and Wn = 0. Let us express several quantities related to the tangential gradient matrix of tangential vector fields on Γ in terms of the covariant derivatives and the local orthonormal frame.

Lemma 5.C.3. Let $\{\tau_1, \tau_2\}$ be a local orthonormal frame of the tangent plane of Γ on a relatively open subset U of Γ . For $X, Y \in C^1(\Gamma, T\Gamma)$ we have

$$\operatorname{div}_{\Gamma} X = \sum_{i=1,2} \overline{\nabla}_i X \cdot \tau_i, \qquad (5.C.9)$$

$$\nabla_{\Gamma} X : W = \sum_{i=1,2} \overline{\nabla}_i X \cdot W \tau_i = \sum_{i=1,2} W \overline{\nabla}_i X \cdot \tau_i, \qquad (5.C.10)$$

$$\nabla_{\Gamma} X : \nabla_{\Gamma} Y = \sum_{i=1,2} \overline{\nabla}_i X \cdot \overline{\nabla}_i Y + W X \cdot W Y, \qquad (5.C.11)$$

$$\nabla_{\Gamma} X : (\nabla_{\Gamma} Y) P = \sum_{i=1,2} \overline{\nabla}_i X \cdot \overline{\nabla}_i Y, \qquad (5.C.12)$$

$$W\nabla_{\Gamma}X: (\nabla_{\Gamma}Y)P = \sum_{i=1,2} \overline{\nabla}_{W\tau_i}X \cdot \overline{\nabla}_iY$$
(5.C.13)

on U, where $\overline{\nabla}_i := \overline{\nabla}_{\tau_i}$ for i = 1, 2.

Proof. Throughout the proof we carry out calculations on U. First note that

$$(\nabla_{\Gamma} X)^{T} \tau_{i} = (\tau_{i} \cdot \nabla_{\Gamma}) X = \overline{\nabla}_{i} X + (WX \cdot \tau_{i}) n, \quad i = 1, 2,$$

$$(\nabla_{\Gamma} X)^{T} n = (n \cdot \nabla_{\Gamma}) X = 0$$
(5.C.14)

by (5.C.2) and $n \cdot \nabla_{\Gamma} X_j = 0$ for j = 1, 2, 3. Since $\{\tau_1, \tau_2, n\}$ forms an orthonormal basis of \mathbb{R}^3 , we have

$$\operatorname{div}_{\Gamma} X = \operatorname{tr}[\nabla_{\Gamma} X] = \sum_{i=1,2} (\nabla_{\Gamma} X)^T \tau_i \cdot \tau_i + (\nabla_{\Gamma} X)^T n \cdot n$$

The equality (5.C.9) follows from the above equality and (5.C.14). We also get (5.C.10) by using (5.C.14), $W^T = W$, and Wn = 0 to the right-hand side of

$$\nabla_{\Gamma} X : W = (\nabla_{\Gamma} X)^T : W^T = \sum_{i=1,2} (\nabla_{\Gamma} X)^T \tau_i \cdot W^T \tau_i + (\nabla_{\Gamma} X)^T n \cdot W^T n$$

Let us prove (5.C.11). By (5.C.2) and $\overline{\nabla}_i X \cdot n = \overline{\nabla}_i Y \cdot n = 0$,

$$\nabla_{\Gamma}X : \nabla_{\Gamma}Y = (\nabla_{\Gamma}X)^{T} : (\nabla_{\Gamma}Y)^{T} = \sum_{i=1,2} (\nabla_{\Gamma}X)^{T} \tau_{i} \cdot (\nabla_{\Gamma}Y)^{T} \tau_{i} + (\nabla_{\Gamma}X)^{T} n \cdot (\nabla_{\Gamma}Y)^{T} n$$
$$= \sum_{i=1,2} \left\{ \overline{\nabla}_{i}X + (WX \cdot \tau_{i})n \right\} \cdot \left\{ \overline{\nabla}_{i}Y + (WY \cdot \tau_{i})n \right\}$$
$$= \sum_{i=1,2} \overline{\nabla}_{i}X \cdot \overline{\nabla}_{i}Y + \sum_{i=1,2} (WX \cdot \tau_{i})(WY \cdot \tau_{i}).$$

Here the second term on the last line is equal to $WX \cdot WY$, since WX and WY are tangential vector fields on Γ and $\{\tau_1, \tau_2\}$ is an orthonormal basis of the tangent plane of Γ . Hence (5.C.11) is valid. Also, noting that $P^T = P$, $W^T = W$, and

$$P(\nabla_{\Gamma}Y)^{T}\tau_{i} = P\left\{\overline{\nabla}_{i}Y + (WY \cdot \tau_{i})n\right\} = \overline{\nabla}_{i}Y,$$
$$(\nabla_{\Gamma}X)^{T}W\tau_{i} = (W\tau_{i} \cdot \nabla_{\Gamma})X = \overline{\nabla}_{W\tau_{i}}X + (WX \cdot W\tau_{i})n$$

by (5.C.2), Pn = 0, and the fact that $\overline{\nabla}_i Y$ is tangential on Γ , we can prove (5.C.12) and (5.C.13) by the same calculations as above.

Next we give an integration by parts formula for integrals over Γ of the covariant derivatives along vector fields of a local orthonormal frame.

Lemma 5.C.4. Let $\{\tau_1, \tau_2\}$ be a local orthonormal frame of the tangent plane of Γ on a relatively open subset U of Γ and $\overline{\nabla}_i := \overline{\nabla}_{\tau_i}$ for i = 1, 2. Suppose that $X \in C^2(\Gamma, T\Gamma)$ and $Y \in C^1(\Gamma, T\Gamma)$ are compactly supported in U. Then we have

$$\sum_{i=1,2} \int_{\Gamma} \left(\overline{\nabla}_i \overline{\nabla}_i X - \overline{\nabla}_{\overline{\nabla}_i \tau_i} X \right) \cdot Y \, d\mathcal{H}^2 = -\sum_{i=1,2} \int_{\Gamma} \overline{\nabla}_i X \cdot \overline{\nabla}_i Y \, d\mathcal{H}^2. \tag{5.C.15}$$

Proof. The proof is basically the same as that of [50, Proposition 34]. We define a tangential vector field Z on Γ by

$$Z := \begin{cases} \sum_{i=1,2} \left(\overline{\nabla}_i X \cdot Y \right) \tau_i & \text{on } U, \\ 0 & \text{on } \Gamma \setminus U. \end{cases}$$

Since $X \in C^2(\Gamma, T\Gamma)$ and $Y \in C^1(\Gamma, T\Gamma)$ are compactly supported in U, we have $Z \in C^1(\Gamma, T\Gamma)$. Moreover, since $\{\tau_1, \tau_2\}$ forms an orthonormal basis of the tangent plane of Γ , we see by (5.C.3) that $Z \cdot V = \overline{\nabla}_V X \cdot Y$ on Γ for all $V \in C(\Gamma, T\Gamma)$. From this fact and (5.C.5) we deduce that

$$\overline{\nabla}_i Z \cdot \tau_i = \tau_i \cdot \nabla_{\Gamma} (Z \cdot \tau_i) - Z \cdot \overline{\nabla}_i \tau_i = \tau_i \cdot \nabla_{\Gamma} \left(\overline{\nabla}_i X \cdot Y \right) - \overline{\nabla}_{\overline{\nabla}_i \tau_i} X \cdot Y$$

on U for i = 1, 2. Applying (5.C.5) again to the first term on the right-hand side we get

$$\overline{\nabla}_i Z \cdot \tau_i = \left(\overline{\nabla}_i \overline{\nabla}_i X - \overline{\nabla}_{\overline{\nabla}_i \tau_i} X\right) \cdot Y + \overline{\nabla}_i X \cdot \overline{\nabla}_i Y \quad \text{on} \quad U.$$

By this equality and (5.C.9) we see that

$$\operatorname{div}_{\Gamma} Z = \sum_{i=1,2} \left\{ \left(\overline{\nabla}_i \overline{\nabla}_i X - \overline{\nabla}_{\overline{\nabla}_i \tau_i} X \right) \cdot Y + \overline{\nabla}_i X \cdot \overline{\nabla}_i Y \right\} \quad \text{on} \quad U.$$
(5.C.16)

Since X, Y, and Z are supported in U, we may assume that (5.C.16) holds on the whole surface Γ . Thus, integrating both sides of (5.C.16) over Γ and noting that the integral of the surface divergence of Z over Γ vanishes by the Stokes theorem (note that Z is tangential and Γ has no boundary), we conclude that the formula (5.C.15) is valid.

The formula (5.C.15) gives a relation between two Laplacians acting on tangential vector fields on Γ . For $X \in C^2(\Gamma, T\Gamma)$, $Y \in C^1(\Gamma, T\Gamma)$, and $Z \in C(\Gamma, T\Gamma)$, the second covariant derivative is defined as

$$\overline{\nabla}_{Z,Y}^2 X := \overline{\nabla}_Z \overline{\nabla}_Y X - \overline{\nabla}_{\overline{\nabla}_Z Y} X.$$

The trace of the second covariant derivative is called the connection Laplacian, i.e.

$$\mathrm{tr}\overline{\nabla}^2 X := \sum_{i=1,2} \overline{\nabla}_{\tau_i,\tau_i}^2 X = \sum_{i=1,2} \left(\overline{\nabla}_i \overline{\nabla}_i X - \overline{\nabla}_{\overline{\nabla}_i \tau_i} X \right), \quad X \in C^2(\Gamma, T\Gamma).$$

Here $\{\tau_1, \tau_2\}$ is a local orthonormal frame and $\overline{\nabla}_i = \overline{\nabla}_{\tau_i}$ for i = 1, 2. On the other hand, the Bochner Laplacian $\Delta_B X$ of $X \in C^2(\Gamma, T\Gamma)$ is defined as a tangential vector field on Γ satisfying

$$\int_{\Gamma} \Delta_B X \cdot Y \, d\mathcal{H}^2 := -\sum_{i=1,2} \int_{\Gamma} \overline{\nabla}_i X \cdot \overline{\nabla}_i Y \, d\mathcal{H}^2 \tag{5.C.17}$$

for all $Y \in C^1(\Gamma, T\Gamma)$, where we assume that X and Y are appropriately localized by a partition of unity of Γ . (Note that there are other definitions of the connection and Bochner Laplacians in which one takes the opposite sign.) Then by (5.C.15) we observe that these two Laplacians agree, i.e.

$$\operatorname{tr}\overline{\nabla}^2 X = \Delta_B X, \quad X \in C^2(\Gamma, T\Gamma).$$

When the surface Γ is embedded in \mathbb{R}^3 , we can also consider the Laplace–Beltrami operator acting on each component of a vector field $X = (X_1, X_2, X_3)$ on Γ :

$$\Delta_{\Gamma} X = (\Delta_{\Gamma} X_1, \Delta_{\Gamma} X_2, \Delta_{\Gamma} X_3), \quad \Delta_{\Gamma} = \sum_{i=1}^3 \underline{D}_i^2.$$

Let us give a relation between the two Laplacians Δ_B and Δ_{Γ} .

Lemma 5.C.5. For $X \in C^2(\Gamma, T\Gamma)$ we have

$$\Delta_B X = P \Delta_\Gamma X + W^2 X \quad on \quad \Gamma. \tag{5.C.18}$$

Proof. The relation (5.C.18) is proved in [43, Appendix B]. Here we give another proof of it. By a localization argument with a partition of unity of Γ , it is sufficient to show (5.C.18) on a relatively open subset U of Γ on which we can take a local orthonormal frame $\{\tau_1, \tau_2\}$. Let $Y \in C^1(\Gamma, T\Gamma)$ be an arbitrary vector field compactly supported in U. By (5.C.11) and (5.C.17) we have

$$\int_{\Gamma} \Delta_B X \cdot Y \, d\mathcal{H}^2 = -\sum_{i=1,2} \int_{\Gamma} \overline{\nabla}_i X \cdot \overline{\nabla}_i Y \, d\mathcal{H}^2$$

$$= -\int_{\Gamma} \nabla_{\Gamma} X : \nabla_{\Gamma} Y \, d\mathcal{H}^2 + \int_{\Gamma} W X \cdot W Y \, d\mathcal{H}^2.$$
(5.C.19)

To the first integral on the last line we use (5.2.18) to get

$$\begin{split} \int_{\Gamma} \nabla_{\Gamma} X : \nabla_{\Gamma} Y \, d \, \mathcal{H}^2 &= \sum_{i,j=1}^3 \int_{\Gamma} (\underline{D}_i X_j) (\underline{D}_i Y_j) \, d\mathcal{H}^2 \\ &= -\sum_{i,j=1}^3 \int_{\Gamma} \{ (\underline{D}_i^2 X_j) Y_j + (\underline{D}_i X_j) Y_j H n_i \} \, d\mathcal{H}^2 \\ &= -\int_{\Gamma} \Delta_{\Gamma} X \cdot Y \, d\mathcal{H}^2 - \int_{\Gamma} \{ (n \cdot \nabla_{\Gamma}) X \cdot Y \} H \, d\mathcal{H}^2 \end{split}$$

Noting that the second integral on the last line vanishes by $(n \cdot \nabla_{\Gamma})X = 0$, we apply this equality and $W^T = W$ to (5.C.19) to obtain

$$\int_{\Gamma} \Delta_B X \cdot Y \, d\mathcal{H}^2 = \int_{\Gamma} (\Delta_{\Gamma} X + W^2 X) \cdot Y \, d\mathcal{H}^2.$$

Now we set Y := Pv in this equality with $v \in C^1(\Gamma)^3$ compactly supported in U. Then since $\Delta_B X$ and $W^2 X$ are tangential on Γ (note that PW = W), we have

$$\int_{\Gamma} \Delta_B X \cdot v \, d\mathcal{H}^2 = \int_{\Gamma} (P \Delta_{\Gamma} X + W^2 X) \cdot v \, d\mathcal{H}^2$$

for all $v \in C^1(\Gamma)^3$ compactly supported in U. Hence by the fundamental lemma of calculus of variations we conclude that (5.C.18) holds on U.

Note that the normal component of $\Delta_{\Gamma} X$ does not vanish in general even if X is tangential on Γ . Indeed, using $X \cdot n = 0$ and $-\nabla_{\Gamma} n = W$ we observe by direct calculations that

 $\Delta_{\Gamma} X \cdot n = \operatorname{div}_{\Gamma}(WX) + W : \nabla_{\Gamma} X = \operatorname{div}_{\Gamma} W \cdot X + 2W : \nabla_{\Gamma} X \quad \text{on} \quad \Gamma.$

When Γ is a flat domain in \mathbb{R}^2 , we have $W = -\nabla_{\Gamma} n = 0$ by n = (0, 0, 1). Hence the normal component of $\Delta_{\Gamma} X$ vanishes and $\Delta_B X = \Delta_{\Gamma} X$ reduces to the usual Laplacian on \mathbb{R}^2 acting on each component of $X = (X_1, X_2)$.

Remark 5.C.6. By Lemma 5.2.6 the function space $C^2(\Gamma, T\Gamma)$ is dense in $H^m(\Gamma, T\Gamma)$, m = 0, 1, 2. Hence the formulas given in this appendix are also valid (a.e. on Γ) if we replace $C^m(\Gamma, T\Gamma)$, m = 0, 1, 2 with $H^m(\Gamma, T\Gamma)$.

References

- R. A. Adams and J. J. F. Fournier, *Sobolev spaces*, 2nd ed., Pure and Applied Mathematics (Amsterdam), vol. 140, Elsevier/Academic Press, Amsterdam, 2003.
- [2] R. Aris, Vectors, tensors, and the basic equations of fluid mechanics, Dover Publications, 1989.
- [3] T. Aubin, Some nonlinear problems in Riemannian geometry, Springer Monographs in Mathematics, Springer-Verlag, Berlin, 1998.
- [4] J. W. Barrett, H. Garcke, and R. Nürnberg, Stable numerical approximation of two-phase flow with a Boussinesq-Scriven surface fluid, Commun. Math. Sci. 13 (2015), no. 7, 1829–1874.
- [5] H. Beirão Da Veiga, Regularity for Stokes and generalized Stokes systems under nonhomogeneous slip-type boundary conditions, Adv. Differential Equations 9 (2004), no. 9-10, 1079–1114.
- [6] D. Bothe and J. Prüss, On the two-phase Navier-Stokes equations with Boussinesq-Scriven surface fluid, J. Math. Fluid Mech. 12 (2010), no. 1, 133–150.
- [7] M. J. Boussinesq, Sur l'existence d'une viscosité superficielle, dans la mince couche de transition séparant un liquide d'un autre fluide contigu, Ann. Chim. Phys. 29 (1913), 349–357.
- [8] F. Boyer and P. Fabrie, Mathematical tools for the study of the incompressible Navier-Stokes equations and related models, Applied Mathematical Sciences, vol. 183, Springer, New York, 2013.
- [9] B.-Y. Chen, Total mean curvature and submanifolds of finite type, 2nd ed., Series in Pure Mathematics, vol. 27, World Scientific Publishing Co. Pte. Ltd., Hackensack, NJ, 2015. With a foreword by Leopold Verstraelen.
- [10] P. Constantin and C. Foias, Navier-Stokes equations, Chicago Lectures in Mathematics, University of Chicago Press, Chicago, IL, 1988.
- [11] G. Dziuk and C. M. Elliott, Finite element methods for surface PDEs, Acta Numer. 22 (2013), 289–396.
- [12] D. G. Ebin and J. Marsden, Groups of diffeomorphisms and the motion of an incompressible fluid., Ann. of Math. (2) 92 (1970), 102–163.
- [13] L. C. Evans, *Partial differential equations*, 2nd ed., Graduate Studies in Mathematics, vol. 19, American Mathematical Society, Providence, RI, 2010.
- [14] G. P. Galdi, An introduction to the mathematical theory of the Navier-Stokes equations, 2nd ed., Springer Monographs in Mathematics, Springer, New York, 2011. Steady-state problems.

- [15] D. Gilbarg and N. S. Trudinger, *Elliptic partial differential equations of second order*, Classics in Mathematics, Springer-Verlag, Berlin, 2001. Reprint of the 1998 edition.
- [16] J. K. Hale and G. Raugel, A damped hyperbolic equation on thin domains, Trans. Amer. Math. Soc. 329 (1992), no. 1, 185–219.
- [17] J. K. Hale and G. Raugel, Reaction-diffusion equation on thin domains, J. Math. Pures Appl. (9) 71 (1992), no. 1, 33–95.
- [18] M. Higaki, Navier wall law for nonstationary viscous incompressible flows, J. Differential Equations 260 (2016), no. 10, 7358–7396.
- [19] L. T. Hoang, A basic inequality for the Stokes operator related to the Navier boundary condition, J. Differential Equations 245 (2008), no. 9, 2585–2594.
- [20] L. T. Hoang, Incompressible fluids in thin domains with Navier friction boundary conditions (I), J. Math. Fluid Mech. 12 (2010), no. 3, 435–472.
- [21] L. T. Hoang and G. R. Sell, Navier-Stokes equations with Navier boundary conditions for an oceanic model, J. Dynam. Differential Equations 22 (2010), no. 3, 563–616.
- [22] L. T. Hoang, Incompressible fluids in thin domains with Navier friction boundary conditions (II), J. Math. Fluid Mech. 15 (2013), no. 2, 361–395.
- [23] C. Hu, Navier-Stokes equations in 3D thin domains with Navier friction boundary condition, J. Differential Equations 236 (2007), no. 1, 133–163.
- [24] D. Iftimie, The 3D Navier-Stokes equations seen as a perturbation of the 2D Navier-Stokes equations, Bull. Soc. Math. France 127 (1999), no. 4, 473–517 (English, with English and French summaries).
- [25] D. Iftimie and G. Raugel, Some results on the Navier-Stokes equations in thin 3D domains, J. Differential Equations 169 (2001), no. 2, 281–331. Special issue in celebration of Jack K. Hale's 70th birthday, Part 4 (Atlanta, GA/Lisbon, 1998).
- [26] D. Iftimie, G. Raugel, and G. R. Sell, Navier-Stokes equations in thin 3D domains with Navier boundary conditions, Indiana Univ. Math. J. 56 (2007), no. 3, 1083–1156.
- [27] W. Jäger and A. Mikelić, On the roughness-induced effective boundary conditions for an incompressible viscous flow, J. Differential Equations 170 (2001), no. 1, 96–122.
- [28] T. Jankuhn, M. A. Olshanskii, and A. Reusken, Incompressible fluid problems on embedded surfaces: Modeling and variational formulations, arXiv:1702.02989 [math-ph].
- [29] S. Jimbo and K. Kurata, Asymptotic behavior of eigenvalues of the Laplacian on a thin domain under the mixed boundary condition, Indiana Univ. Math. J. 65 (2016), no. 3, 867–898.
- [30] J. Jost, Riemannian geometry and geometric analysis, 6th ed., Universitext, Springer, Heidelberg, 2011.
- [31] H. Koba, C. Liu, and Y. Giga, Energetic variational approaches for incompressible fluid systems on an evolving surface, Quart. Appl. Math. 75 (2017), no. 2, 359–389.
- [32] H. Koba, C. Liu, and Y. Giga, Errata to "Energetic variational approaches for incompressible fluid systems on an evolving surface", Quart. Appl. Math. 76 (2018), no. 1, 147–152.
- [33] S. Kobayashi and K. Nomizu, Foundations of differential geometry. Vol. II, Wiley Classics Library, John Wiley & Sons, Inc., New York, 1996. Reprint of the 1969 original; A Wiley-Interscience Publication.
- [34] D. Krejčiřík, Spectrum of the Laplacian in narrow tubular neighbourhoods of hypersurfaces with combined Dirichlet and Neumann boundary conditions, Math. Bohem. 139 (2014), no. 2, 185–193.
- [35] O. A. Ladyzhenskaya, The mathematical theory of viscous incompressible flow, Second English edition, revised and enlarged. Translated from the Russian by Richard A. Silverman and John Chu. Mathematics and its Applications, Vol. 2, Gordon and Breach, Science Publishers, New York-London-Paris, 1969.
- [36] M. Lewicka and S. Müller, The uniform Korn-Poincaré inequality in thin domains, Ann. Inst. H. Poincaré Anal. Non Linéaire 28 (2011), no. 3, 443–469 (English, with English and French summaries).
- [37] J.-L. Lions and E. Magenes, Non-homogeneous boundary value problems and applications. Vol. I, Springer-Verlag, New York-Heidelberg, 1972. Translated from the French by P. Kenneth; Die Grundlehren der mathematischen Wissenschaften, Band 181.
- [38] J.-L. Lions, R. Temam, and S. H. Wang, New formulations of the primitive equations of atmosphere and applications, Nonlinearity 5 (1992), no. 2, 237–288.
- [39] J.-L. Lions, R. Temam, and S. H. Wang, On the equations of the large-scale ocean, Nonlinearity 5 (1992), no. 5, 1007–1053.

- [40] J.-L. Lions, R. Temam, and S. H. Wang, Mathematical theory for the coupled atmosphere-ocean models. (CAO III), J. Math. Pures Appl. (9) 74 (1995), no. 2, 105–163.
- [41] M. Mitrea and M. Taylor, Navier-Stokes equations on Lipschitz domains in Riemannian manifolds, Math. Ann. 321 (2001), no. 4, 955–987.
- [42] T.-H. Miura, Zero width limit of the heat equation on moving thin domains, Interfaces Free Bound. 19 (2017), no. 1, 31–77.
- [43] T.-H. Miura, On singular limit equations for incompressible fluids in moving thin domains, Quart. Appl. Math. 76 (2018), 215–251.
- [44] I. Moise, R. Temam, and M. Ziane, Asymptotic analysis of the Navier-Stokes equations in thin domains, Topol. Methods Nonlinear Anal. 10 (1997), no. 2, 249–282. Dedicated to Olga Ladyzhenskaya.
- [45] S. Montgomery-Smith, Global regularity of the Navier-Stokes equation on thin three-dimensional domains with periodic boundary conditions, Electron. J. Differential Equations (1999), No. 11, 19.
- [46] T. Nagasawa, Construction of weak solutions of the Navier-Stokes equations on Riemannian manifold by minimizing variational functionals, Adv. Math. Sci. Appl. 9 (1999), no. 1, 51–71.
- [47] C. L. M. H. Navier, Mémoire sur les lois du mouvement des fluides, Mem. Acad. R. Sci. Inst. France 6 (1823), 389–440.
- [48] J. Nečas, Direct methods in the theory of elliptic equations, Springer Monographs in Mathematics, Springer, Heidelberg, 2012. Translated from the 1967 French original by Gerard Tronel and Alois Kufner; Editorial coordination and preface by Šárka Nečasová and a contribution by Christian G. Simader.
- [49] I. Nitschke, A. Voigt, and J. Wensch, A finite element approach to incompressible two-phase flow on manifolds, J. Fluid Mech. 708 (2012), 418–438.
- [50] P. Petersen, Riemannian geometry, 2nd ed., Graduate Texts in Mathematics, vol. 171, Springer, New York, 2006.
- [51] M. Prizzi, M. Rinaldi, and K. P. Rybakowski, Curved thin domains and parabolic equations, Studia Math. 151 (2002), no. 2, 109–140.
- [52] M. Prizzi and K. P. Rybakowski, On inertial manifolds for reaction-diffusion equations on genuinely high-dimensional thin domains, Studia Math. 154 (2003), no. 3, 253–275.
- [53] G. Raugel and G. R. Sell, Navier-Stokes equations on thin 3D domains. I. Global attractors and global regularity of solutions, J. Amer. Math. Soc. 6 (1993), no. 3, 503–568.
- [54] G. Raugel, Dynamics of partial differential equations on thin domains, Dynamical systems (Montecatini Terme, 1994), Lecture Notes in Math., vol. 1609, Springer, Berlin, 1995, pp. 208–315.
- [55] A. Reusken, Stream function formulation of surface Stokes equations, preprint, IGPM report 478 (2018).
- [56] W. Rudin, Functional analysis, 2nd ed., International Series in Pure and Applied Mathematics, McGraw-Hill, Inc., New York, 1991.
- [57] M. Schatzman, On the eigenvalues of the Laplace operator on a thin set with Neumann boundary conditions, Appl. Anal. 61 (1996), no. 3-4, 293–306.
- [58] L. E. Scriven, Dynamics of a fluid interface Equation of motion for Newtonian surface fluids, Chemical Engineering Science 12 (1960), no. 2, 98–108.
- [59] G. R. Sell and Y. You, Dynamics of evolutionary equations, Applied Mathematical Sciences, vol. 143, Springer-Verlag, New York, 2002.
- [60] V. A. Solonnikov and V. E. Ščadilov, On a boundary value problem for a stationary system of Navier-Stokes equations, Proc. Steklov Inst. Math. 125 (1973), 186–199.
- [61] H. Sohr, The Navier-Stokes equations, Modern Birkhäuser Classics, Birkhäuser/Springer Basel AG, Basel, 2001. An elementary functional analytic approach; [2013 reprint of the 2001 original] [MR1928881].
- [62] M. E. Taylor, Analysis on Morrey spaces and applications to Navier-Stokes and other evolution equations, Comm. Partial Differential Equations 17 (1992), no. 9-10, 1407–1456.
- [63] M. E. Taylor, Partial differential equations I. Basic theory, 2nd ed., Applied Mathematical Sciences, vol. 115, Springer, New York, 2011.
- [64] R. Temam, Navier-Stokes equations, Revised edition, Studies in Mathematics and its Applications, vol. 2, North-Holland Publishing Co., Amsterdam-New York, 1979. Theory and numerical analysis; With an appendix by F. Thomasset.

- [65] R. Temam and M. Ziane, Navier-Stokes equations in three-dimensional thin domains with various boundary conditions, Adv. Differential Equations 1 (1996), no. 4, 499–546.
- [66] R. Temam and M. Ziane, Navier-Stokes equations in thin spherical domains, Optimization methods in partial differential equations (South Hadley, MA, 1996), Contemp. Math., vol. 209, Amer. Math. Soc., Providence, RI, 1997, pp. 281–314.

Chapter 6

Hamilton–Jacobi equations on an evolving surface

6.1 Introduction

In this chapter we are concerned with the existence, uniqueness and numerical approximation of Hamilton–Jacobi equations on moving hypersurfaces. Let $\Gamma(t)$, $t \in [0, T]$ be a family of smooth, closed, connected and oriented hypersurfaces in \mathbb{R}^3 and $S_T := \bigcup_{t \in (0,T)} \Gamma(t) \times \{t\}$. We consider the following Hamilton–Jacobi equation on the evolving surfaces $\Gamma(t)$

$$\partial^{\bullet} u + H(x, t, \nabla_{\Gamma} u) = 0 \quad \text{on} \quad S_T.$$
 (6.1.1)

In the above, $\partial^{\bullet} u = u_t + v_{\Gamma} \cdot \nabla u$ denotes the material derivative, v_{Γ} denotes the velocity of a parametrisation of $\Gamma(t)$, and $\nabla_{\Gamma} u = (I_3 - \nu \otimes \nu) \nabla u$ the tangential gradient of u, where ν is a unit normal field of $\Gamma(t)$ respectively. The precise definitions and assumptions on $H: S_T \times \mathbb{R}^3 \to \mathbb{R}$ will be given in Sections 2 and 3. The well-posedness theory is developed using the concept of viscosity solutions and existence is achieved through finite volume discretisations on evolving triangulations.

6.1.1 A motivating example

It is the purpose of this chapter to study the natural development of a theory of viscosity solutions to first order equations on evolving surfaces. One motivation for considering Hamilton–Jacobi equations of the form (6.1.1) is the following application. Consider the motion of a closed curve $\gamma(t) \subset \Gamma(t)$ according to the evolution law

$$V_{\mu}(x,t) = F(x,t) + \beta(x,t) \cdot \mu(x,t), \qquad x \in \gamma(t), \tag{6.1.2}$$

where V_{μ} denotes the velocity of $\gamma(t)$ in the direction of the conormal μ and $F: S_T \to \mathbb{R}$, and $\beta: S_T \to \mathbb{R}^3$ are a given function and vector field. Let us assume that

$$\gamma(t) = \{(x,t) \in S_T \mid u(x,t) = r\}$$

for some $r \in \mathbb{R}$ with a function $u: N_T \to \mathbb{R}$ satisfying $\nabla_{\Gamma} u(\cdot, t) \neq 0$ on $\gamma(t)$, where N_T is an open neighbourhood of S_T . Choosing parametrizations $\varphi(\cdot, t): S^1 \to \mathbb{R}^3$ of $\gamma(t)$ we have that $u(\varphi(s,t),t) = r$ for $s \in S^1, t \in (0,T)$. If we differentiate both sides with respect to t we obtain

$$u_t(\varphi(s,t),t) + \varphi_t(s,t) \cdot \nabla u(\varphi(s,t),t) = 0,$$

or equivalently

$$0 = \partial^{\bullet} u(\varphi(s,t),t) + (\varphi_t(s,t) - v_{\Gamma}(\varphi_t(s,t),t)) \cdot \nabla u(\varphi(s,t),t) = \partial^{\bullet} u(\varphi(s,t),t) + (\varphi_t(s,t) - v_{\Gamma}(\varphi_t(s,t),t)) \cdot \nabla_{\Gamma} u(\varphi(s,t),t),$$

since $\varphi(s,t) \in \Gamma(t)$ implies that $(\varphi_t(s,t) - v_{\Gamma}(\varphi(s,t),t)) \cdot \nu(\varphi(s,t),t) = 0$. Using that $\mu = \nabla_{\Gamma} u / |\nabla_{\Gamma} u|$ we obtain from (6.1.2) that at $x = \varphi(s,t)$

$$F(x,t) + \beta(x,t) \cdot \mu(x,t) = V_{\mu}(x,t) = \varphi_t(s,t) \cdot \frac{\nabla_{\Gamma} u(x,t)}{|\nabla_{\Gamma} u(x,t)|}$$
$$= -\frac{\partial^{\bullet} u(x,t)}{|\nabla_{\Gamma} u(x,t)|} + v_{\Gamma}(x,t) \cdot \frac{\nabla_{\Gamma} u(x,t)}{|\nabla_{\Gamma} u(x,t)|}.$$

Formally the above calculations then show that the level sets of a solution u of (6.1.1) with

$$H(x,t,p) = F(x,t) |p| + \beta(x,t) \cdot p - v_{\Gamma}(x,t) \cdot p$$

$$(6.1.3)$$

evolve according to the evolution law (6.1.2).

6.1.2 Background

Partial differential equations on time evolving hypersurfaces arise in many applications in biology, fluids and materials science, for example see [6, 12, 13, 16] and the references cited therein. The theory of parabolic equations has been considered in [2,3,9,26]. Existence and uniqueness of first order scalar conservation laws on moving hypersurfaces and Riemannian manifolds has been proved in [11, 20]. Viscosity solutions of Hamilton–Jacobi equations on Riemannian manifolds are considered in [24]. See [7] and [23] for level set approaches to the motion of curves on a stationary surface. Numerical transport on evolving surfaces by level set methods was considered in [1, 27]. The numerical analysis of advection diffusion equations on evolving surfaces via the evolving surface finite element method began in [9], see also [10, 19]. Finite volume schemes for diffusion and conservation laws on moving surfaces have been considered, respectively, in [21] and [14]. Another approach is to use diffuse interfaces, see [25].

6.1.3 Outline

The paper is organized as follows. We begin in Section 2 by establishing some notation and concepts relating to moving surfaces. In Section 3 we generalise the classical definition of viscosity solution (see e.g. [4, 5, 15]) to moving curved domains using surface derivative operators instead of the usual derivatives. In this setting we show that a comparison principle holds which yields uniqueness of a viscosity solution. As in the seminal work [8] we approach existence via a discretisation in space and time. To do so, we approximate the moving surfaces by triangulated surfaces so that we need to formulate our numerical scheme an unstructured meshes. Numerical schemes for Hamilton–Jacobi equations on unstructured meshes on flat domains have been proposed in [18] and [22]. In order to guarantee monotonicity of their schemes the authors in [18], [22] have to assume that the underlying triangulation is acute, which is a rather strong requirement and difficult to realise in the case of moving surfaces where the triangulation will vary from time step to time step. In order to address this issue we construct in Section 4 a finite volume scheme by adapting an idea introduced by Kim and Li in [17] to the case of evolving hypersurfaces. With this construction we are able to prove monotonicity and consistency assuming only regularity of the triangulation. In Section 5 we prove that the sequence of discrete solutions obtained via our scheme converges to a viscosity solution if the discretization parameters tend to zero. At the same time this gives an existence result for the Hamilton–Jacobi equation. Finally, we prove in Section 6 an $O(\sqrt{h})$ error bound between the viscosity solution and the numerical solution extending well–known error estimates for the flat case to the case of moving hypersurfaces.

6.2 Preliminaries

6.2.1 Tangential derivatives of functions on fixed surfaces

Let Γ be a smooth, closed (i.e. compact without boundary) and orientable hypersurface in \mathbb{R}^3 with outward unit normal field ν . For a differentiable function f on Γ we define the tangential gradient by

$$\nabla_{\Gamma} f(x) := P_{\Gamma}(x) \nabla \tilde{f}(x), \quad x \in \Gamma,$$
(6.2.1)

where \tilde{f} is a smooth extension of f to an open neighbourhood N of Γ satisfying $\tilde{f} = f$ on $N \cap \Gamma$ and $P_{\Gamma}(x) := I_3 - \nu(x) \otimes \nu(x)$ is the orthogonal projection onto the tangent plane of Γ at x. Here I_3 is the (3×3) identity matrix and $\nu \otimes \nu = (\nu_i \nu_j)_{i,j}$ is the tensor product of ν . It is well-known that $\nabla_{\Gamma} f(x)$ is independent of the particular extension \tilde{f} . Furthermore, we define by $\Delta_{\Gamma} f := \nabla_{\Gamma} \cdot \nabla_{\Gamma} f$ the Laplace–Beltrami operator of f. We denote by d the signed distance function to Γ oriented in such a way that it increases in the direction of ν . There exists an open neighbourhood U of Γ such that d is smooth in U and such that for every $x \in U$ there exists a unique $\pi(x) \in \Gamma$ with

$$x = \pi(x) + d(x)\nu(\pi(x))$$
 and $\nabla d(x) = \nu(\pi(x)).$ (6.2.2)

For a given function $f: \Gamma \to \mathbb{R}$ we can define $f_c: U \to \mathbb{R}$ via $f_c(x) := f(\pi(x))$, which extends f constantly in the normal direction to Γ . It is not difficult to verify that

$$\nabla f_c(x) = \nabla_{\Gamma} f(x), \quad x \in \Gamma, \tag{6.2.3}$$

$$\|\nabla f_c\|_{B(U)} \le c \|\nabla_{\Gamma} f\|_{B(\Gamma)},\tag{6.2.4}$$

$$\|\nabla^2 f_c\|_{B(U)} \le c \left(\|\nabla_{\Gamma} f\|_{B(\Gamma)} + \|\nabla_{\Gamma}^2 f\|_{B(\Gamma)} \right), \tag{6.2.5}$$

provided that the derivatives of f exist. Here, $||f||_{B(D)} := \sup_{x \in D} |f(x)|$.

6.2.2 Time dependent surfaces

Let us next turn to the case of time dependent surfaces and assume that Γ_0 is a closed, connected, oriented and smooth hypersurface in \mathbb{R}^3 . We consider a family $\{\Gamma(t)\}_{t\in[0,T]}$, T > 0 of evolving hypersurfaces given via a smooth flow map $\Phi \colon \Gamma_0 \times [0,T] \to \mathbb{R}^3$ such that $\Phi(\cdot,t)$ is a diffeomorphism of Γ_0 onto $\Gamma(t)$ satisfying

$$\frac{\partial \Phi}{\partial t}(X,t) = v_{\Gamma}(\Phi(X,t),t), \quad \Phi(X,0) = X, \tag{6.2.6}$$

for all $X \in \Gamma_0, t \in (0, T)$. Here we say that v_{Γ} is the velocity field of $\Gamma(t)$. Let $d(\cdot, t)$ be the signed distance function to $\Gamma(t)$ increasing in the direction of $\nu(\cdot, t)$, where $\nu(\cdot, t)$ is the unit

outward normal of $\Gamma(t)$. For each $t \in [0,T]$ there exists a bounded open subset $N(t) \subset \mathbb{R}^3$ such that d is smooth in $N_T := \bigcup_{t \in (0,T)} (N(t) \times \{t\})$ and such that for every $x \in N(t)$ there exists a unique $\pi(x,t) \in \Gamma(t)$ satisfying (6.2.2).

Next, for a differentiable function f on S_T , the material derivative of f along the velocity v_{Γ} is defined as

$$\partial^{\bullet} f(\Phi(X,t),t) = \frac{d}{dt} \Big(f(\Phi(X,t),t) \Big), \quad (X,t) \in \Gamma_0 \times (0,T).$$

The material derivative is also expressed as

$$\partial^{\bullet} f(x,t) = \partial_t \tilde{f}(x,t) + v_{\Gamma}(x,t) \cdot \nabla \tilde{f}(x,t), \quad (x,t) \in S_T, \tag{6.2.7}$$

where \tilde{f} is an arbitrary extension of f to N_T satisfying $\tilde{f}|_{S_T} = f$.

6.2.3 Triangulated surface

In order to approximate the evolving surfaces $\Gamma(t)$ we choose a family of triangulations $(\mathcal{T}_h(t))_{0 \le h \le h_0}$ of $\Gamma(t)$ and set

$$\Gamma_h(t) := \bigcup_{K(t) \in \mathcal{T}_h(t)} K(t) \quad \text{and} \quad h := \max_{t \in [0,T]} \max_{K(t) \in \mathcal{T}_h(t)} h_{K(t)}$$

where $h_{K(t)} = \operatorname{diam} K(t)$ for each triangle K(t). We assume that there exists a constant $\gamma > 0$ such that

$$\forall t \in [0, T] \ \forall K(t) \in \mathcal{T}_h(t) \qquad h_{K(t)} \le \gamma \rho_{K(t)}, \tag{6.2.8}$$

where $\rho_{K(t)}$ is the radius of the largest circle contained in K(t). We denote by $\nu_h(\cdot, t)$ the unit normal to $\Gamma_h(t)$ oriented in the direction in which the signed distance $d(\cdot, t)$ increases. It is well-known that for all $K(t) \subset \Gamma_h(t)$

$$\|d(\cdot,t)\|_{B(K(t))} \leq Ch_{K(t)}^2, \tag{6.2.9}$$

$$\|\nu_{h|K(t)} - \nu(\cdot, t)\|_{B(K(t))} \leq Ch_{K(t)}, \qquad (6.2.10)$$

where we can think of $\nu(\cdot, t)$ as being extended to a neighbourhood of $\Gamma(t)$ via $\nu(x, t) = \nabla d(x, t)$ (cf. (6.2.2)).

We assume that the vertices of the triangulation are advected with the velocity v_{Γ} and thus the number of the vertices, which we refer to as $M \in \mathbb{N}$, is fixed in time. For $i = 1, \ldots, M$ we call the *i*-th vertex simply *i* and write $x_i^0 \in \Gamma(0)$ for its point at t = 0. By the assumption on the motion of the vertices, the position of *i* at time $t \in [0, T]$ is given by $x_i(t) = \Phi(x_i^0, t) \in \Gamma(t)$ so that the triangulated surfaces $\Gamma_h(t)$ are interpolations of $\Gamma(t)$. In particular, $\Gamma_h(t) \subset N(t)$ if h_0 is sufficiently small and we assume that $\pi_h(\cdot, t) := \pi(\cdot, t)|_{\Gamma_h(t)}$ is a homeomorphism of $\Gamma_h(t)$ onto $\Gamma(t)$ for each $t \in [0, T]$. Writing $\pi_h^{-1}(\cdot, t)$ for the inverse map, we define the lift of a function $\eta : \Gamma_h(t) \to \mathbb{R}$ onto $\Gamma(t)$ by

$$\eta^l(x) := \eta(\pi_h^{-1}(x,t)), \quad x \in \Gamma(t).$$

For each $t \in [0, T]$ we introduce the finite element space

 $V_h(t) = \{ u_h \in C^0(\Gamma_h(t)) \mid u_h|_{K(t)} \text{ is linear affine for each } K(t) \in \mathcal{T}_h(t) \}$

together with its standard nodal basis $\chi_1(\cdot, t), \ldots, \chi_M(\cdot, t)$, where $\chi_i(\cdot, t) \in V_h(t)$ satisfies $\chi_i(x_j(t), t) = \delta_{ij}$.

For a function $\eta \in C^0(\Gamma(t))$ we define the linear interpolation $I_h^t \eta \in V_h(t)$ by

$$I_h^t \eta(x) := \sum_{i=1}^M \eta(x_i(t)) \chi_i(x, t), \quad x \in \Gamma_h(t)$$

Lemma 6.2.1. Suppose that $\eta: \Gamma(t) \to \mathbb{R}$, $t \in [0,T]$ is Lipschitz continuous, i.e. there exists a constant $L_U > 0$ such that

$$|\eta(x) - \eta(y)| \le L_U |x - y|, \quad x, y \in \Gamma(t).$$
 (6.2.11)

Then we have

$$\|\eta - [I_h^t \eta]^l\|_{B(\Gamma(t))} \le Ch.$$
(6.2.12)

Proof. Fix $x \in \Gamma(t)$. Then there exists $\tilde{x} \in \Gamma_h(t)$ such that $x = \pi_h(\tilde{x}, t)$, say $\tilde{x} \in K(t)$ for some $K(t) \in \mathcal{T}_h(t)$. Assuming for simplicity that the vertices of K(t) are $x_1(t), x_2(t)$ and $x_3(t)$ we may write

$$\eta(x) - [I_h^t \eta]^l(x) = \eta(x) - \sum_{i=1}^3 \eta(x_i(t))\chi_i(\tilde{x}, t) = \sum_{i=1}^3 (\eta(x) - \eta(x_i(t)))\chi_i(\tilde{x}, t),$$

since $\sum_{i=1}^{3} \chi_i(\tilde{x}, t) = 1$. Combining this relation with the fact that $\chi_i(\tilde{x}, t) \ge 0$, (6.2.11), (6.2.2) and (6.2.9) we deduce that

$$\begin{aligned} |\eta(x) - [I_h^t \eta]^l(x)| &\leq \max_{i=1,2,3} |\eta(x) - \eta(x_i(t))| \\ &\leq L_U \max_{i=1,2,3} |x - x_i(t)| = L_U \max_{i=1,2,3} |\pi(\tilde{x}, t) - x_i(t)| \\ &\leq L_U \max_{i=1,2,3} (|\tilde{x} - x_i(t)| + |d(\tilde{x}, t)|) \\ &\leq L_U (h_{K(t)} + Ch_{K(t)}^2) \leq Ch_{K(t)} \leq Ch. \end{aligned}$$

6.3 Viscosity solutions: Uniqueness

We consider the Hamilton–Jacobi equation

$$\begin{cases} \partial^{\bullet} u(x,t) + H(x,t,\nabla_{\Gamma} u(x,t)) = 0, & (x,t) \in S_T, \\ u(x,0) = u_0(x), & x \in \Gamma(0). \end{cases}$$
(6.3.1)

Here $H: \overline{S_T} \times \mathbb{R}^3 \to \mathbb{R}$ is a Hamiltonian and $u_0: \Gamma(0) \to \mathbb{R}$ is an initial value. Throughout this chapter we suppose that there exist positive constants $L_{H,1}$ and $L_{H,2}$ such that

$$|H(x,t,p) - H(y,s,p)| \le L_{H,1}(|x-y| + |t-s|)(1+|p|), \tag{6.3.2}$$

$$|H(x,t,p) - H(x,t,q)| \le L_{H,2}|p-q|$$
(6.3.3)

for all $(x,t), (y,s) \in \overline{S_T}$ and $p, q \in \mathbb{R}^3$. Furthermore, we assume for the velocity field that $v_{\Gamma} \in C^1(\overline{S_T})$. Note that the Hamiltonian in (6.1.3) satisfies the above conditions provided that F and β are Lipschitz on $\overline{S_T}$.

For $\Gamma = \Gamma(t)$ with each fixed $t \in [0, T]$ or $\Gamma = \overline{S_T}$, we denote by $USC(\Gamma)$ (resp. $LSC(\Gamma)$) the set of all upper (resp. lower) semicontinuous functions on Γ . In what follows we shall work in the framework of discontinuous viscosity solutions. **Definition 6.3.1.** Let u_0 be a function on $\Gamma(0)$. A locally bounded function $u \in USC(\overline{S_T})$ (resp. $u \in LSC(\overline{S_T})$) is called a viscosity subsolution (resp. supersolution) of (6.3.1) if $u(x,0) \leq u_0(x)$ (resp. $u(x,0) \geq u_0(x)$) for all $x \in \Gamma(0)$ and, for any $\varphi \in C^1(\overline{S_T})$, if $u - \varphi$ takes a local maximum (resp. minimum) at $(x_0, t_0) \in \overline{S_T}$ with $t_0 > 0$, then

$$\partial^{\bullet}\varphi(x_0, t_0) + H(x_0, t_0, \nabla_{\Gamma}\varphi(x_0, t_0)) \le 0 \quad (\text{resp.} \ge 0).$$
 (6.3.4)

If u is a sub- and supersolution, then we call u a viscosity solution to (6.3.1).

By the definition above, a viscosity solution is continuous and satisfies $u(x,0) = u_0(x), x \in \Gamma(0)$. In Section 6.5 we prove that the upper and lower weak limits of a sequence of approximate solutions are a subsolution and supersolution, respectively, and then obtain a viscosity solution by showing that the upper weak limit agrees with the lower weak limit. For this argument and the uniqueness of a viscosity solution the following comparison principle is crucial.

Theorem 6.3.2. Let u be a subsolution and v be a supersolution of (6.3.1). Suppose that $u(\cdot, 0) \leq v(\cdot, 0)$ on $\Gamma(0)$. Then $u \leq v$ on $\overline{S_T}$.

Proof. We essentially use a standard argument that is e.g. outlined in [5, Section 5]. Let us define for $\eta > 0$ the function $u_{\eta}(x,t) := u(x,t) - \eta t$. Clearly, $u_{\eta} \in USC(\overline{S_T})$ and $u_{\eta}(\cdot,0) \leq v(\cdot,0)$ on $\Gamma(0)$. Since $v \in LSC(\overline{S_T})$ we have $u_{\eta} - v \in USC(\overline{S_T})$ so that $\sigma_{\eta} := \max_{\overline{S_T}}(u_{\eta} - v)$ exists. Let us suppose that $\sigma_{\eta} > 0$. We use the doubling of variables technique and define for $0 < \alpha \ll 1$

$$\Psi_{\alpha}(x,t,y,s) := u_{\eta}(x,t) - v(y,s) - \frac{|x-y|^2 + |t-s|^2}{\alpha^2}, \quad (x,t,y,s) \in \overline{S_T} \times \overline{S_T}.$$

The function Ψ_{α} is upper semicontinuous on $\overline{S_T} \times \overline{S_T}$ and hence attains a maximum at some point $(\bar{x}, \bar{t}, \bar{y}, \bar{s}) \in \overline{S_T} \times \overline{S_T}$, where we suppress the dependence on α . It is shown in [5, Lemma 5.2] that

$$\frac{|\bar{x}-\bar{y}|^2}{\alpha^2}, \ \frac{|\bar{t}-\bar{s}|^2}{\alpha^2} \to 0, \qquad \text{as } \alpha \to 0, \tag{6.3.5}$$

$$\bar{t}, \bar{s} > 0,$$
 for small $\alpha > 0.$ (6.3.6)

We define for $(x, t), (y, s) \in \mathbb{R}^4$ the functions

$$\varphi^1(x,t) := v(\bar{y},\bar{s}) + \frac{|x-\bar{y}|^2 + |t-\bar{s}|^2}{\alpha^2}, \ \varphi^2(y,s) := u_\eta(\bar{x},\bar{t}) - \frac{|\bar{x}-y|^2 + |\bar{t}-s|^2}{\alpha^2}$$

Clearly, the restriction of φ^i , i = 1, 2 to $\overline{S_T}$ belongs to $C^1(\overline{S_T})$. Since u is a subsolution to (6.3.1) and $u(x,t) - (\varphi^1(x,t) + \eta t) = (u_\eta - \varphi^1)(x,t) = \Psi_\alpha(x,t,\bar{y},\bar{s})$ takes a maximum at $(x,t) = (\bar{x},\bar{t}) \in \overline{S_T}$ with $\bar{t} > 0$, we have

$$\partial^{\bullet}\varphi^{1}(\bar{x},\bar{t}) + H(\bar{x},\bar{t},\nabla_{\Gamma}\varphi^{1}(\bar{x},\bar{t})) \leq -\eta.$$

Observing that by (6.2.1) and (6.2.7)

$$\nabla_{\Gamma}\varphi^{1}(x,t) = \frac{2}{\alpha^{2}}P_{\Gamma}(x,t)(x-\bar{y}), \quad \partial^{\bullet}\varphi^{1}(x,t) = \frac{2}{\alpha^{2}}(t-\bar{s}) + \frac{2}{\alpha^{2}}v_{\Gamma}(x,t)\cdot(x-\bar{y})$$

we deduce

$$\frac{2(\bar{t}-\bar{s})}{\alpha^2} + \frac{2}{\alpha^2} v_{\Gamma}(\bar{x},\bar{t}) \cdot (\bar{x}-\bar{y}) + H(\bar{x},\bar{t},\frac{2}{\alpha^2} P_{\Gamma}(\bar{x},\bar{t})(\bar{x}-\bar{y})) \le -\eta.$$
(6.3.7)

Since v is a supersolution and $(v - \varphi^2)(y, s) = -\Psi_{\alpha}(\bar{x}, \bar{t}, y, s)$ takes a minimum at $(y, s) = (\bar{y}, \bar{s}) \in \overline{S_T}$ with $\bar{s} > 0$, it follows that

$$\partial^{\bullet}\varphi^{2}(\bar{y},\bar{s}) + H(\bar{y},\bar{s},\nabla_{\Gamma}\varphi^{2}(\bar{y},\bar{s})) \ge 0$$

and we obtain similarly as above

$$\frac{2(\bar{t}-\bar{s})}{\alpha^2} - \frac{2}{\alpha^2} v_{\Gamma}(\bar{y},\bar{s}) \cdot (\bar{x}-\bar{y}) - H(\bar{y},\bar{s},\frac{2}{\alpha^2} P_{\Gamma}(\bar{y},\bar{s})(\bar{x}-\bar{y})) \le 0.$$
(6.3.8)

We deduce from (6.3.7) and (6.3.8) that

$$\bar{A} := \frac{2}{\alpha^2} \{ v_{\Gamma}(\bar{x}, \bar{t}) - v_{\Gamma}(\bar{y}, \bar{s}) \} \cdot (\bar{x} - \bar{y})$$

+ $H\left(\bar{x}, \bar{t}, \frac{2}{\alpha^2} P_{\Gamma}(\bar{x}, \bar{t})(\bar{x} - \bar{y}) \right) - H\left(\bar{y}, \bar{s}, \frac{2}{\alpha^2} P_{\Gamma}(\bar{y}, \bar{s})(\bar{x} - \bar{y}) \right) \le -\eta.$ (6.3.9)

Since v_{Γ} , P_{Γ} are smooth on $\overline{S_T}$ we obtain with the help of (6.3.2), (6.3.3) and (6.3.5)

$$\begin{split} |\bar{A}| &\leq \frac{2}{\alpha^2} \left(|v_{\Gamma}(\bar{x},\bar{t}) - v_{\Gamma}(\bar{y},\bar{s})| + L_{H,2} |P_{\Gamma}(\bar{x},\bar{t}) - P_{\Gamma}(\bar{y},\bar{s})| \right) |\bar{x} - \bar{y} \\ &+ L_{H,1} (|\bar{x} - \bar{y}| + |\bar{t} - \bar{s}|) \left(1 + \frac{2}{\alpha^2} |P_{\Gamma}(\bar{x},\bar{t})(\bar{x} - \bar{y})| \right) \\ &\leq C \frac{|\bar{x} - \bar{y}|^2 + |\bar{t} - \bar{s}|^2}{\alpha^2} + \alpha^2 \to 0, \quad \alpha \to 0 \end{split}$$

contradicting (6.3.9). Hence, $\sigma_{\eta} \leq 0$, so that $u_{\eta} \leq v$ on $\overline{S_T}$. The result now follows upon sending $\eta \to 0$.

Corollary 6.3.3 (Uniqueness of a viscosity solution). For any initial value $u_0 \in C(\Gamma(0))$ there exists at most one viscosity solution to (6.3.1).

6.4 Finite volume scheme

Let us next turn to the approximation of (6.3.1). As mentioned already in the introduction, our scheme is based on the finite volume scheme for Hamilton–Jacobi equations in a flat and stationary domain introduced by Kim and Li in [17].

Let N be a positive integer, $\tau := T/N$ a time step, $t^n = n\tau, n = 0, ..., N$ and $x_i^n = x_i(t^n), V_h^n = V_h(t^n)$. In order to derive our scheme we start from the following viscous approximation of (6.3.1)

$$\partial^{\bullet} u(x,t) + H(x,t,\nabla_{\Gamma} u(x,t)) = \varepsilon \Delta_{\Gamma} u(x,t), \quad (x,t) \in S_T,$$
(6.4.1)

where $0 < \epsilon \ll 1$. Let us fix $i \in \{1, \ldots, M\}$ and consider a time-dependent set $V_i(t) \subset \Gamma(t)$ centered at $x_i(t)$. Integrating (6.4.1) for $t = t^n$ over $V_i(t^n)$ we find that

$$\int_{V_i(t^n)} \partial^{\bullet} u \, d\mathcal{H}^2 + \int_{V_i(t^n)} H(\cdot, t^n, \nabla_{\Gamma} u) \, d\mathcal{H}^2 = \varepsilon \int_{V_i(t^n)} \Delta_{\Gamma} u \, d\mathcal{H}^2.$$
(6.4.2)

Here, \mathcal{H}^n is the *n*-dimensional Hausdorff measure. Let us consider the first term on the left-hand side of (6.4.2). Using the transport theorem (see e.g. [10, Theorem 5.1]) and
approximating $\int_{V_i(t)} u \, d\mathcal{H}^2$ by $u(x_i(t), t)|V_i(t)| \ (|V_i(t)| = \mathcal{H}^2(V_i(t)))$ we obtain

$$\int_{V_i(t^n)} \partial^{\bullet} u \, d\mathcal{H}^2 = \frac{d}{dt} \int_{V_i(t)} u \, d\mathcal{H}^2_{|t=t^n} - \int_{V_i(t^n)} \nabla_{\Gamma} \cdot v_{\Gamma} \, u \, d\mathcal{H}^2$$
$$\approx \frac{u(x_i^{n+1}, t^{n+1})|V_i(t^{n+1})| - u(x_i^n, t^n)|V_i(t^n)|}{\tau} - \int_{V_i(t^n)} \nabla_{\Gamma} \cdot v_{\Gamma} \, u \, d\mathcal{H}^2.$$

Since $\frac{d}{dt}|V_i(t)| = \int_{V_i(t)} \nabla_{\Gamma} \cdot v_{\Gamma} d\mathcal{H}^2$ we may approximate

$$|V_i(t^{n+1})| \approx |V_i(t^n)| + \tau \int_{V_i(t^n)} \nabla_{\Gamma} \cdot v_{\Gamma} \, d\mathcal{H}^2$$

so that

$$\int_{V_i(t^n)} \partial^{\bullet} u \, d\mathcal{H}^2 \approx \frac{u(x_i^{n+1}, t^{n+1}) - u(x_i^n, t^n)}{\tau} |V_i(t^n)|.$$

Finally, after applying the Gauss theorem for hypersurfaces to the integral on the right-hand side of (6.4.2) we obtain

$$\frac{u(x_i^{n+1}, t^{n+1}) - u(x_i^n, t^n)}{\tau} |V_i(t^n)| + \int_{V_i(t^n)} H(\cdot, t^n, \nabla_{\Gamma} u) \, d\mathcal{H}^2 \approx \varepsilon \int_{\partial V_i(t^n)} \frac{\partial u}{\partial \mu} \, d\mathcal{H}^1, \quad (6.4.3)$$

where μ denotes the outer unit conormal to $\partial V_i(t^n)$. In order to turn (6.4.3) into a numerical scheme we construct a suitable discrete version $V^{n,i} \subset \Gamma_h(t^n)$ of $V_i(t^n)$. Let $\mu_i \in \mathbb{N}$ be the number of triangles that have the common vertex i, which is independent of n. The other vertices of the triangles with common vertex i are denoted by i_j , $j = 1, \ldots, \mu_i$, which we enumerate in clockwise direction. We write $T_j^{n,i} \in \mathcal{T}_h(t^n)$ for the triangle with vertices i, i_j , and i_{j+1} and $E_j^{n,i}, \bar{E}_j^{n,i}$ for the edges of $T_j^{n,i}$ connecting the vertices i and i_j and the vertices i_j and i_{j+1} , respectively (see Figure 6.1, left).



Figure 6.1:

Let $d_{j}^{n,i}$ be the length from the vertex *i* to the contact point on $E_{j}^{n,i}$ of the inscribed circle of $T_{j}^{n,i}$ and $d^{n,i} := \min\{d_{j}^{n,i} \mid j = 1, \ldots, \mu_i\}$. We define the volume $V^{n,i} \subset \Gamma_h(t^n)$ as a polygonal region surrounded by line segments perpendicular to each edge $E_{j}^{n,i}$ and whose distances from the vertex *i* are all equal to $d^{n,i}$. The parts of the edge of $V^{n,i}$ perpendicular to $E_{j}^{n,i}$ and lying in $T_{j-1}^{n,i}$ and $T_{j}^{n,i}$ are denoted by $e_{j,L}^{n,i}$ and $e_{j,R}^{n,i}$ with their length $h_{j,L}^{n,i}$ and $h_{j,R}^{n,i}$, respectively (see Figure 6.1, right). Note that in view of (6.2.8) there exist constants $0 < \alpha_1 < \alpha_2$ and C > 0 such that

$$\alpha_1 \le \frac{h_{j,L}^{n,i} + h_{j,R}^{n,i}}{|E_j^{n,i}|} \le \alpha_2, \quad h_{T_j^{n,i}} \le Cd^{n,i}$$
(6.4.4)

for all n = 0, 1, ..., N, i = 1, ..., M, and $j = 1, ..., \mu_i$. If we look for a discrete solution of the form $u_h^n = \sum_{i=1}^M u_i^n \chi_i(\cdot, t^n) \in V_h^n$, then (6.4.3) motivates the following relation:

$$\frac{u_i^{n+1} - u_i^n}{\tau} |V^{n,i}| + \sum_{j=1}^{\mu_i} |V^{n,i} \cap T_j^{n,i}| H\left(x_i^n, t^n, \nabla_{\Gamma_h} u_h^n|_{T_j^{n,i}}\right) = \varepsilon_i^n \sum_{j=1}^{\mu_i} \frac{u_{i_j}^n - u_i^n}{|E_j^{n,i}|} (h_{j,L}^{n,i} + h_{j,R}^{n,i})$$

for suitably chosen $\varepsilon_i^n > 0$. Here, $\nabla_{\Gamma_h} u^n = (I_3 - \nu_j^{n,i} \otimes \nu_j^{n,i}) \nabla u_h^n$ with $\nu_j^{n,i} = \nu_{h|T_j^{n,i}}$. Note that $\nu_j^{n,i}$ and hence $\nabla_{\Gamma_h} u^n$ is constant on $T_j^{n,i}$. To summarize, our numerical scheme for the Hamilton–Jacobi equation (6.3.1) looks as follows. For a given $u_0 \colon \Gamma(0) \to \mathbb{R}$, set

$$u_h^0 := I_h^0 u_0 = \sum_{i=1}^M u_i^0 \chi_i(\cdot, 0) \in V_h^0, \quad u_i^0 := u_0(x_i^0).$$
(6.4.5)

For n = 0, 1, ..., N - 1, if $u_h^n = \sum_{i=1}^M u_i^n \chi_i(\cdot, t^n) \in V_h^n$ is given, then we define

$$u_h^{n+1} = S_h^n(u_h^n) := \sum_{i=1}^M u_i^{n+1} \chi_i(\cdot, t^{n+1}) \in V_h^{n+1},$$
(6.4.6)

where

$$u_i^{n+1} = [S_h^n(u_h^n)]_i := u_i^n - \tau H_i^n(u_i^n, u_{i_1}^n, \dots, u_{i_{\mu_i}}^n), \quad i = 1, \dots, M.$$
(6.4.7)

Here $H_i^n(u_i^n, u_{i_1}^n, \dots, u_{i_{\mu_i}}^n)$ is the numerical Hamiltonian given by

$$H_{i}^{n}(u_{i}^{n}, u_{i_{1}}^{n}, \dots, u_{i_{\mu_{i}}}^{n}) = \sum_{j=1}^{\mu_{i}} \frac{|V^{n,i} \cap T_{j}^{n,i}|}{|V^{n,i}|} H\left(x_{i}^{n}, t^{n}, \nabla_{\Gamma_{h}} u_{h}^{n}|_{T_{j}^{n,i}}\right) - \frac{\varepsilon_{i}^{n}}{|V^{n,i}|} \sum_{j=1}^{\mu_{i}} \frac{u_{i_{j}}^{n} - u_{i}^{n}}{|E_{j}^{n,i}|} (h_{j,L}^{n,i} + h_{j,R}^{n,i}). \quad (6.4.8)$$

Let us derive several properties of the finite volume scheme (6.4.5)-(6.4.8). It is easy to see that the scheme is invariant under translation with constants, i.e.

$$S_h^n(u_h^n + c) = S_h^n(u_h^n) + c ag{6.4.9}$$

for any $u_h^n \in V_h^n$ and $c \in \mathbb{R}$. We proceed by proving that the scheme is montone.

Lemma 6.4.1 (Monotonicity). There exist positive constants C_1 and C_2 depending only on γ and $L_{H,2}$ such that, if

$$\varepsilon_i^n = C_1 \max_j h_{T_j^{n,i}}, \quad \tau \le C_2 \min_{i,j} |E_j^{n,i}|$$
(6.4.10)

and u_h^n , $v_h^n \in V_h^n$ satisfy $u_h^n \leq v_h^n$ on $\Gamma_h(t^n)$, then $S_h^n(u_h^n) \leq S_h^n(v_h^n)$ on $\Gamma_h(t^{n+1})$.

Proof. Let $u_h^n, v_h^n \in V_h^n$ be of the form

$$u_h^n = \sum_{i=1}^M u_i^n \chi_i(\cdot, t^n), \quad v_h^n = \sum_{i=1}^M v_i^n \chi_i(\cdot, t^n) \quad \text{on} \quad \Gamma_h(t^n).$$

Note that $u_h^n \leq v_h^n$ on $\Gamma_h(t^n)$ is equivalent to $u_i^n \leq v_i^n$ for all $i = 1, \ldots, M$ since the nodal basis functions χ_i are piecewise linear affine and satisfy $\chi_i(x_j(t), t) = \delta_{ij}$. By the same reason it is sufficient to establish that

$$[S_h^n(u_h^n)]_i \le [S_h^n(v_h^n)]_i \quad \text{for all} \quad i = 1, \dots, M$$
(6.4.11)

in order to prove our claim. For i = 1, ..., M, by (6.4.7) and (6.4.8) we have

$$[S_h^n(v_h^n)]_i - [S_h^n(u_h^n)]_i = v_i^n - u_i^n + \tau(I_1 + I_2 + I_3),$$
(6.4.12)

where $I_1 + I_2 + I_3 = -H_i^n(v_i^n, v_{i_1}^n, \dots, v_{i_{\mu_i}}^n) + H_i^n(u_i^n, u_{i_1}^n, \dots, u_{i_{\mu_i}}^n)$ with

$$I_{1} := -\sum_{j=1}^{\mu_{i}} \frac{|V^{n,i} \cap T_{j}^{n,i}|}{|V^{n,i}|} \left\{ H\left(\nabla_{\Gamma_{h}} v_{h}^{n}|_{T_{j}^{n,i}}\right) - H\left(\nabla_{\Gamma_{h}} u_{h}^{n}|_{T_{j}^{n,i}}\right) \right\},$$

$$I_{2} := \frac{\varepsilon_{i}^{n}}{|V^{n,i}|} \sum_{j=1}^{\mu_{i}} \frac{v_{ij}^{n} - u_{ij}^{n}}{|E_{j}^{n,i}|} (h_{j,L}^{n,i} + h_{j,R}^{n,i}),$$

$$I_{3} := -\frac{\varepsilon_{i}^{n} (v_{i}^{n} - u_{i}^{n})}{|V^{n,i}|} \sum_{j=1}^{\mu_{i}} \frac{(h_{j,L}^{n,i} + h_{j,R}^{n,i})}{|E_{j}^{n,i}|}.$$

In the definition of I_1 we suppressed x_i^n and t^n of H. Let us estimate I_1 , I_2 , and I_3 . By (6.3.3) and an inverse inequality

$$\begin{aligned} \left| H\left(\nabla_{\Gamma_{h}} v_{h}^{n}|_{T_{j}^{n,i}}\right) - H\left(\nabla_{\Gamma_{h}} u_{h}^{n}|_{T_{j}^{n,i}}\right) \right| &\leq L_{H,2} \left| \nabla_{\Gamma_{h}} v_{h}^{n}|_{T_{j}^{n,i}} - \nabla_{\Gamma_{h}} u_{h}^{n}|_{T_{j}^{n,i}} \right| \\ &\leq C |\nabla(v_{h}^{n} - u_{h}^{n})|_{T_{j}^{n,i}}| \leq C |E_{j}^{n,i}|^{-1} \left\| v_{h}^{n} - u_{h}^{n} \right\|_{B(T_{j}^{n,i})} \\ &\leq C |E_{j}^{n,i}|^{-1} \{ (v_{i}^{n} - u_{i}^{n}) + (v_{ij}^{n} - u_{ij}^{n}) + (v_{ij+1}^{n} - u_{ji+1}^{n}) \}, \end{aligned}$$

since u_h^n, v_h^n are linear on $T_j^{n,i}$ and $v_h^n - u_h^n \ge 0$. Using that $\sum_{j=1}^{\mu_i} |V^{n,i} \cap T_j^{n,i}| / |V^{n,i}| = 1$ as well as

$$|V^{n,i} \cap T_j^{n,i}| = \frac{1}{2} d^{n,i} (h_{j,R}^{n,i} + h_{j+1,L}^{n,i}) \le |E_j^{n,i}| \max_j h_{T_j^{n,i}}, \quad j = 1, \dots, \mu_i,$$
(6.4.13)

we get

$$|I_1| \le \frac{C}{\min_j |E_j^{n,i}|} (v_i^n - u_i^n) + \frac{C}{|V^{n,i}|} \max_j h_{T_j^{n,i}} \sum_{j=1}^{\mu_i} (v_{i_j}^n - u_{i_j}^n).$$
(6.4.14)

Next, from (6.4.4) and the fact that $u_{i_j}^n \leq v_{i_j}^n$ for $j = 1, \ldots, \mu_i$ we infer that

$$I_2 \ge \frac{\alpha_1 \varepsilon_i^n}{|V^{n,i}|} \sum_{j=1}^{\mu_i} (v_{i_j}^n - u_{i_j}^n).$$
(6.4.15)

In view of the relation $|V^{n,i}| = \sum_{j=1}^{\mu_i} \frac{1}{2} d^{n,i} (h_{j,L}^{n,i} + h_{j,R}^{n,i})$ and (6.2.8) we obtain

$$\begin{aligned} \frac{1}{|V^{n,i}|} \sum_{j=1}^{\mu_i} \frac{h_{j,L}^{n,i} + h_{j,R}^{n,i}}{|E_j^{n,i}|} &\leq \frac{1}{|V^{n,i}|} \frac{1}{\min_j |E_j^{n,i}|} \sum_{j=1}^{\mu_i} (h_{j,L}^{n,i} + h_{j,R}^{n,i}) = \frac{2}{d^{n,i}} \frac{1}{\min_j |E_j^{n,i}|} \\ &\leq \frac{C}{\max_j h_{T_j^{n,i}} \min_j |E_j^{n,i}|} \leq \frac{CC_1}{\varepsilon_i^n} \frac{1}{\min_j |E_j^{n,i}|}, \end{aligned}$$

where we used (6.4.10) in the last step. Hence

$$I_3 \ge -\frac{CC_1}{\min_j |E_j^{n,i}|} (v_i^n - u_i^n).$$
(6.4.16)

From (6.4.12), (6.4.14), (6.4.15), and (6.4.16) it follows that

$$\begin{split} [S_h^n(v_h^n)]_i &- [S_h^n(u_h^n)]_i \\ &\geq \left(1 - \frac{\tau C(1+C_1)}{\min_j |E_j^{n,i}|}\right) (v_i^n - u_i^n) + \frac{1}{|V^{n,i}|} (\alpha_1 \varepsilon_i^n - C \max_j h_{T_j^{n,i}}) \sum_{j=1}^{\mu_i} (v_{i_j}^n - u_{i_j}^n) \end{split}$$

which yields (6.4.11) if we choose $C_1 = C/\alpha_1$ and $C_2 = 1/C(1+C_1)$ in (6.4.10). \Box In what follows we write $I_h^n \varphi$ instead of $I_h^{t^n} \varphi$, i.e.

$$I_h^n \varphi = \sum_{i=1}^M \varphi_i^n \chi_i(\cdot, t^n) \in V_h^n, \quad \varphi_i^n = \varphi(x_i^n, t^n).$$

Lemma 6.4.2 (Consistency). Suppose that (6.4.10) is satisfied. Then there exists a constant $C_3 > 0$ depending only on γ , $L_{H,2}$ such that

$$\left| \frac{\varphi_i^{n+1} - [S_h^n(I_h^n \varphi)]_i}{\tau} - \left\{ \partial^{\bullet} \varphi(x_i^n, t^n) + H(x_i^n, t^n, \nabla_{\Gamma} \varphi(x_i^n, t^n)) \right\} \right| \\ \leq C_3 h \left(\|\nabla_{\Gamma} \varphi\|_{B(\overline{S_T})} + \|\nabla_{\Gamma}^2 \varphi\|_{B(\overline{S_T})} + \|(\partial^{\bullet})^2 \varphi\|_{B(\overline{S_T})} \right) \quad (6.4.17)$$

for all $\varphi \in C^2(\overline{S_T})$, n = 0, 1, ..., N - 1, and i = 1, ..., M. Here, $(\partial^{\bullet})^2 \varphi$ is the second-order material derivative of φ .

Proof. Using (6.4.7) and (6.4.8) we have

$$\frac{\varphi_i^{n+1} - [S_h^n(I_h^n \varphi)]_i}{\tau} = \frac{\varphi_i^{n+1} - \varphi_i^n}{\tau} + H_i^n(\varphi_i^n, \varphi_{i_1}^n, \dots, \varphi_{i_{\mu_i}}^n).$$

Let us set

$$\begin{split} I_1 &:= \frac{\varphi_i^{n+1} - \varphi_i^n}{\tau} - \partial^{\bullet} \varphi(x_i^n, t^n), \\ I_2 &:= \sum_{j=1}^{\mu_i} \frac{|V^{n,i} \cap T_j^{n,i}|}{|V^{n,i}|} H(x_i^n, t^n, \nabla_{\Gamma_h} I_h^n \varphi|_{T_j^{n,i}}) - H(x_i^n, t^n, \nabla_{\Gamma} \varphi(x_i^n, t^n)), \\ I_3 &:= -\frac{\varepsilon_i^n}{|V^{n,i}|} \sum_{j=1}^{\mu_i} \frac{\varphi_{i_j}^n - \varphi_i^n}{|E_j^{n,i}|} (h_{j,L}^{n,i} + h_{j,R}^{n,i}), \end{split}$$

so that

$$\frac{\varphi_i^{n+1} - [S_h^n(I_h^n \varphi)]_i}{\tau} - \{\partial^{\bullet}\varphi(x_i^n, t^n) + H(x_i^n, t^n, \nabla_{\Gamma}\varphi(x_i^n, t^n))\} = I_1 + I_2 + I_3 \qquad (6.4.18)$$

and estimate I_1 , I_2 , and I_3 separately. From $\varphi_i^n = \varphi(x_i^n, t^n) = \varphi(\Phi(x_i^0, t^n), t^n)$ and the definition of the material derivative it follows that

$$\frac{\varphi_i^{n+1} - \varphi_i^n}{\tau} = \frac{1}{\tau} \int_{t^n}^{t^{n+1}} \frac{d}{dt} \left(\varphi(\Phi(x_i^0, s), s) \right) ds = \frac{1}{\tau} \int_{t^n}^{t^{n+1}} \partial^{\bullet} \varphi(\Phi(x_i^0, s), s) \, ds.$$

Applying the definition of the material derivative again we obtain

$$I_1 = \frac{1}{\tau} \int_{t^n}^{t^{n+1}} \{\partial^{\bullet} \varphi(\Phi(x_i^0, s), s) - \partial^{\bullet} \varphi(\Phi(x_0^i, t^n), t^n)\} ds$$
$$= \frac{1}{\tau} \int_{t^n}^{t^{n+1}} \int_{t^n}^s (\partial^{\bullet})^2 \varphi(\Phi(x_i^0, \tilde{s}), \tilde{s}) d\tilde{s} ds.$$

Since $\varphi \in C^2(\overline{S_T})$, the second-order material derivative $(\partial^{\bullet})^2 \varphi$ is bounded on $\overline{S_T}$. Hence by the above equality, $t^{n+1} - t^n = \tau$, and (6.4.10) we obtain

$$|I_1| \le \frac{(t^{n+1} - t^n)^2}{\tau} \|(\partial^{\bullet})^2 \varphi\|_{B(\overline{S_T})} = \tau \|(\partial^{\bullet})^2 \varphi\|_{B(\overline{S_T})} \le Ch \|(\partial^{\bullet})^2 \varphi\|_{B(\overline{S_T})}.$$
 (6.4.19)

Next we estimate I_2 . From now on, we suppress t^n in all functions and x_i^n in the Hamiltonian. Clearly,

$$I_{2} = \sum_{j=1}^{\mu_{i}} \frac{|V^{n,i} \cap T_{j}^{n,i}|}{|V^{n,i}|} \left\{ H\left(\nabla_{\Gamma_{h}} I_{h}^{n} \varphi|_{T_{j}^{n,i}}\right) - H(\nabla_{\Gamma} \varphi(x_{i}^{n})) \right\}.$$
 (6.4.20)

For each $j = 1, \ldots, \mu_i$, the inequality (6.3.3) yields

$$\left| H\left(\nabla_{\Gamma_h} I_h^n \varphi|_{T_j^{n,i}} \right) - H(\nabla_{\Gamma} \varphi(x_i^n)) \right| \le L_{H,2} \left| \nabla_{\Gamma_h} I_h^n \varphi|_{T_j^{n,i}} - \nabla_{\Gamma} \varphi(x_i^n) \right|.$$
(6.4.21)

Abbreviating $\varphi^{-l}(x) := \varphi(\pi_h(x)), x \in \Gamma_h$ we may write

$$\nabla_{\Gamma_h} I_h^n \varphi|_{T_j^{n,i}} - \nabla_{\Gamma} \varphi(x_i^n)$$

= $\left(\nabla_{\Gamma_h} I_h^n \varphi|_{T_j^{n,i}} - \nabla_{\Gamma_h} \varphi^{-l}(x_i^n) \right) + \left(\nabla_{\Gamma_h} \varphi^{-l}(x_i^n) - \nabla_{\Gamma} \varphi(x_i^n) \right) \equiv A + B. \quad (6.4.22)$

Since $I_h^n \varphi_{|T_j^{n,i}|}$ is the linear interpolation of $\varphi_{|T_i^{n,i}|}^{-l}$ we obtain

$$|A| \le Ch \|\nabla_{\Gamma_h}^2 \varphi^{-l}\|_{B(T_j^{n,i})} \le Ch \left(\|\nabla_{\Gamma}\varphi\|_{B(\overline{S_T})} + \|\nabla_{\Gamma}^2\varphi\|_{B(\overline{S_T})}\right).$$
(6.4.23)

On the other hand, we infer from (4.18) in [10] and the relations $\pi_h(x_i^n) = x_i^n, d(x_i^n) = 0, \nu(x_i^n) \cdot \nabla_{\Gamma} \varphi(x_i^n) = 0$ that

$$B = (I_3 - \nu_j^{n,i} \otimes \nu_j^{n,i}) \nabla_{\Gamma} \varphi(x_i^n) - \nabla_{\Gamma} \varphi(x_i^n) = \left(\nu(x_i^n) \otimes \nu(x_i^n) - \nu_j^{n,i} \otimes \nu_j^{n,i}\right) \nabla_{\Gamma} \varphi(x_i^n),$$

so that by (6.2.10)

$$|B| \le 2 \|\nu - \nu_h\|_{B(T_j^{n,i})} \|\nabla_{\Gamma}\varphi\|_{B(\overline{S_T})} \le Ch \|\nabla_{\Gamma}\varphi\|_{B(\overline{S_T})}.$$
(6.4.24)

Combining (6.4.20)–(6.4.24) we obtain

$$|I_2| \le Ch\left(\|\nabla_{\Gamma}\varphi\|_{B(\overline{S_T})} + \|\nabla_{\Gamma}^2\varphi\|_{B(\overline{S_T})} \right).$$
(6.4.25)

Finally, let us write

$$I_3 = \frac{\varepsilon_i^n}{|V^{n,i}|} (J_1 + J_2), \tag{6.4.26}$$

where

$$J_{1} := -\sum_{j=1}^{\mu_{i}} \frac{h_{j,L}^{n,i} + h_{j,R}^{n,i}}{|E_{j}^{n,i}|} \{ (\varphi_{i_{j}}^{n} - \varphi_{i}^{n}) - \nabla_{\Gamma}\varphi(x_{i}^{n}) \cdot (x_{i_{j}}^{n} - x_{i}^{n}) \},$$

$$J_{2} := -\sum_{j=1}^{\mu_{i}} \left(\nabla_{\Gamma}\varphi(x_{i}^{n}) \cdot \frac{x_{i_{j}}^{n} - x_{i}^{n}}{|E_{j}^{n,i}|} \right) (h_{j,L}^{n,i} + h_{j,R}^{n,i}).$$
(6.4.27)

Extending φ constantly in normal direction via φ_c and recalling (6.2.3) we have

$$\begin{split} \varphi_{i_{j}}^{n} - \varphi_{i}^{n} - \nabla_{\Gamma}\varphi(x_{i}^{n}) \cdot (x_{i_{j}}^{n} - x_{i}^{n}) &= \varphi_{c}(x_{i_{j}}^{n}) - \varphi_{c}(x_{i}^{n}) - \nabla\varphi_{c}(x_{i}^{n}) \cdot (x_{i_{j}}^{n} - x_{i}^{n}) \\ &= \int_{0}^{1} \{\nabla\varphi_{c}(x_{i}^{n} + s(x_{i_{j}}^{n} - x_{i}^{n})) - \nabla\varphi_{c}(x_{i}^{n})\} \, ds \cdot (x_{i_{j}}^{n} - x_{i}^{n}) \\ &= \int_{0}^{1} \left(\int_{0}^{s} \nabla^{2}\varphi_{c}(x_{i}^{n} + \tilde{s}(x_{i_{j}}^{n} - x_{i}^{n}))(x_{i_{j}}^{n} - x_{i}^{n}) \, d\tilde{s} \right) ds \cdot (x_{i_{j}}^{n} - x_{i}^{n}). \end{split}$$

Thus, we deduce from (6.2.5) and (6.4.4) that

$$|J_1| \le C \left(\|\nabla_{\Gamma}\varphi\|_{B(\overline{S_T})} + \|\nabla_{\Gamma}^2\varphi\|_{B(\overline{S_T})} \right) \sum_{j=1}^{\mu_i} |E_j^{n,i}|^2.$$
(6.4.28)

To estimate J_2 we observe that

$$0 = \int_{V^{n,i}} \operatorname{div}_{\Gamma_h} p \, d\mathcal{H}^2 = \sum_{j=1}^{\mu_i} \int_{V^{n,i} \cap T_j^{n,i}} \operatorname{div}_{\Gamma_h} p \, d\mathcal{H}^2 \tag{6.4.29}$$

for the constant vector $p = \nabla_{\Gamma} \varphi(x_i^n) \in \mathbb{R}^3$. For each $j = 1, \ldots, \mu_i, V^{n,i} \cap T_j^{n,i}$ is a flat quadrilateral whose sides consist of the edges $e_{j,R}^{n,i}, e_{j+1,L}^{n,i}$, and

$$S_{j,L}^{n,i} := E_j^{n,i} \cap \partial(V^{n,i} \cap T_j^{n,i}), \quad S_{j,R}^{n,i} := E_{j+1}^{n,i} \cap \partial(V^{n,i} \cap T_j^{n,i}).$$

The unit outward co-normal $\mu_j^{n,i}$ to $\partial(V^{n,i} \cap T_j^{n,i})$ (i.e. the unit outward normal to $\partial(V^{n,i} \cap T_j^{n,i})$ that is tangent to $T_j^{n,i}$) is given by

$$\mu_{j}^{n,i} = \begin{cases} \mu_{j,E}^{n,i} & \text{on } e_{j,R}^{n,i}, \\ \mu_{j+1,E}^{n,i} & \text{on } e_{j+1,L}^{n,i}, \\ \mu_{j,L}^{n,i} & \text{on } S_{j,L}^{n,i}, \\ \mu_{j+1,R}^{n,i} & \text{on } S_{j,R}^{n,i}, \end{cases}$$



Figure 6.2:

where (see Figure 6.2)

$$\mu_{j,E}^{n,i} := \frac{x_{i_j}^n - x_i^n}{|x_{i_j}^n - x_i^n|} \quad \text{and} \quad \mu_{j,L}^{n,i} := \nu_j^{n,i} \times \mu_{j,E}^{n,i}, \quad \mu_{j,R}^{n,i} := -\nu_{j-1}^{n,i} \times \mu_{j,E}^{n,i}. \tag{6.4.30}$$

Here, \times denotes the vector product in \mathbb{R}^3 . Using the divergence theorem for integrals over a flat quadrilateral we have

$$\begin{split} \int_{V^{n,i} \cap T_{j}^{n,i}} \operatorname{div}_{\Gamma_{h}} p \, d\mathcal{H}^{2} &= \int_{e_{j,R}^{n,i}} p \cdot \mu_{j,E}^{n,i} \, d\mathcal{H}^{1} + \int_{e_{j+1,L}^{n,i}} p \cdot \mu_{j+1,E}^{n,i} \, d\mathcal{H}^{1} \\ &+ \int_{S_{j,L}^{n,i}} p \cdot \mu_{j,L}^{n,i} \, d\mathcal{H}^{1} + \int_{S_{j,R}^{n,i}} p \cdot \mu_{j+1,R}^{n,i} \, d\mathcal{H}^{1} \\ &= p \cdot \{h_{j,R}^{n,i} \, \mu_{j,E}^{n,i} + h_{j+1,L}^{n,i} \, \mu_{j+1,E}^{n,i} + d^{n,i} (\mu_{j,L}^{n,i} + \mu_{j+1,R}^{n,i})\}, \end{split}$$

since $|e_{j,R}^{n,i}| = h_{j,R}^{n,i}$, $|e_{j+1,L}^{n,i}| = h_{j+1,L}^{n,i}$ and $|S_{j,L}^{n,i}| = |S_{j,R}^{n,i}| = d^{n,i}$ by the definition of the volume $V^{n,i}$. Summing up both sides of the above equality over $j = 1, \ldots, \mu_i$ we obtain from (6.4.29)

$$0 = \sum_{j=1}^{\mu_i} (p \cdot \mu_{j,E}^{n,i}) (h_{j,L}^{n,i} + h_{j,R}^{n,i}) + d^{n,i} \sum_{j=1}^{\mu_i} p \cdot (\mu_{j,L}^{n,i} + \mu_{j,R}^{n,i})$$
$$= -J_2 + d^{n,i} \sum_{j=1}^{\mu_i} \nabla_{\Gamma} \varphi(x_i^n) \cdot (\mu_{j,L}^{n,i} + \mu_{j,R}^{n,i}).$$

Here the last line follows from $p = \nabla_{\Gamma} \varphi(x_i^n)$, (6.4.30), and (6.4.27). Hence

$$|J_2| = |d^{n,i} \sum_{j=1}^{\mu_i} \nabla_{\Gamma} \varphi(x_i^n) \cdot (\mu_{j,L}^{n,i} + \mu_{j,R}^{n,i})| \le C d^{n,i} \|\nabla_{\Gamma} \varphi\|_{B(\overline{S_T})} \max_j |\mu_{j,L}^{n,i} + \mu_{j,R}^{n,i}|.$$
(6.4.31)

Note that, contrary to the case of a flat stationary domain considered in [17], the equality $\mu_{j,L}^{n,i} = -\mu_{j,R}^{n,i}$ does not hold in general because the triangles $T_{j-1}^{n,i}$ and $T_j^{n,i}$ do not lie in the same plane. Instead we deduce from (6.4.30) and $|\mu_{j,E}^{n,i}| = 1$

$$\begin{aligned} |\mu_{j,L}^{n,i} + \mu_{j,R}^{n,i}| &= |(\nu_j^{n,i} - \nu_{j-1}^{n,i}) \times \mu_{j,E}^{n,i}| \le |\nu_j^{n,i} - \nu_{j-1}^{n,i}| \\ &\le |\nu_j^{n,i} - \nu(x_i^n, t^n)| + |\nu(x_i^n, t^n) - \nu_{j-1}^{n,i}| \le Ch \end{aligned}$$

$$(6.4.32)$$

by (6.2.10). Inserting (6.4.28), (6.4.31) with (6.4.32) into (6.4.26) and taking into account (6.4.4) as well as (6.4.10) we derive

$$|I_{3}| \leq C \frac{\varepsilon_{i}^{n}}{|V^{n,i}|} \left(\sum_{j=1}^{\mu_{i}} |E_{j}^{n,i}|^{2} + d^{n,i}h \right) \left(\|\nabla_{\Gamma}\varphi\|_{B(\overline{S_{T}})} + \|\nabla_{\Gamma}^{2}\varphi\|_{B(\overline{S_{T}})} \right)$$

$$\leq Ch \left(\|\nabla_{\Gamma}\varphi\|_{B(\overline{S_{T}})} + \|\nabla_{\Gamma}^{2}\varphi\|_{B(\overline{S_{T}})} \right).$$
(6.4.33)

The result now follows from (6.4.18) together with (6.4.19), (6.4.25) and (6.4.33).

6.5 Convergence to viscosity solutions

The purpose of this section is to prove that the approximate solution generated by the scheme (6.4.5)–(6.4.8) converges to a viscosity solution of the Hamilton–Jacobi equation (6.3.1) providing at the same time an existence result for this problem. We start with a technical result that compares the nodal values of a solution of the scheme with those at the initial time, see Lemma 2.3 in [17] for a similar result in the flat case.

Lemma 6.5.1. Suppose that $v_h^n = \sum_{i=1}^M v_i^n \chi(\cdot, t^n) \in V_h^n$ is a solution of $v_h^{n+1} = S_h^n(v_h^n), n = 0, \ldots, N-1$ with initial data $v_h^0(x_i^0) = v_0(x_i^0), i = 1, \ldots, M$, where $v_0 : \Gamma(0) \to \mathbb{R}$ is Lipschitz continuous with constant L_0 . If (6.4.10) holds, then there exists a constant $C_4 > 0$ depending on γ , H and L_0 such that

$$\max_{i=1,\dots,M} |v_i^n - v_i^0| \le C_4 t^n, \quad n = 0, 1, \dots, N.$$
(6.5.1)

Proof. Let us denote by v_0^{\sharp} the push-forward of v_0 i.e. $v_0^{\sharp}(x,t) := v_0(\Phi^{-1}(x,t)), (x,t) \in \overline{S_T}$ and by $I_h^n v_0^{\sharp} \in V_h^n$ its interpolant. Since $x_i^n = \Phi(x_i^0, t^n)$ we have

$$[I_h^n v_0^{\sharp}]_i = I_h^n v_0^{\sharp}(x_i^n) = v_0^{\sharp}(x_i^n, t^n) = v_0(\Phi^{-1}(x_i^n, t^n)) = v_0(x_i^0), \ i = 1, \dots, M.$$
(6.5.2)

Note that the right-hand side is independent of n. We claim that there exists a constant $R \ge 0$ such that

$$|\nabla I_h^n v_0^{\sharp}| \le R \quad \text{on } \Gamma_h(t^n). \tag{6.5.3}$$

To see this, let us fix a triangle $K(t^n) \subset \Gamma_h(t^n)$ whose vertices are denoted for simplicity by x_1^n, x_2^n and x_3^n . By transforming onto the unit triangle, using (6.5.2), the Lipschitz continuity of v_0 and Φ^{-1} as well as (6.2.8) we obtain

$$\begin{aligned} |\nabla I_h^n v_{0|K(t^n)}^{\sharp}| &\leq \frac{C}{\rho_{K(t^n)}} \max_{i=2,3} |I_h^n v_0^{\sharp}(x_i^n) - I_h^n v_0^{\sharp}(x_1^n)| = \frac{C}{\rho_{K(t^n)}} \max_{i=2,3} |v_0(x_i^0) - v_0(x_1^0)| \\ &\leq \frac{CL_0}{\rho_{K(t^n)}} \max_{i=2,3} |x_i^0 - x_1^0| = \frac{CL_0}{\rho_{K(t^n)}} \max_{i=2,3} |\Phi^{-1}(x_i^n, t^n) - \Phi^{-1}(x_1^n, t^n)| \\ &\leq \frac{CL_0}{\rho_{K(t^n)}} \max_{i=2,3} |x_i^n - x_1^n| \leq CL_0 \gamma =: R \end{aligned}$$

proving (6.5.3). Recalling the definition (6.4.8) of the numerical Hamiltonian we deduce with

the help of (6.5.3) and (6.4.10) that

$$\begin{aligned} |H_{i}^{n}([I_{h}^{n}v_{0}^{\sharp}]_{i},[I_{h}^{n}v_{0}^{\sharp}]_{i_{1}},\ldots,[I_{h}^{n}v_{0}^{\sharp}]_{i_{\mu_{i}}})| \\ &\leq \sum_{j=1}^{\mu_{i}} \frac{|V^{n,i} \cap T_{j}^{n,i}|}{|V^{n,i}|} \Big| H\Big(x_{i}^{n},t^{n},\nabla_{\Gamma_{h}}I_{h}^{n}v_{0}^{\sharp}|_{T_{j}^{n,i}}\Big)\Big| + \frac{\varepsilon_{i}^{n}}{|V^{n,i}|} \sum_{j=1}^{\mu_{i}} \frac{|[I_{h}^{n}v_{0}^{\sharp}]_{i_{j}} - [I_{h}^{n}v_{0}^{\sharp}]_{i|}}{|E_{j}^{n,i}|} (h_{j,L}^{n,i} + h_{j,R}^{n,i}) \\ &\leq \max_{(x,t)\in\overline{S_{T}},|p|\leq R} |H(x,t,p)| + C \frac{\max_{j}(h_{j}^{n,i})^{2}}{|V^{n,i}|} \leq :C_{4} \quad (6.5.4) \end{aligned}$$

where C_4 can be chosen independently of *i* and *n*. Now let us show by induction with respect to n = 0, 1, ..., N that

$$v_i^n \le [I_h^n v_0^{\sharp}]_i + C_4 t^n \quad \text{for all} \quad i = 1, \dots, M.$$
 (6.5.5)

Since $v_i^0 = v_0(x_i^0) = [I_h^0 v_0^{\sharp}]_i$ the inequality (6.5.5) holds for n = 0. Let us assume that (6.5.5) is true for some $n \in \{0, 1, \ldots, N-1\}$ so that $v_h^n \leq I_h^n v_0^{\sharp} + C_4 t^n$ on $\Gamma_h(t^n)$. Applying Lemma 6.4.1 together with (6.4.9) we infer that

$$v_h^{n+1} = S_h^n(v_h^n) \le S_h^n(I_h^n v_0^{\sharp} + C_4 t^n) = S_h^n(I_h^n v_0^{\sharp}) + C_4 t^n$$

on $\Gamma_h(t^{n+1})$, and hence by (6.4.7), (6.4.8), and (6.5.4)

$$v_i^{n+1} \leq [S_h^n(I_h^n v_0^{\sharp})]_i + C_4 t^n = [I_h^n v_0^{\sharp}]_i - \tau H_i^n([I_h^n v_0^{\sharp}]_i, [I_h^n v_0^{\sharp}]_{i_1}, \dots, [I_h^n v_0^{\sharp}]_{i_{\mu_i}}) + C_4 t^n \\ \leq [I_h^n v_0^{\sharp}]_i + C_4 \tau + C_4 t^n = [I_h^{n+1} v_0^{\sharp}]_i + C_4 t^{n+1}$$

for all i = 1, ..., M, where we used (6.5.2) in the last step. Hence we see by induction that (6.5.5) holds for all n = 0, 1, ..., N. By the same argument we can show that $[I_h^n v_0^{\sharp}]_i - C_4 t^n \leq v_i^n$ for all n = 0, 1, ..., N and i = 1, ..., M. Finally, (6.5.2), (6.5.5), and the above inequality yield (6.5.1).

Let us denote by $u_h^n = \sum_{i=1}^M u_i^n \chi_i(\cdot, t_n) \in V_h^n$, $n = 0, 1, \ldots, N$ the finite element function on $\Gamma_h(t_n)$ given by the numerical scheme (6.4.5)–(6.4.8). Now we define an approximate solution $u_h^l : \overline{S_T} \to \mathbb{R}$ by

$$u_{h}^{l}(x,t) = \sum_{i=1}^{M} u_{i}^{n} \chi_{i}^{l}(x,t), \quad t \in [t^{n}, t^{n+1}), \ x \in \Gamma(t)$$
(6.5.6)

for n = 0, 1, ..., N-1 (we include $t = t^N = T$ when n = N-1), where u_0 is a given function on $\Gamma(0)$. For $(x, t) \in \overline{S_T}$ set

$$\bar{u}(x,t) := \limsup_{\substack{h \to 0\\\overline{S_T} \ni (y,s) \to (x,t)}} u_h^l(y,s), \quad \underline{u}(x,t) := \liminf_{\substack{h \to 0\\\overline{S_T} \ni (y,s) \to (x,t)}} u_h^l(y,s).$$
(6.5.7)

It follows from [4, Section V.2.1, Proposition 2.1] that $\bar{u} \in USC(\overline{S_T})$ and $\underline{u} \in LSC(\overline{S_T})$. Our aim is to show that \bar{u} (resp. \underline{u}) is a subsolution (resp. supersolution) to (6.3.1). As a first step we prove

Lemma 6.5.2. Let \bar{u} and \underline{u} be given by (6.5.6)–(6.5.7). Assume that (6.4.10) is satisfied and that $u_0 \in C(\Gamma(0))$. Then $\bar{u}(\cdot, 0) = \underline{u}(\cdot, 0) = u_0$ on $\Gamma(0)$. *Proof.* Fix $x_0 \in \Gamma(0)$. By (6.5.7) it immediately follows that $\underline{u}(x_0, 0) \leq \overline{u}(x_0, 0)$. Therefore, if the inequality

$$\bar{u}(x_0,0) \le u_0(x_0) \le \underline{u}(x_0,0)$$
(6.5.8)

holds, then we get $\bar{u}(x_0, 0) = \underline{u}(x_0, 0) = u_0(x_0)$. Let us prove (6.5.8). Since $\Gamma(0)$ is compact in \mathbb{R}^3 , the function $u_0 \in C(\Gamma(0))$ is bounded and uniformly continuous on $\Gamma(0)$. Hence setting

$$\omega_0(r) := \sup\{|u_0(x) - u_0(x_0)| \mid x \in \Gamma(0), |x - x_0| \le r\}, \quad r \in [0, \infty),$$

we see that $\omega_0(0) = 0$ and ω_0 is bounded, nondecreasing, continuous at r = 0. From this fact and the proof of [15, Lemma 2.1.9 (i)] there exists a bounded, nondecreasing, and continuous function ω on $[0, \infty)$ satisfying $\omega(0) = 0$ and $\omega_0 \leq \omega$ on $[0, \infty)$. Fix an arbitrary $\delta > 0$. By the above properties of ω we may take a constant $A_{\delta} > 0$ such that $\omega(r) \leq \delta + A_{\delta}r^2$ for all $r \in [0, \infty)$. From this inequality and $|u_0(x) - u_0(x_0)| \leq \omega_0(|x - x_0|) \leq \omega(|x - x_0|)$ it follows that

$$u_0(x) \le u_0(x_0) + \delta + A_\delta |x - x_0|^2 \quad \text{for all } x \in \Gamma(0).$$
(6.5.9)

Now we construct $v_h^n = \sum_{i=1}^M v_i^n \chi_i(\cdot, t^n) \in V_h^n$, $n = 0, 1, \ldots, N$ by (6.4.5)–(6.4.8) from the initial value $v_0(x) := A_{\delta}|x - x_0|^2$, $x \in \Gamma(0)$. Then by interpolating both sides of (6.5.9) on $\Gamma_h(0)$ and observing that $u_0(x_0) + \delta$ is constant we have

$$u_h^0 \le u_0(x_0) + \delta + v_h^0$$
 on $\Gamma_h(0)$.

Combining this inequality with Lemma 6.4.1 and (6.4.9) we obtain

$$u_h^1 = S_h^0(u_h^0) \le S_h^0(u_0(x_0) + \delta + v_h^0) = u_0(x_0) + \delta + S_h^0(v_h^0) = u_0(x_0) + \delta + v_h^1 \quad \text{on} \quad \Gamma_h(t^1)$$

and then inductively $u_h^n \leq u_0(x_0) + \delta + v_h^n$ on $\Gamma_h(t^n)$ for $n = 0, 1, \dots, N$, or

$$u_i^n \le u_0(x_0) + \delta + v_i^n \le u_0(x_0) + \delta + v_i^0 + C_4 t^n$$
(6.5.10)

for n = 0, 1, ..., N, i = 1, ..., M, where we applied Lemma 6.5.1 to v_h^n . Multiplying both sides by $\chi_i^l(\cdot, t), t \in [t^n, t^{n+1})$ and summing them over i = 1, ..., M we infer with the help of (6.5.2) (with t instead of t^n)

$$u_h^l(x,t) \le u_0(x_0) + \delta + [I_h^t v_0^{\sharp}]^l(x) + C_4 t \quad \text{for all} \quad (x,t) \in \overline{S_T}.$$
 (6.5.11)

Since $v_0^{\sharp}(x_0, 0) = v_0(x_0) = 0$ and v_0^{\sharp} is Lipschitz continuous on $\overline{S_T}$ we may estimate

$$\begin{split} |[I_h^t v_0^{\sharp}]^l(x)| &\leq |[I_h^t v_0^{\sharp}]^l(x) - v_0^{\sharp}(x,t)| + |v_0^{\sharp}(x,t) - v_0^{\sharp}(x_0,0)| \\ &\leq \|v_0^{\sharp}(\cdot,t) - [I_h^t v_0^{\sharp}]^l\|_{B(\Gamma(t))} + C(|x-x_0|+t) \leq C(h+|x-x_0|+t), \end{split}$$

where we also used Lemma 6.2.1. Combining this estimate with (6.5.11) we infer

$$\bar{u}(x_0,0) = \limsup_{\substack{h \to 0\\\overline{S_T} \ni (x,t) \to (x_0,0)}} u_h^l(x,t) \le u_0(x_0) + \delta.$$

Since $\delta > 0$ is arbitrary, it follows that $\bar{u}(x_0, 0) \leq u_0(x_0)$. By the same argument we can show $u_0(x_0) \leq \underline{u}(x_0, 0)$. Hence (6.5.8) is valid and the lemma follows.

Lemma 6.5.3. Under the same assumptions as in Lemma 6.5.2, \bar{u} (resp. \underline{u}) is a subsolution (resp. supersolution) to (6.3.1).

Proof. We know from Lemma 6.5.2 that $\bar{u}(x,0) = \underline{u}(x,0) = u_0(x), x \in \Gamma(0)$ so that it remains to verify (6.3.4).

Let us suppose first that $\varphi \in C^2(\overline{S_T})$ and that $\overline{u} - \varphi$ takes a local maximum at $(x_0, t_0) \in \overline{S_T}$ with $t_0 > 0$. Since \overline{u} is bounded on $\overline{S_T}$ we may assume by a standard argument that $\overline{u} - \varphi$ has a strict global maximum at (x_0, t_0) . Let φ_h^l be given by

$$\varphi_h^l(x,t) := \sum_{i=1}^M \varphi_i^n \chi_i^l(x,t), \quad t \in [t^n, t^{n+1}), \ x \in \Gamma(t), \tag{6.5.12}$$

where $\varphi_i^n := \varphi(x_i^n, t^n), i = 1, \dots, M$ and we include $t = t^N = T$ if n = N - 1. We claim that

$$(\bar{u} - \varphi)(x, t) = \limsup_{\substack{h \to 0\\ \overline{S_T} \ni (y, s) \to (x_0, t_0)}} (u_h^l - \varphi_h^l)(y, s).$$
(6.5.13)

In order to see this, we note that in view of the Lipschitz continuity of φ on $\overline{S_T}$ it is sufficient to show that $\varphi_h^l \to \varphi$ uniformly on $\overline{S_T}$. But,

$$\begin{split} \|\varphi_{h}^{l} - \varphi\|_{B(\overline{S_{T}})} &\leq \sup_{t \in [0,T]} \|\varphi(\cdot,t) - [I_{h}^{t}\varphi]^{l}\|_{B(\Gamma(t))} \\ &+ \max_{n=0,\dots,N-1} \sup_{x \in \Gamma(t), t^{n} \leq t \leq t^{n+1}} \Big| \sum_{i=1}^{M} (\varphi(x_{i}(t),t) - \varphi(x_{i}^{n},t^{n}))\chi_{i}^{l}(x,t) \Big| \\ &\leq Ch + \max_{n=0,\dots,N-1} \sup_{i=1,\dots,M, t^{n} \leq t \leq t^{n+1}} |\varphi(x_{i}(t),t) - \varphi(x_{i}^{n},t^{n})| \leq C(h+\tau) \end{split}$$

by Lemma 6.2.1, the fact that $x_i(t) = \Phi(x_i^0, t)$ and the Lipschitz continuity of φ and Φ . Thus, (6.5.13) holds so that there exist $h_j > 0$ and $(y_j, s_j) \in \overline{S_T}$, $j \in \mathbb{N}$ with $h_j \to 0$, $(y_j, s_j) \to (x_0, t_0)$, and $(u_{h_j}^l - \varphi_{h_j}^l)(y_j, s_j) \to (\bar{u} - \varphi)(x_0, t_0)$ as $j \to \infty$. For each $j \in \mathbb{N}$, the function $u_{h_j}^l - \varphi_{h_j}^l$ is of the form

$$(u_{h_j}^l - \varphi_{h_j}^l)(x, t) = \sum_{i=1}^M (u_i^n - \varphi_i^n) \chi_i^l(x, t), \ x \in \Gamma(t), t \in [t^n, t^{n+1}), n = 0, \dots, N-1.$$

Let us choose $n_j \in \{0, 1, \dots, N\}$ and $i_j \in \{1, \dots, M\}$ such that

$$u_{i_j}^{n_j} - \varphi_{i_j}^{n_j} = \max\{u_i^n - \varphi_i^n \mid n = 0, \dots, N, i = 1, \dots, M\}$$

and use $\chi_i(x,t) \ge 0$, i = 1, ..., M and $\sum_{i=1}^M \chi_i^l(x,t) = 1$ to get

$$(u_{h_j}^l - \varphi_{h_j}^l)(x, t) \le (u_{i_j}^{n_j} - \varphi_{i_j}^{n_j}) \sum_{i=1}^M \chi_i^l(x, t) = (u_{h_j}^l - \varphi_{h_j}^l)(x_{i_j}^{n_j}, t^{n_j})$$
(6.5.14)

for all $(x,t) \in \overline{S_T}$. In particular, for all $j \in \mathbb{N}$,

$$(u_{h_j}^l - \varphi_{h_j}^l)(y_j, s_j) \le (u_{h_j}^l - \varphi_{h_j}^l)(x_{i_j}^{n_j}, t^{n_j}).$$

Since $(x_{i_j}^{n_j}, t^{n_j})$ belongs to the compact set $\overline{S_T}$, we may assume (up to a subsequence) that there exists $(\bar{x}, \bar{t}) \in \overline{S_T}$ such that $(x_{i_j}^{n_j}, t^{n_j}) \to (\bar{x}, \bar{t})$ as $j \to \infty$. Then by the above inequality and (6.5.13) we have

$$(\bar{u} - \varphi)(x_0, t_0) = \lim_{j \to \infty} (u_{h_j}^l - \varphi_{h_j}^l)(y_j, s_j) \le \limsup_{j \to \infty} (u_{h_j}^l - \varphi_{h_j}^l)(x_{i_j}^{n_j}, t^{n_j}) \le (\bar{u} - \varphi)(\bar{x}, \bar{t})$$

where the last inequality follows from the fact that $\bar{u} - \varphi \in USC(\overline{S_T})$. Recalling that $\bar{u} - \varphi$ takes a strict global maximum at (x_0, t_0) we infer that $(\bar{x}, \bar{t}) = (x_0, t_0)$. In particular, since $\lim_{j\to\infty} t^{n_j} = \bar{t} = t_0 > 0$ we have for sufficiently large j that $t^{n_j} > 0$ i.e. $n_j \ge 1$. Thus we can set $(x, t) = (x_i^{n_j-1}, t^{n_j-1})$ in (6.5.14) to obtain

$$(u_{h_j}^l - \varphi_{h_j}^l)(x_i^{n_j-1}, t^{n_j-1}) \le \delta_j := u_{i_j}^{n_j} - \varphi_{i_j}^{n_j},$$

or equivalently, $u_i^{n_j-1} \leq \varphi_i^{n_j-1} + \delta_j$ for $i = 1, \ldots, M$. From this we see that

$$u_{h_j}^{n_j-1} \leq I_{h_j}^{n_j-1}\varphi + \delta_j \quad \text{on} \quad \Gamma_{h_j}(t^{n_j-1}),$$

and then by Lemma 6.4.1 and (6.4.9)

$$u_{h_j}^{n_j} = S_{h_j}^{n_j - 1}(u_{h_j}^{n_j - 1}) \le S_{h_j}^{n_j - 1}(I_{h_j}^{n_j - 1}\varphi + \delta_j) = S_{h_j}^{n_j - 1}(I_{h_j}^{n_j - 1}\varphi) + \delta_j \text{ on } \Gamma_{h_j}(t^{n_j}).$$

Inserting $x = x_{i_j}^{n_j} \in \Gamma_{h_j}(t^{n_j})$ into this inequality we get

$$u_{i_j}^{n_j} \le [S_{h_j}^{n_j-1}(I_{h_j}^{n_j-1}\varphi)]_{i_j} + \delta_j = [S_{h_j}^{n_j-1}(I_{h_j}^{n_j-1}\varphi)]_{i_j} + u_{i_j}^{n_j} - \varphi_{i_j}^{n_j}$$

by the definition of δ_i and hence,

$$\varphi_{i_j}^{n_j} - [S_{h_j}^{n_j-1}(I_{h_j}^{n_j-1}\varphi)]_{i_j} \le 0.$$
(6.5.15)

Since $\varphi \in C^2(\overline{S_T})$, we can combine (6.5.15) with Lemma 6.4.2 to derive

$$\partial^{\bullet}\varphi(x_{i_j}^{n_j-1}, t^{n_j-1}) + H(x_{i_j}^{n_j-1}, t^{n_j-1}, \nabla_{\Gamma}\varphi(x_{i_j}^{n_j-1}, t^{n_j-1})) \le C_{\varphi}h_j$$
(6.5.16)

and observing that

$$|(x_{i_j}^{n_j}, t^{n_j}) - (x_{i_j}^{n_j - 1}, t^{n_j - 1})| \le C\tau \le Ch_j \to 0, \ j \to \infty$$

we obtain (6.3.4) by sending $j \to \infty$ in (6.5.16).

Finally, let $\varphi \in C^1(\overline{S_T})$ and suppose that $\overline{u} - \varphi$ takes a local maximum at $(x_0, t_0) \in \overline{S_T}, t_0 > 0$. As in the first part of the proof, we may assume that $\overline{u} - \varphi$ takes a strict global maximum at (x_0, t_0) . Let us approximate φ by a sequence $(\varphi_{\delta}) \subset C^2(\overline{S_T})$ such that $\varphi_{\delta} \to \varphi$ in $C^1(\overline{S_T})$ as $\delta \to 0$. For a suitable subsequence there exist $(x_{\delta}, t_{\delta}) \in \overline{S_T}$ such that $(x_{\delta}, t_{\delta}) \to (x_0, t_0)$ and $\overline{u} - \varphi_{\delta}$ takes a global maximum at (x_{δ}, t_{δ}) . In particular, $t_{\delta} > 0$ for sufficiently small $\delta > 0$. It follows from the first part of the proof that

$$\partial^{\bullet}\varphi_{\delta}(x_{\delta}, t_{\delta}) + H(x_{\delta}, t_{\delta}, \nabla_{\Gamma}\varphi_{\delta}(x_{\delta}, t_{\delta})) \leq 0.$$

Letting $\delta \to 0$ in the above inequality we see that φ satisfies (6.3.4) at (x_0, t_0) , so that \bar{u} is a subsolution to (6.3.1). In the same way one shows that \underline{u} is a supersolution.

Finally, let us prove the existence of a viscosity solution to (6.3.1).

Theorem 6.5.4. Suppose that $u_0 \in C(\Gamma(0))$. Then there exists a unique viscosity solution to (6.3.1).

Proof. The uniqueness of a viscosity solution was already shown in Corollary 6.3.3. Let us prove the existence. Since $u_0 \in C(\Gamma(0))$, Lemmas 6.5.2 and 6.5.3 imply that \bar{u} and \underline{u} constructed by (6.5.6)–(6.5.7) are a subsolution and supersolution to (6.3.1), respectively, and satisfy $\bar{u}(\cdot, 0) = \underline{u}(\cdot, 0) = u_0$ on $\Gamma(0)$. Hence we can apply the comparison principle (see Theorem 6.3.2) to the subsolution \bar{u} and the supersolution \underline{u} to get $\bar{u} \leq \underline{u}$ on $\overline{S_T}$. Moreover, by (6.5.7) we easily see that $\underline{u} \leq \bar{u}$ on $\overline{S_T}$. Therefore, $u := \bar{u} = \underline{u}$ is a viscosity solution to (6.3.1).

6.6 Error bound

In this section we prove an error estimate between the viscosity solution to (6.3.1) and the numerical solution given by the scheme (6.4.5)-(6.4.8).

Theorem 6.6.1. Let u be the viscosity solution to (6.3.1) with initial value u_0 . For h > 0and n = 0, 1, ..., N, let $u_h^n = \sum_{i=1}^M u_i^n \chi_i(\cdot, t^n) \in V_h^n$ be the finite element function constructed from u_0 by (6.4.5)–(6.4.8). Assume that (6.4.10) is satisfied and that u is Lipschitz continuous on $\overline{S_T}$ in the sense that

$$|u(x,t) - u(y,s)| \le L_U(|x-y| + |t-s|)$$
(6.6.1)

for all $(x,t), (y,s) \in \overline{S_T}$, where $L_U > 0$ is a constant independent of (x,t) and (y,s). Then there exist $h_0 > 0$ and a constant C > 0 independent of h such that

$$\max_{1 \le i \le M, \, 0 \le n \le N} |u(x_i^n, t^n) - u_i^n| \le Ch^{1/2} \quad for \ all \quad h \in (0, h_0).$$
(6.6.2)

Proof. The argument is similar to that in the proof of the comparison principle (see Theorem 6.3.2). Let us define

$$\Psi(x,t,i,n) := u(x,t) - \rho\sqrt{h}t - u_i^n - \frac{|x - x_i^n|^2 + |t - t^n|^2}{\sqrt{h}}$$
(6.6.3)

for $(x,t) \in \overline{S_T}$, $i \in \{1, \ldots, M\}$ and $n \in \{0, 1, \ldots, N\}$. Here, the constant $\rho > 0$ is subject to $\rho\sqrt{h} \leq 1$ and will be chosen later. Clearly,

$$\max_{1 \le i \le M, 0 \le n \le N} (u(x_i^n, t^n) - u_i^n) = \max_{1 \le i \le M, 0 \le n \le N} (\Psi(x_i^n, t^n, i, n) + \rho \sqrt{h} t^n)$$
$$\leq \max_{(x,t) \in \overline{S_T}, i=1,\dots,M, n=0,\dots,N} \Psi(x, t, i, n) + \rho \sqrt{h} T \qquad (6.6.4)$$
$$= \Psi(x_0, t_0, i_0, n_0) + \rho \sqrt{h} T$$

for some $(x_0, t_0) \in \overline{S_T}$, $i_0 \in \{1, \dots, M\}$ and $n_0 \in \{0, 1, \dots, N\}$. In particular, we have $\Psi(x_{i_0}^{n_0}, t^{n_0}, i_0, n_0) \leq \Psi(x_0, t_0, i_0, n_0)$, i.e.

$$u(x_{i_0}^{n_0}, t^{n_0}) - \rho\sqrt{h} t^{n_0} - u_{i_0}^{n_0} \le u(x_0, t_0) - \rho\sqrt{h} t_0 - u_{i_0}^{n_0} - \frac{|x_0 - x_{i_0}^{n_0}|^2 + |t_0 - t^{n_0}|^2}{\sqrt{h}}.$$

From this, (6.6.1), and the fact that $\rho\sqrt{h} \leq 1$ it follows that

$$\frac{|x_0 - x_{i_0}^{n_0}|^2 + |t_0 - t^{n_0}|^2}{\sqrt{h}} \le u(x_0, t_0) - u(x_{i_0}^{n_0}, t^{n_0}) + \rho\sqrt{h}(t^{n_0} - t_0)$$
$$\le L_U(|x_0 - x_{i_0}^{n_0}| + |t_0 - t^{n_0}|) + |t_0 - t^{n_0}|$$
$$\le C(|x_0 - x_{i_0}^{n_0}|^2 + |t_0 - t^{n_0}|^2)^{1/2}$$

and hence

$$\frac{(|x_0 - x_{i_0}^{n_0}|^2 + |t_0 - t^{n_0}|^2)^{1/2}}{\sqrt{h}} \le C.$$
(6.6.5)

Now let us consider several possible cases.

Case 1: $t_0 > 0$ and $n_0 \ge 1$. By exploiting the fact that u is a subsolution we obtain as in (6.3.7)

$$\frac{2}{\sqrt{h}}(t_0 - t^{n_0}) + \frac{2}{\sqrt{h}}v_{\Gamma}(x_0, t_0) \cdot (x_0 - x_{i_0}^{n_0}) + H\left(x_0, t_0, \frac{2}{\sqrt{h}}P_{\Gamma}(x_0, t_0)(x_0 - x_{i_0}^{n_0})\right) \le -\rho\sqrt{h}.$$
 (6.6.6)

On the other hand, since $\Psi(x_0, t_0, i, n_0 - 1) \le \Psi(x_0, t_0, i_0, n_0), i = 1, ..., M$ we infer

$$\varphi_i^{n_0-1} - u_i^{n_0-1} \le \varphi_{i_0}^{n_0} - u_{i_0}^{n_0}, \quad i = 1, \dots, M,$$

where

$$\varphi_i^n = \varphi(x_i^n, t^n)$$
 and $\varphi(x, t) = -\frac{|x_0 - x|^2 + (t_0 - t)^2}{\sqrt{h}}.$

Hence, $I_h^{n_0-1}\varphi \leq u_h^{n_0-1} + \varphi_{i_0}^{n_0} - u_{i_0}^{n_0}$ on $\Gamma_h(t^{n_0-1})$ so that we deduce with the help of Lemma 6.4.1, (6.4.9) and the definition of the scheme

$$S_h^{n_0-1}(I_h^{n_0-1}\varphi) \le S_h^{n_0-1}(u_h^{n_0-1}) + \varphi_{i_0}^{n_0} - u_{i_0}^{n_0} = u_h^{n_0} + \varphi_{i_0}^{n_0} - u_{i_0}^{n_0}$$

Evaluating the above inequality for $x = x_{i_0}^{n_0}$ we find that

$$[S_h^{n_0-1}(I_h^{n_0-1}\varphi)]_{i_0} \le \varphi_{i_0}^{n_0},$$

from which we infer that

$$-\partial^{\bullet}\varphi(x_{i_0}^{n_0}, t^{n_0}) - H(x_{i_0}^{n_0}, t^{n_0}, \nabla_{\Gamma}\varphi(x_{i_0}^{n_0}, t^{n_0})) \le A + B,$$
(6.6.7)

where

$$\begin{split} A &= -\partial^{\bullet}\varphi(x_{i_{0}}^{n_{0}-1}, t^{n_{0}-1}) - H\left(x_{i_{0}}^{n_{0}-1}, t^{n_{0}-1}, \nabla_{\Gamma}\varphi(x_{i_{0}}^{n_{0}-1}, t^{n_{0}-1})\right) + \frac{\varphi_{i_{0}}^{n_{0}} - [S_{h}^{n_{0}-1}(I_{h}^{n_{0}-1}\varphi)]_{i_{0}}}{\tau}, \\ B &= [\partial^{\bullet}\varphi(x_{i_{0}}^{n_{0}-1}, t^{n_{0}-1}) - \partial^{\bullet}\varphi(x_{i_{0}}^{n_{0}}, t^{n_{0}})] \\ &+ [H\left(x_{i_{0}}^{n_{0}-1}, t^{n_{0}-1}, \nabla_{\Gamma}\varphi(x_{i_{0}}^{n_{0}-1}, t^{n_{0}-1})\right) - H\left(x_{i_{0}}^{n_{0}}, t^{n_{0}}, \nabla_{\Gamma}\varphi(x_{i_{0}}^{n_{0}}, t^{n_{0}})\right)]. \end{split}$$

We deduce from Lemma 6.4.2 that

$$|A| \le C_3 h \left(\|\nabla_{\Gamma}\varphi\|_{B(\overline{S_T})} + \|\nabla_{\Gamma}^2\varphi\|_{B(\overline{S_T})} + \|(\partial^{\bullet})^2\varphi\|_{B(\overline{S_T})} \right) \le C\sqrt{h}$$
(6.6.8)

since

$$\partial^{\bullet}\varphi(x,t) = -\frac{2}{\sqrt{h}}(t-t_0) - \frac{2}{\sqrt{h}}v_{\Gamma}(x,t) \cdot (x-x_0), \qquad (6.6.9)$$

$$\nabla_{\Gamma}\varphi(x,t) = -\frac{2}{\sqrt{h}}P_{\Gamma}(x,t)(x-x_0).$$
(6.6.10)

Using (6.6.9), (6.6.10), (6.3.2), (6.3.3) and the Lipschitz continuity of v_{Γ} we further obtain

$$|B| \leq \left(\frac{C}{\sqrt{h}} + L_{H,1} \left(1 + \|\nabla_{\Gamma}\varphi\|_{B(\overline{S_T})}\right)\right) \left(|x_{i_0}^{n_0} - x_{i_0}^{n_0-1}| + |t^{n_0} - t^{n_0-1}|\right) + L_{H,2} |\nabla_{\Gamma}\varphi(x_{i_0}^{n_0}, t^{n_0}) - \nabla_{\Gamma}\varphi(x_{i_0}^{n_0-1}, t^{n_0-1})|$$

$$\leq \frac{C}{\sqrt{h}} \tau \leq C\sqrt{h},$$
(6.6.11)

where we used (6.4.10) for the last inequality. If we insert (6.6.8) and (6.6.11) into (6.6.7) and use again (6.6.9), (6.6.10) we obtain

$$-\frac{2}{\sqrt{h}}(t_0 - t^{n_0}) - \frac{2}{\sqrt{h}}v_{\Gamma}(x_{i_0}^{n_0}, t^{n_0}) \cdot (x_0 - x_{i_0}^{n_0}) - H(x_{i_0}^{n_0}, t^{n_0}, \frac{2}{\sqrt{h}}P_{\Gamma}(x_{i_0}^{n_0}, t^{n_0})(x_0 - x_{i_0}^{n_0})) \le C\sqrt{h}.$$
 (6.6.12)

We sum up both sides of (6.6.6) and (6.6.12) and employ the Lipschitz continuity of v_{Γ} as well as (6.3.2), (6.3.3) to get

$$\begin{split} \rho\sqrt{h} &\leq C\sqrt{h} + \frac{2}{\sqrt{h}} \{ v_{\Gamma}(x_{i_{0}}^{n_{0}}, t^{n_{0}}) - v_{\Gamma}(x_{0}, t_{0}) \} \cdot (x_{0} - x_{i_{0}}^{n_{0}}) \\ &+ H\left(x_{i_{0}}^{n_{0}}, t^{n_{0}}, \frac{2}{\sqrt{h}} P_{\Gamma}(x_{i_{0}}^{n_{0}}, t^{n_{0}})(x_{0} - x_{i_{0}}^{n_{0}})\right) - H\left(x_{0}, t_{0}, \frac{2}{\sqrt{h}} P_{\Gamma}(x_{0}, t_{0})(x_{0} - x_{i_{0}}^{n_{0}})\right) \\ &\leq C\sqrt{h} + \frac{C(|x_{0} - x_{i_{0}}^{n_{0}}| + |t_{0} - t^{n_{0}}|)|x_{0} - x_{i_{0}}^{n_{0}}|}{\sqrt{h}} \\ &+ L_{H,1}(|x_{0} - x_{i_{0}}^{n_{0}}| + |t_{0} - t^{n_{0}}|)\left(1 + \frac{2}{\sqrt{h}}|P_{\Gamma}(x_{0}, t_{0})(x_{0} - x_{i_{0}}^{n_{0}})|\right) \\ &+ \frac{2L_{H,2}}{\sqrt{h}}|P_{\Gamma}(x_{0}, t_{0}) - P_{\Gamma}(x_{i_{0}}^{n_{0}}, t^{n_{0}})||x_{0} - x_{i_{0}}^{n_{0}}| \\ &\leq C\sqrt{h} + C\frac{|x_{0} - x_{i_{0}}^{n_{0}}|^{2} + |t_{0} - t^{n_{0}}|^{2}}{\sqrt{h}} + C\left(|x_{0} - x_{i_{0}}^{n_{0}}| + |t_{0} - t^{n_{0}}|\right) \\ &\leq C\sqrt{h} \end{split}$$

in view of (6.6.5). Choosing $\rho > C$ we obtain a contradiction so that this case cannot occur. **Case 2:** $t_0 = 0$ and $n_0 \ge 0$. Since $u(x_0, t_0) = u(x_0, 0) = u_0(x_0)$ we obtain with the help of (6.6.1), Lemma 6.5.1 and (6.6.5)

$$\Psi(x_{0}, t_{0}, i_{0}, n_{0}) = \Psi(x_{0}, 0, i_{0}, n_{0}) \leq u(x_{0}, 0) - u_{i_{0}}^{n_{0}} = u_{0}(x_{0}) - u_{0}(x_{i_{0}}^{0}) + u_{i_{0}}^{0} - u_{i_{0}}^{n_{0}}$$

$$\leq L_{U}|x_{0} - x_{i_{0}}^{0}| + C_{4}t^{n_{0}} \leq C(|x_{0} - x_{i_{0}}^{n_{0}}| + |x_{i_{0}}^{n_{0}} - x_{i_{0}}^{0}|) + C_{4}t^{n_{0}} \qquad (6.6.13)$$

$$\leq C(|x_{0} - x_{i_{0}}^{n_{0}}| + |t_{0} - t^{n_{0}}|) \leq C\sqrt{h}.$$

Case 3: $t_0 \ge 0$ and $n_0 = 0$. Using once more (6.6.1) and (6.6.5) we derive

$$\Psi(x_0, t_0, i_0, n_0) = \Psi(x_0, t_0, i_0, 0)
\leq u(x_0, t_0) - u_{i_0}^0 = u(x_0, t_0) - u(x_{i_0}^0, 0)
\leq L_U(|x_0 - x_{i_0}^0| + t_0) = L_U(|x_0 - x_{i_0}^{n_0}| + |t_0 - t^{n_0}|)
< C\sqrt{h}.$$
(6.6.14)

In conclusion we infer that from (6.6.4), (6.6.13), (6.6.14) and the fact that Case 1 cannot occur that

$$\max_{1 \le i \le M, \, 0 \le n \le N} (u(x_i^n, t^n) - u_i^n) \le C\sqrt{h}.$$

In an analogous way we bound $\max_{1 \le i \le M, 0 \le n \le N} (u_i^n - u(x_i^n, t^n))$ which completes the proof of the theorem.

References

- D. Adalsteinsson and J. A. Sethian, Transport and diffusion of material quantities on propagating interfaces via level set methods, J. Comput. Phys. 185 (2003), no. 1, 271–288.
- [2] A. Alphonse, C. M. Elliott, and B. Stinner, An abstract framework for parabolic PDEs on evolving spaces, Port. Math. 72 (2015), no. 1, 1–46.
- [3] A. Alphonse, C. M. Elliott, and B. Stinner, On some linear parabolic PDEs on moving hypersurfaces, Interfaces Free Bound. 17 (2015), no. 2, 157–187.
- [4] M. Bardi and I. Capuzzo-Dolcetta, Optimal control and viscosity solutions of Hamilton-Jacobi-Bellman equations, Systems & Control: Foundations & Applications, Birkhäuser Boston, Inc., Boston, MA, 1997. With appendices by Maurizio Falcone and Pierpaolo Soravia.
- [5] G. Barles, An introduction to the theory of viscosity solutions for first-order Hamilton-Jacobi equations and applications, Hamilton-Jacobi equations: approximations, numerical analysis and applications, Lecture Notes in Math., vol. 2074, Springer, Heidelberg, 2013, pp. 49–109.
- [6] J. W. Barrett, H. Garcke, and R. Nürnberg, On the stable numerical approximation of two-phase flow with insoluble surfactant, ESAIM Math. Model. Numer. Anal. 49 (2015), no. 2, 421–458.
- [7] L.-T. Cheng, P. Burchard, B. Merriman, and S. Osher, Motion of curves constrained on surfaces using a level-set approach, J. Comput. Phys. 175 (2002), no. 2, 604–644.
- [8] M. G. Crandall and P.-L. Lions, Two approximations of solutions of Hamilton-Jacobi equations, Math. Comp. 43 (1984), no. 167, 1–19.
- [9] G. Dziuk and C. M. Elliott, Finite elements on evolving surfaces, IMA J. Numer. Anal. 27 (2007), no. 2, 262–292.
- [10] G. Dziuk and C. M. Elliott, Finite element methods for surface PDEs, Acta Numer. 22 (2013), 289–396.
- [11] G. Dziuk, D. Kröner, and T. Müller, Scalar conservation laws on moving hypersurfaces, Interfaces Free Bound. 15 (2013), no. 2, 203–236.
- [12] C. Eilks and C. M. Elliott, Numerical simulation of dealloying by surface dissolution via the evolving surface finite element method, J. Comput. Phys. 227 (2008), no. 23, 9727–9741.
- [13] C. M. Elliott, B. Stinner, and C. Venkataraman, Modelling cell motility and chemotaxis with evolving surface finite elements, Journal of The Royal Society Interface 9 (2012), no. 76, 3027–3044.
- [14] J. Giesselmann and T. Müller, Geometric error of finite volume schemes for conservation laws on evolving surfaces, Numer. Math. 128 (2014), no. 3, 489–516.
- [15] Y. Giga, Surface evolution equations: A level set approach, Monographs in Mathematics, vol. 99, Birkhäuser Verlag, Basel, 2006.
- [16] T. Jankuhn, M. A. Olshanskii, and A. Reusken, *Incompressible fluid problems on embedded surfaces: Modeling and variational formulations*, Tech. Report 462, RWTH Aachen University (2017).

- [17] K. Kim and Y. Li, Convergence of finite volume schemes for Hamilton-Jacobi equations with Dirichlet boundary conditions, J. Comput. Math. 33 (2015), no. 3, 227–247.
- [18] G. Kossioris, Ch. Makridakis, and P. E. Souganidis, *Finite volume schemes for Hamilton-Jacobi equations*, Numer. Math. 83 (1999), no. 3, 427–442.
- [19] B. Kovács and C. A. P. Guerra, Error analysis for full discretizations of quasilinear parabolic problems on evolving surfaces, Numer. Methods Partial Differential Equations 32 (2016), no. 4, 1200–1231.
- [20] D. Lengeler and T. Müller, Scalar conservation laws on constant and time-dependent Riemannian manifolds, J. Differential Equations 254 (2013), no. 4, 1705–1727.
- [21] M. Lenz, S. F. Nemadjieu, and M. Rumpf, A convergent finite volume scheme for diffusion on evolving surfaces, SIAM J. Numer. Anal. 49 (2011), no. 1, 15–37.
- [22] X.-G. Li, W. Yan, and C. K. Chan, Numerical schemes for Hamilton-Jacobi equations on unstructured meshes, Numer. Math. 94 (2003), no. 2, 315–331.
- [23] C. B. Macdonald and S. J. Ruuth, Level set equations on surfaces via the closest point method, J. Sci. Comput. 35 (2008), no. 2-3, 219–240.
- [24] C. Mantegazza and A. C. Mennucci, Hamilton-Jacobi equations and distance functions on Riemannian manifolds, Appl. Math. Optim. 47 (2003), no. 1, 1–25.
- [25] K. E. Teigen, X. Li, J. Lowengrub, F. Wang, and A. Voigt, A diffuse-interface approach for modeling transport, diffusion and adsorption/desorption of material quantities on a deformable interface, Commun. Math. Sci. 7 (2009), no. 4, 1009–1037.
- [26] M. Vierling, Parabolic optimal control problems on evolving surfaces subject to point-wise box constraints on the control—theory and numerical realization, Interfaces Free Bound. 16 (2014), no. 2, 137–173.
- [27] J.-J. Xu and H.-K. Zhao, An Eulerian formulation for solving partial differential equations along a moving interface, J. Sci. Comput. 19 (2003), no. 1-3, 573–594.